NIVERSITY OF HAWAI'I WEST O'AHU

VEST · O'AH

F.HARA

IVERS



FINAL ENVIRONMENTAL IMPACT STATEMENT

Prepared for: University of Hawai'i

Prepared by:



December 2006



UNIVERSITY OF HAWAI'I WEST O'AHU FINAL ENVIRONMENAL IMPACT STATEMENT

Submitted Pursuant to Chapter 343, Hawai'i Revised Statutes and Title 11, Chapter 200, Hawai'i Administrative Rules

> Prepared for: State of Hawai'i University of Hawai'i West O'ahu

> > Prepared by:



This Final Environmental Impact Statement and all ancillary documents have been prepared under my direction or supervision and the information submitted, to the best of my knowledge, fully addresses document content requirements as set forth in Chapter 343, Hawai'i Revised Statutes and Section 11-200-17, Hawai'i Administrative Rules.

Que E. Qual-

Gene Awakuni, Chancellor University of Hawai'i West Oahu

December 2006

TABLE OF CONTENTS

EXEC	EXECUTIVE SUMMARY vii			
1.0	INTR	ODUCT	ION	1
	1.1	Projec	T PROFILE	1
	1.2	Applic,	ant and Landowner	2
	1.3	Planni	ING CONSULTANT	2
	1.4	Accept	ING AUTHORITY	2
	1.5	DESCRI	PTION OF THE PROPERTY	,
	1.6 1.7	Surro Backg	unding Land Uses Round	
2.0	DESC		N OF THE PROJECT	5
	2.1	Projec	t Goals and Objectives	5
	2.2	Need f	or the Project	5
	2.3	UHW	est Oʻahu Strategic Plan	7
	2.4	UHW	est Oʻahu Long Range Development Plan	
		2.4.1	Conceptual Land Use Plan	8
		2.4.2	UH West Oʻahu Lands	10
	- -	2.4.3	Private Development Lands	
	2.5	INFRAS	TRUCTURE IMPROVEMENTS	
		2.5.1	On-site Infrastructure	
			2.5.1.1 Koduways and Hame	17
			2.5.1.3 Wastewater Collection and Transmission	
			2.5.1.4 Drainage Facilities	18
		2.5.2	Off-site Infrastructure	
			2.5.2.1 Roadways	19
			2.5.2.2 Water Supply and Distribution	
			2.5.2.3 Wastewater Collection and Transmission	
	26	SUSTAN	2.5.2.4 Drainage facilities	
	2.0			
	2.7	ORDER	of Magnitude Costs	
3.0	ASSE	SSMENT	OF THE NATURAL ENVIRONMENT, POTENTIAL IMPAC	CTS, AND
	ΜΙΤΙ	GATION	MEASURES	
	3.1	CLIMAT	Έ	
	3.2	Topog	RAPHY	
	3.3	Drain/	AGE	
	3.4	Soils.		
	3.5	Groun	NDWATER RESOURCES/HYDROLOGY	

UNIVERSITY OF HAWAI'I WEST O'AHU

FINAL DRAFT ENVIRONMENTAL IMPACT STATEMENT

	3.6	NATUR	al Hazards	
	3.8	Fauna		
4.0	ASSES	GATION	OF THE HUMAN ENVIRONMENT, POTENTIAL IMPAC	1S, AND
	4.1	Archa	eological and Historic Resources	41
	4.2	CULTU	RAL RESOURCES	
	4.3	Agricu	JLTURAL IMPACT	
	4.4	NOISE.		
	4.5	Air Qu	ALITY	54
	4.6	Man- <i>n</i>	IADE HAZARDS	58
	4.7	Visual	Resources and Open Space	
	4.8	Social	CHARACTERISTICS	63
		4.8.1	Population	63
		4.8.2	Housing	65
		4.8.3	Lifestyle/Character of the Community	66
	4.9	Econo	MIC CHARACTERISTICS	67
		4.9.1	Employment, Personal Income, and Expenditures	
		4.9.2	Economic Factors/Government Revenues	70
	4.10	INFRAS	FRUCTURE AND UTILITIES	70
		4.10.1	Roadways and Traffic	
		4.10.2	Water Supply Facilities	83
		4.10.3	Wastewater Facilities	85
		4.10.4	Drainage Facilities	
		4.10.5	Solid Waste Disposal Facilities	
		4.10.0	Electrical Facilities	
	1 1 1	4.10./ Dublic	SEDVICES AND EACH TIES	
	4.11	1 11 1	Educational Eacilities	
		4.11.1	Polico Protoction	95
		4.11.2	Fire Protection	
		4 11 4	Medical Facilities	
		4.11.5	Recreational Facilities	
		4.11.6	Public Transit	
		4.11.7	Community Services	
5.0	RELA	TIONSH	IIP TO LAND USE PLANS, POLICIES, AND CONTROLS	105
	51			105
	5.7	STATE	OF HAWAII	105
	5.2	521	Environmental Impact Statement Law Chapter 343 Hawaii Revis	ed Statutes
		5.2.1	Environmental impact outernent Law, enapter 5 15, Hawan Revis	
		5.2.2	State Environmental Policy, Chapter 344, Hawaii Revised Statute	es106
		5.2.3	State Land Use Law, Chapter 205, Hawaii Revised Statutes	
		5.2.4	Coastal Zone Management Program, Chapter 205A, Hawaii Revis	ed Statutes
				111
		5.2.5	Hawaii State Plan, Chapter 226, Hawaii Revised Statutes	115

UNIVERSITY OF HAWAI'I WEST O'AHU

FINAL DRAFT ENVIRONMENTAL IMPACT STATEMENT

		5.2.6 Hawaii State Functional Plans	.123				
		5.2.7 103D-407 Hawaii Revised Statutes	<u>.125</u>				
	БЭ	5.2.8103D-408 Hawali Kevised Statutes	<u>. 126</u> 127				
	5.5	CITY AND COUNTY OF ΠΟΝΟLULU	127				
		5.3.2 Fwa Development Plan	130				
		5.3.3 Land Use Ordinance	.144				
		5.3.4 Special Management Area	.147				
	5.4	Other Required Permits and Approvals	147				
6.0	ALTE	RNATIVES TO THE PROPOSED ACTION	149				
	6.1	Alternatives Considered	149				
	6.2	ALTERNATIVE SITES	149				
	6.3	"No-Action" Alternative	151				
	6.4	Alternatives Related to Different Designs or Details of the Proposed Act	IONS				
		Which Would Present Different Environmental Impacts	153				
	6.5	Actions of a Significantly Different Nature Which Would Provide Sim	1ILAR				
		Benefits With Different Environmental Impacts	154				
	6.6	The Alternative of Postponing Action Pending Further Study	154				
7.0	CONTEXTUAL ISSUES						
	7.1	Relationship Between Local Short-term Uses of the Environment and	THE				
		Maintenance of Long-term Productivity	157				
	7.2	CUMULATIVE AND SECONDARY IMPACTS	158				
	,	7.2.1 Impacts on the Physical Environment	.159				
		7.2.2 Impacts on the Human Environment	.160				
		7.2.3 Impacts on the Socio-economic Environment	.161				
		7.2.4 Impacts on Infrastructure Facilities	.162				
		7.2.5 Impacts on Public Facilities and Services	.163				
	7.3	IRREVERSIBLE AND IRRETRIEVABLE COMMITMENTS OF RESOURCES	164				
	7.4	PROBABLE ADVERSE ENVIRONMENTAL EFFECTS THAT CANNOT BE AVOIDED	165				
		7.4.1 Rationale for Proceeding with the Project Notwithstanding Unavoid	lable				
		Effects	.167				
		7.4.2 Unresolved Issues	.16/				
8.0	CON	SULTATION	169				
	8.1	PRE-CONSULTATION EFFORTS	169				
		8.1.1 Individuals/Organizations Consulted During the Site Selection Process	169				
	8 2	8.1.2 Individuals/Organizations Consulted During the Pre-consultation Proc	<i>ess1</i> /2 175				
0.0			173				
9.0		UF FREFARERJ	1//				
10.0	GLOS	5SARY	179				
11.0	REFE	RENCES	183				

UNIVERSITY OF HAWAI'I WEST O'AHU

FINAL DRAFT ENVIRONMENTAL IMPACT STATEMENT

12.0	COMMENTS	ON	THE	DRAFT	ENVIRONMENTAL	IMPACT	STATEMENT	AND
	RESPONSES	•••••	•••••			•••••		187

LIST OF TABLES

Table	Page
<u>2.1</u>	Relative Need for Postsecondary Education (Based on Projected
	Change in Population (#), 2000-2020)
2.2	Relative Need for Postsecondary Education (Based on Projected
	<u>Change in Population (%), 2000-2020)7</u>
<u>2.3</u> 2.1	Proposed Land Use Acreage9
<u>2.4</u> 2.2	Preliminary Phasing Plan23
3.1	Alien (Introduced) Birds Observed on the Project Site
4.1	Hazardous Waste Sites61
4.2	Intersection Level of Service Summary79
4.3	Intersection Level of Service Summary (Phase 1)80
<u>4.4</u>	Proposed UH West Oʻahu Parking Stalls
<u>4.5</u> 4.4	Public School Enrollment93
<u>4.6</u> 4.5	Possible Public Schools for East Kapolei Students
<u>4.7</u> 4.6	'Ewa Parks99
<u>5.1</u>	Proposed PRU Permit/Zone Change Land Use Acreage145
<u>5.2</u>	Required Approvals and Permits
8.1	Pre-consultation Comment Letters172
8.2	EISPN Comment Letters174
<u>12,1</u>	DEIS Comment Letters 187

LIST OF FIGURES

Figure		Follows Page
1.1	Location Map	
1.2	Aerial Photograph	4
1.3	Тах Мар Кеу	4
1.4	Existing Tenant Farms	4
1.5	Kapolei Area Long Range Master Plan	4
1.6	State Land Use District Boundary Map	4
<u>1.7</u>	Surrounding Land Uses	<u>4</u>
2.1	Conceptual Land Use Plan	
2.2	Campus Plan	12
2.3	Proposed Off-Site Infrastructure	
2.4	Regional Roadway Network	
2.5	North-South Road Typical Section	
2.6	Conceptual Land Use Plan – Phase 1 Development	24
3.1	Flood Insurance Rate Map	
3.2	Detailed Land Classification	
3.3	Soil Survey	
3.4	Agricultural Land of Importance to the State of Hawai'i	
3.5	Abutilon menziesii Plant Clusters Map	
4.1	Visual Resources	62
<u>4.1a</u>	Visual Resources	
4.2	Conceptual Land Use Plan	
<u>4.3</u> 4.2	Ewa Development Plan – Public Facilities Map	100
5.1	Ewa Development Plan – Urban Land Use Map	
5.2	Existing Zoning Map	144
<u>5.3</u>	Proposed Zoning Map	<u>146</u>
6.1	Alternative Sites Location Map	150

LIST OF APPENDICES

Appendix A	Letter from the U.S. Department of the Army
Appendix B	Botanical Survey
Appendix C	Habitat Conservation Plan for Abutilon menziesii at Kapole
Appendix D	Avifaunal and Feral Mammal Field Survey
Appendix E	Archaeological Reconnaissance and Assessment
Appendix F	Noise Technical Report
Appendix G	Air Quality Technical Report
Appendix H	Traffic Study
Appendix I	Infrastructure Study

(This page intentionally left blank.)

EXECUTIVE SUMMARY

EXECUTIVE SUMMARY

DESCRIPTION OF THE ACTION

This Environmental Impact Statement (EIS) has been prepared in support of the proposed University of Hawai'i – West O'ahu (UH West O'ahu). The project involves the development of the UH West O'ahu campus and residential (including student housing and work force/affordable housing), commercial, mixed, and public uses on a 500.327-acre property makai of the H-1 Freeway, in 'Ewa, O'ahu. In 2005, the University decided to seek a private development rights to the non-campus/income-generating lands. The UH West O'ahu Lands comprise 213.8 acres and the Private Development Lands comprise approximately 286.5 acres of the approximately 500-acre property.

The UH West O'ahu project reflects a mixture of instructional, administrative, recreational, residential, and commercial uses that will replace vacant and cultivated lands. In support of this project, major infrastructure facilities will be developed, including access and circulation (roadways and pedestrian paths), drainage improvements, distribution lines for potable water, collection lines for wastewater, and communication/utility systems.

Plans for the approximately 500-acre property are presented in the University of Hawai'i West O'ahu Long Range Development Plan (PBR HAWAII, 2004) and the University of Hawai'i West O'ahu Long Range Development Plan Update (PBR HAWAII, 2006). This EIS provides a description of the proposed project and the natural environment in the project area. It also provides a detailed assessment of how the proposed project may impact existing environmental conditions, and recommends mitigation measures to minimize such impacts.

Additionally, the project involves the development of off-site infrastructure (i.e., water, wastewater, and drainage facilities) to accommodate the campus and community.

SIGNIFICANT BENEFICIAL AND ADVERSE IMPACTS

The current UH West O'ahu is a two-year, upper division, baccalaureate degree-granting campus located on the Leeward Community College campus in Pearl City. University of Hawai'i strategic plans, academic plans, and master plans have for many years documented the need for a new UH West O'ahu campus, in response to the future population growth planned for the 'Ewa and Central O'ahu regions, and the limited expansion area available at the University of Hawai'i - Mānoa (UH Mānoa). The *Ewa Development Plan* (City and County of Honolulu, 1997) also calls for the development of a University of Hawai'i campus, although not on the currently proposed project site. This commitment by the City and County of Honolulu (City) reflects its determination to provide additional higher education opportunities in 'Ewa during the 20-year planning horizon.

Development of the approximately 500-acre property will impact its physical resources; however, past and current uses of the property and adjacent properties (including historical sugarcane cultivation and limited diversified agriculture) have significantly altered the natural landscape. Major facility improvements proposed for the project include the installation of new potable water systems, wastewater collection systems, drainage improvements, transportation improvements, electrical and communication systems, and the development of administrative, instructional, support, residential, and commercial buildings. Regional transportation improvements planned by the State of Hawai'i (State) and the City include the construction of North-South Road and H-1 Freeway interchange, widening of Farrington Highway, and extension of Kapolei Parkway.

Potential environmental impacts (i.e., noise, soil erosion, fugitive dust and exhaust emissions, and temporary traffic disruption) will primarily occur during the construction period. Mitigation measures will be implemented, and once developed, the project will have no significant impact on noise, air quality, water quality, and archaeological and cultural resources. No endangered or threatened avian or mammalian species are expected to be impacted by the project, as none were observed on the property during a biological survey. A botanical survey of the property suggested that seeds of the endangered species, *Abutilon menziesii*, may be present in the soils where an individual plant once lived. A few individual plants of this species were also identified along the project's southeastern boundary. These plants and seeds would be removed with the proposed development; however, the *Habitat Conservation Plan for Abutilon menziesii at Kapolei* (HCP) was developed in 2004 to protect the species. Successful implementation of the HCP would significantly increase the number of new plants on O'ahu and improve the quality of the existing population in Kapolei.

Currently, portions of the property are leased under revocable permits and provide employment and generate income; however, no employment or income is generated from the vacant lands. With project development, the property will be transformed into a masterplanned University campus and community, providing opportunities to live, work, learn, play, and shop within the community.

PROPOSED MITIGATION MEASURES

Various mitigation measures will be implemented throughout the development of the project to minimize environmental impacts. All major infrastructure improvements will incorporate appropriate engineering and design methods to meet basic infrastructure requirements.

Drainage/Flood Control/Water Quality/Soil Erosion. <u>Runoff will be conveyed to a drainage</u> channel within the 300-foot-wide utility, drainage, and access corridor along the east side of the proposed North-South Road. The channel will divert flow in Kalo'i Gulch away from the UH West O'ahu property and into a detention basin at the downstream end of the channel above the Ewa Villages Golf Course. A proposed box drain system will divert the flow from Hunehune Gulch to the 11.2-acre detention basin proposed at the southern boundary of the property. The flow would then discharge into the regional detention basin through box

culverts under the North-South Road. Coordination between the University and its private development partner is required to determine an efficient way of integrating drainage systems.

<u>Flood control detention areas will be required until a downstream drainage connection to the</u> <u>Pacific Ocean is established at Oneula Beach Park.</u>

During construction, the grassing and watering of graded areas and the installation of dust screens and a silt fence will reduce fugitive dust emissions, mitigate potential soil erosion, and protect water quality. All grading operations will be conducted in full compliance with dust and erosion control requirements of the City and County of Honolulu's Grading Ordinance. All construction activities will comply with the State's fugitive dust regulations, and a grading permit and National Pollutant Discharge Elimination System (NPDES) permit will be obtained before construction begins.

A watering program will be implemented to minimize soil loss through fugitive dust emissions during construction. Other dust and erosion control measures include cleaning job-site construction equipment and establishing ground cover as quickly as possible after grading. During the public review period, the Department of Health Clean Air Branch recommended that a dust control management be prepared. The DOH CAB also provided additional examples of measures that can be implemented during construction:

- where possible, for areas involving mixed land uses, buffer zones be established to alleviate potential dust nuisance problems;
- planning the different phases of construction, focusing on minimizing the amount of dust-generating materials and activities, centralizing on-site vehicular traffic routes, and locating potential dust-generating equipment in areas of the least impact;
- providing an adequate water resource and watering program at the site prior to startup of construction activities;
- landscaping and providing rapid covering of bare areas, including slopes, starting from the initial grading phase;
- minimizing dust from shoulders and access roads;
- providing adequate dust control measures during weekends, after hours, and prior to daily start-up of construction activities; and
- controlling dust from debris being hauled away from the project site.

<u>Mitigation measures generally associated with best (water borne soil erosion) management</u> <u>practices include:</u>

- Early construction of drainage control features;
- Construction of temporary sediment basins to trap silt;
- Use of temporary berms and cut-off ditches where needed; and
- Use of temporary silt fences or straw bale barriers to trap silt.

A 5-acre detention basin near the campus and an 11.2-acre detention basin at the southern boundary of the property will be developed in Phase 1 to maintain existing runoff quantities

throughout project development and operation. The 5-acre detention basin near the campus will be expanded to 10 acres at full build out. These flood control dentention areas will be required until a downstream drainage connection to the Pacific Ocean is established at Oneula Beach Park.

Flora and Fauna. In accordance with both State and Federal regulations, mitigation measures have been identified in the *Habitat Conservation Plan for Abutilon menziesii at Kapolei* (HCP) prepared in consultation with the State DLNR. The major strategy designed to mitigate impacts and to benefit the species is the creation of three protected off-site wild populations on O'ahu from the single degraded Kapolei population. The new off-site populations will protect existing individuals as well as the genetic diversity of the existing population.

Under DLNR's interim management program, three new populations of ko'oloa'ula have been initiated at Koko Crater Botanical Garden, Ka'ena Point State Park, and the Honouliuli Unit of the Pearl Harbor National Wildlife Refuge. The Ka'ena Point State Park and Pearl Harbor National Wildlife Refuge are wild sites, and additional sites will continue to be pursued to yield three successful wild sites for propagation of ko'oloa'ula.

If an off-site preserve can be established, some of the plants would remain *in situ* for many years. Because the protection, propagation, and relocation of the ko'oloa'ula plant is a long-term undertaking, final implementation of the HCP will extend well into the project's construction period. Once construction and build-out of the campus is complete, use of the ko'oloa'ula in project landscaping and continued use of the ko'oloa'ula nursery for propagation will ensure a much larger and vigorous population of the ko'oloa'ula than would have occurred without development of the UH West O'ahu property. The HCP also proposes long-term management of the populations to occur concurrently with project development, over a period of approximately 20 years. The successful implementation of the HCP would significantly increase the numbers of new plants on O'ahu and improve the quality of existing populations on adjacent properties.

To mitigate the impact of planned Kapolei developments on the endangered species, Abutilon menziesii, the Habitat Conservation Plan for Abutilon menziesii at Kapolei (HCP) was prepared in consultation with the State Department of Land and Natural Resources (DLNR). Three new populations of Abutilon menziesii have been initiated at outplanting sites and are tentatively expected to persist.

No native, threatened, or endangered species were observed on the project site and none are expected to be impacted by the UH West O'ahu development.

Archaeological Resources. <u>Previously, the State Historic Preservation Division concurred that</u> <u>t</u>+he presence of any significant archaeological sites on the surface or subsurface of the property is unlikely due to the disruption caused by continuous sugarcane cultivation for nearly 70 years. This was confirmed by an archaeological reconnaissance and assessment, and a review of existing literature, historic records, and maps. <u>During the public review</u>

period both SHPD and the Office of Hawaiian Affairs (OHA) requested the preparation of an archaeological inventory survey. UH West O'ahu has contracted such a survey and the consulting archaeologist is in the process of conducting fieldwork. The survey and report will be completed prior to the submittal of any grading or building permits. In accordance with applicable laws and regulations, if any artifacts or human remains are discovered during construction activities, the contractor will be instructed to halt work and notify the State Historic Preservation Division (SHPD).

Noise. Construction noise impacts will be relatively short-term during daytime hours, and all construction activities will comply with State Department of Health (DOH) noise regulations. Traffic noise will increase, with or without the project, but is expected to remain within the limits of general noise guidelines. No residences will be built within 75 feet of Farrington Highway, and any residences within 225 feet of Farrington Highway will require some type of noise mitigation. Residences bordering North-South Road should be at least 100 feet from the edge of pavement, and any residences within 100 feet of North-South Road will require some type of noise mitigation. Effective noise mitigation measures could include constructing barrier walls and/or earth berms along roadways, installing air conditioners in buildings, and acoustically softening interior spaces.

Air Quality. <u>State air pollution control regulations require that there be no visible fugitive</u> dust emissions at the property line. Hence, an effective dust control plan must be implemented to ensure compliance with State regulations. Fugitive dust emissions can be controlled to a large extent by watering active work areas, using wind screens, keeping adjacent paved roads clean, and covering open-bodied trucks. Other dust control measures could include limiting the area that can be disturbed at any given time and/or mulching or chemically stabilizing inactive areas that have been worked. Paving and landscaping of project areas early in the construction schedule will also reduce dust emissions. Monitoring dust at the project boundary can help to evaluate the effectiveness of the dust control program. Exhaust emissions can be mitigated by moving construction equipment and workers to and from the project site during off-peak traffic hours.

All construction activities on the property will comply with State Air Pollution Control regulations and the provisions of Section 11-60.1-33, HAR, on Fugitive Dust. An effective dust control plan for the construction phase will be implemented, and particular care will be taken when construction activities take place near existing homes, businesses, or highways.

Measures to mitigate long-term, traffic-related air pollution include improvements to roadways, reduction of traffic, and reduction of individual vehicular emissions. Given that the more stringent State standards would likely be met during worst-case conditions, implementing any air quality mitigation measures for long-term traffic-related impacts is probably unnecessary and unwarranted. Nevertheless, the UH West O'ahu will incorporate public transit into the design of its campus, as it is located near planned rail transit nodes. The addition of buffer zones between walkways and roadways can help mitigate potential air quality impacts, and the UH West O'ahu will include acres of parks and open space. In

addition, all internal roads are currently planned to include planting strips or tree wells.

Short term impacts from fugitive dust and exhaust emissions may occur during the construction phase. An effective dust control plan will be implemented, and all construction activities will comply with State Air Pollution Control regulations and the provisions of Section 11-60.1-33, HAR, on Fugitive Dust. After construction, motor vehicles traveling to and from the proposed development will result in a long-term increase in air pollution emissions in the project area. Given that the more stringent State standards would likely be met during worstcase conditions, implementing any air quality mitigation measures for long-term traffic-related impacts is probably unnecessary and unwarranted.

Traffic. Regional transportation improvements planned by the State and City include the construction of North-South Road and H-1 Freeway interchange, the widening of Farrington Highway, and the extension of Kapolei Parkway. By the year 2015, much of the future roadway network in the project vicinity will be in place. All intersections in the project vicinity are projected to operate acceptably during the peak hours of traffic, with or without the proposed development.

<u>Future transit service near the UH West O'ahu property is expected to increase significantly.</u> <u>The DTS is currently conducting an Alternative Analysis Study for the Honolulu High-Capacity Transit Project (HHCTP), which explores the feasibility of several alternative ways to implement a high-capacity transit line within a 23-mile corridor extending from Kapolei to UH Mānoa.</u>

The campus is being planned to be transit-ready and the use of public transportation systems will be encouraged. With current plans to include UH Mānoa and UHWO at or near the end of each rail transit alignment alternative, the development of the UH West O'ahu campus is expected to provide greater ridership numbers, than if the site was developed as originally planned (mostly lower-density residential). Ridership is not expected just from students, but from faculty and staff living outside of the project area.

Proposed residential units and bicycle facilities within the approximately 500-acre property will encourage walking and bicycling within the area. This will help to reduce vehicular traffic in the vicinity of the campus. Ridership on rail transit would be especially attractive if the mode of transit could accommodate temporary bicycle storage, allowing riders to carry their bicycles with them.

The UH West O'ahu campus will be a higher education facility and major employment center. West O'ahu residents will have the opportunity to enroll in University courses and work near their homes. This opportunity will help to re-orient travel to private Universities, UH Mānoa, and workplaces outside of West O'ahu to UH West O'ahu. As such, UH West O'ahu would help to manage the growth in vehicular travel on routes into and within Honolulu and would result in better traffic conditions than if it did not occur.

Visual Resources. Development of the UH West O'ahu will comply with the City's Land Use Ordinance (LUO), which controls the type of land use, height, density, building setbacks, and landscaping of all properties throughout O'ahu. The visual appearance of the property will change from agricultural/vacant land to a master-planned campus and community that will provide employment and higher education opportunities. The campus will be oriented to capitalize on natural landforms, and views from internal roadways will be considered as practical. Extensive landscaping will mitigate the visual impact of the campus and integrate the built environment with surrounding land uses. Major roadways, parking areas, and areas surrounding all major structures will also be landscaped in accordance with a landscape master plan.

Infrastructure/Public Facilities/Utilities. New major backbone infrastructure (i.e., water, wastewater, and drainage systems) will be sized in accordance with project requirements. <u>UH West O'ahu and/or a private developer will be required to install the necessary water</u> system improvements to serve the proposed development. Water lines within dedicable roads are planned to be dedicated to the BWS where practicable. <u>The property will connect</u> to a non-potable water system when one is developed for East Kapolei.

All wastewater plans will conform to applicable provisions of Chapter 11-62, HAR "Wastewater Systems," and the DOH Wastewater Branch reserves the right to review the detailed wastewater plans for conformance to applicable rules. In addition, UH West O'ahu will work with the Board of Water Supply and utilize recycled water for irrigation and other non-potable water purposes in the open spaces and for landscaping areas to the extent practicable. During the DEIS Public Review Period, the Department of Planning, Site Development Division, Wastewater Branch wrote that the "...projected average flow of 1.68 mgd is included in the approved Wastewater Master Plan for East Kapolei, dated June 2006."

The UH West O'ahu campus will be a public higher education institution. The campus will ultimately support 7,600 students, providing educational opportunities and employment opportunities for Hawai'i's residents. Additionally, a 12-acre public elementary school site is proposed at the southern portion of the property (although there is a possibility that additional facilities, such as a "UH Lab School" may be developed to meet State Department of Education requirements). The school will be located near proposed residential parcels and the State Department of Hawaiian Home Lands (DHHL) residential development proposed south of the property. Several parks will be located within residential parcels.

Electrical and communication improvements necessary to support the requirements of this project will be served from existing and proposed utility systems. Two Hawaiian Electric Company (HECO) substations are proposed – one within the UH West O'ahu Lands and the other within the Private Development Lands. Sustainability guidelines have been established for the UH West O'ahu. The guidelines strive to achieve a Leadership in Energy and Environmental Design (LEED) NC silver standard for campus development and community performance standards based on the LEED NC v2.2 rating system.

DBEDT recommended that UHWO consult the City and County of Honolulu Energy code early in the project. DBEDT also noted that HECO may also have demand-side management programs that offer rebates for installation of energy-efficient technologies.

Based on DBEDT's recommendations, UHWO's mechanical and electrical consultants, in consultation with UHWO's sustainability consultant, will be directed to review the City and County of Honolulu's Energy code early in the project and to consult with HECO on demand-side management programs that offer rebates for installation of energy-efficient technologies.

While current plans for the development of the UHWO campus includes both energy efficient design and the use of electricity developed and sold by HECO, the site is suited for the use of renewable energy technologies such as wind turbines and photovoltaics.

Solid Waste. A construction waste recycling plan will be prepared before construction is initiated. All solid waste generated during project construction shall be directed to a Department of Health permitted solid waste disposal or recycling facility. Also, all highway and road construction improvement projects funded by the State or a county or roadways that are to be accepted by the State or a county as public roads shall utilize a minimum of ten percent crushed glass aggregate as specified by the Department of Transportation in all base-course (treated or untreated) and sub-base when the glass is available to the quarry or contractor at a price no greater than that of the equivalent aggregate.

Recycling shall be encouraged within the project including the reuse and recycling of green waste generated during construction clearing and grubbing activities, the use of recycled construction and demolition wastes and the use of materials made from recycled products, the use of locally produced compost as available for landscaping, and the provision of space for recycling bins in the detailed design of the community.

At full build-out, the solid waste generated by the project is estimated to average approximately 80,926 pounds per day. It should be noted that this estimate does not account for solid waste that would be recycled, which would be a considerable amount. Sustainability guidelines have been established for the UH West O'ahu Lands. The goal for waste management is to appropriately reduce, reuse and recycle materials, to minimize generation of solid waste and achieve diversion from landfills. The Private Development Lands will strive to achieve the applicable design criteria and the recommended LEED community performance standards. UH West O'ahu will promote the optimal use of solid wastes through programs of waste prevention, energy resource recovery, and recycling with the goal that all of its wastes are utilized pursuant to Chapter 344-4.2, HRS.

ALTERNATIVES CONSIDERED

The University of Hawai'i West O'ahu Campus Site Selection Study (2002) was conducted to assist the University of Hawai'i Administration and Board of Regents in the selection of a

permanent site for the West O'ahu campus. After a thorough review of 16 potential sites, the Kapolei Makai site (the proposed site) was selected based on: 1) site development potential and campus character; 2) schedule for project development; 3) costs; and 4) community concerns.

UNRESOLVED ISSUES

Funding. The University is currently entertaining a number of funding strategies for the initial 1,520 student campus. These strategies include a development partnership, land sale and/or legislative funding, or a combination of these options. Funding for campus development beyond the initial phase is currently an unresolved issue. The initial campus for 1,520 students will be developed by the University's development partner. Construction of the campus beyond the initial phase will occur on an as needed basis through a public private partnership and/or funding from the State.

Public/Private Partnership. The University and its private development partner are currently working on an agreement for the UH West O'ahu development. <u>The exact acreage allocated to the Private Developer shown in this EIS is approximate and will not be known until the agreement with the private development partner is finalized. In any case, adequate land area will be set aside for a campus than can eventually accommodate up to 7,600 students in the future.</u>

Completion of Archaeological Inventory Survey Report. OHA recommends that the project area undergo an Archaeological Inventory Survey (AIS), as a new study or supplement to the previous, in which all aspects of applicable 6E administrative rules and revised statutes are satisfied. The AIS will include a substantial subsurface inventory as much of the surface architecture may have been destroyed during previous commercial agriculture activities.

Dedication of Roads, Drainage Facilities and Their Related Infrastructure. Dedication of Roads, Drainage Facilities and Their Related Infrastructure. During the public review period, the City and County of Honolulu Department of Facility Maintenance wrote: *"Roads, drainage facilities and their related infrastructure should be owned and maintained by the State of Hawaii for this State facility, similarly to that at the Manoa Campus. The City and County of Honolulu will not accept the dedication or maintenance of this facility."* This will be identified as an unresolved issue since the Department of Planning and Permitting is specifying roads within the 500-acre project to provide regional connectivity, and the non-campus lands will include residential and commercial uses that will generate property tax revenue to the City and County of Honolulu, it is felt that the non-campus roads should be allowed to be dedicated to the County.

COMPATIBILITY WITH LAND USE PLANS AND POLICIES

The proposed project does not require an amendment to either the *City and County of Honolulu General Plan* or the *Ewa Development Plan*. In 1997, the City Council adopted the *Ewa Development Plan*, which permits the development of UH West O'ahu – upon approval of the Plan Review Use (PRU) permit application reviewed by the City. The current conceptual plan for the approximately 500-acre UH West O'ahu property has been designed to address the future educational, residential, commercial, and recreational needs of the 'Ewa region.

The Kapolei area, as shown in the *Kapolei Area Long Range Master Plan* prepared by The Estate of James Campbell, is located on the 'Ewa Plain. The 'Ewa Plain encompasses approximately 32,000 acres, or about eight percent of the island of O'ahu. The City of Kapolei and surrounding developments have been master-planned to provide for commercial, industrial, resort, recreational, and residential uses. Existing and planned developments in 'Ewa, including the UH West O'ahu, will enable Kapolei residents to live, work, play, shop, and attend an institution of higher learning close to their homes and businesses. The Estate of James Campbell transferred the approximately 500-acre property to the State, on the condition that it be developed as the UH West O'ahu campus.

LISTING OF PERMITS AND APPROVALS REQUIRED

The following is a preliminary list of major permits and approvals required for the project.

PERMIT OR APPROVAL							
Environmental Impact	Action	51/(105					
Statement (EIS) in compliance	Governor, State of Hawai'i						
with Chapter 343, HRS	,						
Motion to Amend Decision	Land Lise Commission	Motion to be filed at the					
and Order		time the FEIS is submitted.					
		<u>Application to be</u>					
	Department of Planning and	submitted after the					
Plan Review Use (PRU) Permit	Permitting / City Council	acceptance of the FEIS					
		<u>Change application</u>					
		<u>Change application.</u>					
		Application to be submitted after the					
Zone Change	Department of Planning and	acceptance of the EEIS					
Zone enange	Permitting / City Council	concurrently with the PRU					
		Permit application.					
		Application to be					
Culturini ora Americanal	Department of Planning and	submitted after the Zone					
Subdivision Approval	Permitting	Change application is					
		approved.					
		Application to be					
Park Dedication	Department of Planning and	submitted with an					
	Permitting	application for Subdivision					
		<u>Approval.</u>					
	Department of Planning and	Application to be filed					
Building/Grading Permits	Permitting	after the Zone Change					
		<u>application is approved.</u>					
Installation of Power Lines and	State Public Litilities	with HECO Will be					
Substations	Commission	approved prior to					
		occupancy.					
National Pollutant Discharge		Application to be					
Elimination System (NPDES)	State Department of Health	submitted prior to Grading					
Permit		Permits.					

Required Approvals and Permits

(This page intentionally left blank.)

1.0 Introduction

1.0 INTRODUCTION

1.1 **PROJECT PROFILE**

The following summary describes the existing entitlements and proposed actions.

Project Name:	University of Hawaiʻi – West Oʻahu (UH West Oʻahu)
Location:	Kapolei, Oʻahu, Hawaiʻi
Judicial District:	'Ewa
Landowner:	University of Hawai'i
Applicant:	University of Hawai'i
Тах Мар Кеу:	9-1-016: 120, 127, and 129
Property Area:	500.327 acres
Existing Use ¹ :	The majority of the approximately 500-acre site is presently under revocable leases to Aloun Farms, Inc. and A.M. Enterprise, Inc.
	A small portion (0.826 acres) of the approximately 500-acre site is presently encumbered by a revocable permit to Kapolei People's Inc. (Kapolei Golf Course) for parking lot purposes.
Proposed Use:	7,600-student campus, student housing, workforce/affordable housing, mixed use (retail, office, and residential), residential (low-, medium-, and high-density), elementary school(s), parks, electrical substations, detention basin, and roads
Land Use Designations:	 State Land Use: Urban District Ewa Development Plan: Low and Medium Density Residential, High Density Residential, and Transit Node Zoning: AG-1 Restricted Agriculture Special Management Area: Outside of the Special Management Area

¹ The site was formerly planned for residential development as part of the East Kapolei Master Plan by the State Housing and Community Development Corporation of Hawaii (HCDCH). The site was to include 343.6 acres of single-family residences (up to 2,749 units), 88.5 acres of multi-family residential units (up to 1,062 units), 10 acres of commercial use, 2 elementary schools on a total of 24 acres, and 9 acres of park space.

Permits/Approvals	Compliance with Chapter 343, Hawaii Revised Statutes (HRS)
Required:	Motion to Amend the Decision and Order
-	Plan Review Use (PRU) Permit
	Zone Change
	Subdivision Approval
	Park Dedication
	Building/Grading Permits
	Public Utilities Commission approval of substations and powerlines
	National Pollutant Discharge Elimination System (NPDES) Permit

EIS Accepting Authority: Governor, State of Hawai'i

1.2 APPLICANT AND LANDOWNER

The University of Hawai'i is the applicant for required land use entitlements and the landowner of the property.

Contact Person: Mr. Gene Awakuni, Chancellor University of Hawai'i – West O'ahu Office of the Chancellor 96-129 Ala Ike Pearl City, Hawai'i 96782 Telephone: (808) 454-4750 Facsimile: (808) 453-6076

1.3 PLANNING CONSULTANT

The University of Hawai'i has contracted PBR HAWAII to prepare applicable environmental documents in compliance with Chapter 343, *Hawaii Revised Statutes* (HRS).

Contact Person: Mr. Vincent Shigekuni, Vice President PBR HAWAII 1001 Bishop Street ASB Tower, Suite 650 Honolulu, Hawai'i 96813 Telephone: (808) 521-5631 Facsimile: (808) 523-1402

1.4 ACCEPTING AUTHORITY

In accordance with Section 11-200-4, *Hawaii Administrative Rules* (HAR), "the governor, or an authorized representative, whenever an action proposes the use of state lands or the use of state funds, or, whenever a state agency proposes an action within section 11-200-6(b)," shall

be the final authority to accept an environmental impact statement. Consequently, the Governor of the State of Hawai'i is the accepting authority for the *University of Hawai'i West O'ahu Environmental Impact Statement*.

Contact Person: Honorable Linda Lingle, Governor c/o State of Hawai'i Office of Environmental Quality Control 235 South Beretania Street, Suite 702 Honolulu, Hawai'i 96813 Telephone: (808) 586-4185 Facsimile: (808) 586-4186

1.5 DESCRIPTION OF THE PROPERTY

The UH West O'ahu property includes approximately 500 acres of land in the 'Ewa district and Honouliuli ahupua'a (see Figures 1.1 and 1.2). The property is bounded to the north by Farrington Highway, to the east by vacant land for the future North-South Road, to the south by the proposed Department of Hawaiian Home Lands (DHHL) East Kapolei Development Parcel B residential subdivision (currently being constructed), and to the west by Kapolei Golf Course and a small portion of the Villages of Kapolei. The property is comprised of three parcels identified as Tax Map Key numbers 9-1-016: 120, 127, and 129 (see Figure 1.3). It is generally flat to slightly sloping, and traversed by Kalo'i Gulch and Hunehune Gulch (a tributary to Kalo'i Gulch).

The northeastern portion of the property is being leased to Aloun Farms, Inc., and the northwestern portion of the property is being leased to A.M. Enterprise, Inc. for agricultural purposes (see Figure 1.4). A small portion (0.826 acres) of the property is currently encumbered by a revocable permit to Kapolei People's, Inc. (Kapolei Golf Course) for parking purposes.

1.6 SURROUNDING LAND USES

The Kapolei area, as shown in the *Kapolei Area Long Range Master Plan* prepared by The Estate of James Campbell, is located on the 'Ewa Plain (see Figure 1.5). The 'Ewa Plain encompasses approximately 32,000 acres, or about eight percent of the island of O'ahu. The City of Kapolei and surrounding developments have been master-planned to provide commercial, industrial, resort, recreational, and residential uses. With the proposed UH West O'ahu development, 'Ewa residents will have the opportunity to live, work, learn, play, and shop close to their homes.

The property is bounded to the north by Farrington Highway, to the east by vacant land for the future North-South Road, to the south by the proposed DHHL East Kapolei Development Parcel B residential subdivision, and to the west by Kapolei Golf Course and a small portion

of the Villages of Kapolei. The approximately 500-acre property is located within the Urban State Land Use District (see Figure 1.6).

Off-site infrastructure required for the development is proposed to be located mauka and makai of the property. This area includes the H-1 Freeway, Farrington Highway, the North-South Road, lands formerly planned for development by the HCDCH, lands currently planned for development by the DHHL and D.R. Horton – Schuler Division, and another UH West O'ahu property (the former campus site). The majority of land mauka of the property is within the Agricultural State Land Use District (see Figure 1.6) and is currently leased to Sugarland Farms and Rocker G. Livestock. According to Section 205-4.5, *Hawaii Revised Statutes*, public, private, and quasi-public utility lines, major water storage tanks, and appurtenant small buildings such as booster pumping stations are permissible uses within the Agricultural State Land Use District. The land makai of the property is within the Urban State Land Use District.

1.7 BACKGROUND

To identify an appropriate site for the UH West O'ahu campus, the University conducted the *University of Hawai'i West O'ahu Alternative Sites Evaluation* and the *University of Hawai'i West O'ahu Campus Site Selection Study* (Site Selection Study) in 2002. The study evaluated 16 potential sites, and on September 13, 2002, the University of Hawai'i Board of Regents selected the approximately 500-acre site as the permanent site for the campus.

With limited capital funds available for construction of a new campus, the University is forming a partnership with a private partner and its development team to construct most of the first phase of the campus with private funds in exchange for the development rights for non-campus-related lands within the approximately 500-acre property.





Area to be included in the PRU

Figure 1.1 Location Map University of Hawai'i West O'ahu



Figure 1.2 Aerial Photograph University of Hawai'i West O'ahu





Figure 1.3 Tax Map Key

University of Hawai'i West O'ahu





Figure 1.4 Existing Tenant Farms University of Hawai'i West O'ahu





Source: The Estate of James Campbell



Legend



Agricultural District Conservation District Urban District

Figure 1.6 State Land Use District Boundary Map **University of Hawai'i West O'ahu**







Figure 1.7 Surrounding Land Uses University of Hawai'i West O'ahu East Kapolei, O'ahu





2.0 Description of the Project
2.0 DESCRIPTION OF THE PROJECT

2.1 **PROJECT GOALS AND OBJECTIVES**

The project objectives established by the University of Hawai'i are to create an integrated campus community that will: 1) provide educational opportunities to the communities within the UH West O'ahu geographic service area; 2) accommodate the overall educational program needs of the University for an initial campus of 1,520 students developed by a private developer and in operation by the fall of 2009; 3) provide lands for campus expansion and growth to accommodate a population of 7,600 students; 4) accommodate an appropriate mix of campus and regional supporting land uses including a mixed-use University Village, commercial uses, residential uses (including student housing and work force/affordable housing), and other community-supporting land uses.

2.2 **NEED FOR THE PROJECT**

The UH West O'ahu occupies a unique niche within the University of Hawai'i system. It is currently a two-year, upper division, baccalaureate degree-granting campus located adjacent to Leeward Community College (LCC). The UH West O'ahu is fully accredited by the Western Association of Schools and Colleges (WASC) and had a Fall 2005 semester enrollment of 858 students. The UH West O'ahu emphasizes access for all residents of Hawai'i by offering classes during the day, evening, and weekend. It delivers classes to five islands via the internet, interactive television, video streaming, and in-person. About 80 percent of students live on O'ahu, and the remaining students live on the neighbor islands and study via distance education options.

The UH West O'ahu student body has an average age of 33, and most are working adults. At UH West O'ahu, a substantial number of students are Hawaiian/part Hawaiian (19 percent). The UH West O'ahu curriculum is based on a firm foundation in the liberal arts and allows students to specialize in 13 discipline areas within four degree programs. Student demographics are expected to change dramatically as the UH West O'ahu becomes a four-year campus. The average age of students is expected to decrease, and additional support services are likely to be provided since younger students tend to expect more campus activities and guidance than older students.

Existing UH West O'ahu facilities consist of 30,000 square feet of space in 29 portable buildings that include seven classrooms, faculty and staff offices, a computer lab, a writing center, and a faculty and student lounge. The UH West O'ahu campus has its own small library that is separate from, but housed within, the LCC library. It also uses other LCC facilities, including the bookstore and cafeteria. In March 2003, a WASC team evaluated the physical plant of the UH West O'ahu campus and expressed concern that the facility was

UNIVERSITY OF HAWAI'I WEST O'AHU Final Draft Environmental Impact Statement

nearly at capacity with little room for expansion of enrollment or academic programs. The proposed development of a new campus in Kapolei will greatly enhance the growth and expansion potential of the existing UH West O'ahu campus at LCC.

The proposed UH West O'ahu will be a four-year university in the 'Ewa region, where the greatest residential and population growth is planned on O'ahu. According to the City Department of Planning and Permitting (DPP), the year 2000 population for the Ewa Development Plan area was 68,718, compared to the 1990 population of 42,931 (DPP, 2004). This represents a population increase of about 60 percent in 'Ewa over 10 years, compared to a population increase of about 4.8 percent for O'ahu over the same 10 years. In the *Annual Report on the Status of Land Use on Oahu: Fiscal Year 2003*, the DPP projected the year 2010, 2020, and 2025 populations of the Ewa Development Plan area to be 96,332; 141,864; and 164,462, respectively. The proposed UH West O'ahu campus will improve access to higher education, as it will be located in an area that has undergone rapid population growth and is projected to continue growing.

With the expected population increase in the 'Ewa region, the proposed UH West O'ahu campus is expected to target students in the surrounding regions to alleviate the burden on the UH Mānoa campus. The relative need for postsecondary education based on projected changes in population from the year 2000–2020, are shown in Tables 2.1 and 2.2.

Table 2.1 Relative Need for Postsecondary Education (Based on Projected Change in Population (#), 2000-2020



UNIVERSITY OF HAWAI'I WEST O'AHU Final Draft Environmental Impact Statement

Table 2.2 Relative Need for Postsecondary Education (Based on Projected Change in Population (%), 2000-2020



Several meetings with civic groups and presentations to neighborhood boards in 'Ewa, Kapolei, and other areas of West O'ahu were made during the campus site selection process. These groups endorsed building and locating the UH West O'ahu campus in the Kapolei area and overwhelmingly supported the selection of the currently proposed site for the permanent campus.

2.3 UH WEST O'AHU STRATEGIC PLAN

The University of Hawai'i - West O'ahu Strategic Plan 2002-2010 was adopted by the University's Board of Regents in November 2002. The plan establishes a vision and mission for the campus, identifies strategic priorities (goals) for accomplishing the mission, and lists critical success factors and key performance indicators for measuring the success of the school in fulfilling its mission.

As specified in the mission statement, UH West O'ahu is committed to the continuing development of the region through both innovative educational offerings and public service activities. The mission statement emphasizes quality teaching and flexible class schedules to foster life-long learning, enabling students to pursue career-related education coupled with the values, ideas, and challenges of the liberal arts. By offering approximately half of all courses in the evening or during the weekend, UH West O'ahu strives to meet the educational needs of recent high school graduates and non-traditional students. The academic program structure for the institution stresses the exploration of interdisciplinary

studies, cross-cultural and international studies, and communication skills.

The vision, mission statement, goals, and priorities established in the strategic plan served as the foundation for the *University of Hawai'i* - *West O'ahu Long Range Development Plan* (PBR HAWAII, 2004). The plan for the new campus ensures that the growing population of West O'ahu, as well as neighbor islands, has access to a comprehensive range of baccalaureate degree opportunities without having to relocate or commute undue distances.

2.4 UH WEST O'AHU LONG RANGE DEVELOPMENT PLAN

The Long Range Development Plan (LRDP) for the UH West O'ahu was adopted by the Board of Regents in July 2004. Since the adoption of the LRDP, the University accomplished a major milestone, in July 2005, with the selection of a development partner for the first phase of the campus. The University has been working with the private development partner on a comprehensive plan and development implementation strategy for the entire UH West O'ahu community.

During the course of refining the plan and implementation strategy, numerous changes and updates were made to the 2004 LRDP. As such, the University has prepared the 2006 LRDP Update, which is a comprehensive plan for the UH West O'ahu campus and surrounding lands within the University's approximately 500-acre property. The 2006 LRDP Update serves as the foundation for this EIS.

2.4.1 Conceptual Land Use Plan

The conceptual land use plan envisioned in the 2006 LRDP Update focuses on creating an integrated campus community (see Figure 2.1). Within the approximately 500-acre property, 213.8 acres of land to be owned by the University are designated for the development of a 7,600-student campus (including a possible "Lab School"), University Village, mixed uses, student housing or campus expansion, work force/affordable housing, roads, and a Hawaiian Electric Company, Inc. (HECO) electrical substation. These lands are herein referred to as "UH West O'ahu Lands".

The remaining 286.5 acres (herein referred to as "Private Development Lands") will include residential uses, mixed uses, roads, an elementary school site, a detention basin, and a HECO electrical substation. <u>(The exact acreage allocated to the Private Developer shown in this EIS is approximate and will not be known until the agreement with the private development partner is finalized. In any case, adequate land area will be set aside for a campus than can eventually accommodate up to 7,600 students in the future.)</u>

Proposed land uses are shown in Table 2.3 and described in the following pages.



Figure 2.1 Conceptual Land Use Plan University of Hawai'i West O'ahu



UNIVERSITY OF HAWAI'I WEST O'AHU FINAL DRAFT ENVIRONMENTAL IMPACT STATEMENT

Table 2.3 – Proposed Land Use Acreage							
LAND USE	ACREAGE	UNITS/ ACRE	RESIDENTIAL UNITS (approximate)	COMMERCIAL SQUARE FOOTAGE			
UH WEST O'AHU LANDS							
UH West O'ahu Campus/Mixed-Use/Housing							
7,600 Student Campus (including 10-acre detention basin <u>and possible "Lab School"</u>)	103.5	-	-	-			
Student Housing/ Mixed Use or Campus Expansion Parcel A (531 student housing units, 115 residential units)	38.0	17	646	331,056			
Student Housing or Campus Expansion Parcel B	12.1	19	230	-			
HECO Substation	1.0	-	-	-			
Mixed Use and Affordable Housing							
Mixed Use (Retail, Office, Residential)	15.1	10	151	164,439			
Work Force/Affordable Housing (Multi Family) Campus Expansion/Multi-Family Housing/Mixed Use	22.2	16	355	-			
Mixed Use Parcel C	10.2	10	102	111,078			
Roads							
Farrington Highway Expansion	4.4	-	-	-			
Roads A, B, C, and D	7.3	-	-	-			
Total for UH West O'ahu Lands	213.8	-	1,484	606,573			
PRIVATE	E DEVELOPMEN	it Lands					
Residential							
Residential Parcel A (High Density, Multi- Family)	33.1	16	530	-			
Residential Parcel B (Medium Density, Multi Family)	20.1	12	241	-			
Residential Parcel C (Low Density, Single Family)	30.5	6	183	-			
Residential Parcel D (Low Density, Single Family)	30.3	6	182	-			
Residential Parcel E (Medium-Low Density, Multi-Family)	56.1	10	561	-			
Residential Parcel F (High Density, Multi Family)	24.7	16	395	-			
Residential Parcel G (Medium Density, Multi Family)	20.7	12	248	-			
Commercial							
Mixed Use Parcel A	10.5	10	105	114,345			
Mixed Use Parcel B	11.2	10	112	121,968			
Other							
Elementary School	12.0	-	-	-			
Detention Basin	11.2	-	-	-			
HECO Substation	1.0	-	-	-			
Farrington Highway Expansion	1.1	-	-	-			
Roads C, D, E, F, and G	24.0	-	-	-			
Total for Private Development Lands	286.5	-	2,557	236,313			
TOTAL	500.3	-	4,041	842,886			

2.4.2 UH West O'ahu Lands

The UH West O'ahu Lands comprise 213.8 acres and include a 7,600-student campus₇; University Village₇; mixed-use parcels₇; student housing; multi-family housing and/or campus expansion₇; work force/affordable housing; a HECO substation site₇; and roadways. <u>Most of</u> the development of the UH West O'ahu Lands will comply with LUO development standards for the proposed underlying zoning districts, however, the campus lands will likely require "variances" from the developments of the underlying zoning districts, as other colleges and universities do (through the Plan Review Use process).

UH West O'ahu 7,600 Student Campus. The 103.5-acre UH West O'ahu campus will accommodate a population of 7,600 students. The campus is situated at the center of the approximately 500-acre property to facilitate "town and gown" interaction between the campus and adjacent land uses. Within the campus, land uses are organized to accommodate campus functional requirements, with consideration for future campus expansion. An approximately 10-acre detention basin is also included within the campus.

The vision for the UH West O'ahu campus follows:

A CAMPUS FOR ALL

Located within the ahupua'a of Honouliuli, the UH West O'ahu campus will serve as the premier four-year public university serving the Leeward and Central O'ahu region. The UH West O'ahu campus is envisioned as a sustainable campus environment which provides a caring and nurturing academic setting for its diverse faculty and student population, the surrounding community, and the region. Viewed from the H-1 Freeway and surrounding communities, the UH West O'ahu campus will be a distinctive landmark while remaining respectful of its neighbors. Upon entering the campus, there is a feeling of ho'okipa (hospitality) where students, faculty, staff, visitors and the community will feel that the campus is welcoming and accessible to them. The community is especially invited to the campus to encourage life long learning and build stronger ties with the region. The campus will strive to embrace all elements of sustainability, and its architecture, open space, and landscape will work in unison to foster a Hawaiian sense of place that is also reflective of the 'Ewa region, both its natural attributes and its cultural history.

ARCHITECTURE

Buildings on the campus will be designed to be flexible and accommodate diverse space needs, creating an environment that comfortably supports innovative teaching, learning opportunities, and cultural venues offered by the University. A unified architectural theme will be established for the campus to ensure that the buildings will be scaled to reflect a distinct sense of place on the 'Ewa Plain.

OPEN SPACE

A hierarchy of outdoor open spaces or "outdoor rooms" linked through a pedestrian network will be provided on the campus to create outdoor learning environments that promote interaction between students, faculty and the community. The spaces will include: small outdoor spaces to encourage quiet student interaction; larger spaces for informal group gatherings; a major open space for large campus functions, such as graduations and community-related activities; and possibly, an ethno-botanical garden that will incorporate native Hawaiian plantings.

LANDSCAPE

As part of the outdoor learning environment, the campus will incorporate xeriscape techniques aimed at creating a sustainable landscape that complements the dry climate, pays tribute to the region's agricultural past, and incorporates planting of native vegetation. The landscape will also serve to create a pedestrian- and bicycle-friendly environment for students, providing greenery and shade from the hot climatic conditions of the 'Ewa plain.

SUSTAINABILITY

The UH West O'ahu campus's sustainability guidelines establish a vision of a distinct campus within a vibrant neighborhood. The plan is to create a development that is environmentally and financially sustainable, with all the urban qualities distinguishing it within the context as a unique, vibrant, attractive and healthy community.

Based on the vision described above, the concept for the UH West O'ahu campus plan (see Figure 2.2) is to create a "sustainable campus community" that is welcoming and accessible, and creates a feeling of *ho'okipa* (hospitality) toward students, faculty, staff, and the community. Establishing links between the campus, the University Village, and the Private Development Lands is essential to creating this "sustainable campus community". The University Village will provide the necessary services and activities for faculty and students, and will be accessible via a variety of modes of transportation (i.e., walking, biking, driving, and public transportation). The campus plan is organized around a gateway plaza located near the terminus of Roads A and B, and linked to the mixed-use University Village and the intermodal transit facilities. To emphasize the importance of this plaza to the campus plan concept, it is referred to by a name derived from the site itself. The "Kalo'i Plaza" name references Kalo'i Gulch on the property.²

The plaza will be designed to accommodate a large amount of pedestrian activity and will cater to both large formal gatherings and small informal gatherings. Radiating outward from Kalo'i Plaza is a series of pedestrian malls that extend into the campus and surrounding mixed-use village.

To encourage interaction between faculty and students, the campus will be designed to provide a hierarchy of open spaces from formal plazas to intimate gathering areas. An interconnected series of smaller pedestrian paths extends outward from the malls and major pathways, linking the various campus activities. A vehicular loop roadway is provided so as not to disrupt pedestrian traffic within the core of the campus.

² The actual naming of plazas and pedestrian malls, like buildings and other campus improvements, is under the authority of the Board of Regents, per Board of Regents Policies, Chapter 11, Section 11-4.

Student Housing/Mixed Use or Campus Expansion Parcel A. Student housing will be provided adjacent to and north of the campus, at the intersection of Farrington Highway and the future North-South Road. This 38-acre parcel will provide approximately 646 residential units (531 student housing units and 115 residential units) at a density of 17 units per acre. As an alternative, this parcel or portions of the parcel could also serve as lands for campus expansion, should the University decide to lower resident housing requirements for the campus.

Student Housing or Campus Expansion Parcel B. Located adjacent to the Farrington Highway campus entry (Road A), this 12.1-acre parcel could accommodate approximately 230 residential units at a density of 19 units per acre. As an alternative, this parcel or portions of the parcel could also serve as lands for campus expansion, should the University decide to lower resident housing requirements for the campus.

Together with units in the 38-acre Student Housing/ Mixed Use or Campus Expansion Parcel A (previously described), this parcel could help fulfill the need for approximately 2,280 beds for the 7,600-student resident population.

During the Draft EIS public review period, the State Office of Environmental Quality Control wrote: "Is there provision for any kind of temporary lodging on campus for visiting faculty or conference-goers? That would be a logical accessory service for a university campus. The possibility of such a land use was discussed with the Department of Planning and Permitting (DPP) and at least preliminarily, DPP believes an extended-stay facility is possible within the campus area and permitted with a Plan Review Use approval. As noted by OEQC and UHWO, this use is necessary in this area as the closest resort-zoned area is Ko Olina, and the type of lodging facilities available in Ko Olina are more expensive than the budget usually allowed visiting professors, University conference/seminar attendees, vendors and other visitors to the campuses of the University of Hawai'i.

HECO Substation. A 1-acre HECO substation site is located at the northern boundary of the property, adjacent to Farrington Highway and Student Housing/ Mixed Use or Campus Expansion Parcel A.

Mixed Use. Located at the northeastern boundary of the property, University Village is comprised of two parcels – the 15.1-acre Mixed Use parcel and the adjacent 38-acre Student Housing/ Mixed Use or Campus Expansion Parcel A (previously described). These two parcels comprise a total of 53.1 acres and will provide approximately 797 residential units.

University Village will serve as the University's town center and will be the transition between the campus and the community. University Village will have a mix of land uses that are closely related or cater to the University and its diverse student population. Retail establishments, such as bookstores, copy centers, coffee shops, and specialty food item stores, along with small start-up offices, are envisioned for this development.



Figure 2.2 Campus Plan University of Hawai'i West O'ahu







UNIVERSITY OF HAWAI'I WEST O'AHU FINAL DRAFT ENVIRONMENTAL IMPACT STATEMENT

University Village will be located in close proximity to the Central Plaza and the "heart" of the UH West O'ahu campus, to encourage interaction between the campus and the University Village community. University Village will also promote multi-modal transportation with tree-lined, pedestrian-friendly roadways, and biking and jogging paths. Regional transit access to the University Village could be provided via a City and County-proposed elevated rail transit node³ to be located either in or near the University Village (in the vicinity of the North-South Road and Farrington Highway intersection) and through municipal bus service. From the proposed transit node, the University will be within easy walking distance for students, residents, and visitors.

The initial phases of development of the University Village include the 15.1-acre parcel, which will provide approximately 151 residential units and approximately 164,000 gross square feet of commercial space, with housing proposed for the upper floors of the mixed-use structures. Depending on future University program requirements and market demand for its expansion, the 15.1-acre Mixed Use parcel will extend into the 38-acre Student Housing/Mixed Use or Campus Expansion parcel. The 38-acre parcel could accommodate campus expansion and/or a mix of land uses, including commercial, office, and residential uses as well as student housing. The parcel is currently planned to accommodate approximately 331,000 gross square feet of commercial space and approximately 646 residential units (including student housing).

Work Force/Affordable Housing. Located to the west of the campus and adjacent to Farrington Highway, this 22.2 acre-parcel is envisioned as a multi-family residential development providing approximately 355 units, a large portion of which will be work force/affordable housing units, at a density of 16 units per acre. To partially address the City's affordable housing requirements, this parcel will provide affordable housing for qualified University faculty and staff as well as for the general public. The remainder of the affordable housing units required for the development will be distributed throughout the residential parcels in the Private Development Lands.

Campus Expansion/Multi-Family Housing/Mixed Use. Depending on future University program requirements and market demands, a 22.2 acre parcel located to the west of the campus and adjacent to Farrington Highway has been earmarked to accommodate campus expansion, and/or multi-family housing, and/or mixed use. The parcel is currently planned to accommodate approximately 355 units, at a density of 16 units per acre.

Mixed Use Parcel C. A 10.2-acre mixed-use parcel, located at the intersection of Road F and Farrington Highway, will cater to the needs of the residential community and surrounding neighborhoods. The parcel will provide approximately 102 residential units (at a density of 10 units per acre) and approximately 111,000 gross square feet of commercial space. Commercial uses within this parcel could include such land uses as supermarkets, drugstores,

³ This was indicated in the City's *Ewa Development Plan* (August 1997) and via discussions with the City Department of Transportation Services.

retail establishments, specialty food item stores, general offices, medical facilities, restaurants, personal services, along with multi-family residential uses catering to residents with a range of incomes.

Farrington Highway Expansion. Farrington Highway is a major arterial roadway that provides regional and sub-regional mobility. The highway runs in an east-west direction adjacent to the UH West O'ahu property. During the public review period, the Department of Planning and Permitting, Traffic Review Branch wrote that the Department of Design and Construction is currently working on the roadway cross-section for the widening of Farrington Highway fronting the Campus property. Approximately 4.4 acres within the northern portion of the property will be used for improvements to Farrington Highway.

Roads A, B, C, and D. A hierarchical network of roadways and a pedestrian and bikeway system will be provided within the property. Roadways within the UH West O'ahu Lands comprise approximately 7.3 acres. Road A will provide access to the campus from Farrington Highway. Road B will provide access to the campus from the North-South Road. Road C will also provide access to the campus from an internal road off of Farrington Highway (Road F). Road D will connect the UH West O'ahu Lands with the Private Development Lands. These roadways are further described in Section 4.10.1.

2.4.3 Private Development Lands

The Private Development Lands are envisioned as a sustainable community, integrating commercial, residential, and public uses. Comprising 286.5 acres, the community will include residential parcels, mixed-use parcels, parks, an elementary school, a detention basin, and a HECO substation site. (The exact acreage allocated to the Private Developer shown in this EIS is approximate and will not be known until the agreement with the private development partner is finalized. In any case, adequate land area will be set aside for a campus than can eventually accommodate up to 7,600 students in the future.) The vision for the privately-developed lands is to:

- Enhance community identity,
- Create a mix of land uses,
- Provide housing opportunities and choices,
- Foster walkable neighborhoods,
- Develop an energy and resource efficient community, and
- Provide a variety of transportation choices.

High Density Residential Parcels A and F. High-density, multi-family residences will be developed along the eastern boundary of the property, between the campus and North-South Road. High-density buildings will be 6-plex and 3 stories in height. Units will be clustered around common green areas, which will provide open space, recreation, and visual relief. Each building will be alley-loaded for vehicles and will front the main landscaped thoroughfare. Parcels A and F will provide a total of approximately 925 residential units at a density of 16 units per acre.

Medium Density Residential Parcels B and G. Medium-density, multi-family residences will be developed south of the campus, along Farrington Highway and North-South Road. Parcels B and G will include a mixture of 4- and 6-plex units at a density of 12 units per acre. These units will be organized around common green areas, similar to the high-density areas. Approximately 489 residential units will be provided within these parcels.

Medium-Low Density Residential Parcel E. Medium-low density residences will consist of duplex units with detached carports at a density of 10 units per acre. Duplexes will be clustered around small parks or green areas to reinforce a sense of community and provide visual relief. Vehicles will be accessed by alleys to emphasize the pedestrian-oriented community vision of these neighborhoods. Approximately 561 units will be provided in Parcel E.

Low Density Residential Parcels C and D. Low-density, single-family residences will be developed along the southwestern boundary of the property, adjacent to the elementary school and the proposed DHHL residential subdivision. Parcels C and D will include bungalow-styled units at a density of 6 units per acre. These primarily alley-loaded units will provide entry porches and lanais. Clusters of homes will be organized around passive parks or green areas. Landscaped streets and wide sidewalks shall connect these green areas, creating a walkable integrated neighborhood. Parcels C and D will provide 365 residential units.

Mixed Use Parcels A and B. Mixed Use Parcel A (10.5 acres) and Mixed Use Parcel B (11.2 acres) are located at the intersection of Road F and North-South Road. Like the University Village, these mixed-use parcels will include a mix of land uses, but will focus on catering more to the needs of the residential community and surrounding neighborhoods. The parcels will contain approximately 236,000 gross square feet of commercial space comprised of land uses such as supermarkets, drugstores, retail establishments, specialty food item stores, general offices, medical facilities, restaurants, personal services; along with approximately 217 multi-family residential units catering to residents with a range of incomes (possibly including affordable housing).

Elementary School. A 12-acre elementary school site will be located adjacent to low-density residential parcels and near the DHHL residential subdivision. The school will help to address the demand for public educational facilities in the region and will be set behind the commercial area to buffer it from North-South Road, while making it directly accessible to the community. The school is within walking or biking distance of residences and accessible via an internal pedestrian path. During the public review period, the Department of Transportation Services (DTS) wrote: *"The school should be situated where potential pedestrian conflicts with vehicles will be minimized. We would prefer pathways and/or sidewalks that do not traverse major roadways. Planning for school access should be done early so that an access plan that is agreeable to all parties can be formulated." The Draft EIS did not mention/emphasize that the siting of the elementary school was driven by two*

UNIVERSITY OF HAWAI'I WEST O'AHU Final Draft Environmental Impact Statement

important factors: 1) it was intended that the elementary school site was sited as close as possible to the makai boundary of the UHWO site in order to better serve the adjoining DHHL East Kapolei 1 residential development immediately makai; 2) siting of the school at the makai boundary is constrained by an air installation compatible use zone (AICUZ) emanating from the former Barbers Point Naval Air Station (NAS). While there are efforts to remove the AICUZ (and efforts by some parties to retain the AICUZ), until the issue of the AICUZ is resolved, the siting of the school is not final. UHWO has coordinated with DOE on the siting of the elementary school as currently shown, and assumes that DOE was comfortable with the location of the elementary school in terms of the potential for pedestrian and vehicular conflicts. UHWO has met with City and County of Honolulu agencies on a number of occasions, including at least one with DTS present, and to date, there had been no concerns raised about the location of the elementary school site. It is possible that a "Lab School" may be developed on the campus lands in order to address DOE requirements.

Detention Basin. An 11.2-acre drainage detention basin will be located at the southern boundary of the property.

HECO Substation. A 1-acre potential HECO substation site is located at the southern portion of the property, adjacent to the detention basin and Mixed Use Parcel B.

Farrington Highway Expansion. During the public review period, the Department of Planning and Permitting, Traffic Review Branch wrote that The Department of Design and Construction is currently working on the roadway cross-section for the widening of Farrington Highway fronting the Campus property. Within the Private Development Lands, 1.1 acres will be used for improvements to Farrington Highway.

Roads C, D, E, F, and G. Roads within the Private Development Lands total 24 acres. Road C will also provide access to the campus from an internal road off of Farrington Highway (Road F). Road D will connect the Private Development Lands with the UH West O'ahu Lands. Road E will provide another access to the campus from North-South Road. Road F will connect Farrington Highway and North-South Road through the Private Development Lands. Road G will provide access to the Private Development Lands from the DHHL residential subdivision to the south. These roadways are further described in Section 4.10.1. Per the State Department of Transportation's comments during the public review period, the developer(s) of the Private Development Lands will be informed that they will need to coordinate with DOT's Highways Division regarding the responsibility for the provision of any required roadway improvements to mitigate impacts on the State roadways in the area.

2.5 INFRASTRUCTURE IMPROVEMENTS

In support of the UH West O'ahu development, infrastructure facilities to be expanded or improved include water transmission lines, wastewater collection lines, drainage systems, access and circulation roadways, and electrical/communication systems. All infrastructure improvements will be designed and sized to accommodate the campus and community. Construction will begin with development of infrastructure after applicable grading permits have been issued.

2.5.1 On-site Infrastructure

Major on-site infrastructure improvements required for the project include facilities for water transmission, wastewater collection, traffic circulation, drainage, and electrical and communication systems.

2.5.1.1 Roadways and Traffic

The only existing public roadway providing access to property is Farrington Highway. During the public review period, the Department of Planning and Permitting, Traffic Review Branch wrote that the Department of Design and Construction is currently working on the roadway cross-section for the widening of Farrington Highway fronting the Campus property. Approximately 5.5 acres of land within the property has been set aside for improvements to Farrington Highway. A loop road will provide the major access to various areas of the campus. A loop road encircling the major University buildings will provide access throughout the campus. The campus loop road will have a right-of-way width of 56 feet. Seven other internal roadways are planned within the property, with right-of-way widths varying from 66 feet to 104 feet. Roads within the UH West Oahu Lands will either be controlled by the University or dedicated to the City. Roads within the Private Development Lands will be dedicated to the City as determined during the entitlement phase of the project. It is proposed that all roads (and water, sewer and drainage systems) within the project site will be dedicated to the City and County of Honolulu, except Road "A" and the portion of Road "B" westward of the intersection with Road "D". During the public review period, the City and County of Honolulu Department of Facility Maintenance wrote: "Roads, drainage facilities and their related infrastructure should be owned and maintained by the State of Hawaii for this State facility, similarly to that at the Manoa Campus. The City and County of Honolulu will not accept the dedication or maintenance of this facility." This will be identified as an unresolved issue since the Department of Planning and Permitting is specifying roads within the 500-acre project to provide regional connectivity, and the noncampus lands will include residential and commercial uses that will generate property tax revenue to the City and County of Honolulu, it is felt that the non-campus roads should be allowed to be dedicated to the County.

2.5.1.2 Water Supply and Distribution

Currently, there is no water system to serve the approximately 500-acre property, which

spans two Board of Water Supply (BWS) service pressure zones at elevations of 215 feet and 440 feet. On-site potable water systems will connect to transmission mains in the North-South Road (currently under construction) and Farrington Highway. The property will connect to the dual (non-potable and potable) supply system planned for the 215-foot system.

2.5.1.3 Wastewater Collection and Transmission

Currently, there is no sewer service to the property. A portion of the major regional sewer trunk system will run through the property. The sewer trunk line will enter the property from Farrington Highway and exit the property to North-South Road. Smaller sewers will branch off the major trunk line to service the property. Flows from the property and properties to the north will be conveyed to the trunk line in North-South Road.

2.5.1.4 Drainage Facilities

Currently, there are no drainage improvements on the property. Runoff is carried by Kalo'i Gulch and Hunehune Gulch, which converge on the property, or sheet flows over land. The proposed drainage system for the campus will consist of grated inlets in parking lots and landscaped areas, curb inlets along roadways, underground pipe/box drains, and a 10-acre detention/water quality basin with a flow control structure. The proposed drainage system for the Private Development Lands will include an 11.2-acre detention basin at the southern boundary of the property. Storm water detained in the basin will be released at a controlled rate after the storm and will be directed to the regional East Kapolei drainage system being built as part of the construction of the North-South Road.

2.5.2 Off-site Infrastructure

Primary off-site infrastructure required to accommodate the development includes water storage, transmission, and booster pump facilities; wastewater collection facilities; and drainage facilities. Potable water facilities are proposed to be located mauka of the approximately 500-acre property, at the 215- and 440-foot elevations (see Figure 2.3). These sites have been previously studied under formerly proposed projects for the State of Hawai'i Housing and Community Development Corporation of Hawaii (HCDCH) and the University of Hawai'i. The HCDCH studied a 1,300-acre area (which included the proposed 215-foot elevation reservoirs site) in the *East Kapolei Master Plan Final Environmental Impact Statement* (1998) (herein referred to as "HCDCH East Kapolei EIS"). The University conducted several studies on an approximately 1,000-acre area, at the formerly proposed UH West O'ahu campus site mauka of the H-1 Freeway.

Wastewater facilities to accommodate the development are proposed to be located within the North-South Road right-of-way (east of the property) and to continue to Renton Road (south of the property). The North-South Road has been previously studied by the State Department of Transportation (DOT) and the City Department of Transportation Services (DTS) in the *North-South Road and Kapolei Parkway Final Environmental Assessment* (2004), and a Finding of No Significant Impact (FONSI) was issued by the Federal Highway Administration in January



SOURCE: ENGINEERING CONCEPTS, INC.

Figure 2.3 Proposed Off-Site Infrastructure

University of Hawai'i West O'ahu



2005. The portion of the proposed wastewater collection line between the proposed Kapolei Parkway Extension and Renton Road has been previously studied in the HCDCH East Kapolei EIS.

Flood waters from the Kalo'i Gulch watershed are planned to be diverted to the 300-footwide utility, drainage, and access corridor along the proposed North-South Road. The North-South Road has been previously studied in the *North-South Road and Kapolei Parkway Final Environmental Assessment/FONSI*.

2.5.2.1 Roadways

The <u>City and County of Honolulu Department of Design and Construction (DTSDC)</u> plans to widen Farrington Highway to two lanes in each direction with a right-of-way width of 100 feet between the Kapolei Golf Course and Fort Weaver Road, taking land along the UH West O'ahu property frontage. Access to the property is planned via two intersections at Farrington Highway and three DOT-approved intersections at the proposed North-South Road. Figure 2.4 shows existing and proposed roadways in the project area.

2.5.2.2 Water Supply and Distribution

To accommodate the planned developments in East Kapolei, the existing 215-foot Kapolei potable water system will be upgraded and a new 440-foot potable water system will be constructed. For the 215-foot potable water system, a 4.0-million gallon (MG) reservoir and a transmission main in North-South Road will be completed with ongoing developments in the area. For the 440-foot potable water system, one 5.0-MG reservoir or two 2.5-MG reservoirs and a transmission main in North-South Road will be constructed concurrently with the development of the UH West O'ahu.

<u>A dual potable and non-potable water system is planned for the 215-foot elevation service</u> <u>zone.</u> The 215-foot non-potable water system is expected to be completed by the Board of Water Supply (BWS) before completion of the UH West O'ahu development.

2.5.2.3 Wastewater Collection and Transmission

A proposed 24-inch trunk line will convey flows from the property to the sanitary system proposed within the North-South Road right-of-way. Sewer trunk lines within the North-South Road are planned to be as large as 42 inches in diameter and will convey flows from all of East Kapolei to the proposed 42-inch Kapolei Interceptor Sewer, which will connect to the Honouliuli Wastewater Treatment Plant via Renton Road.

2.5.2.4 Drainage Facilities

Runoff will be conveyed to a drainage channel within the 300-foot-wide utility, drainage, and access corridor along the east side of the proposed North-South Road (see Figure 2.5). The channel will divert flow in Kalo'i Gulch away from the UH West O'ahu property and into a

detention basin at the downstream end of the channel above the Ewa Villages Golf Course.

2.6 SUSTAINABILITY

Sustainability guidelines have been established for the UH West O'ahu. The guidelines strive to achieve a Leadership in Energy and Environmental Design (LEED) NC silver standard for campus development and community performance standards based on the LEED NC v2.2 rating system. The Private Development Lands will also strive to achieve the applicable design criteria in the UH West O'ahu LRDP and the recommended community performance standards.

The sustainability guidelines for the UH West O'ahu are structured around the following focus area and goals.

- Site Planning and Land Use
 - Harmonize with existing site elements through appropriate use of land to create a compact, sustainable and vibrant campus.
- Landscaping
 - Create habitat that relies on native species, enhances the campus experience and celebrates the region's agrarian roots.
- Water Management
 - *Reduce overall potable water consumption and control quantity and quality of storm water.*
- Energy Use
 - Utilize passive design strategies, increase design for energy efficiency, reduce energy consumption and demand, and explore energy generation from renewable resources.
- Transportation and Circulation
 - Plan access to a wide range of efficient environmentally sensitive and convenient means of transportation.
- Environmentally-appropriate Materials
 - Use environmentally appropriate materials with reduced embodied energy in the design and construction of infrastructure and buildings.
- Waste Management
 - Appropriately reduce, reuse and recycle materials, to minimize generation of solid waste and achieve diversion from landfills.
- Social Benefits and Community Development
 - Create a vibrant and sustainable community for the students, the faculty, the staff, the visitors and the neighborhood. Design for all disabilities and promote sustainable values in art and education.
- Operations and Maintenance
 - Incorporate into the operations, maintenance and administrative policies of the campus.



Source: Parsons Brinckerhoff Quade & Douglas





Source: North-South Road and Kapolei Parkway Project Final Environmental Assessment Figure 2.1-3 Parsons Brinckerhoff Quade & Douglas

Disclaimer: This graphic has been prepared for general planning purposes only.

Figure 2.5 North-South Road Typical Section University of Hawai'i West O'ahu



NOT TO SCALE



- Financing and Justification
 - Promote future sustainable projects on campus, and use appropriate techniques to justify investments in those endeavors.
- Third Party Certification
 - Strive for certification using the LEED[™] rating system. Achieve a standard of LEED[™]-Silver across the campus.

During the DEIS public review period, the State Department of Business, Economic Development & Tourism (DBEDT), Strategic Industries Division, wrote to call attention to: 1) State energy conservation goals; and 2) energy and resource efficiency and renewable energy and resource development.

DBEDT noted that project buildings, activities, and site grounds should be designed and/or retrofit with energy saving "considerations." DBEDT further noted that the mandate for such consideration is found in Chapter 344, HRS ("State Environmental Policy") and Chapter 226, HRS ("Hawaii State Planning Act"). In particular, DBEDT noted HRS 226 18(c) (4) which includes a State objective of promoting all cost-effective energy conservation through adoption of energy-efficient practices and technologies.

DBEDT noted two important directives affecting State agencies. The first is Administrative Directive No. 06-01, which is also reflected in Act 96, 2006 Session Laws of Hawaii (SLH), which encourages State agencies and programs to increase their leadership commitment to implement innovative and resource efficient operations and management and to design and construct buildings to meet and receive certification for U.S. Green Building Council's Leadership in Energy and Environmental Design (LEED), among others. The other directive DBEDT noted was Act 160, 2006 SLH, relating to the State Budget, which requires that each executive department shall plan for or install energy reduction, energy savings or energyproducing efforts and technologies to lessen electrical consumption or to increase efficiencies in using electrical energy. As stated above:

Sustainability guidelines have been established for the UH West O'ahu. The guidelines strive to achieve a Leadership in Energy and Environmental Design (LEED) NC silver standard for campus development and community performance standards based on the LEED NC v2.2 rating system.

DBEDT recommended that UHWO consult the City and County of Honolulu Energy code early in the project. DBEDT also noted that HECO may also have demand-side management programs that offer rebates for installation of energy-efficient technologies.

Based on DBEDT's recommendations, UHWO's mechanical and electrical consultants, in consultation with UHWO's sustainability consultant, will be directed to review the City and County of Honolulu's Energy code early in the project and to consult with HECO on demand-side management programs that offer rebates for installation of energy-efficient technologies.

2.7 PHASING AND TIMING OF ACTION

The initial phase and the build-out phase of the UH West O'ahu development are shown in Table 2.4 and described in this section.

UNIVERSITY OF HAWAI'I WEST O'AHU Final Draft Environmental Impact Statement

Table 2.4 – Preliminary Phasing Plan							
LAND USE	INITIAL PHA	SE (PHASE 1)	BUILD-O	UT PHASE			
		Res. Units		Res. Units			
	Acreage	(approx.)	Acreage	(approx.)			
UH WEST O'AHU LANDS							
UH West O'ahu Campus							
7,600 Student Campus (including 5-acre							
detention basin in Phase 1 and 10-acre	22.0	-	103.5	-			
detention basin in Build-out Phase)							
Student Housing/ Mixed Use or Campus			20.0	C A C			
Expansion Parcel A	-	-	50.0	040			
Student Housing or Campus Expansion			12.1	230			
Parcel B	-	-	12.1	230			
HECO Substation	1.0	-	1.0	-			
Mixed Use and Affordable Housing							
Mixed Use (Retail, Office, Residential)	6.0	60	15.1	151			
Work Force/Affordable Housing (Multi							
Family) Campus Expansion/Multi-Family	-	-	22.2	355			
<u>Housing/Mixed Use</u>							
Mixed Use Parcel C	10.2	102	10.2	102			
Roads							
Farrington Highway Expansion	4.4	-	4.4	-			
Roads A, B, C, and D	6.3	-	7.3	-			
Total for UH West O'ahu Lands	49.9	162	213.8	1,484			
PRIVATE DEVELOPMENT LANDS							
Residential							
Residential Parcel A (High Density, Multi-			22.1	530			
Family)	-	-	55.1	550			
Residential Parcel B (Medium Density,	_	_	20.1	241			
Multi Family)	_	_	20.1	271			
Residential Parcel C (Low Density, Single	_	_	30.5	183			
Family)			30.5	105			
Residential Parcel D (Low Density, Single	13.8	83	30.3	182			
Family)	1010	00	00.0	102			
Residential Parcel E (Medium-Low Density,	8.5	85	56.1	561			
Multi-Family)	0.0	00	0011				
Residential Parcel F (High Density, Multi	9.8	157	24.7	395			
Family)							
Residential Parcel G (Medium Density,	10.8	130	20.7	248			
Multi Family)							
Commercial	Γ	т г	10 5	405			
Mixed Use Parcel A	-	-	10.5	105			
Mixed Use Parcel B	-	-	11.2	112			
	Γ	<u>ј</u>	12.0	[
Elementary School	-	-	11.0	-			
	11.2	-	11.2	-			
HECO Substation	-	-	1.0	-			
Farrington Highway Expansion	1.1	-	1.1	-			
Koads C, D, E, F, and G	5.4	-	24.0	-			
I otal for Private Development Lands	60.6	454	286.5	2,557			
TOTAL	110.5	616	500.3	4,041			

Initial Phase. Phase 1 development of the UH West O'ahu Lands includes construction of a 1,520-student campus (including a 5-acre detention basin); a portion of University Village; a HECO substation; Roads A and B, and a portion of Road D; and the expansion of Farrington Highway (see Figure 2.6). Phase 1 development of the Private Development Lands includes construction of portions of Residential Parcels D, E, F, and G; an 11.2-acre detention basin; portions of Roads C, D, and E; and the expansion of improvements at intersections with Farrington Highway. Approximately 110.5 acres of the 500-acre property will be developed in Phase 1, providing 616 residential units. <u>The current estimated date for opening of the campus is Fall 2009.</u>

Build-Out Phase. Ultimately, the UH West O'ahu Lands will include a 7,600-student campus (including a 10-acre detention basin); University Village (student housing/mixed use or campus expansion); work force/affordable multi-family housing; a HECO substation; and roads. The Private Development Lands will include single- and multi-family residential uses; mixed-use areas; a detention basin; an elementary school(s); a HECO substation; and roads. The current estimated year for completion of the project is 2015.

2.8 ORDER OF MAGNITUDE COSTS

The current estimated cost for Phase 1, including earthwork; landscaping; internal roadways; sewer, potable water, non-potable water, and drainage facilities; and improvements to Farrington Highway for access is approximately \$150 million.

Due to limited State resources, a private developer was sought out to alleviate the financial burden on the University. Final agreements for private construction have not been finalized, however, the University and the private developer are taking the necessary steps to ensure that the projected phasing of the project remain on schedule.



Figure 2.6 Conceptural Land Use Plan Phase 1 Development



3.0

Assessment of the Natural Environment, Potential Impacts, and Mitigation Measures

3.0 ASSESSMENT OF THE NATURAL ENVIRONMENT, POTENTIAL IMPACTS, AND MITIGATION MEASURES

This chapter describes the natural environment associated with the UH West O'ahu property and surrounding area. Potential impacts that may result from the project and mitigation measures to address such impacts have been identified for areas such as soils, drainage, flora, fauna, groundwater, air quality, noise, visual resources, archaeological resources, etc.

3.1 CLIMATE

Existing Conditions

Based on more than 50 years of data collected at the 'Ewa Plantation, average annual daily minimum and maximum temperatures in the project area are 65 and 84 degrees Fahrenheit, respectively. Extreme minimum and maximum temperatures were recorded at 47 and 93 degrees Fahrenheit.

The most representative long-term wind data available for the UH West O'ahu property was collected at Kalaeloa (formerly Naval Air Station Barbers Point), located to the southwest. Wind frequency data from Kalaeloa reported that the annual prevailing wind direction for this area of O'ahu is east-northeast about 40 percent of the time, at approximately 10 knots (12 miles per hour). Winds from the south are associated with Kona storms and are infrequent, occurring only a few days during the year and mostly in the winter. The 'Ewa region experiences light rainfall (mean annual rainfall of about 20 inches per year), most of which occurs between November and April.

Anticipated Impacts and Mitigation Measures

With build-out of the UH West O'ahu campus and community, there may be some localized increases in temperature as a result of paved surfaces, roofs, and reduced vegetation. However, extensive landscaping and necessary irrigation will mitigate potential temperature increases. Shade trees will be incorporated into the landscaping of the property.

Sustainability guidelines have been established for the development, and the UH West O'ahu Lands will strive to incorporate sustainable design and building concepts. In addition, all new construction and retrofit projects by the University will strive meet or exceed the targets set by the Leadership in Energy and Environmental Design (LEED) NCv2.2 rating system. The Private Development Lands will strive to achieve the recommended LEED community performance standards.

3.2 **TOPOGRAPHY**

Existing Conditions

The island of O'ahu originated as a volcano from two volcanoes: Wai'anae and Ko'olau, and is characterized by underlying basaltic flows. The 'Ewa Plain is an emerged coral reef formed during the Pleistocene Period, when the ocean level was at a higher elevation. For the most part, the 'Ewa Plain is flat with a few isolated bluffs eroded by Honouliuli Stream. The plain is underlain by material that has been modified over the millennia and is hard but extremely permeable. This hard, permeable caprock of sedimentary deposits forms a wedge that retards the seaward movement of fresh groundwater from the inland basaltic aquifer. At higher elevations, the ground surface is made of alluvium and sedimentary deposits that washed down over the millennia.

The UH West O'ahu property is located in the middle to lower portion of the Kalo'i Gulch watershed, on the southern slopes of the Wai'anae Mountain Range. The elevation at the lower boundary of the property is 80 feet mean sea level (msl) and rises to about 155 feet msl at the upper boundary, over a 6,500-foot distance. The property is relatively flat, with an average slope of about one to two percent.

Since much of the property has been historically utilized for sugarcane cultivation (and portions of land are currently used for vegetable and fruit cultivation), the property has been extensively modified with cane haul roadways, a furrow irrigation system, and other appurtenant agricultural structures. There is an existing potential for erosion throughout the property.

Anticipated Impacts and Mitigation Measures

The proposed 7,600-student campus will generally follow the existing terrain. Existing slopes will accommodate buildings, walkways, and parking lots without requiring any significant reshaping of the land. The proposed grading concept is intended to conform to the surrounding grades of the East Kapolei development and the North-South Road, as well as the preliminary *East Kapolei Drainage Master Plan*. It is anticipated that due to the flat terrain, the campus will require fill to provide adequate slopes for drainage and to elevate the campus above the Hunehune Gulch and Kalo'i Gulch storm water elevations.

During Phase 1 of the development, grading for the 1,520-student campus will consist of shallow embankments for the parking lots, immediate campus, and primary access roads, Road A and B. Future campus expansion and development of the surrounding areas will require additional amounts of shallow embankment.

All grading operations will be conducted in full compliance with dust and erosion control requirements of the City's Grading Ordinance. All construction activities will comply with the provisions of Section 11-60.1-33, HAR, on Fugitive Dust. A grading permit is required to

UNIVERSITY OF HAWAI'I WEST O'AHU Final Draft Environmental Impact Statement

modify the topography, and a National Pollutant Discharge Elimination System (NPDES) permit is required before construction begins to address non-point source discharges.

3.3 DRAINAGE

Existing Conditions

The UH West O'ahu property is located in the mid-level portion of the Kalo'i Gulch and Hunehune Gulch watersheds. The property is bisected by Kalo'i Gulch and Hunehune Gulch, which are characterized as dry ditches within the flat terrain of the property. At times, significant precipitation causes direct runoff, and since there are no existing drainage improvements within the property, runoff flows overland via the two gulches toward the 'Ewa Villages Golf Course. The existing gulches are not adequate to handle large storms and excess runoff flows overland.

Kalo'i Gulch and Hunehune Gulch converge within the property. Based on City and County standards, runoff calculations indicate a peak flow of over 1,200 cubic feet per second (cfs) from the Hunehune Gulch watershed, which crosses Farrington Highway and enters the property. A peak flow of over 5,900 cfs enters the East Kapolei lands from the Kalo'i Gulch watershed.

In 2000, an emergency overflow channel was constructed to allow overflows from a very large, infrequent storm event to overtop the detention/retention basins of the Coral Creek Golf Course (located to the southeast of the property) and proceed toward the Pacific Ocean in the vicinity of Oneula Beach Park. However, a site inspection revealed that there are no physical indications that storm flows have ever overtopped the Coral Creek drainage system and discharged to the Pacific Ocean, as storm water management systems associated with developments within the lower reaches of the Kalo'i Gulch watershed have been engineered such that the upper reaches of Kalo'i Gulch are effectively isolated from the Pacific Ocean. The Department of the Army (DA) has determined that the upper reaches of the U.S. (see Appendix A).

The Federal Emergency Management Agency classifies flood hazard zones on the Flood Insurance Rate Map (FIRM) as part of the Flood Insurance Program for the City. The property is located in Zone D, which includes areas in which flood hazards are undetermined (see Figure 3.1). This was confirmed by the State Department of Land and Natural Resources (DLNR) Engineering Division, which also stated in its pre-consultation comment letter that if future studies determine that the property is within the flood zone, the project must comply with the rules and regulations of the National Flood Insurance Program (see Chapter 8.0).

Anticipated Impacts and Mitigation Measures

Existing runoff volume will increase as a result of the addition of impermeable surfaces (i.e., roadways, sidewalks, parking areas, buildings, and courtyards). However, proposed drainage improvements, further discussed in Section 4.10.4, are intended to control flooding and provide adequate storm water disposal. A 10-acre detention basin is proposed within the campus and an 11.2-acre detention basin is proposed at the southern boundary of the property. Storm water runoff from the property will be conveyed to the East Kapolei regional drainage system. A drainage corridor is planned within the 300-foot-wide utility, drainage, and access corridor along the proposed North-South Road (see Figure 2.5). Flood control detention areas will be required until a downstream drainage connection to the Pacific Ocean is established at Oneula Beach Park.

As discussed above, a Department of the Army permit pursuant to Section 404 of the Clean Water Act is not required for the proposed fill of Kalo'i Gulch.

3.4 SOILS

Existing Conditions

Three soil suitability studies have been prepared for Hawai'i. The principal focus of these studies is to describe the physical attributes and relative productivity of different land types for agricultural production. The three studies include the *Detailed Land Classification* by the University of Hawai'i Land Study Bureau, a soil survey by the U.S. Department of Agriculture Natural Resources Conservation Service, and the Agricultural Lands of Importance to the State of Hawai'i (ALISH) system by the State Department of Agriculture.

Detailed Land Classification. The *Detailed Land Classification* (1963) was developed for each island by the University of Hawai'i Land Study Bureau. This series of reports was intended to develop a land inventory and productivity evaluation based on statewide standards of crop yields and levels of management. A five-class productivity rating is applied using the letters A, B, C, D, and E, with A representing the soils with the highest productivity and E the lowest. Soils on the UH West O'ahu property are rated mostly A and B (see Figure 3.2). This reflects the property's past and present agricultural use under irrigated conditions. A small northern portion of the property is rated E.



Legend

Zone X: Areas Determined to be Outside 500-Year Floodplain

Zone D: An Area of Undetermined but Possible Flood Hazards

Zone AE: An Area Inundated by 1% Annual Chance Flooding For Which Base Flood Elevations Have Been Determined

Zone AEF: Floodway Area in Zone AE

Figure 3.1 Flood Insurance Rate Map University of Hawai'i West O'ahu





Legend







Natural Resources Conservation Service Soil Survey. The Natural Resources Conservation Service identified six general soil types and a water reservoir within the property (see Figure 3.3). The soil types include:

- Honouliuli Clay, 0 to 2 percent slopes (HxA);
- Honouliuli Clay, 2 to 6 percent slopes (HxB);
- Waialua Silty Clay, 0 to 3 percent slopes (WkA);
- Ewa Silty Clay Loam, 3 to 6 percent slopes (EaB);
- Ewa Stony Silty Clay, 6 to 12 percent slopes (EwC); and
- Waipahu Silty Clay, 0 to 2 percent (WzA).

Honouliuli Clay, 0 to 2 percent slopes (HxA). The majority of soils within the UH West O'ahu property are of this type. Permeability for this soil is moderately slow, runoff is slow, and the erosion hazard is no more than slight.

Honouliuli Clay, 2 to 6 percent slopes (HxB). This soil is found throughout the central and northern portions of the property. Runoff for this soil is slow and the erosion hazard is slight.

Waialua Silty Clay, 0 to 3 percent slopes (WkA). This soil is found throughout the property. Permeability for this soil is moderate, runoff is slow, and the erosion hazard is no more than slight.

Ewa Silty Clay Loam, 3 to 6 percent slopes (EaB). This soil is found along the western boundary of the property. Permeability for this soil is moderate, runoff is slow, and the erosion hazard is slight.

Ewa Stony Silty Clay, 6 to 12 percent slopes (EwC). A small amount of this soil is found at the western boundary of the property. Runoff for this soil is slow to medium and the erosion hazard is slight to moderate.

Waipahu Silty Clay, 0 to 2 percent (WzA). This soil is found in the northern portion of the property. Permeability for this soil is moderately slow. Runoff is slow or very slow, and the erosion hazard is none to slight.

Agricultural Lands of Importance to the State of Hawai'i. The Agricultural Lands of Importance to the State of Hawai'i (ALISH) (1977) system maps and identifies three broad classes of agricultural land based, in part, on the criteria established by the Natural Resources Conservation Service. Prime Agricultural Land is defined as "land best suited for the production of food, feed, forage, and fiber crops." This class of land has the soil quality, growing season, and moisture supply needed to economically produce sustained high yields of crops when treated and managed according to modern farming methods. Prime Agricultural Land gives the highest yields with the lowest inputs of energy or money, and with the least damage to the environment. The two other classes of the ALISH system are Unique Agricultural Land and Other Important Agricultural Land, identifying successively less

productive soils. The entire UH West O'ahu property is designated Prime Agricultural Land (see Figure 3.4).

Anticipated Impacts and Mitigation Measures

Due to the relatively flat terrain, grading for the campus will not be conspicuous. The proposed 7,600-student campus and surrounding uses will generally follow the existing terrain. Existing slopes will accommodate buildings, walkways, and parking lots without requiring any significant re-shaping of the land. However, the campus will require fill to provide adequate slopes for drainage and to elevate the campus above the Hunehune Gulch and Kalo'i Gulch storm water elevations.

All grading operations will be conducted in full compliance with dust and erosion control requirements of the City's Grading Ordinance. All construction activities will comply with the provisions of Section 11-60.1-33, HAR, on Fugitive Dust. A grading permit is required to modify the topography, and a National Pollutant Discharge Elimination System (NPDES) permit is required before construction begins to address non-point source discharges. Prior to the issuance of a grading permit by the City DPP, an erosion control plan and best management practices required for the NPDES permit will be submitted to describe the implementation of appropriate erosion control measures.

A watering program will be implemented to minimize soil loss through fugitive dust emissions during construction. Other dust and erosion control measures include cleaning job-site construction equipment and establishing ground cover as quickly as possible after grading. During the public review period, the Department of Health Clean Air Branch recommended that a dust control management be prepared. The DOH CAB also provided additional examples of measures that can be implemented during construction:

- where possible, for areas involving mixed land uses, buffer zones be established to alleviate potential dust nuisance problems;
- planning the different phases of construction, focusing on minimizing the amount of dust-generating materials and activities, centralizing on-site vehicular traffic routes, and locating potential dust-generating equipment in areas of the least impact;
- providing an adequate water resource and watering program at the site prior to startup of construction activities;
- landscaping and providing rapid covering of bare areas, including slopes, starting from the initial grading phase;
- minimizing dust from shoulders and access roads;
- providing adequate dust control measures during weekends, after hours, and prior to daily start-up of construction activities; and
- controlling dust from debris being hauled away from the project site.



Legend



Water Reservoir

Honouliuli Clay, 0-2 Percent Slopes (HxA) Waipahu Silty Clay, 0-2 Percent Slopes (WzA) Honouliuli Clay, 2-6 Percent Slopes (HxB) Waialua Silty Clay, 0-3 Percent Slopes (WkA) Ewa Silty Clay Loam, 3-6 Percent Slopes (EaB) Ewa Stony Silty Clay, 6-12 Percent Slopes (EwC)




Legend



Unclassified Agricultural Land

Prime Agricultural Land

Other Agricultural Land

Figure 3.4 Agricultural Land of Importance to the State of Hawai'i (ALISH)

University of Hawai'i West O'ahu



Mitigation measures generally associated with best <u>(water borne soil erosion)</u> management practices include:

- Early construction of drainage control features;
- Construction of temporary sediment basins to trap silt;
- Use of temporary berms and cut-off ditches where needed; and
- Use of temporary silt fences or straw bale barriers to trap silt.

Once construction is complete, the soils on the property will be permanently lost for soilbased agriculture. However, since much of the property currently consists of exposed soil and scrub vegetation, which resulted from past and present agricultural uses, overall soil loss will likely be reduced significantly after development. Landscaping, groundcover, and drainage improvements will also mitigate soil loss after build-out.

No significant geological and/or geotechnical constraints that would preclude the proposed development are expected to be encountered. The primary geotechnical considerations for the development of this property appear to be site drainage, abandoned agricultural infrastructure, expansive soil conditions, and possible backfilled, undocumented ditches or reservoirs associated with past use of the land. Due to the relatively dry climatic conditions of the 'Ewa Plain, it is important that proper site drainage and landscaping, along with a permanent irrigation system, be developed to help control the expansive soils and reduce the potential for significant erosion of soil materials.

3.5 **GROUNDWATER RESOURCES/HYDROLOGY**

Existing Conditions

The 'Ewa region of O'ahu overlies the Southern Oahu Basal Aquifer (SOBA), a designated Sole Source Aquifer. The gently sloping topography of the 'Ewa Plain is comprised of terrestrial alluvium, which is made up of clay and mud eroded from volcanic rock and interlayered with coral limestone deposited during periods when the area was covered by the ocean. This geologic feature is commonly referred to as caprock, which is approximately 1,000 feet thick near the shoreline. Water in the caprock is too saline to be potable.

The potential for surface contamination of water under the caprock is low due to artesian conditions and the relative impermeability of the caprock. Infiltration of surface water mauka of the caprock could contaminate potable groundwater resources since the caprock is not present to function as a barrier in this area. Hydrological modeling predicts that the salinity of the caprock aquifer will increase as fresh water recharge declines with the reduction in sugarcane irrigation.

The State Department of Health (DOH) has established the Underground Injection Control (UIC) line at North Hanson Road to protect potential sources of drinking water by not allowing wastewater injection wells or cesspools mauka of the line. The UH West O'ahu property is mauka of the UIC line but will not include injection wells or cesspools.

The property is bisected by Kalo'i Gulch and Hunehune Gulch. At times, significant precipitation causes direct runoff, and since there are no existing drainage improvements within the property, runoff flows overland via the two gulches toward the 'Ewa Villages Golf Course. The existing gulches are not adequate to handle large storms and excess runoff flows overland.

Anticipated Impacts and Mitigation Measures

With the termination of irrigation for sugarcane in the project area, groundwater recharge has decreased. When a non-potable water system is developed for East Kapolei, non-potable water from the Honouliuli Water Recycling Facility will be used for the irrigation of common areas, in accordance with State DOH regulations.

3.6 NATURAL HAZARDS

Existing Conditions

Natural hazards are events such as tsunami, floods, hurricanes, earthquakes, soil slippage, and volcanic hazards. The UH West O'ahu property is located outside of the tsunami evacuation area and within Zone D, where flood hazards are undetermined but possible (see Figure 3.1). Kalo'i Gulch and Hunehune Gulch traverse the property; however, with the development of drainage improvements for the Villages of Kapolei (including a golf course, a lined drainage ditch, and injection wells), the drainage conditions impacting the property have been significantly altered.

Earthquakes in the Hawaiian Islands are associated with volcanic eruptions or tectonic movements. Volcanic hazards in the area are considered minimal due to the extinct status of former volcanoes.

The State has been affected twice since 1982 by devastating hurricanes, 'Iwa in 1982 and 'Iniki in 1992. While it is difficult to predict these natural occurrences, it is reasonable to assume that events could occur, given the past record. However, the project area is no more vulnerable to the destructive winds and torrential rains associated with hurricanes and cyclones than other parts of the island.

Anticipated Impacts and Mitigation Measures

Currently, the property has no on-site drainage facilities, and Kalo'i and Hunehune Gulches are not adequate to handle large storms and excess runoff flows overland. However, proposed drainage improvements, further discussed in Section 4.10.4, are intended to control flooding and provide adequate storm water disposal. A 5-acre detention basin near the campus and an 11.2-acre detention basin at the southern boundary of the property will be developed in Phase 1. The 5-acre campus detention basin will be expanded to 10 acres at

full build-out. The Storm water runoff from the property will be conveyed to the East Kapolei regional drainage system. A drainage corridor is planned within the 300-foot-wide utility, drainage, and access corridor along the proposed North-South Road (see Figure 2.5). A Department of the Army permit is not required to fill Kalo'i Gulch, as the upper reaches of the gulch within the project site do not have a regulated tributary connection to waters of the U.S. (see Appendix A).

Potential damage caused by hurricanes and earthquakes will be mitigated by compliance with the Uniform Building Code adopted by the City.

3.7 FLORA

Existing Conditions

The property was historically used for sugarcane cultivation, and portions of the property are currently on month-to-month revocable leases to Aloun Farms, Inc. and A.M. Enterprise, Inc. for vegetable and fruit farming (see Figure 1.4). These agricultural activities have significantly disrupted the original vegetation. Several botanical surveys on the property and adjacent lands have identified sugarcane fields, ruderal fields, and fallow fields. A survey of the 1,300-acre East Kapolei Master Plan area, which included the UH West O'ahu property, was undertaken by K.M. Nagata between mid-September and early October 1996. To verify the findings of this survey and to more accurately inventory and map the plants, Char & Associates conducted a survey in December 1996 and reported findings in January 1997. In June 2003, a botanical survey of the approximately 500-acre UH West O'ahu property was conducted by Char & Associates. This report is included in Appendix B.

K.M. Nagata Survey (1996). The 1996 field survey of the 1,300-acre East Kapolei Master Plan area discovered the endangered plant species, ko'oloa'ula or Abutilon menziesii. The survey report is included in the Habitat Conservation Plan for Abutilon menziesii at Kapolei (see Appendix C). According to the survey report, the property was classified as one of lowland shrub with a coastal fringe of kiawe trees. These communities were characterized by actively cultivated sugarcane fields, abandoned cane fields, common "weedy" introduced plants, and lowland wayside species. The vegetation of these lands is entirely secondary and is determined by the history of cultivation (or disturbance) on each individual parcel. Eight plant communities were recognized, with each community existing as a continuum with one blending into another. Within the eight plant communities identified, 99 different plant species were recorded, two indigenous ('ilima and pa'uohi'iaka), two probably indigenous ('uhaloa and hoary abutilon), and one endemic (ko'oloa'ula, or Abutilon menziesii). Except for ko'oloa'ula, all of the native species on the site were common lowland species in Hawai'i. At least 38 individuals of the ko'oloa'ula were recorded throughout the 1,300-acre survey area. The ko'oloa'ula is now a federally listed endangered species, once endemic to Lāna'i, Maui, O'ahu, and Hawai'i. It may now be extinct on Hawai'i.

Char & Associates (1997). As previously noted, in September 1996, Nagata found at least 38 ko'oloa'ula plants within the 1,300-acre East Kapolei Master Plan area. After the unusually heavy rains in November 1996, Char & Associates conducted an intensive inventory of the plants in December and flagged and inventoried a total of 88 plants occurring in three colonies. The survey report is included in the *Habitat Conservation Plan for Abutilon menziesii at Kapolei* (see Appendix C).

Char & Associates (2003). A botanical survey of the UH West O'ahu property was conducted in June 2003 by Char & Associates to: 1) prepare a general description of the vegetation on the site; 2) inventory the flora; 3) search for threatened and endangered species as well as species of concern; and 4) identify areas of potential environmental problems or concerns and propose appropriate mitigation measures (see Appendix B). Three vegetation types were recognized on the property: agricultural/farm, scrub vegetation, and gulch vegetation. The species observed in the field survey are indicative of the environmental conditions at the time of the survey. Findings of the botanical survey are summarized below.

Agricultural/Farm. Actively cultivated fields compose the agricultural farm land vegetation type, which covers the majority of the property. Short stumps of koa haole shrubs (*Leucaena leucocephala*) were scattered throughout these fields. Various crops covered large areas of the planted fields on the eastern portion of the property. A few weedy species were found growing among the crop plants. Most of the weedy plants, however, occur along the uncultivated areas that border the fields. They receive runoff from the cultivated fields and tend to be lush and green. The native 'ilima (*Sida fallax*) is locally common in some places, and one new species not recorded from the island of O'ahu, Russian thistle or tumbleweed (*Salsola tragus*), was collected and deposited at the Bishop Museum herbarium.

Gulch Vegetation. Kalo'i Gulch and Hunehune Gulch cross the property. The vegetation within the gulches (sides and bottom) is characterized by dense, robust clumps of Guinea grass, five- to ten-feet tall. The dense Guinea grass cover tended to exclude other species, but a few patches of California grass (*Brachiaria mutica*), sourbush, castor bean (*Ricinus communis*), wild bittermelon (*Momordica charantia*), comby hyptis (*Hyptis pectinata*), and cocklebur (*Xanthium strumarium*) were found where the Guinea grass cover was thin and the soil was exposed. Along the top banks of the gulches, buffelgrass formed a thick mat and Guinea grass was also abundant in places. Koa haole shrubs were scattered to very dense, forming small thickets, especially along the eastern section of Kalo'i Gulch. Tangled mats of coccinia vine (*Coccinia grandis*) were frequently observed climbing up and over the koa haole shrubs. A few kiawe trees were also found along the top of the gulches.

Scrub Vegetation. This vegetation type occupies the southern portion of the property and is usually one- to three-feet tall. Long dead stalks of sugar cane were scattered throughout this area, and at the time of this survey, the property was very dry with plant cover between 50 and 60 percent. Bare soil areas with large, knee-deep cracks

were prominent. Swollen fingergrass was the most abundant species, forming fairly large patches. In some places, buffelgrass (*Cenchrus ciliaris*) was locally abundant and formed a thick mat. Four herbaceous species were abundant to common, and small shrubs of hoary abutilon (*Abutilon incanum*), 'uhaloa (*Waltheria indica*), and 'ilima were abundant. Taller shrubs of koa haole and sourbush (*Pluchea carolinensis*) were scattered throughout this scrub cover. Other woody components were found in small numbers, and a few species usually used as landscaping material had established themselves within these former cane fields. Vegetation was somewhat denser along the old, crushed coral-covered cane haul roads and irrigation ditches, and the Koa haole shrubs and Guinea grass were common. Other species forming fairly large patches here include saltbush (*Atriplex suberecta*), 'uhaloa, slender mimosa (*Desmanthus pernambucan*), *Macroptilium atropurpureum*, Natal redtop grass (*Melinis repens*), 'ilima, and swollen fingergrass.

Ko'oloa'ula or Abutilon menziesii. Ko'oloa'ula is a member of the hibiscus or mallow family (*Malvaceae*). It is a branched shrub covered by velvety, silvery hairs. The heart-shaped leaves are silvery-green and the small 'ilima-like flowers range in color from pale peach to dark red. *Abutilon menziesii* is found in dry, lowland habitats on the islands of O'ahu, Maui, Lāna'i, and Hawai'i. In 1986, the species was federally listed as endangered and is protected under the provisions of the Endangered Species Act of 1973, as amended, and Chapter 195D, HRS, as amended. In its natural habitat the plants are threatened by browsing animals, competition from weedy introduced species, fires, predation by insects, loss of native pollinators, and development (U.S. Fish and Wildlife Service 1994).

A single plant of the endangered ko'oloa'ula was found within the property in the 1997 survey by K.M. Nagata, but has since died. The plant was identified as Cluster D, and seeds may be present in the soil or around the cluster (see Figure 3.5). A few individual plants exist along the southeastern boundary of the property. These plants represent the most mauka extension of the Cluster C population.

Native Species. Some native species including 'ilima, hoary abutilon, and 'uhaloa are common to abundant throughout the scrub vegetation on the property and elsewhere.

Discussion and Recommendations. The UH West O'ahu property was under sugarcane cultivation for nearly a century with the last harvest occurring in 1994, prior to permanent closure of Oahu Sugar Company, Ltd. (OSCo) in1995. Today, only dead stalks of sugar cane and faint traces of planting furrows remain. Weedy scrub vegetation consisting of a mixture of swollen fingergrass and buffelgrass, herbaceous species, and small shrubs covers the former cane fields on the southern half of the property, while the northern portion is actively cultivated for various fruit and vegetable crops by Aloun Farms, Inc. Kalo'i Gulch and Hunehune Gulch support dense Guinea grass and stands of koa haole shrubs.

The vegetation on the property is dominated by introduced or alien species. A total of 95 plant species were observed during this study. Of these, 89 (94 percent) are introduced. Four species are indigenous or presumably indigenous, including 'ilima (*Sida fallax*), hoary abutilon (*Abutilon incanum*), 'uhaloa (*Waltheria indica*), and popolo (*Solanum americanum*). Two endemic species are the endangered ko'oloa'ula (*Abutilon menziesii*) and pa'uohi'iaka (*Jacquemontia ovalifolia subsp. sandwicensis*).

Habitat Conservation Plan for *Abutilon menziesii* **at Kapolei (2004).** In 1997, the Hawai'i State Legislature enacted and passed House Bill 1292 to provide private landowners with incentives to promote the conservation and recovery of threatened and endangered species and their habitats. House Bill 1292 was signed into law by the Governor and now augments Chapter 195D, HRS. The Act provides for the preparation and implementation of Habitat Conservation Plans and Safe Harbor Agreements under the Federal Endangered Species Act and the State of Hawai'i endangered species law, and for the State law to complement the Federal law. In addition, the Act provides new incentives for private landowners to recover and protect threatened and endangered species on their lands, while providing a dynamic and flexible framework to allow creative solutions, increase public appreciation and understanding of endangered species issues, and encourage partnerships to help Hawai'i recover its unique natural heritage. This legislation makes Hawai'i law generally consistent with the Federal Endangered Species Act.

In accordance with both State and Federal regulations, mitigation measures have been identified in the *Habitat Conservation Plan for Abutilon menziesii at Kapolei* (HCP) prepared in consultation with the State DLNR. The HCP is included in Appendix C. The goal of the HCP is to initiate and sustain a program that would result in an overall net gain in the number of *Abutilon menziesii* on O'ahu. Three objectives have been identified: 1) describe the existing conditions of the Kapolei population; 2) describe the potential impacts of the Kapolei projects on *Abutilon menziesii*; and 3) describe the strategies and actions to mitigate the impacts. The major strategy designed to mitigate impacts and to benefit the species is the creation of three protected off-site wild populations on O'ahu from the single degraded Kapolei population. The new off-site populations will protect existing individuals as well as the genetic diversity of the existing population.

The primary funding mechanism is a trust fund for endangered species as promulgated in Chapter 195D-31, HRS. The State DOT will provide the funding to implement the HCP. The North-South Road, as planned, would affect approximately 25 percent of the population, and the other Kapolei projects (i.e., Kapolei Parkway Extension, DHHL developments, and DLNR developments) would affect the remaining 75 percent. As agreed by the DOT and the DLNR in a Memorandum of Agreement (MOA), the DOT on March 14, 2001 made available funds in the amount of \$250,000 for the initial five years of HCP implementation. These initial funds are being utilized (since approximately August 2001) by DLNR to manage and implement the HCP strategies. The MOA also stipulates that the subsequent 15 years will also be funded by DOT. An additional \$750,000 will also be delegated to DLNR upon the approval of the HCP by the Board of Land and Natural Resources (BLNR) (or the Hawai'i



Legislature, as appropriate) and the Governor's release of the funds for a total not to exceed \$1 million over a maximum period of 20 years. The DOT has developed a process for third party developers ("cooperators") to utilize the Incidental Take License through a Certificate of Inclusion. The cooperators would pay into a contingency fund, to be established by the DOT, for the following purposes: 1) to finance unanticipated costs incurred by DLNR in the implementation of the HCP measures; and 2) to fund the management and monitoring of three wild populations beyond 20 years. The total initial amount of the contingency fund is \$200,000. To ensure this fund DOT will deposit the full \$200,000 amount following the approval of the HCP.

On April 8, 2004, the BLNR unanimously approved the HCP and accompanying Incidental Take License and Certificate of Inclusion. The University is currently working with the DOT to be included in the Certificate of Inclusion for the Incidental Take License.

Anticipated Impacts and Mitigation Measures

Seeds that may be present in the soils and the few individual plants along the southeastern boundary of the property will be removed. However, the long-term impact on the ko'oloa'ula population in Kapolei would be beneficial, as proposed mitigation measures in the HCP will ensure the future propagation of new plants.

Under DLNR's interim management program, three new populations of ko'oloa'ula have been initiated at Koko Crater Botanical Garden, Ka'ena Point State Park, and the Honouliuli Unit of the Pearl Harbor National Wildlife Refuge. The Ka'ena Point State Park and Pearl Harbor National Wildlife Refuge are wild sites, and additional sites will continue to be pursued to yield three successful wild sites for propagation of ko'oloa'ula.

If an off-site preserve can be established, some of the plants would remain *in situ* for many years. Because the protection, propagation, and relocation of the ko'oloa'ula plant is a long-term undertaking, final implementation of the HCP will extend well into the project's construction period. Once construction and build-out of the campus is complete, use of the ko'oloa'ula in project landscaping and continued use of the ko'oloa'ula nursery for propagation will ensure a much larger and vigorous population of the ko'oloa'ula than would have occurred without development of the UH West O'ahu property. The HCP also proposes long-term management of the populations to occur concurrently with project development, over a period of approximately 20 years. The successful implementation of the HCP would significantly increase the numbers of new plants on O'ahu and improve the quality of existing populations on adjacent properties.

3.8 FAUNA

Existing Conditions

In April 2005, a wildlife survey was conducted by Phillip L. Bruner to document the species

of birds and mammals present on the project site and to note any habitat features utilized by native and migratory birds. This report is included in Appendix D.

Native Land Birds. No native land birds were observed during the survey. The only species known to exist in this region is the Hawaiian Owl, or Pueo (*Asio fammeus sandwichensis*), which is listed as an endangered species on O'ahu by the State of Hawai'i. In previous surveys at Barbers Point and Campbell Industrial Park, Pueo were seen on several occasion over the past ten years.

Native Waterbirds and Seabirds. The habitat provided on the project site is not suitable for native waterbirds or seabirds. As expected, none were seen during the survey.

Migratory Birds. Several Pacific Golden-Plover, or Kolea (*Pluvialis fulva*), were observed during the survey. This bird is not listed as a threatened or endangered species.

Alien Birds. Sixteen species of alien birds were observed during the survey (see Table 3.1). None of these species are listed as threatened or endangered. The array and relative abundance of alien birds observed is typical for this region of O'ahu.

COMMON NAME	SCIENTIFIC NAME	
Cattle Egret	Bubulcus ibis	
Ring-necked Pheasant	Phasianus colchicus	
Spotted Dove	Streptopelia chinensis	
Zebra Dove	Geopelia striata	
Sky Lark	Alauda arvensis	
Red-vented Bulbul	Pycnonotus cafer	
Northern Mockingbird	Mimus polyglotos	
Common Myna	Acridotheres tristis	
Red-crested Cardinal	Paroaria coronata	
Northern Cardinal	Cardinalis cardinalis	
House Finch	Carpodacus mexicanus	
Common Waxbill	Estrilda astrild	
Red Avadavat	Amandava amandava	
Nutmeg Mannikin	Lonchura cantans	
Chestnut Munia	Lonchura atricapilla	
Java Sparrow	Padda oryzivora	

Table 3.1 – Alien (Introduced) Birds Observed On-Site

Feral Mammals. The only feral mammals observed during the survey were the Small Indian Mongoose (*Herpestes auropunctatus*) and feral cats (*Felis catus*). Rats (*Rattus spp.*) and House Mouse (*Mus musculus*) likely occur in this area. The endangered Hawaiian Hoary Bat (*Lasiurus cinereus semotus*) is rarely observed on O'ahu and none were observed during this survey.

Anticipated Impacts and Mitigation Measures

No native, threatened, or endangered land birds were observed on the project site and none are expected to be impacted by the UH West O'ahu development. The habitat provided by the site is not suitable for native waterbirds or seabirds, and as expected, none were observed during the survey.

The UH West O'ahu development will change the landscape and habitat of the project site. As a result of this change, the number of some alien birds found on the site, such as Common Myna and Red-vented Bulbul, will likely increase. The number of alien species that prefer open grasslands and agricultural fields will likely decrease; however, nearby fields and landscaped areas within the project may provide similar habitat. Additionally, none of the alien species observed on the project site are listed as threatened or endangered.

The number of wintering migratory Pacific Golden-Plover may increase, as the proposed development will provide landscaped lawns. Feral cats may also increase in number, especially if they are fed, which often happens in public parks and around schools.

(This page intentionally left blank.)

4.0 Assessment of the Human Environment, Potential Impacts, and Mitigation Measures

4.0 ASSESSMENT OF THE HUMAN ENVIRONMENT, POTENTIAL IMPACTS, AND MITIGATION MEASURES

4.1 ARCHAEOLOGICAL AND HISTORIC RESOURCES

Existing Conditions

The earliest detailed map of the project area shows no habitation closer than the western edge of West Loch in the vicinity of Papapapuhi Point. The M.D. Monsarrat Surveyors 1878 Map of Honouliuli Taro Land documents substantial settlement at the Honouliuli Taro Lands (approximately 1.7 miles northeast of the UH West O'ahu property) in the Papapapuhi Point area, which appears to have been the focus of population in the Honouliuli ahupua'a. Fishponds, taro lo'i, shellfish collecting, and salt drying in the area would have focused population here in prehistoric times. The place name Papapapuhi Point must have been applied to the entire ahupua'a secondarily. The richness of the coastal Papapapuhi area is a contrast to the dry 'Ewa Plain environment in which the UH West O'ahu site is located.

While very little prehistory is known about the area in the immediate vicinity of the UH West O'ahu property, there is no indication of human occupation or any other utilization of the land. No remains of any prehistoric activity are believed to exist in the area, and no sites listed on the Hawai'i or National Register of Historic Places are found on the property.

The presence of any significant archaeological sites on the surface or subsurface of the UH West O'ahu property is unlikely due to the disruption caused by continuous sugarcane cultivation for nearly 70 years. This was confirmed by the *Archaeological Reconnaissance and Assessment of the HFDC – East Kapolei Development Project* (Scientific Consultant Services, Inc., 1996) and a review of existing literature including relevant archaeological research previously conducted in the project area, historic records, and maps (see Appendix E). A letter from the State DLNR Historic Preservation Division (SHPD) regarding the *East Kapolei Master Plan Environmental Impact Statement Preparation Notice* for the 1,300 acre area including the UH West O'ahu property stated:

A review of our records shows that there are no known historic sites at the project location...our records indicate that theses lands were commercially cultivated with sugar cane which altered the land for many years. The depth of cane cultivation exceeded the expected depth of historic sites in the area, based on site patterns in similar environmental contexts. Therefore, it is unlikely that significant historic sites will be found in the project area and we believe that this project will have "no effect" on such sites.

This letter is also included in Appendix E.

During the DEIS Public Review Period, the Office of Hawaiian Affairs (OHA) wrote:

A brief search of OHA's records generated the following:

The 'Ewa plain has historically been known to contain sinkholes in which human skeletal and avi-faunal remains have been encountered. These sinkholes can continue to exist in areas that have been graded or heavily cultivated for agricultural uses.

<u>According to records at the Bishop Museum pertaining to inventories conducted for</u> <u>compliance with the Native American Graves Protection and Repatriation Act of 1990, burial</u> <u>sites in Honouliuli and in 'Ewa in general have been documented in the past including:</u>

In 1938, human remains representing six individuals from Honouliuli, 'Ewa, O'ahu were collected by Kenneth P. Emory and William A. Lessa and acquired by the Bishop Museum. Museum documentation indicates these remains were in a shallow crypt burial one mile from the coast;

In 1933, human remains representing three individuals from stone pits at 'Ewa, O'ahu were collected by J.W. Barrington and Edwin H. Bryan;

In 1942, human remains representing two individuals from Kualakai, 'Ewa Beach, O'ahu were donated to the Bishop Museum;

In 1959, human remains representing seven individuals from 'Ewa, O'ahu were donated to the Bishop Museum by the Anthropology Club of the University of Hawai'i (from Standard Oil Refinery land);

In 1980, human remains representing nine individuals from Honouliuli, O'ahu were collected and donated to the Bishop Museum by Albert, Borthwick and Folk. Donor information indicates these human remains were recovered from coral sinkholes.

In the last decade, unmarked burial sites have been found in the area of St. Francis West, West Loch Estates, Old Fort Weaver Road, Kalaeloa, One'ula Beach, Campbell Estate, Ko'Olina and other areas in the vicinity of this project.

The depth of grading activities and the likelihood of adversely impacting any sub-surface cultural sites or deposits is contingent upon understanding the original surface grade as it may have existed prior to agricultural activities such as sugarcane.

Native Hawaiian burial sites have been found just on and under the surface to depths of eightor nine feet depending upon the nature of the terrain. Furthermore, the nature of documented interments in the 'Ewa area (stone pits, sinkholes, crypts, etc.) could lead to the survival of these sites despite intensive agricultural activities on the surface...

OHA is recommending that the project area undergo an Archaeological Inventory Survey (AIS), as a new study or supplement to the previous, in which all aspects of applicable 6E administrative rules and revised statutes are satisfied. It is our recommendation that the AIS include substantial subsurface inventory as much of the surface architecture may have been destroyed during previous commercial agriculture activities.

Also during the DEIS public review period, the State Historic Preservation Division wrote:

"...portions of Honouliuli ahupuaa have been demonstrated, by various reports on file at the State Historic Preservation Division (SHPD), to contain historically-significant resources, traditional and historic, including human remains/burials. Also, we believe that the area of potential effect (APE) associated with the current undertaking has a moderate to high potential of containing historically-significant remnants of sugar plantation infrastructure (e.g., irrigation systems)."

For these reasons, an archaeological inventory survey is warranted prior to the issuance of permits for the proposed undertaking.

Anticipated Impacts and Mitigation Measures

According to the archaeological reconnaissance and assessment, literature review, and the SHPD, there are no significant archaeological resources associated with the UH West O'ahu property and the larger HCDCH East Kapolei Development Project area. The project is not expected to impact any archaeological resources; however, should construction uncover any archaeological sites, work in the area will stop and the SHPD will be notified in accordance with applicable State regulations.

As recommended by OHA and SHPD, an archaeologist has been contracted to conduct an archaeological inventory survey. This AIS will be completed prior to the issuance of grading and building permits for the proposed project.

4.2 CULTURAL RESOURCES

Existing Conditions

The 1996 archaeological reconnaissance and assessment for the HCDCH East Kapolei Development Project area (referenced in Section 4.1 above), including the UH West O'ahu property, confirmed that no cultural resources are located in Hunehune Gulch, Kalo'i Gulch, or the broad agricultural plains. A field inspection found no cultural resources and it was determined that no further cultural resource work was required (see Appendix E).

According to the *North South Road and Kapolei Parkway Final Environmental Assessment* (DOT and DTS, 2004), the gathering of native plant resources may have been a major traditional cultural practice associated with the project area. However, the accessibility of Honouliuli lands to Hawaiians for gathering or other cultural purposes was radically curtailed due to cattle grazing and later commercial sugar cane cultivation.

A cultural impact assessment for the UH West O'ahu development was conducted and is

included below in its entirety.

The proposed University of Hawai'i West O'ahu project area is located in the *ahupua'a* of Honouliuli, in the 'Ewa district of O'ahu. Honouliuli is the largest *ahupua'a* in the moku (district) of 'Ewa, and includes the lands from the western boundary of Pu'uloa (Pearl Harbor) westward to the 'Ewa/Wai'anae district boundary. The coastline includes 12 miles of shallow reef, offering rich marine resources. For years, this coastline was used as an air station for the U.S. Navy. The area once known as Kalaeloa was renamed Barber's Point.

Honouliuli translates literally as, "dark water," or "blue harbor." There are several other accounts of the naming of this moku. In their book *Sites of Oahu*, Sterling and Summers, cite a story told by Simeon Nawa'a associating Honouliuli with the gods Kāne and Kanaloa:

When Kane and Kanaloa were surveying the islands they came to Oahu and when they reached Red Hill saw below them the broad plains of what is now Ewa. To mark boundaries of the land they would throw a stone and where the stone fell would be the boundary line. When they saw the beautiful land lying below them, it was their thought to include as much of the flat level land as possible. They hurled the stone as far as the Waianae range and it landed somewhere, in the Waimanalo section. When they went to find it, they could not locate the spot where it fell. So Ewa (strayed) became known by the name.

Another account of the name, Honouliuli, can be found in the legend of Lepeamoa, the "chicken" girl of Kapālama. In this legend, Honouliuli is the name of the husband of the chiefess Kapālama and grandfather of Lepeamoa (Thrum 1923:164-184). "Her grandfather gave his name, Honouliuli to a land district west of Honoulul..."

Fragments of History of Honouliuli

Honouliuli is a very sacred area to Hawaiians. The following Hawaiian historians and cultural preservationists were asked if they were aware of any current cultural practices in the project area: Ms. Momi Kamahele, Mr. Shad Kāne, Ms. Arline Eaton and Mr. Poni Kamau'u. Ms. Kamahele was unavailable for comment. Poni Kamau'u provided the following comments about the project site.

When asked about the area of study, Poni Kamau'u, a recognized historian, shared many experiences about the area of Pu'u Ku'ua. According to Mr. Kamau'u, many burial sites exist at Pu'u Ku'ua and Pu'u o Kapolei, as well as areas that have already been developed such as Ko Olina and Kapolei. Pu'u Ku'ua was home to the kauwā people (outcaste), slave caste of Hawai'i. In their publication, *Sites of Oahu*, Sterling and Summers translate an article from the Hawaiian newspaper, *Ka Loea Kālai'āina*, confirming that many Kauwā resided in the area:

... If you are above Pu'uloa, you will see Pu'u-o-Kapolei, a small hill. Lying below and

back of that hill is the government road going to Wai'anae. Above that is also a small hill and back of that, is a big hill and above it is a large hollow. That is Pu'u-Kuua where the very dirty ones lived.

In another article Pu'u Ku'ua is mentioned in a story of the gods Kāne and Kanaloa:

The two gods (Kāne and Kanaloa) looked down on the hollow and saw how thickly populated it was. The mode of living here was so that chiefs and commoners mixed freely and they were so like the lowest of people (Kauwā). That was what these gods said and that was the time when the term kauwā was first used, and was used for many years afterwards.

Most of Kalaeloa and Kapolei were once heavily populated and were prime spots for the Hawaiian spirits to "lele" or leap to the next world, which may explain why so many burial sites can be found on this side of the island. "Pu'u Ku'ua was also a battleground, and there's a few heiau out there." Mr. Kamau'a mentions, "Kaupe'a plains, which kind of near to Pu'u Ku'u, is famous for its wandering souls." He adds, "A few Hawaiians today still go out to the leeward coast to speak to their *kūpuna* (elders)." According to the book, *Ka Po'e Kahiko*, there are different stages of death:

... There were three realms (ao) for the spirits of the dead... There were, first, the realm of the homeless souls, the ao kuewa; second, the realm of the ancestral spirits, the ao 'aumakua; and third, the ream of Milu, ke ao o Milu ...

The ao kuewa, the realm of homeless souls, was also called the ao 'auwana, the realm of wandering souls. When a man who had no rightful place in the 'aumakua realm (kanaka kuleana 'ole) died, his soul would wander about and stray amongst the underbrush on the plain of Kama'oma'o on Maui, or in the wiliwili grove of Kaupe'a on Oahu. If his soul came to Leilono [in Halawa, 'Ewa near Red Hill], there he would find the breadfruit tree of Leiwalo, ka'ulu o Leiwalo. If it was not found by an 'aumakua soul who knew it (i ma'a mau iaia), or one who would help it, the soul would leap upon the decayed branch of the breadfruit tree and fall down into endless night, the pō pau 'olo o Milu. Or, a soul that had no rightful place in the 'aumakua realm, or who had no relative or friend (makamaka) there who would watch out for it and welcome it, would slip over the flat lands like a wind, until it came to a leaping place of souls, a leina a ka 'uhane... [Kamakau 1991].

On the plain of Kaupe'a, beside Pu'uloa [Pearl Harbor], wandering souls could go to catch moths (pulelehua) and spiders (nanana). However, wandering souls could not go far in the places mentioned earlier before they would be found catching spiders by 'aumakua souls, and be helped to escape... [Kamakau 1991].

Mr. Kamau'u says that Honouliuli and Pu'u o Kapolei are also mentioned quite often in the epic story of Kamapua'a. There are many references to Kamapua'a, his grandmother

Kamaunuaniho (who resides at Pu'u o Kapolei), as well as his lover/enemy Pele and her sister Kapo.

Kamapua'a subsequently conquered most of the island of O'ahu, and, installing his grandmother [Kamaunuaniho] as queen, took her to Pu'uokapolei, the lesser of the two hillocks forming the southeastern spur of the Wai'anae Mountain Range, and made her establish her court there. This was to compel the people who were to pay tribute to bring all the necessities of life from a distance, to show his absolute power over all [Nakuina 1904:50-51].

Along the highway from Pu'u Ku'ua through Nānākuli, Mr. Kamau'u recounts a story of a white dog with reddish eyes (said to be Poki, the pet dog of Pele), whose size is comparable to an automobile. In times of danger, Poki would block the pathway to restrict people from entering the danger zone. When Pele's sister, Kapo would visit O'ahu, this white dog would take the form of clouds adorning Pele's beloved sister in the shape of a lei, which is according to Mr. Kamau'u, one of the stories of how Kapolei acquired its name.

Kamukila Campbell, the wife of Mr. James Campbell shared many stories with Mr. Kamau'u about Honouliuli as well as Pu'u Makakilo, a hill located just north of the project site. Mrs. Campbell informed Mr. Kamau'u of the many *ki'i pōhaku* (petroglyphs) that point the way to the *kilo* (stargazing sites, also a reader of omens). The kilo site provides a way for the people on earth to communicate with the heaven and the sky above. When the stones are aligned properly, four winds are said to appear over the site. As Mr. Kamau'u stood there near the stones, he noticed the alignment of the stones acted as a compass. Makakilo has been a common place for Hawaiian astronomy, hence, the name *Makakilo* (the observing eyes). Makakilo, is also believed to be closely watched and guarded by the eyes of a demigod that takes the form of a *pueo* (owl).

Shad Kāne is another noted cultural preservationist and has recently recommended some appropriate street names for the area of Kapolei. While Pu'u o Kapolei was not one of his recommended street names, he mentions that this practice of *kilo* was also observed on the hill known as Pu'u o Kapolei. The sun, the moon, stars and constellations with reference to geographical features, were used to determine the time of year. Pu'u o Kapolei is now in the center, the "piko," of the new city of Kapolei.

Land Uses

In traditional times, there was an abundance of kalo (taro) fields in Honouliuli. Archaeological data collected by Dicks, Haun, and Rosendahl shows evidence of widespread wetland systems (*lo'i*).

Ms. Arline Eaton, affectionately referred to as, Aunty Arline, is a *kupuna* (elder) teacher at Iroquois Point Elementary. She was raised in the traditional Hawaiian style, in an area once known as Keahi and Kūpaka (now known as Iroquois Point/'Ewa), with Hawaiian as her first

language. She was referred by Shad Kāne, and is considered one of today's most knowledgeable *kūpuna* of the Honouliuli area.

She shared stories of going down to the *lo'i* to "pull" taro. Although the lo'i was not as abundant by the time she was born, she used to go upland to gather taro. She also mentions that there were many dryland taro where she lived in Keahi, near the beach. "The kalo we used to pick was more red. The ones up there was the wetland kalo, more purple. Poi is poi, it's still *'ono* (delicious)". Aunty Arline explained that there were different colors to the taro of different areas. She assumes it was because of the different soils.

Aunty Arline was a good friend of Bernice Kahanumoku, sister of the legendary Duke Kahanumoku. She and her *tūtū papa* (grandfather) would travel by canoe to Kou (Waikīkī), to visit her dear friend. The trip would be long, and they would normally stay overnight before returning home. "Back then, never have *ka'a* (car), so we use the ocean. He would paddle and I would just go with him."

During their visits they would bring fish and *limu* (seaweed). "They had fish in Waikīkī, but we would bring them fish and *limu*. This place had plenty *limu* all over in this area, you could smell it in the air." She claims that when an influx of foreigners came to visit the area they would often take more than they needed, and often sell the *limu*. "That's not the way to do it. We take only what we need."

In 1877, most of the lands in Honouliuli were purchased by James Campbell for the amount of \$95,000. Mr. Campbell purchased the lands from the *ali'i* (high chiefess) Miriam Keahikuni Kekau'ōnohi, who was granted the land during the *māhele*, land division in 1845 and 1846, when Hawai'i was first introduced to private property and land ownership. Until 1889, the majority of Campbell's lands were used for cattle grazing, resulting in open and sparsely vegetated plains in the 'Ewa region. At one point the land was so dry and full of bottomless fissures that water would be lost and an irrigation system would seem nearly impossible.

The first crop, 2,849 tons of sugar, was harvested in 1892, Ewa was the first allartesian plantation and, in spite of early troubles...it gave an impressive demonstration of the part artesian wells were to play in the subsequent industry of Hawaiian sugar industry [Kuykendall 1967:69].

Existing Conditions and Potential Impacts

A field inspection was conducted for the proposed North-South Road and Kapolei Parkway Extension in April 2004 and in June 1996. Both surveys were done by Cultural Surveys Hawai'i. Findings show that most of the project area comprises of land that was used for sugar cultivation. Since the cessation of sugar production, various developments have taken place or are proposed including: 'Ewa Villages (located southeast of the project site) and the Villages of Kapolei (located southwest of the project site). According to the surveys, there

were no properties related to traditional Hawaiian culture. Two plantation-era structures, however, were observed (Chiojioji et al, 2004).

Most of the project area has been deserted and no longer produces commercial sugar. Sugar production would have destroyed any historic properties related to Hawaiian culture, and it is unlikely that the project will impact current cultural practices such as gathering. There should be minimal direct impact upon native Hawaiian cultural practices and beliefs. Based on the statement of Mr. Poni Kamau'u, that Kapolei was once very heavily populated, and perhaps once a battlefield, special care should be observed in the event a Hawaiian burial is discovered.

Anticipated Impacts and Mitigation Measures

The UH West O'ahu property was used to produce commercial sugar, and portions of the property continue to be used for agricultural purposes under revocable leases. Sugar production would have destroyed any cultural resources on the property. As such, there should be minimal direct impact upon native Hawaiian cultural practices and beliefs. However, should native iwi or native Hawaiian cultural or traditional deposits be found during ground disturbance, work will cease, and the appropriate agencies will be contacted.

Despite past agricultural practices, many landforms surrounding the project area still exist, including Pu'u Ku'ua and Pu'u o Kapolei. The campus will be oriented to capitalize on views of these landforms, and the project will be designed to include references to the mo'olelo (legends and myths) of the area.

4.3 AGRICULTURAL IMPACT

Existing Conditions

The majority of the property is under month-to-month revocable leases to Aloun Farms, Inc. (northeastern portion of the property) and A.M. Enterprise, Inc. (northwestern portion of the property) (see Figure 1.4). Aloun Farms, Inc. leases lands to cultivate vegetables and fruits. Additionally, a small portion of the property (0.826 acres) is currently encumbered by a revocable permit to Kapolei People's, Inc. (Kapolei Golf Course) for parking lot expansion purposes.

Anticipated Impacts and Mitigation Measures

<u>Portions of the c</u>urrently cultivated lands within the property will be withdrawn from agricultural production <u>as portions of the project site are developed</u> for the proposed development, which would have a negative impact on the above listed agricultural lessees. This will result in some loss in revenues, jobs, or-and payroll generated by agricultural activities. However, the State and City have long planned for new development in the project area, and the property is within the Urban State Land Use District. <u>Tenants have been fully</u>

aware, for quite some time, that the proposed project area would be used to accommodate for future development in the region. It is our understanding with the closure of the Del Monte Fresh Produce, there are 3,100 acres of vacant lands in Kunia available for cultivation.

The *East Kapolei Master Plan Development Project: Impact on Agriculture* (Decision Analysts Hawaii, Inc., 1996) was prepared for the HCDCH East Kapolei EIS. The report stated that ample agricultural land will be available to accommodate the projected growth of diversified agriculture on O'ahu and statewide. This conclusion was based on the following:

- 1. a vast amount of agricultural land and water is currently available, having been freed from sugar and pineapple production;
- 2. additional sugarcane acreage and water may be freed in coming years;
- 3. some of the remaining sugar companies, if not most of them, are willing to make their lands available for profitable replacement crops to the extent that such crops exist; and
- 4. projected land requirements for diversified agriculture are modest compared to the land supply.

The report also stated that the limiting factor to the growth of diversified agriculture will not be the land supply, but rather the size of the market for those crops that can be grown profitably in Hawai'i. A potential use for former sugar cane and pineapple lands in Central O'ahu and elsewhere is the use of crops for ethanol production.

The UH West O'ahu will provide numerous employment opportunities during construction and long-term operation (see Section 4.9.1).

4.4 NOISE

Existing Conditions

A noise assessment report for the UH West O'ahu property was conducted in June 2006 by D.L. Adams Associates, Ltd. This report is included in Appendix F.

Noise Measurement

Short- and long-term noise was measured at three different locations of the project site: 1) adjacent to Farrington Highway, about 45 feet makai of the edge-of-pavement (Location S1); 2) at the makai end of the project site, in a field near the proposed North-South Road (Location L1); at the mauka end of the project site, about 100 feet from the edge of Farrington Highway (Location L2).

Location S1. An approximate 30 minute ambient noise level measurement was conducted in the morning at 8 am and in the evening at 5 pm. The daytime (8:00 am to 8:30 am) short-term noise measurement resulted in a 71.4 A-weighted decibel (dBA). The nighttime (5:00 pm to 5:30 pm) short-term noise measurement resulted in a 64.9 dBA. Simultaneous traffic

counts were also conducted. These figures were to be used in calculating the long-term noise levels for Locations L1 and L2.

Location L1. Sound levels at Location L1 were typical of a rural ambient environment. Daytime (7:00 am to 10:00 pm) noise levels ranged from 42 to 57 A-weighted decibels (dBA). Average nighttime (10:00 pm to 7:00 am) noise levels ranged from 34 to 55 dBA; however, a meter overload occurred between 3:00 pm and 5:00 am possibly due to rain or excessive winds. Therefore, the average day-night noise level was estimated to be 55 dBA. Dominant noise sources at this location included wind, birds, aircraft flyovers, farm workers, and possible operation of farming equipment. Secondary noise sources included vehicular traffic from Farrington Highway and the H-1 Freeway.

Location L2. Daytime sound levels at Location L2 ranged from 48 to 63 dBA. Nighttime sound levels ranged from 39 to 59 dBA. The average day-night noise level was calculated to be 59 dBA. Dominant noise sources at this location included vehicular traffic from Farrington Highway and wind. Secondary noise sources included birds, farming equipment, electrical lines, and vehicular traffic from the H-1 Freeway.

Ambient Noise

Vehicular noise from Farrington Highway and the H-1 Freeway located mauka of the property dominate the ambient environment in the vicinity of these roadways. Additionally, a quarry and recycling plant located near the property may contribute to some of the ambient noise.

Airport Noise Contours

The UH West O'ahu property is located several miles from the Kalaeloa Airport and the Honolulu International Airport, and is outside of the 55 L_{dn} noise contour for both airports. According to the DOT Airports Division, property will be outside of the high noise contours generated by aircraft activity at Kalaeloa (see Section 8.2).

Anticipated Impacts and Mitigation Measures

Construction Noise

Construction of the UH West O'ahu will involve excavation, grading, and other typical construction activities, which may generate a significant amount of noise. Since much of the land surrounding the property is in agricultural use, the project-generated noise is not expected to impact adjacent properties. However, residences built during the initial phase of development may be impacted by construction noise from subsequent development within the property. Earth-moving equipment (i.e., bulldozers and diesel-powered trucks) will likely be the loudest equipment used during construction. Construction noise impacts will be relatively short-term during daytime hours. Construction activities will comply with State DOH noise regulations. <u>The UH West O'ahu will monitor construction of the proposed site</u>

to ensure that contractors comply with applicable noise standards.

When construction noise exceeds or is expected to exceed the State's maximum permissible property line noise levels, a permit must be obtained from the State DOH to allow the operation of vehicles, cranes, construction equipment, and power tools, which emit noise levels in excess of maximum permissible levels. The use of mufflers on diesel and gasoline engines, and the use of properly tuned and balanced machines are common mitigation measures.

Primarily Traffic-Related Noise Level Predictions

Noise levels were predicted (based on traffic counts) at four locations: A) along Farrington Highway; B) along the proposed North-South Road; C) along the proposed Road B; and D) along the proposed Road F, near the proposed elementary school.

Location A. At Location A, parcels adjacent to Farrington Highway are expected to experience a vehicular traffic noise level increase of 2 to 4 dB in the future, without the UH West O'ahu development. The increase in traffic noise due to the project is less than 2 dB. In general, an increase of 3 dB or less is not considered to be significant. Approximately 225 feet makai of Farrington Highway, noise levels are expected to be at the Federal Highway Administration (FHWA)/DOT maximum of 67 dBA.

Location B. Future traffic projections show that traffic noise levels at Location B (at least 100 feet from the proposed North-South Road) are expected to be slightly below the FHWA/DOT maximum noise limit of 67 dBA, with or without the UH West O'ahu development. The increase in traffic noise due to the UH West O'ahu development is expected to be less than 2 dB. In general, an increase of 3 dB or less is not considered to be significant.

Locations C and D. At approximately 25 feet from proposed Roads B and F, noise levels due to vehicular traffic are predicted to be well below the FHWA/DOT maximum noise limit of 67 dBA.

Surrounding Community. Residences located southwest of the UH West O'ahu property and adjacent to Farrington Highway are beyond the route of heavy trucks that dominate the UH West O'ahu property. Existing noise levels in this area are estimated to be lower than 67 dBA during peak traffic hours. Without the project, vehicular traffic noise levels are expected to increase by 2 to 4 dB. The increase in traffic noise due to the project is less than 2 dB. In general, an increase of 3 dB or less is not considered to be significant.

Compliance with Noise Regulations

State Community Noise Control. Noise emanating from commercial uses could significantly impact adjacent noise-sensitive residential areas. Stationary mechanical equipment typically used for commercial buildings must meet State noise rules. For commercial areas, maximum

noise levels are 60 dBA during the day and 50 dBA during the night. For residential areas, maximum noise levels are 55 dBA during the day and 45 during the night. Noisy equipment will be located away from neighbors and residential units, as practical.

For commercial areas to be compatible with adjacent residential areas, noise mitigation measures should be implemented. Typical noise mitigation for stationary equipment (i.e., air conditioners, ventilation equipment, refrigerators, and compressors) includes the use of mufflers, silencers, acoustical enclosures, and noise barrier walls. Other sources of noise include non-stationary equipment such as trucks loading and unloading supplies. Ambulance sirens and back-up alarms on trucks and forklifts also generate noise but are exempt from DOH noise regulations. Noisy activities, such as traffic access and loading areas, should be located away from nearby residential areas. Restrictions may need to be placed on commercial uses to control the development of potential noise-producing industries in commercial areas. The noise assessment report recommended that sale and lease documents for commercial property disclose and emphasize the significance of the DOH noise regulations with respect to the abutting residential areas. For mixed-use areas, DOH regulations specify that the primary land use designation in a mixed zoning district shall be used to determine the maximum permissible sound level.

All project activities will comply with Chapter 11-46, HAR "Community Noise Control".

FHWA/DOT Limits. The FHWA/DOT maximum noise limit is 67 dBA during peak traffic hours. With the UH West O'ahu development, vehicular traffic noise is predicted to be at or below the FHWA/DOT maximum noise limit.

Vehicular traffic noise from Farrington Highway may significantly impact the proposed development. The noise assessment report stated that residences constructed on parcels bordering Farrington Highway should be at least 225 feet from the edge of pavement so as not to exceed the FHWA's maximum exterior noise limit. Any homes within 225 feet of Farrington Highway will require some type of noise mitigation to meet the criteria. No homes should be built within 75 feet of Farrington Highway, even if noise mitigation treatments are planned.

Similarly, vehicular traffic noise from the North-South Road may significantly impact the proposed development. The noise assessment report stated that residences constructed on parcels bordering North-South Road should be at least 100 feet from the edge of pavement so as not to exceed the FHWA's maximum exterior noise limit of 67 dBA. Any homes within 100 feet of North-South Road will require some type of noise mitigation to meet the criteria.

Effective noise mitigation measures could include:

- Constructing barrier walls and/or earth berms along roadways;
- Installing air conditioners in buildings instead of relying on natural ventilation;
- Acoustically softening interior spaces by adding thick carpet with a padding underlay, an acoustical tile ceiling, louvered closet doors, etc.;

- Using exterior wall constructions that exhibit high noise reductions; or
- Reducing the elevation of roadways relative to adjacent lands.

Environmental Protection Agency (EPA)/U.S. Department of Housing and Urban Development (HUD) Guidelines. The EPA's existing design goal is for a calculated day-night noise level to be less than or equal to 65 dBA for exterior noise levels. However, the EPA's future design goal is for a calculated day-night noise level to be less than or equal to 55 dBA for exterior noise levels. Noise levels at homes 225 feet from Farrington Highway and 100 feet from North-South Road are expected to be below the existing EPA design goal but above the future EPA design goal. The HUD exterior design goal is for a calculated day-night noise level less than or equal to 65 dBA for residential units at least 225 feet from a roadway. Day-night noise levels are expected to be within the HUD noise guidelines for residential units 225 feet from Farrington Highway and 100 feet from North-South Road.

It should be noted that the EPA and HUD noise guidelines are design goals and not enforceable regulations.

Federal Aviation Administration (FAA)/DOT Guidelines. The project will not be impacted by aircraft noise, as the property is located outside of the 55 L_{dn} noise contour for the Kalaeloa Airport and the Honolulu International Airport. No noise mitigation to attenuate aircraft noise is necessary.

During the pre-consultation process, the DOT Airports Division stated that the UH West O'ahu campus will be outside of the high noise contours generated by the aircraft activity in the Kalaeloa Airport (see Section 8.1).

Honolulu High-Capacity Transit Project and Federal Transit Administration (FTA) Guidelines. The DTS is evaluating alternatives for high-capacity transit service between Kapolei and Mānoa. Two of the alignment alternatives run along the North-South Road and will include at least two stations in the vicinity of the UH West O'ahu property. Noise impacts will occur if the rail transit noise exceeds 58 dBA, and severe noise impacts occur if the rail transit noise exceeds 64 dBA. Noise can be effectively mitigated at the noise source (rail car) and along the noise path (tracks) through resilient rubber wheels, vehicle skirts, wheel truing, rail grinding, or undercar absorption. Additionally, sound barrier walls close to the guideway are very effective, often reducing noise by 6 to 10 dB. Ballast can be laid to reduce noise by 3 dB at-grade and 5 dB on arterial guideways.

Board of Education (BOE) Guidelines. BOE Policy 6700 requires air conditioners to be installed for schools exposed to an exterior noise level of 65 dBA. An elementary school at the presently proposed location is not likely to experience a noise level greater than 65 dBA, and noise mitigation is not required. However, temporary noise mitigation measures will be required if construction activities occur in the vicinity of the elementary school. Currently, it is not known when construction of the elementary school will take place. If the elementary school is constructed prior to completion of development of nearby uses, noise mitigation

measures will be implemented.

4.5 AIR QUALITY

An air quality study was conducted in June 2006 by B.D. Neal and Associates to describe existing air quality in the project area and to assess the potential short- and long-term direct and indirect air quality impacts that could result from construction and use of the proposed facilities. The study also discussed potential impacts on the project from nearby air pollution sources and proposed measures to mitigate impacts by the project or on the project. The study is included in Appendix G.

Existing Conditions

The present air quality of the project area appears to be reasonably good based on nearby air quality monitoring data. Both Federal and State standards have been established to maintain ambient air quality. At the present time, seven parameters are regulated including: particulate matter, sulfur dioxide, hydrogen sulfide, nitrogen dioxide, carbon monoxide, ozone and lead. State air quality standards are comparable to the Federal standards, except those for nitrogen dioxide and carbon monoxide, which are more stringent than the Federal standards.

Air quality data from the nearest monitoring stations operated by the State Department of Health (DOH) suggest that all Federal air quality standards are currently being met, although occasional exceedances of the more stringent State standards for carbon monoxide may occur near congested roadway intersections.

Pollution Sources

Air quality in the project area is mostly affected by air pollutants from motor vehicles, industrial sources, agricultural operations, and natural sources to a lesser extent. Much of the particulate emissions on O'ahu originate from area sources, such as the mineral products industry and agriculture. Sulfur oxides are emitted almost exclusively by point sources, such as power plants and refineries. Nitrogen oxides emissions emanate predominantly from industrial point sources, although area sources (mostly motor vehicle traffic) also contribute a significant share. The majority of carbon monoxide emissions occur from area sources (motor vehicle traffic), while hydrocarbons are emitted mainly from point sources. Based on previous emission inventories that have been reported for O'ahu, emissions of particulate and nitrogen oxides may have increased during the past ten years, while emissions of sulfur oxides, carbon monoxide, and hydrocarbons have probably declined.

The H-1 Freeway is a major arterial roadway that presently carries moderate to heavy levels of vehicle traffic during peak traffic hours. Emissions from motor vehicles using this roadway, primarily nitrogen oxides and carbon monoxide, will tend to be carried away from the project site by the prevailing winds.

Several sources of industrial air pollution are located in the Campbell Industrial Park, which is located a few miles from the project site toward the southwest. Industries currently operating at Campbell Industrial Park include the Chevron and BHP refineries, the Honolulu Program of Waste Energy Recovery (H-POWER) facility, Kalaeloa Partners, Applied Energy Services, Hawaiian Cement, and others. The HECO Kahe Generating Station is located a few miles to the northwest at Kahe Point. These industries emit large amounts of sulfur dioxide, nitrogen oxides, particulate matter, carbon monoxide, and other air pollutants; however, prevailing winds from the east or northeast carry these emissions away from the UH West O'ahu property most of the time.

Until recently, air pollution in the project area originating from agricultural sources could mainly be attributed to sugar cane operations near the project site. Emissions from both the mill and the canefield operations in the area have now been eliminated with the closure of Oahu Sugar Company, Ltd. (OSCo). Much of the former sugarcane lands are currently being used as pastureland or for diversified agriculture.

Natural sources of air pollution emissions that could also affect the project area but cannot be quantified very accurately include the ocean (sea spray), plants (aero-allergens), wind-blown dust, and perhaps distant volcanoes on the island of Hawai'i.

Kapolei Air Quality Monitoring Station

During the 2000 to 2004 period, sulfur dioxide was monitored by the DOH at an air quality station located at Kapolei. Concentrations monitored were consistently low and there were no exceedances of the State/Federal 3-hour or 24-hour ambient air quality standards (AAQS) for sulfur dioxide during the 5-year period.

Particulate matter was also measured at the Kapolei monitoring station. Average annual concentrations reported were within the State and Federal AAQS.

Carbon monoxide measurements were also made at the Kapolei monitoring station. No exceedances of the State or Federal 1-hour or 8-hour AAQS were reported.

Nitrogen dioxide was also monitored by the DOH at the Kapolei monitoring station. Annual average concentrations of this pollutant were within the State and Federal AAQS.

Sand Island Air Quality Monitoring Station

The nearest available ozone measurements were obtained at Sand Island (about 10 miles east of the project area). The second-highest 8-hour concentrations measured between 2002 and 2004 were well within the State and Federal standards.

Downtown Honolulu Air Quality Monitoring Station

The nearest and most recent measurements of ambient lead concentrations reported were made at the downtown Honolulu monitoring station between 1996 and 1997. Average quarterly concentrations were near or below the detection limit, and no exceedances of the State AAQS were recorded. Monitoring for this parameter was discontinued in 1997.

Based on this data, it appears likely that the State AAQS for sulfur dioxide, nitrogen dioxide, particulate matter, ozone, and lead are currently being met at the project site. While carbon monoxide measurements at the Kapolei monitoring station suggest that concentrations are within the State and Federal standards, local "hot spots" may exist near traffic-congested intersections.

Anticipated Impacts and Mitigation Measures

It may be inevitable that some short- and/or long-term impacts on air quality will occur either directly or indirectly as a result of project construction and use.

Short-term Impacts and Mitigation

Short-term impacts from fugitive dust will likely occur during the project construction phase. To a lesser extent, exhaust emissions from stationary and mobile construction equipment, from the disruption of traffic, and from workers' vehicles may also affect air quality during the period of construction.

State air pollution control regulations require that there be no visible fugitive dust emissions at the property line. Hence, an effective dust control plan must be implemented to ensure compliance with State regulations. Fugitive dust emissions can be controlled to a large extent by watering active work areas, using wind screens, keeping adjacent paved roads clean, and covering open-bodied trucks. Other dust control measures could include limiting the area that can be disturbed at any given time and/or mulching or chemically stabilizing inactive areas that have been worked. Paving and landscaping of project areas early in the construction schedule will also reduce dust emissions. Monitoring dust at the project boundary can help to evaluate the effectiveness of the dust control program. Exhaust emissions can be mitigated by moving construction equipment and workers to and from the project site during off-peak traffic hours.

All construction activities on the property will comply with State Air Pollution Control regulations and the provisions of Section 11-60.1-33, HAR, on Fugitive Dust. An effective dust control plan for the construction phase will be implemented, and particular care will be taken when construction activities take place near existing homes, businesses, or highways.

Long-term Impacts and Mitigation

After construction, motor vehicles traveling to and from the proposed development will result in a long-term increase in air pollution emissions in the project area. To assess the impact of emissions from these vehicles, an air quality modeling study was undertaken to estimate current ambient concentrations of carbon monoxide at intersections in the project vicinity and to predict future levels both with and without the proposed project. During worst-case conditions, model results indicated that in the year 2015 without the project 1-hour and 8hour carbon monoxide concentrations would comply with both the State and Federal AAQS. With the project in the year 2015, carbon monoxide concentrations were estimated to increase at some locations in the project area by 20 percent or more. Even with these higher concentrations, worst-case concentrations should remain well within the Federal standards but may approach the more stringent State standards.

Measures to mitigate long-term, traffic-related air pollution include improvements to roadways, reduction of traffic, and reduction of individual vehicular emissions. Given that the more stringent State standards would likely be met during worst-case conditions, implementing any air quality mitigation measures for long-term traffic-related impacts is probably unnecessary and unwarranted. Nevertheless, the UH West O'ahu will encourage incorporate public transit into the design of its campus, as it is located near planned rail transit nodes. The addition of buffer zones between walkways and roadways can help mitigate potential air quality impacts, and the UH West O'ahu will include acres of parks and open space. In addition, all internal roads are currently planned to include planting strips or tree wells.

Predicted Worst-case Carbon Monoxide Concentrations. To evaluate the potential long-term indirect ambient air quality impact of increased roadway traffic associated with the project, computerized emission and atmospheric dispersion models were used to estimate ambient carbon monoxide concentrations along roadways leading to and from the project. Maximum carbon monoxide concentrations typically coincide with peak traffic periods; therefore, air quality impacts were assessed during the morning and afternoon peak traffic periods evaluated by the traffic study (see Section 4.10.1 and Appendix H). Air quality analyses were conducted for the following intersections:

- North-South Road and Farrington Highway;
- North-South Road and Road B;
- North-South Road and Road F;
- Farrington Highway and Road A; and
- Farrington Highway and Road F.

In 2015, without the project, the highest worst-case 1-hour carbon monoxide concentration was predicted to occur during the morning at the intersection of the North-South Road and Farrington Highway. A value of 7.9 milligrams per cubic meter (mg/m³⁾ was predicted to occur at this location and time. Peak-hour worst-case values at the other locations and times studied for the 2015 without-project scenario ranged between 4.1 and 6.3 mg/m³, with higher

concentrations occurring during the morning at all three locations studied. All projected worstcase concentrations for this scenario remained within the State and Federal standards.

In 2015, with the proposed project, the predicted highest worst-case 1-hour concentration continued to occur during the morning at the intersection of the North-South Road and Farrington Highway with a value of 9.6 mg/m³, which is about 22 percent higher compared to the without-project scenario. Other concentrations for this scenario ranged between 3.8 and 7.9 mg/m³. Although the predicted concentrations increased at all of the locations studied compared to the without-project scenario, the values remained within the State and Federal standards, but the State standard was met by only a small margin.

Several assumptions were made concerning both traffic movement and worst-case meteorological conditions in this air quality analysis. The analysis assumed a wind speed of 1 meter per second with a steady direction for one hour will occur. A steady wind of 1 meter per second blowing from a single direction for an hour is extremely unlikely and may occur only once a year or less. With wind speeds of 2 meters per second, for example, computed carbon monoxide concentrations would be only about half the values given above.

Electrical Demand and Solid Waste Generation. Depending on the demand levels, long-term impacts on air quality are also possible due to indirect emissions associated with a development's electrical power and solid waste disposal requirements. Based on the estimated demand levels and emission rates involved, any impacts will likely be negligible. Sustainability guidelines have been established for the UH West O'ahu. The guidelines strive to achieve a Leadership in Energy and Environmental Design (LEED) NC silver standard for campus development and community performance standards based on the LEED NC v2.2 rating system. The UH West O'ahu will strive to incorporate energy conservation design features and promote conservation and recycling programs to mitigate potential long-term air quality impacts.

Impacts on the Project

Due to the close proximity of Campbell Industrial Park, emissions emanating from industrial facilities may occasionally impact the project through coincidental occurrences of industry malfunctions and southwesterly winds, both of which are relatively infrequent events. Increased scrutiny by the DOH, an air quality task force mandated by the State legislature, and the modernization by some industrial park tenants should help to mitigate future impacts on the proposed project.

4.6 MAN-MADE HAZARDS

Existing Conditions

Environmental legislation and laws regulate hazardous and toxic wastes and materials, and the manufacture, generation, use, storage, release, and disposal of such materials. As a

consequence of these laws and regulations, numerous agencies collect and disseminate information for use in evaluating recognized environmental conditions (RECs).

PBR HAWAII used Environmental Data Resources, Inc. (EDR) to search major Federal, State, and local regulatory agency lists of RECs in January 2003. No hazardous sites were identified on the UH West O'ahu property. <u>During the public review period the DOH Hazard Evaluation & Emergency Response Office (HEER) recommended that a Phase I ESA be conducted and the property be characterized for arsenic contamination. The UH West O'ahu will conduct a Phase I ESA and soils testing once an agreement with the private developer is reached.</u>

Federal American Society for Testing and Materials (ASTM) Standard

Resources Conservation and Recovery Information System-Small Quantity Generator (*RCRIS-SQG*). A review of the RCRIS-SQG list revealed that there is one site within approximately 0.75 miles of the UH West O'ahu property. This is the Makakilo Quarry at 91-920 Farrington Highway (owned by Grace Pacific Corporation). According to the EDR radius map report, this facility has no violations.

State ASTM Standard

State Hazardous Waste Sites (SHWS). A review of the SHWS list revealed that there are two sites within approximately 1.5 miles of the property. One of these sites is the Pacific Concrete and Rock Landfill at 91-402 Farrington Highway, west-northwest of the property and mauka of the H-1 Freeway. According to the EDR radius map report, the Pacific Concrete and Rock Landfill is a solid waste disposal facility.

The other site on the SHWS is the Ewa Sugar Mill at Renton Road (formerly owned by Oahu Sugar Company). On May 7, 1997, a fallen drum was located at the west end of Renton Road. Black liquid seeped into the ground near the fallen drum. On August 23, 1997, explosions and chemical fumes from the mill were reported. On April 20, 1998, a resident noticed the odor of TCA or TCE at an excavation site. The site was covered by pieces of concrete. The Comprehensive Environmental Response, Compensation, and Liability Information System (CERCLIS) status of the Ewa Sugar Mill is No Further Remedial Action Planned (NFRAP). CERCLIS sites are potentially hazardous waste sites that have been reported to the U.S. Environmental Protection Agency. As of February 1995, CERCLIS sites designated NFRAP have been removed from CERCLIS.

Leaking Underground Storage Tank (LUST). A review of the LUST list revealed that one LUST site exists within approximately one mile of the property. This site is the Makakilo Quarry. According to the EDR radius map report, this facility has completed its site cleanup.

Underground Storage Tank (UST). A review of the State's UST list revealed that Makakilo Quarry has one UST currently in use. The tank was installed on April 10, 1980 and is used to

store used oil. Makakilo Quarry also has two USTs that were permanently closed on April 13, 1996. These tanks were installed on April 10, 1976 and were used to store diesel.

The EDR radius map report noted other sites in the 'Ewa region that could not be mapped due to poor or inadequate address information. These sites, and the databases in which they are listed, are shown in Table 4.1.

UNIVERSITY OF HAWAI'I WEST O'AHU FINAL DRAFT ENVIRONMENTAL IMPACT STATEMENT

Table 4.1 – Hazardous Waste Sites			
SITE NAME	LOCATION	DATABASE	
Barbers Point Naval Air Station, NE	Building 829 AST	SHWS, SPILLS	
Barbers Point Naval Air Station, CO	Barbers Point	SHWS, SPILLS	
Barbers Point Navy Public Works Center	Public Works Center	SHWS	
HECO Kahe Generator	89-900 Farrington Highway	SHWS, SPILLS	
Gentry Homes Ltd. Ewa by Gentry	Geiger Road and Kapolei Parkway	RCRIS-SQG, FINDS	
Coral Wastepit	n/a	SHWS	
Jackson Construction Landfill	North End of Hananui Street	CERCLIS, SHWS, FINDS	
Pacific Concrete & Rock Landfill	91-402 Farrington Highway	CERC-NFRAP	
West Loch Fairways Sewage Pump Station	91-126 Haiea Place	UST	
Brewer Chemical Corporation	Kaomi Loop Road	SHWS	
Ewa Sugar Mill Fumigant Storage Area	0.8 miles west of Farrington Highway	CERC-NFRAP	
Ewa Sugar Mill Pesticide Mixing and Loading Site	0.8 miles west of Farrington Highway	CERCLIS, FINDS	
DHHL East Kapolei Affordable Housing Brownfield Project	0.8 miles west of Farrington Highway	<u>Awaiting HEER</u>	
Ewa Sugar Mill Coral Waste Pit	0.2 miles west of Farrington Highway	CERCLIS, FINDS	
KSSK-FM Transmitter Site	165 Palehua Ridge	UST	
200 Puaina Place	200 Puaina Place	SPILLS	
West Beach #2 Wastewater Pump Station	95-159 Waipehe Place	UST	
Hawaii Project Management	Kaomi Loop	SHWS	
Barbers Point Harbor	Barbers Point Harbor	SHWS, SPILLS	
Hawaii Metal Recycling Company	91-056 Hanua Street, Campbell Industrial Park	CERCLIS, RCRIS-SQG, FINDS	
Hanua Street Fugitive Oil	Hanua Street	SHWS, SPILLS	
91-325 Komohana Street	91-325 Komohana Street	ERNS	
Texaco Malakole Street Pipeline Excavation	Malakole Street	SHWS, SPILLS	
Deep Draft Harbor Pier 5 Crude Oil	Pier 5 Barbers Point Deep Draft Harbor	SHWS	
Single Buoy Mooring Barbers Point Harbor	Single Buoy Mooring Barbers Point Harbor	SHWS	
Hawaiian Sugar Planter's Association	Kunia Road	SHWS	
Aeronautical Radio, Inc.	Makakilo on Palehua Ridge	UST	
Palehua Observatory	Palehua Road, Building 210	UST	
Oahu Sugar Company Old Oil Pit Excavation	Off Renton Road	SPILLS	

.... . C:4

ERNS (Emergency Response Notification System) - Information on reported releases of oil and hazardous substances

FINDS (Facility Index System) – Facility information

SPILLS (State or Local Release Notifications) – Releases of hazardous substances to the environment

Anticipated Impacts and Mitigation Measures

Although several recognized environmental conditions have been identified in the 'Ewa region, no sites are located on the UH West O'ahu property. None of the sites identified in the 'Ewa region are listed in the National Priorities List.

The Ewa Sugar Mill Pesticide Mixing and Loading Site is located east of North-South Road, outside of the UH West O'ahu property. According to the DLNR Engineering Division, the site is contaminated with dioxins at elevated levels and has been identified by the State DOH, Hazard Evaluation and Emergency Response Office as a high priority (see Section 8.1).

No industrial facilities handling explosive or fire-prone materials (i.e., liquid propane, gasoline, or other storage tanks) are adjacent to or visible from the UH West O'ahu property. As such, no impacts from hazardous industrial operations are anticipated and no mitigation measures are proposed. Neither people nor buildings will be exposed to explosive or flammable fuels or chemical containers as a result of the project.

Since the majority of the site is currently undeveloped, was historically used for sugarcane cultivation and is cultivated for diversified agriculture, it is not expected that there may be asbestos on site. However, if asbestos is found, the applicant or subsequent developers will contact the Asbestos Abatement Office in the Noise, Radiation and Indoor Air Quality Branch prior to construction/demolition.

4.7 VISUAL RESOURCES AND OPEN SPACE

Existing Conditions

The UH West O'ahu property is relatively flat and was cultivated for sugarcane production for several decades by the Oahu Sugar Company, Ltd. (OSCo). The property is bounded to the north by Farrington Highway, to the west by Kapolei Golf Course and the Villages of Kapolei, to the south by the proposed DHHL Parcel B development, and to the east by vacant land proposed for the North-South Road and its related utility and transit corridor. All of these adjoining lands are similarly flat. Figure 4.1 shows the existing visual conditions at the UH West O'ahu property.

As shown in the *Ewa Development Plan* (Ewa DP), panoramic views of areas to the southeast (including the UH West O'ahu property) are offered from portions of the H-1 Freeway. Significant views and vistas noted in the Ewa DP and offered from the property include:

- Views of na pu'u at Kapolei, Palailai, and Makakilo;
- Mauka and makai views; and
- Views of central Honolulu and Diamond Head.

Due to the flat topography of the project area, no ocean views are offered from the property.



A. View looking mauka toward the proposed University of Hawai'i West O'ahu Campus Site. Note the old cane haul road and scrub vegetation characteristic of the entire subject property.



B. View looking west toward the proposed North/South Road corridor. Note the existing power lines which also illustrate the location of the corridor. This view also depicts the relatively flat topography and undefined drainageways that have caused flooding in the past.



Figure 4.1 Visual Resources University of Hawai'i West O'ahu



NOT TO SCALE




C. View looking makai toward the proposed University of Hawai'i West O'ahu Campus site from H-1 Freeway (180 degree panoramic view)



D. View looking makai toward the proposed University of Hawai'i West O'ahu Campus site from Farrington Highway (180 degree panoramic view)





Visual Resources

University of Hawai'i West O'ahu

East Kapolei, O'ahu



NOT TO SCALE

The most prominent views from within the property are of the Wai'anae Mountain Range. The Ko'olau Mountain Range and Diamond Head are also visible from the project site, as are the 138-kV power lines that run along the site from Farrington Highway toward Renton Road.

The UH West O'ahu property is highly visible from portions of the H-1 Freeway, Farrington Highway, the Villages of Kapolei, and the Kapolei Golf Course. The site would also be visible from the proposed North-South Road and the proposed DHHL East Kapolei Parcel B.

To assess visual resources on O'ahu, the City conducted a comprehensive view shed assessment documented in the *1987 Coastal View Study, City and County of Honolulu Department of Land Utilization.* Existing visual resources for the entire O'ahu coastline are inventoried, prioritized, and documented in this study, which describes the 'Ewa view shed as generally flat terrain absent of predominant land features. Views are decentralized with no particular focus. The only significant roadway view identified by the study is of the makai area, from Farrington Highway and portions of the H-1 Freeway.

Anticipated Impacts and Mitigation Measures

The visual appearance of the UH West O'ahu property will change from vacant scrub and cultivated vegetation to a campus and mixed-use community. Distant views of the shoreline from the H-1 Freeway may be impacted by the proposed development; however, the UH West O'ahu will serve as an important visual landmark for the 'Ewa region. Within the property, the siting of buildings will impact mauka and makai views, as the property is relatively flat. The campus will be oriented to capitalize on views of landforms such as Pu'u Kapolei, Pu'u Palailai, Pu'u Makakilo, the Wai'anae Mountain Range, and Central O'ahu. Views from future internal roadways will be considered to the extent possible. Extensive landscaping, campus view corridors, and thoughtful architectural design will add to the visual character of the area. <u>HECO has yet to determine whether power lines to the substations and power lines extending out from the substations will be overhead or underground. Should HECO decide on implementing overhead power lines, there will be possible visual impacts on the project property.</u>

4.8 SOCIAL CHARACTERISTICS

4.8.1 Population

Existing Conditions

According to the DPP, the population of the *Ewa Development Plan* Area (Ewa DPA) grew from 42,931 in 1990 to 68,718 in 2000, representing an increase of 60.1 percent (DPP 2004). In comparison, the population for the City as a whole increased only 4.8 percent from 836,231 in 1990 to 876,156 in 2000. Residential development in the 'Ewa region continues to increase in population, and the City expects the population of the Ewa DPA to reach

141,864 and 164,462 by 2020 and 2025 as the secondary urban center, respectively. The City's population is expected to reach 1,037,250 and 1,078,050 in 2020 and 2025, respectively.

Within the Ewa DPA, the population of children between the ages of 5 and 17 increased from 8,999 in 1990 to 15,496 in 2000, a 72.2 percent increase in the number of potential college students. Neighboring Central O'ahu and Wai'anae DPAs also saw increases in the number of children between the ages of 5 and 17 over the same time period (11.5 percent and 10.0 percent respectively). In 2000, Central O'ahu had the second highest population of children between the ages of 5 and 17, after the Primary Urban Center, which had 29,416 children between 5 and 17. Younger families tend to buy more affordably-priced housing in the 'Ewa region, and the median age will decrease as newer residents move into the region. This reflects the 'Ewa region's relatively high population growth and results in the need for higher education facilities in the area.

According to the *North South Road and Kapolei Parkway FEA/FONSI*, rapid population growth in the Ewa DPA has balanced the ethnic characteristics of 'Ewa residents to be closer to that of the general population of O'ahu and the State. The median age of residents in selected neighborhoods of the Ewa DPA is slightly lower than the median age of O'ahu residents.

Anticipated Impacts and Mitigation Measures

At full build-out, the UH West O'ahu development will introduce new residents to the 'Ewa region and will provide approximately 4,041 residential units (including 761 student housing units). Using an estimate of 2.9 persons per household, the 3,280 residential units (not including student housing units) will provide homes to approximately 9,512 people. This population increase is consistent with stated governmental policies of directing future growth toward the 'Ewa Plain. Within the property, the proposed elementary school <u>facility(ies)</u> and UH West O'ahu campus will provide for the educational needs of students residing within the property and the surrounding area.

The proposed UH West O'ahu campus will support 8,640 students, faculty, and staff. The campus will provide much needed higher education opportunities (and employment opportunities) for the growing college-aged population of West O'ahu. UH West O'ahu will be a four-year university offering a broad array of educational opportunities for traditional and non-traditional students. With the addition of the first two years of bachelor degree programs, recent high school graduates will be able to enroll at the UH West O'ahu. The average age of students will decrease, and support services provided by the University will likely increase since younger full-time college students tend to expect more campus activities and guidance. The expanded campus will likely attract students who do not want to commute to the UH Mānoa campus, cannot afford local private institutions, or do not want to leave Hawai'i.

To meet the educational needs of school-aged children, a 12-acre elementary school is

proposed within the southern portion of the property (see Figure 2.1). The location of the school would be in close proximity to the proposed residential areas within the property, as well as the proposed DHHL East Kapolei Parcel B residential subdivision directly south of the property. If the DOE requires additional public school facilities, than UHWO may develop a "Lab School" within the campus. Additionally, the DHHL is proposing an elementary school and a separate middle school site on its property immediately east of the UH West O'ahu. A DOE high school site is currently being considered on DHHL land immediately mauka of the UH West O'ahu property, between the H-1 Freeway and Farrington Highway.

4.8.2 Housing

Existing Conditions

Most of O'ahu's new housing stock is planned for development in the 'Ewa and Central O'ahu regions. This trend is expected to continue over the next several decades to accommodate the projected population growth and housing demand. Large-scale housing development projects in the 'Ewa region provide an opportunity for fee-simple home ownership at relatively affordable (and "market") prices. Such projects have been undertaken by the City, the HCDCH, and some private developers to ensure that houses being built are affordable to households of low- and moderate-income and "gap-group" ranges.

Lower interest rates in recent years and the pricing of 'Ewa area housing units have expanded opportunities for home ownership. Consequently, the market seems to be directed toward areas that provide moderately-priced housing rather than the generally more expensive market-priced housing available in the Primary Urban Center (PUC) and East Honolulu.

In 2000, 64 percent of housing units within the Ewa DPA were owner-occupied and 27 percent were renter-occupied, while islandwide percentages were 49 percent owner-occupied and 41 percent renter-occupied. The larger proportion of owner-occupied housing units in the Ewa DPA is attributable to the large number of affordably-priced new single-family residential developments in the region.

Anticipated Impacts and Mitigation Measures

The UH West O'ahu property will occupy approximately 500 acres of land previously planned for residential development as part of the HCDCH East Kapolei Master Plan. The development will provide approximately 4,041 units in a range of housing types listed below and further described in Sections 2.4.2 and 2.4.3:

- UH West O'ahu Lands
 - o 761 student housing units,
 - o 355 work force/affordable multi-family housing units, and
 - o 368 multi-family units in mixed-use parcels;
- Private Development Lands
 - o 365 low-density, single-family units,

- o 561 medium-low-density, multi-family units,
- o 489 medium-density, multi-family units,
- o 925 high-density, multi-family units, and
- o 217 multi-family units in mixed-use parcels.

Park areas and open space will be provided throughout the property in compliance with the City's Park Dedication Ordinance. <u>The University and/or the private developer will meet</u> with affected City departments to discuss specific plans to satisfy the park dedication requirements. To meet the City's affordable housing requirements, approximately 355 units, a large portion of which will be affordable <u>housing units</u>, will be provided within the UH West O'ahu Lands. The remainder of the affordable units required for the development will be distributed throughout residential parcels (and not concentrated in one or two areas) in the Private Development Lands.

The project will address the provision of affordable housing by coordinating with the Department of Planning and Permitting and the Department of Community Services. UHWO is particularly motivated that the project provides workforce housing for faculty and staff as it is difficult to attract the best candidates without it. In addition, UHWO will eventually include approximately 760 student housing units with approximately 3 beds each. Thus, there will be approximately 2,280 less students that would be competing for affordable rentals in the open market. The University and/or the private developer will meet with affected City departments, including the Department of Community Services to discuss specific plans to satisfy the affordable housing requirements. During the public review period, the Department of Planning and Permitting wrote the following regarding the potential of developing "granny flat"-type rental units in proposed housing development: "The DPP considers the potential benefits (i.e. low cost student rentals and increased housing unit affordability) available through this type of housing to be substantial...It appears that an opportunity for this type of unit exists where the "alley loaded" low-density residential development on Parcels C and D where such units could take the form of studio units above the garage. This approach could at least partially fulfill the need for approximately 2,280 beds for the 7,600-student resident population at little or no cost to the University...to construct "granny flats," lots would need to meet underlying residential district standards for two (2) dwellings, or a developer could use the Planned Development Housing (PD-H) option."

With the proposed UH West O'ahu campus project, elementary school <u>facilities</u>, parks, and commercial opportunities within mixed-use areas, residents will be able to live, work, learn, play, and shop within the property.

4.8.3 Lifestyle/Character of the Community

Existing Conditions

Many of the components of a developing secondary urban center have emerged in Kapolei. Kapolei now includes a regional park, a preschool, a private school, a library, a police

station, a State office building, a City office building, a high school, and a middle school. Nearby, the Ko Olina Resort and Marina is a 652-acre major resort with hotel, timeshare, commercial, recreational, and residential uses. The *Ewa Development Plan* projects that by 2020, the City of Kapolei will house more than 7,000 residents and provide work sites for 25,000 private jobs and 5,000 City and State jobs, many of which will be located in the Civic Center and the State office building. Although many people live and work in the Kapolei area, most still commute to work or to the UH Mānoa campus. However, as new employment centers are established, the 'Ewa region will evolve into a self-contained community with less dependence on Honolulu for essential services and employment opportunities. One challenge to attracting businesses to 'Ewa has been that most of the housing inventory provided has been affordably- and moderately-priced housing, rather than executive housing. If more executive housing were available, business owners buying executive housing in 'Ewa would likely relocate their businesses to 'Ewa.

Anticipated Impacts and Mitigation Measures

The regional identity of the project area will change as the UH West O'ahu property changes from agricultural land to a campus community. The UH West O'ahu will be the focus of the 'Ewa region and will help to transform the region from a bedroom community to a second city.

The UH West O'ahu will provide a range of educational and employment opportunities for Leeward and Central O'ahu residents. The proposed campus will support 8,640 students, faculty, and staff, and the proposed elementary school <u>(on non-campus lands)</u> will support 610 students, faculty, and staff. <u>(Additional students, faculty and staff would be associated with a "Lab School" if one is developed on the campus lands)</u>. Additionally, the development will provide approximately 4,041 residential units (including student, market, and affordable housing), with parks and open space throughout the property. Improved and diversified community resources and facilities (i.e., campus libraries, commercial areas, parks, cultural events, and possible non-credit educational classes) will be offered to the community with proposed development.

With the UH West O'ahu, traffic on Farrington Highway in the vicinity of the property will increase. However, with roadway and traffic improvements planned for the area, the future roadway system will accommodate traffic generated by the proposed development (see Section 4.10.1). It is important to note that by providing opportunities for residents to live, work, learn, play, and shop in 'Ewa, the UH West O'ahu may help to reduce island-wide traffic toward the UH Mānoa campus and downtown Honolulu.

4.9 **ECONOMIC CHARACTERISTICS**

Sugar cultivation and processing were the dominant economic activities in the 'Ewa region from the late 1800s to the 1980s. The economic base in 'Ewa shifted in the 1980s and 1990s to urban development of mostly residential and commercial uses.

4.9.1 Employment, Personal Income, and Expenditures

Existing Conditions

Presently, the UH West O'ahu property produces employment and income in the form of short-term revocable agricultural leases to Aloun Farms, Inc. and A.M. Enterprise, Inc.

The City DPP estimates that approximately 64,000 jobs will be provided in the Ewa DPA in 2020. Most of these jobs will occur at existing major employment areas in 'Ewa, including the City of Kapolei, James Campbell Industrial Park, Kapolei Business-Industrial Park, Kalaeloa Barbers Point Harbor and its surrounding industrial land uses, and Ko Olina Resort.

According to the 2000 Census, the median household income for the Ewa DPA was \$60,811 (in 1999 dollars). Incomes were equal to or higher than the median household income for O'ahu (\$51,914), with the exception of the 'Ewa Villages neighborhood. The percentage of residents living below the poverty line was 5 percent for the Ewa DPA, compared to 10 percent for O'ahu. Income differences within the Ewa DPA were greater than that for O'ahu in general.

On O'ahu, annual jobs in construction grew from 16,500 in 1999 to 20, 900 in 2004 (27 percent). The average hourly wage for construction workers in Hawai'i rose from \$21.33 in 1999 to \$27.21 in early 2006 (about 27.5 percent). Continued growth in construction jobs is projected; however, some observers feel that projected growth jobs will not be enough to meet the demand over the next several years due to low unemployment, a depleted pool of construction workers, and the demanding construction work.

Anticipated Impacts and Mitigation Measures

The proposed UH West O'ahu development will generate direct, indirect, and induced jobs (during construction and operation of the project) within the site and throughout the island. Direct jobs are immediately involved with construction of a project or its operations. Direct jobs are not necessarily on-site, as construction supports construction company personnel in offices and base yards. Generally, 20 percent of direct construction jobs are off-site. Indirect jobs are created as businesses directly involved with a project purchase goods and services in the local economy. Induced jobs are created as workers spend their income on goods and services. Indirect and induced jobs are created throughout the State and are likely to be concentrated in commercial and/or industrial centers rather than near a job site.

Construction Employment

Construction of the UH West O'ahu is expected to begin in 2008 and end in 2015. Approximately 14,275 person-years of employment are estimated to be generated by the project. This amounts to about 1,784 jobs in O'ahu over eight years of construction.

Direct construction employment for the UH West O'ahu Lands will generate 6,058 personyears of employment, which will support 8,216 person-years of indirect and induced employment. This amounts to about 757 direct jobs and 1,027 indirect and induced jobs over eight years of construction. Construction employment for the Private Development Lands will generate nearly three times the level generated by the UH West O'ahu Lands.

Workforce income associated with construction will amount to approximately \$803 million in direct wages and \$1,610 million in indirect and induced wages. The total direct, indirect, and induced income associated with construction will be about \$2,413 million.

Operations Employment

UH West O'ahu Lands are estimated to generate approximately 3,451 person-years of direct employment (822 direct jobs) between 2009 and 2015. The Private Development Lands are estimated to generate about 7,965 person-years of direct employment (1,680 direct jobs) between 2009 and 2015. In total, about 26,954 person-years of employment (5,835 direct, indirect, and induced jobs) are expected to be generated by operation of the project between 2009 and 2015.

Payroll for the UH West O'ahu campus is estimated to be as high as \$48.9 million annually (including inflation) by 2015. These wages and salaries will support another \$118.3 million in indirect and induced employment in the same year. The campus will generate about \$683.4 million in operations employment (direct, indirect, and induced jobs) between 2009 and 2015.

Payroll for commercial operations is estimated to reach about \$40.2 million in 2015. These wages will support another \$71.9 million in indirect and induced employment in the same year. In total, commercial operations will create about \$525.9 million between 2009 and 2015.

Labor Supply

The UH West O'ahu project is anticipated to feel the pinch of current labor shortages, but this may change. Contractors have been paying higher wages and increasing benefits to attract workers, and unions have initiated return-to-work programs for expatriate members. Unions have also increased recruiting efforts to attract new workers.

Agricultural Impact

Currently cultivated agricultural lands on the property will be withdrawn from agricultural production for the development of the UH West O'ahu. This will result in some loss in revenues, jobs, or payroll; however, the UH West O'ahu development will provide numerous employment opportunities. No long-term agricultural activity is planned within the Ewa DPA, and additional agricultural land is available on O'ahu. Based on the ample supply of

land suitable for diversified agriculture on O'ahu and the relative lack of market demand (compared to the supply available), no mitigation measures are proposed to replace the lost agricultural production associated with the property.

4.9.2 Economic Factors/Government Revenues

Existing Conditions

Presently, the UH West O'ahu property generates revenue in the form of relatively low lease rent from Aloun Farms, Inc., Kapolei Golf Course, and A.M. Enterprise, Inc. to the State; sales taxes from the sale of produce; and income taxes from employees of the farm.

Anticipated Impacts and Mitigation Measures

The UH West O'ahu property will occupy approximately 500 acres of land previously planned for residential development as part of the HCDCH East Kapolei Master Plan. The project will provide employment, retail, and residential opportunities, thus generating income, excise, and property taxes for the State and City.

The proposed development represents a major commitment of public funds. However, new revenues associated with construction and operation of the development will serve to offset that expenditure. The project will result in substantial contributions to the local government from taxes on direct and induced spending (excise tax), personal income taxes, and corporate income taxes. The State is estimated to receive new revenues totaling \$196.4 million (in 2006 dollars) by 2015.

4.10 INFRASTRUCTURE AND UTILITIES

This section describes the potential impact of the UH West O'ahu development on existing infrastructure. A traffic study for the project was prepared by Parsons Brinckerhoff Quade & Douglas, Inc. (see Appendix H). Engineering Concepts, Inc. prepared an infrastructure study (see Appendix I). Proposed infrastructure improvements are discussed in the following sections.

Existing Infrastructure and Proposed Improvements

On-site Facilities. There are currently no significant on-site infrastructure facilities within the UH West O'ahu property. Farrington Highway is the major transportation access to the property, and the City plans to widen Farrington Highway to two lanes in each direction between the Kapolei Golf Course and Fort Weaver Road. The State Department of Transportation (DOT) is currently constructing the North-South Road and has plans for future construction of a portion of Kapolei Parkway.

On-site infrastructure, including facilities for water transmission and distribution; wastewater

collection; traffic circulation; drainage; and electrical and communication systems, will be required to accommodate the UH West O'ahu development.

Off-site Facilities. Extensive improvements to off-site infrastructure will be required to accommodate 'Ewa and Kapolei as Oahu's Second City. Infrastructure improvements include those to transportation systems, water and wastewater facilities, and electrical/communication systems.

<u>Sustainability</u>

Sustainability guidelines have been established for the UH West O'ahu. The guidelines strive to achieve a Leadership in Energy and Environmental Design (LEED) NC silver standard for campus development and community performance standards based on the LEED NC v2.2 rating system. All new construction and retrofit projects by the University will strive to meet or exceed the target set by the LEED NC v2.2 rating system.

The Private Development Lands will strive to achieve the applicable design criteria in the UH West O'ahu LRDP and the recommended community performance standards. The sustainability guidelines are further discussed in Section 2.6.

In compliance with Chapter 344, HRS "State Environmental Policy" and Chapter 226, HRS "Hawaii State Planning Act," (which is discussed in Section 5.2.5 of this EIS), project buildings, activities, and site grounds will be designed with energy-saving considerations. Energy efficiency goals and requirements of the State will be met by the project, and hot water use on campus will be subject to Administrative Directive No. 98-03, "Policy Governing the Use of Solar Water Heating Systems for State Facilities." The State Department of Business, Economic Development and Tourism (DBEDT) Strategic Industries Division (formerly the Energy, Resources and Technology Division) encourages recycling and the use of recycled-content products (i.e., steel, concrete aggregate fill, drywall, carpet, glass tile). The Honolulu Energy Code will be consulted during the design of the campus, as will HECO for demand-side management programs offering rebates and/or incentives for energy-efficient technologies. Some energy-saving design practices include:

- Establishing the building on an east-west axis;
- Minimizing east- and west-facing glass;
- Using natural ventilation to increase occupant comfort;
- Using daylighting;
- Maximizing the use of natural lighting without heat gain;
- Using high efficiency compact fluorescent lighting;
- Using insulation/radiant barriers for an equivalent R-19 value in ceiling;
- Using ceiling fans;
- Using solar water heating;
- Using landscaping to control dust and minimize heat gain to area; and
- Using photovoltaics, fuel cells, and other renewable energy sources.

Guidelines for Sustainable Building Design in Hawaii: A planner's checklist and A Contractor's Waste Management Guide will be consulted during the design and construction of the project. The project will not conflict with the goals of Act 77, Session Laws of Hawaii 2002, "Relating to Energy Resources".

4.10.1Roadways and Traffic

Existing Conditions

<u>Roadways</u>

The only existing roadway providing access to the property is Farrington Highway. Once completed, the future North-South Road will provide access to the property from the east. Major roadways in the project area are described below and shown in Figure 2.4.

The H-1 Freeway is a 6-lane freeway in the vicinity of the UH West O'ahu property. The Makakilo Interchange is located approximately two miles west of the property, and the Kunia Interchange is located approximately two miles east of the property. The posted speed limit on the H-1 Freeway in the vicinity of the property is 60 miles per hour (mph).

Farrington Highway is a major arterial roadway that provides east-west mobility through the 'Ewa region. It runs along the northwestern boundary of the property as a 2-lane, undivided roadway. Farrington Highway is a 4-lane, divided roadway from the Kapolei Golf Course access road to Kamokila Boulevard in the City of Kapolei. Farrington Highway is also a 4-lane, divided roadway near Fort Weaver Road. The posted speed limit on Farrington Highway in the project area is 35 mph.

Fort Weaver Road/Kunia Road is currently the principal north-south arterial roadway serving the 'Ewa and 'Ewa Beach communities. The roadway is located east of the UH West O'ahu property and is named Fort Weaver Road south of Farrington Highway, and Kunia Road north of Farrington Highway. Fort Weaver Road/Kunia Road is a 6-lane expressway between the H-1 Freeway and Laulanui Street, with interchanges at the H-1 Freeway and Farrington Highway. It is a 4-lane principal arterial from Farrington Highway to the future North-South Road, and a 2-lane minor arterial through the rest of 'Ewa Beach. The posted speed limit on Fort Weaver Road is 45 mph mauka of Geiger Road and reduces to 35 mph makai of Geiger Road.

Fort Barrette Road/Makakilo Drive is a major north-south roadway serving Makakilo and Kapolei. The roadway provides access to the H-1 Freeway and Farrington Highway. Fort Barrette Road/Makakilo Drive is located west of the UH West O'ahu property. Fort Barrette Road is a 2-lane major arterial road with a posted speed limit varying between 25 and 40 mph. The roadway is planned to be widened to four lanes in the near future. Fort Barrette Road becomes Makakilo Drive north of Farrington Highway. Makakilo Drive is a 4-lane divided roadway with a posted speed limit varying between 25 and 40 mph.

The Oahu Regional Transportation Plan (ORTP), prepared for the Oahu Metropolitan Planning Organization (OMPO), identifies roadway concepts necessary to support future development in the 'Ewa Plain. Major roadway improvements identified by the ORTP include completion of the Kapolei Parkway and the North-South Road. The North-South Road is currently under construction.

<u>Public Transit</u>

Oahu Transit Services is contracted by the DTS to operate TheBus on a schedule subject to the availability of resources. In Kapolei, service to Honolulu is provided by buses traveling from Mākaha (Routes C, 40, 93, and 93A), passing on Farrington Highway near the campus. Within the project vicinity, bus service is provided to and from Campbell Industrial Park (Route 413), Makakilo (Routes 411, 412, and 414), the Villages of Kapolei and 'Ewa Beach (Route 41), and the U.S. Veteran's Housing area (Route 415).

The Handi-Van provides para-transit service for semi-ambulatory and non-ambulatory persons with disabilities.

<u>Traffic</u>

A traffic study for the project was prepared by Parsons Brinckerhoff Quade & Douglas, Inc. in May 2006 to identify existing traffic conditions. This report is included in Appendix H.

Automatic traffic recorders were placed along Farrington Highway in the vicinity of the future North-South Road on April 15, 2004. The AM peak hour of traffic occurred between 7:00 AM and 8:00 AM, and the PM peak hour of traffic occurred between 4:30 PM and 5:30 PM, respectively. In the vicinity of the UH West O'ahu property, Farrington Highway operated at <u>what the traffic engineering consultant considers</u> an acceptable Level of Service (LOS) E during both AM and PM peak hours of traffic. The majority of traffic on Farrington Highway in the vicinity of the property is traveling between Kapolei and Waipahu. Regional traffic to and from the Primary Urban Center generally travels on the H-1 Freeway.

Anticipated Impacts and Mitigation Measures

As requested by the Department of Planning and Permitting (DPP) Traffic Review Branch, a comprehensive transportation master plan, which incorporates various modes of travel, including transit, vehicle, bicycle and pedestrian will be developed and included with the zone change and PRU applications to DPP. This plan will be designed to establish and promote a safe and efficient balance between the various modes of travel, such as grade separated bicycle and pedestrian facilities, convenient and centrally located transit stops

and terminals, traffic calming devices and other methods, as necessary.

<u>Roadways</u>

The roadway network in the vicinity of the UH West O'ahu property is expected to change significantly in the future. Phase 1 of North-South Road and its interchange with the H-1 Freeway will be constructed by late 2008. Consistent with the ORTP, the H-1 Freeway and Farrington Highway will be widened. The DOT and DTS have been consulted in the timing of these projects. It is assumed that the portion of Kapolei Parkway between Kapolei Middle School and the North-South Road will be timed to be completed by the Department of Hawaiian Home Lands with the completion of the first phase of North-South Road. Similarly, it assumed that the segment of Kapolei Parkway through 'Ewa Villages will be completed which will allow 'Ewa residents an alternate (to Geiger Road/Roosevelt Avenue) route to Fort Barrette Road. A description of project improvements at particular corridors and a timeframe as to when the anticipated improvements are scheduled to occur will be included in the zone change and PRU applications. UH West O'ahu and its private development partner will be requesting an amendment to Ordinance 02-52 to include widening and improvements of Farrington Highway from two to four lanes between Fort Weaver Road and the Kapolei Golf Course Access Road (including additional street lighting, if warranted), and the construction of the "East-West Connector" (Road F, a new east-west road that extends from Farrington Highway through the UHWO site and beyond, eventually connecting with Fort Weaver Road) when Ordinance 02-52 undergoes its five-year review in 2007.

<u>All surrounding roadways controlled by the City, affected by the proposed project, will be</u> <u>analyzed for adequacy of street lighting illumination per Illuminated Engineering Society of</u> <u>North America standards for roadway lighting.</u>

Access to the UH West O'ahu property is planned via two intersections at Farrington Highway and three intersections at North-South Road. The University is coordinating with the DOT on roadway improvements to these intersections.

<u>HOV lanes on</u> <u>T</u>the H-1 Freeway is <u>are</u> planned to be extended from the Wai'awa Interchange to the Makakilo Interchange, according to the *Oahu Regional Transportation Plan* (2025).

North-South Road is currently being constructed by the DOT as an arterial roadway between the H-1 Freeway and Kapolei Parkway. North-South Road will border the length of the entire eastern side of the UH West O'ahu property and will provide additional access to the H-1 Freeway for the 'Ewa region. North-South Road will also provide sub-regional accessibility for developments in the vicinity of the UH West O'ahu. North-South Road will include three vehicular lanes with paved shoulders in each direction, a 28-foot-wide median that could accommodate and exclusive rapid transit corridor, and sidewalks on both sides (see Figure 2.5). A new interchange connecting North-South Road with the H-1 Freeway is planned to be completed by late 2008.

Farrington Highway, between the Kapolei Golf Course access road and Fort Weaver Road, is planned to be widened from a 2-lane, undivided roadway to a 4-lane, divided roadway. This road widening would make Farrington Highway a continuous 4-lane, divided roadway between Kamokila Boulevard and the Wai'awa Interchange. Approximately 5.5 acres of land within the property will be used for improvements to Farrington Highway. <u>The design of Farrington Highway frontage improvements and access intersections will be coordinated with DDC, and has undergone preliminary review by the DPP Traffic Review and Subdivision branches.</u>

As recommended by the Department of Planning and Permitting Traffic Review Branch (during the public review period), UHWO will incorporate various design elements, such as separated bicycle paths within the Campus and chanellized auxiliary lanes and pedestrian islands at major intersections. A hierarchy of roadways, bikeways, and pedestrian paths will be designed within the property. A loop road will provide the major access to various areas of the campus. A hierarchical network of roadways and a pedestrian and bikeway system will be provided within the property. A loop road encircling the major University buildings will provide access throughout the campus. Seven internal roadways comprising a total of 31.3 acres are planned within the property (see Figure <u>4.2</u> 2.1) and are described below.

Road A will provide access to the property from Farrington Highway. It will be a major entry roadway to the campus and will include bike lanes.

Road B will provide access to the property from North-South Road. Road B will be a major entry roadway to the campus and will include parking lanes and bike lanes. Road B will narrow between the campus entry and the campus drop-off area.

Road C will serve as a secondary access to the campus. It will provide access to the campus from an internal road off of Farrington Highway (Road F). Road C will include parking lanes and bike lanes.

Road D will service the regional roadway network, providing access within the approximately 500-acre property as well as to the surrounding 'Ewa/Kapolei region. Road D will connect Roads B, E, and G, and will include bike lanes.

Road E will provide access to the property from North-South Road, with limited rightin/right-out access to North-South Road. Road E will serve as a secondary access to the campus and will have bike lanes.

Road F will provide access to the property from Farrington Highway and North-South Road. It will service the regional roadway network, providing access within the approximately 500-acre property as well as to the surrounding 'Ewa/Kapolei region. Road F will have a right-of-way width of 104 feet, with four travel lanes (two in each direction), a 16-foot-wide median, bike lanes, 10-foot-wide planting strips, and 6-footwide sidewalks. <u>This road is seen as the western end of the planned "East-West</u> <u>Connector" (connecting Farrington Highway and Fort Weaver Road).</u>

Road G will provide access to the property from the DHHL residential subdivision to the south. Road G will be a collector roadway with parking lanes. <u>This road will provide an alternative north-south route connecting the campus with Kapolei Parkway</u>.

Roads within the UH West Oahu Lands will be controlled by the University, and roads within the Private Development Lands will be dedicated to the City and County of Honolulu. It is proposed that all roads (and water, sewer and drainage systems) within the project site will be dedicated to the City and County of Honolulu, except Road "A" and the portion of Road "B" westward of the intersection with Road "D". During the public review period, the City and County of Honolulu Department of Facility Maintenance wrote: *"Roads, drainage facilities and their related infrastructure should be owned and maintained by the State of Hawaii for this State facility, similarly to that at the Manoa Campus. The City and County of Honolulu will not accept the dedication or maintenance of this facility." This will be identified as an unresolved issue since the Department of Planning and Permitting is specifying roads within the 500-acre project to provide regional connectivity, and the non-campus lands will include residential and commercial uses that will generate property tax revenue to the City and County of Honolulu, it is felt that the non-campus roads should be allowed to be dedicated to the County.*

During the public review period the Department of Transportation Services wrote: *"The design of the project should ensure that TheHandi-Van vehicles have access to the project buildings. Presently, the tallest vehicle is 127 inches high. In addition to facilitating TheHandi-Van vehicle circulation, mobility features should be integrated into the design of public spaces on campus and among the private development land areas to promote accessibility for persons with disabilities..." As TheHandi-Van provides para-transit service for semi-ambulatory and non-ambulatory persons with disabilities, the campus is relatively flat, the campus will be designed to accommodate fire engines within 150 feet of all buildings, and all campus parking areas will include wheelchair accessibility and ADA accessibility to project buildings.*



Figure 4.2 Conceptual Land Use Plan University of Hawai'i West O'ahu



<u>Public Transit</u>

Future transit service near the UH West O'ahu property is expected to increase significantly. The DTS is currently conducting an Alternative Analysis Study for the Honolulu High-Capacity Transit Project (HHCTP), which explores the feasibility of several alternative ways to implement a high-capacity transit line within a 23-mile corridor extending from Kapolei to UH Mānoa. The HHCTP is further discussed in Section 4.11.6.

At this time, two three of the four possible alignments in the Kapolei//Ewa section of the HHCTCP Fixed Guideway Alternative would be located immediately adjacent to the proposed UH West O'ahu, either on Farrington Highway or on North-South Road. Both All three alignments include one two stations at adjacent to the UH West O'ahu campus proposed project site. The alignments that are adjacent to the UH West O'ahu would either be at-grade with limited grade crossings or on elevated structures. The University's preferred option includes an alignment that travels along Farrington Highway and down the North-South Road, next to the UH West O'ahu property. If this alignment is selected, the DTS has indicated that in the vicinity of the University property, the transit corridor would be elevated and located within the median of the North-South Road right-of-way. The DTS has also indicated that two transit stops may be considered for the UH West O'ahu property, one located near the intersection of Farrington Highway and the North-South Road, and the other in the southern portion of the property, near the southern access from the North-South Road. During the public review period, the Department of Planning and Permitting, Development Plan and Zone Change Zone Branch wrote: "...consideration should be given to having a transit station on campus."

The Alternative Analysis Study is also evaluating major upgrades in bus service within the project area. The current bus route can be modified or supplemented to accommodate service to the proposed development, while still servicing the current route. An example route would be to continue east of Farrington Highway after servicing the Villages of Kapolei. It would then stop at the park-and-ride at the corner of the Farrington Highway and North-South Road intersection. After the park-and-ride, the bus route would travel makai on the North-South Road and could then enter the campus through Road B. From the drop-off, the bus would turn makai onto Road D and service the proposed residential and commercial retail developments along Road D and exit using Road E or continue down to Road F and service the Hawaiian Home Lands further makai via Road G. Leaving the property, the bus would turn east on Kapolei Parkway and travel on its current path once it reached Geiger Road. Additionally, University Express routes could access the University via the H-1 Freeway/North-South Road Interchange, down the North-South Road and into the campus property (via Road B) to the drop-off fronting the Kalo'i Plaza and Campus Center.

It is likely that UHWO will be providing a shuttle service on campus as an alternative form of transportation, when there is a critical mass of students, faculty and staff.

<u>Traffic</u>

Project-generated Traffic. At full build-out (year 2015), the proposed UH West O'ahu development is estimated to generate 3,075 vehicular trips (1,578 entering and 1,497 exiting) during the AM peak hour. The development is estimated to generate 4,818 vehicular trips (2,369 entering and 2,449 exiting) during the PM peak hour. <u>All intersections operated at an average no worse than Level of Service⁴ (LOS) D, with or without the UH West O'ahu, during both AM and PM peak hours of traffic.</u> The estimated trip generation was based on *Trip Generation*, 7th edition (Institute of Transportation Engineers, 2003) and the proposed land uses for the UH West O'ahu. Regional travel patterns for vehicles entering and exiting the property were based on the Oahu Metropolitan Planning Organization (OMPO) travel demand model.

Commercial retail trips were assumed to originate from 'Ewa, the Villages of Kapolei, Royal Kunia, Kalaeloa, Makakilo, and Wai'anae. Trips associated with the proposed elementary school were assumed to originate from residential areas within the property and the DHHL subdivision to the south. (Not included was any traffic generated by a "Lab School" if one is developed on the campus lands).

Background Traffic. Background traffic volumes are not directly associated with the proposed development. They include sub-regional traffic on Farrington Highway and the future North-South Road, future trips associated with the DHHL East Kapolei developments makai and east of the UH West O'ahu property, and future trips associated with D.R. Horton's proposed Ho'opili mixed-use development. The projected background traffic volumes for the year 2015 are shown in Figure 8 of the traffic study (see Appendix H).

Total Traffic. Background traffic levels and traffic generated by the UH West O'ahu development are shown in Figure 9 of the traffic study. For this traffic analysis, background traffic was reassigned to utilize new roadway network connections provided by Roads F and G within the property. <u>This traffic analysis also includes the projected traffic from other proposed projects by adjacent landowners including the DHHL East Kapolei I and 2 residential development.</u>

By the year 2015, much of the future roadway network in the project vicinity is assumed to be in place. All intersections analyzed are projected to operate acceptably (LOS D or better) during the peak hours of traffic, with or without the proposed development (see Table 4.2). For this analysis, North-South Road was assumed to be built to its ultimate 6-lane cross-

⁴ The Highway Capacity Manual defines six Levels of Service (LOS), with A indicating the best condition and F indicating the worst condition. The LOS for signalized and unsignalized intersections is defined in terms of average user delays. LOS A indicates little or no delay. LOS B indicates short traffic delays. LOS C indicates average traffic delays. LOS D indicates long traffic delays. LOS E indicates very long traffic delays. LOS F indicates that demand volume exceeds capacity, resulting in extreme delays that may cause severe congestion and may affect other movements at the intersection.

section with left-turn lanes at intersections. The intersections of Farrington Highway with Roads A and F, and the intersections of North-South Road with Roads B and F were expected to be signalized.

The North-South Road and Farrington Highway intersection will be channelized to provide exclusive right-turn lanes and double left-turn lanes on all approaches. The ultimate configuration of North-South Road as a 6-lane, divided roadway is sufficient for the projected 2015 traffic demand with the proposed UH West O'ahu development. The ultimate configuration of Farrington Highway as a 4-lane, divided roadway is also sufficient for the projected 2015 traffic demand with the proposed development. Internal roadways will be designed to accommodate the projected year 2015 traffic demand with the proposed development.

In general, the proposed UH West O'ahu development will increase traffic volumes in the area as the campus population increases and the project achieves build-out. Initial development of the campus will accommodate only 1,520 students, and build-out of the 7,600-student campus will occur over several years. Currently, 23 percent of undergraduate students at the UH Mānoa campus reside on the west side of Red Hill, and traffic on the H-1 Freeway is noticeably heavier when the UH Mānoa campus is in session. Providing an additional campus and employment center in Kapolei for Leeward and Central O'ahu residents could divert enough vehicles to improve eastbound traffic flow on the H-1 Freeway.

	WITHOUT UH WEST O'AHU				WITH UH WEST O'AHU				
INTERSECTION		AM Peak		PM Peak		AM Peak		PM Peak	
		Hour		Hour		Hour		Hour	
		Delay	LOS	Delay	LOS	Delay	LOS	Delay	
North-South Road/H-1 Westbound	П	36.9	C	31.0	П	42.5	П	413	
Off-Ramp Terminus		50.9	C	51.0	U	42.5		41.5	
North-South Road/H-1 Eastbound Off- Ramp Terminus		10.2	В	11.7	В	11.8	В	13.9	
									North-South Road/Farrington Highway
North-South Road/Road B	С	28.5	В	19.0	D	38.9	D	54.9	
North-South Road/Road E		Unsignalized Right-In/Right-				Unsignalized Right-In/Right-			
		Out			Out				
North-South Road/Road F		28.0	С	26.8	D	47.0	D	48.4	
Farrington Highway/Road A	No	No Intersection in Scenario		С	20.4	С	22.4		
Farrington Highway/Road F	No Intersection in Scenario		С	28.2	C	34.6			

Table 4.2 – Intersection Level of Service Summary

Source: Parsons Brinckerhoff Quade & Douglas, Inc., Traffic Study University of Hawaii West Oahu (June 2006)

Phase 1 Traffic. Phase 1 development includes portions of four residential parcels, a mixeduse parcel, a portion of University Village, part of the UH West O'ahu campus, and a detention basin. By Phase 1, North-South Road is anticipated to be constructed as a 6-lane, divided roadway between the H-1 Freeway and Road B, tapering to a 4-lane, undivided roadway makai of Road B. Farrington Highway is assumed to remain as a 2-lane, undivided roadway in the vicinity of the property in Phase 1. Turn movements at the intersection of North-South Road and Farrington Highway are assumed to be channelized with exclusive turning lanes. Roads A and B are assumed to be constructed in their entirety in Phase 1. Road D will only extend from Road B to the Residential Parcel F, and Road F will only be constructed between Farrington Highway and Residential Parcel D (see Figure 2.6).

Phase 1 of the proposed UH West O'ahu development (year 2008) is estimated to generate 665 vehicular trips (367 entering and 298 exiting) during the AM peak hour. The development is estimated to generate 1,040 vehicular trips (492 entering and 548 exiting) during the PM peak hour.

Year 2008 background traffic volumes were estimated based on traffic volumes included in the North-South Road and Kapolei Parkway EA, and the projected traffic generated by the DHHL East Kapolei Parcel B residential subdivision (south of the property), which is anticipated to be completed by late 2007. All intersections assumed to be completed by 2008 were analyzed and are projected to operate acceptably (LOS D or better) during the peak hours of traffic, with or without the proposed development (see Table 4.3).

	2008 WITH UH WEST O'AHU PHASE 1					
INTERSECTION	AM Pe	ak Hour	PM Peak Hour			
	LOS	Delay	LOS	Delay		
North-South Road/H-1 Westbound Off-Ramp Terminus	С	34.4	D	40.1		
North-South Road/H-1 Eastbound Off-Ramp Terminus	А	9.4	В	13.4		
North-South Road/Farrington Highway	D	43.8	D	48.7		
North-South Road/Road B	С	28.6	С	27.6		
North-South Road/Road F	С	26.3	С	27.9		
Farrington Highway/Road A	В	15.5	С	27.6		
Farrington Highway/Road F	В	17.5	С	29.1		

Table 4.3 – Intersection Level of Service Summary (Phase 1)

Source: Parsons Brinckerhoff Quade & Douglas, Inc., Traffic Study University of Hawaii West Oahu (June 2006)

Phase 1 of the North-South Road will be completed by late 2008 and is currently planned to include half of the ultimate roadway cross-section between Kapolei Parkway and Farrington Highway, the ultimate North-South Road and Farrington Highway intersection, the 6-lane cross section between Farrington Highway and the H-1 Freeway, and the ultimate H-1 Freeway Interchange. Based on an evaluation of Phase 1 traffic volumes, the UH West O'ahu traffic study recommended that the ultimate cross section of the North-South Road be extended to Road B before tapering to the half cross section width between Road B and Kapolei Parkway. This would allow an exclusive left-turn lane to service both mauka- and makai-bound left turns into Road B.

During Phase 1 of the UH West O'ahu development, Farrington Highway is assumed to only have two lanes. However, it is assumed that full channelization will be implemented at its intersection with North-South Road to provide exclusive turn lanes on all approaches. The

UH West O'ahu traffic study determined that this configuration is sufficient for the projected Phase 1 traffic demand the project.

Internal roadways to be developed during Phase 1 are also expected to be sufficient for the projected Phase 1 traffic demand the project.

Traffic Demand Management

During the DEIS public review period, State Representative Rida Cabanilla suggested that classes do not commence until after 9:00 a.m.. The University of Hawai'i will study this suggested change in operation and assess the impact it might have on its operations.

<u>Parking</u>

The City's Land Use Ordinance (LUO) establishes a minimum of one parking stall for every ten students, plus one stall for every 400 square feet of office space. Based on this standard, a ratio of one parking stall for every two students was utilized to accommodate the parking needs of students, faculty, and administrative staff. For the initial 1,520-student campus, 760 parking stalls (at 0.5 per student head count) are estimated to be required. The ultimate 7,600-student campus is expected to require 2,850 parking stalls (at 0.375 per student head count). The proposed number of parking stalls exceeds the LUO requirement and the number of stalls suggested by the Institute of Transportation Engineers. The University will continue to work with the City during the Plan Review Use (PRU) permitting process to ensure that the design of future parking facilities adequately meets future needs. Future factors that may affect parking demand are: 1) plans to provide student housing equivalent to 30 percent of the ultimate full time equivalent student population, and 2) plans for rail and bus transit.

For the initial 1,520-student campus, 760 parking stalls (at .5 per student head count) are estimated to be required. This requirement is higher than both the Institute for Traffic Engineers (ITE) standard and the ratio utilized for the ultimate 7,600 student campus. This higher parking requirement was utilized to compensate for the lack of development of alternative modes of transportation and lack of student housing in the immediate vicinity of the campus, which would result in higher automobile use during the initial phase. For the ultimate 7,600 student campus approximately 2,812 parking stalls (at .37 per student headcount) will be provided. This number is equal to the Institute of Traffic Engineers (ITE) suggested parking requirement of 2,812 stalls.

Factors that may impact the estimated parking requirement for the 7,600 student campus include: 1) the level of transit service to the UH West O'ahu campus (less parking required if a significant portion of the UHWO students, faculty and staff use transit), 2) the percentage of distance learning students, 3) the percentage of students enrolled in non-daytime courses, and 4) the amount of nearby and on-campus student housing which is projected to be equal to 30 percent of the total student population. Parking standards could be adjusted, as necessary, to accommodate for any of the factors identified above. For example, parking standards for the

ultimate development of the campus could be reduced should transit access to the area be enhanced through the City's proposed HHCTP. The University will continue to work with the City during the Plan Review Use (PRU) permitting process to ensure that the design of future parking facilities adequately meets future needs.

Table 4.4 shows the initial phase student population and the ultimate student population and the proposed number of parking stalls that is being proposed for each phase of campus development. Table 4.4 also provides the estimated number of parking stalls using methodology provided by the Institute of Traffic Engineers and if one were to compare with the number of stalls provided for a comparable number of students at the UH Mānoa Campus.

-			Q	-
<u>Year</u>	<u>Student</u>	<u>UHWO</u>	<u>ITE</u>	<u>UH Mānoa based</u>
	Population	<u>Stalls</u>		on Head Count
		Provided		<u>(Stalls)</u>
	(Students)	(Stalls)	<u>(Stalls)</u>	
<u>2009</u>	<u>1,520</u>	<u>760</u>	<u>562</u>	<u>464</u>
<u>2015</u>	7,600	2,812	<u>2,812</u>	<u>2,318</u>

Table 4.4 - Proposed UH West O'ahu Parking Stalls

It is possible that the proposed UHWO campus is more likely to have impacts on on-street parking on existing residential neighborhoods nearby. For the initial 1.520 student campus. the project site has adequate land available if each of the students were to take their own cars. In the longer term, the ultimate 7,600-student campus is expected to include 2,812 parking stalls (at 0.37 per student head count). The proposed number of parking stalls equals the number of stalls suggested by the Institute of Transportation Engineers (at 0.37 per student head count). Since the nearest existing residential neighborhoods are located in the Villages of Kapolei and the Villages of Kapolei is separated from UHWO by the Kapolei Golf Course, without any connecting streets, any students who park in the Villages of Kapolei would have to either walk, bike or moped onto either Kapolei Parkway or Farrington Highway to access the campus. The closest proposed residential neighborhoods are the private development lands surrounding the campus and it is hoped that students from these areas will either walk or bike to the campus. It is possible that in the long-term future, if there is inadequate parking provided on the UHWO campus, commuting students will park on-street in the private development lands or on surrounding proposed projects (DHHL East Kapolei 1 - which is immediately makai of the 500 acre UHWO site, and DHHL East Kapolei 2 and the Ho'opili project – which are both across the proposed North-South Road from UHWO).

Development of the UH West O'ahu Campus Lands will comply with the City's requirements for off-street loading, pursuant to Section 21-6.100, LUO.

4.10.2Water Supply Facilities

An infrastructure study for the UH West O'ahu was completed by Engineering Concepts, Inc. in June 2006. This report is included in Appendix I.

Existing Conditions

Potable Water

The UH West O'ahu campus lies over the Pearl Harbor Aquifer and the 'Ewa Caprock Aquifer, and is within the BWS' 440- and 215-foot elevation service zones. Based on discussions with the BWS, water is available to both service zones from the existing 215-foot Kapolei potable water system. Two major water transmission mains (30- and 36-inch) in Farrington Highway provide water to the 215-foot Kapolei and Barbers Point reservoirs via the Honouliuli line booster and Kapolei line booster. A 4-million gallon (MG) reservoir for the 215-foot potable water system is planned to accommodate East Kapolei developments (including portions of the UH West O'ahu property). Currently, there are no existing 440-foot potable water system facilities in the project area and a new system will need to be constructed. Water will be conveyed from the 215-foot system to a proposed 440-foot system reservoir through booster pumps. The UH West O'ahu water system will be part of the East Kapolei regional water system, which will accommodate East Kapolei and the future demand of the Kalaeloa area. The water master plan for East Kapolei has been submitted to and approved by the BWS.

Non-potable Water

The Honouliuli Water Recycling Facility (HWRF) produces R-1 recycled water for irrigation and reverse osmosis (RO) recycled water for industrial uses. R-1 recycled water is the highest quality recycled water, having gone through filtration and disinfection to make it safe for use on lawns, golf courses, parks, and other places that people frequent. The HWRF currently produces 12 million gallons of recycled water per day as a non-potable water source for the 'Ewa area. The BWS plans to extend the existing 215-foot non-potable water system and install irrigation systems along the North-South Road corridor.

Anticipated Impacts and Mitigation Measures

The proposed on-site water system for the UH West O'ahu will be designed to conform to the BWS' Water System Standards (2002) and will be based on guidelines established in the *Ewa Water Master Plan* (prepared by Belt Collins in 1987). The UH West O'ahu water system will be part of the East Kapolei regional water system, which will accommodate East Kapolei and the future demand of the Kalaeloa area. The water master plan for East Kapolei has been submitted to and approved by the BWS. Coordination with the BWS and the DLNR Land Division will take place to incorporate this project into the *Water Use and Development Plan* and the *State Water Projects Plan*.

Projected Water Demand

A dual potable and non-potable water system is planned for the 215-foot elevation service zone. Based on conventional development, the average potable water demand for the UH West O'ahu project is estimated to be 2.584 million gallons per day (MGD) (1.971 MGD from the 440-foot system and 0.613 MGD from the 215-foot system). The average nonpotable water demand for the project is estimated to be 0.324 gallons per day (gpd). The use of non-potable water will be coordinated with the BWS, and non-potable water will be used (to the extent practicable) as permitted by the State Department of Health. The UH West O'ahu has established sustainability guidelines (see Section 2.6), and with the implementation of water-saving measures, water usage will likely be less than that currently estimated.

Required storage for project is estimated to be 0.92 MG for the 215-foot potable water system and 4.55 MG for the 440-foot potable water system. Coordination with the BWS and adjacent developers is necessary for the provision of non-potable water.

The total booster pump capacity needed is estimated at 5,000 gallons per minute (gpm).

Proposed Improvements

The UH West O'ahu property will be served by the 215- and 440-foot potable water systems. The existing 215-foot system will need to be upgraded, and a new 440-foot elevation system will need to be constructed. A dual system (potable and non-potable water) is planned for the 215-foot service zone.

Upgrades to the 215-foot potable water system include the installation of a 4.0-MG reservoir and transmission main in North-South Road. These upgrades will be completed with ongoing developments in the area. The new 440-foot elevation system will include a 5.0-MG reservoir or two 2.5-MG reservoirs and a transmission main in North-South Road. These facilities will be completed concurrently with development of the UH West O'ahu. Construction of the 215-foot non-potable water system to the UH West O'ahu property is expected to be completed by the BWS before completion of the UH West O'ahu. <u>UH West</u> <u>O'ahu and/or a private developer will be required to pay the BWS Water System Facilities</u> <u>Charges for resource development and transmission.</u>

The on-site water system will consist of pipes ranging in size from 8 to 24 inches in diameter, laid out in loops. Loops are designed into water systems to provide more reliable flows and provide adequate pressures. <u>Careful consideration will be taken in the design and operation of the proposed water systems to prevent the cross-connection of these systems and prevent the possibility of backflow of water from the non-potable water system to the potable water system. The two water systems will be clearly labeled and physically separated by air gaps or reduced pressure principle backflow prevention devices to avoid contaminating the potable water supply. Back flow devices will be tested periodically to assure their proper operation</u>

and all non-potable spigots and irrigated areas will be clearly labeled with warning signs to prevent inadvertent consumption of non-potable water. All water system facilities plans will conform to applicable provision of Chapter 11-21, HAR "Cross Connection and Backflow <u>Control."</u>

<u>UH West O'ahu and/or a private developer will be required to install the necessary water</u> system improvements to serve the proposed development. Water lines within dedicable roads are planned to be dedicated to the BWS where practicable. Water lines within the campus will not be dedicated to the BWS, unless water line easements are established for such purposes. The campus is intended to be metered by several master water meters, which will be able to provide the adequate fire flow. A backflow prevention device will be required by the BWS at all meter locations.

Fire Protection

In compliance with Honolulu Fire Department (HFD) requirements, a water system in which all appurtenances, hydrant spacing, and fire flow requirements meet the BWS standards will be provided. Water infrastructure shall be designed and installed in accordance with the Uniform Fire Code, Section 903.2, as amended. Fire hydrants will be spaced throughout the campus within 150 feet of all sides of unsprinklered buildings and 150 feet of the face of sprinklered buildings. All multi-story buildings are assumed to be sprinklered. An HFD access road will be provided within 150 feet of the first floor of the most remote structure. Fire apparatus access roads shall be designed and constructed in accordance with the Uniform Fire Code, Section 902.2.1. The access shall have a minimum vertical clearance of 13 feet 6 inches, be constructed of an all-weather driving surface complying with DTS standards, be capable of supporting the minimum 60,000-pound weight of fire apparatus, and have a gradient not to exceed 20 percent. The unobstructed width of the fire apparatus access road shall meet City requirements. All dead end fire apparatus access roads in excess of 150 feet in length shall be provided with an approved turnaround having a radius complying with DTS standards. Civil drawings will be submitted to HFD for review and approval.

4.10.3Wastewater Facilities

An infrastructure study for the UH West O'ahu was completed by Engineering Concepts, Inc. in June 2006. This report is included in Appendix I.

Existing Conditions

The UH West O'ahu property is within the service area of the Honouliuli Wastewater Treatment Plant (WWTP), which has a treatment capacity of 38 MGD. There are future plans to expand the plant's capacity to 51 MGD. The Makakilo Interceptor Sewer and the recently completed Kapolei Interceptor Sewer currently transport wastewater from existing developments west of the UH West O'ahu property to the WWTP. According to the City

DPP, the capacity of the WWTP is limited by the capacity of the solids handling treatment units, which have a current capacity of approximately 27 to 29 MGD average flow (see Section 8.2). <u>A planned project to add anaerobic digesters, which will increase the solids treatment capacity, is tentatively scheduled to be completed by early 2007. After completion of this project, the overall WWTP solids handling capacity will be 38 MGD.</u>

Currently, there is no sewer service to the UH West O'ahu property. A 30-inch stub-out was provided on the Kapolei Interceptor Sewer for future connection.

The Honouliuli Water Reclamation Facility was purchased by the BWS and provides 12 MGD of recycled water to the West Loch and 'Ewa Villages Golf Course, 'Ewa Mahiko District Park, and Fort Weaver Road.

Anticipated Impacts and Mitigation Measures

The proposed sewer system for the UH West O'ahu has been coordinated with the *Wastewater Master Plan for East Kapolei* (2004), and pipes have been sized to accommodate the ultimate East Kapolei development.

Projected Wastewater Flow

The average wastewater flow for the UH West O'ahu development is projected to be 1.68 MGD.

Proposed Improvements

A portion of the major trunk sewer system for East Kapolei will run through the UH West O'ahu property. The major trunk sewer line will enter the property from Farrington Highway, run along Road D, and exit the property to North-South Road (through Road F). The sewer line, ranging from 21 to 24 inches in diameter, will carry wastewater from the property (and properties located north of the UH West O'ahu property) to the 24-inch trunk line in the North-South Road. The DHHL currently plans to construct the off-site sewer system to the southeast corner of the property at Road F.

Smaller sewer lines ranging from 8 to 12 inches in size will branch off the major trunk line to serve the property. Sewer lines within dedicable roads are planned to be dedicated to the City where practical. Sewer lines within the campus will not be dedicated to the City, unless sewer easements are established for such purposes.

All wastewater plans will conform to applicable provisions of Chapter 11-62, HAR "Wastewater Systems," and the DOH Wastewater Branch reserves the right to review the detailed wastewater plans for conformance to applicable rules. <u>In addition, UH West O'ahu</u> will work with the Board of Water Supply and utilize recycled water for irrigation and other non-potable water purposes in the open spaces and for landscaping areas to the extent

<u>practicable.</u> As recommended by the City Department of Design and Construction, the City DPP will be consulted to determine sewer adequacy, and the City Department of Environmental Services will be consulted to determine the adequacy of the Honouliuli WWTP. During the DEIS Public Review Period, the Department of Planning, Site Development Division, Wastewater Branch wrote that the "...projected average flow of 1.68 mgd is included in the approved Wastewater Master Plan for East Kapolei, dated June 2006. A Site Development Master Application for Sewer Connection is required. This project may also be liable for payment of a Wastewater System Facility Charge."

4.10.4Drainage Facilities

An infrastructure study for the UH West O'ahu was completed by Engineering Concepts, Inc. in June 2006. This report is included in Appendix I.

Existing Conditions

The UH West O'ahu property is located within the middle reaches of the Kalo'i Gulch watershed. Kalo'i Gulch and Hunehune Gulch, which converge within the property, are ephemeral, flowing in response to storm events that are significant enough to generate direct runoff. Kalo'i Gulch enters the property from the east, about 900 feet south of Farrington Highway. Hunehune Gulch enters the property from the north, about 1,200 feet west of the Farrington Highway and proposed North-South Road intersection. Hunehune Gulch enters, converges with Kalo'i Gulch about 1,200 feet north of the southern boundary of the property.

Anticipated Impacts and Mitigation Measures

Design guidelines contained in the City's *Storm Drainage Standards* were used to evaluate the proposed drainage facilities. The *Erosion and Sediment Control Guide for Hawaii* and the *East Kapolei Drainage Master Plan* were used as references UHWO's civil engineering consultant, Engineering Concepts, Inc., used commonly accepted professional civil engineering methodology in determining existing and proposed runoff characteristics. Per comments received by the State Department of Health Clean Water Branch, any discharges related to project construction or operation activities shall comply with the applicable State Water Quality Standards as specified in Chapter 11-54, HAR. Further, the DOH Clean Water Branch wrote that the Hawaii Revised Statues, Subsection 342D-50(a), requires that "[n]o person, including any public body, shall discharge any water pollutants into state waters, or cause or allow any water pollutant to enter state waters except in compliance with this chapter, rules adopted pursuant to this Chapter, or a permit or variance issued by the director."

Projected Peak Flows

Projected peak flows for the UH West O'ahu campus are 849 cubic feet per second (cfs) (10-year storm) and 1,462 cfs (100-year storm).

Proposed Improvements

Runoff will be conveyed to a drainage channel within the 300-foot-wide utility, drainage, and access corridor along the east side of the proposed North-South Road (see Figure 2.5). The channel will divert flow in Kalo'i Gulch away from the UH West O'ahu property and into a detention basin at the downstream end of the channel above the Ewa Villages Golf Course. A proposed box drain system will divert the flow from Hunehune Gulch to the 11.2-acre detention basin proposed at the southern boundary of the property. The flow would then discharge into the regional detention basin through box culverts under the North-South Road. Coordination between the University and its private development partner is required to determine an efficient way of integrating drainage systems.

The proposed drainage system for the campus will consist of grate inlets in parking lots and landscaped areas, curb inlets along roadways, underground pipe/box drains, and a 10-acre detention/water quality basin with a flow control structure. In addition, as suggested by the UH Environmental Center during the Public Review period, UHWO will consider the installation of permeable paving surfaces on parking lots, walkways and courtyards to assist in the management of storm water runoff while increasing infiltration of rain water into the water table. A release structure will be designed to control the discharge rates to the existing peak flows for the 10- and 100-year storm events. A perforated pipe riser will be designed to meet the City-specified releasing time period for water quality control.

An 11.2-acre detention basin is proposed at the southern boundary of the property to accommodate the Private Development Lands. The calculated basin size would need to be approximately 16 feet deep with three horizontal to one vertical side slopes to hold a storm runoff and water quality design volume of 96.76 acre-feet or 156,107 cubic yards. A 100-foot long spillway and six 11-foot wide by 3-foot high concrete box culverts were used in this analysis to limit the detention basin's outflow rate to the pre-existing site runoff rate.

Flood control detention areas will be required until a downstream drainage connection to the Pacific Ocean is established at Oneula Beach Park. <u>According to comments received by Haseko (Ewa), Inc. (HASEKO), an ocean outlet at Oneula Beach Park appears to be the City's preferred regional drainage solution for the Kaloi Gulch Watershed. HASEKO has agreed to process the permits required for this outlet to the ocean with City and State agencies. In this regard, a Final Environmental Impact Statement describing a proposed surface channel over Oneula Beach Park has been accepted by the City (December 2005).</u>

According to the Department of the Army (DA), the upper reaches of Kalo'i Gulch within the project site do not have regulated tributary connection to waters of the U.S. (see Appendix A). As such, a DA permit is not required for the proposed filling of Kalo'i Gulch.

4.10.5Solid Waste Disposal Facilities

Existing Conditions

On O'ahu, most residential and general commercial trash is disposed of at the Honolulu Program of Waste Energy Recovery (H-POWER) facility, the City's waste-to-energy plant located at Campbell Industrial Park. The facility processes over 600,000 tons of solid waste annually, reducing the volume of solid waste going into landfills by 90 percent. Under a purchase power agreement with HECO, the H-POWER facility provides 46 MW of renewable energy that supplies power to between 40,000 and 45,000 homes on O'ahu each day. Ash and non-processibles are transported and buried at the Waimānalo Gulch Landfill. Currently, H-POWER has two boilers and one turbine/generator. A proposed third boiler would enable H-POWER to supply electricity to 20 percent more homes each year.

Waimānalo Gulch Landfill, which opened in 1989, is located approximately five miles northwest of the project site. The land is owned by the City and the landfill is operated by Waste Management, Inc. The site accepts ash and residue from the H-POWER facility, industrial wastes, and non-combustible construction and demolition debris. Commercial haulers pay \$72.75 per ton to dispose solid waste at the facility.

Anticipated Impacts and Mitigation Measures

Solid waste will be generated during construction and operation of various functions within the UH West O'ahu. The amount of waste generated during construction will vary, depending on the construction activity, campus enrollment, number of residences, and amount of commercial space. Construction will conform to the State DOH and the City Department of Public Works program goals and objectives of the Integrated Solid Waste Management Act, Chapter 342G, *Hawaii Revised Statues*. Construction will also comply with the City's approved integrated solid waste management plans in a schedule and time frame satisfactory to the DOH. Special disposal may be required for certain materials used on the campus, depending on the research facilities and laboratories.

Per comments received from the State Office of Environmental Quality Control during the Public Review Period, a construction waste recycling plan will be prepared before construction is initiated. All solid waste generated during project construction shall be directed to a Department of Health permitted solid waste disposal or recycling facility. Also, all highway and road construction improvement projects funded by the State or a county or roadways that are to be accepted by the State or a county as public roads shall utilize a minimum of ten percent crushed glass aggregate as specified by the Department of Transportation in all base-course (treated or untreated) and sub-base when the glass is available to the quarry or contractor at a price no greater than that of the equivalent aggregate.

Recycling shall be encouraged within the project including the reuse and recycling of green

waste generated during construction clearing and grubbing activities, the use of recycled construction and demolition wastes and the use of materials made from recycled products, the use of locally produced compost as available for landscaping, and the provision of space for recycling bins in the detailed design of the community.

At full build-out, the solid waste generated by the project is estimated to average approximately 80,926 pounds per day⁵. It should be noted that this estimate does not account for solid waste that would be recycled, which would be a considerable amount. Sustainability guidelines have been established for the UH West O'ahu Lands (see Section 2.6). The goal for waste management is to appropriately reduce, reuse and recycle materials, to minimize generation of solid waste and achieve diversion from landfills. The Private Development Lands will strive to achieve the applicable design criteria and the recommended LEED community performance standards. Therefore, the increase in solid waste generated by the UH West O'ahu is expected to be accommodated by existing solid waste disposal facilities. <u>UH West O'ahu will promote the optimal use of solid wastes through programs of waste prevention, energy resource recovery, and recycling with the goal that all of its wastes are utilized pursuant to Chapter 344-4.2, HRS.</u>

As some point in the future, the H-POWER plant will need to be expanded or increased recycling efforts will be needed to dispose of the solid waste generated in O'ahu. According to the City, approximately 15 percent of residential solid waste is recyclable newspaper, aluminum, glass, and plastic; and 25 to 30 percent is compostable yard trimming.

4.10.6Electrical Facilities

Existing Conditions

Electricity for the area surrounding the UH West O'ahu property is currently provided by the Hawaiian Electric Company, Inc. (HECO). HECO owns and maintains a pole line along Farrington Highway that supports two 138-kilo-volt (KV) lines and one 12.47-KV line (with provisions for a 46-KV line in the future). The pole line runs along Farrington Highway from HECO's 'Ewa Nui Substation to Palehua Road. <u>Should HECO have existing facilities/easements on the project property, UHWO and/or the private developer will allow continued access to HECO for maintenance purposes.</u> The 12.47-KV line continues along Farrington Highway to Kapolei. Both of the 138-KV lines turn south at Palehua Road and follow the alignment of the future North-South Road to the OR&L right-of-way. Two other 138-KV pole lines from the Kahe Power Plant pass to the north of the UH West O'ahu

Elementary School: 550 students x 2.0 pounds/person/day = **1,100 pounds/day**

 ⁵ University Students: 3,800 students x 2.0 pounds/person/day = 7,600 pounds/day University Faculty and Staff: 1,040 persons x 2.0 pounds/person/day = 2,080 pounds/day On-campus Resident Students: 2,280 students x 3.5 pounds/person/day = 7,980 pounds/day Commercial: 842,886 square feet (at 3 persons/1,000 square feet) = 2,528.66 persons 2,528.66 persons x 3.5 pounds/person/day = 8,850.3 pounds/day Residential: 3,281 units (at 3.25 persons/unit, not including student housing) = 10,663 persons 10,663 persons x 5.0 pounds/person/day = 53,316 pounds/day

property, mauka of the H-1 Freeway. An existing 46-KV circuit that is supported by wood poles follows the alignment of the H-1 Freeway. On the Honolulu-side of Pu'u Makakilo, a spur continues across the H-1 Freeway to the Pacific Concrete and Rock Substation, while the main branch of the pole line continues to parallel the H-1 Freeway alignment and heads toward the 'Ewa Nui Substation. HECO also has a 12.47-KV wood pole line that crosses the southern end of the North-South Road corridor to service the existing 'Ewa Villages Golf Course maintenance facility. There are also two existing 46-KV lines that run along the railroad tracks, connecting several substations.

The existing 138-KV pole at the intersection of Farrington Highway and the proposed North-South Road may need to be relocated with the future widening of Farrington Highway, but the 138-KV pole line will remain.

Anticipated Impacts and Mitigation <u>Measures</u>

Phase 1 of the development is expected to generate an electrical demand of 18 megavolt amperes (MVA) for conventional development (without Energy Star appliances and other energy-saving design measures). At full build-out, the estimated electrical demand is estimated to be 55 MVA. HECO calculations indicate that there is insufficient capacity from the existing 'Ewa Nui, Kamokila, and Fort Weaver substations to serve the projected load. To accommodate a portion of this increase in load, HECO is proposing to construct two new system distribution substations on the UH West O'ahu property. Land for a 1-acre substation site has been set aside at the northern boundary of the property, and another 1-acre substation site will be sited near the southern boundary of the property. <u>To serve the UHWO</u> development and surrounding electrical loads, there will be power lines bringing power to the substations, and power lines to the substations will be overhead or underground. <u>UHWO's preference is for the power lines to be installed underground, subject to PUC</u> approval. The installation of power lines and substations will require approval by the State Public Utilities Commission, as well as other applicable state and city approvals and permits.

HECO is planning to build a 110-megawatt (MW) simple cycle combustion turbine generator power plant in 2009. In addition, some of the projected demand would be offset if the solid waste generated daily by the development were burned at the City's H-POWER facility. An estimated 40.46 tons of solid waste will be produced daily by the development (assuming no recycling), and at 1 MW of electricity per 35 tons of solid waste, approximately 1.16 MW will be generated from the property. Under a purchase power agreement with HECO, the H-POWER facility provides 46 MW of renewable energy that supplies power to between 40,000 and 45,000 homes on O'ahu each day. A proposed third boiler at the facility would enable H-POWER to supply electricity to 20 percent more homes each year.

During the DEIS public review period, the State Department of Business, Economic Development & Tourism (DBEDT), Strategic Industries Division, wrote to call attention to: 1) State energy conservation goals; and 2) energy and resource efficiency and renewable energy and resource development.

DBEDT noted that project buildings, activities, and site grounds should be designed and/or retrofit with energy saving "considerations." DBEDT further noted that the mandate for such consideration is found in Chapter 344, HRS ("State Environmental Policy") and Chapter 226, HRS ("Hawaii State Planning Act"). In particular, DBEDT noted HRS 226 18(c) (4) which includes a State objective of promoting all cost-effective energy conservation through adoption of energy-efficient practices and technologies.

DBEDT noted two important directives affecting State agencies. The first is Administrative Directive No. 06-01, which is also reflected in Act 96, 2006 Session Laws of Hawaii (SLH), which encourages State agencies and programs to increase their leadership commitment to implement innovative and resource efficient operations and management and to design and construct buildings to meet and receive certification for U.S. Green Building Council's Leadership in Energy and Environmental Design (LEED), among others. The other directive DBEDT noted was Act 160, 2006 SLH, relating to the State Budget, which requires that each executive department shall plan for or install energy reduction, energy savings or energyproducing efforts and technologies to lessen electrical consumption or to increase efficiencies in using electrical energy. As stated in Section 2.6 of the Draft EIS:

Sustainability guidelines have been established for the UH West O'ahu. The guidelines strive to achieve a Leadership in Energy and Environmental Design (LEED) NC silver standard for campus development and community performance standards based on the LEED NC v2.2 rating system.

DBEDT recommended that UHWO consult the City and County of Honolulu Energy code early in the project. DBEDT also noted that HECO may also have demand-side management programs that offer rebates for installation of energy-efficient technologies.

Based on DBEDT's recommendations, UHWO's mechanical and electrical consultants, in consultation with UHWO's sustainability consultant, will be directed to review the City and County of Honolulu's Energy code early in the project and to consult with HECO on demand-side management programs that offer rebates for installation of energy-efficient technologies.

While current plans for the development of the UHWO campus includes both energy efficient design and the use of electricity developed and sold by HECO, the site is suited for the use of renewable energy technologies such as wind turbines and photovoltaics.

4.10.7Telephone/Communication Facilities

Existing Conditions

Hawaiian Telcom provides telephone service to the 'Ewa area and owns and maintains a pole line along Farrington Highway. This pole line is substandard; however, Oceanic Time Warner Cable and Pacific Lightnet have an agreement with Hawaiian Telcom for use of its poles and haves attached cables to extend their facilities to Kapolei. AT&T has a fiber cable buried within the southern shoulder of the existing Farrington Highway right-of-way. The Federal government also owns a buried joint tactical support cable within the Farrington Highway right-of-way.

Anticipated Impacts and Mitigation Measures

The proposed development will increase the demand for telephone/communication service. Hawaiian Telcom and Oceanic Time Warner Cable will need to extend their trunking facilities from Farrington Highway to serve the UH West O'ahu development. No significant impacts to existing telephone or cable service are anticipated, and no mitigation measures are proposed for the expansion of existing service. The existing telephone pole line along Farrington Highway is substandard, and Hawaiian Telcom, Oceanic Time Warner Cable, and Pacific Lightnet will have to relocate their lines to new poles along Farrington Highway in the future.

4.11 **PUBLIC SERVICES AND FACILITIES**

4.11.1Educational Facilities

Existing Conditions

Public schools in the vicinity of the UH West O'ahu property include Kapolei High School and Kapolei Middle School along Kapolei Parkway (to the south). The 2004 to 2005 school year enrollment and the 2005 to 2006 Fall enrollment for these schools as well as others in the project area are listed in Table 4.54. It should be noted that Fall enrollment is generally higher than school year enrollment.

Table $43\underline{3}4 = 1$ done sensor Enrolment							
	2004-2005 SCHOOL YEAR	2005-2006 SCHOOL YEAR					
SCHOOL NAME	ENROLLMENT	FALL ENROLLMENT					
Barbers Point Elementary	413	529					
Campbell High School	1,837	2,283					
'Ewa Elementary	791	933					
'Ewa Beach Elementary	557	665					
Holomua Elementary	1,302	1,442					
Ilima Intermediate	1,155	1,201					

Table 4.54 – Public School Enrollment

SCHOOL NAME	2004-2005 SCHOOL YEAR ENROLLMENT	2005-2006 SCHOOL YEAR FALL ENROLLMENT
Kaimiloa Elementary	600	679
Kapolei Elementary	1,043	1,126
Kapolei Middle School	1,493	1,580
Kapolei High School	1,872	2,333
Makakilo Elementary	460	509
Mauka Lani Elementary	503	577
Pohakea Elementary	497	551

Source: State Department of Education, School Status and Improvement Report School Year 2004-05 (June 2006) State Department of Education, 2005-06 Official Enrollment (September 2005)

In its comment letter on the EISPN, the State Department of Education (DOE) provided a table of possible public schools for East Kapolei students. This table has been reproduced and included as Table 4.<u>6</u>-5.

School	Capacity 2003	Active Enrollment 2003	Active Enrollment 2004	Additional Students 2004	Projected Enrollment 2005	Projected Enrollment 2007	Projected Enrollment 2009	2009 Enrollment Less 2003 Capacity		
Middle and	Middle and High Schools – Kapolei Complex									
Kapolei High	1,853	1,928	2,162	234	2,412	2,668	2,712	859		
Kapolei Middle	1,387	1,698	1,699	1	1,704	1,730	1,715	328		
Elementary	Elementary Schools – Kapolei Complex									
Barbers Point	636	381	510	129	422	445	426	-210		
Kapolei	1,198	1,165	1,187	22	1,228	1,240	1,246	48		
Elementary Schools – Campbell Complex										
'Ewa	659	850	896	46	863	899	893	234		
Holomua	1,184	1,344	1,437	93	1,424	1,462	1,457	273		
Iroquois Point	878	455	485	30	409	367	343	-535		
Kaimiloa	740	691	686	-5	633	621	618	-122		
Pohakea	635	563	560	-3	578	656	826	191		
Ocean Pointe	650					462	674	24		

Table 4.65 – Possible Public Schools for East Kapolei Students

Source: State Department of Education, EISPN Comment Letter (March 2005)

The West O'ahu area has a higher proportion of high school graduates than the rest of the City. It also has a lower share of citizens with baccalaureate degrees. This suggests an issue of access to higher education facilities, as there is presently available capacity at the UH Mānoa campus. The UH Mānoa and UH Community Colleges currently has ve a smaller enrollment than they did several years ago. However, these campuses should not be expected to accommodate additional enrollment without additional resources. Budget cuts have reduced the lecturer and faculty resources that were available several years ago when

enrollments were higher.

Anticipated Impacts and Mitigation Measures

The UH West O'ahu will likely increase the number of students enrolled in public schools. Based on the table provided by the DOE (Table 4.<u>65</u>), in 2009, three elementary schools (Barbers Point, Iroquois Point, and Kaimiloa) will be under capacity by approximately 867 students, and Kapolei Middle School and Kapolei High School will be over capacity by approximately 328 and 859 students, respectively. During the public review period the DOE wrote that the data provided in the DEIS illustrates the growing enrollment in the schools and the projected number of students that will grow beyond the capacity of the existing schools' classrooms. The DOE estimated that when the UH project is mature, that there will be approximately 1,771 public school students living within the project and that would be enough elementary school students to fill one elementary school.

<u>UH has held preliminary discussions with DOE to address the demand of future school-aged</u> <u>children that would be generated by the proposed project.</u> To help meet the demand for public educational facilities in the area, a 12-acre elementary school is proposed within the southern portion of the UH West O'ahu property (see Figure 2.1). The location of the school would be in close proximity to the proposed residential areas within the property and the planned DHHL residential subdivision directly south of the property. The school is projected to accommodate <u>a typical DOE elementary school for</u> 550 students and 60 faculty and staff. <u>During the public review period, the DOE also wrote that it estimated that a school fair-share</u> <u>contribution for the proposed project would include sufficient acreage for an elementary</u> <u>school and a cash contribution. In addition, DOE wrote that it "recognizes UH's</u> <u>acknowledgement of the need for a contribution to offset the impacts of the proposed project</u> <u>and we look forward to working together." As such, one possibility is the development of a</u> <u>"Lab School" on the campus site.</u>

During the public review period, the DTS wrote that the proposed elementary school should be situated where potential pedestrian conflicts with vehicles will be minimized. DTS stated a preference that pathways and/or sidewalks do not traverse major roadways. As such, UHWO will continue to coordinate with DOE, DTS, DPP and other appropriate agencies so that an agreeable access plan can be formulated.

The DHHL is proposing an elementary school and a separate middle school site on its property immediately east of the UH West O'ahu. The DOE is contemplating a high school on DHHL land immediately north of the UH West O'ahu and has been in discussion with developers for additional school sites in East Kapolei.

Additionally, school-aged children living in the UH West O'ahu development may attend private schools in the 'Ewa and Kapolei area, including:

- Friendship Christian Schools (preschool and kindergarten to 12th grade);
- Island Pacific Academy (kindergarten to 12th grade);
 Lanakila Baptist Schools High School (7th grade to 12th grade);
 Messiah Lutheran School (kindergarten to 8th grade); and
- Our Lady of Perpetual Help School (kindergarten to 8th grade). •

The UH West O'ahu campus will help to meet the demand for higher education facilities in the 'Ewa region. Residents of West O'ahu would have the opportunity to obtain postsecondary degrees without commuting to other areas. Available demographic data and the enrollments at nearby public schools reflect the number of college-aged students living in the region, as well as the overall growth of housing units and population in this area. Demographic data also indicate that the existing UH Mānoa campus and UH Community College system may not be able to accommodate projected undergraduate enrollment if the UH West O'ahu campus is not built.

4.11.2Police Protection

Existing Conditions

The proposed UH West O'ahu campus falls within District 8 of the City and County of Honolulu Police Department (HPD). This police district includes the Wai'anae Coast, Makakilo, the 'Ewa Plain, and the City of Kapolei (approximately 128 square miles and approximately 35 miles of coastline), and has about 100 field officers in 18 beats. Response time for the entire district ranges between five and seven minutes. In 2000, the Kapolei District Station was opened at 1100 Kamokila Boulevard to meet the growing needs of the 'Ewa Plain communities. A storefront station is located at the intersection of Fort Weaver Road and Renton Road. The only district substation is located in Wai'anae.

Anticipated Impacts and Mitigation Measures

After reviewing the EISPN, the HPD stated, "we believe that calls for service to the area will be impacted and that there will be vehicular and pedestrian traffic concerns" (see Section 8.2). With development of the UH West O'ahu, there will be an occasional and unavoidable increase in demand for police service. On-site security would be provided for the campus; however, officers at the Regional Kapolei Police Station would respond to calls for police service. These occurrences are not expected to significantly impact the police protection provided to the rest of the community. Vehicular and pedestrian traffic facilities will be designed to avoid conflicts. According to the Ewa Development Plan, the 'Ewa Villages Substation is planned to service the East 'Ewa region, including the UH West O'ahu. The service date for this substation has not been determined.
4.11.3Fire Protection

Existing Conditions

Fire protection in the 'Ewa area is provided by the City and County of Honolulu Fire Department (HFD) 'Ewa Beach Fire Station (engine company), Makakilo Fire Station (engine company), and Kapolei Fire Station (engine and ladder company). Kapolei Fire Station, previously known as Campbell Industrial Park Fire Station, is the Battalion 4 Headquarters located in Kapolei Business Park.

Portions of the UH West O'ahu property are comprised of vacant scrub vegetation, which, combined with the low rainfall characteristic of the 'Ewa region, creates a potential fire hazard.

Anticipated Impacts and Mitigation Measures

Urban structures and landscaping for the UH West O'ahu will eliminate the potential fire hazard posed by the existing scrub vegetation. Although all buildings will be equipped with modern fire control devices and access for fire apparatus, water supply, and building construction will conform to existing codes and standards, an occasional and unavoidable increase in demand for fire protection services is likely to result. According to the *Ewa Development Plan*, to meet the projected population and economic growth in 'Ewa by 2020, three fire stations are planned at 'Ewa Villages, Ko Olina, and Makaīwa Hills, but service dates have not been determined. The UH West O'ahu would be serviced by the proposed 'Ewa Villages Fire Station in Tenney Village. As the population of 'Ewa grows and the planned fire stations are established, the UH West O'ahu development will be adequately protected from the unavoidable occurrence of fire and as such, no significant impacts to fire protection facilities or services are expected to result from the development.

As required by the HFD, a private water system in which all appurtenances, hydrant spacing, and fire flow requirements meet BWS standards will be provided. <u>Water infrastructure shall</u> be designed and installed in accordance with the Uniform Fire Code, Section 903.2, as <u>amended.</u> A fire department access road within 150 feet of the first floor of the most remote structure will also be provided. <u>Fire apparatus access roads shall be designed and constructed in accordance with the Uniform Fire Code, Section 902.2.1</u>. This access will have a minimum vertical clearance of 13 feet and 6 inches, be constructed of an all-weather driving surface that complies with the City DTS standards, be capable of supporting the minimum 60,000-pound weight of fire apparatus, and be of a gradient not exceeding 20 percent. The unobstructed width of the access road will meet City requirements, and all dead-end fire apparatus access roads in excess of 150 feet in length will be provided with an approved turnaround having a radius complying with DTS standards. Fire hydrants will be spaced throughout the campus within 150 feet of all sides of unsprinklered buildings and 150 feet of the face of sprinklered buildings. All multi-story buildings <u>are</u> assumed to be sprinklered. In addition, civil drawing<u>s</u> will be submitted to the HFD for review and

approval.

4.11.4Medical Facilities

Existing Conditions

St. Francis Medical Center – West is the nearest hospital facility to the UH West O'ahu property. Ambulance service is coordinated with the City, and the hospital is equipped with a helipad. St. Francis Medical Center – West offers general hospital services, including emergency care, outpatient care, lab and imaging services, and medical offices. The hospital has 79 licensed beds available, and bed capacity will soon be increased to 84 beds. The hospital is operating at about 80 percent occupancy and has space available for a total of 136 beds. Emergency medical and surgical services are also be provided at Pali Momi Medical Center (116 beds) in 'Aiea and Wahiawa General Hospital (162 beds, of which 93 are for long-term care).

The HFD Emergency Medical Services (EMS) Division staff and trucks are located at the Wai'anae Fire Station. New EMS units have recently been established in Nanakuli and Kapolei, and are designated as Advanced Life Support units.

Non-emergency medical services are offered at the Kaiser Permanente Punawai Clinic in Waipahu and major hospital facilities in urban Honolulu, an approximately 30-minute drive from the property. Additionally, Kapolei Medical Park, located across Kapolei Shopping Center at the corner of Farrington Highway and Fort Barrette Road, opened in 2000. The 50,000-square-foot facility provides rental space for tenants including Ambulatory Services, Inc., Hawaii Medical Services Association, Kaiser Permanente, and Straub Kapolei Family Health Center.

Anticipated Impacts and Mitigation Measures

There will be an occasional and unavoidable demand for emergency medical services by students, faculty, staff, or visitors within the UH West O'ahu development. However, it is unlikely that this demand will impact the level of service provided to other O'ahu residents. With planned transportation improvements in the area (i.e., construction of North-South Road, extension of Kapolei Parkway, and widening of Farrington Highway), adequate access would be provided to the UH West O'ahu property. Existing medical and healthcare facilities within 'Ewa and nearby should be able to accommodate the anticipated increase in demand.

4.11.5Recreational Facilities

Existing Conditions

Public recreational facilities in the 'Ewa area include regional parks, community parks, neighborhood parks, beach/shoreline parks, and golf courses. Regional parks are large recreational complexes. According to the City Department of Parks and Recreation (DPR),

community parks serve an approximate population of 10,000 people and normally include play fields, courts, and a recreation building. Neighborhood parks serve an approximate population of 5,000 people and normally include play fields, courts, and a comfort station. Beach/shoreline parks are day-use parks primarily for swimming, sunbathing, and picnicking. Table 4.<u>76</u> shows existing and planned parks in the 'Ewa area. The Kapolei Golf Course is located adjacent to and west of the proposed UH West O'ahu property. The 'Ewa Villages Golf Course is located east of the property, and the Coral Creek Golf Course is located near Kalaeloa.

NAME	SIZE (ACRES)
Barbers Point Beach Park	7.39
'Ewa Beach Community Park	13.25
'Ewa Beach Park	4.88
'Ewa Mahiko Neighborhood Park	6.33
Kahe Point Beach Park	4.47
Kamokila Park	5.89
Kapolei Community Park	12.00
Kapolei Regional Park	69.39
Makakilo Community Park	8.50
Makakilo Neighborhood Park	4.01
Mauka Lani Neighborhood Park	4.40
Oneula Beach Park	30.00
Pu'uloa Neighborhood Park	4.34
West Beach Shoreline Park, North	10.00
West Beach Shoreline Park, South	18.26

Table	4.76 -	'Fwa	Parks
Table	T •/ U =	Lvva	ιαικэ

Source: State Comprehensive Outdoor Recreation Plan (May 2003), State Department of Land and Natural Resources

Anticipated Impacts and Mitigation Measures

The demand for recreational facilities in 'Ewa will increase as the population grows. The proposed UH West O'ahu development will provide additional recreational facilities in compliance with the City's Park Dedication Ordinance. <u>The University and/or the private developer will meet with affected City departments to discuss specific plans to satisfy the park dedication requirements, as required by Section 22-7, Revised Ordinances of Honolulu.</u> New recreational facilities within the proposed campus are planned for use primarily by students, faculty, and staff. Such facilities could include tennis courts, basketball/volleyball courts, and jogging paths.

Existing regional park facilities in West O'ahu, Central O'ahu, and Waipi'o may be visited by new residents of the development; however, other park facilities are planned in 'Ewa. The DHHL residential subdivision adjacent to and south of the UH West O'ahu property proposes 4.5 acres for park use and 10.7 acres for a community center to be developed and operated by the Salvation Army. The proposed center could include a swimming pool, performing arts center, child care center, gymnasium, education center, game and recreation areas, and other

multi-use spaces, which would be available to the general public. Additional recreational facilities for the 'Ewa region are planned in Kalaeloa and Ho'opili (D.R. Horton's project located east of the UH West O'ahu property).

4.11.6Public Transit

Existing Conditions

Oahu Transit Services is contracted by the DTS to operate TheBus on a schedule subject to the availability of resources. In Kapolei, service to Honolulu is provided by buses traveling from Mākaha (Routes C, 40, 93, and 93A), passing on Farrington Highway near the campus. Within the project vicinity, bus service is provided to and from Campbell Industrial Park (Route 413), Makakilo (Routes 411, 412, and 414), the Villages of Kapolei and 'Ewa Beach (Route 41), and the U.S. Veteran's Housing area (Route 415).

According to the *Ewa Development Plan*, a rapid transit corridor is planned to connect the City of Kapolei to Waipahu and onward to the Primary Urban Center of Honolulu. Conceptually, the transit corridor would extend from the City of Kapolei along Kapolei Parkway (and its proposed extension) and up the proposed North-South Road, turning eastward along Farrington Highway to Waipahu. Two transit nodes are shown in the vicinity of the UH West O'ahu property (see Figure 4.3).

The DTS' Honolulu High-Capacity Transit Corridor Project (HHCTCP) is presently evaluating transit alternatives for the 23-mile long corridor between Kapolei and UH Mānoa. Four alternatives are being analyzed (DTS 2006):

- Alternative 1. The No-Build Alternative includes existing bus and highway facilities and other committed transportation projects identified in the *Draft Oahu Regional Transportation Plan* (2030) by the Oahu Metropolitan Planning Organization.
- Alternative 2. The Transportation System Management Alternative provides an enhanced bus system based on a hub-and-spoke route network.
- Alternative 3. The Managed Lanes Alternative involves construction of a new twolane viaduct starting at the H-1 Wai'awa Interchange and ending at Pacific Street in lwilei. An additional access point to enter and exit the Managed Lanes would be provided near Aloha Stadium. Buses, high-occupancy vehicles (with two or three or more passengers), and toll-paying single-occupant vehicles would be allowed to use the Managed Lanes.
- Alternative 4A. The Fixed Guideway Alternative is a high-capacity transit line with a Kamokila Boulevard / Salt Lake Boulevard / King Street / Hotel Street / Kapi'olani Boulevard alignment.
- Alternative 4B. The Fixed Guideway Alternative is a high-capacity transit line with a North-South Road / Camp Catlin Road / King Street / Queen Street / Kapi'olani Boulevard alignment.
- Alternative 4C. The Fixed Guideway Alternative is a high-capacity transit line with a Fort Weaver Road / Farrington Highway / Kamehameha Highway / Dillingham



Source: City & County of Honolulu Department of Planning and Permitting

Figure 4.3 Ewa Development Plan Public Facilities Map **University of Hawai'i West O'ahu**





PBR

Boulevard / Kaʻaʻahi Street / Beretania Street / King Street / Kaialiu Street alignment.

• Alternative 4D. The Fixed Guideway Alternative is a high-capacity transit line with a North-South Road / Farrington Highway / Airport / Dillingham Boulevard / Hotel Street / Kapi'olani Boulevard with a Waikīkī spur alignment.

By November 1, 2006. The results of the alternatives analysis will be have been submitted to the Honolulu City Council for its use in selecting the Locally Preferred Alternative (LPA). It is anticipated that the City Council will make this decision prior to January 1, 2007. After this decision is made, the City will prepare the environmental disclosure document for the LPA.

At this time, two three of the four possible alignments in the Kapolei/'Ewa section of the HHCTCP Fixed Guideway Alternative would be located immediately adjacent to the proposed UH West O'ahu, either on Farrington Highway or on North-South Road. Both <u>All</u> three alignments include one two stations at adjacent to the UH West O'ahu campus proposed project site.

The Fixed Guideway Alternative alignments that are adjacent to the UH West O'ahu would either be at-grade with limited grade crossings or on elevated structures. The fixed guideway could be as narrow as 25 feet wide. If the structures are elevated, they would be supported by columns about 30 feet tall.

At this time, no decision has been made regarding the technology that would be used for the Fixed Guideway Alternative. This decision will be made at a later stage of project development, subsequent to the decision on the LPA. Should the City Council select the Fixed Guideway Alternative that follows either Farrington Highway or North-South Road adjacent to the UH West O'ahu campus, continued coordination of the two projects will be integral. According to the HHCTC website, the near-term schedule includes selection of a locally preferred alternative (LPA) by the City Council before the end of 2006. Following the Council's decision, the City will prepare a draft EIS and request approval from Federal Transit Administration to begin preliminary engineering (project design). The earliest that construction could begin on a selected alternative would be 2009. Due to the size and cost of the overall project, it is likely to be built in several phases lasting several years.

Anticipated Impacts and Mitigation Measures

As a result of the UH West O'ahu development, traffic on local, collector, and regional transportation systems will increase as students, faculty, and staff commute to the proposed campus on a daily basis. However, the UH West O'ahu property is bounded to the north by Farrington Highway and to the east by the proposed North-South Road, near planned transit nodes and corridors. As such, the campus is being planned to be transit-ready and the use of public transportation systems will be encouraged. With current plans to include UH Mānoa and UHWO at or near the end of each rail transit alignment alternative, the development of the UH West O'ahu campus is expected to provide greater ridership numbers, than if the site was developed as originally planned (mostly lower-density residential). Ridership is not

expected just from students, but from faculty and staff living outside of the project area. In addition, traffic conditions in the project vicinity are expected to improve with the future completion of Kapolei Parkway and construction of North-South Road, which will connect the areas between the Kapolei Parkway and Farrington Highway to the H-1 Freeway.

Proposed residential units and bicycle facilities within the approximately 500-acre property will encourage walking and bicycling within the area. This will help to reduce vehicular traffic in the vicinity of the campus. <u>Ridership on rail transit would be especially attractive if the mode of transit could accommodate temporary bicycle storage, allowing riders to carry their bicycles with them.</u>

The UH West O'ahu campus will be a higher education facility and major employment center. West O'ahu residents will have the opportunity to enroll in University courses and work near their homes. As such, traffic conditions into and within Honolulu are also expected to improve, as fewer people would commute to the UH Mānoa, other UH Community Colleges, and workplaces outside of West O'ahu. This opportunity will help to re-orient travel to private Universities, UH Mānoa, and workplaces outside of West O'ahu to UH West O'ahu. As such, UH West O'ahu would help to manage the growth in vehicular travel on routes into and within Honolulu and would result in better traffic conditions than if it did not occur.

4.11.7Community Services

Existing Conditions

Community services and public facilities in the vicinity of the UH West O'ahu include:

- Schools (with Kapolei Elementary, Middle, and High Schools nearby) [see Section 4.11.1];
- Kapolei Police Station [see Section 4.11.2];
- Three fire stations in the 'Ewa region ('Ewa Beach, Kapolei, and Makakilo) [see Section 4.11.3];
- Churches;
- Shopping centers in Kapolei and Makakilo;
- Public libraries in Kapolei and 'Ewa Beach;
- Post offices in Kapolei and 'Ewa;
- Hospitals/healthcare facilities (St. Francis Medical Center West and Kapolei Medical Park) [Section 4.11.4];
- Recreational facilities [see Section 4.11.5]; and
- Public transit facilities [see Section 4.11.6].

Anticipated Impacts and Mitigation Measures

The proposed UH West O'ahu development is not expected to have any adverse impact on community services in the area. Anticipated impacts on public schools, HPD and HFD

facilities, hospitals and healthcare facilities, recreational facilities, and public transportation facilities were previously discussed in this chapter. Mitigation measures were also identified to minimize potential impacts.

The UH West O'ahu will offer a variety of programs and services to the West O'ahu community. Adult education and continuing education classes will be offered, and cultural activities and programs may also be developed and offered to the community. The proposed development will provide numerous opportunities for education, employment, recreation, and housing.

(This page intentionally left blank.)

5.0 Relationship to Land Use Plans, Policies, and Controls

5.0 RELATIONSHIP TO LAND USE PLANS, POLICIES, AND CONTROLS

5.1 FEDERAL

Abutilon menziesii. *Abutilon menziesii,* or ko'oloa'ula, is a federally listed endangered species that was discovered on the property during a botanical survey conducted in 1996 for the HCDCH East Kapolei EIS. A recent survey of the UH West O'ahu property (Char 2003) identified a few individual plants along the southeastern boundary of the property. Seeds may also be present in the soil where a single plant once lived. *Abutilon menziesii* has been federally listed as an endangered species since 1986 and is now protected under the provisions of the Federal Endangered Species Act of 1973, as amended. In accordance with both State and Federal regulations, mitigation measures have been identified in the HCP prepared in consultation with the DLNR (see Appendix C). The University is currently working with the DOT to be included in the Certificate of Inclusion for the Incidental Take License.

Americans with Disabilities Act. The Americans with Disabilities Act (ADA) of 1990 establishes guidelines for accessibility to buildings and facilities by individuals with disabilities. To the extent required by regulations issued by federal agencies, these guidelines will be applied to the UH West O'ahu Lands during the design and construction of new facilities covered by the ADA. It should be noted that the site is relatively flat and thus is naturally conducive to accessibility. <u>The design of public spaces on campus and among the private development land areas will be integrated to promote accessibility for persons with disabilities.</u>

National Pollution Discharge Elimination System. A National Pollution Discharge Elimination System (NPDES) permit, which is administered by the State DOH, will be required for this development.

5.2 STATE OF HAWAII

5.2.1 Environmental Impact Statement Law, Chapter 343, Hawaii Revised Statutes

This EIS has been prepared pursuant to Chapter 343, *Hawaii Revised Statutes* (HRS) and Subchapter 4, Section 11-200-4, *Hawaii Administrative Rules* (HAR), which states that, "the governor, or an authorized representative, whenever an action proposes the use of state lands or the use of state funds, or, whenever a state agency proposes an action within section 11-200-6(b)" shall be the final authority to accept an environmental impact statement. Since the proposed project will require the use of state lands and funds, the Governor of the State of

Hawai'i is the accepting authority for the University of Hawai'i West O'ahu Environmental Impact Statement.

Various agencies and individuals were consulted during the Site Selection Study and the preconsultation phase for this EIS. Comment letters on the EISPN and responses are included in Section 8.2. A 45-day review period will commence upon publication of this Draft EIS in the Office of Environmental Quality Control (OEQC) *The Environmental Notice*. Comments on this Draft EIS and applicable responses will be provided in the Final EIS.

5.2.2 State Environmental Policy, Chapter 344, Hawaii Revised Statutes

Chapter 344, *Hawaii Revised Statutes* (HRS) establishes a State policy to encourage productive and enjoyable harmony between people and their environment, promote efforts that prevent or eliminate damage to the environment and biosphere and stimulate the health and welfare of humanity, and enrich the understanding of ecological systems and natural resources important to the people of Hawai'i. Guidelines of the policy that are applicable to the UH West O'ahu development are discussed below.

LAND, WATER, MINERAL, VISUAL, AIR, AND OTHER NATURAL RESOURCES

- (A) Encourage management practices which conserve and fully utilize all natural resources.
- (B) Promote irrigation and waste water management practices which conserve and fully utilize vital water resources.
- (*C*) Promote the recycling of waste water.
- (G) Promote the optimal use of solid wastes through programs of waste prevention, energy resource recovery, and recycling so that all our wastes become utilized.

DISCUSSION: Sustainability guidelines have been established for the UH West O'ahu. The guidelines strive to achieve a Leadership in Energy and Environmental Design (LEED) NC silver standard for campus development and community performance standards based on the LEED NC v2.2 rating system. The Private Development Lands will also strive to achieve the applicable design criteria in the UH West O'ahu LRDP and the recommended community performance standards.

FLORA AND FAUNA

- (A) Protect endangered species of indigenous plants and animals and introduce new plants or animals only upon assurance of negligible ecological hazard.
- (B) Foster the planting of native as well as other trees, shrubs, and flowering plants compatible to the enhancement of our environment.

DISCUSSION: A single plant of endangered ko'oloa'ula (Abutilon menziesii) was found within the property in a 1997 flora survey but has since died. In comparison larger plant clusters were found within the alignment of the North-South Road and adjoining DHHL properties. Since the North-South Road was the first project to be constructed which would affect the Abutilon menziesii plant clusters, DOT prepared a Habitat Conservation Plan. In accordance with both State and Federal regulations, mitigation measures have been identified in the Habitat Conservation Plan for Abutilon menziesii at Kapolei (HCP) prepared in consultation with the State DLNR. On April 8, 2004, the BLNR unanimously approved the HCP and accompanying Incidental Take License and Certificate of Inclusion. The University of Hawai'i is currently working with the DOT to be included in the Certificate of Inclusion for the Incidental Take License. Long term impact on the Abutilon menziesii population in Kapolei will be beneficial, as proposed mitigation measures in the HCP will ensure the future propagation of new plants. If an off-site preserve can be established, some of the plants would remain *in situ* for many years. Because the protection, propagation, and relocation of the ko'oloa'ula plant is a long-term undertaking, final implementation of the HCP will extend well into the project's construction period. Once construction and build-out of the campus is complete, use of the ko'oloa'ula in project landscaping and continued use of the ko'oloa'ula nursery for propagation will ensure a much larger and vigorous population of the ko'oloa'ula than would have occurred without development of the UH West O'ahu property. The HCP also proposes long-term management of the populations to occur concurrently with project development, over a period of approximately 20 years. The successful implementation of the HCP would significantly increase the numbers of new plants on O'ahu and improve the quality of existing populations on adjacent properties.

As noted in Section 2.6 of this Draft EIS, one of the sustainability guidelines for UHWO is to "create habitat that relies on native species, enhances the campus experience and celebrates the region's agrarian roots."

PARKS, RECREATION, AND OPEN SPACE

- (A) Establish, preserve and maintain scenic, historic, cultural, park and recreation areas, including the shorelines, for public recreational, educational, and scientific uses.
- (C) Promote open space in view of its natural beauty not only as a natural resource but as an ennobling, living environment for its people.

Discussion: A hierarchy of open spaces will be provided within the campus and surrounding residential developments. The campus open space elements include a central plaza, informal courtyards, and open spaces. Two pedestrian malls enhanced with lighting (for safety), consistent paving patterns and materials, and site furnishings will also help to define the open space network. The campus will be connected to regional open space systems adjacent to the campus through the Hunehune and Kalo'i greenway, which will extend from the campus to the surrounding community. Mauka-makai views from the property will be enhanced, focusing on views of the Wai'anae and Ko'olau Mountains, and landscape treatments and

building design will mitigate visual impacts of the campus.

ECONOMIC DEVELOPMENT

(B) Promote and foster the agricultural industry of the State; and preserve and conserve productive agricultural lands.

Discussion: Currently cultivated lands within the property will be withdrawn from agricultural production for the proposed development. This will result in some loss in revenues, jobs, or payroll. However, the State and City have long planned for new development in the project area, and the property is within the Urban State Land Use District. Tenants have been fully aware, for quite some time, that the proposed project area would be used to accommodate for future development in the region.

TRANSPORTATION

- (A) Encourage transportation systems in harmony with the lifestyle of the people and environment of the State.
- (B) Adopt guidelines to alleviate environmental degradation caused by motor vehicles.

Discussion: The plan for UHWO addresses the need to expand and improve the transportation system and close the gap between where people live and work through decentralization, mixed zoning, and related initiatives. The UH West O'ahu will provide education and employment opportunities in the 'Ewa region. The campus is conveniently located near major transportation facilities and residential developments, thus encouraging public transportation and walking or bicycling. West O'ahu residents will be able to live, work, and attend school within this region, alleviating traffic to other universities and workplaces in Honolulu.

The UH West O'ahu campus will help to alleviate environmental degradation by reducing commuting time and distances for students, staff and faculty residing in the 'Ewa region. West O'ahu residents will have better access to higher education facilities, and ultimately, the campus will connect to a transit corridor planned adjacent to the North-South Road. Within the campus, pedestrian walkways, facilities for bicycles, and facilities for distance education will encourage non-motorized vehicular forms of transportation and reduce fossil fuel consumption.

ENERGY

(A) Encourage the efficient use of energy resources.

Discussion: The UH West O'ahu campus will help to conserve existing energy resources by reducing commuting time and distances for students and faculty residing in the 'Ewa region. West O'ahu residents will have better access to higher education facilities, and ultimately, the

campus will connect to a transit corridor planned adjacent to the North-South Road. Within the campus, pedestrian walkways, facilities for bicycles, and facilities for distance education will encourage non-motorized vehicular forms of transportation and reduce fossil fuel consumption. The use of energy-efficient equipment for lighting systems and hot water heating and cooling will also be encouraged within the campus. The State's Model Energy Code will be considered during the detailed design phases of project development.

Sustainability guidelines have been established for the UH West O'ahu Lands (see Section 2.6). The guidelines strive to achieve a Leadership in Energy and Environmental Design (LEED) NC silver standard for campus development and community performance standards based on the LEED NC v2.2 rating system. The Private Development Lands will strive to achieve the applicable design criteria in the UH West O'ahu LRDP and the recommended community performance standards.

COMMUNITY LIFE AND HOUSING

- (A) Foster lifestyles compatible with the environment; preserve the variety of lifestyles traditional to Hawaii through the design and maintenance of neighborhoods which reflect the culture and mores of the community.
- (B) Develop communities which provide a sense of identity and social satisfaction in harmony with the environmental and provide internal opportunities for shopping, employment, education, and recreation.
- (*C*) Encourage the reduction of environmental pollution which may degrade a community.
- (D) Foster safe, sanitary, and decent homes.
- (E) Recognize community appearances as major economic and aesthetic assets of the counties and the State; encourage green belts, plantings, and landscape plans and designs in urban area; and preserve and promote mountain-to-ocean vistas.

Discussion: The proposed project presents a unique opportunity to create a new campus that addresses the need to create more of a resident campus with more student housing opportunities than normally provided by most universities and a "town-gown"/campus town lifestyle, as well as housing opportunities to attract the best faculty and staff. The university campus-focus of a residential mixed-use project will attract residents who are interested in not only in their childrens' education, but in cultural events and life-long learning opportunities that UHWO will present. The project has been designed to provide internal opportunities for shopping, employment, education and recreation.

As noted above, one of the goals of the project is to achieve LEED NC silver standard as a way to reduce environmental pollution. All housing on the project site will be designed to foster safe, sanitary and decent homes.

Kalo'i Greenway and Hunehune Greenway will include a pedestrian/bike path that will connect the campus with residential and mixed-use commercial developments surrounding the campus. The greenway system is also envisioned to link the two proposed transit stations on the eastern portion of the property (adjacent to the North-South Road) with the University and surrounding mixed-use and residential communities. The greenway system, for the most part, will be unimpeded by vehicular traffic.

<u>Plazas, courtyards, other landscaped areas, and the 10-acre detention basin will provide open</u> <u>space within the campus, allowing views towards the Wai'anae and Ko'olau mountain</u> <u>ranges.</u>

EDUCATION AND CULTURE

(A) Foster culture and the arts and promote their linkage to the enhancement of the environment.

Discussion: While the residential population within Kapolei has grown as planned, it lacks much of the culture and the arts which are available within the Primary Urban Center. The university campus-focus of a residential mixed-use project will attract residents who are interested in not only in their childrens' education, but in cultural events and life-long learning opportunities that UHWO will present.

CITIZEN PARTICIPATION

(A) Encourage all individuals in the State to adopt a moral ethic to respect the natural environment; to reduce waste and excessive consumption; and to fulfill the responsibility as trustees of the environment for the present and succeeding generations.

Discussion: The University of Hawai'i has been a leader in recycling and will continue this effort at its UHWO campus.

5.2.3 State Land Use Law, Chapter 205, Hawaii Revised Statutes

Chapter 205, HRS establishes the State Land Use Commission (LUC) and classifies all lands in the State into four districts: Urban, Rural, Agricultural, and Conservation. Ongoing urbanization of the 'Ewa region conforms to the official policies established by the LUC. The UH West O'ahu property is located within the Urban District and the proposed land uses are permitted (see Figure 1.6). <u>During the public review period, the Land Use Commission wrote: "…the Land Use Commission (LUC) urbanized the project site under LUC Docket No. A99-728/Housing and Community Development Corporation of Hawaii, State of Hawaii (HCDCH). The LUC imposed 27 conditions upon the reclassification pursuant to Findings of Fact, Conclusions of Law, and Decision and Order (Decision and Order) dated September 8, 1999. Given that the conditions run with the land and are binding upon the HCDCH and each and every subsequent owner, lessee, sub-lessee, transferee, grantee, assignee, or development Lands is subject to these conditions. As such, we request that there be</u>

discussion of the proposed uses on these lands and their impacts within the context of applicable conditions."

The original intent of the State Land Use District Boundary Amendment was to entitle approximately 1,300 acres of land in East Kapolei, which would then be sold to private developers. It was intended that the proceeds of the land sales would be used to fund the construction of a UHWO campus site mauka of the H-1 Freeway. Subsequent to the LUC approval of LUC Docket No. A99-728, Hawai'i's economy suffered a serious downturn, and there was little or no market demand for new residential development. During this period, it was determined that the construction of a new campus mauka of the H-1 Freeway would be more expensive than the current site (makai of Farrington Highway). Since the marketability of the petition area was non-existent, and a site selection study determined that the current site was best suited for a new UHWO campus, UHWO was allocated 500 acres and most of the remainder of the 1,300 acres which were "urbanized" was granted to DHHL. The subsequent owners (UHWO, DHHL, DOT, DLNR and eventually DOE) then, are significantly different from those originally envisioned (private residential developers).

UH West O'ahu intends to request an amendment to the Decision and Order within a few months.

5.2.4 Coastal Zone Management Program, Chapter 205A, Hawaii Revised Statutes

The objectives of the Coastal Zone Management (CZM) Program (Section 205A-2, HRS) are to provide the public with recreational opportunities, protect historic and prehistoric resources, protect scenic and open space resources, protect coastal ecosystems, provide facilities for economic development, reduce hazards, and manage development. Objectives of the CZM Program that are applicable to the UH West O'ahu development are discussed below.

RECREATIONAL RESOURCES

Objective: Provide coastal recreational opportunities accessible to the public. **Policies:**

- (B) Provide adequate, accessible, and diverse recreational opportunities in the coastal zone management area by:
 - (i) Protecting coastal resources uniquely suited for recreation activities that cannot be provided in other areas.
 - (iii) Providing and managing adequate public access, consistent with conservation of natural resources, to and along shorelines with recreational value.
 - (iv) Providing an adequate supply of shoreline parks and other recreational facilities suitable for public recreation.
 - (vii) Developing new shoreline recreational opportunities, where appropriate, such as artificial lagoons, artificial beaches, and artificial reefs for surfing and fishing.
 - (viii) Encouraging reasonable dedication of shoreline areas with recreational value for public use as part of discretionary approvals or permits by the land use commission, board of land and natural resources, and county authorities; and crediting such dedication against the requirements of section 46-6.

Discussion: The UH West O'ahu property is located in Zone D, areas in which flood hazards are undetermined (see Figure 3.1). Uncontrolled flooding and associated water quality impacts on coastal resources can result from the existing undeveloped condition of the property. However, with the proposed development and associated drainage improvements, surface flows will be better managed. A 10-acre detention basin within the campus and an 11.2-acre detention basin at the southern boundary of the property will collect silt on-site before it is discharged off-site through control structures. Off-site drainage improvements will comply with all applicable State and City rules and regulations. Since the property is located inland, the project will not impact access to coastal resources.

HISTORIC RESOURCES

Objective: Protect, preserve, and where desirable, restore those natural and man made historic and prehistoric resources in the coastal zone management area that are significant in Hawaiian and American history and culture.

Policies:

- (A) Identify and analyze significant archaeological resources;
- (B) Maximize information retention through preservation of remains and artifacts or salvage operations; and
- (*C*) Support state goals for protection, restoration, interpretation, and display of historic resources.

Discussion: The State Historic Preservation Division has concluded that no significant archaeological sites are likely to exist in the UH West O'ahu project area. However, should any archaeologically significant artifacts, bones, or other indicators of previous on-site activity be uncovered during construction of the campus, they will be treated in compliance with the DLNR requirements.

SCENIC AND OPEN SPACE RESOURCES

Objective: Protect, preserve, and where desirable, restore or improve the quality of coastal scenic and open space resources.

Policies:

- (B) Ensure that new developments are compatible with their visual environment by designing and locating such developments to minimize the alteration of natural landforms and existing public views to and along the shoreline.
- (C) Preserve, maintain, and where desirable, improve and restore shoreline open space and scenic resources.
- (D) Encourage those developments that are not coastal dependent to locate in inland areas.

Discussion: The UH West O'ahu property is located outside of the Special Management Area and approximately three to four miles mauka of coastal areas. As such, coastal scenic resources will not be significantly affected by the campus. The relatively flat topography of the property and the 'Ewa Plain prevents views of the shoreline from within the site, and applicable PRU regulations that restrict heights and density of development will further mitigate visual impacts of structures on the property. Generous provisions for open space

within the campus have also been provided. Detention basins within the campus and at the southern boundary of the property will serve as open space.

COASTAL ECOSYSTEMS

Objective: Protect valuable coastal ecosystems, including reefs, from disruption and minimize adverse impacts on all coastal ecosystems.

Policies:

- (B) Improve the technical basis for natural resource management.
- (C) Preserve valuable coastal ecosystems, including reefs, of significant biological or economic importance.
- (D) Minimize disruption or degradation of coastal water ecosystems by effective regulation of stream diversions, channelization, and similar land and water uses, recognizing competing water needs.
- (E) Promote water quantity and quality planning and management practices that reflect the tolerance of fresh water and marine ecosystems and maintain and enhance water quality through the development and implementation of point and nonpoint source water pollution control measures.

Discussion: The UH West O'ahu property does not front the natural shoreline. Nevertheless, the project will incorporate necessary measures to mitigate any water quality impacts from surface run-off in accordance with applicable State DOH and City drainage regulations. Because rainfall in the area is relatively low, proposed drainage improvements have been designed to control the quantity and quality of surface water by detaining it on-site to facilitate groundwater recharge. Detaining surface water on-site will mitigate potential impacts to coastal resources by improving water quality before runoff leaves the site and restricting runoff quantities to current levels. Similarly, construction-related impacts will be mitigated by the implementation of best management practices to control erosion.

ECONOMIC USES

Objective: Provide public or private facilities and improvements important to the State's economy in suitable locations.

Policies:

- (A) Ensure that coastal dependent development such as harbors and ports, and coasted related development such as visitor industry facilities and energy generating facilities, are located, designed, and constructed to minimize adverse social, visual, and environmental impacts in the coastal zone management area.
- (B) Direct the location and expansion of coastal dependent developments to areas presently designated and used for such developments and permit reasonable long-term growth at such areas, and permit coastal dependent development outside presently designated areas when:
 - (ii) Adverse environmental effects are minimized.

Discussion: The UH West O'ahu campus will provide employment and educational opportunities in the 'Ewa region, where population is growing fastest in O'ahu. The campus is located a few miles mauka of the shoreline and does not propose any coastal-dependent

development. The land uses planned for the project are well inland of coastal areas and are appropriate for the Urban State Land Use District. Coastal resources will not be affected and visual impacts will be minimized, as the campus will be integrated with the surrounding topography, to the extent possible.

COASTAL HAZARDS

Objective: Reduce hazard to life and property from tsunami, storm waves, stream flooding, erosion, subsidence, and pollution.

Policies:

- (B) Control development in areas subject to storm wave, tsunami, flood, erosion, hurricane, wind, subsidence, and point and nonpoint source pollution hazards.
- (C) Ensure that developments comply with requirements of the Federal Flood Insurance Program.
- (D) Prevent coastal flooding from inland projects.

Discussion: The proposed UH West O'ahu will be designed and constructed in compliance with all applicable Federal, State, and City environmental protection, design, and building standards and regulations, including the Federal Flood Insurance Program. The property is not located within the Special Management Area (SMA), and any potential impact of surface runoff on near-shore ecosystems will be mitigated by the implementation of best management practices during construction. Proposed drainage improvements will mitigate the existing flood hazard potential, and drainage systems will comply with City design standards to ensure the safe conveyance and discharge of storm runoff.

MANAGING DEVELOPMENT

Objective: Improve the development review process, communication and public participation in the management of coastal resources and hazards.

Policies:

- (A) Use, implement, and enforce existing law effectively to the maximum extent possible in managing present and future coastal zone development;
- (B) Facilitate timely processing of applications for development permits and resolve overlapping or conflicting permit requirements; and
- (C) Communicate the potential short- and long-term impacts of proposed significant coastal developments early in their life-cycle and in terms understandable to the public to facilitate public participation in the planning and review process.

Discussion: Applications for required land use entitlements for the proposed project will be reviewed by both City and State land use planning agencies. Individuals, community groups, and appropriate City agency personnel have been consulted about the proposed project, and comments received have been incorporated into the planning process. No coastal areas or resources are expected to be affected by the proposed campus, which is not located near the shoreline.

PUBLIC PARTICIPATION

Objective: Stimulate public awareness, education, and participation in coastal management. **Policies:**

- (A) Promote public involvement in coastal zone management processes.
- (C) Organize workshops, policy dialogues, and site-specific mediations to respond to coastal issues and conflicts.

Discussion: As a State-sponsored project, the proposed UH West O'ahu development has received public comments and will undergo thorough examination during the legislative and public hearing process. The campus is located well inland from the ocean; therefore, no impact on coastal resources is anticipated.

BEACH PROTECTION

Objective: Protect beaches for public use and recreation. **Policies:**

- (A) Locate new structures inland from the shoreline setback to conserve open space, minimize interference with natural shoreline processes, and minimize loss of improvements due to erosion.
- (*C*) Minimize the construction of public erosion-protection structures seaward of the shoreline.

Discussion: The proposed UH West O'ahu will not be located on or near a beach, nor will it impact coastal resources.

MARINE RESOURCES

Objective: Promote the protection, use, and development of marine and coastal resources to assure their sustainability.

Policies:

- (A) Ensure that the use and development of marine and coastal resources are ecologically and environmentally sound and economically beneficial.
- (B) Coordinate the management of marine and coastal resources and activities to improve effectiveness and efficiency.

Discussion: Since the UH West O'ahu property is located away from the shoreline, no coastal or marine resources will be impacted.

5.2.5 Hawaii State Plan, Chapter 226, Hawaii Revised Statutes

The *Hawaii State Plan* (Chapter 226, HRS), serves as a guide for the future long-range development of the State; identifies the goals, objectives, policies, and priorities for the State; provides a basis for determining priorities and allocating limited resources; improves coordination of Federal, State, and County plans, policies, programs, projects, and regulatory activities; and establishes a system for plan formulation and program coordination to provide for an integration of all major State and County activities. Sections of the *Hawaii State Plan*

applicable to the UH West O'ahu project are discussed in the following pages.

Section 226-4 State goals: In order to guarantee, for present and future generations, those elements of choice and mobility that insure that individuals and groups may approach their desired levels of self-reliance and self-determination, it shall be the goal of the State to achieve:

- (1) A strong, viable economy, characterized by stability, diversity, and growth, that enables the fulfillment of the needs and expectations of Hawaii's present and future generations.
- (2) A desired physical environment, characterized by beauty, cleanliness, quiet, stable natural systems, and uniqueness, that enhances the mental and physical well- being of the people.
- (3) Physical, social, and economic well-being, for individuals and families in Hawaii, that nourishes a sense of community responsibility, of caring, and of participation in community life.

Section 226-5 Objective and policies for population:

- (a) It shall be the objective in planning for the State's population to guide population growth to be consistent with the achievement of physical, economic, and social objectives contained in this chapter.
- (b) To achieve the population objective, it shall be the policy of this State to:
 - (3) Promote increased opportunities for Hawaii's people to pursue their socio-economic aspirations throughout the islands.

Section 226-6 Objective and policies for the economy – in general:

- (a) Planning for the State's economy in general shall be directed toward achievement of the following objectives:
 - (1) Increased and diversified employment opportunities to achieve full employment, increased income and job choice, and improved living standards for Hawaii's people.

Discussion: The proposed UH West O'ahu project will provide for a diverse range of employment and economic opportunities for Hawai'i residents, both during and after project construction. Short-term construction-related jobs as well as permanent long-term operational jobs will be offered directly and indirectly, increasing employment throughout the region and State. Other socio-economic benefits to West O'ahu residents include opportunities to attend an institution of higher education and obtain a baccalaureate degree without commuting to other areas in O'ahu, and the myriad of social and cultural events associated with universities. With greater economic and educational opportunities, the overall living standards and lifestyles will be enhanced for West O'ahu residents, who will be able to live, work, and attend an institution of higher education within the Second City.

Section 226-11 Objectives and policies for the physical environment – land-based, shoreline, and marine resources:

- (a) Planning for the State's physical environment with regard to land-based, shoreline, and marine resources shall be directed towards achievement of the following objectives.
 - (2) Effective protection of Hawaii's unique and fragile environmental resources.
- (b) To achieve the land-based, shoreline, and marine resources objectives, it shall be the policy of this State to:

- (1) Exercise an overall conservation ethic in the use of Hawaii's natural resources.
- (2) Ensure compatibility between land-based and water-based activities and natural resources and ecological systems.
- (3) Take into account the physical attributes of areas when planning and designing activities and facilities.
- (4) Manage natural resources and environs to encourage their beneficial and multiple use without generating costly or irreparable environmental damage.
- (6) Encourage the protection of rare or endangered plant and animal species and habitats native to Hawaii.
- (8) Pursue compatible relationships among activities, facilities, and natural resources.
- (9) Promote increased accessibility and prudent use of inland and shoreline areas for public recreational, educational, and scientific purposes.

Discussion: In accordance with both State and City policies, most new development on O'ahu is being directed toward the 'Ewa Plain, as its physical attributes are compatible with urban development. The UH West O'ahu property was surveyed during the Site Selection Study to ensure that the physical, environmental, and cultural attributes of the property were compatible with the land uses proposed. The survey found that most natural features on the property have been extensively modified by past agricultural activities. A few individuals of the endangered plant, *Abutilon menziesii* were found along the southeastern portion of the property, and seeds may be present in the soil where another individual once lived. The HCP has been prepared to ensure that the species is protected and proliferates. No other unique or fragile environmental resources are known to exist on the property. This EIS identifies existing natural and physical site conditions (i.e., slope, soils, drainage characteristics, archaeological sites, flora and fauna, public services and infrastructure) and potential impacts resulting from the UH West O'ahu development, and proposes several measures to mitigate potential impacts.

Section 226-12 Objective and policies for the physical environment – scenic, natural beauty, and historic resources:

- (a) Planning for the State's physical environment shall be directed towards achievement of the objective of enhancement of Hawaii's scenic assets, natural beauty, and multi-cultural/historical resources.
- (b) To achieve the scenic, natural beauty, and historic resources objective, it shall be the policy of this State to:
 - (1) Promote the preservation and restoration of significant natural and historic resources.
 - (3) Promote the preservation of views and vistas to enhance the visual and aesthetic enjoyment of mountains, ocean, scenic landscapes, and other natural features.
 - (4) Protect those special areas, structures, and elements that are an integral and functional part of Hawaii's ethnic and cultural heritage.
 - (5) Encourage the design of developments and activities that complement the natural beauty of the islands.

Discussion: No significant archaeological sites have been identified on the UH West O'ahu property. However, in response to requests from the State Historic Preservation Division and

the Office of Hawaiian Affairs, UH West O'ahu has contracted the preparation of a new archaeological inventory survey. Should any subsurface archaeological features be uncovered during construction, the State Historic Preservation Division will be notified in accordance with State requirements. These features will be protected or preserved as recommended by the project archaeologist and other affected groups, as applicable. The endangered plant species, *Abutilon menziesii*, will be protected with implementation of the HCP. Due to the flat topography of the property and adjacent lands, there are no ocean views from within the property and the most prominent views are of the Wai'anae Mountain Range. The proposed campus will provide viewing opportunities of the mountain range, as the site is currently inaccessible to the public.

Section 226-13 Objectives and policies for the physical environment – land, air, and water quality:

- (a) Planning for the State's physical environment with regard to land, air, and water quality shall be directed towards achievement of the following objectives:
- (1) Maintenance and pursuit of improved quality in Hawaii's land, air, and water resources.
- (b) To achieve the land, air, and water quality objectives, it shall be the policy of this State to:
 - (2) Promote the proper management of Hawaii's land and water resources.
 - (3) Promote effective measures to achieve desired quality in Hawaii's surface, ground, and coastal waters.
 - (4) Encourage actions to maintain or improve aural and air quality levels to enhance the health and well-being of Hawaii's people.
 - (5) Reduce the threat to life and property from erosion, flooding, tsunamis, hurricanes, earthquakes, volcanic eruptions, and other natural or man-induced hazards and disasters.
 - (6) Encourage design and construction practices that enhance the physical qualities of Hawaii's communities.
 - (7) Encourage urban developments in close proximity to existing services and facilities.

Discussion: Planning and design of the proposed UH West O'ahu development recognizes the importance and value of the area's land, air, and water resources. The campus will take advantage of the existing aesthetic qualities of the area, while enhancing the physical attributes of the region. Design guidelines will also be developed to ensure compatibility between structures within the campus. The potential flood hazard on the property will be mitigated by the development of a system of detention facilities that comply with City drainage regulations and ensure that pre-development off-site runoff quantities are maintained. Non-potable water will be provided by from the 215-foot elevation dual water system planned by the BWS. Natural hazards such as hurricanes, earthquakes, and volcanic eruptions exist, but are no more likely to affect the property than any other location in the 'Ewa Plain.

The proposed UH West O'ahu property is adjacent to Farrington Highway and the planned North-South Road. All necessary infrastructure will be sized and engineered to accommodate the development.

Section 226-15 Objectives and policies for facility systems – solid and liquid wastes:

- (a) Planning for the State's facility systems with regard to solid and liquid wastes shall be directed towards achievement of the following objectives:
 - (1) Maintenance of basic public health and sanitation standards relating to treatment and disposal of solid and liquid wastes.

Section 226-16 Objective and policies for facility systems – water:

- (a) Planning for the State's facility systems with regard to water shall be directed towards achievement of the objective of the provision of water to adequately accommodate domestic, agricultural, commercial, industrial, recreational, and other needs within resource capacities.
- (b) To achieve the facility systems water objective, it shall be the policy of this State to:
 - (1) Coordinate development of land use activities with existing and potential water supply.

Section 226-17 Objectives and policies for facility systems – transportation:

(b) To achieve the transportation objectives, it shall be the policy of this State to:

- (6) Encourage transportation systems that serve to accommodate present and future development needs of communities.
- (11) Encourage safe and convenient use of low-cost, energy-efficient, non-polluting means of transportation.
- (13) Encourage diversification of transportation modes and infrastructure to promote alternate fuels and energy efficiency.

Section 226-18 Objectives and policies for facility systems – energy:

- (c) To further achieve the energy objectives, it shall be the policy of this State to:
 - (4) Promote all cost-effective conservation of power and fuel supplies through measures including:
 - (A) Development of cost-effective demand-side management programs;
 - (B) Education; and
 - (C) Adoption of energy-efficient practices and technologies;
 - (7) Promote alternate fuels and energy efficiency by encouraging diversification of transportation modes and infrastructure.

Discussion: Improvements to the wastewater system, such as the construction of wastewater transmission lines, will be completed in accordance with project phasing requirements. Solid waste produced at the campus will be recycled, used to generate electricity at the H-POWER facility in Campbell Industrial Park, and/or transferred to the Waimānalo Gulch Landfill. Wastewater from the campus will be collected, treated, and disposed of in accordance with applicable State DOH and City regulations at facilities owned and operated by the City. No significant impacts on groundwater resources or the quality of surface water are anticipated to result from the project. Construction activities will comply with all applicable DOH regulations.

The UH West O'ahu property is located within the BWS' 215- and 440-foot elevation water service zones. Upgrades to the existing 215-foot potable water system and a new 440-foot

potable water system will be required to accommodate East Kapolei developments. Potable water for the UH West O'ahu will be provided by the 215- and 440-foot systems. Irrigation water will be derived from the 215-foot dual system, when constructed, to conserve potable water for human consumption. All applicable governmental regulations will be observed to ensure public safety and health.

The proposed project will promote greater energy self-sufficiency, as it is located near essential services, employment centers, and existing and planned residential areas. Pedestrian walkways and bikeways within the campus are planned to encourage alternate forms of transportation. The University plans to adopt energy-efficient design practices and incorporate, as applicable, technologies such as efficient lighting and water heating systems, computerized energy management systems, roof installation, radiant barriers, landscaping, and energy-efficient windows. It will also explore financing opportunities for energy-efficient technologies with HECO and implement the State's Model Energy Code in applicable phases of project design. As additional employment opportunities become available in the 'Ewa region, fewer West O'ahu residents will commute to work outside of the region. This will eventually result in a reduction of energy consumed by transportation vehicles.

During the DEIS public review period, the State Department of Business, Economic Development & Tourism (DBEDT), Strategic Industries Division, wrote to call attention to: 1) State energy conservation goals; and 2) energy and resource efficiency and renewable energy and resource development.

DBEDT noted that project buildings, activities, and site grounds should be designed and/or retrofit with energy saving "considerations." DBEDT further noted that the mandate for such consideration is found in Chapter 344, HRS ("State Environmental Policy") and Chapter 226, HRS ("Hawaii State Planning Act"). In particular, DBEDT noted HRS 226 18(c) (4) which includes a State objective of promoting all cost-effective energy conservation through adoption of energy-efficient practices and technologies. This particular subsection was addressed in the Draft EIS.

DBEDT noted two important directives affecting State agencies. The first is Administrative Directive No. 06-01, which is also reflected in Act 96, 2006 Session Laws of Hawaii (SLH), which encourages State agencies and programs to increase their leadership commitment to implement innovative and resource efficient operations and management and to design and construct buildings to meet and receive certification for U.S. Green Building Council's Leadership in Energy and Environmental Design (LEED), among others. The other directive DBEDT noted was Act 160, 2006 SLH, relating to the State Budget, which requires that each executive department shall plan for or install energy reduction, energy savings or energy-producing efforts and technologies to lessen electrical consumption or to increase efficiencies in using electrical energy. As stated in Section 2.6 of the Draft EIS:

Sustainability guidelines have been established for the UH West O'ahu. The guidelines strive to achieve a Leadership in Energy and Environmental Design

(LEED) NC silver standard for campus development and community performance standards based on the LEED NC v2.2 rating system.

DBEDT recommended that UHWO consult the City and County of Honolulu Energy code early in the project. DBEDT also noted that HECO may also have demand-side management programs that offer rebates for installation of energy-efficient technologies.

Based on DBEDT's recommendations, UHWO's mechanical and electrical consultants, in consultation with UHWO's sustainability consultant, will be directed to review the City and County of Honolulu's Energy code early in the project and to consult with HECO on demand-side management programs that offer rebates for installation of energy-efficient technologies.

Section 226-52 Statewide planning system.

- (a) The statewide planning system shall consist of the following policies, plans, and programs:
 - (2) The priority guidelines established in this chapter shall provide guidelines for decisionmaking by the State and the counties for the immediate future and set priorities for the allocation of resources. The formulation and amendment of state functional plans shall be in conformance with the priority guidelines.
- (b) The statewide planning system shall also consist of several implementation mechanisms, including:
 - (2) The state budgetary, land use, and other decision making processes. The state budgetary, land use, and other decision making processes shall consist of:
 - (D) Land use decision making processes of state agencies. Land use decisions made by state agencies shall be in conformance with the overall theme, goals, objectives, and policies, and shall utilize as guidelines the priority guidelines contained within this chapter, and the state functional plans adopted pursuant to this chapter. The rules adopted by appropriate state agencies to govern land use decision making shall be in conformance with the overall theme, goals, objectives, and policies contained within this chapter.

Discussion: The proposed project complies with the guidelines established by the *Hawaii State Plan* and *State Functional Plans* regarding the Statewide planning system and the land use decision-making process. A 45-day review period will commence upon publication of this Draft EIS in *The Environmental Notice*. Additionally, the review and approval process for required permits will ensure that the development complies with applicable land use policies and regulations, and will allow the public ample opportunity to review the proposed plans and development program.

Section 226-103 Economic priority guidelines.

- (f) Priority guidelines for energy use and development:
 - (1) Encourage the development, demonstration, and commercialization of renewable energy resources.
 - (2) Initiate, maintain, and improve energy conservation programs aimed at reducing energy waste and increasing public awareness of the need to conserve energy.

Discussion: Solar water heating will be encouraged and solid waste from the development will be used to generate electricity at the H-POWER facility. Overall, energy will be used more efficiently as residents can live, work, and attend a public university without commuting to Honolulu.

Section 226-104 Population growth and land resources priority guidelines.

- (b) Priority guidelines for regional growth distribution and land resource utilization:
 - (1) Encourage urban growth primarily to existing urban areas where adequate public facilities are already available or can be provided with reasonable public expenditures, and away from areas where other important benefits are present, such as protection of important agricultural land or preservation of lifestyles.
 - (6) Direct future urban development away from critical environmental areas or impose mitigating measures so that negative impacts on the environment would be minimized.
 - (10) Identify critical environmental areas in Hawaii to include but not be limited to the following: watershed and recharge areas; wildlife habitats (on land and in the ocean); areas with endangered species of plants and wildlife; natural streams and water bodies; scenic and recreational shoreline resources; open space and natural areas; historic and cultural sites; areas particularly sensitive to reduction in water and air quality; and scenic resources.
 - (12) Utilize Hawaii's limited land resources wisely, providing adequate land to accommodate projected population and economic growth needs while ensuring the protection of the environment and the availability of the shoreline, conservation lands, and other limited resources for future generations.

Discussion: The proposed UH West O'ahu campus will expand access to, and help meet the projected demand for, higher education facilities in the 'Ewa region. Employment opportunities for residents of this region will be directly and indirectly stimulated by construction and operations of the campus and commercial lands. Various public facilities exist within the City of Kapolei, and several new facilities are planned to accommodate the projected population growth for the 'Ewa area. The property is within the State Urban Land Use District, and transit nodes located near the property will encourage the use of public transportation and help to alleviate traffic in the area and toward Honolulu.

The property is located away from the shoreline and outside of an environmentally-sensitive area. During the site feasibility and site selection phases, the project area was surveyed to identify critical environmental areas that would be potentially impacted by the project. Based on these surveys, it has been determined that:

- No significant potable groundwater resources or recharge areas are associated with the property;
- No significant wildlife habitats exist on the property;
- Existing individuals and seeds of an endangered plant species (*Abutilon menziesii*) will be removed from the property, and the HCP has been developed to protect the species;
- No significant scenic or recreational shoreline resources will be impacted;
- Open space agricultural resources will be affected (the majority of the property is currently under revocable agricultural leases, and a portion of the property consists of vacant scrub vegetation); and

• Historic and cultural sites will not be destroyed or altered by the project.

When available, non-potable water from the BWS' planned 215-foot elevation dual system will be used for landscape irrigation. All construction activities will comply with DOH regulations to mitigate potential erosion and air and water quality impacts. The property will be transformed from an agricultural setting to a university campus and community that will accommodate the 'Ewa region's projected population and economic growth and educational needs. Should any historic and cultural sites be discovered during construction, work will halt and the SHPD will be contacted.

5.2.6 Hawaii State Functional Plans

The *Hawaii State Plan* is primarily guided by the *State Functional Plans* (Chapter 226, HRS) and implemented by the Department of Budget and Finance and the State LUC. *State Functional Plans* are prepared by various state agencies, with community input, and focus on a specific area: agriculture, conservation lands, education, employment, energy, health, higher education, historic preservation, housing, human services, recreation, tourism, and transportation. The following pages describe how the proposed UH West O'ahu project comply with applicable *State Functional Plans*.

State Agriculture Functional Plan. The objectives, policies, and implementation actions in this plan address the broad issues of agricultural research, marketing, development, regulations, and protection; however, they do not directly apply to the proposed UH West O'ahu project, since the property is within the Urban State Land Use District and zoned AG-1 Restricted Agriculture by the City.

State Employment Functional Plan. The policies and recommended actions of this plan center around the development and improvement of career/job training programs, the expansion of the labor pool, and the improvement of the quality of life for workers. In accordance with this plan, development of the proposed UH West O'ahu project will enhance career/job training programs, expand the labor pool, and improve the quality of life for workers. Construction of the project and operation of the campus and commercial areas will provide short- and long-term employment opportunities, both directly and indirectly. According to the socio-economic impact assessment report, approximately 14,624 person-years of employment are estimated to be generated by the project (1,828 jobs in O'ahu over eight years of construction). Additionally, an estimated 5,835 direct, indirect, and induced operations jobs will be created by the UH West O'ahu project between 2009 and 2015.

State Energy Functional Plan. This plan develops policies to promote energy efficiency, displace fossil fuel consumption, support public education and legislation on energy, and improve the development and management of energy. The UH West O'ahu campus will help to conserve existing energy resources by reducing commuting time and distances for students and faculty residing in the 'Ewa region. West O'ahu residents will have better access to higher education facilities, and ultimately, the campus will connect to a transit corridor

planned adjacent to the North-South Road. Within the campus, pedestrian walkways, facilities for bicycles, and facilities for distance education will encourage non-motorized vehicular forms of transportation and reduce fossil fuel consumption. The use of energy-efficient equipment for lighting systems and hot water heating and cooling will also be encouraged within the campus. The State's Model Energy Code will be considered during the detailed design phases of project development.

Sustainability guidelines have been established for the UH West O'ahu Lands (see Section 2.6). The guidelines strive to achieve a Leadership in Energy and Environmental Design (LEED) NC silver standard for campus development and community performance standards based on the LEED NC v2.2 rating system. The Private Development Lands will strive to achieve the applicable design criteria in the UH West O'ahu LRDP and the recommended community performance standards.

State Historic Preservation Functional Plan. According to this plan, "the preservation of historic properties involves three major areas of activity: the identification, protection, and management and treatment of historic properties." The policies in this plan are aimed primarily toward government action to improve the State's inventory, preservation systems, public access, and public awareness programs regarding archaeological matters. The proposed project will comply with the policies of this plan <u>(and a new archaeological inventory survey is being conducted by UH West O'ahu)</u>, and, in accordance with State regulations, should any archaeological sites be uncovered during development of the campus, construction will halt and the SHPD of the DLNR will be notified.

State Higher Education Functional Plan/University of Hawai'i Strategic Plan. The University of Hawai'i Strategic Plan has replaced the State Higher Education Functional Plan. The University of Hawai'i Strategic Plan, 1997-2007, sets forth University goals and assumptions, strategic planning principles, and action strategies to direct campus and unit planning efforts and the development of the University of Hawai'i system into the 21st century.

Relevant roles and missions for UH West O'ahu are:

- b. Provide educational and cultural services and function as a community resource for the residents of West O'ahu. In cooperation with the rest of the system, participate in providing distance education on O'ahu and the neighbor islands.
- c. Continue to carry out its research mission primarily through scholarship and creative activities that support the instructional mission.
- d. In cooperation with Leeward Community College and UH Mānoa, prepare a longrange academic development plan for implementation when construction begins at the new site.
 - (1) In the first five to twenty years at the site, the campus will develop as a four-year baccalaureate institution offering a more comprehensive array of liberal arts programs and selected undergraduate professional programs for which there is a sustainable demand. The campus will collaborate with other parts of the system to support the delivery of graduate and other undergraduate programs needed by the West O'ahu community.

Relevant priorities for UH West O'ahu are:

b. Give priority to the continued accreditation of the institution by the Western Association of Schools and Colleges.

Each of the roles, missions, and priorities identified in the *University of Hawai'i Strategic Plan* will be implemented through development of the proposed UH West O'ahu campus. Educational and cultural services in the region will be enhanced and facilities for distance education will provide all residents with greater access to higher education. Campus facilities will be developed in accordance with a long-range academic development plan to ensure continued accreditation of the institution.

State Recreation Functional Plan. This plan addresses government action toward the acquisition of recreation areas and accesses, the establishment and development of areas and facilities, and the management and usage of recreation resources. New recreational opportunities will be created by the UH West O'ahu. Recreational facilities within the campus may include tennis courts, basketball/volleyball courts, and jogging paths. Parks will be provided within the property to meet the requirements of the City's Park Dedication Ordinance.

State Transportation Functional Plan. This plan discusses the need to expand and improve the transportation system and close the gap between where people live and work through decentralization, mixed zoning, and related initiatives. The UH West O'ahu will provide education and employment opportunities in the 'Ewa region. The campus is conveniently located near major transportation facilities and residential developments, thus encouraging public transportation and walking or bicycling. West O'ahu residents will be able to live, work, and attend school within this region, alleviating traffic to other universities and workplaces in Honolulu.

5.2.7 103D-407 Hawaii Revised Statutes

Hawaii Revised Statues 103D-407 requires the use of recycled glass in paving materials wherever possible, specifically:

(a) When purchasing roadway materials or other high-value, end-use applications for public projects, state agencies shall, and county agencies may, purchase materials with minimum recycled glass content meeting specifications adopted by the policy board which, at a minimum, shall provide for:

(1) A minimum recycled glass content of ten per cent crushed aggregate in treated or untreated basecourse in paving materials that shall not reduce the quality standards for highway and road construction; and

(2) The use of one hundred per cent aggregate in nonstructural capital improvement applications.

(b) All highway and road construction and improvement projects funded by the State or a county or roadways that are to be accepted by the State or a county as public roads shall utilize a minimum of ten per cent crushed glass aggregate as specified by the department of transportation in all basecourse (treated or untreated) and subbase when the glass is available to the quarry or contractor at a price no greater than that of the equivalent aggregate.

(c) All state and county construction projects calling for nonstructural backfill shall utilize one hundred per cent crushed glass when available at a cost equal to or lower than the equivalent aggregate.

It is the University of Hawaii's intent to comply with the above law.

5.2.8 103D-408 Hawaii Revised Statutes

Hawaii Revised Statutes 103D-408 requires the use of native Hawaiian flora whenever and wherever possible, specifically:

(a) Wherever and whenever feasible, all plans, designs, and specifications for new or renovated landscaping of any building, complex of buildings, facility, complex of facilities, or housing developed by the State with public moneys shall incorporate indigenous land plant species as defined in section 195D-2, and plant species brought to Hawaii by Polynesians before European contact, such as the kukui, noni, and coconut; provided that:

(1) Suitable cultivated plants can be made available for this purpose without jeopardizing wild plants in their natural habitat; and

(2) Wherever and whenever possible, indigenous plants shall be used for landscaping on the island or islands on which the species originated.

(b) Each plant or group of plants used pursuant to subsection (a) shall be clearly identified with signs for the edification of the general public. [L 1999, c 149, pt of §2]

It is the University of Hawaii's intent to comply with the above law and to further comply with sustainability principles by: 1) selecting plants suited to the climate of the project site, 2) specifying plant materials, that would be irrigated, with similar irrigation requirements; and 3) using non-drinkable water for irrigation, where feasible.

5.3 CITY AND COUNTY OF HONOLULU

5.3.1 General Plan

As required by the City Charter, the *General Plan* for the City and County of Honolulu establishes long-range social, economic, environmental, and design objectives for the general welfare and prosperity of the people of O'ahu. It also establishes broad policies to achieve the objectives. Those objectives and policies applicable to the UH West O'ahu project are discussed in the following pages.

POPULATION

Objective C, Policy 2: Encourage development within the secondary urban center at Kapolei and the Ewa and Central Oahu urban-fringe areas to relieve developmental pressures in the remaining urban-fringe and rural areas and to meet housing needs not readily provided in the primary urban center.

Objective C, Policy 4: Seek a year 2010 distribution of Oahu's residential population...

Discussion: According to the DPP, the population of the *Ewa Development Plan* Area (Ewa DPA) grew from 42,931 in 1990 to 68,718 in 2000, representing an increase of 60.1 percent (DPP 2004). In comparison, the population for the City as a whole increased only 4.8 percent from 836,231 in 1990 to 876,156 in 2000. Residential development in the 'Ewa region continues to increase in population, and the City expects the population of the Ewa DPA to reach 141,864 and 164,462 by 2020 and 2025 as the secondary urban center, respectively. The City's population is expected to reach 1,037,250 and 1,078,050 in 2020 and 2025, respectively.

At full build-out, the UH West O'ahu development will introduce new residents to the 'Ewa region and will provide approximately 4,041 residential units (including 761 student housing units). Using an estimate of 2.9 persons per household, the 3,280 residential units (not including student housing units) will provide homes to approximately 9,512 people. This population increase is consistent with stated governmental policies of directing future growth toward the 'Ewa Plain. Within the property, the proposed elementary school and UH West O'ahu campus will provide for the educational needs of students residing within the property and the surrounding area.

NATURAL ENVIRONMENT

Objective A, Policy 4: Require development projects to give due consideration to natural features such as slope, flood and erosion hazards, water-recharge areas, distinctive land forms, and existing vegetation.

Discussion: The 'Ewa Plain has long been planned for urban development in accordance with both the City and State planning policies. The UH West O'ahu development will be compatible with existing land use patterns. Potential hazards include flooding during intense storms and soil erosion during construction. Since much of the property currently consists of exposed soil and scrub vegetation, overall soil loss will likely be reduced after development. Proposed improvements, including extensive drainage control measures, will mitigate the potential flood hazard and control off-site drainage flows. There are no slope hazards or distinctive landforms on the property, which consists of actively cultivated land and scrub vegetation.

TRANSPORTATION AND UTILITIES

Objective A, Policy 2: Provide transportation services to people living within the Ewa, Central Oahu, and Pearl City-Hawaii Kai corridors primarily through a mass transit system including exclusive right-of-way rapid transit and feeder-bus components as well as through the existing highway system with limited improvements as may be appropriate.

Objective D, Policy 5: Require the installation of underground utility lines wherever feasible.

Discussion: The UH West O'ahu development will provide education and employment opportunities near the City of Kapolei, and will help reduce or eliminate the need for many students, faculty, and administrative personnel to commute into Honolulu. Transit nodes generally consisting of future mass transit stations are planned along the proposed North-South Road and Kapolei Parkway. These bus stops and future mass transit stations (as envisioned by the City's *Ewa Development Plan*) would be located at the intersections of North-South Road with Farrington Highway and Kapolei Parkway. The City has also proposed to locate a park-and-ride bus facility on the Koko Head side of North-South Road, between Farrington Highway and the H-1 Freeway.

Where feasible, new utility lines servicing the UH West O'ahu development will be placed underground.

PHYSICAL DEVELOPMENT AND URBAN DESIGN

Objective A, Policy 4: Require new developments to provide or pay the cost of all essential community services, including roads, utilities, schools, parks, and emergency facilities that are intended to directly serve the development.

Objective A, Policy 5: Provide for more compact development and intensive use of urban lands where compatible with the physical and social character of existing communities.

Objective C: Develop a secondary urban center in Ewa with its nucleus in the Kapolei area.

Objective D, Policy 1: Develop and maintain urban-fringe areas as predominantly residential areas characterized by generally low rise, low density development which may include

significant levels of retail and service commercial uses as well as satellite institutional and public uses geared to serving the needs of households.

Objective E, Policy 3: Encourage distinctive community identities for both new and existing districts and neighborhoods.

Objective E, Policy 5: Require new developments in stable, established communities and rural areas to be compatible with the existing communities and areas.

Discussion: The UH West O'ahu property is located within the secondary urban center at Kapolei and the proposed project is consistent with the urban growth policy established for the 'Ewa Plain. The development will create an important regional center and facilitate social and cultural events typically associated with a major university. It will also support the needs of the projected population and improve West O'ahu residents' access to higher education facilities. Nearby employment centers currently exist or are planned in the City of Kapolei, Campbell Industrial Park, and Kapolei Business Park. On-site infrastructure required for the project will be provided by the University of Hawai'i.

PUBLIC SAFETY

Objective B, Policy 2: Require all developments in areas subject to floods and tsunamis to be located and constructed in a manner that will not create any health or safety hazard.

Discussion: The study area is not subject to tsunami. With extensive drainage improvements planned for the property, storm water runoff will be managed and all potential health and/or safety hazards will be mitigated. Storm water flows will be controlled to limit off-site discharges and permit on-site detention and recharge of storm water.

HEALTH AND EDUCATION

Objective B: To provide a wide range of educational opportunities for the people of Oahu.

Policy 5: Facilitate the appropriate location of learning institutions from the preschool through the university levels.

Objective C: To make Honolulu the center of higher education in the Pacific.

Policy 1: Encourage continuing improvement in the quality of higher education in Hawaii.

Policy 2: Encourage the development of diverse opportunities in higher education.

Policy 3: Encourage research institutions to establish branches on Oahu.

Discussion: The proposed UH West O'ahu campus will provide higher education opportunities to the residents of 'Ewa, Wai'anae, and Central O'ahu. The mission statement of the UH West O'ahu emphasizes quality teaching and flexible class schedules to foster life-

long learning, enabling students to pursue career-related education coupled with the values, ideas, and challenges of the liberal arts. By offering approximately half of all courses in the evening or during the weekend, UH West O'ahu strives to meet the educational needs of both the recent high school graduate and non-traditional student. The academic program structure stresses the exploration of interdisciplinary studies, cross-cultural and international studies, and communication skills.

5.3.2 Ewa Development Plan

The *Ewa Development Plan* (Ewa DP) was updated and officially adopted by the City Council in August 1997. The plan presents a vision for 'Ewa's future development and provides conceptual land use plans that will serve as a policy guide for more detailed zoning maps and regulations, and public and private sector investment decisions. Provisions of the Ewa DP that are applicable to the UH West O'ahu development are discussed in the following pages.

VISION STATEMENT

The vision for 'Ewa, as expressed in the Ewa DP, embraces two horizons: the first from the present to the year 2020 and the second horizon extending beyond 2020, when 'Ewa would be fully developed. In support of the City *General Plan* policies, the Ewa DP:

- Provides a secondary employment center with its nucleus in the City of Kapolei to supplement the Primary Urban Center (PUC) and to divert commuter traffic from the PUC, and
- Concentrates primary employment activities at industrial and resort areas and at government service and higher education centers near the City of Kapolei, creating secondary markets for office and retail activities.

The Ewa DP projects significant population growth and will help to establish 'Ewa as a Secondary Urban Center for O'ahu. Population is expected to grow from 43,000 in 1990 to 125,000 in 2020, and by then, nearly 28,000 new housing units will have been built in a series of master-planned communities. Employment opportunities will grow from 17,000 jobs to more than 64,000 jobs. Beyond 2020, implementation of the Ewa DP will have established an Open Space Network within an Urban Growth Boundary.

The City of Kapolei will evolve into the Secondary Urban Center for O'ahu, supporting a mix of commercial, office, and residential uses. According to the Ewa DP, "by 2020, it is projected that the City of Kapolei will house more than 7,000 residents and provide 25,000 private sector jobs and 7,000 public sector jobs." Ko Olina and 'Ewa Marina will also add over 3,700 visitor units by the year 2020, providing basic jobs which will support office and commercial jobs in the City of Kapolei.
Master-planned communities envisioned by the Ewa DP are East Kapolei, 'Ewa by Gentry, 'Ewa Marina, 'Ewa Villages, Ko Olina, Laulani, Makaīwa Hills, Makakilo, and the Villages of Kapolei. These communities will interact and support the vision for development of the entire 'Ewa region.

Discussion: The UH West O'ahu development represents one of the major land use elements envisioned by the Ewa DP and is a logical extension of much needed higher education opportunities, which are presently provided at a temporary campus adjacent to the Leeward Community College in Pearl City. The UH West O'ahu campus will support a total population of 8,640 students, faculty, and staff. The campus is envisioned to include an extensive open space system that will ultimately connect with future development at Barbers Point through pedestrian walkways, bike paths, and a transit corridor along the proposed North-South Road.

KEY ELEMENTS OF THE VISION

The Ewa DP establishes 12 key elements to ensure that the vision statement for the 'Ewa region is achieved. The following describes those key elements that are applicable to the UH West O'ahu project.

Urban Growth Boundary. The entire UH West O'ahu property is located within the Urban Growth Boundary of the Ewa DP Urban Land Use Map (see Figure 5.1). The property was used in the past for plantation agriculture; however, the tremendous demand for irrigation water, lost profitability, and proximity of the site to future urban growth areas directed this area to be located within the Urban Growth Boundary. The proposed project is also consistent with the conceptual maps of the Ewa DP, which depict Open Space, Urban Land Use, Public Facilities, and Phasing concepts.

Network of Open Space and Greenways. Kalo'i Greenway and Hunehune Greenway will include a pedestrian/bike path that will connect the campus with residential and mixed-use commercial developments surrounding the campus. The greenway system is also envisioned to link the two proposed transit stations on the eastern portion of the property (adjacent to the North-South Road) with the University and surrounding mixed-use and residential communities. The greenway system, for the most part, will be unimpeded by vehicular traffic.

Plazas, courtyards, other landscaped areas, and the 10-acre detention basin will provide open space within the campus. The 11.2-acre detention basin at the southern boundary of the property will serve as open space, and parks will be provided within the development to meet the requirements of the City's Park Dedication Ordinance.

The Secondary Urban Center. The Secondary Urban Center would function as the primary employment generator for the Ewa DP area by the year 2020. Major new employment centers are planned at the City of Kapolei, Kapolei Hale (the Kapolei Civic Center completed

in 2001), Ko Olina, the deep draft harbor, Campbell Industrial Park, Barbers Point Naval Air Station (with civilian reuse), and the UH West O'ahu development. Although most major employment opportunities generated by the project will be during the construction period, the operational employment after build-out will also be substantial. This would facilitate the larger goal of developing a "self-contained" second city whereby residents can live, work, play, attend school, and shop within one community, instead of enduring long commute times. The Ewa DP states, "Many of the jobs in the City of Kapolei will be supported by development of the UH West O'ahu Campus which is expected to have 7,600 students and 800 staff and faculty by 2020." At this stage of planning, the UH West O'ahu campus is expected to have 7,600 students and 1,040 faculty and staff by the target enrollment date of 2015.

Communities Designed to Support Non-automotive Travel. According to the *Ewa Development Plan,* master-planned residential communities will be designed to support pedestrian, bicycle, and public transit uses. A rapid transit corridor along the proposed North-South Road will link the City of Kapolei, Villages of Kapolei, UH West O'ahu campus, and Waipahu. High-density residential developments will be built along the corridor within walking distance of the major nodes and transit stops. Since the UH West O'ahu will provide employment and educational opportunities to nearby residents, it will be connected to all residential communities. The proposed University Village will be located at the intersection of Farrington Highway and the proposed North-South Road. University Village is envisioned as a sustainable, mixed-used village closely integrated with the campus. It will serve as the University's town center and provide a mix of uses.

Preservation and Enhancement of Cultural Resources. The Ewa DP states:

Ewa's Historic and Cultural Resources will be preserved and enhanced by...retaining visual landmarks and significant vistas, including:

- Distant views of the shoreline from the H-1 Freeway above the Ewa Plain;
- Views of the ocean from Farrington Highway between Kahe Point and the boundary of the Waianae Development Plan Area;
- Views of the Waianae Range from H-1 Freeway between Kunia Road and Kalo'i Gulch and from Kunia Road;
- Views of na puu at Kapolei, Palailai, and Makakilo;
- Mauka and makai views; and
- Views of central Honolulu and Diamond Head.

Structures within the proposed UH West O'ahu campus could impact distant views of the shoreline from the H-1 Freeway (which is at a higher elevation than the project site); however, the UH West O'ahu will serve as an important visual landmark for the 'Ewa region. Views of the Ko'olau Mountain Range, however, would be created from the property, which is currently inaccessible to the public. The visual appearance of the UH West O'ahu property will change from vacant scrub and cultivated vegetation to a built campus environment. The campus will be oriented to capitalize on views of landforms such as Pu'u Kapolei, Pu'u Palailai, Pu'u Makakilo, the Wai'anae Mountain Range, and Central O'ahu. Views from future internal roadways will be considered to the extent possible; however, because the site



Source: City & County of Honolulu Department of Planning and Permitting

EWA DEVELOPMENT PLAN

Urban Land Use Map

 Low and Medium Density Residential

 High Density Residential

 Community Commercial Center

 City of Kapolei (High Density Residential and Commercial)

 Resort/Recreation Area

 Industrial

 Military

 Public Institution

 Agricultural and Preservation

 Parks and Golf Courses

 Transit Node (High Density Residential and Commercial)

 Urban Growth Boundary

EXISTING FUTURE

.

0

(III)

胷

Ð

Δ

Â

人

-

0-0-0

smN

Þ

Щ

Civic Center

Electric Power Plant

Wastewater T.P.

Intermediate School

High School

U.H. West Oahu

Hospital

Δ

Small Boat Marina

Commercial Harbor

Airfield

Highways, Arterial & Major Streets

Historic Railway

Figure 5.1 Ewa Development Plan Urban Land Use Map University of Hawai'i West O'ahu

NOT TO SCALE



is relatively flat, views that do exist will be impacted primarily by the siting of future buildings, landscaping, and other campus features. Extensive landscaping, campus view corridors, and thoughtful architectural design will add to the visual character of the area.

Phased Development. The Ewa DP has set forth a phasing program to ensure that future development more efficiently utilizes existing and planned infrastructure. Major projects are categorized by the time period or phase in which they can apply for a zoning change. As a public facility, the UH West O'ahu campus is a permitted land use within the existing AG-1 Restricted Agricultural District, with a PRU approval. However, a change of zoning is required for development of the non-campus lands within the property.

LAND USE POLICIES, PRINCIPLES, AND GUIDELINES

According to the Ewa DP, the vision for development of 'Ewa will be implemented through the application of land use general policies, principles, and guidelines. The following describes those provisions applicable to the UH West O'ahu development.

Open Space Preservation and Development. Kalo'i Greenway and Hunehune Greenway will include a pedestrian/bike path that will connect the campus with residential and mixed-use commercial developments surrounding the campus. The greenway system is also envisioned to link the two proposed transit stations on the eastern portion of the property (adjacent to the North-South Road) with the University and surrounding mixed-use and residential communities. The greenway system, for the most part, will be unimpeded by vehicular traffic.

Plazas, courtyards, landscaped areas, and detention basins within the property will serve as open space, and parks will be provided within the development to meet the requirements of the City's Park Dedication Ordinance.

Historic and Cultural Resources. According to an archaeological reconnaissance and assessment and a letter from SHPD, there are no significant historic or cultural resources associated with the property (see Appendix E). However, should any archaeologically significant artifacts, bones, or other indicators of previous on-site activity be uncovered during construction phases of development, construction will halt and archaeological resources will be treated in strict compliance with the requirements of the SHPD.

UNIVERSITY OF HAWAI'I WEST O'AHU

Section 3.7.6.1 of the Ewa DP contains general policies, planning principles, and guidelines for development of the UH West O'ahu campus.

General Policies. According to the Ewa DP, the UH West O'ahu campus should evoke a unique sense of place that distinguishes it as an important civic and cultural institution in 'Ewa. The Ewa DP states that the projected enrollment is 2,800 students by the end of

construction, which is to begin no later than the end of December 2011, and 7,600 students by 2020, and the projected faculty and staff are 800. However, an initial campus for 1,520 students is currently planned to be occupied by the Fall 2009 semester. The ultimate 7,600-student campus would be developed over several years following the initial development.

The Ewa DP also says that the campus should be oriented to support pedestrian access and transit usage from a major transit node located on the North-South Road. Development of the UH West O'ahu campus should include plans to provide shuttle bus service to the transit node at the corner of Farrington Highway and the proposed North-South Road, and the campus should be designed so that open space areas can be used for flood detention as part of the Kalo'i Gulch watershed master plan.

The proposed UH West O'ahu campus will be designed to be compatible with the character and context of the region. It will also incorporate measures to enhance pedestrian access and transit usage.

Planning Principles. The following general planning principles will be used as a framework for design of the UH West O'ahu campus:

Cultural Sensitivity. University development should be environmentally and culturally sensitive to the site and reflective of the Hawaiian culture and of the heritage of Ewa.

Discussion: The proposed campus will be environmentally and culturally sensitive to the site. These concepts will be implemented during the review of the UH West O'ahu LRDP and the City's PRU processing procedures. No culturally-significant sites were identified within the property, and therefore, no impacts on cultural resources are anticipated.

Regional Integration. The campus should function as a fully integrated community within the context of the broader regional community. The campus should include housing, support services, community and business facilities, in addition to the required academic facilities.

Discussion: The proposed campus will be designed to include academic facilities and support services, including housing, as warranted in the future. The campus will also include community and business facilities and extensive recreational facilities.

Community Orientation and Service. The campus should be community-oriented and should serve the Kapolei area and West Oahu as an urban park and cultural center, providing community services, cultural opportunities, and remedial educational opportunities.

Discussion: In addition to providing higher education facilities, the UH West O'ahu campus will provide cultural activities. Extensive open space and cultural facilities (i.e., performing arts theater/auditorium) have been incorporated into the educational specifications for the campus.

Functional and Accessible Design. Campus design should reflect appropriate functional

relationships, internal compactness, and accessibility between academic functions and supporting facilities, providing a pleasant and efficient study environment.

Discussion: Functional relationships, internal compactness, and accessibility were considered in the design of the UH West O'ahu campus. Educational specifications and functional relationships were identified and coordinated to create a well-functioning campus. Parking will be conveniently located near major campus structures, and an internal loop road will enhance accessibility while minimizing the need for vehicular transportation within the internal pedestrian portions of the campus.

Drainage Impacts. A large portion of the campus lies within the Kaloi Gulch watershed. In order to reduce the downstream impact of major storm events, the campus open space system should incorporate flood detention and retention capability. For example, sports playing fields could be designed to act as flood detention basins during major storm events.

The drainage plans for the Campus should not increase storm water flows or velocity above the design levels used in designing the water retention areas of the Ewa Villages Golf Course and the drainage systems for earlier developments in the Kaloi Gulch watershed.

Discussion: The proposed drainage system for the UH West O'ahu development will ensure that existing downstream storm water runoff quantities are maintained. A 10-acre detention basin within the campus and an 11.2-acre detention basin at the southern boundary of the property will maintain existing off-site runoff quantities throughout project development and operation.

Guidelines.

Architectural Forms

- Specific activity areas and structures should be sited and designed to accommodate required internal academic or support relationships. This would include siting of buildings or facilities so as to promote academic continuity, provide spatial definition to public areas and allow easy access to needed support areas (housing, business/food services, recreation, and parking).
- Buildings and structures should reflect a sensitivity to the local environmental conditions as well as to Hawaiian regional styles.
- Structures should not visually dominate the site. Rather, low rise academic structures with more emphasis on regional architectural forms and human scale should prevail.

Discussion: The UH West O'ahu development will incorporate the architectural guidelines listed above. Activity areas and structures will be sited to take full advantage of internal academic and support relationships. The campus will be designed to provide a hierarchy of open spaces from formal plazas to intimate gathering areas, encouraging interaction. An interconnected series of smaller pedestrian paths will extend outward from the malls and major pathways, linking the various campus activities. The University Village is envisioned as a sustainable, mixed-use village, closely integrated with the campus. University Village will serve as the University's town center and will be the transition between the campus and

the community. Housing and recreational facilities will be clearly defined by circulation patterns, topography, and building type and design. Low- to mid-rise buildings and structures will be designed to reflect an appropriate architectural character. Extensive landscaping will soften the distinction between the built environment and the surrounding natural features and plant materials. As such, buildings will not visually dominate the property, especially from off-site public viewing points.

Landscape Forms

- Trees and other landscape materials should be used throughout the campus to provide welcome shade and visual relief.
- Street trees and accent plantings should be used to feature gateways, define circulation corridors or enhance special activity areas. The intensity or selection of landscape treatments should be used to further define, identify or buffer various campus land uses.
- Landscape materials should be used which reflect climate conditions, limited water resources, and maintenance issues.
- Use of native/indigenous species should be incorporated into landscape treatments to the greatest extent possible.

Discussion: Landscaping for the UH West O'ahu development will be designed to incorporate the guidelines listed above. Landscaping that is uniquely representative of UH West O'ahu will reinforce the organization and hierarchy of campus elements. Areas drawing upon a common focus (i.e., malls, plazas, or courtyards) would be planted with the same species of trees or palms. These theme plantings will create a strong focus and make common areas easily identifiable. Plants will also be used in formal patterns to complement the layout of the buildings, entries, malls, and plazas. Informal arrangements of trees can be placed in open areas between buildings, creating a softer character more conducive to student relaxation. The consistent use of a selected palette of paving materials, site furniture, public art, lighting, and plant material will establish a common theme, create a strong sense of place, and unify the campus environment.

The 1991 Hawai'i State Legislature passed Act 73, which "encourage(s) the propagation of Hawai'i's indigenous species of land plants by requiring that they be employed, where feasible, in the landscaping of public buildings, facilities, and housing projects developed by the State." Native plants existing on the site or in surrounding State lands, as well as other native species found in the dry to mesic forests of the 'Ewa Plain and Wai'anae Mountain Range, would be used in the landscape design.

Circulation

- Circulation patterns should provide for easily accessed routes to, within, and around the campus. Conflicts between cars, bikes, and pedestrians should be minimized.
- The hierarchy of roadway, bikeway, and pedestrian circulation patterns should be highlighted by a distinctive design treatment for each element of the system.
- Potential visual impacts from vehicle corridors and parking lots should be minimized through appropriate site design and placement.
- Provisions for public transportation with ties to the regional system and transit corridor should be an integral part of the campus plan.

Discussion: Circulation guidelines have been incorporated into the design of the UH West O'ahu development. A hierarchical interconnected roadway network is proposed for the project. The hierarchy of roadway, bikeway, and pedestrian circulation patterns will utilize design methods (i.e., different paving materials and plant materials) to distinguish each element of the system. The fire access plan will be reviewed by the HFD during the PRU process. Access to the campus will be provided from Farrington Highway and North-South Road. A loop roadway system is provided so as not to disrupt pedestrian traffic within the core of the campus. Parking lots will be located between the internal loop road and campus structures to mitigate potential visual impacts. Landscape buffers will also be utilized to screen parking areas from Farrington Highway. The internal circulation system will also incorporate provisions for public transportation by connecting to the regional system and transit corridor in the proposed North-South Road.

Open Space/Views

- Open space components should be integrated and blended throughout the campus in the form of passive landscape areas, courtyards, mall spaces, and multi-purpose recreation fields or community spaces.
- The internal campus open space system should provide links with the adjoining regional open space systems of the adjacent developments.
- Development of campus gateways and enhancement of internal view corridors should be an integral part of the open space elements within the campus.
- Campus development should preserve and enhance mauka-makai views within major open spaces and through building siting.
- Visual buffering through landscape treatments or building design should occur between conflicting or unsightly functions.

Discussion: The UH West O'ahu development will incorporate the open space/views guidelines listed above. A hierarchy of open spaces will be provided within the campus and surrounding residential developments. The campus open space element includes a central plaza, informal courtyards, and open spaces. Two pedestrian malls enhanced with lighting (for safety), consistent paving patterns and materials, and site furnishings will also help to define the open space network. The campus will be connected to regional open space systems adjacent to the campus through the Hunehune and Kalo'i greenway, which will extend from the campus to the surrounding community. Mauka-makai views from the property will be enhanced, focusing on views of the Wai'anae and Ko'olau Mountains, and landscape treatments and building design will mitigate visual impacts of the campus.

OTHER COMMUNITY FACILITIES

Other existing and proposed community facilities shown on the Ewa DP Urban Land Use Map (see Figure 5.1) include hospitals, colleges, correctional facilities, and cemeteries. The map depicts the location of the formerly planned UH West O'ahu campus (mauka of Farrington Highway) and the St. Francis Medical Center – West. According to the Ewa DP, "colleges and hospitals should generally be located in urban areas near transit nodes,

commercial centers, or high-density residential areas."

Discussion: In 1967, the UH Board of Regents initiated efforts to establish a four-year college in Central or West O'ahu. Over the years, the Regents approved varying sites: Wai'awa Ridge (1970), Honouliuli/Ewa (1973) and in 1993, Kapolei Makai (the current site). In 1996, it was decided that the West O'ahu campus site would be moved to 991 acres mauka of the H-1 Freeway in Kapolei. This was based on a strategy by the State to entitle approximately 1,300 acres of land in East Kapolei, which would then be sold to private developers. As a result, the Housing and Community Development Corporation of Hawaii (HCDCH) initiated a master planning effort for 1,300 acres including the UHWO Kapolei Makai site for mostly residential uses. It was intended that the proceeds of the land sales would be used to fund the construction of a UHWO campus site mauka of the H-1 Freeway. On March 5, 1999, HCDCH filed a Petition for Land Use District Boundary Amendment to reclassify approximately 1,300 acres, including the subject 500 acres, from Agricultural use to Urban use for the master-planned community known as East Kapolei. By Findings of Fact, Conclusions of Law and Decision and Order filed on September 8, 1999, the State Land Use Commission granted this petition. (This mauka campus location is reflected in the current Ewa DP Land Use Map.) Subsequently, however, Hawai'i's economy suffered a serious downturn, and there was little or no market demand for new residential development. During this period, it was determined that the construction of a new campus mauka of the H-1 Freeway would be more expensive than the current site (makai of Farrington Highway).

During the fall of 2001, the UH Administration, the Estate of James Campbell, and the Governor explored the possibility of relocating the UH West O'ahu campus from the 991acre mauka site to the City of Kapolei.

Since the marketability of the petition area was non-existent, and a site selection study determined that the current site was best suited for a new UHWO campus, UHWO was allocated 500 acres and most of the remainder of the 1,300 acres which were "urbanized" was transferred to DHHL.

On September 13, 2002, the UH Board of Regents approved the 500-acre Kapolei Makai site as the permanent location for the new UH West O'ahu campus and requested the transfer of title to the 500 acres from DLNR. The Board of Land and Natural Resources approved the transfer of title of the 500- acre site to the UH in fee simple.

The University of Hawai'i still retains the 991-acre site mauka of the H-1 Freeway, but has no current plans for it other than being generally viewed as potentially being developed in the future as an expansion area for the UHWO campus.

Although the UH West O'ahu development is proposed in a different location than depicted on the Urban Land Use Map, it is still consistent with the Ewa DP. Currently, the unofficial public review draft of the Ewa DP Five-Year Review adds language reflecting the relocation of the UH West O'ahu to the currently proposed site at the Farrington Highway and North-South

Road intersection. Within the property, several acres of commercial and residential uses are planned along the proposed North-South Road, which is planned as a major regional roadway servicing the project area. The North-South Road right-of-way will include a mass transit corridor in its ultimate roadway median. The major roadway network, although modified, follows that proposed in the Ewa DP.

PUBLIC FACILITIES AND INFRASTRUCTURE POLICIES AND PRINCIPLES

The Public Facilities Map (see Figure 4.3) and policies and principles of the Ewa DP are provided to guide the planning and construction of proposed facilities and infrastructure systems. These guidelines have been incorporated into the planning for the UH West O'ahu development.

Public Facilities Map. This map depicts the public facilities and infrastructure needed to implement the vision statement for the Ewa DP. The UH West O'ahu campus is shown in its previously proposed location mauka of Farrington Highway. Currently, the unofficial public review draft of the Ewa DP Five-Year Review adds language reflecting the relocation of the UH West O'ahu to the currently proposed site.

Discussion: In its currently proposed location, the UH West O'ahu development would be consistent with major improvements planned and shown on the Public Facilities Map (i.e., an interchange at the H-1 Freeway, construction of the North-South Road, development of a transit node at the intersection of Farrington Highway and North-South Road, and the makai entry to the campus).

On the currently proposed UH West O'ahu site, a high school is shown in the Public Facilities Map; however, this school was intended to accommodate the anticipated population growth resulting from the HCDCH East Kapolei Development Project. The DOE is currently considering a high school site mauka of the UH West O'ahu site, between the H-1 Freeway and Farrington Highway.

Planned Extensions of the Roadway Network. Planning and development of major roadways is the shared responsibility of the State DOT and the City DTS. Planning and use of the Federal transportation funds is coordinated through the Oahu Metropolitan Planning Organization (OMPO), a joint City-State agency. The OMPO is updating the *2020 Oahu Regional Transportation Plan* based on year 2030 traffic volumes projected to be generated by land uses approved under the previous Development Plan Special Provisions and Land Use Map. In 2002, the City Council passed Ordinance 02-52, which identifies major roadway improvements required in 'Ewa and the cost each residential/commercial developer is required to pay. The roads listed in the ordinance will be required by 2020 to properly serve the anticipated developments. Major roadway improvements include:

• Kapolei Parkway, which is planned as a major east-west corridor, connecting the eastern parts of 'Ewa with the City of Kapolei and employment areas to the west; and

• North-South Road, which will link Kapolei Parkway with Farrington Highway and the H-1 Freeway, and eventually extend mauka of the H-1 Freeway interchange to become part of Makakilo Drive.

Discussion: Major planned and proposed roadway elements, and other transportation system features, have been incorporated into the project planning for the UH West O'ahu development. This includes improvements to Farrington Highway, the proposed North-South Road, and transit nodes at the intersections of the North-South Road with Farrington Highway and the proposed East-West Road.

Ordinance 02-52 is scheduled for review and possible revision next year. Farrington Highway and North-South Road will eventually provide access to the campus. As such, distinct landscaping along the campus entry will be established to reinforce the identity and aesthetic attributes, while reducing the dominance of paved surfaces.

Transit. The Ewa DP states that the City should increase transit service in 'Ewa in order to provide suitable service for peak-hour commuting and to enhance circulation among 'Ewa communities and between 'Ewa and the adjacent Wai'anae and Central O'ahu areas.

Discussion: The UH West O'ahu is conveniently located adjacent to Farrington Highway and the proposed North-South Road, both of which are planned in the Ewa DP as alternative rail transit corridors. The campus is also located near various residential subdivisions and will be designed to incorporate linkages. This will encourage public transportation and walking and bicycling.

Planned Rapid Transit Corridor. As shown on the Public Facilities Map, a rapid transit corridor is planned to connect the City of Kapolei with Waipahu and the Primary Urban Center. The corridor could provide an 'Ewa shuttle service, which could travel back and forth on the transit corridor between Ko Olina, the City of Kapolei, the UH West O'ahu campus, and Waipahu. A commuter service could provide peak-hour express bus service to and from the Primary Urban Center. The corridor could carry express bus service or even higher-speed dedicated transit service during peak-hour commuting. By connecting to the Primary Urban Center via Waipahu, the corridor could provide for a future high-speed connection between UH West O'ahu and Leeward Community College, Honolulu Community College, and UH Mānoa. The 'Ewa rapid transit corridor is planned to run from Waipahu along the Farrington Highway right-of-way, turning south at the North-South Road and west again in the Kapolei Parkway right-of-way to the City of Kapolei.

Discussion: The UH West O'ahu is located adjacent to Farrington Highway and the proposed North-South Road for transit access. Provisions for two transit stops are also incorporated into the campus plan.

Bikeway System. The *Kapolei Area Bikeway Plan* (KABP), published by Campbell Estate in 1991, establishes a comprehensive bikeway network to serve the 'Ewa Plain. The network

would include 56 miles of bikeway facilities, including bike paths (separated from the roadway), bike lanes (4 to 6 feet in width), and bike routes (shared curbside vehicle lane, with minimum width of 12 feet). The KABP is part of the *City of Kapolei Urban Design Plan*, which was adopted by the City Council in 1995. The KABP covers all of 'Ewa, except for military bases in the area. Elements of the KABP have been adopted by the State DOT as part of the State's *Bike Plan Hawaii* (1994).

Discussion: Major bike paths will be designed to run along the access road to the UH West O'ahu campus. Within the campus, bikeways will be incorporated into major roadways and there will be an extensive network of multi-use pedestrian/bike paths.

Water Allocation and System Development. The BWS projects that an additional 35 million gallons per day (MGD) of potable water will be needed in 'Ewa by 2020 to meet projected growth in residential and commercial demand. Long-term demand for non-potable water for existing and new urban irrigation and other urban purposes is estimated to be approximately 26 MGD. Meeting this demand will require reallocation of water within the islandwide system, as well as development of new sources.

Discussion: Based on conventional development, the average potable water demand for the UH West O'ahu project is estimated to be 2.584 million gallons per day (MGD) (1.971 MGD from the 440-foot system and 0.613 MGD from the 215-foot system). The average non-potable water demand for the project is estimated to be 0.324 gallons per day (gpd). Coordination with the BWS and the DLNR Land Division will take place to incorporate this project into the *Water Use and Development Plan* and the *State Water Projects Plan*.

Wastewater Treatment. The City Department of Environmental Services estimates treatment/disposal capacity at the Honouliuli Wastewater Treatment Plant will need to be increased from existing capacity for primary treatment of 38 million gallons per day (MGD) to almost 51 MGD by 2020 to meet projected population and economic growth in 'Ewa and Central O'ahu. In addition, the capacity of specific sewer lines and pump stations will need to be increased.

Discussion: The property has no existing sewer service. A portion of the major trunk sewer system for East Kapolei will run through the UH West O'ahu property. The major trunk sewer line will enter the property from Farrington Highway, run along Road D, and exit the property to North-South Road (through Road F). The sewer line, ranging from 21 to 24 inches in diameter, will carry wastewater from the property (and properties located north of the UH West O'ahu property) to the 24-inch trunk line in the North-South Road. The DHHL currently plans to construct the off-site sewer system to the southeast corner of the property at Road F. The average wastewater flow for the UH West O'ahu development is projected to be 1.68 MGD.

Electrical Power Development. HECO forecasts that increased demand and the proposed retirement of the Honolulu Power Plant from service will create a need for additional island-

wide power generation capacity by 2020.

Discussion: Electricity for the area surrounding the UH West O'ahu property is provided by HECO, which has an available generation capacity (peak demand) of approximately 1,687 megawatts (MW). Peak electrical demand for the project is expected to be 18 MVA during Phase 1 and 55 MVA at full build-out. This increase in demand will occur gradually as the project achieves build-out. Additionally, much of this demand will be offset if the solid waste generated daily by the project is burned at the H-POWER plant, which can produce 1 MW of electricity for every 35 tons of solid waste burned.

Solid Waste Handling and Disposal. Two major solid waste handling and disposal facilities are located in 'Ewa. The H-POWER plant at Campbell Industrial Park is operating at maximum capacity, receiving over 600,000 tons of solid waste each year. The Waimānalo Gulch Landfill is the major active waste disposal site on O'ahu and is projected to run out of capacity within 10 to 25 years. However, the landfill was identified as having potential for expansion.

Discussion: Sustainability guidelines have been established for the UH West O'ahu development. The campus will "appropriately reduce, reuse, and recycle materials to minimize generation of solid waste and achieve diversion from landfills". Solid waste generated by the project is estimated to average approximately 40.46 tons per day after project build-out.

Drainage Systems. The Ewa DP states that public and private agencies should employ methods of retaining or detaining storm water for gradual release into the ground as the preferred strategy for management of storm water.

Discussion: As a result of the UH West O'ahu development, there will be an increase in storm runoff from impermeable surfaces (i.e., roadways, parking areas, and buildings) established on-site. To mitigate this increase in runoff, a 10-acre detention basin within the campus and an 11.2-acre detention basin at the southern boundary of the property will maintain existing off-site runoff quantities. Storm water detained in the basin will be released at a controlled rate after the storm and will be directed to the regional East Kapolei drainage system in the North-South Road.

School Facilities. The DOE faces an enormous shortfall in funding to meet projected needs for new classrooms. As a result, the DOE is asking for developer "fair- share" contributions, exploring alternative school financing options such as lease/purchase agreements, and seeking to increase the number of schools operating year-round and with multi-tracking or double shifts. The Ewa DP states that developers should pay their fair share of all costs needed to insure provision of adequate school facilities for the children living in their developments.

Discussion: The UH West O'ahu will include a new four-year University of Hawai'i campus

initially designed to accommodate 1,520 students (Phase 1), with an ultimate target population of 7,600 students. Additionally, a 12-acre elementary school site is proposed in the southern portion of the property to meet the educational needs of children living in residences within and outside of the property. <u>Also being contemplated is the development of a "Lab School" on the campus site.</u>

Public Safety Facilities. Adequate staffing and facilities are needed to ensure public safety. The Ewa DP states that new development should be approved only if staffing and facilities will be adequate to provide fire and police protection when development is completed. According to the Ewa DP, three fire stations at 'Ewa Villages, Ko Olina, and Makaīwa Hills are planned, but service dates have not been determined.

Discussion: In order to meet the growing public safety needs of the 'Ewa Plain communities, the City constructed the Regional Kapolei Police Station at 1100 Kamokila Boulevard. Fire services in the 'Ewa area are provided by the 'Ewa Beach Fire Station, Makakilo Fire Station, and Kapolei Fire Station. The project will comply with all HFD requirements for access, water supply, and building construction. The HPD and HFD have been and will continue to be consulted in this EIS process (see Sections 8.1 and 8.2).

PUBLIC FACILITY INVESTMENT PRIORITIES

The Ewa DP states that Capital Improvement Projects shall support the development of High Priority Areas during the first eight years (1997 to 2005). The Capital Improvement Projects listed as having the highest priority include:

- The UH West O'ahu campus in the vicinity of Pu'u Kapuai and north of the H-1 Freeway;
- A dedicated Rapid Transit Corridor linking the City of Kapolei, Kapolei Village, the UH West O'ahu campus, and Waipahu;
- The North-South Road and other elements of the 'Ewa Regional Highway Transportation Plan;
- Drainage Plans for Kalo'i Gulch, Kapolei, and West Loch Watersheds;
- New potable and non-potable water sources; and
- Expanded wastewater treatment plant capacity, the reclamation of effluent from the Honouliuli Wastewater plant for non-potable water uses.

Discussion: The directed growth strategy for the 'Ewa region requires the cooperation of both public and private agencies in planning, financing, and constructing infrastructure. These high-priority projects are at various levels of implementation by either State or City agencies. This EA reflects the UH West O'ahu project, although the currently proposed site differs from the previously proposed site mauka of the H-1 Freeway. In February 2005, a Finding of No Significant Impact (FONSI) was issued for the North-South Road and Kapolei Parkway project.

Drainage plans for Kalo'i Gulch are being prepared, and water and wastewater systems in the region are being improved.

Land Use Map. The currently proposed U.H. West O'ahu property is designated for Low and Medium Density Residential, and High Density Residential uses by the 'Ewa Development Plan Urban Land Use Map (see Figure 5.1). A symbol for the UH West O'ahu campus is shown mauka of the H-1 Freeway, reflecting a previously proposed campus site. Currently, it is believed that the Ewa DP Five-Year Review will show the UH West O'ahu at the currently proposed site.

Public Facilities Map. Public facilities identified on the *Ewa Development Plan* Public Facilities Map (see Figure 4.3) include a planned transit connection and a grade-separated interchange at the H-1 Freeway. According to this map, major proposed infrastructure and public facilities impacting the study area include a transit corridor parallel to the proposed North-South Road and a transit node at the intersection of Farrington Highway and North-South Road, adjacent to the property.

5.3.3 Land Use Ordinance

The UH West O'ahu property is presently zoned Ag-1 Restricted Agriculture by the City Land Use Ordinance (LUO) (see Figure 5.2). Within this zoning classification, "public uses and structures" are permitted with a PRU permit approval. Specifically, Section 3.1.60-1 of the Land Use Ordinance states:

Plan review use approval shall be required for the following public and private uses; hospitals, prisons, airports, colleges and universities (except business schools and business colleges), trade or convention centers, and those golf courses described in subsection (e).

Additionally, public uses and structures are defined by the LUO as:

...uses conducted by or structures owned or managed by the federal government, the State of Hawai'i or the city to fulfill a governmental function, activity or service for public benefit and in accordance with public policy. Excluded are uses which are not purely a function, activity or service of government and structures leased by government to private entrepreneurs or to nonprofit organizations. Typical public uses and structures include: libraries, base yards, satellite city halls, public schools and post offices.

The UH West Oahu campus is permitted as a public use and structure and is consistent with the existing zoning classification for the property upon approval of the PRU permit application. However, a change in zoning will be required for development of the non-campus lands within the property.

Under the City's LUO, the UH West Oahu property is currently zoned AG 1, Restricted Agriculture (see Figure 5.2). Universities and colleges are permitted with a Plan Review Use (PRU) approval for any zoning district. The University will seek a PRU permit for the campus and student housing; however, a zone change will be required for the remaining lands within the property.



Legend



- Ag-1 Restricted
- Ag-2 General
- R-3.5 Residential
- **R-5** Residential
- P-2 General
- A-1 Apartment

Figure 5.2 Existing Zoning Map **University of Hawai'i West O'ahu**



Under the City's LUO, the UH West O'ahu property is currently zoned AG-1, Restricted Agriculture (see Figure 5.2). The University will seek a PRU permit for the campus and student housing lands. PRU approval from the Honolulu City Council will only apply to the University and uses directly related to the university. Additionally, DPP requires that the "campus lands" be rezoned to a district other than the currently zoned AG-1; and as such, a zone change will be required to be processed concurrently with the PRU. Rezoning approval will also be needed for the portion of the project that will be developed privately, and which is not related to the university.

Figure 5.3 is a proposed zoning map that shows the proposed zoning districts and corresponding acres for the approximately 500-acre UHWO property. Table 5.1 identifies, lists and discusses the zoning districts to which both the campus and the non-campus lands are proposed to be rezoned. With rezoning and PRU approval and at full development, the proposed project will include 4,041 housing units.

LAND USE	<u>ACREAGE</u>	<u>UNITS/</u> <u>ACRE</u>	<u>RESIDENTIAL</u> <u>UNITS</u> (approximate)	COMMERCIAL SQUARE FOOTAGE	
UH WEST O'AHU LANDS					
Lands Covered Under the Plan Review Use Peri	nit (to be Zone	ed R-5)			
7,600 Student Campus (including 10-acre detention basin and possible "Lab School")	<u>103.5</u>	-	Ξ	Ξ	
<u>Student Housing or Campus Expansion</u> <u>Parcel B</u>	<u>12.1</u>	<u>19</u>	<u>230</u>	Ξ	
Roads	3.7	-	-	-	
Subtotal for PRU (Zoned R-5)	<u>119.3</u>	-	<u>230</u>	-	
Business Mixed Use District (BMX-3)					
Mixed Use (Retail, Office, Residential)	<u>15.1</u>	<u>10</u>	<u>151</u>	<u>164,439</u>	
<u>Campus Expansion/Multi-Family</u> <u>Housing/Mixed Use</u>	<u>22.2</u>	<u>16</u>	<u>355</u>	Ξ	
Student Housing/ Mixed Use or Campus Expansion Parcel A (531 student housing units, 115 residential units)	<u>38.0</u>	<u>17</u>	<u>646</u>	<u>331,056</u>	
Mixed Use Parcel C	<u>10.2</u>	<u>10</u>	<u>102</u>	<u>111,078</u>	
HECO Substation	<u>1.0</u>	-	-	-	
Roads	8.0	-	-	-	
Subtotal for BMX-3 District	<u>94.5</u>	-	<u>1,254</u>	<u>606,573</u>	
Total for UH West O'ahu Lands	<u>213.8</u>	-	<u>1,484</u>	<u>606,573</u>	
PRIVATE DEVELOPMENT LANDS					
Business Mixed Use Community District (BMX-3)					
Mixed Use Parcel A	<u>10.5</u>	<u>10</u>	<u>105</u>	<u>114,345</u>	
Mixed Use Parcel B	11.2	10	112	121,968	
Detention Basin	11.2	-	-	-	
HECO Substation	<u>1.0</u>	-	-	-	

Table 5.1 – Pro	posed Plan Review	Use Permit/Zone	Change Land U	Jse Acreage

LAND USE	<u>ACREAGE</u>	<u>UNITS/</u> <u>ACRE</u>	RESIDENTIAL UNITS (approximate)	COMMERCIAL SQUARE FOOTAGE
Roads	<u>3.9</u>	- 1	=	=
Subtotal Total BMX-3 District	<u>37.8</u>	-	<u>217</u>	<u>236,313</u>
Apartment Medium Density District (A-2)				
<u>Residential Parcel A (High Density, Multi-</u> <u>Family)</u>	<u>33.1</u>	<u>16</u>	<u>530</u>	=
<u>Residential Parcel B (Medium Density, Multi Family)</u>	<u>20.1</u>	<u>12</u>	<u>241</u>	-
<u>Residential Parcel F (High Density, Multi</u> <u>Family)</u>	<u>24.7</u>	<u>16</u>	<u>395</u>	=
<u>Residential Parcel G (Medium Density, Multi Family)</u>	<u>20.7</u>	<u>12</u>	<u>248</u>	=
Roads	<u>9.9</u>	-	-	-
Subtotal A-2 District	<u>108.5</u>	-	<u>1,414</u>	-
<u>Residential District (R-5)</u>			1	ſ
<u>Residential Parcel C (Low Density, Single</u> <u>Family)</u>	<u>30.5</u>	<u>6</u>	<u>183</u>	Ξ
<u>Residential Parcel D (Low Density, Single</u> Family)	<u>30.3</u>	<u>6</u>	<u>182</u>	Ξ
Elementary School	12.0	-	-	_
Roads	4.3	-	-	-
Subtotal R-5 District	<u>77.1</u>	•	<u>365</u>	-
<u>Residential District (R-3.5)</u>				
<u>Residential Parcel E (Medium-Low</u> <u>Density, Multi-Family)</u>	<u>56.1</u>	<u>10</u>	<u>561</u>	Ξ
Roads	<u>7.0</u>	-	<u>-</u>	<u>-</u>
Subtotal R-3.5 District	<u>63.1</u>	-	<u>561</u>	-
Total Private Development Lands	286.5	<u> </u>	2,557	236,313
GRAND TOTAL	500.3	-	4,041	842,886

During the public review period, the Department of Planning and Permitting, Land Use Permits Division, Zoning Regulations and Permits Branch wrote: *"The FEIS should discuss how development of the campus lands will or will not comply with LUO development standards, such as for height, setbacks and density, for the proposed underlying zoning districts in which the university and university-related uses will be located."* While the exact "exemptions" and "variances" are not known at this time, UHWO's architect, John Hara & Associates, Inc. has provided the following preliminary opinion on how the campus design may or may not comply with LUO development standards for R-5 zoning. Building heights on campus will vary, but will typically exceed the 25-30 foot maximum height required by R-5 zoning. Buildings in the first phase will be approximately 60 feet high at the ridge, with the campus tower reaching approximately 85 feet above finish grade. Setbacks of buildings will meet R-5 zoning requirements. The buildings will be set back more than 30 feet at the front and more than 15 feet at the side and rear property



Proposed Zoning Map University of Hawai'i West O'ahu



lines.

The height setbacks will also be met. Any portion of a structure exceeding 15 feet will be set back from every side and rear buildable area boundary line at least one foot for each two feet of additional height over 15 feet. Any portion of a structure exceeding 20 feet will be set back from the front buildable area boundary line at least one foot for every two feet of additional height over 20 feet.

Density of the site will meet R-5 zoning requirements. The maximum building areas will not exceed 50 percent of the zoning lot.

5.3.4 Special Management Area

The UH West O'ahu property is located outside of the Special Management Area established by the City.

5.4 OTHER REQUIRED PERMITS AND APPROVALS

The following is an approximate list of major approvals and permits required for the implementation of the proposed project. From the earliest stages of the planning process, the University of Hawai'i has worked with all affected agencies to obtain their comments and necessary approval of plans and specifications.

Table 5.2 – Required Approvals and Permits

PERMIT OR APPROVAL	AUTHORITY	STATUS	
Environmental Impact			
Statement (EIS) in compliance	Governor, State of Hawai'i		
with Chapter 343, HRS			
Motion to Amend Decision	Land Use Commission	Motion to be filed at the	
and Order		time the FEIS is submitted.	
		Application to be	
	Department of Planning and	submitted after the	
Plan Review Use (PRU) Permit	Permitting / City Council	acceptance of the FEIS	
	0. /	concurrently with the Zone	
		Change application.	
		Application to be	
Zana Changa	Department of Planning and	submitted alter the	
Zone Change	Permitting / City Council	acceptance of the FEIS	
		Concurrently with the FKO	
		Application to be	
	Department of Planning and	submitted after the Zone	
Subdivision Approval	Permitting	Change application is	
	i cinimang	approved	
		Application to be	
	Department of Planning and	submitted with an	
Park Dedication	Permitting	application for Subdivision	
		Approval.	
		Application to be filed	
Building/Grading Permits	Department of Planning and	after the Zone Change	
	Permitting	application is approved.	
		Currently coordinating	
Installation of Power Lines and	State Public Utilities	with HECO. Will be	
<u>Substations</u>	<u>Commission</u>	approved prior to	
		occupancy.	
National Pollutant Discharge		Application to be	
Elimination System (NPDES)	State Department of Health	submitted prior to Grading	
Permit		Permits.	

6.0 Alternatives to the Proposed Action

6.0 ALTERNATIVES TO THE PROPOSED ACTION

6.1 ALTERNATIVES CONSIDERED

In compliance with Section 11-200-17(f), HAR, this EIS describes the alternatives that could attain the objectives of the action, regardless of cost, and explains why they were rejected. The environmental impacts of each of the alternatives have also been evaluated. The following sections describe the alternative sites and alternative site uses that have been investigated.

6.2 ALTERNATIVE SITES

In 1967, the UH Board of Regents initiated efforts to establish a four-year college in Central or West O'ahu. Over the years, the Regents approved varying sites: Wai'awa Ridge (1970), Honouliuli/'Ewa (1973) and in 1993, Kapolei Makai (the current site). In 1996, it was decided that the West O'ahu campus site would be moved to 991 acres mauka of the H-1 Freeway in Kapolei. This was based on a strategy by the State to entitle approximately 1,300 acres of land in East Kapolei, which would then be sold to private developers. As a result, the Housing and Community Development Corporation of Hawaii (HCDCH) initiated a master planning effort for 1,300 acres including the UHWO Kapolei Makai site for mostly residential uses. It was intended that the proceeds of the land sales would be used to fund the construction of a UHWO campus site mauka of the H-1 Freeway. On March 5, 1999, HCDCH filed a Petition for Land Use District Boundary Amendment to reclassify approximately 1,300 acres, including the subject 500 acres, from Agricultural use to Urban use for the master-planned community known as East Kapolei. By Findings of Fact, Conclusions of Law and Decision and Order filed on September 8, 1999, the State Land Use Commission granted this petition. Subsequently, however, Hawai'i's economy suffered a serious downturn, and there was little or no market demand for new residential development. During this period, it was determined that the construction of a new campus mauka of the H-1 Freeway would be more expensive than the current site (makai of Farrington Highway).

During the fall of 2001, the UH Administration, the Estate of James Campbell, and the Governor explored the possibility of relocating the UH West O'ahu campus from the 991acre mauka site to the City of Kapolei.

The University of Hawai'i conducted the *University of Hawai'i West O'ahu Campus Site Selection Study* (2002) (Site Selection Study) to assist the Administration and Board of Regents in the selection of a permanent site for the UH West O'ahu campus. Sites considered for a permanent campus included: Based upon the site evaluation criteria, three sites were initially listed: the City of Kapolei, Leeward Community College, and the Kapolei Sports Complex, the City of Kapolei and while the Kapolei Makai (the proposed site) (see Figure 6.1) was

<u>added based on community input.</u> A detailed evaluation of these sites included: 1) the compilation of information on the characteristics of each site studied; 2) an evaluation of the physical development potential of each site; 3) order-of-magnitude cost estimates and scheduling; and 4) recommendations on a selected site for the UH West O'ahu campus. <u>Six community meetings were held which indicated strong support for the 500-acre Kapolei</u> <u>Makai site.</u>

Conceptual site plans were created to provide site-specific analyses of development feasibility, costs, and time schedules for development on each site. These plans were based on the assumption that the University of Hawai'i would develop a campus that could accommodate an initial student population of 2,750 and an ultimate student population of 7,600. Because the educational program for the new campus was in its initial phases of preparation and no program information was available to the planning team, the site plans were developed based on very general assumptions developed through discussions with the University, the program planning consultant, and the project team.

A major portion of the Site Selection Study was dedicated to implementing a community outreach/presentation process to brief the community on the status of the project and to solicit input for the community's concerns and preferences related to each site. The community review process included presentations to the Makakilo/Kapolei/Honokai Hale Neighborhood Board and its Planning and Zoning Committee, the Villages of Kapolei Community Association, and the 'Ewa Neighborhood Board and its Planning and Zoning Committee. This process also included meetings with representatives from City and State agencies to obtain information and discuss issues related to site selection.

Leeward Community College. The Leeward Community College site was eliminated from consideration as an alternative site because of concerns about poor site access for vehicles exiting the H-1 Freeway. The limited land area for expansion and lack of a separate campus identity between the Leeward Community College and UH West O'ahu were other concerns that eliminated this site from further consideration.

Kapolei Sports Complex. The Kapolei Sports Complex site received strong community opposition due to concerns regarding the possible increase in traffic along Kapolei Parkway. Other community concerns included the potential for traffic accidents involving students crossing the road from Kapolei Middle and High Schools, the limited size of this site for a campus and the limited expansion potential, and the proposed height of buildings on this site. In addition, access, flooding, and regional drainage issues, U.S. Navy easements, and State Land Use Commission conditions made this site the least feasible to develop. After the community review process, this site was eliminated from the study and is now owned by the DHHL and proposed as a major regional shopping center.

City of Kapolei Site. The 16.2-acre City of Kapolei site was eliminated because of its limited expansion potential (beyond a 7,600 student population). Expansion of the campus would require acquisition of additional lands from the City and The Estate of James Campbell.



Figure 6.1 Alternative Sites Location Map University of Hawai'i West O'ahu



Vertical expansion would also be difficult due to the five to six-story building height limits. Since this site would require a higher density campus, special considerations for vertical circulation (reliance on elevators) would need to be addressed. This site lacked community support, although The Estate of James Campbell was the main supporter.

Kapolei Makai Site (proposed site). The Kapolei Makai site was the former UH West O'ahu campus site prior to the location of the campus mauka of the H-1 Freeway. Based on its size, the Kapolei Makai site offered opportunities for campus expansion beyond the targeted 7,600 student population and greater flexibility in the type of academic programs offered at the campus. The site offered potential ease of access with its proximity to the City of Kapolei and residential communities in the Kapolei and 'Ewa region, as well as the proposed North-South Road along the eastern boundary. Although the site had a number of environmental and physical considerations (i.e., realignment of Kalo'i Gulch, habitat conservation for the endangered plant species *Abutilon menziesii*, and easement requirements), there was ample land to accommodate a campus with a 7,600-student population and other university-related land uses.

Preferred Site. From a community support and site development perspective, the Kapolei Makai site was the preferred site. It was unanimously supported by both the 'Ewa and Kapolei communities, and offered greater expansion potential for the campus to grow beyond its ultimate targeted 7,600-student population. The Kapolei Makai site offered greater flexibility to accommodate additional academic programs with larger land area requirements.

Since the petition area was not marketable, and a site selection study determined that the current site was best suited for a new UHWO campus, UHWO was allocated 500 acres and most of the remainder of the 1,300 acres which were "urbanized" was transferred to DHHL.

On September 13, 2002, the UH Board of Regents approved the 500-acre Kapolei Makai site as the permanent location for the new UH West O'ahu campus and requested the transfer of title to the 500 acres from DLNR. The Board of Land and Natural Resources approved the transfer of title of the 500- acre site to the UH in fee simple.

The University of Hawai'i still retains the 991-acre site mauka of the H-1 Freeway, but has no current plans for it other than being generally viewed as developable in the future as an expansion area for the UHWO campus.

6.3 "NO-ACTION" ALTERNATIVE

The "no-action" alternative would not attain the objectives of the proposed action. As discussed in Section 2.2, Need for the Project, with the expected population increase in the 'Ewa region, the proposed UH West O'ahu campus is expected to target students in the surrounding regions to alleviate the burden on the UH Mānoa campus. The relative need for postsecondary education based on projected changes in population from the year 2000–2020, are shown in Tables 2.1 and 2.2. The "no-action" alternative would not be consistent

with stated governmental policies of directing future growth toward the 'Ewa Plain (as in the *Ewa Development Plan*). It would also be inconsistent with the *University of Hawai'i - West O'ahu Strategic Plan 1997-2007*, which seeks to establish new higher education and employment opportunities in the region. Under this alternative, the property would remain as agricultural and vacant land, surrounded by planned urban developments in the area. After weighing the socio-economic benefits of the campus against the loss of actively cultivated land on portions of the property, the "no-action" alternative was rejected.

Under one of the conditions and terms of agreement set forth between James Campbell Company, LLC, and the State of Hawai'i, in regards to the acquisition of land for the proposed UH West O'ahu campus, the State shall have secured funding, entered into contracts and commenced the construction of all of the administration, classroom and related support facilities necessary to accommodate an enrollment of 2,750 students, provided that such commencement date shall be extended to December 31, 2011, if the State Legislature appropriates funds for the planning and design of the UH West O'ahu campus on the mauka property by December 31, 2006.

The State has appropriated funds for the planning and design of the property for the UH West O'ahu campus, however, if construction of the UH West O'ahu campus is not commenced by December 31, 2011, then the State, at its election shall either:

- 1. <u>Convey to Campbell Estate a single, agreed upon parcel of land that is located makai</u> of H-1, zoned for agriculture with urban potential, and that has value equal to the <u>1994 value of the 200 acres of land conveyed to the State in 1994; or</u>
- 2. <u>Pay the Campbell Estate the current fair market value of the 200 acres of land conveyed to the State in 1994.</u>

6.4 ALTERNATIVES RELATED TO DIFFERENT DESIGNS OR DETAILS OF THE PROPOSED ACTIONS WHICH WOULD PRESENT DIFFERENT ENVIRONMENTAL IMPACTS

The property could be developed to encompass a range of viable land uses, as it is entirely owned by the State of Hawai'i and located adjacent to a major new roadway (North-South Road), a proposed interchange on the H-1 Freeway, and proposed DHHL developments. Viable land uses include primarily residential and commercial uses, presenting environmental impacts different from the proposed UH West O'ahu project and dependent on the program, timing, and design of the development. Environmental impacts from primarily residential uses were described in the HCDCH East Kapolei EIS and the DHHL East Kapolei Development Parcel B Draft Environmental Assessment. A model of a similar type of development would be the Villages of Kapolei, which has a variety of residential and golf course land uses located on lands with similar characteristics (flat terrain).

One alternative to the proposed action that would represent a different design or detail is an alternative land use emphasis of the non-campus lands from residential to commercial. The original land use plan for the non-campus land was primarily focused towards commercial development (with very little housing). While commercial development would provide needed employment opportunities in the 'Ewa region (and reduce traffic to/from Downtown Honolulu) and provide a source of long-term income to fund operations and maintenance and pay down construction debt, UH decided to seek a solution where a private residential developer would be able to provide a significant amount of immediate cash to help offset the cost of constructing the first phase of the campus. As a result, the plan now reflects significantly more residential than commercial. In that regard, there should be significantly less impact on the housing inventory in the 'Ewa region, than if UHWO implemented its original land use plan. UHWO is particularly motivated that the project provides workforce housing for faculty and staff as it is difficult to attract the best candidates without it.

Environmental impacts of the proposed UH West O'ahu development would differ depending on the density and design capacity. For example, a campus and community with a higher density would reduce the buildable area and quantity of surface runoff. However, buildings would be larger and taller, and parking structures would replace the currently proposed ongrade parking lots. Spreading the campus over a larger land area would reduce the density, but consume more land, increase the quantity of runoff, and require a much larger investment in grading and infrastructure. The quantities of potable water, wastewater, and traffic generated by the project would not be significantly different since the number of students and faculty served would not change; however, the cost of infrastructure would be more for a lower density development.

The land uses selected for the current plan were organized to capitalize on synergistic relationships between the campus and community, creating an integrated community where people can live, work, learn, play, and shop.

6.5 ACTIONS OF A SIGNIFICANTLY DIFFERENT NATURE WHICH WOULD PROVIDE SIMILAR BENEFITS WITH DIFFERENT ENVIRONMENTAL IMPACTS

There are no known actions significantly different to those discussed that would provide the same educational opportunities as the proposed UH West O'ahu project. Enhancement of Leeward Community College and the existing UH West O'ahu facilities could provide improved educational services and access to such services; however, spatial constraints, traffic and access issues, and the distance between these facilities and major population growth areas limit the potential benefits of this alternative.

There are no other known land uses that would provide similar university-level educational benefits without creating environmental impacts. Agricultural use of the property would result in continued soil erosion and use of fertilizers and chemicals, and would provide limited economic benefits.

Other institutional land uses, such as major medical facilities, are already in place in the 'Ewa region, and most of the major hospitals are looking to develop smaller satellite clinics within rapidly growing areas. A prison or another major institutional public facility would likely be objectionable to residents in the surrounding community.

6.6 THE ALTERNATIVE OF POSTPONING ACTION PENDING FURTHER STUDY

Further study of any proposed development on the property would not be consistent with the *Ewa Development Plan* or the State's objective to provide new educational opportunities within an area planned for major population growth. Development of the property and the 'Ewa Plain as O'ahu's Second City has been studied and planned for many years, and postponing the proposed action would not achieve the overall project objectives of providing education and employment opportunities.

Under conditions and terms of agreement set forth between the James Campbell Company, LLC and the State of Hawai'i, in regards to the acquisition of land for the proposed UH West O'ahu campus: 1) UH must designate at least 200 acres of the mauka property for use solely for University Purposes by the earlier to occur of December 31, 2005, or the date of any conveyance of any portion of the mauka property to any third party; and 2) the State shall have secured funding, entered into contracts and commenced the construction of all of the administration, classroom and related support facilities necessary to accommodate an enrollment of 2,750 students, provided that such commencement date shall be extended to December 31, 2011, if the State Legislature appropriates funds for the planning and design of the UH West O'ahu campus on the mauka property by December 31, 2006.

The State has appropriated funds for the planning and design of the property for the UH West O'ahu campus, however, if construction of the UH West O'ahu campus is not commenced by December 31, 2011, then the State, at its election shall either:

- 2. <u>Convey to Campbell Estate a single, agreed upon parcel of land that is located makai</u> of H-1, zoned for agriculture with urban potential, and that has value equal to the 1994 value of the 200 acres of land conveyed to the State in 1994; or
- 2. <u>Pay the Campbell Estate the current fair market value of the 200 acres of land conveyed to the State in 1994.</u>

(This page intentionally left blank.)

7.0 Contextual Issues

7.0 CONTEXTUAL ISSUES

7.1 RELATIONSHIP BETWEEN LOCAL SHORT-TERM USES OF THE ENVIRONMENT AND THE MAINTENANCE OF LONG-TERM PRODUCTIVITY

As discussed in previous sections of this EIS, the project area consists entirely of former sugar land recently cultivated with various "truck" crops. Other than development of the property proposed in this EIS, the most logical use of the property would be residential and/or residential/commercial mixed uses (and supportive urban uses) because of the high demand for residential product and the financial feasibility of such a use. There is also a need to provide jobs for these residential developments.

Allowing the property to remain in its current state (vacant with portions leased for diversified agriculture) would not be consistent with its Urban State Land Use District designation. Continuing agricultural use would also be inconsistent with the property's *Ewa Development Plan* designation for Low and Medium Density Residential and High Density Residential. Long-term productivity would be minimal if the property continued as agricultural land, and the Kapolei area, West O'ahu, and the City and County of Honolulu would not experience the overall positive economic benefits discussed throughout this EIS.

Through careful site planning, the property will complement the urban character of the growing 'Ewa region. At build-out, the UH West O'ahu has the potential for long-term productive use of the property. The development will generate significant social and economic benefits to the community through employment, education, and home ownership opportunities, and the provision of infrastructure and other community services and facilities. The UH West O'ahu campus will likely attract students who do not want to commute to the UH Mānoa campus, cannot afford local private institutions, or do not want to leave Hawai'i

Long-term impacts from continued agricultural use of the property are negative when compared to potential positive impacts of the UH West O'ahu development. Without the proposed development, the existing UH West O'ahu campus would remain at its current location and would continue to share land, facilities, and services with Leeward Community College. With UH West O'ahu student enrollment expected to increase, existing facilities would be further strained, possibly jeopardizing the university's accreditation. Additionally, without the proposed development, short-term construction-related jobs would not be created, and industries supporting construction or servicing construction workers (restaurant and retail industries) would not benefit economically.

Studies performed in preparation of this EIS indicate that the project will not adversely affect the existing environment. The physical attributes of the property are appropriate for the land uses proposed, and grading will not be significant due to the relatively flat terrain. Short-term construction impacts can be mitigated by the implementation of best management practices and other techniques, and long-term land uses will generate substantial economic benefits and create a sustainable community.

The project is expected to have significant beneficial impacts on the City and State's fiscal and economic resources. Short-term uses of environmental resources to develop and construct the project would promote long-term productivity through employment and increased income and excise tax revenues. The project will not foreclose future options, narrow the range of beneficial uses, or pose long-term risks to health or safety.

7.2 CUMULATIVE AND SECONDARY IMPACTS

In general, many new residents, employees, and visitors (i.e., students, shoppers, etc.) will be introduced to the East Kapolei area by proposed projects including the UH West O'ahu. The majority of these people will be O'ahu residents relocating to the East Kapolei area. The project's regional impact may be significant, and at the island-scale, the impacts may likewise be significant, at least to commuters to the Primary Urban Center from Central and West O'ahu. Due to growth of the existing 'Ewa population and in-migration into 'Ewa and Central O'ahu, a new university campus is needed for Leeward and Central O'ahu residents. The proposed UH West O'ahu will provide educational, employment, residential, recreational, and commercial opportunities and will enhance the establishment of Kapolei as the Second City.

Several other developments are planned near the UH West O'ahu property, including construction of North-South Road and H-1 Freeway interchange, widening of Farrington Highway, extension of Kapolei Parkway, development of the DHHL East Kapolei properties, and development of a new community, Ho'opili, proposed by D.R. Horton – Schuler Division. The proposed DHHL East Kapolei projects include a residential subdivision with commercial and recreational uses (known as Parcel B) south of and adjacent to the UH West O'ahu property, DOE school sites, and the proposed Kroc Center. D.R. Horton – Schuler Division recently acquired the area between UH West O'ahu and Fort Weaver Road (to the west and east) and the H-1 Freeway and 'Ewa Villages (to the north and south). D.R. Horton – Schuler Division is in the process of master planning this property (consisting of approximately 1,600 acres). The property is slated for a mixed residential commercial development and while no definitive plans have been established, the development is anticipated to include DOE school sites and a district park site.

This chapter identifies secondary and cumulative impacts that may result from the proposed development of UH West O'ahu and East Kapolei. Mitigation measures are also identified and discussed in this chapter.

7.2.1 Impacts on the Physical Environment

Climate, Topography, and Soils. Planned developments in the project area are not expected to adversely impact regional climate, topography, and soils. However, within their respective project sites, construction will impact topographic features and soils, and new buildings may affect the respective micro-climate of each building site (by retaining and/or reflecting solar energy and heat, creating shade where none presently exists, or creating "wind tunnels"). Grading operations will comply with DOH regulations and are not expected to adversely impact air and water quality. Increased impervious surfaces (i.e., roofs, roadway surfaces, and sidewalks) will increase the potential for runoff, although existing and proposed detention basins throughout East Kapolei will effectively manage drainage. Additionally, landscaping introduced by the developments will reduce the potential for soil erosion, compared to current conditions on vacant and actively cultivated lands.

Drainage and Groundwater. Drainage from Kalo'i Gulch above Farrington Highway will be realigned to the North-South Road utility corridor (see Figure 2.5). Regional infrastructure will be developed to control drainage in the project area. The planned developments will introduce impervious surfaces through the construction of buildings, roadways, parking areas, and walkways; however, no significant impacts on existing drainage conditions south of the project area are anticipated, as UH West O'ahu facilities and regional (East Kapolei) facilities will be developed and sized to control runoff. All drainage facilities will comply with the City's Storm Drainage Standards.

No injection wells or cesspools are proposed by any of the developments. Any wastewater generated will be collected and treated at the Honouliuli WWTP. A non-potable water system will be developed for East Kapolei, to facilitate the recharge of caprock water and to reduce the demand for potable water from the BWS system.

Natural Hazards. It is unlikely that any of the proposed structures and uses associated with the planned developments will affect the potential occurrence of earthquakes, hurricanes, or tsunami. The proposed projects will likely be designed to address potential damage from natural hazards.

Flora and Fauna. Although agricultural activities in the past significantly altered natural landforms and removed plant species, some native species were established in the project area after agricultural operations ceased. According to a botanical survey, native species on the UH West O'ahu property are common to abundant throughout scrub vegetation elsewhere. An endangered plant species, *Abutilon menziesii* (ko'oloa'ula), is found in certain areas of Kapolei and will be relocated as part of an approved HCP (see Figure 3.5 and Appendix C). Within each of the proposed project sites, existing plant species will be removed. The long-term impact on the ko'oloa'ula population in Kapolei will be beneficial, as proposed mitigation measures in the HCP will ensure the future propagation of new plants.

The cumulative impact of the planned developments in East Kapolei is the reduction of habitat. Past and current agricultural activities in the project area have altered natural habitats, and existing fauna are primarily introduced species. During construction, these species are likely to vacate the site for suitable habitats nearby. After construction, these species may return, depending on their preference of habitat and the landscaping provided.

7.2.2 Impacts on the Human Environment

Archaeological/Historic and Cultural Resources. Although very little prehistory is known about the project area, there is no indication of human occupation or any other utilization of the land. No remains of any prehistoric activity are believed to exist in the area, and no sites listed on the Hawai'i or National Register of Historic Places are found on the property. According to the archaeological reconnaissance and assessment conducted for the HCDCH East Kapolei EIS, the presence of any significant archaeological sites on the surface or subsurface is unlikely due to the disruption caused by continuous sugarcane cultivation for nearly 70 years.

Agriculture. Lands in the project area were historically used for sugarcane cultivation and most recently for diversified agriculture. Much of the project area is actively cultivated through revocable leases to Aloun Farms, Inc., A.M. Enterprise, Inc., Sugarland Farms, Rocker G. Livestock, and Garst Seed (see Figure 1.4). With the proposed developments, land will be withdrawn from production, resulting in some loss in revenues, jobs, and payroll, unless replacement farm leases are established elsewhere. Ample agricultural land will be available to accommodate the projected growth of diversified agriculture on O'ahu and statewide, and the limiting factor to the growth of diversified agriculture is not the land supply, but the size of the market for those crops that can be grown profitably in Hawai'i. Many acres of former sugarcane and pineapple land lie fallow statewide; therefore, the loss of agricultural land on the site will not preclude the potential for crops suitable for the production of ethanol.

Noise. Construction activities in East Kapolei will increase existing on-site noise levels. However, most of the project area is open space that is currently vacant or leased for agricultural activities. Noise impacts will be relatively short-term during daytime hours, and construction activities will comply with State DOH noise regulations.

Over the long term, as the project area develops and the 'Ewa population grows, noise levels will increase above current conditions. New roadways and improvements to existing roadways are proposed, and vehicles traveling to, from, through, and within the East Kapolei area will be a significant contributor of noise. Educational, residential, recreational, and commercial facilities will also increase noise, as could aircraft noise from Kalaeloa. However, noise-sensitive uses occurring in buildings are likely to be insulated and equipped with air conditioners, mitigating noise from traffic, wind, wildlife, and aircraft.
UNIVERSITY OF HAWAI'I WEST O'AHU Final Draft Environmental Impact Statement

Air Quality. Emissions from cars and equipment used to generate electricity can affect air quality. The planned developments in East Kapolei will attract traffic and use electricity, but are not expected to significantly affect regional air quality, as new technologies, increasingly stringent air pollution control regulations and improving automobile technology, and alternative forms of transportation (i.e., bicycling, walking, public transportation) may offset potential increases in air pollution. Predominant tradewinds during the afternoon provide dispersal of emissions.

Man-made Hazards. No industrial facilities handling explosive or fire-prone materials (i.e., liquid propane, gasoline, or other storage tanks) exist in the project area. However, there are two hazardous sites in and near the East Kapolei area – the former 'Ewa Sugar Mill, which is surrounded by the 'Ewa Villages residential and golf project and has a current status of No Further Remedial Action Planned, and the 'Ewa Sugar/O'ahu Sugar Company Pesticide Mixing and Loading Site, which is located to the east of North-South Road. The land uses proposed by the planned developments in East Kapolei are not likely to expose people or buildings to explosive or flammable fuels or chemical containers.

Visual Resources. The planned developments in East Kapolei will change the open space character of the area and introduce urban structures including a university campus, various types of residences, commercial centers, schools, and transportation facilities. These developments will include landscaping to reduce the bulk of buildings and enhance the visual character of the sites. Due to the flat topography of the project area, there are no ocean views offered. However, new views will be created from currently inaccessible project sites and future roadways. The campus buildings will be oriented to capitalize on views of the Wai'anae and Ko'olau mountain ranges.

7.2.3 Impacts on the Socio-economic Environment

Population, Housing, and Community Character. The planned developments will help to transform the East Kapolei area into a part of the Second City by providing educational, employment, housing, recreational, and improved transportation opportunities. No homes or residents will be displaced by the projects, which will be developed on land that is currently undeveloped.

Land uses proposed in the HCDCH East Kapolei Master Plan area included single- and multifamily residential, commercial, and public facilities (schools, parks, and a sports complex). In addition to providing a campus for 7,600 students, the proposed UH West O'ahu project will include single- and multi-family residential units (including student housing and work force/affordable housing), an elementary school site <u>(and a possible "Lab School"</u>, and parks.

The UH West O'ahu development is not expected to impact the Gentry Investment Properties' Ewa Makai development, which proposed the development of 7,000 homes (of which 5,383 homes had been completed as of January 31, 2003) on 1,000 acres of land.

UNIVERSITY OF HAWAI'I WEST O'AHU Final Draft Environmental Impact Statement

Employment, Income, and Government Revenues. Portions of the UH West O'ahu property and the proposed DHHL East Kapolei development are currently under revocable agricultural leases. These lands will be withdrawn from agricultural production, resulting in some loss of revenues, jobs, and payroll although there is plentiful vacant agricultural land further north of East Kapolei and in Kunia, Central O'ahu, and the North Shore. However, construction of the planned developments will provide numerous employment opportunities. Long-term operation of UH West O'ahu will also provide faculty, staff landscaping, service, and maintenance positions. Local businesses will profit from the developments in East Kapolei, and money spent by workers and businesses will flow through the West O'ahu regional economy especially in the City of Kapolei. The City and State will benefit from the developments through the generation of income and general excise tax revenues.

7.2.4 Impacts on Infrastructure Facilities

For many years, the City and State have planned major new development and investment in public infrastructure within the 'Ewa region.

Water. The overall demand for potable water will increase with population growth in the East Kapolei area. New demand for potable water will be addressed by the installation of appropriate water source, storage, and distribution improvements. A non-potable water system will be developed for East Kapolei by the BWS.

Wastewater. Wastewater flows will increase with the proposed developments and the expected population growth of the 'Ewa region. Regional wastewater facilities will be installed to accommodate these developments. Expansion of the municipal water reclamation facility will ensure water reuse and less the impact on potable water sources.

Drainage. Planned and future developments will increase the amount of impervious surfaces by construction of buildings, roadways, parking areas, and walkways. Detention basins are planned throughout the East Kapolei area to control storm water runoff.

Solid Waste. As O'ahu's population increases, the capacity of the Waimānalo Gulch Landfill will decrease. Solid waste will be generated during construction and operation of the planned developments in East Kapolei. The UH West O'ahu campus has established sustainability guidelines and will incorporate reduce, reuse, and recycle materials to minimize generation of solid waste and divert it from landfills. A third boiler at the H-POWER plant is being discussed and would save space in the landfill and supply electricity to 20 percent more homes each year.

Roadways and Traffic. Vehicular traffic in the project area would increase with the proposed developments. The UH West O'ahu campus would generate traffic in the area, but would also divert traffic to/from the existing UH West O'ahu campus and the UH Mānoa campus. Planned transportation improvements including construction of North-South Road and H-1 Freeway interchange, extension of Kapolei Parkway, and widening of Farrington Highway

will help to mitigate traffic impacts from the proposed developments. Additionally, proposed mixed-use communities and the proposed rapid transit system will encourage walking, bicycling, and the use of public transportation, relieving vehicular traffic to and from Honolulu. As Kapolei becomes O'ahu's Second City, more employment centers and schools will be provided and vehicular traffic toward workplaces and schools in Honolulu will improve.

7.2.5 Impacts on Public Facilities and Services

Electrical and Communication Facilities. Steam units, diesel units, and gas turbines used to generate electricity also generate emissions and affect air quality. Any increase in demand for electricity not generated by renewable resources will have an indirect impact on air quality. Two new electrical substations are proposed within the UH West O'ahu property. The UH West O'ahu has established sustainability guidelines that strive to achieve targets set by the LEED NCv2.2 rating system (see Section 2.6). Other developments increasingly incorporate energy-saving features in their design.

The planned developments in East Kapolei will also increase the demand for communications services (i.e., cable television and telephone service), but are unlikely to have an adverse effect on existing communications systems.

Recreational Facilities. The 'Ewa region is the fastest growing region in O'ahu, and the proposed developments will help to meet the demand for additional recreational facilities. Parks will be provided to meet the requirements of the City's Park Dedication Ordinance. New recreational facilities (i.e., tennis courts, basketball/volleyball courts, and jogging paths) within the campus are also planned for use primarily by students, faculty, and staff. The DHHL residential subdivision adjacent to and south of the UH West O'ahu property proposes 4.5 acres for park use and 10.7 acres for a community center to be developed and operated by the Salvation Army. The proposed center could include a swimming pool, performing arts center, child care center, gymnasium, education center, game and recreation areas, and other multi-use spaces. The facilities would be available to the general public. Additionally, Ho'opili will provide recreational facilities in accordance with the City's Park Dedication Ordinance.

Existing regional park facilities in West O'ahu, Central O'ahu, and Waipi'o may be visited by new residents of the East Kapolei developments; however, other park facilities are planned in 'Ewa, especially in Kalaeloa.

Medical Facilities. Within the East Kapolei project sites, accidents requiring medical attention will be occasional. The demand on existing medical facilities will increase as the population of 'Ewa grows. The proposed UH West O'ahu student center is programmed to include a student health clinic.

Educational Facilities. The UH West O'ahu will be a higher education institution in the 'Ewa

UNIVERSITY OF HAWAI'I WEST O'AHU Final Draft Environmental Impact Statement

region, enabling Leeward and Central O'ahu residents to obtain post-secondary degrees without commuting to the UH Mānoa campus or other higher education institutions in Honolulu. A large number of college-aged students live in Leeward and Central O'ahu, and without the proposed new campus, the existing UH West O'ahu campus will not be able to accommodate prospective students.

The demand for educational facilities will increase as the population of 'Ewa grows. The State DOE has expressed concern about schools in Kapolei exceeding their capacity. Within the project area, the DOE anticipates the need for two elementary schools, one middle school, and one high school. A 12-acre elementary school is proposed at the southern boundary of the UH West O'ahu property (near residential uses within the property and the proposed DHHL residential subdivision). The DHHL is proposing an elementary school and a separate middle school site on its property immediately east of the UH West O'ahu site. The DOE is also contemplating a high school on DHHL land immediately north of the UH West O'ahu site. The DOE has been in discussion with developers for additional school sites in East Kapolei. UHWO is contemplating the possible development of a "Lab School" on the campus lands in order to meet any additional DOE requirements.

Police Protection. Incidents requiring police protection or service will be occasional. As the population of 'Ewa grows, additional officers and staff may be needed. During construction, off-duty police officers could be periodically employed to manage traffic near project sites. Within the UH West O'ahu campus, on-site security systems or personnel will be employed. Home security systems may also be installed by individual residents.

Fire Protection. There will be an occasional and unavoidable need for fire protection or emergency service from the HFD. As the population of 'Ewa grows, additional fire fighters and staff may be needed. However, to prevent fires, all proposed facilities would be designed to meet fire code requirements, and design plans would be coordinated with the HFD. Emergency and fire vehicle access would be provided within each project site.

7.3 IRREVERSIBLE AND IRRETRIEVABLE COMMITMENTS OF RESOURCES

The proposed UH West O'ahu development will result in the irreversible and irretrievable commitment of land, water, energy, and fiscal resources from the State and private sectors. As previously discussed, the University has limited capital funds available and has therefore selected a development partner to construct the campus with private funds in exchange for the development rights to a portion of the approximately 500-acre property.

The UH West O'ahu property was historically used for the cultivation of sugarcane, and much of the property continues to be used for diversified agriculture under revocable leases (see Figure 1.4). The ALISH system classifies the land on the property as Prime Agricultural Land (see Figure 3.4); however, a substantial commitment of water and energy are required to realize the property's agronomic qualities. Furthermore, replacement lands that are more appropriately suited for agricultural purposes (in relation to State and City plans to create a

second city in 'Ewa) currently exist elsewhere on O'ahu and throughout the State.

The UH West O'ahu development will require on-site infrastructure (i.e., transportation, drainage, water, and wastewater facilities), which will be provided by the State and its private development partner. The development will increase the use of existing public infrastructure (i.e., police and fire facilities), but will also provide new residential, educational, employment, retail, cultural, and recreational facilities for area residents.

Labor and materials, which are mostly non-renewable and irretrievable resources, will be required during construction. After construction, operation of the project will require the continued use of water and petroleum-generated electricity. However, the proposed development will introduce urban structures and significantly reduce the irrigated land area compared to existing conditions under which large quantities of water are used for agricultural purposes. Within the UH West O'ahu development, non-potable water will be used for irrigation purposes where allowed under DOH regulations. In addition, the nearly flat project site will be designed to encourage walking and biking within the property and to surrounding areas.

The proposed UH West O'ahu development will require an irreversible and irretrievable commitment of resources, but will provide substantial socio-economic benefits to the community and State. Educational, employment, and residential opportunities will become available through the development of the UH West O'ahu. Infrastructure to support the development will be installed on the property, which currently requires usage of water, fertilizers, and pesticides for agricultural purposes. The proposed use of the property is consistent with future development planned by the City and State, and with its proximity to the City of Kapolei and major transportation facilities planned, the UH West O'ahu campus would provide West O'ahu residents with improved access to higher education facilities and workplaces. Development of the UH West O'ahu campus and community on the proposed site will help transform Kapolei into the Second City, enabling residents to live, work, play, and attend school without commuting to areas outside of the 'Ewa region.

7.4 PROBABLE ADVERSE ENVIRONMENTAL EFFECTS THAT CANNOT BE AVOIDED

Environmental impacts anticipated to result from development of the 500-acre UH West O'ahu property are typical of any large-scale development and have been discussed throughout this EIS. Although past and current uses of the property and adjacent properties (i.e., sugarcane cultivation, grazing, and limited diversified agriculture) have significantly altered the natural landscape, under the proposed project, the existing landscape will be transformed for urban development.

Construction-related Impacts. Potential environmental impacts (i.e., noise, soil erosion, fugitive dust and exhaust emissions, and temporary traffic disruption) will primarily occur during the construction period. Noise levels will temporarily increase during construction;

UNIVERSITY OF HAWAI'I WEST O'AHU Final Draft Environmental Impact Statement

however, all construction activities will occur during daylight hours and will comply with DOH noise regulations. Noise mitigation will be implemented as necessary. There is a potential for soil erosion and fugitive dust during construction; however, all construction activities will comply with State DOH regulations, and after construction, soil erosion will likely be reduced compared to current conditions under which soil is exposed on vacant land and on land actively cultivated.

Infrastructure Impacts. Solid waste and wastewater will be generated on-site, and energy and water will be consumed for operation of the UH West O'ahu. These impacts are typical of large-scale urban developments and will be mitigated through site design and development standards. Major infrastructure improvements proposed include the installation of new potable and non-potable water systems, wastewater collection systems, major drainage improvements, and off-site transportation facility improvements, onelectrical/communication infrastructure, and related administrative, instructional, and support buildings. In addition, regional transportation improvements planned by the State and City (i.e., North-South Road and interchange with H-1 Freeway, improvements to Farrington Highway, and extension of Kapolei Parkway) will help to alleviate traffic in the 'Ewa region.

Visual Impacts. The on-site visual environment will significantly change as agriculturerelated structures and vegetation are replaced by urban structures. However, the project site and area are relatively flat, providing no shoreline views. Views from future internal roadways will be given consideration to the extent possible, and extensive landscaping and thoughtful architectural design will add to the visual character of the area. Views within the campus will capitalize on the Wai'anae and Ko'olau mountain ranges.

Faunal Impacts. The number of some introduced birds (i.e., Common Myna and Red-vented Bulbul) will likely increase, while the number of introduced birds that prefer open grasslands and agricultural fields will likely decrease. The number of Pacific Golden-Plover may also increase as lawns are installed throughout the campus and residential areas. No native, threatened, or endangered species were observed during the survey, and as such, none are expected to be impacted by the proposed project.

Floral Impacts. A botanical survey of the property identified a few individual plants of the endangered species, *Abutilon menziesii* along the southeastern boundary of the property. The survey also suggested that seeds may be present in the soils in the central portion of the property where a single plant once lived. The existing plants and seeds would be removed for the proposed development; however, implementation of the HCP would significantly increase the number of new plants on O'ahu and improve the quality of the existing population in Kapolei.

Agricultural Impacts. The public currently benefits from employment and income generated through short-term revocable agricultural leases on portions of the property. These cultivated lands will be withdrawn from agricultural production for the development of the UH West O'ahu; however, there is an ample supply of land suitable for diversified agriculture on O'ahu

and a relative lack of market demand. No public benefits result from the vacant lands within the property. The proposed development will provide new employment opportunities, as well as educational, recreational, and residential opportunities.

7.4.1 Rationale for Proceeding with the Project Notwithstanding Unavoidable Effects

The UH West O'ahu will provide socio-economic opportunities to residents of Hawai'i. The approximately 500-acre property will include the UH West O'ahu campus, which will accommodate an ultimate population of 7,600 students. The property will also include a mixed-use University Village, commercial areas, single-family residential units, multi-family residential units (including student housing and work force/affordable housing), and a<u>t least one</u> 12-acre elementary school site. The project will generate direct, indirect, and induced jobs during construction and operation, enabling residents to live, work, learn, shop, and play in one area.

7.4.2 Unresolved Issues

In compliance with Section 11-200-17(n), HAR, this chapter describes the unresolved issues associated with the UH West O'ahu development. These unresolved issues primarily deal with future actions and decisions of governmental bodies that cannot be determined at this time. The following sections discuss how such issues will be resolved and what overriding reasons exist for proceeding with the project.

There are three <u>six</u> currently unresolved issues for the proposed project. Each issue is in the process of being resolved by the University of Hawai'i or other agencies with jurisdiction. None of the unresolved issues require acceptance of this EIS prior to permit approval or construction.

Funding. The University is currently entertaining a number of funding strategies for the initial 1,520 student campus. These strategies include a development partnership, land sale and/or legislative funding, or a combination of these options. Funding for campus development beyond the initial phase is currently an unresolved issue. The initial campus for 1,520 students will be developed by the University's development partner. Construction of the campus beyond the initial phase will occur on an as-needed basis through a public-private partnership and/or funding from the State.

Public/Private Partnership. The University and its private development partner are currently working on an agreement for the UH West O'ahu development. The exact acreage allocated to the Private Developer shown in this EIS is approximate and will not be known until the agreement with the private development partner is finalized. In any case, adequate land area will be set aside for a campus than can eventually accommodate up to 7,600 students in the future.

Completion of Archaeological Inventory Survey Report. OHA recommends that the project area undergo an Archaeological Inventory Survey (AIS), as a new study or supplement to the

UNIVERSITY OF HAWAI'I WEST O'AHU Final draft Environmental Impact Statement

previous, in which all aspects of applicable 6E administrative rules and revised statutes are satisfied. The AIS will include a substantial subsurface inventory as much of the surface architecture may have been destroyed during previous commercial agriculture activities. UH West O'ahu has contracted a professional archaeological firm and the firm is currently conducting its field survey.

Dedication of Roads, Drainage Facilities and Their Related Infrastructure. During the public review period, the City and County of Honolulu Department of Facility Maintenance wrote: *"Roads, drainage facilities and their related infrastructure should be owned and maintained by the State of Hawaii for this State facility, similarly to that at the Manoa Campus. The City and County of Honolulu will not accept the dedication or maintenance of this facility." This will be identified as an unresolved issue since the Department of Planning and Permitting is specifying roads within the 500-acre project to provide regional connectivity, and the non-campus lands will include residential and commercial uses that will generate property tax revenue to the City and County of Honolulu, it is felt that the non-campus roads should be allowed to be dedicated to the County.*

Plan Review Use (PRU) Permit. Pursuant to Section 3.1.60-1, LUO, the University will seek a PRU Permit for the campus and student housing lands. As required by DPP, the University will submit a PRU Permit application concurrently with the Zone Change application.

Zone Change. The UH West O'ahu property is currently zoned AG-1, Restricted Agriculture. The Department of Planning and Permitting requires that the "campus lands" be rezoned to a district other than the currently zoned AG-1; and as such, a zone change will be required to be processed concurrently with the PRU. Rezoning approval will also be needed for the portion of the project that will be developed privately, and which is not related to the university.

8.0 Consultation

8.0 CONSULTATION

8.1 **PRE-CONSULTATION EFFORTS**

8.1.1 Individuals/Organizations Consulted During the Site Selection Process

In 1967, the UH Board of Regents initiated efforts to establish a four-year college in Central or West O'ahu. Over the years, the Regents approved varying sites: Wai'awa Ridge (1970), Honouliuli//Ewa (1973) and in 1993, Kapolei Makai (the current site). In 1996, it was decided that the West O'ahu campus site would be moved to 991 acres mauka of the H-1 Freeway in Kapolei. This was based on a strategy by the State to entitle approximately 1,300 acres of land in East Kapolei, which would then be sold to private developers. As a result, the Housing and Community Development Corporation of Hawaii (HCDCH) initiated a master planning effort for 1,300 acres including the UHWO Kapolei Makai site for mostly residential uses. It was intended that the proceeds of the land sales would be used to fund the construction of a UHWO campus site mauka of the H-1 Freeway. In 1998, an EIS Preparation Notice covering the 991-acre mauka site was published in the Environmental On March 5, 1999, HCDCH filed a Petition for Land Use District Boundary Notice. Amendment to reclassify approximately 1,300 acres, including the subject 500 acres, from Agricultural use to Urban use for the master-planned community known as East Kapolei. By Findings of Fact, Conclusions of Law and Decision and Order filed on September 8, 1999, the State Land Use Commission granted this petition. Subsequently, however, Hawai'i's economy suffered a serious downturn, and there was little or no market demand for new residential development. During this period, it was determined that the construction of a new campus mauka of the H-1 Freeway would be more expensive than the current site (makai of Farrington Highway). As a result, a Draft EIS for the 991-acre mauka site was not published.

During the fall of 2001, the UH Administration, the Estate of James Campbell, and the Governor explored the possibility of relocating the UH West O'ahu campus from the 991acre mauka site to the City of Kapolei.

Since the marketability of the petition area was non-existent, and a site selection study determined that the current site was best suited for a new UHWO campus, UHWO was allocated 500 acres and most of the remainder of the 1,300 acres which were "urbanized" was transferred to DHHL.

On September 13, 2002, the UH Board of Regents approved the 500-acre Kapolei Makai site as the permanent location for the new UH West O'ahu campus and requested the transfer of title to the 500 acres from DLNR. The Board of Land and Natural Resources approved the transfer of title of the 500- acre site to the UH in fee simple. As a result, 1998 EIS Preparation Notice was withdrawn and new one submitted in February 2005. The University of Hawai'i still retains the 991-acre site mauka of the H-1 Freeway, but has no current plans for it other than being generally viewed as potentially being developed in the future as an expansion area for the UHWO campus.

The University of Hawai'i and its consultants made presentations to various community groups, neighborhood boards, community associations, and government agencies as described below:

- University of Hawai'i Board of Regents;
- Makakilo/Kapolei/Honokai Hale Neighborhood Board Planning and Zoning Committee (June 6, 2002) – Issues raised during the meeting focused on the positive and negative aspects of each site, including concerns related to impacts from surrounding land uses, building heights, traffic, roadways, parcel size, expansion potential, feasibility of development on other sites, parking on campus, and phasing. Other issues raised included concerns regarding project funding, student housing, and the proposed UH West O'ahu curriculum. The Committee noted that they would like to have a presentation given to the Makakilo/Kapolei/Honokai Hale Neighborhood Board.
- Villages of Kapolei Association (June 26, 2002) Issues raised during the meeting focused on the positive and negative aspects of each site under consideration, including concerns regarding traffic, roadways, pollution impacts, parcel size and development costs. Other issues raised included concerns regarding community input from other areas, the ultimate campus size, and the UH West O'ahu curriculum. Although no vote was taken, the Kapolei Makai (current) site appeared to be the most popular choice.
- 'Ewa Neighborhood Board (July 11, 2002) Issues raised during the meeting focused on the positive and negative aspects of each site under consideration, including concerns regarding traffic and roadways, campus growth and expansion, costs, and building heights. Other issues raised included concerns regarding student housing, the ultimate campus size, and the UH West O'ahu curriculum. The issue was referred to the 'Ewa Neighborhood Board Planning and Zoning Committee.
- 'Ewa Neighborhood Board Planning and Zoning Committee (July 23, 2002); and _______ Issues raised during the meeting focused on the positive and negative aspects of each site under consideration, including concerns regarding traffic, development cost, other sites previously considered, roadways, infrastructure, timing of construction, building heights, and future campus expansion. Other issues raised included concerns regarding the UH West O'ahu curriculum. The Committee voted in support of the Kapolei Makai (current) site.
- Makakilo/Kapolei/Honokai Hale Neighborhood Board (July 31, 2002) <u>- Issues</u> regarding various sites for consideration, including future campus size and expansion potential, timing of construction, traffic, building heights, endangered plants, costs, parking, and funding, were raised during the meeting. Other issues raised included concerns regarding community input from other areas, and student housing. At the

UNIVERSITY OF HAWAI'I WEST O'AHU Final Draft Environmental Impact Statement

time, no vote was taken but Board members gave their opinions about the proposed sites, with the majority favoring the Kapolei Makai (current) site.

- Makakilo/Kapolei Honokai Hale Neighborhood Board (August 28, 2002) The Board decided to take a position on the location of the UH West O'ahu Campus. The Board voted in support of the Kapolei Makai (current) site.
- University of Hawai'i Board of Regents (September 13, 2002) The UH Board of Regents approved the designation of the 500-acre Kapolei Makai (current) site as the location for a permanent UH West O'ahu Campus.
- <u>'Ewa Neighborhood Board (February 12, 2004)</u> Issues discussed during the meeting included the feasibility of various sites for the location of the campus, UH West O'ahu curriculum, and funding initiatives for the campus. The Board voted in support of a resolution strongly endorsing the need to immediately commence building a permanent four-year UH West O'ahu Campus.
- University of Hawai'i Board of Regents (July 16, 2004) The UH Board of Regents approved in principle the 2004 Long Range Development Plan (LRDP).
- <u>Makakilo/Kapolei/Honokai Hale Neighborhood Board (August 25, 2004)</u> –Issues raised during this meeting included concerns regarding the location of the campus, phasing, project funding, development partnerships, student housing, roadways, infrastructure, and future campus expansion.
- <u>Makakilo/Kapolei/Honokai Hale Neighborhood Board (January 1, 2005)</u> –Issues raised during this meeting included concerns regarding roadways, student housing, project funding, the UH West O'ahu curriculum, accreditation, increasing attendance, development partnership, expansion on other UH West O'ahu property, and providing support at the legislature for the campus.
- Wai'anae Neighborhood Board (March 1, 2005) Issues raised during the meeting included concerns regarding project-generated traffic, available residential/work force housing, student housing, and the location of the campus.
- University of Hawai'i Board of Regents (July 20-21, 2006) The UH Board of Regents approved an update of the LRDP for the UH West O'ahu Campus.
- <u>Makakilo/Kapolei/Honokai Hale Neighborhood Board (July 26, 2006)</u> The University of Hawai'i West O'ahu and its consultants gave a presentation regarding the UH West O'ahu's intent to pursue a Plan Review Use and Zone Change Application. Issues raised during the meeting included concerns regarding roadways, parking on campus, development of other UH West O'ahu property, campus size, campus architecture, open space and detention areas, project funding, available student/work force housing, information on public presentations, the proposed DOE school, and the proposed UH West O'ahu curriculum. The Board voted in favor to support the UH West O'ahu Campus.
- 'Ewa Neighborhood Board (August 10, 2006) The University of Hawai'i West O'ahu and its consultants gave a presentation regarding the UH West O'ahu's intent to pursue a Plan Review Use and Zone Change Application. Issues raised during the meeting included concerns regarding roadways, opportunities for community input, and affordable housing.

8.1.2 Individuals/Organizations Consulted During the Pre-consultation Process

In December 2002, pre-consultation letters were sent to the following individuals and organizations. Those who provided comments are identified in **bold** lettering. Comment letters (and appropriate responses) are included in this chapter.

	AGENCY	DATE OF COMMENTS
Stat	e	
1	Department of Business, Economic Development and Tourism	
2	Department of Business, Economic Development and Tourism – Land Use Commission	01-21-03
3	Department of Business, Economic Development and Tourism – Office of Planning	
4	Department of Business, Economic Development and Tourism – Energy, Resources and Technology Division	01-22-03
5	Department of Education	01-15-03
6	Department of Hawaiian Home Lands	01-08-03
7	Department of Health – Environmental Planning Office	01-27-03
8	Department of Health – Office of Environmental Quality Control	01-02-03
9	Department of Human Services – Housing and Community Development Corporation of Hawaii	
10	Department of Land and Natural Resources	01-18-03
11	Department of Land and Natural Resources – Historic Preservation Division	
12	Department of Transportation	02-11-03
13	Department of Transportation – Airports Division	01-17-03
14	State Legislature – Representative Mark Moses	
15	State Legislature – Senator Brian Kanno	
Fed	eral	
16	Department of the Army – Army Engineer District	
17	Department of Homeland Security – U.S. Coast Guard	
18	Department of the Interior – Fish and Wildlife Service	
19	Department of the Navy	
City		
20	Board of Water Supply	01-10-03
21	Department of Community Services	
22	Department of Design and Construction	03-06-03
23	Department of Environmental Services	
24	Department of Facility Maintenance	
25	Department of Parks and Recreation	01-13-03
26	Department of Planning and Permitting	
27	Department of Transportation Services	01-14-03

Table 8.1 – Pre-consultation Comment Letters

UNIVERSITY OF HAWAI'I WEST O'AHU

FINAL DRAFT ENVIRONMENTAL IMPACT STATEMENT

	AGENCY	DATE OF COMMENTS
28	'Ewa Neighborhood Board, No. 23	
29	Fire Department	01-10-03
30	Makakilo/Kapolei Neighborhood Board, No. 34	
31	Police Department	01-02-03
Oth	er Organizations	
32	Haseko Homes, Inc.	
33	Hawaiian Electric Company, Inc.	02-11-03
34	Palehua Community Association	02-06-03
35	Schuler Homes	
36	The Estate of James Campbell	03-06-03
37	The Gentry Companies	01-03-03
38	Villages of Kapolei Association	

(This page intentionally left blank.)

January 27, 2005 Ms. Genevieve Salmonson, Director State of Hawaii Office of Environmental Quality Control	255 South Beretania Street, Suite 702 Honolulu, Hawaii 96813	SUBJECT: UNIVERSITY OF HAWAII – WEST OAHU DRAFT ENVIRONMENTAL ASSESSMENT/ENVIRONMENTAL IMPACT STATEMENT PREPARATION NOTICE (EISPN) Dear Ms. Salmonson:	Thank you for your letter dated January 2, 2003. We have reviewed your letter and offer the following responses:	 The EISPN will describe area roads and the Draft EIS will include a Traffic Impact Assessment Report. 	2. The current plans for the UH West Oahu include one elementary school site and will meet the park dedication requirements.	3. The Draft EIS will assess the impacts of any student or faculty housing.	 The planning for the UH West Oahu Campus has and will continue consultations with area landowners and residents, though public meetings and presentations at area Neighborhood Board meetings. 	Again, thank you for your comments and your participation in the EIS process. If you have any questions, please do not hesitate to contact me. Sincerely.	PBR HAWAII	Vincent Shigekuni O.JOB1311322.05Elwironmental Assessment-EISPNPre-consultation Response LinsbL-01 OEQC.doc
HAW ALT LAND FLANSING LAND FLANSING LAND FLANSING LAND FLANSING LAND FLANSING LAND FLANSING LAND FLANSING LAND FLANSING	THOMAS S. WITTEN, ASLA	R. STAN DUNCAN. ASI.A Executive Vice-Pressideut Russell, Y. J. Chung, ASI.A Executive Vice-President Vincent Shiegkun PRANPAL	JAMES LEONARD, AICP PRINCIPAL HILO OFFICE	GRANT MURAKAMI, A[CP Senior Associate	TOM SCHNELL, AICP ASSOCIATE RAYNOND T. HIGA, ASLA ASSOCIATE	Kevin Nishikawa, ASLA Associate	Travenueri Chastros	LIANUJAAN OFFILE LIANUJAAN STREET ADD TOWER STREET ADD TOWER STREEGO HEAAOUJUL HAWAT 96813-348 TEL (2003) 5531-6531 FAAC (2003) 573-1402 E-MALL: 9ysaftrin@phthawill.com	III.10 OPPER 101 AUGOS CEPTER SERET 2020 AUGOS CEPTER SERET 2020 AUGOS CONTENTES 2020 AUGOS AUGOS 2023 2021 AUGOS AUGOS AUGOS 2021 AUGOS 2021 AUGOS	E-MAL: phrilo@juva.net WALIJKU OFFICS 2123 K.Cont Brace That: (2023 22.2375 That: (2023 22.2375 That: (2023 22.2375 BrAna: phranai@java.net
A CARACTERISTIC OFFICE OF ENVIRONMENT OUTLINE OFFICE OFFI	January 2, 2003	Mr. Vincent Shigekuni PBR Hawaii 1001 Bishop Street Pauahi Tower, Suite 650 Honolulu, HI 96813	<pre>Subject: University of Hawaii - West Oahu Draft Environment Assessment/Environmental Impact Statement Notice of Preparation</pre>	Dear Mr. Shigekuni:	We have reviewed the information provided for the proposed University of Hawaii - West Oahu campus. We have the following suggestions:	 Address the traffic impacts of the area. Address the impacts with the elimination of the 2 	elementary schools and 9 acres parks on the surrounding communities. 3. If the project will provide student or facility housing, please address the impacts. 4. Consult with the surrounding instances and area	Tesidents. Tesidents. We have no other comments to offer at this time, but will reserve further comments when the documents are submitted.	Should you have any questions, please feel free to call our office at 586-4185. Sincerely,	Var Lut. Génevieve Salmonson Director

LINDA LINGLE GOVERNOR

ШS
MPAN
Ö ≿
BENTE



Provide Stream waters and JAN - G

January 3, 2003

Mr. Vince Shigekuni, Principal Pacific Tower, Suite 650 Honolulu, HI 96813 1001 Bishop Street PBR Hawaii

Dear Mr. Shigekuni,

Thank you for allowing us to provide comments on the proposed UH West Oahu campus and its potential impacts on any of our existing or proposed projects.

employment and educational center in the Ewa region, thereby supporting our vision of West Oahu as the "greatest place to live, work and play." We are therefore in favor of this project, particularly in its current planned location makai of the H-1 freeway. From our standpoint, the proposed UH West Oahu campus will serve as a major

When preparing the Draft EA/EIS, we request that you address two issues: transportation and drainage. With respect to transportation, we support expeditious development of the planned North-South Road and Interchange, as well as completion of Kapolei Parkway, so that homes in the nearby communities of Ewa/Ewa Beach will have easy access to the new campus. We would also like to see alternative modes of transportation (e.g., buses, bikeways, etc.) addressed in the DEA/DEIS in order to alleviate the anticipated increase in traffic that will be generated by the proposed West Oahu campus. With regard to drainage, it appears that the campus site is located within the Kaloi Ditch drainage shed. Please thoroughly address the issue of surface water run-off and ensure that downstream communities will be adequately protected from potential flooding and that any degradation of drainage run-off is minimized.

Thank you again for the opportunity to provide our comments.

Sincerely,

GENTRY HOMES, LTD.

Š

Senior Vice President - Planning Fosh Hosoda

560 N. Nimitz Hwy., Honolutu, Hawaii 96817 P.O. Box 295, Honolulu, Hawaii 96809 (808) 599-5558



LAND PLANNING ANDSCAPE ARCHITECTURI ENVIRONAENTAL STUDIES

P. O. Box 295 WM. FRANK BRANDT, FASLA CHAIRMAN

THOMAS S. WITTEN, ASLA PRESIDENT

VINCENT SHIDEKUNI PRINCIPAL

TOM SCHNELL, AICP ASSOCIATE

KEVIN NISHIKAWA, ASLA ASSOCIATE

Vires R. Suge

Howozuzu OPFICE HOWOZUZU (1001 BERNET ASB TOWER, SUITE 660 HOWOZUZU HAW AT 96813-3484 TZ 26803 523-642 FXZ 68803 523-1402 FXZ 68803 523-1402

HIJJA OPFICE DI AUTON DENERT BLIG LAGOON CENTER, STREF HILD, HAWAT 96720-4262 The (2003) 901-4303 FALE (2003) 901-4039 E-MAIL: Phylilo@law.net

WAILUKU OPFICE 2123 KAOIIU STREET WAILIKU, HAWAYI 96793-2204 TEL (808) 242-2878 FACE (808) 242-2902 EMAILE phyrnauf@lava.rfel

January 27, 2005

Mr. Tosh Hosoda, Senior Vice President - Planning The Gentry Companies

Honolulu, Hawaii 96809

DRAFT ENVIRONMENTAL ASSESSMENT/ENVIRONMENTAL UNIVERSITY OF HAWAII – WEST OAHU SUBJECT:

R. STAN DUNCAN, ASLA Executive Vice-Presiden

RUSSELL Y.J. CHUNG, ASLA EXECUTIVE VICE-PRESIDENT

Dear Mr. Hosoda:

JAMES LEONARD, AICP PRINCIPAL HILO OFFICE

Thank you for your letter dated January 3, 2003. We have reviewed your letter and the University of Hawaii greatly appreciates The Gentry Companies' support of the project. As

IMPACT STATEMENT PREPARATION NOTICE (EISPN)

requested, the Draft EIS will include information on transportation and drainage.

Again, thank you for your comments and your participation in the EIS process. If you have any questions, please do not hesitate to contact me.

GRANT MURAKAMI, AICP SEVIOR ASSOCIATE

RAYMOND T. HIGA, ASLA ASSOCIATE

Sincerely,

PBR HAWAII

Vincent Shigekuni Principal O:UOB1311322.05\Environmental Assessment-EISPNVPre-consultation Response Ltrs\BL-02 Gentry.doc

January 27, 2005	Mr. Boisse P. Correa, Police Chief Police Department	City and County of Honoiulu 801 South Beretania Street Honoiulu, Hawaii 96813	Attn: Ms. Carol Sodetani	SUBJECT: UNIVERSITY OF HAWAII – WEST OAHU DRAFT ENVIRONMENTAL ASSESSMENT/ENVIRONMENTAL IMPACT STATEMENT PREPARATION NOTICE (EISPN)	Dear Chief Correa:	Thank you for your letter dated January 2, 2003 (your reference number CS-KP). We have reviewed your letter, and please be assured that the Police Department will remain a	consulted party during the entire EIS process. The Draft EIS should include more information from which the Police Department will be to evaluate the impact of the proposed project. The principles of crime prevention through environmental design was used in	developing the Long Range Development Plan for the campus. Again, thank you for your comments and your participation in the EIS process. If you have any questions, please do not hesitate to contact me.	Sincerely,	PBK HAWAII	Vurezto K. Hugh- Vincent Shigekuni Principal	OAOB13N1322.05KEnvironmental Assessment-EISPARPre-consultation Response LursbL-03 HPD.doc
PBR BAWATH	LAND PLANDARMO LANDSCAPE AKCHITECTURE BAVENORMANAN LANDARMONIN	WM. FRANK BRANDT, FASLA Charraan Thomas S. Witten, ASLA	R. STAN DUNCAN, ASLA Executive Vice-President	RUSSELL Y.J. CHUNG, ASLA Executive Vice: Pressdent VINCENT SHIGEKUN PRINCIPAL	James Leonard, Aicp Principal Hilo Office	GRANT MURAKAMI, AICP SENIOR ASSOCIATE	TOM SCHNELL, AICP Associate Raymond T. Higa, ASLA	ASSOCIATE KEVIN NISHIKAWA, ASLA ASSOCIATE	HOMOLULU OPPICE 1001 BISIOP STRUET ASB TOWER, SUITE 650 HONOLULI, HAWAT 958 1-3424 Th2: 4003 521-5031	FAX: (808) 523-1402 E-MAIL: syszáttisin@plathawali.com	HILG OFFUES 101 AANNAS STRUET BIALALAGONA CERTUS, STATTS 10 HILG, HANNAT 95720-4262 TD2: (808) 961 -3233 FAC: (808) 961 -3333 FAC: (808) FAC: (808) 761 -3333 FAC: (808) 761 -3333 FAC: (808) 761 -333	WALUKU OFFICE 123 KAOHU OFFICE 2123 KAOHU STREFT VALIKU HAVAN 19579-2264 Thu (609) 242-2563 FACE (808) 242-2903 EAAUL phymatio di Avalant
CITV AND COUNTV OF HONOLULU HONOLULU, HAWAII 95813 - AREA CODE (808) 529 3111 honoLULU, HAWAII 95813 - AREA CODE (808) 529 3111	JEREMY HARAIS MAVOR AAVOR AAVOR AAVOR AAU D. PUCKING AAU D. PUCKIN	OUR REFERENCE CS-KP	January 2, 2003	Mr. Vincent Shigekuni, Principal PBR Hawaii	Pacific Tower, Suite 650 1001 Bishop Street Honolulu. Hawaii 96813-3429	Dear Mr. Shigekuni:	Thank you for the opportunity to review and comment on the University of Hawaii, West Oahu Draft Environmental Assessment/Environmental Impact Statement Notice of Preparation.	Because of the sheer size of the proposed project, we believe that calls for police service to the area will inevitably be impacted. In addition, we know that there will be both vehicular and pedestrian traffic concerns in and around the immediate area. However, we would like to reserve some of our comments until more specific plans for the proposal are developed and studies relative to various impact items are conducted.	But, as a start, we would like to recommend that the principles of crime prevention through environmental design be used as a guide to help minimize criminal activity in planning for and designing the proposed campus.	If there are any questions, please call Ms. Carol Societani of the Support Services Bureau at 529-3658.	Sincerely, LEE D. DONOHUE Chief of Police	By Year Conservices Bureau

Serving and Protecting with Aloha

TATIAL TATAL	 Mr. Vincent Shigekuni, Principal Page 2 January 10, 2003 3. Submit civil drawings to the HFD for review and approval. Should you have any questions, please call Battalion Chief Lloyd Rogers of our Fire Prevention Bureau at 831-7778. 	Sincerely, ATTILIO K. LEONARDI Fire Chief	cc: Allan Ah San, Research Corporation of the University of Hawai'i Sam Callejo, University of Hawai'i	
	Hitans Financial States Financial States Fina	Mr. Vincent Shigekuni, Principal PBR HAWAII 1001 Bishop Street Pacific Tower, Suite 650 Honolulu, Hawaii 96813-3429 Dear Mr. Shigekuni:	 Subject: University of Hawai'i - West O'ahu Draft Environmental Assessment/ Environmental Impact Statement Notice of Preparation We received your letter dated December 23, 2002, requesting our comments on the above- mentioned subject. The Honolulu Fire Department (HFD) requires that the following be complied with: 1. Provide a private water system where all appurtenances, hydrant spacing, and fire flow requirements meet Board of Water Supply standards. 2. Provide a fire department access road within 150 feet of the first floor of the most remote structure. Such access shall have a minimum vertical clearance of 13 feet 6 supporting the minimum 60,000-pound weight of our fire apparatus, and with a gradient not to exceed 20%. The unobstructed width of the fire apparatus access road shall meet the requirements of the appropriate county jurisdiction. All dead-end fire apparatus access road in avail. Such access road shall be provided with an approved turnaround having a radius complying with DTS standards. 	

LINA LARGE LINA LARGE RECENTANCE RECENTANCE TARGE OF LAW OF LAW POPOLO STATE OF LAW POPOLO POP	Mr. Vincent Shigekuni, Principal PBR Hawaii 1001 Bishop Street Pacific Tower, Suite 650 Honolulu, Hawaii 96813-3429 Dear Mr. Shigekuni:	<pre>Subject: University of Hawaii - West Oahu Draft Environmental Assessment/Environmental Impact Statement Notice of Preparation Thank you for your December 23, 2002, request for comments on the potential impacts of the proposed project on Department of Hawaiian Home Lands (DHHL) projects, policies, or programs. DHHL is supportive of the proposed project. As adjacent landowners, we should consider working together in the design and construction of offsite improvements, such as drainage systems, water distribution, and wastewater collection lines, which are used jointly by the proposed project and surrounding projects. A collaborative effort could achieve a significant cost reduction for our respective projects.</pre>	We appreciate the opportunity to comment on the proposed project and look forward to reviewing the draft EIS upon its completion. Aloha and mahaon Micah A. Kane, Chairman Designate Hawaiian Homes Commission
January 27, 2005 A W A I I A W A I I I A W A I I I I I I I I I I I I I I I I I	PRESIDENT FINAL ALLA ALLI: Battalion Chief Lloyd Rogers ETV 2. CHING. ASLA ALLI: Battalion Chief Lloyd Rogers ETV 2. CHING. ASLA SUBJECT: UNIVERSETY OF HAWAII – WEST OAHU ETME WEE-PRESIDENT INTER FRESIDENT INTER FRESIDENT	HILD OFFICE TID COPPORT Thank you for your letter dated January 10, 2003. We have reviewed your letter and will include the information you provided regarding fire prevention requirements in the Draft EIS. SERVELATC Again, thank you for your comments and your participation in the EIS process. If you have any questions, please do not hesitute to contract me. ASSOCIAT Again, thank you for your comments and your participation in the EIS process. If you have any questions, please do not hesitute to contract me. ASSOCIAT Sincerely. PBR HA MAI PBR HA MAI THAN SOLIT Market Sincerely. PBR HAWIN PIN HAWIN Thread Sincerely. PIN HAWIN PIN LEADED Market Sincerely. PI	ILLOFFICE UNDER OUR 13/132.05/Environmental Assessment-EISPNVPre-consultation Response Lin3BL-04 HFD.doe 0.4 Alternation at the consultation Response Lin3BL-04 HFD.doe 0.4 Alternation at the consultation at

DEPARTMENT OF PARKS AND RECREATION CITY AND COUNTY OF HONOLULU 1000 ULUDHIN STREET 6017 2009 6325 5131 • INTERNET MAXAAA MAAAAAAAAAAAAAAAAAAAAAAAAAAAAAA	JERENY HARRIS JANOR January 13, 2003	Mr. Vincent Shigekuni, Principal PBR Hawaii 1001 Bishop Street, Suite 650 Honolulu, Hawaii 96813 Dear Mr. Shigekuni:	Subject: University of Hawaii-West Oahu Draft Environmental Assessment Environmental Impact Statement Notice of Preparation Thank you for the opportunity to review and comment on the Draft Environmental Assessment relating to the University of Hawaii West Oahu Campus.	(The Department of Parks and Reoreation is unable to evaluate the impact of the proposed West Oahu Campus on any of our existing or proposed projects based upon the limited site development map submitted with your request.	We do wish to remain a consulted party to the EIS process and ask that you send us a map of the proposed campus that includes significantly more of the surrounding community at a larger scale to assist us in our evaluation of possible impact to our existing facilities.	Should you have any questions, please contact Mr. John Reid, Planner, at 692-5454. Sincerely,	WILLIAM D. BALFOUR, JR. WILLIAM D. BALFOUR, JR. Director (1970)	cc: Mr. Don Griffin, Department of Design and Construction
PBR January 27, 2005	LAND FLANSTRIC LAND FLANSTRIC ENVERTING THE AND ALL AND ALL KAINE, Chairman Designate Hawaiian Homes Commission WAL FRANK BRANDT FASLA CUMMUN THOMAS S. WITTEL ASLA P. O. Box 1879	Treasured HONOLULI, HAWAII 96805 R. STAN DUKCAN, ASLA R. STAN DUKCAN, ASLA Subgret 7.1. Church ASLA Subgret 7.1. Church ASLA Subgret 7.1. Church ASLA DRAFT ENVIRONMENTAL ASSESSMENT/ENVIRONMENTAL BRAFT ENVIRONMENTAL ASSESSMENT/ENVIRONMENTAL DRAFT ENVIRONMENTAL ASSESSMENT/ENVIRONMENTAL DRAFT ENVIRONMENTAL ASSESSMENT/ENVIRONMENTAL DRAFT ENVIRONMENTAL ASSESSMENT/ENVIRONMENTAL DRAFT ENVIRONMENTAL DRAFT ENVIRONMENTAL ASSESSMENT/ENVIRONMENTAL DRAFT ENVIRONMENTAL DRAFT ENVIRONMENTAL ASSESSMENT/ENVIRONMENTAL	JAMES LEGNARD, AICP Deal PUL, FALLO. <i>PRINCIPAL</i> <i>HILD OFFICE</i> Thank you for your letter dated January 8, 2003. We have reviewed your letter and the <i>HILD OFFICE</i> University of Hawaii is very appreciative of DHHL's support. We believe a University of GRANT MURAKAM. AICP Hawaii campus at this location will greated that the opportunities for university-level <i>semon Associate</i> education of DHHL's beneficiaries in Kapolei and in Nanakuli. We note that the University of Hawaii and vour Denartment have had onen discussions on the nossihility of ionity	RAYMOND T. HEAL ASLA development of required infrastructure. Associate KENININHIKAMA. ASLA Again, thank you for your comments and your participation in the EIS process. If you have ASSOCIATE any questions, please do not hesitate to contact me.	Sincerely,	100 Biss Street 100 Bi	III.0 Ожнак Vincent Shigekuni 10 Анния Биата Нид. Намату Батала Та.: (898) 541-333 Б.А.А.Ц. ранію виманся Б.А.А.Ц. ранію виманся Б.А.А.Ц. ранію виманся	Wurku Officts 12.21.2.2.2.2.2.2.2.2.2.2.2.2.2.2.2.2.2

BOARD OF WATER SUPPLY CITY AND COUNTY OF HONOLIAU So SOUTH REFETANIA STREET ONOLUUL, HI 96843 CHARLES A STED, Yoe-Chaimen	January 10, 2003 JAN DAROLVA ISK KAOUA, SR. AROLVA ISK. AKAOUA, SR. AROLVA ISK. AKAOUA, SR. AROLVA ISK. ALERONO SAROLVA ISK. AKAOSKAO, EXORICO AMALE Engineer DOWNA, FAY K. KYYOSKAO Deputy Amager and Chef Engineer	Mr. Vincent Shigekuni, Principal PBR Hawaii	1001 Bishop Sfreet Pacific Tower, Suite 650	Honolulu, Hawaii 90813-5429 Dear Mr. Shigekuni:	Subject: Your Letter Dated December 23, 2002 Regarding the University of Hawan-West Oanu Draft Environmental Assessment/Environmental Impact Statement Notice of Preparation	Thank you for the opportunity to comment on the proposed development.	A water master plan for the proposed development should be submitted for our approval. I ar master plan should include the proposed development layout, along with the proposed water facilities and necessary hydraulic calculations. The developer shall address source requirements, as well as provide transmission and storage facilities to service the development.	Board of Water Supply Rules and Regulations require the use of nonpotable water for the irrigation of large landscaped areas if a suitable supply is available. The applicant should research and address the availability and use of nonpotable water for irrigation.	If you have any questions, please contact Joseph Kaakua at 527-6123.	Very truly yours,	CLIFFORD S. IAMILE Manager and Chief Engineer			Pure Water our greatest need – use it wisely
PBR January 27, 2005	W. PAANDER AND FLANDER AND FLANDER Mr. Lester K.C. Chang, Director Department of Parks and Recreation W. PAANK BRANDT, FASLA City and County of Hanolulu	стамкам поли описация эле эли тамая. миттам, Ast. A Rapolei, Hawaii 96707 Рякатоемт	R. STAN DUWCAN, ASLA Attn: Mr. John Reid Execute Vice-President	RUSSELL Y.I. CHUNG. ASLA SUBJECT: UNIVERSITY OF HAWAII – WEST OAHU Executive Vice-President VICET PREPARATION NOTICE (EISPN) PRAFT ENVIRONMENTAL ASSESSMENT/ENVIRONMENTAL PRAFT PREPARATION NOTICE (EISPN)	JAMES LEONARD. AICP Dear Mr. Chang: PRANCTAL Dear Mr. Chang: HiLD OFFICE	Thank you for your letter dated January 13, 2003. We have reviewed your letter and please GRANT MURAKANI, AICP be assured that the DPR will remain a consulted party during the entire BIS process. The Draft FIS should include more information from which DPR will be able to evaluate the	TOM SCHWELL ALCP DIAL OF THE PROPOSED PROJECTION INFORMATION AND DEPOSED DPR PROJECTION OF ASSOCIATE Impact of the proposed project on existing or proposed DPR projects. RAYMOND: HIGA.ASLA Again, thank you for your comments and your participation in the EIS process. If you have	KENN RISHEKAA, ASLA AITY QUESHOLIS, PREASE GO HOL RESIDERE IO COLITACI ILE. Associate Sincerely,	Hewerenu Offster PBR HAWAII	AB DEARS STREET HOMELY FAVOUR 78913-3484 The 1889 529-15691 D. Lart 2006 Advention on American American	Vincent Shigekuni HLLO OWNER: Principal HLLO, AGON GENER SWE210	Hind, EliXWW 1957294202 The: (eliB) 95(1-30) Fac: (eliB) 95(1-30) Fac: (eliB) 95(1-40) Fac: (eliB) 95(1-40) Fac: (eliB) 95(1-40) Fac: eliB) 95(1-4	WALDWG OFFICE 2123 KAONIS TRBFT WALDWG 19739-2224 Thu: (200) 242-2302 FAX: (200) 242-202 EAML pBrinaui@Examed	

.

Linda Lingle corenora corenora Strate of HAWAI'I DEPARTMENT of EDUCATION P.C. BCX 280 HONOLIUL, HWAI'I 9804	OFFICE OF BUSINESS SERVICES January 15, 2003	Mr. Vincent Shigekuni PBR Hawai'i Pacific Tower, Suite 650 1001 Bishop Street Honolulu, Hawai'i 96816	Dear Mr. Shigekuni: Subject: University of Hawai'i - West Oahu, Ewa, Oahu TMK: 9-1-016: por. 108	The Department of Education (DOE) appreciates being consulted during the scoping process for the University of Hawai'i (UH), West Oahu, proposed campus location on 500 acres within the Housing and Community Development Corporation of Hawai'i (HCDCH) East Kapolei site.	The DOE is concerned that although the UH West Oahu proposal would eliminate roughly 46 percent of the planned housing for East Kapolei it also eliminates two out of the four school sites designated to serve the area. Our concerns are compounded by the passage of five years since the acceptance of the Final Environmental Impact Statement for East Kapolei Master Plan in 1998. In that time, the Kapolei Intermediate and Kapolei High schools both opened and reached	enuloument capacines very quickty. Inose schools can no longer be expected to serve students living in East Kapolei. Further more, the economic slowdown of the past few years has not only delayed the development of East Kapolei and its schools but also delayed the building of other West Oahu communities and schools. By re-looking at the proposal for East Kapolei and eliminating the homes that would now be used by UH, the area still requires two elementary schools, one middle school, and one high school. The areas remaining in the original East Kapolei plan include space for one elementary school and one middle school. There will still be a need for 12 acres for an elementary school and 50 acres for a high school.	AN AFFIRMATIVE ACTION AND EQUAL OPPORTUNITY EMPLOYER
January 27, 2005 A W A LI A W A LI annuary 27, 2005 Mr. Clifford S. Jamile, Manager and Chief Engineer Board of Water Supply Frank Brantr, FASLA Catabator 630 South Beretania Street	owas wrrten AsLA Honolulu, Hawan Yoo45 Passnew Astronovan Astha Attn: Mr. Joseph Kaakua	MALE AND	Thank you for your letter dated January 10, 2003. We have reviewed your letter and will warmwaxxw.AICP include a Water Master Plan in the Draft EIS. As part of the Water Master Plan, the consulting civil engineer will research and address the availability and use of nonpotable water for irrigation.	when that, ASLA Again, thank you for your comments and your participation in the EIS process. If you have associate any questions, please do not hesitate to contact me.	Sincerely, BROGLIU OFFICE AST TOPIES AND AST TOPIES	Turn Opsize Principal 0.4000 Strates 0.400B13/1322.05/Environmental Assessment-EISPNPre-consultation Response LinstBL-07 BWS.doc 0.4000 Strates 0.4000 Strates 0.4000 Strates 0.4000 Strates 0.4000 Strates 0.400 Strates 0.4000 Strates 0.400 Strates 0.4000 Strates 0.400 Strates	

January 27, 2005	Mr. Raynor M. Minami, Director Facilities and Support Services Branch State of Hawaii Department of Education P. O. Box 2360 Honolulu, Hawaii 96804	SUBJECT: UNIVERSITY OF HAWAII – WEST OAHU DRAFT ENVIRONMENTAL ASSESSMENT/ENVIRONMENTAL IMPACT STATEMENT PREPARATION NOTICE (EISPN) Dear Mr. Minami:	Thank you for your letter dated January 15, 2003. We have reviewed your letter and will include the information you provided on current school conditions in the Kapolei area in the Draft EIS. The University of Hawaii will be contacting DOE to discuss possible ways to mitigate the impact of school-aged children being generated from University-related housing. Current plans for UH West Oahu includes one elementary school site of 12 acres.	Again, thank you for your comments and your participation in the EIS process. If you have any questions, please do not hesitate to contact me. Sincerely	PBR HAWAII Nui-B R. Shurfur Vincent Shigekuni Principal o.00B131322.05Envionmental Assessment-EISPNVPre-consultation Response LisbBL-08 DOE.doe
$\operatorname{PBR}_{{}^{\mathrm{H}}{}_{\mathrm{M}}}$	LAND FLAND FLAND LANDSONADTAL STEDDS BAVRONADTAL STEDDS BAVRONADTAL STEDDS WM. FRANK BRANDT, FASLA CHABMAN THOMAS S. WITTEN ASLA	R. STAN DURCAN, ASILA RECUTIVE VICE-PRESIDEN RUSSEL Y.J. CHUKO, ASLA EXECUTIVE VICE-PRESIDEN VICENT SHOREKUN PRINCIPAL JAMES LEONARD, AICP	HILD OFFICE HILD OFFICE GRANT MURAKAMI, AICP SEWOR ASSOCIATE TOM SCHWELL, AICP ASSOCIATE	RAYMOND T. HIGA, ASI.A ASSOCIATE KEVIN NISHIKAWA, ASI.A ASSOCIATE	HONOLULU OFFICE HONOLULU OFFICE (001 brain-STREET ASET Town, 1981, 3-444 PL: (001) 322-401 PL: (001) 322-401 PL: (001) 322-402 PL: (001) 322-402 PL: (001) 322-402 PL: (001) 322-402 PL: (001) 322-402 PL: (001) 322-204 PL: (001) 322-204 PL: (001) 323-207 PL: (001) 3
Mr. Vincent Shigekuni Page 2 January 15, 2003	The DOE would have no objection to those two campuses being located in appropriate spots within the UT West Oahu 500 acres. Although your Environmental Impact Statement Notice of Preparation (EJSPN) did not mention residential areas being planned within the 500 acres, faculty and student family housing must have been considered in your plans. Schools located in proximity to institutions of higher education could have some mutual benefits.	Thank you for the opportunity to review and comment on this EISPN. Should you have any questions, please call Ms. Heidi Meeker of our branch at 733-4862. Sincerely yours,	して Raynor M. Minami, Director Facilities and Support Services Branch RMM:hy	cc: A. Suga, OBS	

,

•

January 27, 2005 Mr. Edward Y. Hirata, Director City and County of Honolulu Department of Transportation Services	650 South King Street, 3 ¹² Floor Honolulu, Hawaii 96813 Attn: Ms. Faith Miyamoto	SUBJECT: UNIVERSITY OF HAWAII – WEST OAHU DRAFT ENVIRONMENTAL ASSESSMENT/ENVIRONMENTAL IMPACT STATEMENT PREPARATION NOTICE (EISPN) Dear Mr. Hinata:	Thank you for your letter dated January 14, 2003 (your reference number TDP12/02- 19712R). We have reviewed your letter and will include the information you provided on the North-South Road Park-and-Ride in the Draft EIS.	Again, thank you for your comments and your participation in the EIS process. If you have any questions, please do not hesitate to contact me. Sincerely.	PBR HAWAII Nuc-D. R. Suight- Vincent Shigetuni Principal	O:UOB13/1322.05/Enviroamental Assessment-EISPNNre-consultation Response Lins/B1.09 DTS, doe	
LANDSAME AND LANDS	CIMANAN THOMAS S. WITTEN, ASLA Presydent R. Stan Duncan, ASLA Executive Vace-presydent	RUSELL Y.L CHUNG, ASLA Executive Vice-President VINCERT SHOBKUN PRINCPAL IAME LEONNED, ALCP	HILO UFFICE GRANT MURAKAMI, AICP SEMOR ASSOCIATE TOM SCHWELL, AICP	ASSOCIATE RAYMOND T. HIGA, ASLA ASSOCIATE KEVIN NISHIKAWA, ASLA ASSOCIATE	IIANOLULU OFFLE IIANOLULU OFFLE AGS TENEL, STIFFGO HARLT STIFFGO HARD TO AG 2003 231-432 Pr.2 (803) 231-431 Pr.2 (803) 231-431 E-MUL 3 3324-4401	Ifter Orster 101 Autors Struer Hito Locoset 7, 1967 and 1978 Hito Hitor Properties Fire (1979 501-1959 Proc. 1979 501-1959 Ewolt: Phyliol Blon ard	WAILAWG OFFICE VALLAWG OFFICE ACARILSTREET TOLE (1980) 242-2572 FOUR OFFICE FOUR OFFICE FOUR OFFICE FOUR OFFICE FOUR OFFICE
DEPARTMENT OF TRANSPORTATION SERVICES CITY AND COUNTY OF HONOLULU GEORGIF HANG SERVICES GEORGE HANG SEA AT THE AND	January 14, 2003 Provident TPD12/02-19712R	Mr. Vincent Shigekuni, Principal PBR Hawaii 1001 Bishop Street ASB Tower, Suite 550 Hornolulu Hawayi 96813-3429	Dear Mr. Shigekuni: Subject: <u>University of Hawaii-West Oahu</u>	In response to your December 23, 2002 letter, we reviewed the project information provided. We would like to provide you with information on a planned transportation facility in the vicinity of the subject project.	The Primary Corridor Transportation Project (PCTP) proposed by this department includes a North-South Road Park-and-Ride. This park-and-ride is to be located on the Koko Head side of North-South Road, between Farrington Highway and the H-1 Freeway. Its location allows using the future North-South Road Interchange and the H-1 Freeway for bus access. The North-South Road Park-and-Ride will be one component of a multi-modal alternative that will provide a balanced transportation system. The PCTP will improve connections between Kapolei and the Primary Urban Center, and the connections between the subject project and the other University of	rawaii campuses. We look tol wate to community commandiu with the Oniversity of rawaii. Should you have any questions regarding this matter, please contact Faith Miyamoto of the Transportation Planning Division at 527-6976. Sincerely,	Clerch D. Mr CHERYL D. SOON Director

--

LINDA LINGLE Benyamin-j, cavetano governor



JAN 2 Z

ANTHONY J.H. CHING EXECUTIVE OFFICER

> STATE OF HAWAIIProsection DEPARTMENT OF BUSINESS, ECONOMIC DEVELOPMENT & TOURISM LAND USE COMMISSION P.O. Box 2359

P.O. Box 2359 Honolulu, HI 96804-2359 Telephone: 808-587-3822 Fax: 808-587-3827

January 21, 2003

Mr. Vincent Shigekuni, Principal PBR Hawaii Pacific Tower, Suite 650 1001 Bishop Street Honolulu, Hawaii 96813-3429

Dear Mr. Shigekuni:

Subject: University of Hawaii-West Oahu Draft Environmental Assessment/ Environmental Impact Statement Notice of Preparation (EA/EISPN) We are in receipt of your letter dated December 23, 2002, regarding the EA/EISPN for the University of Hawaii-West Oahu project, and have the following comments: We confirm that the 500-acre site proposed for the University of Hawaii-West Oahu campus is located within the 1,300-acre area urbanized by the Land Use Commission (LUC) for the East Kapolei project in LUC Docket No. A99-728/Housing and Community Development Corporation of Hawaii, State of Hawaii. We note that one of the objectives of this project was to generate income from the sale of large lot improved development parcels in East Kapolei to individual developers in support of a 991-acre campus mauka of the H-1 Freeway. Given the new campus location, the DEIS should include a discussion on what brought about the change in location and on the manner in which development of the campus will now be financed. In addition, the DEIS should include an assessment of the impacts of the campus upon the remaining land uses constituting the East Kapolei project as well as proposed projects in the region, including Gentry Investment Properties' Ewa Makai development.

Mr. Vincent Shigekuni, Principal January 21, 2003 Page 2 2) We believe that the proposed campus on the East Kapolei site represents a substantial change in the project from the development originally represented to the LUC. Given the nature and magnitude of the change, a motion to amend the decision and order issued in the above-mentioned docket should be filed with our office pursuant to §15-15-70. Hawaii Administrative Rules. Such a motion, together with supporting affidavits, exhibits, and legal memoranda of points and authorities, would provide the Commission with relevant and reliable information upon which to make a determination in regard to the revised project.

3) We reserve further comments on the subject project in our review of the DEIS.

We have no further comments to offer at this time. Thank you for the opportunity to comment on the subject EA/EISPN. Please feel free to contact Bert Saruwatari of my office at 587-3822, should you require clarification or any further assistance.

Sincerely,

Executive Off ANTHONY

c: Office of Environmental Quality Control

	LINGALINGLE GOTENNIA COTENNIA	STATE OF HAWAII BEPARTMENT OF LAND AND NATURAUFESOURCES DEPARTMENT OF LAND AND NATURAUFESOURCES CONSERVING CONSERVINC CONSERVINC CONSERVINC CONSERVINC CONSERVINC CONSERVINC CONSERVINC CONSERVINC CONSERVINC CONSERVINC CONSERVINC CONSERVINC CONSERVINC CONSERVINC CONSERVINC CONSERVINC CONSERVINC CONSERVINC CON	CALO DATAON CONVACE PAGE POINTER HONOLUU, HAVAII 96809 NATAR VANUERENANNA VANUERENANNA	LD-NAV LD-NAV UHWOCAMPUSPDEABR.RCM L-164/4066/180/104	PBR Hawaii Vincent Shigekuni, Frincipal 1101 Bishop Street Pacific Towers, Suite 650 Honolulu, Hawaii 96013-4329	Dear Mr. Shigekuni:	SUBJECT: Pre-Assessment Consultation - Preparation of a Draft Environmental Assessment for the Proposed University of Hawaii West Oahu Campus Project, Oahu, Hawaii - TMK: 1^{st} / 9-1-16 Portion of 108	Thank your for the opportunity to review and comment on the subject matter.	A copy of your letter dated December 23, 2002 (surmary), and campsite drawing pertaining to the subject matter was distributed to the following Department of Land and Natural Resources' Divisions for their review and comment:	 Division of Aguatic Resources Division of Forestry and Wildhife Division of State Parks Engineering Division Commission on Water Resource Management 	- Land Division Planning and Technical Services - Oahu District Land Office	Attached herewith is a copy of the Division of Forestry and Wildlife, Engineering Division and Commission on Water Resource Management comments.	The Department of Land and Natural Resources has no other comment to offer at this time.	Should you have any guestions, please feel free to contact Nicholas A. Vaccaro of the land Division Support Services Branch at 587-0384.	Very truly yours,	Myadane & Unolui	L DIERDRE S. MOMITA
January 27, 2005	Mr. Anthony J. H. Ching, Executive Officer State of Hawaii Department of Business, Economic Development & Tourism	Land Use Commission P. O. Box 2359 Honolulu, Hawaii 96804-2359	Attn: Bert Saruwatari	SUBJECT: UNIVERSITY OF HAWAH – WEST OAHU DRAFT ENVIRONMENTAL ASSESSMENT/ENVIRONMENTAL IMPACT STATEMENT PREPARATION NOTICE (EISPN)	Dear Mr. Ching: Thank you for your letter dated January 21, 2003. We have reviewed your letter and offer the following resonance:	Conception of the second se	 The Draft EIS will include an assessment of the impacts of the campus upon the remaining land uses constituting the East Kapolei project as well as proposed projects in the region, including Gentry Investment Properties' Bwa Makai development. 	2. As recommended, the University of Hawaii will consult with the LUC regarding the	Approximation to the antion to antion to antion the occision and order issued in DOC DOCKEL No. A99-728/Housing and Community Development Corporation of Hawaii, State of Hawaii with the Land Use Commission pursuant to Subsection 15-15-70, Hawaii Administrative Rules.	We understand the LUC's desire to reserve further comments on the subject project pending your review of the Draft EIS. We will look forward to your comments at that time.	Again, thank you for your comments and your participation in the EIS process. If you have any ouestions, please do not hesitate to contact me.	Sincerelv	PBR HAWAII			v meett Shigekuni Principal	0.VOB1301322.05)Environmental Assessment-EISPNVPre-consultation Response LastBL-10 DBEDT LUC.doc

R. STAN DUNCAN, ASLA EXECTIVE VICE-PRESIDENT RUSSELL Y.J. CHUNG, ASLA EXECUTIVE VICE-PRESIDENT VINCENT SHIGEKUNI PRACTPAL

James Leonard, AiCP Principal Hild Office

GRANT MURAKAMI, AICP Senior Associate

TOM SCHNELL, AICP ASSOCIATE

RAYMOND T. HIGA, ASLA Associate Kevin Nishikawa, ASLA Associate

WM. FRANK BRANDT, FASLA Chairman

LAND PLANNING LANDSCAPE ARCHITECTURE ENVIRONMENTAL STUDIES

BR 1 V M THOMAS S. WITTEN, ASLA PRESIDENT

- 5
- ų

IIAMALALAU OPATES IIOLI BISHOS STREET ASB TAWK, SUTTA 650 HOWALALL, HAWAT 15681 3-348 TEL 6003 523-1403 FAX (803) 523-1403 E-MALE 5ystadmin@phtawaii.com

Sincer

PBR F III.rd OFFICE OID Aureus STENEET HLIG LACCOON CENTRE, SUITE 310 HLIG, HANALTI 95720-4502 The, (1980) 951-4959 E-MAIL: Phyliol@lavea.net

Vincer

WALLUKU OFFICE 2123 KAOHU SPREET WALLUKU, HAWAN 195793-2204 Tha. (808) 242-2878 FAAT (808) 242-2902 EMAIL JDHTRAH @BAVLICE

DEPARTMENT OF LAND AND NATURAL RESOURCES ENGINEERING DIVISION LA/NAV	COMMENTS For your information, the project site is located in Zone D. This is an area in which flood hazards are undetermined.	However, if future studies determine that the project site is within the flood zone, the project must comply with rules and regulations of the National Flood Insurance Program (NFTP). If there are questions regarding the NTP, please contact the State Coordinator, Mr. Sterling Yong, of the Department of Land and Natural Resources at 587-0348. If there are questions regarding flood ordinances, please contact Mr. Robert Summolo at 233-4254 or Mr. Mario Siu Li at 523-4247 of the City and County of Honoluiu, Department of Planning and Permitting. In addition, the draft Barvironmental Assessment should include project water demand and infrastructure required to meet water demands. For your information, the former Ewa Sugar/Oahu Sugar Company Pesticide Mixing and Loading Site is located adjacent and to the East of the proposed UH Postlogment of TrAft: 9-1-State Department of Health, Hazard Evaluation and Emergency Response Office as HIGH priority. Should you have any questions, please call Mr. Andrew Monden of the Planning Branch at 587-0229. Should you have any questions, please call Mr. Andrew Monden of the Planning Branch at 587-0229. NavIDMMAXMISUZIBOAMINUTEDRAMACOMINING	
•			
UNA LINGLE CONTRICT AND	January 2, 2003 LD/NAV LD/NAV Ref.: UHWOCAMPUSDEA.CMT MEMORANDUM: MEMORANDUM:	<pre>TO: XXX Division of Aquatic Resources XXX Division of Forestry & Wildlife Na Ala Hele Trails XXX Division of State Parks XXX Engineering Division of State Parks XXX Engineering Division of State Parks XXX Commission on Mater Resource Management Land Division mater Resource Management Land Division Branches: XXX Oahu District Land Office XXX Oahu District Land Office YXX District Park State XXX Oahu District Land Office YXX District Park State Y Plaaming Mast Oahu Campus, Oahu, Hawaii University of Hawaii (808) 521-5631 MMX: 1*7 9-1-16 Fortion of 108 MMX: 1*7 9-10-16 Forti</pre>	Date:

`-

.

.

. -

Division of Forestry & Wildlife

Land Division FROM: Michael G. Buck, Administrator

Dierdre S. Mamiya, Administrator

THRU:

A: Michael G. Buck, Administrator Division of Forestry and Wildlife

SUBIECT: Pre-Assessment Consultation - Preparation of a Draft EA covering the proposed UH - West Oahu Campus, Oahu, Hawaii Consultant ; PBR Hawaii TMK; 9-1-16; por. 108.

The Division of Forestry & Wildlife (DOFAW) has reviewed the subject document regarding impacts the project may have on DOFAW management programs and provide the following recommendations for your consideration. DOFAW is concerned with three endangered plants that are known to be present in the Kapolei-Kalaeloa-Ewa plains area. The endangered plants are: 1) *Chamaespce skottsbergii var. skottsbergii, 2) Achyranthes splendens var. rotundata,* and 3) *Abutilon menziesii.* In particular, the *Abutilon menziesii* is a species of concern at the proposed UH - West Oahn Campus site. Any endangered plant discovered at this site is protected by State Law, Chapter 195D-4, HRS.

A "Habitat Conservation Plan" (HCP) to protect the endangered plants in this region is currently being developed for the North/South Road Project which is a part of a larger regional effort to address endangered plant issues through the East Kapolei Master Plan. The State Department of Transportation, State Department of Land and Natural Resources, Housing and Community Development Corporation of Hawaii, and University of Hawaii are major stakeholders and landowners involved in the mitigation of endangered plants in this area. However, the current landowner definition of the State's Endangered Species law (Chapter 195D, HRS) prevents public land at UH - West Oahn Campus from being part of that HCP. DOFAW has proposed an administrative Bill in this coming legislative session to amend the HCP to include public lands as landowners.

Nick Vaccaro Page 2 Should this important amended legislation pass, it will allow the University of Hawaii to become eligible for a HCP. Please call Ms. Vickie Caraway, State Botanist at 587-0165 or Mr. Paul Comy, Wildlife Program Manager at 587-4176 if you have questions regarding DOFAW's review of this proposed UH - West Oahu Campus. As always, we appreciate the opportunity to comment on this project.

C: DOFAW, Oahu Branch
 Vickie Caraway, State Botanist
 Paul Conry, Wildlife Program Manager

		102			
LINDA	ALINGLE		EPUC T, HIRANO Active Camareradori	PBR	2005 27 2005
	127 200 2	it o the 2	MEREDITH J. CHING CLATCON W. DELA CRUZ CLATVOME L. FUCKIO, CALZ CLATVOME L. FUCKIO, CALZ ERRAG CM STALS. JR FIERBERT M. RICHWARDS. JR	I I V V I I	
		STATE OF HAWAII DEPARTMENT OF LAND AND NATURAL RESOURCES GOMMISSION ON WATER RESOURCE MANATEMENT	DEAN A. NAKAND Active Using Describe	ENVIRONAENTAL STUDIES	Ms. Dierdre S. Mamiya, Administrator State of Hawaii
	. 1-2 •			WM, FRANK BRANDT, FASLA <i>Chairaian</i>	Department of Land and Natural Resources Land Division
		January 10, 2003		THOMAS S. WITTEN, ASLA President	P. O. Box 621 Honolulu, Hawaii 96809
70: 1		Ms. Dede Mamiya, Administrator Land Division	tef: UH W Oahu Campus.dr	R. STAN DUNCAN, ASLA EXECUTIVE VICE-PRESIDENT	Attn: Mr. Nicholas Vacarro
FROI	Ξ	Dean A. Nakano, Acting Deputy Director AL MARA Commission on Water Resource Management (CWRM)		RUSSELL Y.J. CHUNG, ASLA Executive Vice-President Vincent Shigekuni	SUBJECT: UNIVERSITY OF HAWAII ~ WEST OAHU DRAFT ENVIRONMENTAI, ASSESSMENTRONMENTAI,
SUB,	JECT:	Pre-Assessment Consultation – Preparation of a Draft Environmental As. Proposed University of Hawali West Oahu Campus, Oahu, Hawali	sessment	PRINCIPAL JAMES LEONARD, AICP PRINCIPAL DIA OFFICIPAL	IMPACT STATEMENT PREPARATION NOTICE (EISPN)
FILE	NO.:	UHWOCAMPUSDEA.CMT		nuu urrice	Dear Ms. Mamiya:
are r	Thank) narked bek	you for the opportunity to review the subject document. Our comments rei	ated to water resources	GRANT MURAKAMI, AICP SENIOR ASSOCIATE	Thank you for your letter dated January 18, 2003 (your reference number LD-NAV UHWOCAMPISPDEARP RCM 1.1644066180/104) for circulating our letter among the
arm arm	In gene sures and t ful effects	trail, the CWRM strongly promotes the efficient use of our water resources use of alternative non-potable water resources whenever available, feasible to the ecosystem. Also, the CWRM encources the protection of water re e maintenance of streams and the replenishment of aquifers.	through conservation e, and there are no charge areas, which are	TOM SCHNELL, AICP ASSOCIATE RAYMOND T. HIGA, ASLA ASSOCIATE	various DLNR divisions, and for compiling the responses. We have reviewed the various comments and offer the following responses:
[x]	We recon Plan.	mmend coordination with the county government to incorporate this project into the county's	Water Use and Development	KEVIN NISHIKAWA, ASLA Associate	Engineering Division
[x]	We recon project int	mmend coordination with the Land Division of the State Department of Land and Natural Re- to the State Water Projects Plan.	sources to incorporate this		1. The information provided on flood hazards will be included in the Draft EIS.
Ξ	We are cuthis project requirement	oncerned about the potential for ground or surface water degradation/contamination and rec ct be conditioned upon a review by the State Department of Health and the developer's acc ents related to water quality.	ommend that approvals for splance of any resulting	KONOLULU OFFICE 1001 [Sisting Struggt	2. While the information requested on project water demand and infrastructure was requested for the draft Environmental Assessment, that information will not be available undit the Dack FIG The Dack FIG will include water and information
[]	A Well Co developed	onstruction Permit and/or a Pump Installation Parmit from the Commission would be require d as a source of supply for the project.	d before ground water is	ASB Tower, SHITE 650 HONOLULI, FLAWAY 196813-3484 TEL (808) 521-5331 E. S. 1900, 531-1400	מאמותטוס שווון נווב בומו בנוט. אופינו בוומיני אוו ביוט שווו ווועושפי שווע ווווו מאופיר שוו ווווו מאופיר שני מי master plans.
2	The prop(the Comn	osed water supply source for the project is located in a designated water management area mission would be required prior to use of this source.	. and a Water Use Permit from	E-MAIL: sysadnin@pbthawaii.com	 Information on the Ewa Sugar/Oahu Sugar Company Pesticide Mixing and Loading circum the instructed in the Dark 200
[]	Groundwa	ater withdrawals from this project may affect streamflows, which may require an instream flo	w standard amenóment.	HILO OFFICE	
Ξ	We are constreams v correspon control.	oncerned about the potential for degradation of instream uses from development on highly e within or near the project. We recommend that approvals for this project be conditioned up offing county's Building Department and the developer's acceptance of any resulting require	rodible slopes adjacent to n a review by the ments related to enslon	HLGL LAGONI CENTRA HLGL LAGONI CENTRA, SWIE 310 FILL, FIAMAT'1 96720-4262 FILL, 6005) 961-3333 FLA, 1001 961-4389 F-MARI : Phillice/Devined	Division of Forestry and Wildlife 1. We appreciate the information provided on the Habitat Conservation Plan and we
	If the prop amend the	posed project includes construction of a stream diversion, the project may require a stream e instream flow standard for the affected stream(s).	diversion works permit and		will incorporate it into the Draft EIS.
Ξ	If the prop	posed project alters the bed and banks of a stream channel, the project may require a stream	n channel alleration permit.	WALURU OFFICE 2123 KAONU STRFFT WALLUSU HAWATI 9959-2264 The 16080 347-3578	
×	OTHER:			FAX: (608) 242-2902 EMAIL: pbritati@lava.net	
	demand demand	ommend that the uran Environmental Assessment include the projected p. 25. an anticipated schedule to buildout, and the proposed water sources to 35.	otable and non-potable meet the potable		

.

If there are any questions, please contact Lenore Y. Nakama at 587-0218.



SUBJECT: UNIVERSITY OF HAWAII – WEST OAHU DRAFT ENVIROMENTAL ASSESSMENT/ENVIRONMENTAL IMPACTS STATEMENT PREPARATION NOTICE January 27, 2005 Page 2

Ms. Diedre S. Mamiya, Administrator

Commission on Water Resource Management

- The University of Hawaii will promote the efficient use of water resources through conservation measures and use of alternative non-potable water resources whenever available. feasible and there are no harmful effects to the ecosystem.
- The University of Hawaii is also in accord with the protection of water recharge areas, which are important for the maintenance of streams and the replenishment of aquifers.
- The University of Hawaii will coordinate with the Board of Water Supply to incorporate the proposed UH West Oahu Campus into the City and County of Honolulu's Water Use and Development Plan.
- The University of Hawaii will coordinate with the Land Division of DLNR to incorporate the proposed UH West Oahn Campus into the State Water Projects Plan.
- The information requested on the projected potable and non-potable demands, and an anticipated schedule to buildout will not be available for the EIS Preparation Notice, but will be included in the Draft EIS.

Again, thank you for your comments and your participation in the EIS process. If you have any questions, please do not hesitate to contact me.

Sincerely,

PBR HAWAII

Vires P.Sm

Vincent Shigekuni Principal O:VOB13V1322.05/Environmental Assessment-EISPN/Pre-consultation Response Lirs/BL-11 D1NR.doc

PBR

.

٠

LAND FLANNING LANDSCAPE ARCHITECTURE ENVIRONMENTAL STUDIES

WM. FRANK BRANDT, FASLA CHAIRAIAN

TROMAS S. WITTEN, ASLA PRESIDENT

R. STAN DUNCAN, ASLA Executive Vice-President

RUSSBLL Y.J. CHUNG, ASLA EXECUTIVE VICE-PRESIDENT VINCENT SHIGEKUNI PRINCIPAL

JAMES LEONARD, AICP Principal Hilo Office

GRANT MURAKAMI, AICP Sevior Associate Tom Schnell, AICP Associate RAYMOND T. HIGA, ASLA Associate

KEVIN NISHIKAWA, ASLA Associate

PBR HAWAII

II OKOLULU OPPICZ II OBISCO TSNEST ASIB TOWER, SUITE 550 HONOLULI, HAWAT 19681 2-348 TEL (2002) 2-1-651 TAZE (2003) 221-4021 TAZE (2003) 223-4020 E-44012 systelnin@pohenvuil.com

HILO OFFICE DIA AUVERS ATREET HILO LAROON CENTRA, SUITR 310 HILO, HAWAY 195720-4202 The, (6003) 964 - 9333 FAX: (6003) 964 - 9399 G-MAIL: Phiblio@haw.nel

WALLUKU OPPICE 2123 KAONU STREET 2123 KAONU STREET TILL (2019) 247-2878 FAC (2018) 242-2873 FAC (2018) 242-2972 EMML: phyriaul@lave.net

Mr. Vincent Shigekuni Page 3 January 22, 2003	Please refer to the attached Guidelines for Sustainable Building Design In Hawaii: A planner's checklist and A Contractor's Waste Management Guide " for additional information. Sincerely, Sincerely, Manufer, H. Kaya Buergy, Resources, and Technology Program Administrator Enclosures Concontration of the University of Hawaii) Sam Callejo (University of Hawaii)	
Mr. Vincent Shigekuni Page 2 January 22, 2003	 All plans and designs for new or renovated facilities using state funds or located on state land and incorporating the use of hot water shall include a comparative analysis to determine the cost-benefit of using a conventional water heating system or a solar water heating system. The analysis shall be based on the projected life-cycle costs to purchase and operate the water heating systems. If the life-cycle statistic positive, the facility shall incorporate solar water heating. If water heating entirely by solar is not cost-effective, the analysis shall also evaluate the life-cycle, cost-benefit of solar water heating for preheating water. Each Department shall be responsible for conducting an analysis for every facility that provides hot water. We recommend that you consult the City & County of Honolulu Energy Code early on in your project. Hawaiian Electric Co., Inc., (HECO) may also have demandside management programs that offer rebates and/or incentives for installation of energy efficient technologies. We recommend that you consult the conductions and your project. 	 efficient design practices and technologies the prosidered, as appropriate, include: methods and technologies that could be considered, as appropriate, include: Use of natural ventilation to increase comfort of occupants; Use of daylighting Maximum use of natural lighting without heat gain; Use of high efficiency compact fluorescent lighting; Use of high efficiency compact fluorescent lighting; Use of natural ventilation to increase comfort of occupants; Use of high efficiency compact fluorescent lighting; Use of finite efficiency compact fluorescent lighting; Use of finite fluorescent lighting in the efficiency of the control and to minimize heat gain to area; and the celling tans; Use of photovoltatics, fuel cells and other renewable energy sources. Steery and recycled-content products: Develop a job-site recycling plan for construction and tecycle as much construction and tecycle as much construction waste as possible; Incorporate provisions for recyclables; and construction and tecycle as much construction waste as possible; Incorporate provisions for recyclables; and Specify and use products with recycled content such as: steel, contrate aggregate fill, drywall, carpet and glass tile.

.

.



January 27, 2005

Mr. Maurice H. Kaya, Energy, Resources and Technology Program Administrator

> WM. FRANK BRANDT, FASLA CHAIRMAN CHOMAS S. WITTEN, ASLA President

Department of Business, Economic Development & Tourism

Strategic Industries Division Honolulu, Hawaii 96804

State of Hawaii

P. O. Box 2359

R. STAN DUNCAN, ASLA EXECUTIVE VICE-PRESIDENT

RUSSELL Y.J. CHUNG, ASLA EXECUTIVE VICE-PRESIDENT

VINCENT SHICEKUNI PRINCIPAL

JAMES LEONARD, AICP PRINCIPAL HILO OFFICE

Dear Mr. Kaya:

GRANT MURAKAMI, AICP Sevior Associate TOM SCHNELL, AICP ASSOCIATE

following responses:

<u>..</u>:

KAYMOND T. HIGA, ASLA Associate

The Draft EIS will address the applicable portions of the State Energy Conservation Goals as found in Chapter 344, HRS ("State Environmental Policy") and Chapter

226, HRS ("Hawaii State Planning Act").

Thank you for your letter dated January 22, 2003. We have reviewed your letter and offer the

DRAFT ENVIRONMENTAL ASSESSMENT/ENVIRONMENTAL IMPACT STATEMENT PREPARATION NOTICE (EISPN)

SUBJECT: UNIVERSITY OF HAWAII - WEST OAHU

The Draft EIS will note that hot water use on campus is subject to Administrative

Directive No. 98-03 "Policy Governing the Use of Solar Water Heating Systems for

State Facilities."

e.

for the campus.

5.

4.

The Draft EIS will also address the applicable portions of Act 77, SLH 2002,

"Relating to Energy Resources."

The City and County of Honolulu Energy Code will be consulted during the design

The University of Hawaii will consult with HECO demand-side programs that offer rebates and/or incentives for installation of energy efficient technologies (at its

CEVIN NISHIKAWA, ASLA ASSOCIATE

3

Howen, Jurg OFFICE (201) Bistore STREET ASB TOWER, SUITE 650 Howen, Jurk, Hawkin 7081 5-344 Ta: (5803 523-1402 Fox: (5803 523-1402 Fox: (5803 523-1402

HILA OPPICE 101 Aurents Stream 101 Aurents Carrens, Surre 310 HILA, HAWAT 96720-4262 Taz: (8039 501-4353 FAz: (8039 501-4353 FAz: (8039 501-4353 FAZ: (8039 501-4353)

The Draft EIS will address energy efficient design practices and technologies. 6

existing campuses), and will design the proposed campus accordingly.

7. The Draft EIS will also address recycling and recycled-content products.

WAILJERU OFFICE 2123 KAOHU STWEET WALLINU, HAWN 196793-2204 TEL (808) 242-2878 FAX: (808) 242-2902 EMAIL: phyrnul@lavl.net

Mr. Maurice H. Kaya

SUBJECT: UNIVERSITY OF HAWAII – WEST OAHU DRAFT ENVIRONMENTAL ASSESSMENT/ENVIRONMENTAL IMPACT STATEMENT PREPARATION NOTICE fanuary 27, 2005 Page 2

The Guidelines for Sustainable Building Design in Hawaii: A Planner's Checklist and A Contractor's Waste Management Guide will be consulted during the design and construction of the campus. ø

Again, thank you for your comments and your participation in the EJS process. If you have any questions, please do not hesitate to contact me.

Sincerely,

PBR HAWAII

Vizzek. Suga

Vincent Shigekuni Principal 0:VOB1311322.05/Environmental Assessment-EISPNIPte-consultation Response Lar/BL-13 DBEDT ERTD, doc

Mr. Vincent Shigekuni, Principal January 27, 2003 Page 2	a. Construction activities, including clearing, grading, and excavation that result in the disturbance of equal to or greater than five (5) acres of total land area. The total land area includes a contiguous area where multiple separate and distinct construction activities may be taking place at different times on different schedules under a larger common plan of development or sale. An NPDES permit is required before the commencement of the construction activities,	Note: After March 10, 2003, an NPDES permit will be required for construction activities, including clearing, grading, and excavation that result in the disturbance of one (1) acre or more.	 Discharge of treated effluent from leaking underground storage tank remedial activities; 	c. Discharge of once through cooling water less than one (1) million gallons per day;	d. Discharge of hydrotesting water,	e. Discharge of construction dewatering effluent;	f. Discharge of treated effluent from petroleum bulk stations and terminals;	g. Discharge of treated effluent from well drilling activities;	h. Discharges of treated effluent from recycled water distribution systems;	i. Discharges of storm water from a small municipal separate storm sewer system; and	j. Discharge of circulation water from decorative ponds or tanks.	The CWB requires that a Notice of Intent (NOI) to be covered by a NPDES general permit for any of the above activities be submitted at least 30 days before the commencement of the respective activities. The NOI forms may be picked up at our office or downloaded from our website at http://www.state.hi.us/doh/forms/genl-index.html .	 The applicant may be required to apply for an individual NPDES permit if there is any type of activity in which wastewater is discharged from the project into State waters, 	and/or coverage of the discharge(s) under the NPDES general permit(s) is not permissible. An application for the NPDES permit is to be submitted at least 180 days before the commencement of the activities. The NPDES application forms may also be picked up at our office or downloaded from our website at http://www.state.bi.us/doh/eh/com/forms/ndiv-index.html.
(()) () () () () () () () () () () () ()	STATE OF HAWAII DEPARTMAII Proclosition in www.mananana. Proclosition in the proclosition in the proclosition in the proclosition in the proclement in the p	January 27, 2003	, Principal cific Tower, Suite 650	13-3429		ation for the Preparation of an Environmental Assessment (PEA) and wironmental Inmact Statement Prenaration Notice (FISDN)	ity of Hawaii, West Oahu Campus p Key: 3-1-016:108 (por)	rtunity to review and comment on the subject proposal. The PEA/BISPN	us branches of the Environmental Health Administration. We have the	(<u>a</u> M	s of Engineers should be contacted to identify whether a Federal permit partment of Army permit) is required for this project. Pursuant to	 of the Federal Water Pollution Act (commonly known as the "Clean Section 401 Water Quality Certification is required for "[a]my applicant are or permit to conduct any activity including, but not limited to, the operation of facilities, which may result in any discharge into the "" 	utant Discharge Elimination System (NPDES) general permit coverage is following activities:	sociated with industrial activities, as defined in Title 40, Code of Federal ctions 122.26(b)(14)(j) through 122.26(b)(14)(ix) and 122.26(b)(14)(xi);
LINDA LINGLE LINDA LINGLE COMENCIA OF HAVAU			Mr. Vincent Shigekuni, PBR HAWAII 1001 Bishop Street, Pac	Honolulu, Hawaii 9681	Dear Mr. Shigekuni:	Subject: Consulta Draft En	Universi Tax Mar	Thank you for the oppo	was routed to the variou following comments.	Clean Water Branch (C	 The Army Corp (including a Dep 	Section 401 (a)(Water Act"), a { for Federal licen construction or . navigable water;	 A National Pollt required for the 	Storm water ass Regulations, Ser

 The first distribution of a basic distribution distributin distribution distribution distribution distribution distribu	 The off Advictor for head a forcing interaction for the force. The force advices of the force advices	 The function conduction should be found and a conduction should be found and and and and a conduction should be found and and and and and and and and and a	Mr. Vincent Shigekuni, Principal January 27, 2003 Page 3	Mr. Vincent Shigekuni, Principal January 27, 2003 Page 4
 To control during longer and control and project direct and the project direct of the project direct and the project direct and the provision of the project direct and the during the project direct and the during the project direct and the during during the during the during the during the during the duri	 a constant of the formation of the formation	 Canal tangence, have an event of starting hand a way from the project at the advention of starting hand a way from the project at the advention of starting hand a way from the project at the advention of starting hand a way from the project at the advention of starting hand a way from the project at the advention of starting hand a way from the project at the advention of starting hand a way the advention advention of starting hand a way the advention advention of starting hand a way the advention advention advention of starting hand a way the advention advention advention of starting hand a way the advention advent	4. Hawaii Administrative Rules, Section 11-55-38, also requires the owner to either submit a copy of the NOI or NPDES permit application to the State Department of Land and	e. Provide adequate dust control measures during weekends, after hours, and prior to daily start-up of construction activities; and
If you have any questions, please contract havy Ching at (680) 586-4200. If you have any questions, please contract havy Ching at (680) 586-4200. Massenser plans mass conform to applicable provisions of the Department of Haath's attentions and attentions of the Department of Haath's attention and haloor. Attention and hal	 Tyru law any question, please contract hard y Ching at (600) \$564-500. Wate any question, please contract hard y Ching at (600) \$564-500. Wate any question, please contract hard y Ching at (600) \$564-500. Wate any question, please contract hard y Ching at (600) \$564-500. Wate any question, please contract hard y Ching at (600) \$564-500. Wate any question, please contract hard y the Administrative Khish of the Department of Haddh. Wate any question, please contract hard y the Administrative Khish of the Operational of the Department of Haddh. Wate any question, please contract hard y the Administrative Khish of the Administrative Khish of the Administrative Khish. Wate Haddon and for the and administrative khish. Wate Plank and administrative khish.	 The number of the contraction of contraction contraction of contraction of contraction of contraction o	Natural Kesources, State Enstoric Preservation Division (SHPD) or demonstrate to the statisfaction of the DOH that the project, activity, or site covered by the NOI or annifestion has been or is being requisived by CEDD.	f. Control dust from debris being hauled away from the project site.
 To the net of partition and indoor Air Conditioning that the Administration Relates A thread and conditioning that the Administration Relates Air Conditioning that the Administration Relates Air Conditioning and the Administration Relation Relat	The sequence plane and notices of the Department of Rearth. The sense of plane mate order or propertion. The sense or propertion or propertion. The sense or properition or properition. The sense or properition or properition or properition. The sense or properition or properition or properition. The sense or properition or properition or properition or properition or properior properint properition or properition.	 Not use any queue pare some tar (Not a reduction and the Aminitative Nation Aminitative Natin Aminitative Nation Aminitative Nation Aminitative Nation A		If you have any questions, please contact Barry Ching at (808) 586-4200.
 An instance to parameter of hearthy a diministrative ball or project activities shall comply with the Antinistrative Balles of the Department of Health. An instance activities in a special control or static control or static special control or static s	 A manufactorial parameter of the Dipertment of the Di	 A manufacture and a manufacture and	If you have any questions, please contact the CWB at (808) 586-4309.	Noise. Radiation and Indoor Air Quality (NRIAQ) Branch
Alwaneare parameter of Machines Alwaneare parameter of Machines Alwaneare parameter of Machines Alwaneare parameter parameter of Machines Alwaneare parameter para of commanity Noise Community Noise Com	Anivastance for and f	 Marstaver aplane and control in data must control in data must control in data must control in data must control in a difficultion of data must control in a difficultient of a difficultient of data must control in a difficultient of a difficultient of a difficultient of data must control in a difficultient of a difficultient of	Wastewater Dialicii (WWD)	All project activities shall comply with the Administrative Rules of the Department of Health,
If you have any questions, plases contact the WWB at (2003) 586-4704. Clean Ai Branch (CAB) Chan Ai Branch (CAB) Control of Frightive Dust Control of Frightive Dust There is a significant potential for the provintity to existing contact the NULA of at (800) 586-4701. Sincerely. There is a significant potential for the provintity to existing contact that a beat control management, thereby exact relations and construction activities and and construction activities is warrance. Construction activities must comply with the provisions of Hawaii Administrative Rules. Construction activities is warrance. Construction activities is warrance. Construction activities is warrance. Construction activities of evelopeand with the provisions of Hawaii Administrative Rules. Construction activities is warrance. Construction activities of evelopeant and construction activities is warrance. Construction activities of evelopeant and construction. The contractor activities of evelopeant and construction. Construction activities of evelopeant and construction. Construction activities of evelopeant and construction activities. Construction activities of evelopeant and construction. Construction activities of evelopeant and construction. Construction activities of evelopeant and construction. </td <td> If you have any questions, please contract the WRB at (800) 586-4204. If you have any questions, please contract the WRB at (800) 586-4204. If you have any questions, please contract the WRB at (800) 586-4204. If you have any questions, please contract the WRD at (800) 586-4204. If you have any questions, please contract the WRD at (800) 586-4204. If you have any questions, please contract the WRD at (800) 586-4204. If you have any questions, please contract the WRD at (800) 586-4204. If you have any questions, please contract the WRD at (800) 586-4204. If you have any questions, please and major potential data stage please and major pleases of construction activities that have advected planes and dresses at lactivities that a data stage please. If and the generating contraction activities that model. If and the generating contraction, focusting the amount of (801) generating planes. If and the generating contraction activities that model. If and the generating contraction activities that are not the last inpact. If and the generating contraction activities that are not the last inpact. If and the generating contraction activities that are not the last inpact. If and the generating contraction activities that are not the last inpact. If and the generating contraction activities that are not the last inpact. If and the generating contraction activities that are not the last inpact. If and the generating contraction activities that are not the last inpact. If and the generating contraction actraction activities that are not the last inpact. <l< td=""><td> If you have any quastiona, please contact the WRDs at (805) 386-4204. If you have any quastiona, please contact the WRDs at (805) 386-4204. If you have any quastiona, please contact the WRDs at (805) 386-4204. If you have any quastiona, please contact the WRDs at (805) 386-4204. If you have any quastiona, please contact the WRDs at (805) 386-4204. If you have any quastiona, please contact the WRDs at (805) 386-4204. If you have any quastiona, please contact the WRDs at (805) 386-4204. If you have any quastiona, please at (805) 486-4204. If you have any quastiona during all phases of construction transitient the following. If you have any during provide adquate data control measures during at phases of construction transitient the provision of Have and dataset and and dataset and and dataset and and dataset and dataset and dataset and and dataset and da</td><td>All wastewater plans must conform to applicable provisions of the Department of Health's Administrative Rules, Chapter 11-62, "Wastewater Systems." We reserve the right to review the detailed wastewater plans for conformance to applicable rules.</td><td>Chapter 11-46, on "Community Noise Control", and Chapter 11-39, on "Air Conditioning and Ventilation."</td></l<></td>	 If you have any questions, please contract the WRB at (800) 586-4204. If you have any questions, please contract the WRB at (800) 586-4204. If you have any questions, please contract the WRB at (800) 586-4204. If you have any questions, please contract the WRD at (800) 586-4204. If you have any questions, please contract the WRD at (800) 586-4204. If you have any questions, please contract the WRD at (800) 586-4204. If you have any questions, please contract the WRD at (800) 586-4204. If you have any questions, please contract the WRD at (800) 586-4204. If you have any questions, please and major potential data stage please and major pleases of construction activities that have advected planes and dresses at lactivities that a data stage please. If and the generating contraction activities that model. If and the generating contraction, focusting the amount of (801) generating planes. If and the generating contraction activities that model. If and the generating contraction activities that are not the last inpact. If and the generating contraction activities that are not the last inpact. If and the generating contraction activities that are not the last inpact. If and the generating contraction activities that are not the last inpact. If and the generating contraction activities that are not the last inpact. If and the generating contraction activities that are not the last inpact. If and the generating contraction activities that are not the last inpact. If and the generating contraction actraction activities that are not the last inpact. <l< td=""><td> If you have any quastiona, please contact the WRDs at (805) 386-4204. If you have any quastiona, please contact the WRDs at (805) 386-4204. If you have any quastiona, please contact the WRDs at (805) 386-4204. If you have any quastiona, please contact the WRDs at (805) 386-4204. If you have any quastiona, please contact the WRDs at (805) 386-4204. If you have any quastiona, please contact the WRDs at (805) 386-4204. If you have any quastiona, please contact the WRDs at (805) 386-4204. If you have any quastiona, please at (805) 486-4204. If you have any quastiona during all phases of construction transitient the following. If you have any during provide adquate data control measures during at phases of construction transitient the provision of Have and dataset and and dataset and and dataset and and dataset and dataset and dataset and and dataset and da</td><td>All wastewater plans must conform to applicable provisions of the Department of Health's Administrative Rules, Chapter 11-62, "Wastewater Systems." We reserve the right to review the detailed wastewater plans for conformance to applicable rules.</td><td>Chapter 11-46, on "Community Noise Control", and Chapter 11-39, on "Air Conditioning and Ventilation."</td></l<>	 If you have any quastiona, please contact the WRDs at (805) 386-4204. If you have any quastiona, please contact the WRDs at (805) 386-4204. If you have any quastiona, please contact the WRDs at (805) 386-4204. If you have any quastiona, please contact the WRDs at (805) 386-4204. If you have any quastiona, please contact the WRDs at (805) 386-4204. If you have any quastiona, please contact the WRDs at (805) 386-4204. If you have any quastiona, please contact the WRDs at (805) 386-4204. If you have any quastiona, please at (805) 486-4204. If you have any quastiona during all phases of construction transitient the following. If you have any during provide adquate data control measures during at phases of construction transitient the provision of Have and dataset and and dataset and and dataset and and dataset and dataset and dataset and and dataset and da	All wastewater plans must conform to applicable provisions of the Department of Health's Administrative Rules, Chapter 11-62, "Wastewater Systems." We reserve the right to review the detailed wastewater plans for conformance to applicable rules.	Chapter 11-46, on "Community Noise Control", and Chapter 11-39, on "Air Conditioning and Ventilation."
 Typote nave any questions, prease contact the WMI at (400) 500-4294. Control of Prigrine Dust Control progradi construction activities in lococuting provinding to existing control measures during all phases of construction activities in success the activities in account on measurement and construction activities in success and activities in account in measures include, but are not limiterative Rules, Distro Providen and Construction activities (part and activities in activit	 There any questions, prease contact the WWB at (2003) S60-4204. Comol of Pragitive Dast Comol of Dast Comol of Pragitive Dast Comol of Pragitive Dast Comol of Pragitive Dast Comol of Dast	 The mark and y quastions, prease contact the WM at (100) 360-4204. Comred of Fright we Date: There is a subficient your contact the provision and relates and addresses and major to construction and relates and addresses and		If you have any questions, please contact the NRIAQ at (808) 586-4701.
Clean Air Branch (CAB) Current of Program Dust Current of Program Dust Control of Program Dust Control of Program Dust Control of Program Dust Control of Program Dust There is a significant potential for flagible dust emissions during all phases of construction activities will coording frage problems. It is recommended that east control management path at east control east control of at east management path at east control management path at east control of at east management path at east control management path at east control of at east management path at east control management path at east control east control of at east management path at east management path at east control of at east management path at east management	Clean Air Barnch (CAB) Conrol of Frighter Dara Conrol of Frighter Dara There is a significant potential for flightor data emissions during all phases of construction operating potential dats potential data problems. This recommendance that a dust problems that have a potential part a dust problems that have a potential part a dust problems. This recommended that a dust problems that have a potential data potential data potential data potential on the provisions of Havai Administrative Rules, a manuered during all phases of construction activities is warranted. Control on anyoine masure during all phases of construction activities is warranted. Control masure during all phases of construction activities is warranted. Construction activities is warranted. Construction activities is warranted. Construction activities is warranted. Construction activities is warranted. Construction activities is warranted. Construction activities is warranted. Construction activities is warranted. Construction activities is warranted. Construction activities is warranted. Construction activities of the downs: Part the different phases of construction. These measures tracking the amount of dust. Construction activities is warranted. Construction activities of the downs: Part the different phase of construction activities of the downs: Construction activities of the downs: Construction activities of the activities of the downs: Part the different phase of construction activities of the downs:<	Clean Air Barnch (CAD) Count of Fragine Load Count of Fragine Load Three is a significant potential for flight or date entations during all phases of construction activities is a significant potential for flight or date entations during all phases of construction activities is warranted. Three is a significant potential for flight or date entations during all phases of construction activities is warranted. Three is a significant potential for flight or date entations during all phases of construction activities is warranted. Construction activities is warranted. The ontercomon activities is warranted. The construction activities is warranted. Construction activities is warranted. The ontercomon activities is warranted. The construction activities is warranted. Construction activities is warranted. The activities is warranted. The construction activities is warranted. Construction activities is warranted. The activities is warranted. The construction activities is warranted. Construction activities is warranted. The activities and activities externation activities is warranted. The activities and activities actinactination activities activities actinactinati	If you have any questions, please contact the WWB at (808) 586-4294.	Sincerely
 Control of Fligtive Dast Control of Fligtive Dast There is a significant potential for fligtive dust encodening potential to provinty to existing residences and major throoroghtmes, threely exacelering potential data provints that have a control masterner plan to developed with deartific stat activities that activities th	 Cannol of Fragitive Data Construction device and sensitions during all phases of construction, thereby concertain provisions of the major of the provisions of the major of the provisions of the major of the provisions of the pro	 Control of Frigtive Dist Control of Frigtive Dist There is a significant potential for flighted that emissions during all phases of construction activities in the revolution and potential for flighted that a dust problems. It is recommended that a dust protonomental addresses all activities that have a potential potential optication activities is warranted. There is a significant protein for flighted that potential for flighted that a dust problems. It is recommended that a dust protonomental addresses all activities that have a potential potential optication activities is warranted. Construction activities is warranted. Proton approximation of advantation of adva	<u>Clean Air Branch (CAB)</u>	And F Above Strathers
There is a significant potential for fightive dust musisions during all phases of construction activities will occur in proximity to existing reaidences and major throughdress, through every interversion activities and access and major throughdress, through every interversion activities and access and major throughdress, through every interversion activities is warranted. OVDE IF: HARKLOATH-LUM, MANAGER Environmental Planning Office: Environmental Planning Office: Through every interversion activities is warranted. The contractor manipulation is a control measures during all phases of construction activities is warranted. WMB CAB Construction activities is activities is and activities that a dust. WMB CAB Distribution activities and activities is and activities that a dust. WMB CAB Plan the different phases of construction, frase measures include, but are not limited to, the following: WMB CAB • Plan the different phases of construction, frase measures include, but are not limited to, the following: WMB CAB • Plan the different phases of construction, fousing on minimizing the anount of dust-generating equinate that are sorble east impact; Provide and activities, sentralizing on-site vehicular traffic routes, and locating potential dust-generating equinate that are as including slopes, starting from the initial grading phase; I. Infinitize dust from shoulders and access roude.	There is a significant potential for fugitive dust emissions during all phases of construction, proposed construction activities will occur in proving reviewers and major thromogeneous fugitive dust. This recommended that a day potential to generate fugitive dust. The recommendent that have a potential to generate fugitive dust. The recommendent that have a potential to generate fugitive dust. The recommendent that have a potential to generate fugitive dust. The recommendent that have construction activities must comply with the provisions of Hawaii Administrative Rules, 311-60.1-33 on Fugitive Dust. The contract comply with the provisions of Hawaii Administrative Rules, 311-60.1-33 on Fugitive Dust. The contract comply with the provisions of Hawaii Administrative Rules, 311-60.1-33 on Fugitive Dust. The contract comply with the provisions of Hawaii Administrative Rules, 311-60.1-33 on Fugitive Dust. The contract comply with the provisions of Hawaii Administrative Rules, a Plan the different phases of construction, costing potential dust-generating equipment in areas of the least inpact. The fugition fust-generating equipment in areas of the least inpact. The fugition fust-generating equipment in areas of the least inpact. Thirafig grading phase.	 There is a significant potential for figure dust emissions during all phases of construction. There is a significant potential for figure dust emissions during all phases of construction solutions where a during all phases of construction activities must comply with the provisions of Hawaii Administrative Rules, 191-601-133 on Fugritive Dust. Carsu Complexent and construction activities is writtende. Carsu Complexent activities activities	Control of Fugitive Dust	and the second second second
Proposed construction antivities will construct in antivities will construct in antivities and activities and major c.WB Decontrol management plate determines and addresses all activities is warranded. c. WB Construction activities is warranded. c. AB Construction activities is warranded. c. AB Sill-soll-1-30 on Fugitive Dust. c. AB The contractor should provide adequate measures to control dust from the road areas and during the various phases of construction, activities, put are not limited to, the following: measures during all activities, and activities, and activities, put are not limited to, the following: a. Plan the different phases of construction, frace areas including the amount of dats generating materials and activities, centralizing on-site volticular traffic routes, and locating potential dats-generating explorement in areas of the last impact; c. Landscape and provide rapid covering of bare areas, including slopes, starting from the initial grading phase. d. Minimize dust from shoulders and access roads.	Proposed construction activities will activities that have a control management plane which iterative set and major phoeses of activities that have a control management plane developed which iterative data its accommended that a dust control measures during all phases of developed which iterative data accoss all activities that have a control measures during all phases of developed which iterative data accoss and activities that have a control measures during all phases of developed which iterative data accoss and activities that have a control measures during all phases of developed which iterative kules, gil-60.1-33 on Fugitive Dust. CWB CWB Construction activities is warranted. CAB WMB The contractor should provide adequate measures include, but are not limited to, the following: CAB WMB The contractor should provide adequate measures include, but are not limited to, the following: EAB WAB a. Plan the different phases of construction, focusing on minimizing the amount of thest generating equipment in areas of the least impact; Impact CAB b. Provide an adequate water source at the site prior to start-up of construction activities, entrating from the initial grading phase. Impact Impact c. Indicate dust from shoulders and accoss roads; Impact Impact Impact	Proposed construction activities will ocur in promoted of that the revealung products. The recommended that the revealure studing all the revealures during all the revealures during all prosters fugitive duat. c CWB wwB wrds Proceeding the production activities will court in management plans to control management plan be developed which itentifies and addresses all activities that have a potential to generate flat control measures during all presses of the revealure dust control measures and activities is warranded. c CWB wwB wrds Construction activities must comply with the provisions of Hawaii Administrative Rules. mwB wrds wwB wrds Sill-60.1-33 on Fugitive Dust. File environment of that from the road areas and during the values power of the road areas and during the values provide adequate measures include, but are not innimizing the anount of the following: wwB wrds a. Plant the different planses of construction activities; and advises and during a mount of the following: c wwB wrds b. Provide an adequate water source at the site prior to start up of construction activities; and indices and access to activities; and will be access to activities; and access to activities; and will be access to activities; and acceses to activities; and access to activities; and access t	There is a significant potential for fugitive dust emissions during all phases of construction.	JUNE F. HAKKUGAN-LUM, MANAGEK Environmental Planning Office
 control management plan be developed which identifies and addresses all activities that have a potential to generate allogive dust. Implementation of adequate dust control measures during all phases of development and construction activities is warranted. Construction activities must comply with the provisions of Hawaii Administrative Rules, \$11-60.1-33 on Trogitive Dust. Constructor should provide adequate measures to control dust from the road areas and during the various phases of construction. These measures include, but are not limited to, the following: a. Plan the different phases of construction, focusing on minimizing the amount of dust: b. Provide an adequate water source at the site prior to start-up of construction activities, centralizing on-site vehicular taffic routes, and pocading phase; c. Landscape and provide rapid covering of bare areas, including slopes, starting from the initial grading phase; d. Minimize dust from shoulders and access roads; 	orter of construction activities is warranted. WWB construction activities is warranted. CAB phases of development and construction activities is warranted. CAB Construction activities is warranted. Construction activities is warranted. S11-60.1-33 on Pugitive Dust. Construction activities is warranted. The contractor should provide adequate measures to control dust from the road areas and during the various phases of construction. These measures to control dust from the road areas and during the various phases of construction. These measures include, but are not limited to, the following: CAB a. Plan the different phases of construction. These measures include, but are not limited to, the following: CAB b. Provide and expression activities; Caster phase of construction, focusing on minimizing the amount of dust-generating equipment in areas of the least inpact; CAB b. Provide an adequate water source at the site prior to start-up of construction activities; Caster prior to start-up of construction activities; c. Indigrading phase; Caster prior to start-up of construction activities; Caster prior to start-up of construction activities; d. Minnize dust from shoulders and access roads; Minnize dust from shoulders and access roads; Caster prior to start-up of construction activities;	 control management planes and addresses all activities that have a potential to generate fuggive dust. Indemtides and addresses all activities that have a potential to generate fuggive dust. The provisions of Hawaii Administrative Rules, \$11-60.1-33 on Fugitive Dust. Construction activities mark comply with the provisions of Hawaii Administrative Rules, \$11-60.1-33 on Fugitive Dust. Construction activities mark comply with the provisions of Hawaii Administrative Rules, \$11-60.1-33 on Fugitive Dust. a. Plan the different phases of construction, These measures include, but are not limited to, the following: the various phases of construction. These measures include, but are not limited to the following: a. Plan the different phases of construction, focusing on minimizing the amount of dust form the routes, and control dust form the routes, and control dust form the last impact; b. Provide an adequate measures including slopes, starting from the initial grading phase; d. Minimize dust from shoulders and access roads; 	Proposed construction activities will occur in proximity to existing residences and major thoroughfares, thereby exacerbating potential dust problems. It is recommended that a dust	c: CWB
 Provide an adequate measures include, but are not limited to, the following: a. Plan the different phases of construction. These measures include, but are not limited to, the following: a. Plan the different phases of construction. These measures include, but are not limited to. b. Provide an adequate water source at the site prior to start-up of construction activities; c. Landscape and provide rapid covering of bare areas, including slopes, starting from the initial grading phase; d. Minimize dust from shoulders and access roads; 	 A mathematical material and construction activities must comply with the provisions of Hawaii Administrative Rules, \$31-60.1-33 on Fugitive Dust. The contractor should provide adequate measures to control dust from the road areas and during the various phases of construction. These measures include, but are not limited to, the following: a. Plant he different phases of construction, four should are not limited to, the following: a. Plant he different phases of construction, four should are not limited to, the following: b. Plant he different phases of construction, four should are not unit of dust-generating equipment in areas of the least impact. b. Provide an adequate water source at the site prior to start-up of construction activities; c. Landscape and provide rapid covering of hare areas, including slopes, starting from the initial grading phase. d. Minimize dust from shoulders and access roads; 	 Provide an advantage of the provisions of Hawaii Administrative Rules, Solution activities must comply with the provisions of Hawaii Administrative Rules, Solution adoptate measures to control dust from the road areas and during the various phases of construction. These measures include, but are not limited to, the following: a. Plant he different phases of construction, focusing on minimizing the amount of dust generating materials and activity, centralizing on-site valuate randics. b. Provide an adequate water source at the site prior to start-up of construction. c. Landscape and provide rapid covering of base, including slopes, starting from the initial grading phase; d. Minimize dust from shoulders and access roads; 	control management plan be developed which identifies and addresses all activities that have a potential to generate fugitive dust. Implementation of adequate dust control measures during all these of development and construction construction construction.	WWB CAB NRIAO
 Construction activities must comply with the provisions of Hawaii Administrative Rules, §11-60.1-33 on Fugitive Dust. The contractor should provide adequate measures to control dust from the road areas and during the various phases of construction, These measures include, but are not limited to, the following: a. Plan the different phases of construction, focusing on minimizing the amount of dust-generating materials and activities, centralizing on-site vehicular traffic routes, and locating potential dust-generating equipment in areas of the least impact; b. Provide an adequate water source at the site prior to start-up of construction activities; c. Landscape and provide rapid covering of bare areas, including slopes, starting from the initial grading phase; d. Minimize dust from shoulders and access roads; 	Construction activities must comply with the provisions of Hawaii Administrative Rules, §11-60.1-33 on Pogitive Dust. The contractor should provide adequate measures to control dust from the road areas and during the various phases of construction. These measures include, but are not limited to, the following: a. Plan the different phases of construction, focusing on minimizing the amount of dust- generating materials and activities, centralizing on-site vehicular traffic routes, and locating potential dust-generating equipment in areas of the least impact; b. Provide an adequate water source at the site prior to start-up of construction activities; c. Landscape and provide rapid covering of bare areas, including slopes, starting from the initial grading phase; d. Minimize dust from shoulders and access roads;	 Construction activities must comply with the provisions of Hawaii Administrative Rules, \$11-60.1-33 on Fugitive Dust. The contractor should provide adequate measures to control dust from the road areas and during the various phases of construction. These measures include, but are not limited to, the following: a. Plante different phases of construction on shire vehicular traffic routes, and locating potential dust-generating equipment in areas of the last impact, but are not initiated to the following: b. Provide an adequate water source at the site prior to start-up of construction activities; c. Landscape and provide rapid covering of bare areas, including slopes, starting from the initial grading phase. d. Minimize dust from shoulders and access roads; 	איניצאיי אין הבאבוסאוויביוו שוות בסוואוותביוסוו שמתאוובא וא אשוושוויבח.	
 The contractor should provide adequate measures to control dust from the road areas and during the various phases of construction. These measures include, but are not limited to, the following: a. Plan the different phases of construction, focusing on minimizing the amount of dust-generating materials and activities, centralizing on-site vehicular traffic routes, and locating potential dust-generating equipment in areas of the least impact; b. Provide an adequate water source at the site prior to start-up of construction activities; c. Landscape and provide rapid covering of bare areas, including slopes, starting from the initial grading phase; d. Minimize dust from shoulders and access roads; 	 The contractor should provide adequate measures to control dust from the road areas and during the various phases of construction. These measures include, but are not limited to, the following: a. Plan the different phases of construction, focusing on minimizing the amount of dust-generating materials and activities, centralizing on-site valicular traffic routes, and locating potential dust-generating equipment in areas of the least impact; b. Provide an adequate water source at the site prior to start-up of construction activities; c. Landscape and provide rapid covering of bare areas, including slopes, starting from the initial grading phase; d. Minimize dust from shoulders and access roads; 	 The contractor should provide adequate measures to control dust from the road areas and during the various phases of construction. These measures include, but are not limited to, the following: a. Plan the different phases of construction, focusing on minimizing the amount of dust-generating materials and activities, centralizing on-site vahicular traffic routes, and locating potential dust-generating equipment in areas of the least impact; b. Provide an adequate water source at the site prior to start-up of construction activities; c. Landscape and provide rapid covering of bare areas, including slopes, starting from the initial grading phase; d. Minimize dust from shoulders and access roads; 	Construction activities must comply with the provisions of Hawaii Administrative Rules, §11-60.1-33 on Fugitive Dust.	
 a. Plan the different phases of construction, focusing on minimizing the amount of dust-generating materials and activities, centralizing on-site vehicular traffic routes, and locating potential dust-generating equipment in areas of the least impact; b. Provide an adequate water source at the site prior to start-up of construction activities; c. Landscape and provide rapid covering of bare areas, including slopes, starting from the initial grading phase; d. Minimize dust from shoulders and access roads; 	 a. Plan the different phases of construction, focusing on minimizing the amount of dust-generating materials and activities, centralizing on-site vehicular traffic routes, and locating potential dust-generating equipment in areas of the least impact; b. Provide an adequate water source at the site prior to start-up of construction activities; c. Landscape and provide rapid covering of bare areas, including slopes, starting from the initial grading phase; d. Minimize dust from shoulders and access roads; 	 a. Plan the different phases of construction, focusing on minimizing the amount of dust-generating materials and activities, centralizing on-site valualar traffic routes, and locating potential dust-generating equipment in areas of the least impact; b. Provide an adequate water source at the site prior to start-up of construction activities; c. Landscape and provide rapid covering of bare areas, including slopes, starting from the initial grading phase; d. Minimize dust from shoulders and access roads; 	The contractor should provide adequate measures to control dust from the road areas and during the various phases of construction. These measures include, but are not limited to, the following:	
locating potential dust-generating equipment in areas of the least impact; b. Provide an adequate water source at the site prior to start-up of construction activities; c. Landscape and provide rapid covering of bare areas, including slopes, starting from the initial grading phase; d. Minimize dust from shoulders and access roads;	locating potential dust-generating equipment in areas of the least impact; b. Provide an adequate water source at the site prior to start-up of construction activities; c. Landscape and provide rapid covering of bare areas, including slopes, starting from the initial grading phase; d. Minimize dust from shoulders and access roads;	locating potential dust-generating equipment in areas of the least impact; b. Provide an adequate water source at the site prior to start-up of construction activities; c. Landscape and provide rapid covering of bare areas, including slopes, starting from the initial grading phase; d. Minimize dust from shoulders and access roads;	a. Plan the different phases of construction, focusing on minimizing the amount of dust- generating materials and activities, centralizing on-site vehicular traffic routes, and	
 b. Provide an adequate water source at the site prior to start-up of construction activities; c. Landscape and provide rapid covering of bare areas, including slopes, starting from the initial grading phase; d. Minimize dust from shoulders and access roads; 	 b. Provide an adequate water source at the site prior to start-up of construction activities; c. Landscape and provide rapid covering of bare areas, including slopes, starting from the initial grading phase; d. Minimize dust from shoulders and access roads; 	 b. Provide an adequate water source at the site prior to start-up of construction activities; c. Landscape and provide rapid covering of bare areas, including slopes, starting from the initial grading phase; d. Minimize dust from shoulders and access roads; 	locating potential dust-generating equipment in areas of the least impact;	
 c. Landscape and provide rapid covering of bare areas, including slopes, starting from the initial grading phase; d. Minimize dust from shoulders and access roads; 	c. Landscape and provide rapid covering of bare areas, including slopes, starting from the initial grading phase;d. Minimize dust from shoulders and access roads;	c. Landscape and provide rapid covering of bare areas, including slopes, starting from the initial grading phase;d. Minimize dust from shoulders and access roads;	b. Provide an adequate water source at the site prior to start-up of construction activities;	
d. Minimize dust from shoulders and access roads;	d. Minimize dust from shoulders and access roads;	d. Minimize dust from shoulders and access roads;	c. Landscape and provide rapid covering of bare areas, including slopes, starting from the initial grading phase;	
			d. Minimize dust from shoulders and access roads;	

- -
- PALEHUA COMMUNITY ASSOCIATION	February 6, 2003. Makakilo, Harania Srreat Mar. Vincent Shigekuni, Principal FFR u PRR Hawaii FFR u 1001 Bishop Street Prov. Pacific Towler, Suite 650 Prov. Honolulu, HI 96813-3429 Prov.	SUBJECT: UNIVERSITY OF HAWAII – WEST OAHU DRAFT ENVIRONMENTMENTAL ASSESSMENTJENVIRONMENTAL IMPACT STATEMENT NOTICE OF PREPARATION	Dear Mr. Shigekuni:	In response to your December 23, 2002 letter asking for our comments as to whether the proposed University of Hawaii (UH) – West Oahu campus may have an impact on any of	our existing or proposed projects, policies or programs, we submit the following:	 Your letter was reviewed during our monthly Falehua Community Association meeting on January 23, 2003. The board supports the development of the UH West Oahu campus on the 500-acre parcel mention in your letter. 	2. We are concerned about roadway infrastructure and adequate freeway access, including proper clover leafs, for traffic flow. Specific projects that would help alleviate bottlenecks and affect the flow of traffic on Makakilo Drive include: a. New westbound (Waianae) Makakilo on-ramp. Will help keep traffic	from backing up from up on Makakilo Drive and unnecessary traffic on Farrington Highway in Kapolei. b. A new North-South Road H-1 interchange, with clover leafs. Will reduce the amount of traffic that needs to use the Makakilo Drive and	Farrington Highway roadways through Kapolei. c. Completion of the North-South Road. This again will reduce traffic allow Fort Barrette Road and the Makakilo Drive and Farrington Fishwav interchance	d. Completion of Makakilo Drive, which currently ends at Royal Ridge, to the North-South Road H-1 interchange. This will allow those who live in Makakilo and will work or to to school at UH – West Oahu an	alternative access to the campus. It will reduce the heavy traffic flow at Makakilo Drive H-1 on-ramp (East) to Honolulu and/or to use Farrineton Firehway Fast from Matakilo Drive in Kanolei.	e. Widening Farrington Highway from the Kapolei golf course to Fort Weaver road will also assist in traffic reduction that will bottleneck at Makakilo Drive and Farrington Highway.	f. Repair and maintenance of needs to be upgraded to handle the additional traffic.
January 27, 2005	Ms. June F. Harrigan-Lum, Manager Environmental Planning Office State of Hawaii Department of Health P. O. Box 3378 Honolutu, Hawaii 96801-3378	SUBJECT: UNIVERSITY OF HAWAII WEST OAHU DRAFT ENVIRONMENTAL ASSESSMENT/ENVIRONMENTAL IMPACT STATEMENT PREPARATION NOTICE (EISPN)	Dear Ms. Harrigan-Lum:	Thank you for your letter dated January 27, 2003 (your reference number 03-007/epo). We have reviewed your letter and offer the following responses:	 The Army Corps of Engineers has been contacted to identify whether a Federal permit (including a Department of Army permit) is required for this project. 	2. The EISNOP will state the project will require a NPDES permit.	 The Draft EIS will note that all wastewater plans must conform to applicable provisions of the Department of Health's Administrative Rules, Chapter 11-62. "Wastewater Systems." The Draft EIS will also state that the Department of Health reserves the right to review the detailed wastewater plans for conformance to applicable rules. 	 The Draft EIS will address the potential impact of fugitive dust emissions during construction. Measures to mitigate the impact of fugitive dust will also be addressed in the EIS. 	 The Draft EIS will note that all wastewater plans must comply with the Administrative Rules of the Department of Health, Chapter 11-46, on "Community Noise Control," and Chapter 11-39, on "Air Conditioning and Ventilation." 	Again, thank you for your comments and your participation in the EIS process. If you have any questions, please do not hesitate to contact me.	Sincerely.	Vuizz& R. Buight	Vincent Shigekuni Principal o.yOB13\1322.05\Environmental Assessment-EISPNNPre-consultation Response Lins\BL-14 DOH.doc

LAND PLANNING LANDSCAPE ARCHITECTURE ENVIRONAENTAL STUDIES PBR

WM. FRANK BRANDT, FASLA CHAIRMAN

THOMAS S. WITTEN, ASLA PRESIDENT R. STAN DUNCAN, ASLA EXECUTIVE VICE-PRESIDENT RUSSELLY, J. CHUNG, ASLA EXECUTIVE VICE-PRESIDENT VINCENT SHIGEKUNN PRINCIPAL JAMES LEONARD, AICP HILLO OFFICE

GRANT MURAKAMI, AICP JENNIK ASSOCATE TOM SCHWELL, AICP ASSOCIATE RAYMOND T, HIOA, ASLA ASSOCIATE ASSOCIATE KEVIN NISHIKAWA, ASLA ASSOCIATE

HawoLILU OFFICE (2012) Battor Syster ABST Dewes, Surfrecto ABST Dewes, Surfrecto Howemen, Howart 958(13-384 Th214 (2013) 251-301 Factor (2013) 252-402 Factor 9724 data

HILO OFFICE IOI ALWINGSTREAT BLID.LAURON COFFICE STRFE 310 HILO, FAWAY: 90730-4202 FLIC (980) 501-4329 FAX (803) 951-4329 FAX (803) 951-4329 FAX (803) 951-4329 FAX (803) 951-4329

WAILURU OFFICE 2123 KAOHU STREET 2123 KAOHU STREET WALLIKU, HAWN 196793-2204 The (608) 242-2878 FACE (688) 242-2902 EMAIL JOHTHANI @LIWE, DCT

•

higekuni	PBR	January 27, 2005
5, 2003	H A W A 1 I LAND PLANTING LAND PLANTING LAND CARE AND THE THE	Mr. Mike Golojuch, President Palehua Community Association 92-1091 Palahia Street Makakilo, Hawaii 96707
g. Additions to the current circulator bus routes in Makakilo may be needed and/or modification to the circulator bus routes. e are also concerned that the additional influx of people living and working in akakilo will tax already overcrowded Makakilo elementary schools (Mauka	WA, FRANK BRANDT, FASLA Charmana Thomas S. Witten, ASLA President R. Standora, ASLA Eventuar	SUBJECT: UNIVERSITY OF HAWAII – WEST OAHU DRAFT ENVIRONMENTAL ASSESSMENT/ENVIRONMENTAL IMPACT STATEMENT PREPARATION NOTICE (EISPN) Dear Mr. Golojuch:
mi and Makakilo Elementary) and Kapolei Intermediate and Kapolei High hool. For Makakilo there will be a need for another elementary school. Land d construction costs are problems. review of water and sewer infrastructure needs to be accomplished. Also how n revoled water be used as well as other alternative energy resources. e.z.	LALCUTE VIET ALEVIET A	Thank you for your letter dated February 6, 2003. We have reviewed your letter and offer the following responses: 1. The University of Hawaii appreciates the Palehua Community Association's ennoter of the UTH Ware Ochu commun. It is communities encl as shore that ITH
dar, hydrogen? us know if you need any additional information. If you wish to attend our etings, we usually meet on the fourth Wednesday of the month at our office, 92- thia Street, 7:00 p.m. You can always contact either our Covenants Specialist as at 672-5097 or our Property Manager Wayne Kiritio at 485-0085 to confirm	HILD OFFICE GRANT MURAXANI, AICP SENDE ASSOCIATE TOM SCINEL, AICP ASSOCIATE ASSOCIATE PLANNIN THIA ASI A	 West Oahu was originally intended to serve. We appreciate your association's ideas for alleviating future traffic. Your letter has been transmitted to our traffic consultant. These potential solutions will be addressed in the Traffic Impact Analysis Report (TIAR) which will be appended to the Draft EIS.
idance and/or the date of the meeting. Sincerely,	KEVIN NISHIKAWA, ASLA Associate Associate	 We understand your comments regarding the overcrowding of public schools in the area. Please note that we received similar concerns from the Department of Education (DOE). In response, current plans for the UH West Oahu Campus includes one 12 acre elementary school site.
MIKE GOLOJUCH	HowoLutu OFFICE (100) Bastor STRART ASB Thowas, Structor HowaLut, Ukowa, Vison 503 The (1880) 511-551	 The Draft EIS will include water and sewer master plans. The use of reclaimed water for irrigation and alternative energy sources will also be discussed in the Draft EIS.
	FAC (00) 523-1402 EAGL Syndining Optinwali coun FILO ANTROY SWRST 101 ANTROY SWRST HLO LANTROY SWRST FILO LANTROY SWRST FILO LANTROY SWRST FILO LANTROY SWRST FILO FILO FILO FILO HAU LANTROY SWRST FAMIL PRINC MANAGE	Again, thank you for your comments and your participation in the EIS process. If you have any questions, please do not hesitate to contact me. Sincerely. PBR HAWAII
	WALLING OPPER 2125 KAONI STREAT WALLING ANN 19572254 TEL (2002) 22.252 FACE (2002) 22.222 FACE (2002) 22.222 FACE (2002) 22.2222 FACE (2002) 22.2222 FACE (2002) 22.2222	Vurzes R. Sungle

O:UOB13/1322.05/Environmental Assessment-EISPNPre-consultation Response LtrsIBL-15 Patelrua Comm Assoc.doc

·

January 27, 2005	Mr. Rodney K. Haraga, Director State of Hawaii	Department of Transportation 869 Punchhowl Street Homoluin, Hamoit 06813.5007		Attn: Mr. Ronald Tsuzuki	SUBJECT: UNIVERSITY OF HAWAII – WEST OAHU DRAFT ENVIPONMENTAT, ASSESSMENTENVIPONMENTAL	IMPACT STATEMENT PREPARATION NOTICE (EISPN)	Dear Mr. Haraga:	Thank you for your letter dated January 17, 2003 (your reference number HWY-PS 2.9330). We have reviewed your letter and will include the information you requested on traffic	impacts in the Draft EIS.	Again, thank you for your comments and your participation in the EIS process. If you have any questions, please do not hesitate to contact me.	Sincerely,	PBR HAWAII	Viize R. Bright	Vincent Shigekuni Principal	O:VOB13N1322.05\Environmental Assessment-EISPN1Pre-consultation Response Lurs\B116 DOT.doc			
PBR	H A W A L I A LAND FLAND FLANKS LAND FLAND FLANKS	WM. FRANK BRANDT, FASLA CHAIRMAN	THOMAS S. WITTEN, ASLA PRESIDENT	R. STAN DUNCAN, ASLA Executive Vice-President	RUSSELL Y.J. CHUNG, ASLA EXECUTIVE VICE-PRESIDENT	VINCENT SHIGEKUNI PRINCIPAL	JAMES LEONARD, AJCP PRINCIPAL HILO DEFIFE	GRANT MURAKAMI, AICP Senor Associate	TOM SCHNELL, AICP	ASSOCIATE RAYMOND T. HIGA, ASLA ASSOCIATE	KEVIN NISHIKAVA, ASLA Associate		Hannluku Office 1001 Bistor Street ASB Tower, Sutte 630	Honolitut, HAwri 156813-3454 Thr. (605) 521-5631 EAX: (805) 523-1402 E-MAIL: syyadhin@phfluwail.com	FILLO OFFICE	HLUG LACOON CENTRE, SINTE 310 HLUG LACOON CENTRE, SINTE 310 HLU, HAWAI' 96720-4262 Taz. (608) 951-9333 FAZ: (608) 951-9333 E-46AUL: Phylilo@lava.act		WALLING DEFICE 2125 KAQUID SFREET WALLING HAWAH 96793-2204 TE-46603 942-2378 FAX: (5018) 342-2322 EAAUL phynisui @lava.nel
RODNEY K, HARAGA DIRECTOR Ading Deput Director Ading Deput Director Ading Deput Director	DEPARTMENT ON F TRANSPORTATION FER 1 3 MARENT REFER TO: B69 PUNCHBOWL STREET HONOLULUL, HAWAII 9887-9,697	FFR 11 2003		kuni, Principal		utte 650 Acers	C1006 III	invironmental Impact Statement (EIS) Preparation Notice for University of	fawaii (UH) West Oahu	equesting our input on the EIS Preparation Notice. We have the following	ld like the EIS to address the location, timing, design, and projected traffic at I new intersections with State highways including our future Bwa North-South	he EIS also should address when and how the UTH proposes to fund these ion improvements.	County of Honolulu Ordinance 02-52 requires Ewa developers to contribute ses to nav their fair share of the cost for regional highway immovements	i by the 2010 Ewa Highway Master Plan. We have informed the UH that we are o pay any impact fees assessed to State agencies which have applied for building on the Ewa Regrin. 'However, we may need to reevaluate this offer if the UH is	g significant private development/use of the West Oahu campus.	questions, please contact Ronald Tsuzuki, Head Planning Engineer, Highways -1830.		ARAJA
LINDA LINGLE GOVERNOR				Mr. Vince Shigel	PBR Hawaii 1001 Bichon Stre	Pacific Tower, St	Dear Mr. Shinely	Subject: Er	H	Thank you for re comments:	1. We would proposed	Road. Th intersectiv	2. City and (imnact fee	proposed willing to nermits ir	proposing	If you have any ç Division, at 587-	Very truly yours,	KODNIJYK. HA

.

KODNHY/K. HARAGA Director of Transportation ×

PBRA II A W A T I MORGARE ARMENTICE	BWRRONARSTAU, STUDIES Mr. Karker Senior Environmental Scientist WM. Frank Branur, FASLA P. O. BAX 2750 P. O. BAX 2750	THOMASS. WITTEN ASLA HONOLULU, FLAWAII 96840-0001	R. STAN DIRCAN, ASLA Attn: Mr. William Lane Executive Vice-President	RUSSELT YL CHUNGLASLA SUBJECT: UNIVERSITY OF HAWAII – WEST OAHU EXECUTIVE VICE PRESIDENT DR AFT ENVIRONMENTAL ASSESSMENT/FINVIRONME	VINCENT SHOEKUNI PRINCIPAL PRINCIPAL	James Leoward, AICP Dear Mr. Tomite: Prevented. Hito Office	Thank you for your letter dated February 11, 2003. We have reviewed your let seave Associate University of Hawaii greatly appreciates HECO's interest in participating in the j this project. We concur that there are many enterpy management and co	Tom Schwell, AICP opportunities that can be realized by coordinating with HECO in the planning an RAYMOND T. HIGA, ASLA this project.	KEVIR NISHIKAWA, ASLA Again. thank you for your comments and your participation in the EIS process. ASSOCIATE any questions, please do not hesitate to contact me.	- Succely,	Descutationer (20) Destewsmer Assures, Struckson Hereature, Ideaver (20) Self : JAAWAII	The (1805) 523-5631 Free (1805) 523-1402 BARREL Systelling portuneiticon Vince R. Burghen	Itro Orres: Vincent Sluigekuni MucLeon Arene Stream: MucLeon Stream: In: MucLeon Stream: MucLeon Stream: MucLeon Stream: MucLeon Stream: Principal Four perhoding and Construction Four perhoding and Construction MucLeon Stream: Principal Principal Construction Const	WALLING OFFICE 2123 KAOHLSFREAT WALLING 12622204 The::f0013.0423232204
GEN-6 (EIS)								: on the December 2002 Draft EA / EIS e University of Hawai'i.	lanning of this project. There are many es that can be realized through a his project is Mr. William Lane, Account is Department (543-4618).		david C	ta vironmental Scientist	· · ·	
	February 11, 200		Mr. Vincent Shigekuni, Principal	PBR Hawaii 650 PacificTower	1001 Bishop Street Honolulu, HI 96813-3429	Dear Mr. Shigekuni:	Re: University of Hawai'l - West O'ahu Campus East Kapolei, Oahu TMK (1) 9-1-016: por. 108	Thank you for the opportunity to review and comment Notice of Preparation of the West O'ahu campus of the	HECO is very much interested in participating in the pl energy management and conservation opportuniti coordinated planning effort. Our point of contact for th Manager, Marketing Services Division, Energy Service	Sincaraly	Amonomy free for	Kirk Tamit Senior En	cc: OEQC F. Hirakami W. Lane	

Hawailan Electric Company, Inc. • PO Box 2750 • Honolulu, HI 96840-0001

WINNER OF THE EDISON AWARD FOR DISTNGUISHED INDUSTRY LEADERSHIP

31-ED - 24

N

January 27. 2005 January 27. 2005 Mr. David Rae The Estate of James Campbell 1001 Kamokila Boulevard MMr. Kapolei, Hawaii 96707	THEY AS A SUBJECT: UNIVERSITY OF HAWAII – WEST OAHU BAFT ENVIRONMENTAL ASSESSMENT/ENVIRONMENTA EARSTOR IMPACT STATEMENT PREPARATION NOTICE (EISPN) INVEASIA IMPACT STATEMENT PREPARATION NOTICE (EISPN) Dear Mr. Rae: There are assured that the Estate of James Campbell will remain a consulted party throughout the EIS process. If you have any questions, please do not hesitate to conta me.	ANAL ALP SOCATE Sincerely. H. ACP BR HA WAII HOL ASLA PBR HA WAII HOL ASLA VILLE P. Sturger M. ASLA VILLE P. Sturger Vincent Shigekuni Principal	OPSEE ONDER ONDER ONDER JUI 322.05/Environmental Assessment-EISPNDFe-consultation Response LargebL18.E1 Sameta Sameta Sameta Tastana Sameta Sameta Sameta Ommer Ommer Ommer Ommer Ommer Ommer Official Sameta Sameta Ommer Official Sameta Ommer Official Sameta Ommer Official Sameta Ommer Official Sameta Official Sameta O
H A W A I LANDELANDRACHT	THOMAS S, WITTEN. <i>PRESIDENT</i> R, STAN DUVEVAN A ERECUTIVE VICE-PRE RUSSELLY J, CHIVICA RUSSELLY J, CHIVICA RUSSELLY J, CHIVICA RUSSELLY J, CHIVICA RUSSELY AL PRIMATHAL JANES LEONAD A HILO OFFICE HILO OFFICE	GRATY MURXAM, GRATY MURXAM, TON SCHNELL, AL ASSOCIATE RAYMOND T, HIGA, A ASSOCIATE KEVN NISHKAWA, A ASSOCIATE	Нокенди Ортс 1001 Вакенс-Ултея самат 1980-7581 Ам Полек, Занит Нокенди (1980) 251-147 Беми 2 узайнай ергин Беми 2 узайнай ергин (10) Алтим Ултар Гала, Намот (1977) Гала, Калана, Калан
THE ESTATE OF JAMES CAMPBELL MAR U	Mr. Vincent Shigekuni Principal PBR Hawaii 1001 Bishop Street, Suite 650 Pacific Tower Honoluhu, HI 96813 Dear Mr. Shiehuni:	University of Hawaii - West Oahu Draft Environmental Assessment/Environmental Impact Statement, Notice of Preparation Thank you for the notice and the opportunity to comment. We support PBR's efforts to bring that project to fruition. The development of University of Hawaii - West Oahu campus will have a positive impact on the Kapolei community. Please keep us informed as to your progress.	Very truly yours, Marting And Henry Eng, Alch Community Development Manager

7

1001 Kantokila Boulevud, Kapolei, Hawaii 95707 Phone (808) 674–6674 Facsinile (808) 674–3111 Website: www.kapolei.com

DEPARTMENT OF DESIGN AND CONSTRUCTION CITY AND COUNTY OF HONOLUU 883 SORTH KING STREET 11TH FLOOR 10000. (1001 HWMI 1986) Prove (1001 2001 HWMI 1986) Prove (1001 2001 HWMI 1986) Prove Translessor MMN 1 2 7 Prove Translessor MMN 2 7 Prove Translessor Prove Translessor MMN 2 7 Prove Translessor Prove Tra	e 650 -3429		Hawaii - West Oahu Draft Environmental Assessment/Environmental tent Notice of Preparation	se to your letter of December 23, 2002 requesting comments to assist th environmental impact statement for the University of Hawaii West the following comments:	mpus is included in the Kapolei interceptor sewer tributary area. The
JEREWY HARRIS MAYOR	Mr. Vincent Shigekuni PBR Hawaii Pacific Tower 1001 Bishop Street, Sui Honolulu, Hawaii 9681	Dear Mr. Shigekuni:	Subject: University of Impact Stater	This is in responyour preparation of a dr Oahu campus. We hav	The proposed of

The proposed campus is included in the Kapolei interceptor sewer tributary area. T interceptor is to be constructed by private developers. The Department of Planning and Permitting should be consulted for sewer adequacy, and the Department of Environmental Services should be consulted for adequacy at the Honouliuli Wastewater Treatment Plant. The City is developing the extension of the Kapolei Parkway roadway through the Ewa Villages development, between the connection to the existing Kapolei Parkway in Ewa by Gentry and the northwest boundary of Ewa Villages. The 116-foot night-of-way divided roadway is being developed in phases. The first phase, including the makai lanes between the connection to the existing Kapolei Parkway and Renton Road is presently scheduled to start construction in 2003 with completion in 2004. Commencement and completion of the second phase is subject to the availability of funds. If you have any questions about this project, please contact Mr. Robert Sarae at 523-4071.

Preliminary plans for the widening of Farrington Highway to four lanes are presently being prepared. Planning, transportation, and traffic studies for the West Oahu campus should take this widening under consideration. If you have any questions on this matter, please contact Mr. Albert Miyashiro at 527-5155.

Mr. Vincent Shigekuni Page 2 March 6, 2003

.....

÷

The impact of the new campus will mean elimination of the residential development previously planned in the designated area of Kapolei. Thus, there will no longer be a need for park land, nor will land be set aside for park dedication. If you have any questions about park dedication, please contact Mr. Terry Hildebrand at 523-4696.

Very truly yours,

TIMOTHY B. STEINBERGER, P. E. Acting Director

TES:ei

cc: William D. Balfour, Jr., Department of Parks and Recreation



LAND PLANNING LANDSCAPE ARCHITECTINE ENVIRONMENTAL STUDIES

VM. FRANK BRANDT, FASLA CHAIRMAN

THOMAS S. WITTEN. ASLA PRESIDENT

R. STAN DUNCAN, ASLA EXECUTIVE VICE-PRESIDENT RUSSELL Y.J. CHUNG, ASLA EXECUTIVE VICE-PRESIDENT

VINCENT SHIGEKUNI PRINCIPAL

GRANT MURAKAMI, AICP Senior Associate TOM SCHNELL, AICP ASSOCIATE

RAYMOND T. HIDA, ASLA ASSOCIATE

KEVIN NISHIKAWA, ASLA ASSOCIATE

IIANOLUIU OFFICE IOOI BBUGUS TREET ASB TOWEN, SILTE 660 HONDELL, HAWAH '9 0813-3484 TPL: (606) 523-1402 FAX: (606) 523-1402 FAX: 59364 for the for the formanic control

HILO OFFICE ILLO LAURON STUBER FILLO LACOON CENTRY, SITHE 310 HILO, HAWAN 196720-4302 Taz. (1809) 501-4303 FAX: (1809) 501-4309 E-MAIL: phillo@lawa.net

WAILURU OFFICE 2123 KAOITU STRIBET WAILURU, HAWAT 195793-2204 Taie (302) 242-2023 FAAILE (302) 242-2022 EMAILE photraul@lava.net

January 27, 2005

Department of Design and Construction 650 South King Street, 11th Floor Honolulu, Hawaii 96813 Mr. Timothy Steinberger, P.E. City and County of Honolulu Director

Attn: Mr. Terry Hildebrand

SUBJECT: UNIVERSITY OF HAWAII – WEST OAHU DRAFT ENVIRONMENTAL ASSESSMENT/ENVIRONMENTAL IMPACT STATEMENT PREPARATION NOTICE (EISPN)

JAMES LEONARD, AICP PRINCIPAL HILO OFFICE

Dear Mr. Steinberger:

Thank you for your letter dated March 6, 2003. We have reviewed your letter and offer the following responses:

- The information regarding the Kapolei interceptor sewer tributary area, the Kapolei Parkway, and the widening of Farrington Highway will be included in the Draft Environmental Impact Statement.
- 2. Current plans for the proposed UH West Oahu Campus includes residential use and park areas.

Again, thank you for your comments and your participation in the EIS process. If you have any questions, please do not hesitate to contact me.

Sincerely,

PBR HAWAII

Viz-BR. Sige

Vincent Shigekuni

Principal

O:UOB13/1322.05/Environmental Assessment-EISPNVPre-consultation Response Ltrs/BL-19 DDC.doc

8.2 **EISPN CONSULTATION**

The EISPN was published in the February 8, 2005 issue of the OEQC's *The Environmental Notice* and sent to the parties listed in the following table. Those who provided comments are identified in **bold** lettering. The 30-day public comment period ended on March 9, 2005. Comment letters (and appropriate responses) are included in this chapter.

	AGENCY	EISPN MAIL DATE	DATE OF COMMENTS
Stat	te		
1	Department of Business, Economic Development and Tourism	02-01-05	
2	Department of Business, Economic Development and Tourism – Land Use Commission	02-01-05	
3	Department of Business, Economic Development and Tourism – Office of Planning	02-01-05	03-07-05
4	Department of Business, Economic Development and Tourism – Strategic Industries Division	02-01-05	
5	Department of Defense	02-01-05	
6	Department of Education	02-01-05	03-02-05
7	Department of Hawaiian Home Lands	02-01-05	
8	Department of Health – Environmental Planning Office	02-01-05	
9	Department of Health – Office of Environmental Quality Control	02-01-05	02-08-05
10	Department of Land and Natural Resources	02-01-05	02-07-05
11	Department of Land and Natural Resources – Historic Preservation Division	02-01-05	
12	Department of Transportation	02-01-05	
13	Department of Transportation – Airports Division	02-01-05	
14	Office of Hawaiian Affairs	02-01-05	
15	State Legislature – Representative Mark Moses	02-01-05	
16	State Legislature – Senator Brian Kanno	02-01-05	
17	State Legislature – Senator Will Espero		03-04-05
18	University of Hawai'i – Environmental Center	02-01-05	
19	Hawai'i State Library	02-01-05	
20	'Ewa Beach Public and School Library	02-01-05	
21	Kapolei Public Library	02-01-05	
22	University of Hawai'i – Hamilton Library	02-01-05	
23	Department of Business, Economic Development and Tourism Library	02-01-05	
24	Legislative Reference Bureau	02-01-05	

Table 8.2 – EISPN Comment Letters

UNIVERSITY OF HAWAI'I WEST O'AHU Final Draft Environmental Impact Statement

	AGENCY	EISPN MAIL DATE	DATE OF COMMENTS
25	City and County of Honolulu Department of Customer Services Library (formerly the Municipal Reference and Records Center)	02-01-05	
Fed	eral		
26	Department of the Army – Army Engineer District	02-01-05	03-01-05
27	Department of the Interior – Fish and Wildlife Service	02-01-05	
28	Department of the Navy	02-01-05	
City			
29	Board of Water Supply	02-01-05	02-14-05
30	City Council – Councilmember Todd Apo	02-01-05	
31	Department of Community Services	02-01-05	02-22-05
32	Department of Design and Construction	<u>02-01-05</u>	<u>03-21-05</u>
33	Department of Environmental Services	02-01-05	
34	Department of Facility Maintenance	02-01-05	03-21-05
35	Department of Parks and Recreation	02-01-05	02-14-05
36	Department of Planning and Permitting	02-01-05	03-14-05
37	Department of Transportation Services	02-01-05	03-11-05
38	'Ewa Neighborhood Board, No. 23	02-01-05	
39	Fire Department	02-01-05	02-25-05
40	Makakilo/Kapolei Neighborhood Board, No. 34	02-01-05	
41	Police Department	02-01-05	03-02-05
42	Wai'anae Neighborhood Board, No. 24	02-01-05	
Oth	er Organizations/Individuals		
43	Hawaiian Electric Company, Inc.	02-01-05	
44	Palehua Community Association	02-01-05	
45	The Estate of James Campbell	02-01-05	
46	The Gentry Companies	02-01-05	

ECONOMIC DEVELOPMENT & TOURISM	Mr. Vincent Shigekuni Page 2 March 7, 2005
DFFICE OF PLANNING 135 South Berefania Street, 6th Floor, Honolulu, Hawaii 96813 Aalifig Address: P.O. Box 2359, Honolulu, Hawaii 96804 083.00	The current development concept includes development of the West Oahu Campus along with residential, commercial, recreational and public facility uses.
00000 March 7, 2005	The University of Hawaii has developed two Alternatives for the site – a Residential/Commercial alternative and a Campus Expansion Alternative. Each alternative
Shigekuni Street, Suite 650 Iawaii 96813	provides acreage to develop a campus for approximately 1,520 students, a University Village that will include a mix of residential units and commercial space, and areas for an elementary school, parks, commercial uses, and open-space/retention basins. However, the Residential/Commercial alternative would provide 1,816 residential units over 164.7 acres and 89.2 acres for commercial space, and the Campus Expansion Alternative would provide 960 residential units over approximately 48.5 acres, and 79.2 acres for commercial uses, with approximately 126.2 acres starts area state for cammus excansion. DifHT has distributed a DFA
igekuni:	describing a proposal to develop its East Kapolei Parcel B site (9-1-16: 108 por.) mauka of Kapolei Boulevard to accommodate an ultimate residential population of 1,728.
cct: Environmental Impact Statement Preparation Notice (EISPN) University of Hawaii, West Oahu, Hawaii TMK: 9-1-016: 120,127 and 129	Traffic congestion, groundwater recharge, wastewater treatment/disposal, drainage and flooding are major issues that should be addressed in the Draft Environmental Impact Statement.
University of Hawaii will develop approximately 500.327 acres of land within the Jee Urban District located makin of the H-1 Freeway. The land area is bounded by dighway to the north, Kapolei Golf Course on the west, Kapolei Parkway on the opposed alignment for the North-South Road, and Phase I and Phase II of the East dential development to the east. The subject parcel is a portion of a 1,300-acre land s reclassified from the State Land Use Agricultural District to the Urban District on 3, 1999. According to the Petitioner (Housing and Community Development of Hawaii), the purpose of the reclassification was to establish a land bank from velop a master planned community of land uses that would generate income to Diversity of Hawaii Campus (UHWOC). At that time, the UHWOC appeted marka of the H-1. Freeway on and a velop at master of the set of the velopment of the work on antwork of 991 ² acres or containing the development of the capacity of 991 ² acres or containing the development of the velopment of the velopment of the Volte and Volte Arthune the Volte and Volte Arthure the Volte and Volte Arthure the Volte and Volte Arthure the Volte Arthure the Volte Arthure to the Volte Arthure the Volte Arthure to the Volte Arth	Even with the construction of golf courses that serve as retention/detention basins throughout the region, drainage and flooding are major issues for Oahu's Second City. Developments are examined for the impact a project will have on the area drainage patterns. We understand that drainage from the Kaloi Guldh above Farrington Highway will be realigned to the North-South Road utility corridor. The DEIS should include information regarding the impact of the proposed realignment on the project, and elements within the drainage master plan that will be coordinated with the development of the West Oahu Campus. The Kaloi Gulch drainage system along the subject property will be a private man-made drainage channel. Landowners and developers in the region are responsible for the portion of the system that falls within their project area. The cost of the drainage realignment should be discussed in relation to each landowner.
 of Regents in 1996 for the West Oahu Campus. and uses envisioned for the West Oahu Campus. A mix of single-family, multi-family, commercial, public facility (i.e. school); A sports complex to provide a venue for recreational activities and economic benefits to the State; Open space, recreational land uses within walking distance of each other and future adjoining land uses; and Provision of two hundred (200) acres to the Department of Hawaiian 	Traffic congestion remains a major concern to the Ewa community. There are at a minimum six (6) road improvement projects that are either underway or will be undertaken by the State and the County to relieve <i>existing</i> traffic problems. When the studies for the Department of Transportation's Ewa Highway Impact Fee Program were completed in 2002 by Kaku and Associates, the subject 1,300-acre land area was designated as a land bank targeted for future development that would generate income to support the UHWOC mauka of the H-1 Freeway. Further, data collected for the Kaku study was dependent upon the information provided by the developers. The data and information collected for the study may no longer be relevant as development scenarios and projected outcomes have changed.

Home Lands.

Mr. Vincent Shigekuni Page 3 March 7, 2005 The residential units proposed for UHWOC campus operations, commercial uses, and the 400 units proposed for the Department of Hawaiian Home Lands parcel to the south have the potential to minimize the overall relief anticipated from the planned roadway improvements. This does not take into consideration traffic impacts from the initial 1,520 students or the University's ultimate target of 7,600, employees of the University and commercial operations, or residents from outside the community attracted to the project's commercial sites.

The DEIS Traffic Impact Analysis Report should include an assessment of the anticipated impacts from the two development alternatives (Campus Expansion Alternative and Residential/Commercial Alternative). Further, since vehicle trips per household differ from single-family developments to multifamily developments, each alternative should include an analysis of traffic impacts from these different types of residential developments.

There are currently at least six (6) roadway improvement projects either underway or proposed to be completed by 2010 that alleviate the existing substandard levels of service for major roadways in the region. The projects include the following:

- H-1 Freeway Interchange at Makakilo
 - H-1 Freeway Interchange at Kapolei
- * North South Road from Kapolei to the H-1 Freeway
- Kapolei Parkway completion from Ewa Beach to Ko Olina
 - Fort Barrette widening Road to four lanes
 - Fort Weaver Road widening to six lanes
 - Farrington Highway expansion

The provision of adequate transportation infrastructure and timing within the context of ongoing and future development is the most outstanding concern of the Ewa community. For example, Phase I of the North-South Road will complete the first three lanes from Kapolei Parkway to the H-1 Freeway. The target date for completion of this phase is the end of 2008. No completion date has been determined for Phase II, which will complete the North-South Road improvements to its final six-lane configuration. We recommend the DEIS include a discussion for both the Campus Expansion Alternative and the Residential/Connercial Alternative on how roadway access to and from the site and road development within the site will be coordinated with the aforementioned road improvement projects. Most of the highway improvements are schedule for completion by 2010. The DEIS should discuss how the project will be coordinated with the schedule for roadway improvements, and the impact from the two alternatives. The concept for development of this area presented in Docket A99-728 was to plan all of the land uses within walking distance of each other and from other adjoining land uses. These adjoining land uses now include the Kapolei Golf Course, East Kapolei Phases I and II, Ewa Villages and Golf Course, and the proposed 400-unit residential development proposed by the Department of Hawaiian Home Lands. At this stage, it would appear that given present plans and the need for the six-lane configuration for North-South Road, "walking" to these adjoining uses may no longer be an option. However, the opportunity to promote a sense of community for Oahu's Second City through road

Mr. Vincent Shigekuni Page 4 March 7, 2005 interconnectivity and expansion of travel choices may still be an option. The DEIS should include a description of how the project proposes to facilitate pedestrian paths and biking that will interconnect with adjoining uses. Other issues related to roadways and mitigating traffic congestion that should be addressed within the DEIS include:

- Impact of the project's proposed uses on surrounding traffic patterns
 - Traffic safety issues (vehicles, bikeways, pedestrian)
 - Project circulation layout
 - Access roads locations
- Interconnectivity/connections with adjacent sites
- Parking requirements for the various uses proposed within the project
 - Pedestrian and bikeway connections
- Proposed transit service by the City and County of Honolulu to service the area
 - Use of a community shuttle service
- Coordination and scheduling of project with County plans for public transportation – i.e. Bus Rapid Transit (BRT), rail, and shuttle service

The project should not entirely depend on the implementation of roadway and transit improvements proposed for this region to mitigate the project's potential impacts. Funding and/or lack of community support may delay or even suspend implementation of these improvements. For example, the proposed Bus Rapid Transit (BRT) purported to be a major solution to relieving traffic congestion could be replaced with plans for rail transit. The solution to traffic circulation problems is not providing more roads, but implementing programs and policies to decrease the amount of vehicles using them, and to slow development to keep pace with infrastructure improvements.

We note that the project does not include plans to include a senior residential facility. The project's location and proposed mix of uses and pedestrian oriented community present an ideal opportunity to construct this much-needed housing type. The Office of Planning supports the proposed uses, providing infrastructure and public service needs, including the project's ability to generate income to support the UHWOC. The DEIS should describe which portions of the project will generate funds, the anticipated revenue, and how much of the overall budget it will represent.

The Office of Planning may comment further regarding the project's impacts on drainage patterns, wastewater treatment, traffic, roadways, non-point source pollution, and water availability during the environmental review process when the Draft Environmental Impact Statement is reviewed by our office.

Mr. Vincent Shigekuni Page 5 March 7, 2005	PBR	June 16, 2006
	HAWAII	
Thank you for including us in the review process. If you have any questions, please contact Judith Henry at 587-2803.	LAND PLANNENG LANDSCAPE ARCHITECTURE ENVIRONMENTAL STUDIES	Ms. Mary Lou Kobayashi, Administrator State of Hawai'i
Sincerely,	WM BRANDT EASI A	Department of Business, Economic Development & Tourism Office of Planning
	CHAIRMAN	P.O. Box 2359
Margadow Kilingshi	THOMAS S. WITTEN, ASLA PRESIDENT	Honolulu, Hawai'i 96804
Mary Lou Kobayashi Administrator	R. STAN DUNCAN, ASLA Executive Vice-President	Atta: Ms. Judith Henry
Office of Planning	RUSSELL Y.J. CHUNG, ASLA EXECUTIVE VICE-PRESIDENT	SUBJECT: UNIVERSITY OF HAWAI'I WEST OAHU (UH WEST O'AHU) RNVIRONMENTAL MAPACT STATEMENT BEED A ATTOM
c: Anthony Ching, Land Use Commission	VINCENT SHIGHUNI VICE PRENDENT	NOTICE (EISPN)
Once of Education August Control University of Hawaii, Office of Capital Improvements	GRANT MURAKAMI, AICP Principal	Dear Ms. Kobayashi:
		Since filing the UH West O'ahu EISPN in January 2005, the University has selected a
	TOM SCHNELL, AICP SENIOR ASSOCIATE	development partner to purchase approximately 287 acres of land within the approximately 500-acre property. The University will own the remaining 214 acres of
	RAYMOND T. HIGA, ASLA Senior Associate	land. The Private Development Lands will serve as revenue-producing lands and the profits (sale income) from these lands will be used to construct the first phase of the
	KEVIN NISHIKAWA, ASLA	campus.
	A330CIA18 KIMI MIKAMI YUEN A550CIATE	Thank you for your letter dated March 7, 2005 (your reference number P-10830). We have reviewed your letter and offer the following responses to your comments:
	SCOTT ABRIGO ASSOCIATE	1. The University is no longer considering the two development alternatives previously
	Scott murakamı Associate	described in the EISPN (Residential/Commercial Alternative and Campus Expansion Alternative) and has selected a private developer to construct the first phase of the
		campus. Approximately 1,480 residential units (including student housing) and 606,570 square feet of commercial space are currently proposed in the UH West O'thin 1 and 200,200 are set of commercial 3,560 are set of 200,200 are set of 200,20
	HosoLutu Oppicz	O and Lands. Approximately 2,300 residential units and 236,300 square feet of commercial space are proposed in the Private Development Lands. In total, the
	1001 Bishor STREET ASB TOWER, SUITE 650 HONCOLULU, MAMATI 9681 3-3484 TEL: (808) 521-5631 E-L. 6003, 521-5631	project will provide approximately 4,040 residential units in a range of housing types, including 760 student housing units at 3 beds per unit.
	FAX: (808) 523-1402 3-MALL: sysadmin@pbrhawaii.com	In addition to the 103.5-acre campus, the University is proposing a mixed-use
	HILO OFFICE 101 AUPUNI STREET	University Village, student housing/mixed use or campus expansion, and work force/affordable housing.
	HLO LACCON CENTER, SUITE 310 HLO, HAWAT 96220-4262 TEL: (803) 961-3333 FAX: (808) 961-4389	2. Traffic congestion, groundwater recharge, wastewater treatment/disposal, drainage, and flooding will be addressed in the Draft EIS.
	WAILURU OFFICE 1737 WLI PA LOOR, SUITE 4 WALLURU, FAWAN 129 (5973-127) The (603) 242-2502 FAX: (603) 242-2502	

Ms. Mary Lou Kobayashi SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) ENVIRONMENTAL IMPACT STATEMENT PREPARATION NOTICE (EISPN) June 16, 2006 Page 3	surrounding traffic patterns, traffic safety, traffic circulation, connectivity, parking requirement pedestrian and bikeway connections, proposed transit service, and shuttle service. 9. The UH West O'ahu will not entirely depend on the implementation of roadway and trans improvements proposed for the area. Since the campus is located mear various residential area	and relatively flat and dry land, walking and bicycling will be encouraged. 10. There are currently no plans to include a senior residential facility within the UH West O'ahu. 11. We appreciate the Office of Planning's support for the proposed uses. The Draft FIS wi	describe the portions of the project that will generate funds, the anticipated revenue, and the overall budget. 12. We understand that the Office of Planning may provide additional comments after reviewing the Draft EIS.	Thank you again for your participation in the Environmental Impact Statement process. If you have an questions regarding this project, please do not hesitate to contact me at 521-5631. Sincerely,	PBR HAWAII Vurzed R. Shigh- Vincent Shivekini	Vice President cc Gene Awakuni/UH West O'ahu	G.W.B.R.M.M.B.R.M.B.R.M.B.R.M.B.R.M.B.R.M.B.R.M.B.R.B.M.B.R.B.M.B.B.R.CO.M.B.B.D.R.D.P.R.B.D.B.B.D.R.D.P.R.B.D.B
Ms. Mary Lou Kobayashi SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) ENVIRONMENTAL IMPACT STATEMENT PREPARATION NOTICE (EISPN) Iune 16, 2006 Page 2	3. The proposed realignment of Kalo'i Gulch to North-South Road will reduce the potential for flood within the UH West O'ahu property. The Draft EIS will further discuss the impact of the proposed realignment of Kalo'i Gulch on the project. It will also include information on elements of the drainage master plan that will be coordinated with the development of UH West O'ahu. Landowners and developers in the area will be responsible for the proton of the Kalo'i	 Gulch drainage system that falls within their project area. A. As Ordinance 02-52 (Ewa Highway Impact Fee) is due for a review in 2007, there are a number of studies that are occurring this year that will provide towards updating road improvement projects. These include OMPO, the Ewa Connectivity Study, and the TIAR for UH West O'ahu. 	Traffic data and information collected for the Department of Transportation Ewa Highway Impact Fee Program (2002), when the UH West O'ahu was proposed to be developed mauka of the H-1 Freeway, is no longer relevant as development scenarios and projected outcomes have changed. A Traffic Impact Analysis Report (TIAR) is being prepared for the UH West O'ahu at its currently monosed site and will be indeploded.	5. We acknowledge your assessment that the proposed uses on the UH West O'ahu property and the proposed residential units on the Department of Hawaiian Home Lands East Kapolei Development Parcel B have the potential to minimize the overall relief anticipated from the planned roadway improvements. However, the UH West O'ahu property is part of the area designeded by roticy to be however.	vesuguated by pourcy to be the Second CHY. The six transportation projects fisted in your letter are designed to relieve existing and future traffic congestion. Additionally, the Impact Fee Ordinance (and its planned update in 2007) utilizes projections of future development in the area, and the UH West O'ahu project will comply with this ordinance. Furthermore, the proposed uses within the UH West O'ahu and nearby residential areas will encourage walking, bicycling, and the use of public transportation. The North-South Road is nanned to include redestrim and	bicycle facilities, in addition to an exclusive transit corridor. Future mass transit stations are being discussed along North-South Road and Kapolei Parkway, and the City also plans to locate a park-and-ride near the UH West O'ahu campus. In addition, the UH West O'ahu is expected to alleviate traffic to other colleges, universities, and workplaces in Central Oahu and Honolulu.	 The University is no longer considering the two development alternatives described in the EISPN. A TIAR is being conducted and will assess traffic impacts from different types of residential developments.

and two intersections at Farrington Highway. Access throughout the campus will be provided by a loop road encircling the major University buildings. The TIAR will include an assessment of the anticipated impacts from the development, and how access will be coordinated with the Access to the UH West O'ahu property is planned via three intersections at North-South Road ~

The Draft EIS will include a description of how the UH West O'ahu project will facilitate The Draft EIS will also discuss pedestrian paths and biking to connect adjoining uses. ø

roadway improvement projects proposed or currently underway.

LINDA LINGLE GOVERNON



PATRICIA HAMAMOTO SUPERNTENDENT

> S LALE OF HAWAL' DEPARTMENT OF EDUCATION R.O. BOX 2360 HONOLULU, HAWAI' 96604

> > OFFICE OF BUSINESS SERVICES

March 2, 2005

Mr. Vincent Shigekuni PBR Hawaii ASB Tower, Suite 650 1001 Bishop Street Honolulu, Hawaii 96813-3484

Dear Mr. Shigekuni:

Subject: Environmental Impact Statement Preparation Notice University of Hawaii - West Oahu Ewa, Oahu: TMK, 9-1-016: 120, 127, & 129 The Department of Education (DOE) has reviewed the Environmental Impact Statement Preparation Notice (EISPN) for the University of Hawai'i - West Oahu (UHWO) development in East Kapolei, Oahu. The EJSPN states that the number of possible residential units within the project could range from a low of 960 multit-family units to a high of 1,816 units depending on how the remaining land outside of the initial UHWO campus and the University Village development is developed.

The DOE estimates that the number of public school students residing in the project would range from 493 to 702 depending on the number of units and whether the units on the remaining land are multi-family or single-family. The EISPN shows a site for a 12-acre elementary school in the southeast corner of the project. The site is adjacent to areas identified for on-site detention and retention basins. We would appreciate as much detail as possible regarding these basins and any potential harm they might pose to elementary school-age children.

The EISPN gives no indication of when any of the residential units in the University Village development or the remaining land would be built. We are concerned that students in the project would be required to attend schools that already exceed their facility capacities, particularly Kapolei Middle and Kapolei High schools.

AN AFFIRMATIVE ACTION AND EQUAL OPPORTUNITY EMPLOYER

Mr. Vincent Shigekuni Page 2 March 2, 2005 The EISPN discussion on the impact to the public schools is limited to a list of the 2004-2005 enrollments for 12 different schools in the Kapolei and Campbell school complexes. We have attached a table of the ten schools in closest proximity to UHWO. One of the 10 will not open until the 2006-2007 school year. The table indicates that all but two of the remaining nine schools saw increases in their enrollment this year.

As of 2004, four of the nine schools closest to UHWO have exceeded their facility capacity. By the 2007-2008 school year, six of the nine schools expected to exceed their facility capacities.

In 1999, the State Land Use Commission reclassified the land under the UHWO campus for urban development. As a condition of that reclassification, the developer of the land shall contribute to the development, funding and/or construction of public school facilities to the satisfaction of the DOE. The DOE expects to go into discussions with UH as to the UHWO contributions as a written agreement is required before the City and County of Honolulu zoning can be approved.

If you have any questions, please call me at 586-3444 or Heidi Meeker of the Facilities and Support Services Branch at 733-4862.

Sincerely,

RAMION

kae M. Loui Assistant Superintendent

RML:hy

Attachment

 Mamo Carreira, CAS, Campbell/Kapolei/Waianae Complex Area Office of Environmental Quality Control Jan Yokota, UTH-Office of Capital Improvements

	June 16, 2006 Ms. Patricia Hamamoto, Superintendent State of Hawai'i Department of Education Office of Business Services P.O. Box 2360 Honoluhi Hawai' 96804	Attn: Ms. Heidi Meeker, Facilities and Support Services Branch SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU)	ENVIRONMENTAL IMPACT STATEMENT PREPARATION NOTICE (EISPN) Dear Ms. Hamamoto:	Since filing the UH West O'ahu EJSPN in January 2005, the University has selected a development partner to purchase approximately 287 acres of land within the approximately 500-acre property. The University will own the remaining 214 acres of land. The Private Development Lands will serve as revenue-producing lands and the profits (sale income) from these lands will be used to construct the first phase of the campus.	Thank you for your letter dated March 2, 2005. We have reviewed your letter and offer the following responses to your comments: 1. Under the current master plan, the project will provide approximately 4,040 residential units, including approximately 365 single-family residential units and approximately 3,675 multi-family residential units. However, it should be noted that 760 of the multi-family residential units are student housing units (at 3 beds per unit), which do not generate school-aged students.	2. A 12-acre elementary school is currently proposed at the southern portion of the UH West O'ahu property, adjacent to low-density residential parcels and two internal roadways. The elementary school site is separated from the retention/detention area by one of these roadways and is not expected to pose any harm to elementary school-aged children.	3. Phase 1 of the project will include the construction of approximately 616 residential units (single- and multi-family). In addition to the proposed elementary school within the property, the Department of Hawaiian Home Lands (DHHL) is proposing an elementary school site and a middle school site in its East Kapolei 2 project area. Additionally, three elementary school site, one middle school site, and one high school site are currently planned within the Ho'opili community just east of the UH West O'ahu property. A 50-acre high school site on DHHL land between Farrington Highway and the H-1 Freeway is under discussion.	
Possible Schools for East Kapolei in the Campbell and Kapolei Complexes Additional Capacity Act Enroll Act Enroll students in Projected Projected Projected Projected Projected	LANDERAGEMENT LANDERAGEMENT LANDERAGEMENT LANDERANDE, FASLA CANADAM TRANK BRANDT, FASLA TRANK BRANDT, FASLA	R. STAN DUNCAN, ASLA Eacuttre Vace-Paesideat Russell Y.J. CHUMA, ASLA Executive Vace-Presideat	VINCENT SHIGEKUM UNCE PAESNERT GRANT MURAKAMI, AICP PRINCIPAL	TOM SCHNELL, AICP SENORASEOCATE RAYMON T. FIIGA, ASLA SENOR ASSOCATE KEVIN NISHIKAWA, ASLA ASSOCATE	KINI MIKANI YUEN Assocate Scott Abrigo Assocate Scott NIRAKANI Assocate	Hovenuu OPPECE 1001 Shine Sherer AST Towns, Sum 650 Hovenuu, Maw 1983 33-344 Thu, (689) 231-402 Faxt, 669) 253-1402 E-Mut: syndmine prehmenia com	HLIO OFFICE 101 Auron Strutter HLIO LOCON CENTER, SUITE 310 HLIO LACON CENTER, SUITE 310 HLIO LACON CENTER, SUITE 310 FLA, (830) 901 - 9985 FLA, (830) 914 - 9987 TEL: (630) 342 - 2373 FLA, (830) 342 - 2397 FLA, (830) 242 - 2997	-
	Possible Scl	Capacity	East Ka	polei in the Ca Addit Act. Enroll stude	ampbell and Kapo ional nts in Projected Proj	olei Comple	XES ed 09 Enroll less	

	Capacity 2003	Act. Enroll 2003	Act, Enroll 2004	students in 2004	Projected Enroll 2005	Projected Enroli 2007	Projected Enroli 2009	09 Enroll less '03 Capacity
Kapolei High	1853	1928	2162	234	2412	2668	2712	859
Kapolei Mid	1387	1698	1699	1	1704	1730	1715	328
Elementary so	hools							
Kapolei Compl	lex							
Barbers Point	636	381	510	129	422	445	426	-210
Kapolei El	1198	1165	1187	22	1228	1240	1246	48
Campbell Com	plex							
Ewa	659	850	896	46	863	899	893	234
Holomua	1184	1344	1437	93	1424	1462	1457	273
Iroquois Point	878	455	485	30	409	367	343	-535
Kaimiloa	740	691	686	-5	633	621	618	-122
Pohakea	635	563	560	-3	578	656	826	191
Ocean Point	650					462	674	24

	008 586 4185 DFC ENVIRONMENTAL 102:55:12 02:08-2005	1.12
MS. Patricia Hamamoto SUBJECT: UNIVERSITY OF HAWALI WEST O'AHU (UH WEST O'AHU) ENVIRONMENTAL IMPACT STATEMENT PREPARATION NOTICE (EISPN) June 16, 2006 Page 2	LINDA LINGLE GOVERNON OF HAMAD	NOSNON
4. The table you attached to your letter will be included in the Draft EIS.	STATE OF HAWAII	
5. We acknowledge that a developer must contribute to the development, funding, and/or construction of public school facilities to the DOE's satisfaction, under the condition by which the State Land Use Commission reclassified land including the UH West O'ahu property for urban development. The University will coordinate its contributions with the DOE before change of zoning applications can be approved.	OFFICE OF ENVIRONMENTAL QUALITY CONTROL zis south HERETWAS TREET UNIVERSIA A AND A	
Thank you again for your participation in the Environmental Impact Statement process. If you have any anteriors in the Environmental Impact Statement process.	February 8, 2005	
questions regaranting this project, please do not hesitate to contact me at 521-5631. Sincercly, PBR HAWAII	Jan Yokota University of Hawaii Office of Capital Improvements 1951 East-West Road	
1.544-11	Honolulu, Hawaii 96822	
merce A. Durden-	Dear Ms. Yokota:	
Vincent Shigekuni Vice President	Subject: Environmental Impact Statement (EIS) Preparation Notice, UH - West Oahu	
cc Gene Awakuni/UH West O'ahu	Include the following in the draft EIS:	
O:VOB13V13Z2.05/EJSUDraft EJSUEISPN Commente-Responses/BL-02 DOE response.doc	<u>Community meetings</u> : Section 2.2 notes that meetings were held with neighborhood boards and civic groups. Include the meeting dates and a <i>synopsis</i> of the issues raised. Meeting transcripts are not required, just a synopsis.	
	<u>Terms, abbreviations and acronyms</u> : It would be helpful for the reviewer if the draft EIS included a glossary of terms, abbreviations and acronyms.	
	<u>Selection among alternatives</u> : Section 2.4.2 notes that the preferred alternative will be based on developer interest. §11-200-17 (F) of Hawaii Administrative Rules states the following regarding selection (emphasis added):	
	F. The draft EIS shall describe in a separate and distinct section alternatives which could attain the objectives of the action, regardless of cost, in sufficient detail to explain why they were rejected. The section shall include a rigorous exploration and objective evaluation of the environmental impacts of all such alternative actions. Particular attention shall be given to alternatives that might enhance environmental quality or avoid, reduce, or minimize some or all of the adverse environmental effects, costs, and risks.	
	The basis for selection needs to be environmental factors, as required by HAR 11-200.	
	Air quality technical report: Section 5.5 references the study for the East Kapolei final EIS. Either reproduce the summary and conclusions of this study or, if not lengthy, the entire study.	

June 16, 2006	Ms. Genevieve Salmonson, Director State of Hawai'i	Department of Health Office of Environmental Quality Control 235 South Beretania Street, Suite 702 Honolulu, Hawai'i 96813	Attn: Ms. Nancy Heinrich	 SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) ENVIRONMENTAL IMPACT STATEMENT PREPARATION NOTICE (EISPN) 	Dear Ms. Salmonson:	Since filing the UH West O'ahu EISPN in January 2005, the University has selected a development partner to purchase approximately 287 acres of land within the approximately 500-acre property. The University will own the remaining 214 acres of	and. The Private Development Lands will serve as revenue-producing lands and the profits (sale income) from these lands will be used to construct the first phase of the campus.		Thank you for your letter dated February 8, 2005. We have reviewed your letter and offer the following responses to your comments:	1. The Draft EIS will list some of the community meetings held during the campus site selection study process.	2. The Draft EIS will also include a glossary of terms, abbreviations, and acronyms used.	 In accordance with Section 11-200-17(f), Hawaii Administrative Rules (HAR), the Draft EIS will describe the alternatives that could attain the objectives of the action, regardless of cost, in sufficient detail to explain why they were rejected. Selection must be based on environmental factors, as required by Section 11-200, HAR. 	4. The EISPN made reference to the air quality report prepared for the <i>East Kapolei</i> Master Plan Final Environmental Impact Statement (1996). However, an air quality report is being prepared for the UH West O'ahu project and will be included in the Draft EIS.	An Environmental Data Resources radius map report was prepared and will be included in the Draft EIS. The Draft EIS will also discuss appropriate mitigation measures.
$\operatorname{PBR}_{^{\mathrm{H}\mathrm{A}\mathrm{W}\mathrm{A}\mathrm{H}\mathrm{H}}}$	LAND ELAYNGG LANDSCAFE ARCITTECTURE ENVEROMENTAL STUDIES	WM. FRANK BRANDT, FASL <i>Chairman</i> Thomas S. Witten, ASLA <i>Paesnear</i>	R. STAN DUNCAN, ASLA EXECUTIVE VICE-PRESIDENT	Russell Y.J. Chung, ASL/ Executive Wice-President Vincent Shigeruni Vince President	GRANT MURAKAMI, AICP Principal	TOM SCHNELL, AJCP Senor Associate	RAYMOND T. HIGA, ASLA Sevior Associate	KEVIN NISHIXAWA, ASLA ASSOCIATE	Kimi Mikami Yuen Associate	SCOTT ABRIGO ASSOCIATE SCOTT MURAKAMI ASSOCIATE		HONOLULU OFFICE HONOLULU OFFICE LOI BRANG-FREET NOTE: ANALT PORT, SAITE 600 HONOLUL ANALT 195(2): 3-463 TEL: (200) 253-1403 FACI, 193461010 @phthawdii.co	Нило Ортиск 101 Алгачит Street Нило Locoor Charres, Surre 310 Нило, HAwar 196520–4252 Та.: (808) 961-4399 FAX: (808) 961-4399	WAILUKU OFFICE 1787 WILLEA, LOOR, 5017E 4 WALLUKU, TAAAN T5 00759, 12271 TEL (603) 2422 2273 FAX: (603) 2422 2502
Jan Yokota	reonary 8, 2005 Page 2	<u>Hazardous materials</u> : Since the project site was formerly in agricultural production, there may be pesticides present in the soil. In addition, the Chief Engineer of DLNR's Engineering Division noted in his January 3^{rd} , 2003 memo that a high level of dioxins was present in an adjacent parcel. Include a hazardous materials report or a discussion in the text that indicates how you will	mitigate or remediate these soils.	Funding: Disclose the percentages of Federal, state and county funds involved. This includes any federal funds flowing through the state or county.	alternative transportation inotes. Nater pointy (ritk) 26, 264, 344) requires the promotion of alternative forms of transportation systems that reduce reliance on the private automobile, concernent systems that reduce reliance on the private automobile,	this policy, please discuss what provisions are being made to create bicycle lanes or facilities, promote pedestrian safety and/or encourage other non-motorized modes of transportation.	If you have any questions, call Nancy Heinrich at 586-4185.	Sincerely,	Gennieu Schneen	GENEVIEVE SALMONSON Director	c: Vince Shigekuni			

212 10:55:27 02-08-2005

OFC ENVIRONMENTAL

308 586 4186

PETER 1 YOUNG DAMPGRON BOWD OF UND XXD1XH201RESURGES COMISSION ON WATEN RESOURCE MUNICINEIN	ADJITE ELEXARGE BATTARO CEANARCENARIA BATTARO CEANARCENARIA COMARCENARIA COMARCENARIA CONTRET RESOLUTION CONTRET RESOLUTION CONTRACTOR CONTRACT	NUHORUANE BLAD RESEARE COMMUSSION STATE PLANS	LD-NAV			Notice	on the subject	oject was : of Land and		sion and Dívision of	other comment	las A. Vaccaro		······································
	STATE OF HAWAII	POST OFFICE BOX 221 HONOLULU, HAWAII 55809	J5 LSPN-PBR.RCM	jekuni Street, Suite 650 iwali 96813	.gekuni:	T: University of Hawaii - West Oahu Environmental Impact Statement Preparation Island of Oahu, Hawaii	you for the opportunity to review and comment	of the document pertaining to the subject proor made available to the following Department urces' Divisions for their review and comment	 Engineering Division Division of State Parks Commission on Water Resource Management Office of Conservation and Coastal Lands Land-Oahu District Land Office 	ed please find a copy of the Engineering Divis n Water Resource' comments and a copy of the response.	partment of Land and Natural Resources has no the subject matter.	you have any questions, please contact Nichol Division Support Services Branch at 587-0384.	Very truly yours,	MARREN F. WEGESEND JR. Administrator
СНОА LINGLE СРОЖВИСК РИЧИЛИ		and the second	March 3, 200 UHWESROAHUEI ABD HINNEI	Vincent Shig 1001 Bishop Honolulu, Ha	Dear Mr. Shi	SUBJEC	Thank 1 matter.	A copy transmitted Natural Reso		Enclose Commission o State Parks	The Der to offer on	Should of the Land		
Ms. Genevieve Salmonson SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) ENVJRONMENTAL IMPACT STATEMENT PREPARATION NOTICE (EISPN)	June 16, 2006 Page 2 6. The project will not receive any Federal funding. Currently, all funding is planned through the	7. In accordance with the policies in Chapters 26, 226, 264, and 344, Hawaii Revised Statutes, the Draft EIS will discuss the provisions that will be made to create bicycle lanes or facilities,	Thank you again for your participation in the Environmental Impact Statement process. If you have any unstitums reporting this project plass do not beside to control more Statement process. If you have any	Sincerely,	PBR HAWAII	Nuiz & R. Shight- Vincent Shicehni	vincer original	cc Gene Awakuni/UH West O'ahu Oviden array overeich-of effectery Communication of the second of the second of the second of the second of the	OBQC TESORGEOC					

C: ODLO



Division: burkering

ÉRIC T. HIRANO, CHIEF ENGINEER

Name:

Sciliu Name: Dauled S.

Division: State Parts

EFER YONG FECTIVED FECTIVE FEC	February 17, 2005 Mr. Werran Wanssond In Administrator	rant vesticati veegesend, Jr., Aurimitisuadu Land Division Dean A. Nakano, Acting Deputy Director	Commission on Water Resource Management (CWRM)	University of Hawall West Oahu, Environmental Impact Statement Preparation Notice, Oahu k you for the opportunity to review the subject document. Our comments related to water resources elow.	neral, the CWRM strongly promotes the efficient use of our water resources through conservation d use of alternative non-potable water resources whenever available, feasible, and there are no ts to the ecosystem. Also, the CWRM encourtaces the motaction of water rechards are are no	the maintenance of streams and the replenishment of aquiters.	commend coordination with the Engineering Division of the State Department of Land and Natural Resources to incorporate otect into the State Water Projects Plan.	e concreated about the potential for ground or surface water degradation/contamination and recommend that approvals for olect be conditioned upon a review by the State Department of Health and the developer's acceptance of any resulting sments related to water quality. The supervision of the environment of the environment of the source of any resulting construction Permit and/or a Pump installation Permit from the Commission would be required before ground water is be date as source of suppy for the project.	oposed water supply source for the project is located in a designated water management area, and a Water Use Permit from mimission would be required prior to use of this source. Awater withdrawals from this project may affect streamflows, which may require an instream flow standard amendment.	concerned about the potential for degradation of instream uses from development on highly evoluble stopes adjacent to is within or near the project. We recommend that approvals for this project be conditioned upon a review by the condring countifys Building Department; and the developer's acceptance of any resulting requirements related to erosion	roposed project includes construction of a stream diversion, the project may require a stream diversion works permit and the instream flow standard for the affected stream(s). roposed project alters the bed and banks of a stream channel, the project may require a stream channel alteration permit.	A: Experiment of that the Environmental Impact Statement include the projected potable and non-potable acommends, an anticipated schedule to build out, and the proposed water sources to meet the individual for the works or non-potable water demands, or both, well nuction, pump installation, and water use permits from the Commission would be required. If existing are to be utilized, the developer should consult with the Commission as to the permits that so the easilities of the easilities.
LINDA LINGLE DEVENDENT NUMBER	Ċ	FROM:		SUBJECT: Than are marked b	In ge measures an harmful effec	important for [X] Were Plan.	{X] Were this pr	 We are this preduction that is preducted in the second seco	[] The pr the Co [] Groun	[] We an stream corres contro	[] If the p ameno	(X) OTHE We re water dema dema dema dema dema source source source
FETER 1: VOUNG MARING METER 1: VOUNG MARING MARING MARING COMMESSION ON WITER RESOURCE MANGEMEN COMMESSION ON MARING MARING MARING CONVERSION MARING CONVERI	-	spense Date: 2/18/05		0.ement:	.1. Lands		ation Notice	ing to the subject Division letterhead	bject matter, please	ents by the suspense	ts attached. 2 - 14/05	1111
DEPARTMENT OF HAWAII DEPARTMENT OF LAND AND MATURAL RESOUR	February 7, 2005	OAHUEISPNPBR.CMT Su	NDUM:	XXX Engineering Division XXX Division of State Parks VXXX Commission on Water Resource Mana	XXX Office of Conservation and Coasta XXX Land-Oahu District Land Office	Warren F. Wegesend A., Administrator Land Division	T: University of Hawaii - West Oahu Environmental Impact Statement Prepar Island of Oahu, Hawaii	lease review the document (CD) pertain and submit your comment (if any) on and dated by the suspense date.	nould you need more time to review the su t Nicholas A. Vaccaro at 587-0384.	f this office does not receive your comm we will assume there are no comments.	have no comments. (X) Commen : Una Information Date: d	LENDOR NAVARA Division:
		LD/NA' UHWES'	MEMORU	TO:		FROM:	SUBJEC	t matter signeo	contac	date,	() Wé Signed	Name:

.

If there are any questions, please contact Lenore Y. Nakama at 587-0218.

June 16, 2006	Mr. Warren F. Wegesend, Jr. SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) ENVIRONMENTAL IMPACT STATEMENT PREPARATION NOTICE (EISPN) June 16, 2006 Page 2
Mr. Warren F. Wegesend, Jr., Administrator State of Hawai'i	2. As recommended, coordination with the City and County of Honolulu will take place to incorporate this project into the City's Water Use and Development Plan.
Department of Land and Natural Resources Land Division P.O. Box 621	3. Coordination will also take place with the Engineering Division of the DLNR to incorporate this project into the State Water Projects Plan.
Honolulu, Hawai'i 96809 Attu: Mr. Nicholas A. Vaccaro	4. The Draft EIS will include the projected potable and non-potable water demands, an anticipated schedule to build-out, and the proposed water sources to meet the demands.
SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) ENVIRONMENTAL, IMPACT STATEMENT PERPARATION	Thank you again for your participation in the Environmental Impact Statement process. If you have any questions regarding this project, please do not hesitate to contact me at 521-5631.
NOTICE (EISPN)	Sincerely,
Dear Mr. Wegesend:	PBR HAWAII
Since filing the UH West O'ahu EISPN in January 2005, the University has selected a development partner to purchase approximately 287 acres of land within the approximately 500-acre property. The University will own the remaining 214 acres of land. The Private Development Lands will serve as revenue-producing lands and the proved of the private Development Lands will serve as revenue-producing lands and the proved of the private Development Lands will serve as revenue-producing lands and the proved of the private Development Lands will serve as revenue-producing lands and the proved of the private Development Lands will serve as revenue-producing lands and the proved of the private Development to be a servenue producing lands and the proved of the private Development to be a servenue producing lands and the proved of the private Development to be a servenue producing lands and the produce of the private Development to be a servenue producing lands and the produce of the private Development to be a servenue produce of the private Development to be a servenue produce of the private Development to be a servenue produce of the private Development to be a servenue produce of the private Development to be a servenue produce of the private Development to be a servenue produce of the private Development to be a servenue produce of the private Development to be a servenue produce of the private Development to be a servenue produce of the private Development to be a servenue produce of the private Development to be a servenue produce of the private Development to be a servenue produce of the private Development to be a servenue produce of the private Development to be a servenue produce of the private Development to be a servenue produce of the private Development to be a servenue produce of the private Development to be a servenue produce of the private of t	Vurzek Ruger Vincent Shigekuni Vice President
provins yeare incouncy more these tailos will be used to consuluct the litist phase of the campus.	cc Gene Awakuni/UH West Oʻahu
Thank you for your letter dated March 3, 2005 (reference LD/NAV UHWESTOAHUEISPN-PBR.RCM). We understand that the EISPN was distributed to the following divisions:	O:UOB13N/322.05/EIS/Draft EIS/EISFN Comments-Responses/BL-04 DL/NR response doe

- Engineering Division (no comments in addition to those provided in the preconsultation period); i.

 - Division of State Parks (no comments);
- below);
 - Office of Conservation and Coastal Lands (no comments); and Office of Conservation and Coastal Lands (no con
 Land – Oahu District Land Office (no comments).
 - Howouru.OFFFCE 1001 Bitwo STREET ASB TOWRE, SUITE 650 JONOURU, HAWAT 9481-3-384 TEL: (8093 3:21-5601 FAX: (8093 3:21-3602 AMUL systemin @phhwati.com
- HILO OFFICE 101 AUTUNI STREET IN-LACCON CENTER, SUITE 310 HILO, HAWAY 196720-4262 TEL: (808) 561-3333 FAX: (808) 561-4989 FAX: (808) 561-4989

WALLORU OFFICE 1787 WILI PA LOOP, SUITE 4 VALLUKU, FLAWARI 96793-1271 TEL: (808) 242-2502 FAX: (808) 242-2902

Mr. Warren F. Wegesend, Jr., Administrator Department of Land and Natural Resources Honolulu, Hawai'i 96809 State of Hawai'i Land Division P.O. Box 621

Attn: Mr. Nicholas A. Vaccaro

VINCENT SHIGEKUNI VICE PRESIDENT

Dear Mr. Wegesend:

- Commission on Water Resource Management (see responses to comments રું છું

In response to the Commission on Water Resource Management comments:

water resources whenever available, feasible, and there are no harmful effects to 1. We acknowledge that the CWRM strongly promotes the efficient use of water the ecosystem. The CWRM also encourages the protection of water recharge areas, which are important for the maintenance of streams and the replenishment resources through conservation measures and use of alternative non-potable of aquifers.

LAND FLANNING LANDSCAPE ARCHITECTURE ENVIRONMENTAL STUDIES V M V

VM. FRANK BRANDT, FASLA CHAIRMAN

THOMAS S. WITTEN, ASLA PRESIDENT

R. STAN DUNCAN, ASLA EXECUTIVE VICE-PRESIDENT

UUSSELL Y.J. CHUNG, ASLA EXECUTIVE VICE-PRESIDENT

GRANT MURAKAMI, AICP Principal

TOM SCHNELL, AICP SENIOR ASSOCIATE

RAYMOND T. HIGA, ASLA Senior Associate

KEVIN NISHIKAWA, ASLA ASSOCIATE

KIMI MIKAMI YUEN Associate

SCOTT ABRIGO ASSOCIATE

SCOTT MURAKAMI ASSOCIATE

The Church P. Chird Jergislatture of the start at Aptanci strate corrio. March J. Ergislatture of the start at Aptanci strate corrio. March J. Ergislatture of the start at Aptanci strate corrio. March J. 2005 March 4, 2005 March 4, 2005 March 4, 2005 March 4, 2005 March 4, 2005 March 4, 2005 March 4, 2005 March 4, 2005 March 4, 2005 March 4, 2005 March 4, 2005 March 4, 2005 March 4, 2005 March 4, 2005 March 4, 2005 March 4, 2005 March 4, 2005 Department of Health Leiöpepa A Kamehameha, Suite 702 March 4, 2005 Department of Health HI 96813 March 4, 2005 Dear Ms. Salmonsen: Dear Ms. Salmonsen: March 4, 2005 Jass Street Honolulu, HI 96813 Dear Ms. Salmonsen: March 4, 2005 Dear Ms. Salmonsen: Dear Ms. Salmonsen: March 4, 2005 March 4, 2005 Dear Ms. Salmonsen: Dear Ms. Salmonsen: March 4, 2005 March 4, 2005 Jass March 19 19 19 19 19 19 19 19 19 19 19 19 19	BR ^{June} 16, 2006	AREANDERTING AVE AND THE AND	CHARGE AND	 Itellikawa, ASLA Thank you for your letter dated March 4, 2005. We appreciate your strong suppor Associate public/private partnership to build the UH West O'ahu. We agree with your asses that a four-year institution is the missing link in the Ewa-Kapolei region, and the transmission that a four-year institution is the missing link in the Ewa-Kapolei region, and think campus will alleviate some traffic congestion between Kapolei and Mānoa during transmission. TAMBARANI Thank you again for your participation in the Environmental Impact Statement procession. TANURANANI Thank you again for your participation in the Environmental Impact Statement procession. Soutonee Soutonee Sincerely, BR HAWAII Sincerely, BR HAWAII 	In obsers Vincent Shigekuni www.scars.surg.jo www.scars.surg.surg.scars.scars.scars.scars.scars.scars.scars.scars.scars.
	The Twenty-Third Legislature	of the state of Antunui State of Antunui STATE CAPITOL HONOLULI, HAWANI 98613 HONOLULI, HAWANI 98613 March 4, 2005	Ms. Genevieve Salmonsen, Director Ms. Genevieve Salmonsen, Director Office of Environmental Quality Control Department of Health Leiopapa A Kamehameha, Suite 702 235 S. Beretamia Street Honolulu, HI 96813 Tow Poar Ms. Salmonsen: State Salmonsen:	I am writing to express my strong support for a public/private Reversing to build a new University of Hawal'i – West O'ahu campus. A quality four-year institution is the missing link in the Bwa/Kapolai region. The burgeoning population of Leeward O'ahu makes it imperative that we move forward without any further delay. Scatter many jobs for Hawal'i residents. This new campus will alleviate some of the traffic congestion between Kapolei and Mānoa during peak traffic hours, as well as create many jobs for Hawal'i residents. Please contact me if I can be of any assistance in making UH – West Meeven Reveaulty.	Will Espero Will France Will Espero State Senator

2 /2

15:19:11 03-14-2005

OFC ENVIRONMENTAL

808 586 4186

June 16, 2006		Mr. George P. Young, P.E., Chief U.S. Department of the Army LA U.S. Army Engineer District, Honolulu Regulatory Branch A Fort Shafter, Hawaii 96838-5440	Attn: Ms. Connie Ramsey	 SUBJECT: UNIVERSITY OF HAWAI'I WEST OAHU (UH WEST O'AHU) ENVIRONMENTAL IMPACT STATEMENT PREPARATION NOTICE (EISPN) 	P Dear Mr. Young:	Since filing the UH West O'ahu EISPN in January 2005, the University has selected a development partner to purchase approximately 287 acres of land within the approximately 500-acre property. The University will own the remaining 214 acres of land. The Private Development Lands will serve as revenue-producing lands and the profits (sale income) from these lands will be used to construct the first phase of the campus.	Thank you for your letter dated March 1, 2005 (project file number POH-2005-89). Subsequent to your letter, we understand that the Corps concurs that the upper reaches of	Kato'i Gulch do not have a regulated tributary connection to waters of the U.S. Therefore, a Department of the Arrny permit pursuant to Section 404 of the Clean Water Act will not be required for the proposed fill associated with the UH West O'ahu campus or the North-South Road.	Thank you again for your participation in the Environmental Impact Statement process. If you have any questions regarding this project, please do not hesitate to contact me at 521-5631.	⁴⁴ Sincerely, ²⁰¹ PBR HAWAII	10 United Ringe- Vincent Shigekuni Vice President	cc Gene Awakuni/UH West O'ahu o:VOBI311322.05EISUDafi EISTEISPN Comments-Responses/BL-06 Amy COE response.doc
PBR ILAWALI	LAND FLANNEND LANDSCAFE ARCHITECTURE EAVEOSIMENTAL STUDIES	WAI, FRANK BRANDT, FASI <i>Cianban</i> Thonaas S, Witten, ASL, <i>Presiden</i>	R. STAN DUNCAN, ASLA Executive Vice-Presiden	RUSSELL Y.J. CHUNG, ASL EXECTIVE VICE-PRESIDEN VINCENT SHIGEKUNI VICE PRESIDENT	GRANT MURAKAMI, AICP <i>Principal</i>	TOM SCHNELL, AICP SEWOR ASSOCIATE RAYMOND T. HIGA, ASLA SENDOR ASSOCIATE KEVIN NISHKAWA, ASLA ASSOCIATE	Kimi Mikami Yuen Associate	SCOTT ABRIGO ASSOCIATE SCOTT MURAXAMI ASSOCIATE	Howeldu OFFICE 1001 Bission SFREET	ASB TOWEX, SUTE 650 HOWLUP, HAWATT PERI 3-48 TEL: (605) 523-14501 FAX: (808) 523-1402 E-MALL: systemin@pb/hswaii.co	IILO OFFICE III AUANN SPERE III AUANN SPERES IILO LACONY CENTER IILO LACONY CENTER IILO LACONY CENTER IILO LACONY CENTER FAX: (805) 961-4959 FAX: (805) 961-4959	WAILJIKU ÖFFICE (787 WLIJ PA, LOOR, SUITE 4 WAILJIKU, HAWAH '95679-1271 TEL: (808) 242-2902 FAX: (808) 242-2902
DEPARTMENT OF THE ARMY U.S. ARM ENGINEER DISTRICT. HOROLULU FT. SHAFTER, HAWAII 96053-5440 March 1, 2005	Artentino or Artento or ulatory Branch	cent Shigekuni K Hawaii 3 Tower, Suite 650 1 Bishop Street	olulu, H1 96813	ject: Environmental Impact Statement Preparation Notice (EISPN) for the University of /aii, West O'ahu Campus, Kapolei, Oahu, Hawaii (TMK: (1) 9-1-016: 120, 127, and 129). r Mr. Shigekuni:	This office has reviewed the materials you submitted pursuant to Section 404 of the Clean	er Act (CWA) (33 U.S.C.1344). As mentioned in the EISPN, coordination has already been ated with our office in regard to the Corps' jurisdictional authority over the Kaloi Gulch and Hunehune Gulch, both of which traverse the subject properties. Site inspections were Jucted by my staff on December 21, 2004, and January 11 and February 4, 2005. The field stigations revealed that both features, within and Ianuary 11 and a distictive lack of vegetation strifes, have well-defined channels with a coblect bottom and a distictive lack of vegetation in the invert. Based on the site inspections and a review of reference materials found in our	ces, the Kaloi Gulch and Hunehune Gulch are defined as waters of the U.S. pursuant to Part of the CWA.	Section 404 of the Clean Water Act requires that a DA permit be obtained prior to the ement or discharge of dredged and/or fill material into waters of the U.S. Based on our ex of the information provided, we have determined that the aforementioned project would live work in and/or placement of dredged and/or fill material into waters of the U.S. Juder	regulatory jurisdiction. Therefore, a DA permit would be required prior to any construction urring within the gulches.	If you have any additional questions, please contact Ms. Connic Ramsey at the letterhead css, by telephone at 808-438-2039, by facsimile at 808-438-4060 or by electronic mail at <u>nie.L.Ramsey@usace.army.mil</u> . File number POIH-2005-89 has been assigned to this eet. Thank you for your cooperation with our regulatory program.	Sincerely,	Ceorge F. 1 oung, F.L. Chief, Regulatory Branch ames Yamamoto, R.M. Towill Corp., 420 Waiakamilo Road #411, Honoiulu, HI 96817

ARD OF WATER SUPPLY	MUEI HANNEMARK, MAYOR	
NID COUNTY OF HONOLULU JUTH BERETANIA STREET LULU, H 36843	EDDIE FLORES JR., Chaliman CHARLES A. STED, Ves-Chaiman HERBERT S. K. KANDVA, SR. DAROXIVH I, LEINDA, SR. DAROXIVH I, LEINDA	JULIE 10, 2000
February 14, 2005	LANDKOWEY K. HARAGA, Ex-Officia RODNEY K. HARAGA, Ex-Officia CLIFFORD S. JAMIL E Manager and Chef Engineer DONNA FY K. KIYOSA POWN Manager and Chef Engineer CJAJRWAN	 Mr. Keith S. Shida, Principal Executive City and County of Honolulu Board of Water Supply Customer Care Unit Gastomut Beretating Street LA 630 South Beretating Street
	THOMAS S. WITTEN, ASI Pressident	A Attri- Mr Tocoth Karlen
Mr. Vincent Shigekuni, Principat PBR Hawaii	R. STAN DURCAN, ASL/ Elecuting Vice-Preside	THE SUBJECT: UNIVERSITY OF HAWAI'I WEST OAHU (UH WEST O'AHU)
1001 Bishop Street Honolulu, Hawaii 96813	RUSELL Y_L CHUNG, ASI EXECUTIVE VICE PRESIDE	ENVIRONMENTAL IMPACT STATEMENT PREPARATION NOTICE (EISPN)
Dear Mr. Shigekuni:	VINCENT SHICEKUM VICE PREVENT	Dear Mr. Shida:
Subject: Your Letter Dated January 31, 2005 on the Environmental Preparation Notice for University of Hawaii-West Oahu, TMK: 9-1-16:120, 127, and 129.	GRAAT MURAKANI, AIC PRINCIPAL Tom Scienteit, AICP Tom Scienteit, AICP	^P Since filing the UH West O'ahu EISPN in January 2005, the University has selected a development partner to purchase approximately 287 acres of land within the approximately 500-acre property. The University will own the remaining 214 acres of land. The Private Development Lands will serve as revenue-producine lands and the
Thank you for the opportunity to comment on the proposed development.	SEWOR ASSOCIATE Raynond T, Higa, ASL, SeWord Associate	profits (sale income) from these lands will be used to construct the first phase of the campus.
A water master plan for the East Kapolei development should be submitt	ed for our approval. KEVIN NISHIKANA, ASLA Associate	 Thank you for your letter dated February 14, 2005. We have reviewed your letter and offer the following responses to your comments:
The developer will be required to install the necessary water system impriproposed development, including a reservoir and transmission mains.	overnents to serve the Associate	I. A water master plan has been submitted and approved by the BWS.
Non-potable water should be used for irrigation of large landscape areas.	SCOTT ABRIGO ASSOCIATE SCOTT MILBAXANI	2. Necessary water system improvements to serve the proposed development will be provided.
If you have any questions, please contact Joseph Kaakua at 748-5442.	Alsocarts	The use of non-potable water will be coordinated with the BWS and used as permitted by State Department of Health regulations.
Very truly yours,		
K. Sur	HovoLULU OPPICK 1011 BisHove Strager ASD TOVER, SUITE 600	Thank you agate for your participation in the Environmental Impact Statement process. If you have any questions regarding this project, please do not hesitate to contact me at 521-5631.
KEYTH S. SHIDA Principal Executive Customer Care Unit	TEL (ROY) CALL (ROY A) 243-642 TEL (ROY) 221-601 FAX (ROY) 523-1402 S-MAL: 97940fub @phhwafi	* Sincerely, on PBR HAWAII
	HILLO OFFICE IN LATTON FREET RELO LACOON CORTER, SUITE RELO LACOON CORTER, SUITE RELO MANNY 196729-041 RELO MANNY 196729-041 RELO RADON 201-0495 RELO RADON 201-0495	10 Viized R. Shight
Cc: OEQC University of Hawaii	NALIKU OFFICE 1757 WL 16. Loss, SMITE4	cc Gene Awakuni/Ulf West O'ahu
	WAILURU, HAWATY 96793-127 The (803) 242-2878 Fac: (868) 246-2902	l O.UOB13/1322.05/ElStDrafi ElStEISPN Comments-Responses/BL-07 BWS response doc

BOARD OF WATER SUPPLY

CITY AND COUNTY OF HONOLULU 630 SOUTH BERETANIA STREET HONOLULU, HI 96843

Water for Life Ka Wat Ola

June 16, 2006	IA WAIL Ms. Deborah K. Morikawa, Director IRAH K. MORIKAWA LAURE LANDING Department of Community Services Acring Intercost LAURE LANDING Director JOHN R. SABAS TJ5 South King Street, Suite 311 Honolulu, Hawai'i 96813	WM.FRANK.BRANDT.FASLA Attn: Mr. Keith Ishida CHARMAN	THOMARS, WITTEN, ASLA SUBJECT: UNIVERSITY OF HAWAI'I WEST OAHU (UH 1 PRESIDENT ENVIRONMENTAL IMPACT STATEMENT PREPARA R. STAN DUKOM, ASLA (EISPN) EXECUTIVE UNE-PREDENT (EISPN)	RUSSELL Y.J. CHUNG, ASLA DCar Ms. Morikawa: Protinge Vice Persinent	VINCENT SIDE ALL Since filing the UH West O'ahu EISPN in January 2005, the University VINCENT SIDE ALL VINCENT SIDE ALL Since filing the UH West O'ahu EISPN in January 2005, the University VINCENT SIDE ALL VINCENT SIDE ALL development partner to purchase approximately 287 acres of land within the GRANT MUBARMAN, AICP GRANT MUBARMA, AICP 500-acre property. The University will own the remaining 214 acres of land within the profile (st processor) PRINCIPAL Development Lands will serve as revenue-producing lands and the profile (st these lands will be used to construct the first phase of the campus.	TOMSCHARLAICP Thank you for your letter dated February 22, 2005. We have reviewed your le SENIOR ASSOCIATE following responses to your comments: RAYMOND THALA ASLA	SENTOR ASSOCIATE 1. The Draft EIS will describe the housing elements (especially any faculty o KEVIN NISHIKAWA, ASLA Of the proposed project to the extent possible. ASSOCIATE	RIMM MIXANTYUEN 2. The UH West O'ahu will offer a variety of programs and services to Associate din Associate Community. Students at the existing UH West O'ahu campus at Lee community. Students at the existing UH West O'ahu campus at Lee contract and the average age being 33. The offer a approximately half of all course in the evening or during the wei students cancer expected to change significantly when the UH W becomes a four-year institution. Support services will increase since college and un-year institution. Support services will increase since our sually work while attending college. The goal of the campus is to be an i greater Kapolei community.	However the elderly or disabled. ABP TOPER STORE STORE STORE STORE AND A STORE STOR	TAX: (2003) 251-402 FAX: (2003) 251-402 SAMAL: 9ystemin@phtawaii.com any questions regarding this project, please do not hesitate to contact me at 521-563.	III.0 OFFICE Sincerely, HILO LACON CENTRA SHITE 310 HILO LACON CONTRA SHITE 310 HILO LACON CONTRA SHITE 310 The I (801) 501 - 459 FAX: (801) 501 - 459 The I (801) 501 - 559 The	WALDAU OFFICE Vincent Shigekuni
DEPARTIMENT OF COMMUNITY SERVICES CITY AND COUNTY OF HONOLULUU 715 Southking Street, sure B11 + HOMOLULUU	DEBO		February 22, 2005	/incent Shigekuni, Principal Havvaii	Tower, Suite 650 Tower, Suite 650 Bishop Street Julu, Hawaii 96813 Mr. Shigekuni:	Subject: Environmental Impact Statement Preparation Notice University of Hawaii – West Oahu	The Department of Community Services appreciates the opportunity to review and nent on the subject Environmental Impact Statement Preparation Notice.	We recommend that the draft Environmental Impact Statement describe the housin ants of the proposed project to the extent possible. We would be particularly interest ing details of any faculty or student housing that will be developed as part of the prop ct. The Draft Environmental Impact Statement should also describe any programs an ese to be provided at University of Hawaii – West Oahu that will benefit or involve th community, such as adult education and continuing education classes, services to th y and disabled, and cultural activities and programs.	Thank you for the opportunity to provide these comments, and we look forward to wing the draft Environmental Impact Statement when it is available. Questions regar	natter may be directed to Mr. Keith Ishida at 527-5092. Sincerely,	Odorah K. Markana DEBORAH K. MORIKAWA Director	idk

MUFI HANNEMANN MAYOR

cc: Office of Environmental Quality (Jan Yokota, University of Hawaii

June 16, 2006	Mr. Wayne M. Hashiro, P.E., Director Department of Design and Construction City and County of Honolulu 650 South King Street, 11 th Floor Honolulu, Hannaid, Action	Attn: Mr. Gary Doi	SUBJECT: UNIVERSITY OF HAWAI'I WEST OAHU (UH WEST O'AHU) ENVIRONMENTAL IMPACT STATEMENT PREPARATION NOTICE (EISPN) Door Ma. Hambian	Since filing the UH West O'ahu EISPN in January 2005, the University has selected a	development partner to purchase approximately 287 acres of land within the approximately 500-acre property. The University will own the remaining 214 acres of land. The Private Development 1 ands will serve as revenue-renducing loads and the	profits (sale income) from these lands will be used to construct the first phase of the campus.	Thank you for your letter dated March 21, 2005 (your reference 93836). We understand that the City and County of Honolulu, Department of Parks and Recreation's (DPR) community and district park guidelines indicate that for every 10,000 people there should be two acres	of park space available. A portion of the residential lands on the property will be subject to the DPR's park decication requirements. Depending on meetings with the DPR and the selected developer, these requirements may be fulfilled through the decication of land, payment of a paytenet of a pay	and whit continue to be consulted intolghout the planning process for the UH West Oahu. Thank you again for your participation in the Environmental Impact Statement process. If you have any questions regarding this project, please do not hesitate to contact me at 521- 5631.	Sincerely, PBR HAWAII	Vurze Z. Burgh- Vincent Shigekuni	Vice President cc Gene Awakuni/UH West O'ahu	O:VOB1341322.051E[StDraft E]S1EJSPN Comments-Responses/BL-09 DDC response.doc
PBR	LAND FLANNING LANDSCAFE ACTITICTURE BYVRCONALSTTAL STUDIES	WA. FRANK BRANDT, FASLA CHAIRMAN THOMAS S. WITTEN, ASLA PRESIDENT	R. STAN DUNCAN, ASLA EXECUTVE VICE-PRESIDENT RUSSELL Y.J. CHUNG, ASLA EXECUTIVE VICE-PRESIDENT	VINCENT SHIGEKUNI VICE PRESIDENT	GRANT MURAKAMI, AICP Principal	TOM SCHNELL, AICP Sevior Associate	RAYMOND T, HIGA, ASLA Senior Associate Kevin Mishikawa, ASLA Associate	KIMI MIKAMI YUEN ASSOCIATE SCOTT ABRIGO ASSOCIATE	SCOTT MURAKAMI Associate	I RONGLULU OPPERE 1001 BISROF STREET ASB TOWER, SUTTE 650 HONGLULU HANN 19681 3-3484 TEX: (809) 521-5421 E MART EVERIDES CANAMARIE COM-	III.co OFFICE III.co OFFICE III.co Autoros Stream Auto, Locook Carten, Surra 310 Halo, Carten Active Active Active Active Active Active	Thu: (808) 961.4989 Fax: (808) 961.4989	Waluku Offick 1737 Wll Dan Suite 4 Naluku Hawayi 96793-1271 Tel: (688) 242-2902 Fay: (888) 242-2902
DEPARTMENT OF DESIGN AND CONSTRUCTION CITY AND COUNTY OF HONOLULU BOSOUTH KINS STREET, I THE LOOR Phone: (2001) 523-41664 • Fax: (2003) 523-41667 Workite: www.hornolulu.gov	WAYNE M. HASHIRO, P.E. ACTING DRECTOR ELGENE G. LEE, P.E. DEUTY SIRECTOR	93836 March 21, 2005	skuni 650	t 96813	ai:	ity of Hawaii - West Oahu mental Impact Statement Preparation Notice (EISPN)	sviewed the information provided in the EISPN for the proposed University Oahu campus. We have the following suggestion relating to Section 5-11.5.	nd County of Honolulu, Department of Parks and Recreation's (DPR) strict park guidelines indicate that for every 10,000 people, there should be use available.	vledge the existence of parks in Kapolei as well as in adjacent areas of Ewa and West Beach. In addition, the planned 31.5 acres of open space to service ution area for possible passive park usage is included in the Long Range . This existing and proposed acreage may not meet the DPR's requirements	e curphasize the treed to address the issue of additional park space in ng for the West Oahu campus. Mr. Gary Doi at 527-6699 if there are any questions.	Very truly yours,	For WAYNE M. HASHIRO, P.E. Acting Director	ironmental Quality Control
	MUFI HANNEMANN MAYOB		Mr. Vincent Shigel PBR Hawaii ASB Tower, Suite.	1001 Bishop Street Honolulu, Hawaii	Dear Mr. Shigekun	Subject: Universi Environr	We have re of Hawaii - West C	The City an community and dis 2 acres of park spa	We acknow Beach, Makakilo, a as a retention/deten Development Plan.	voi park space. We subsequent plannin Please call 1			WMH:ei cc: Office of Envir

June 16, 2006	Ms. Laveme Higa, P.E., Director Department of Facility Maintenance City and County of Honolulu : FASLA 1000 Uluohia Street, Suite 215 (ASLA Kapolei, Hawai '1 96707	Attn: Mt. Charles Pignataro Asta SUBJECT: UNIVERSITY OF HAWAI'I WEST OAHU (UH WEST O'AHU Asta ENVIRONMENTAL IMPACT STATEMENT PREPARATION Same NOTICE (EISPN) UNIC LAICP LAICP Dear Ms. Higa: LAICP Since filing the UH West O'ahu EISPN in January 2005, the University has selected 3	development partner to purchase approximately 287 acres of land within the ucr approximately 287 acres of land within the ucr approximately 500-acre property. The University will own the remaining 214 acres of land. The Private Development Lands will serve as revenue-producing lands and the use the profits (sale income) from these lands will be used to construct the first phase of the Astronometer.	Thank you for your letter dated March 21, 2005. We acknowledge that the Departmen of Facility Maintenance has no comments to offer at this time. Thank you again for your participation in the Environmental Impact Statement process If you have any questions regarding this project, please do not hesitate to contact me a S21-5631. Sincerely,	PBR HAWAII PBR HAWAIII PBR HAWAIII PBR HAWAII PBR HAWAII PBR HAWAII PBR HAWAIII PBR HAWAIII PBR HAWAII PBR HAWAII PBR HAWAII PBR HAWAII PB	18 18 1911 - OctoB1311322.08EISUDrafi EISIEISEN Comments-ResponsestEL-10 DFM response.dor 1911 - 191 1911 - 191
PBR	LAND RANDRA LANDSOME ARCHTER BYVIRONABYTAL ST BYVIRONABYTAL ST CLARK BYANDT CLARKAW TTERONAS S. WTTERO,	R. STAN DURVAN, A Executive Vice-Pree Russell V. Churk Viceop Pree Viceop Shideki Viceop Preederadi Prencipal	Tom Schrell, Al <i>Sevior Associat</i> <i>Sevior Associat</i> <i>Sevior Associat</i> <i>Sevior Associat</i> Kevin Nishkawa, /	AISOCARE KINI MIKAN YU AISOCARE SCOTT ABBIO AISOCARE SCOTT MUBAXAA AISOCARE	HIGNOLULU OFFICI (101) BENNES, SERVER ACSIT TORE, SOURCE SOURCE ACSIT TORE, HORSAN 19681 ACSIT TORE, HORSAN 213-140 ACSIT TORE, SUBS, 213-140 FAC, 1683, 213-140 FAC, 1683, 213-140 HILL OFFICE FAC, 1683, 261-335 FAC, 1683, 261-335 FAC, 1683, 261-335	WALLING OFFICE 1787 WLL PA LOOP, SU WALLUNG (AWALT) 9079
DEPARTMENT OF FACILITY MAINTEMANCE AND COUNTY OF HONOLULU DHA STREET, SUITE 215, KAPOLEI, HAWAII 96707 TELEPHONE : (803 057:504 572:857 WEDBIG: ANN-HONDALULAR	Ситемен IIGA, P.E. Астоно вистста кано снаг изванства СЕСКИС К. МИХАНОТО ПРЕИТУ ОВЕСТИ, МИХАНОТО ПРЕИТУ ОВЕСТИЯ ОВ ПОВ 199	March 21, 2005	i – West Oahu ict Statement	o review and comment on the Environment Impact Statement try 2005 for the subject project. We have no comments to add 1s, please call Charles Pignataro of our Division of Road	Very truly yours, X.U.M.M.M.M.M.M.M.M.M. LAVERNE HIGA, P.E. Acting Director and Chief Engineer ality Control	
۵ CITY 1000 ULUC	HANNEDAANN	Mr. Vincent Shigekuni PBR Hawaii 1001 Bishop Street ASB Tower, Suite 650 Honolulu, Hawaii 96813	Dear Mr. Shigekuni: Subject: University of Hawaii Environmental Impa Preparation Notice	Thank you for the opportunity to Preparation Notice dated Janua to the document at this time. Should you have any questior Maintenance at 484-7697.	cc: Office of Environmental Que University of Hawaii Office	

MUFI HANNEMANN MAYOR

	June 16, 2006	Mr. Lester K.C. Chang, Director Department of Parks and Recreation	cury and County of Foreign 209 SLA 1000 Ultuchia Street, Suite 309 Kapolei, Hawai'i 96707	TA SUBJECT: UNIVERSITY OF HAWAI'I WEST OAHU (UH WEST O'AHU)	A ENVIRONMENTAL IMPACT STATEMENT PREPARATION EV NOTICE (EISPN)	BLA Dear Mr. Chang:	Since filing the UH West O'ahu EISPN in January 2005. the University has selected a	cp development partner to purchase approximately 287 acres of land within the	approximately over-tacte property. The University will own the remaining 214 acres of land. The Private Development Lands will serve as revenue-producing lands and the	provide (sale income) from these lands will be used to construct the first phase of the campus.	LA Thank you for your letter dated February 14, 2005. We acknowledge that the	LA Department of Parks and Recreation has no comments to offer at this time.	Thank you again for your participation in the Environmental Impact Statement process. If you have any questions regarding this project, please do not hesitate to contact me at	Sincerely	PBR HAWAIT	. Otb-//	Vuicio K. Otuga-	Vincert. Singekunt Vice President	cc Gene Awakuni/UH West O'ahu	310 0-14OBI 311322.05UEISUDraft EISUEISPN Comments-Responses/BL-11 DFR response.doc	
	PBR	LAVD PLANNING LAVD PLANNING LAVDSCAPE ARCHITECTUR ENVRONNEPTAL STUDIE	WM. FRANK BRANDT, FAS CIMIRMAN	THOMAS S. WITTEN, ASI PRESIDENT	R. STAN DUNCAN, ASL, EXECUTIVE VICE-PRESIDE	RUSSELL Y.J. CHUNG, AS EXECUTIVE VICE-PRESIDE	VINCENT SHIQEKUNI VICE PRESIDENT	GRANT MURAKAMI, AIC Principal		TOM SCHNELL, AICP SENIOR ASSOCIATE	RAYMOND T. HIGA, ASL Senior associate	KEVIN NISHIKAWA, ASL. Associate	KIMI MIKAMI YUEN ASSOCIATE	SCOTT ABRIGO ASSOCIATE	SCOTT MURAKAMI ASSOCIATE		HONOLULU OFFICE 1001 BISHOP STREET ASB TOWER, SUITE 650	HowoLULU, HAWAI'1 96813-34 TEL: (808) 521-5631 FAX: (808) 523-1402 E-MAIL: streadmin@rababasaii.c		HLIO OPYICE THLIO OPYICE FLID LOLACOON CENTR, BUNE HLIO HAWAT 196720-4262 TEL: (903) 961-5333 FAX: (803) 961-5333	WAILJKU OFFICE 1787 Wal PA LOOF, SUTTE 4 WAILVEL HAWAI' 96793-127 TEL (808) 242-2378 FAX: (808) 242-2902
NC	DLULU 96707 Meneluagev	LESTER K. C. CHANG - ACTING DIRECTOR DANA L. TAXAHARA DIAS DEPUTY DIRECTOR										Environmental Impact est Oahu Campus.	he Preparation Notice.	Planner, at 692-5454.		Γ					
PARTMENT OF PARKS AND RECREATION	N D C O U N T Y O F H O N C LUOHIA STREET, SUITE 309 • KAPOLET, HAWAII 009, 6925561 • FAX: 6925131 • INTERNET; WWW	Ø		February 14, 2005						nental Impact Statement Preparation Notice	y of Hawan, west Uanu Campus	opportunity to review and comment on the I tice relating to the University of Hawaii, We	f Parks and Recreation has no comment on t	ny questions, please contact Mr. John Reid,	Sincerely,	list, us tu	LESTER K. C. CHANG				
DEI	CITY A 1000 U PHONE: (8)	UNEMANN OR				Mr. Vincent Shigekuni PBR Hawaii	ASB Tower, Suite 650 1001 Bishop Street	Honolulu, Hawaii 96813	Dear Mr. Shígekuni:	Subject: Environm	Universit	Thank you for the Statement Preparation Not	The Department of	Should you have a				イ 5.00 A	(92789)		

MUFI HANNEMANN MAYOR

wa Development Plan (DP) vision of "place-	itral places for each community and the	ercial uses in those central locations.
as will support the Ewa Development	king", creation of central places for ea	centrating of commercial uses in thos

- Where is the "main street" or town center of the University Village to be located?
- Will mixed-use structures including residential and commercial uses be developed around that center or along that "main street"? .
- Where are other "main streets" or "village centers" to be located in the project

development in the non-campus areas of the project will implement the DP ransit nodes at the corner of Farrington Highway and North South Road intersection and the NS Road -Kapolei Parkway intersection. The Draft vision for development of these two higher density mixed use centers. Transit Nodes. The Ewa Development Plan calls for development of and at a location mid way between the Farrington-North South EIS should discuss how the mixed-use residential commercial

Alternatives to Auto Use. The Draft EIS should discuss how the design of encourage the Ewa Development Plan vision for promoting transit use, the campus and adjacent residential/commercial area will support and walking, and biking as alternatives to automobile use. Connectivity. The Draft EIS should discuss how the design of the campus Scope of Residential Development. Page 7 indicates there will be a 36.2neighborhood cohesion. See the attached materials for additional details. arterials, to allow efficient transit and utility vehicle circulation, to promote has indicated the need for block and street standards and a quarter-mile connectivity for the East Kapolei region. The Ewa DP Five Year Review walking and biking as an alternative to automobile use, and to enhance collector-connector grid to reduce traffic congestion on major regional and the adjacent residential /commercial area will provide needed

acre multi-family residential development. In addition, the site plan shows

Residential Parcel B", containing 128.5 acres. Please address this additional area. Scope of Commercial Development. On page 6, it says commercial lands shows 3 areas, and they total 89.2 acres. Please clarify this in the Draft totaling 79.2 acres, in 2 areas, are proposed. However, the site plan EIS.

The Draft EIS should address how the development of commercial areas in the project will implement the policies of the Ewa DP which call for.

- communities and to be limited in size to 30 acres and 250,000 sq. planned commercial centers outside of the City of Kapolei to provide retail shopping and services for their surrounding fteet at most;
- 250,000 sq. ft to 500,000 sq. ft. of commercial space) or regional shopping centers (over 50 acres and 500,000 sq. ft.) which are planned major community commercial centers (30 to 50 acres, intended to provide for Ewa's regional shopping needs to be located in the City of Kapolei. (See DP Sec. 3.7.1)

In addition, the Plan calls for commercial uses outside the City of Kapolei to be concentrated in central locations rather than in continuous commercial strips along arterial roads.

The proposed University Village mixed-use commercial and residential area at the intersection of North-South Road and Farrington Highway appears to be consistent with these policies.

policies. It is over 75 acres in size, and stretches along the North-South submitted to the Department were big box retail outlets which the Plan However, Commercial Parcel A does not appear to implement these Road. The "major anchor tenants" proposed in earlier materials calls to be located in the City of Kapolei. The proposed lodging facility for Commercial Parcel A would be allowed only if Resort zoning was approved. The General Plan and the Ewa

Development Plan currently designate only the Ko Olina Resort and the	4.6 GROUNDWATER RESOURCES / HYDROLOGY
Ocean Pointe marina area for resort zoning.	<u>Storm Water Quality</u> . The Draft EIS should address how the project will
3.0 REQUIRED APPROVALS AND PERMITS	comply with the storm water quality requirements of the City's Rules
Consistency with City Plans. The Draft EIS report should discuss how the	Related to Storm Drainage Standards.
UH West O'ahu campus addresses the Ewa Development Plan's vision,	4.8 FLORA AND FAUNA
policies, planning principles and guidelines, especially those in Sec. 3.7.6.	Page 16 indicates there is an indigenous species of bird present on the
<u>State Land Use</u> . The Prep Notice indicates the site is within the State	site (black-crowned night heron). Please address project impacts and
Land Use Urban District. The Draft EIS should provide a mapping of State	proposed mitigation to this indigenous species. In addition, where the
Land Use designations for the 3 parcels and the surrounding area.	black-crowned night heron and plover are found, there are generally
<u>Zoning</u> . The Prep Notice indicates zoning is AG-1. A zoning map for the	endangered species of birds including the Hawaiian Stilt, Coot, and
project and surrounding areas should be provided.	Gallinule. These are found near wetlands, and are living on the nearby
Dian Daviant I and The Dreft EIS should alorify the successful sources	Kapolei Golf Course. Please address impacts on these species, if they
riant residential more. The planticle strong dating the proposed dutilified dation and residential more on the second strong str	are found at the project site.
and residential uses. Only index which are unecay serving the university (i.e., student housing) can be covered under the Plan Review Use (PRU)	5.3 ROADWAYS AND TRAFFIC
permit. Housing for the general public (dwellings and apartments), and	Integration with Transit Services. The Draft EIS should explain how the
associated commercial development, do not qualify for processing under a	UH West O'ahu campus and adjacent residential/commercial areas will
PRU. Please show the location of university-related housing and	link with and integrate with the the existing hub and spoke bus and
commercial uses on the site plan, and reference these in the DEIS	express bus service and to future planned rapid transit service.
discussion.	<u>Connectivity</u> . As noted above, analysis done during the Ewa DP Five
Subdivision. A site plan should be provided showing existing property	Year Review indicates the importance of promoting connectivity within and
lines for the 3 parcels. Subdivision action will be required, and proposed	between developments. Our consultants proposed block and street
subdivision lots should be indicated (to the extent known at this time).	standards for residential and commercial areas and called for
According to City records, the site is actually 500.327 acres.	development of a quarter-mile connector-collector grid. The Draft EIS
4.2 TOPOGRAPHY	should provide information on the level of connectivity to be acheived,
Date 11 care the electron of the former housedear is 00.44 and 14 a	both within the campus and residential/commercial areas, and between
r age in says the elevation of the tower boundary is out, thisl, and the upper boundary is 160 ft. msl. However, the site is characterized as	the project and surrounding areas.
"relatively flat" The Draft EIS should provide the average slope and the	Commuting Traffic. The EIS should estimate the potential impact that UH
distance over which there is an 80-ft. elevation differential.	West O'ahu may have on commuter traffic to UH Manoa and central
	Honolulu. If figures are available, the share of UH Manoa students who
	reside in leeward O'ahu should be provided.

·

ŝ

4

5.7 VISUAL RESOURCES AND OPEN SPACE

Significant Views. The Draft EIS should identify significant view corridors (mauka views) and describe how the UH West O'ahu campus plan preserves and frames these important views, particularly from campus 'gathering places'. See the discussion of significant public views in the Ewa DP, Section 3.4 and Section 3.7.6.

5.10 INFRASTRUCTURE

<u>Dual Water Systems</u>. Discussions of Infrastructure master plans included in the Draft EIS should identify how a dual water system will be developed to conserve potable water and to use non-potable water for irrigation and other appropriate uses (See Ewa DP Section 4.2) The possibility of joint development/use of non-potable water system with the DHHL housing development makai of the project should be discussed.

<u>Wastewater Capacity</u>. The Draft EIS should indicate that the capacity of the Honouliuli Wastewater Treatment Plant (WWTP) is limited by the capacity of the solids handling treatment units. The current capacity of the solids handling is approximately 27 to 29 million gallons per day (mgd) average flow. The average daily flow to the WWTP is about 27 mgd. The planned project to add anaerobic digesters, which will increase the solids treatment capacity, is tentatively scheduled to start in 2005 and finish in early 2007. After completion of this project, the overall WWTP capacity will be 38 mgd.

Wastewater infrastructure improvements for the project will need to include a sewer line connecting to the new Kapolei Interceptor Sewer.

5.11.6 PUBLIC TRANSPORTATION

The Draft EIS should discuss how the design and development of the UH West O'ahu campus and residential/commercial areas will encourage use of public transportation, biking and walking.

<u>Shared Parking Facilities</u>. For example; will campus parking or commercial parking be jointly developed with a park and ride facility near

the two transit nodes? Will shared parking development be encouraged for the commercial areas in the University Village and elsewhere? <u>Basis for Parking Ratio</u>. The EIS should indicate proposed parking/student ratio and the determining basis.

Pedestrian/Biking Access. How will the campus and

residential/commercial area design provide pedestrian and bicycle access to the major transit node at the intersection of Farrington Highway and North South Road?



PROPOSED BLOCK STANDARDS

LAND USE	ZONING	MAXIMUM	MID-BLOCK
DISTRICT	DISTRICT	BLOCK SIZE	ALLEY
			REQUIRED FOR
			ANY BLOCK
	;		WITH SIDE OVER
City Center Mixed	BMX-3	400' X 600' or	400'
Use		Sum of Two Sides	
		= 1000	
Community	B-2, BMX-3	600' x 600' or Sum	400'
Commercial		of Two Sides =	
		1200	
Residential	R-3.5, R-5, R-7	300' x 500' or Sum	400'
Neighborhood		of Two Sides =	
ł		800	
IMX/Business	IMX-1	800' x 800' or Sum	400'
		of Two Sides =	
		1,600	
Cul-de-sac streets s	thail not be more than	150 feet in length	

Source: Van Meter Williams Pollack, LLP and Charlier Associates, EWA SMART GROWTH DESIGN CODE (Working Draft, July 19, 2004)

C:\WINDOWS\Desktop\PROPOSED BLOCK STANDARDS.doc

CHARACTERISTICS:

- amenities such as the City Center shops, schools, parks and community facilities, The street network should have provide multiple options for reaching major without needing to access an arterial boulevard.
 - landmarks should be a basic consideration in the arrangement of streets, commercial centers and shared spaces within both residential and mixed use View corridors to the mountains, open space, and other local and regional districts.
- A circulation network master plan over the entire Plan Area should be provided that is comprised of a loose "grid" of arterials at 1/2-1 mile intervals, collector streets at 1/4 mile intervals and connectors roads between individual developments at regular intervals between collectors.
 - New streets, bike ways, paths and trails should connect to existing adjacent neighborhoods.
- Traffic calming measures should be used to eliminate shorts cuts and support a desirable living environment.
- Multiple connecting streets within and between residential neighborhoods should
- Streets, bikeways and walkways should create a unifying circulation network that knit neighborhoods together, not form barriers.
 - provides convenient routes throughout the community
 - See each District for connectivity standards (minimum intersection frequency, maximum dead end length, # of dwellings or buildings on a cul-de-sac, and minimum street spacing) within and between districts.

C:\WINDOWS\Desktop\PROPOSED BLOCK STANDARDS.doc

AND SCHOOL AND TANK BRANDT, FASLA AND FLANDARD ACUTERER ACUTERER ACUTERER ACUTERER ACUTERER ACUTERER ACUTERER ACUTERER ACUTARIA ASLA RESERDENT VICE PRESIDENT VICENT SCHOOL ACUTARIA ASLA RESERDENT VICE PRESIDENT VICENT RESERVENT VICENT RESERVENT VICENT RESERVENT VICENT RESERVENT RANT MUBARAMI, AICP PRINCFAL RANT MUBARAMI, AICP PRINCFA	 Jure 16, 2005 Mr. Henry Erg, Director Mr. Henry Erg, Director Mr. Henry Farg, Director Gö South Kang Street, 77 Flox Honolulu, Hawari '9 6613 SO South Kang Street, 77 Flox HAWAFI WEST OAHU (UH WEST O'AHU) GO South Kang Street, 77 Flox HONONUENTAL. IMPACT STATEMENT PREPARATION SUBJECT: UNIVERSITY OF HAWAFI WEST OAHU (UH WEST O'AHU) SUBJECT: UNIVERSITY OF HAWAFI WEST OAHU (UH WEST O'AHU) SUBJECT: UNIVERSITY OF HAWAFI WEST OAHU (UH WEST O'AHU) SUBJECT: UNIVERSITY OF HAWAFI WEST OAHU (UH WEST O'AHU) SUBJECT: UNIVERSITY OF HAWAFI WEST OAHU (UH WEST O'AHU) SUBJECT: UNIVERSITY OF HAWAFI WEST OAHU (UH WEST O'AHU) SUBJECT: UNIVERSITY OF HAWAFI WEST OAHU (UH WEST O'AHU) SUBJECT: UNIVERSITY OF HAWAFI WEST OAHU (UH WEST O'AHU) SUBJECT: UNIVERSITY OF HAWAFI TARY (ID AND TACT STATEMENT PREPARATION) BEATME O'AHU HIR STATE O'AHU (INI WAR TERPARATION) BEATME O'AHU HIR WILL BERN IN JANAFI TARY (INI HIR STATE AND TARY (INI HIR ST	Mr. Henn STATEM June 16, 4. 6. 8.	 y Eng y Fing T. UNYURSSITY OF HAWAII WEEST OAHU ENVIRONMENTAL IMPACT GENT PREPARATION NOTICE ZUN VIIAge is proposed as 53.1-acre sustainable, mixed-use village, closely integrated with the University Village is providing approximately 797 restential unlis. A 12.1-acre suddent housing parcel is proposed adjacent to the campus to provide family apartment-type setting. As currently planned, the development will not include Gramy family apartment-type setting. As currently planned, the development will not include Gramy family apartment-type setting. As currently planned, the development will not include Gramy family apartment-type setting. As currently planned, the development will not include Gramy family apartment-type setting. As currently planned, the development will not include Gramy family apartment-type setting. As currently planned, the development will not include Gramy family apartment-type setting. As currently planned, the development will not include Gramy family and the concentration of "place-making", creation of connectial acreas will support the <i>Euro Development Plan</i> (Ewa DP) vision of "place-making", creation of connectial acreas will support the <i>Euro Development Plan</i> (Ewa DP). a. The "main street" of University Village will be tun between the concentration of "place-making", creation of connection acreas project. Ho'opili, plans to extend this road through its property, providing connection with the notes are located on the entry road to the entry from the main street of University Village will be under stead and acreas on the maximum street could extend marks into a 38-acre student housing/mixed-use or campus expansion parcel. b. Mixed uses (including retuil, commercial, and residential) will be developed around the main street of University Village with the Ewa DP, as University Village was streed as a targeorthy between the observation of content development around a transit node located at the intersection of North-
-LAcoon CENTER, SUITE 310 LLo, HAWAI 1 96720-4262 TEL: (808) 961-3333 FAX: (808) 961-4989	3. The UH West O'ahu will be constructed in eight phases. Student housing is planned in Phase 4 (Year 2011), with single- and multi-family residential planned in mixed-use and residential areas throughout the property in Phase 1.		sound, North-South Koad and Ho'opill to the east, and Farmgton Eughway to the west. We understand that WOCD has incorporated block and street standards as well as a quarter-mile collector-connector grid in its proposed plans.
WALLORU OPPICE 87 WILT PA LOOP: SUITE 4 LUKU, HAMAI'1 96793-1271 Tel. (306) 242-2972 FAX: (808) 242-2902	Workforce/affordable housing will be developed in Phase 2 (Year 2009) and single- and multi-family residential are planned in Phase 3 (Year 2010). This information will be included in the Draft EIS.	റ്	The Draft EIS will include the revised master plan.

Mr. Henry Eng SUBJECT: UNIVERSITY OF HAWAII WEEST OAHU ENVIRONMENTAL IMPACT STATEMENT PREPARATION NOTICE June 16, 2006 Page 4	14. We understand that only uses directly serving the University can be covered under the Plan Review Use. The Draft EIS will provide additional information about the proposed commercial and residential uses. The Draft EIS will also include the current master plan,	which shows the location of University-telated housing and commercial uses. 15. A Tax Map Key map showing the existing property lines for the three parcels composing the property will be provided in the Draft EIS. The current master plan will also be included in the Draft EIS.	We appreciate the confirmation that the site is 500.327 acres.	16. The elevation of the property ranges from 80 to 160 feet mean sea level. The distance over which this 80-foot elevation differential occurs is approximately 6,500 feet. The average slope of the property is about one to two percent. This will be noted in the Draft EIS.	17. The Draft EIS will address how the project will comply with the storm water quality requirements of the City's Rules Related to Storm Drainage Standards.	18. A wildlife survey of the property will be included in the Draft EIS. The Draft EIS will address potential impacts on indigenous and endangered bird species and will propose mitigation measures. Please note that there are not wellands on the property.	19. The Draft EIS will discuss how the campus and adjacent residential/commercial areas will link with and integrate with the existing hub and spoke bus and express bus service and to future planned rapid transit service.	20. The Draft EIS will provide information on the level of connectivity to be achieved within the campus and residential/commercial areas, and between the property and surrounding areas.	21. A Traffic Impact Analysis Report is being conducted and will be included in the Draft EIS. While the project will increase traffic in and around the project site (in comparison to existing agricultural uses), the UH West O'ahn campus is expected to alleviate commuter traffic between Waipahu and Mānoa, attributed to UH Mānoa. Currently, 23 percent of undergraduate students at the UH Mānoa campus reside on the west side of Red Hill.	22. The Draft EIS will identify significant view corridors and describe how the UH West O'ahu campus plan preserves these important views.	23. A preliminary engineering report will be included in the Draft EIS. Currently, a dual water	system is proposed. Non-pound water from the Honoululu water Kecycling Faculty will be used to supplement the potable water system and meet the total water demand for the project. The non-potable water system will utilize brackish water and/or reclaimed water for irrigation	auto possible all conditioning purposes. Intis system will be completely separate from the potable water system and will be built parallel to the potable water system along North-South Road. The University is currently working with the Department of Hawaiian Home Lands to develop a non-potable water system for East Kapolei.
Mr. Henry Eng SUBJECT: UNIVERSITY OF HAWALI WEEST OAHU ENVIRONMENTAL IMPACT STATEMENT PREPARATION NOTICE June 16, 2006 Page 3	10. The proposed land uses and land use acreages presented in the EISPN have changed. The Draft EIS will include the revised master plan.	file the provided in a mixed-use area shown with the provided in the sources and 220,000 square feet) outside of the City of Kapolei. Approximately 111,000 gross square feet of commercial space will be provided in a mixed-use area along Farrington Highway. This commercial center will cater to the needs of the residential community and surrounding neighborhoods and could include supermarkets, drugstores, retail establishments, specially food item stores, commercial offices mandical featilities restores and scored accurated accurated accurated and control offices mandical featilities restores and accurated accu	sources, structures, transmuss, and personal services. Another mixed-use area along North-South Road will provide approximately 236,000 gross square feet of commercial space.	University Village is envisioned as having a mix of land uses that are closely related or cater to the University and its diverse student population. Retail establishments, such as bookstores, copy centers, coffee shops, and specially food item stores about with small start and otherse	are envisioned for this development. University Village would provide approximately 495,495 gross square feet of commercial space. This is not consistent with the Ewa DP, which calls for major community commercial centers (250,000 to 500,000 source feet) to be horded	in the City of Kapoleti; however, we hope that with the need to reduce vehicular traffic and establish a campus town (which is sorely missing in the UH Mānoa-Mõ'ili'ili area), the current Ewa DP Update can include this proposal.	The proposed commercial centers will not be developed in strips along arterial roads, but will be part of a mixed use area that also includes residential uses. No commercial-only parcels are proposed in the current master plan.	Like the University Village, these mixed-use parcels will include a mix of land uses, but will focus on catering more to the needs of the residential community and surrounding neighborhoods.	They will contain approximately 236,000 gross square feet of commercial space comprised of land uses such as supermarkets, drugstores, retail establishments, specialty food item stores, general offices, medical facilities, restaurants, personal services; along with 217 multi-family residential units catering to residents with a range of incomes (possibly including affordable housing).	11. The Draft EIS will discuss how the campus addresses the Ewa DP's vision, policies, planning principles, and guidelines, especially those in Section 3.7.6 and where revisions to the plan	may be required.	12. The Draft EIS will include a map showing the State Land Use District designation for the property and surrounding area.	13. The Draft EIS will include a map showing the zoning designation for the property and surrounding area.

Mr. Henry Eng SUBJECT: UNIVERSITY OF HAWAII WEEST OAHU ENVIRONMENTAL IMPACT STATEMENT PREPARATION NOTICE June 16, 2006 Page 5	DEPARTMENT OF TRANSPORTATION SERVICES CITY AND COUNTY OF HONOLULU 620 500TH KING STREET, AND FLOOR + HONOLULU, HAMMI 9613 TELEPHONE, 19001 523-4529 - FAX: 8061 523-457 - FAX: 8061 523 - FAX: 806	
24. The wastewater capacity information provided in your letter will be included in the Draft EfS. Infrastructure improvements for the project include a sewer line connecting to the new Kapolei Interceptor Sewer. The wastewater system will consist of a major trunk line along North-South Road and branch lines along connecting roads. The proposed campus sewer system will consist of 8- to 18-inch pipes. In addition, 30-inch sewer pipes are planned near the scondary North-South Road connection to accommodate potential campus expansion and build-out of commercial areas.	ACCOMPACT ACCOMPACT ACCOMPACT ACCOUNTINE ACCOUNTINA ACCOUNTINA ACCOUNTINA ACCOUNTINA ACC	HIRATA ECTOR
25. Mixed-use areas located around proposed rapid transit stops will encourage the use of public transportation. Walking and biking will also be encouraged by a pedestrian and bikeway network throughout the property. Campus parking for students, faculty, and staff will be developed separately from commercial parking.	March 11, 2005 Mr. Vincent Shigekuni	
26. The required number of parking stalls for Phase 1 development of the campus (1,520 students) is 760 (at 0.5 stalls per student headcount). The required number of parking stalls for the ultimate campus development (7,600 students) is 2,850 (at 0.375 stalls per student headcount). The smaller ratio is an listitute of Transportation Engineers (TTE) ratio and the larger ratio accommodates vehicles before transit is available. These numbers will be included in the Draft EIS.	PBR Hawaii ASB Tower, Suite 650 1001 Bishop Street Honolulu, Hawaii 96813 Dear Mr. Shigekuni:	
27. The UH West O'ahu campus is organized around a Central Plaza (Kalo'i Plaza) and linked to the mixed-use University Village and the inter-modal transit facilities through a pedestrian greenway system. Pedestrian and bicycle access to the major transit node at the intersection of Fartrington Highway and North-South Road will be provided by a wide pedestrian mall leading to this intersection. Pedestrian malls and secondary paths will radiate into the campus and University Village mixed-use area from Kalo'i Plaza at the center of the campus.	Subject: <u>University of Hawaii – West Oahu</u> Thank you for your January 31, 2005 letter, requesting our review of and comments on the Environmental Impact Statement (EIS) Preparation Notice for the subject project. The following comments are offered for your consideration as you prepare the draft EIS:	
lhank you again for your participation in the Environmental Impact Statement process. If you have any juestions regarding this project, please do not hesitate to contact me at 521-5631. sincerely,	 The land area reported in Sections 1.1 PROJECT SUMMARY (Page 1), 2.1 PROJECT GOALS AND OBJECTIVES (Page 3) and 2.4 UH WEST O'AHU LONG RANGE DEVELOPMENT PLAN (Page 5) is not consistent with what is shown in Figures 3A and 3B. The 500-acre land area should be verified. 	
	 The land areas for the various uses envisioned for the 500-acre property should be tabulated in Section 2.4 UH WEST O'AHU LONG RANGE DEVELOPMENT PLAN (Pages 5 to 7). 	
/incent Shigekuni /ice President	3. The western portion of Figures 3A and 3B should be expanded to include more of Farrington Highway and the adjacent land uses.	
c Gene Awakuni/UH West O'ahu o:U081311322.051EISUrafi EIS1EISFN Comments-Responses/BL-12 DPP response.doc	4. Figures 3A and 3B show the proposed location of the elementary school that is described on Page 7. The proposed elementary school should be located at a site that minimizes the need for students having to cross a major roadway/intersection to get to school.	
	5. On Page 7, the description of roadways should note the land area that will be required for this use.	

3
Mr. Vincent Shigekuni Page 2 March 11, 2005

- The following comments are related to Section 5.3 ROADWAYS AND TRAFFIC (Page 18):
- a. The Traffic Impact Analysis Report that is being prepared for inclusion in the draft EIS should discuss the impact of the proposed project on the roadway network providing access to the project sile. This analysis should include a phasing plan that includes the completion of various planned roadway projects, such as North-South Road from the H-1 Freeway to Kapolei Parkway, Farrington Highway from the west side of the campus to Fort Weaver Road, and Kapolei Parkway from the Villages of Kapolei to Occan Pointe, with the planned phasing of the proposed project. The west connection of Road E to Farrington Highway as shown on Figures
 - b. The west connection of Road E to Farrington Highway as shown on Figures 3A and 3B should be re-evaluated as it appears to be adjacent to the golf course entrance.
- The streets that will be dedicated to the City and County of Honolulu should be identified.
 - The first sentence in the second paragraph should be revised to read that "Oahu Transit Services is contracted by the Department of Transportation Services to operate TheBus on a schedule..."
- e. The second sentence in the second paragraph should also identify Route 40 buses as buses traveling from Makaha.
- The third sentence in the first paragraph of Section 5.10 INFRASTRUCTURE (Page 24) should be revised to clarify that the portion of Kapolei Parkway through the Ewa Villages development will be constructed by the City.
- Section 5.11.6 Public Transportation (Page 28) should be renamed "Public Transit".
 Our comments regarding this section are as follows:
- The first sentence in the first paragraph should be revised to read that "Oahu Transit Services is contracted by the Department of Transportation Services to operate TheBus on a schedule"
- The second sentence in the first paragraph should also identify Route 40 buses as buses traveling from Makaha.
 In addition to the traffic innearts of the anticipated innacts of a
 - c. In addition to the traffic impacts of the project, the anticipated impacts (i.e., will project require additional buses, etc. and the fiscal impact on the transit system) on the local, collector and regional transit systems should be discussed and proposed mitigation measures described.
 - d. We look forward to working with UH officials in helping to create a transit and pedestrian friendly environment from a regional as well as a local level.

Mr. Vincent Shigekuni Page 3 March 11, 2005

- e. The draft EIS should identify transit stops and/or transit centers. The possibility of locating a transit center on campus should be considered.
 - f. The street system for the proposed development should be adequately designed to accommodate the movement of a 60-foot bus.
- We suggest that the roadway recommendations being proposed in the "Draft Ewa Smart Growth Design Code" be reviewed.

In order to facilitate our review of the EIS, please provide us with two copies of the document. Should you have any questions regarding these comments, please contact Faith Miyamoto of the Transportation Planning Division at 527-6976.

Sincerely,

EDWARD Y. HIƘATA Acting Director 2-UNAL

cc: Ms. Genevieve Salmonson, Director Office of Environmental Quality Control

Ms. Jan Yokota UH – Office of Capital Improvements

June 16, 2006	Melvin N. Kaku SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) ENVIRONMENTAL IMPACT STATEMENT PREPARATION NOTICE (EISPN) June 16, 2006 Pare 2
Mr. Melvin N. Kaku, Director Department of Transportation Services City and County of Honolulu 650 South King Street, 3 rd Floor	 c. Roads within the UH West O'ahu Lands are planned to be controlled by the University. Roads within the Private Development Lands are planned to be dedicated to the City and County of Honolulu.
Honolutu, Hawat'i 96813 Attn: Ms. Faith Miyamoto	d. The Draft EIS will include the following: Oatin Transit Services is contracted by the City Department of Transportation Services (DTS) to operate TheBus on a schedule subject to the availability of resources.
SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) ENVIRONMENTAL IMPACT STATEMENT PREPARATION NOTICE (EISPN)	 The Draft EIS will include the following: In Kapolei, service to Honolulu is provided by buses traveling from Mākaha (Routes C, 40, 93, and 93A), passing on Farrington Highway near the commus.
Dear Mr. Kaku: Since filing the 11H West O'alm FTCDN in Tammer 2005 the Trainwaith the advect	7. The Draft EIS will note that the portion of Kapolei Parkway through 'Ewa Villages will be constructed by the City.
development partner to purchase approximately 287 acres of land within the approximately 500-acre property. The University will own the remaining 214 acres of	8. The section titled "Public Transportation" will be renamed "Public Transit" in the Draft EIS.
land. The Private Development Lands will serve as revenue-producing lands and the profits (sale income) from these lands will be used to construct the first mass of the	a. See 6d above.
campus.	b. See 6e above.
Thank you for your letter dated March 11, 2005 (reference number TP2/05-93213R). We have reviewed your letter and offer the following responses to your comments:	c. The Draft EIS will discuss the anticipated traffic impacts of the proposed project on local, collector, and regional transit systems. Proposed mitigation measures will also be identified.
 The EISPN stated that the property is 500.327 acres, and this was confirmed by the City Department of Planning and Permitting. 	d. The DTS will continue to be consulted in the environmental impact statement process to create a transit- and pedestrian-friendly environment.
2. The currently proposed land uses within the property will be shown in a table and figure to be included in the Draft EIS.	e. Two transit stops are desired along North-South Road. The location of these transit stops will be identified in the Draft EIS.
3. The master plan has changed and revised figures will be included in the Draft EIS.	f. The major campus roads will be designed to accommodate the turning radii required for buses.
4. In the current master plan, the proposed elementary school is located away from North-South Road and near proposed residential areas and the proposed Department of Hawaiian Home Lands residential subdivision to the south.	9. The <i>Draft Ewa Smart Growth Design Code</i> has been reviewed and applicable standards will be incorporated into the design of the campus to the extent possible.
5. The Draft EIS will include a table listing all land use acreages, including proposed roadways.	Thank you again for your participation in the Environmental Impact Statement process. If you have any questions regarding this project, please do not hesitate to contact me at 521-5631.
 a. The TIAR will discuss the impact of the project on the roadway network providing 	Sincerely,
access to the site. This analysis will include a phasing plan (initial development and ultimate development) for the project as well as for various planned roadway projects.	PBR HAWAII Vuized R. Shift
b. The master plan has changed and the Golf Course Expansion area is no longer proposed. The current master plan will be included in the Draft EIS.	Vincent Shigekuni Vice President
	cc Gene Awakuni/UH West O'ahu 0.0008131322.051EISUbnf EISEISPN Comments-Responses/BL-13 DTS response.doe

RAYMOND T. HIGA, ASLA Senior Associate KEVIN NISIIIKAWA, ASLA Associate

TOM SCHNELL, AJCP SEVIOR ASSOCIATE

GRANT MURAKAMI, AICP Principal

VINCENT SHIGEKUNI VICE PRESIDENT

PBR

WM, FRANK BRANDT, FASLA Chairman

LAND PLANNING LANDSCAPE ARCHITECTURE ENVIRONMENTAL STUDIES

TROMAS S. WITTEN, ASLA PRESIDENT R. STAN DUNCAN, ASLA EXECUTIVE VICE-PRESIDENT RUSSELL Y.J. CHUNG, ASLA EXECUTIVE VICE-PRESIDENT

KIMI MIKAMI YUEN Associate

HONOLULU OPFICE (10) Bisitos STRAET ASB TOWER, SUTTE 650 HONOLULI, HAWATI 96813-3484 TEL: (808) 323-1402 FAX: (808) 523-1402 FAX: (808) 523-1402

SCOTT MURAKAMI ASSOCIATE Scott Abrigo Associate

WAILURU OFFICE 1787 WILP A LOOP, SUTE 4 WAILURU, HAWALT 9/579-1271 TEL: (508) 242-2578 FAX: (808) 242-2502

HLO OFFICE 101 AUFUNI STREET 41:0 LAOGON CHOFTER, SUITE 310 HLO, H.WMA(1 95720-4262 Th.2, (BOB) 961-3333 FAX: (802) 961-3399

June 16, 2006	Mr. Kenneth Silva, Fire Chief	Fire Department City and County of Honolulu 3375 Koapaka Street, Suite H425 Honolulu, Hawai 1, 96819-1869	SUBJECT: UNIVERSITY OF HAWAI'I WEST OAHU (UH WEST O'AHU)	ENVIRONMENTAL IMPACT STATEMENT PREPARATION NOTICE (EISPN)	Dear Chief Silva:	Since filing the UH West O'ahu EISPN in January 2005, the University has selected a	development partner to purchase approximately 2677 acres of land within the approximately 500-acre property. The University will own the remaining 214 acres of land. The Private Development I ands will serve as revenue-workneine lands and the	profits (sale income) from these lands will be used to construct the first phase of the campus.	Thank you for your letter dated February 25, 2005. We acknowledge that the Fire	Department has no comments in addition to those provided in a letter dated January 10, 2003.	Thank you again for your participation in the Environmental Impact Statement process. If you have any questions regarding this project, please do not hestlate to contact me at $\frac{2}{2}$	Sincerely,	PBR HAWAII	Vin-B R. Shingh- Vincent Shigekuni Vice President	cc Gene Awakuni/I.H West O'ahu	O:UOB13/1322.05/EIS/Draft EIS/EISFN Comments-Responses/BL-14 HFD response.doc	
PBR	LAND FLANNING LANDSCAFE ARCHITECTURE ENVIRONATERTAL STUDIES	W.M. FRANK BRANDT, FASLA <i>Giarbany</i>	THOMAS S. WITTEN, ASLA President	R. STAN DUNCAN, ASLA Executive Vice-President	RUSSELL Y.J. CHUNG, ASLA EXECUTIVE VICE-PRESIDENT	VINCENT SHIDEKUNI VICE PRESIDENT	Grant Murakami, AICP Principal	Tom Schnell, AICP Senior Associate	RAYMOND T. HIGA, ASLA Senide Associate	KEVIN NISHIKAWA, ASLA ASSOCIATE	KIMI MIKAMI YUEN ASSOCIATE SCOTT ABRIGO	ASSOCIATE Scott murakamı Associate		II ONOLULU OFFICE 1001 BISING STREET ASD TOWN, SUME 559 HOROLUL, HAWAN 195813-348 TEL: (508) 521-4631 E-AAUL: 9574dinb @pthwwii.com		HLO OFFICE (10) ANTONE STREET HLIC LACONC FORTS, SUITE 310 HLIC, HAMAN 195720-322 TEL: (808) 961-3233 FAX: (808) 961-4939	WALUKU OPFICE 1787 WILPA, LOOP, SUTTE 4 WALLUKU, JAAWAT' 95793-1271 TEL; (6093 2422-3573 FAX: (8093 2422-3572
FIRE DEPARTMENT CITY AND COUNTY OF HONOLULU 3735 KOMMA STREET, SUITE 1425 + 1000ULUL HAMIN 9619-1069 TELEPHONE, (3001 831-7761 + 6AX, (3003 831-7762 + 10751 HAMIN 9619-1069 TELEPHONE, (3001 831-7761 + 6AX, (3003 831-7762 + 10751 HAMIN 9619-1069 TELEPHONE, (3001 831-7761 + 6AX, (3003 831-7762 + 10751 HAMIN 9619-1069 TELEPHONE, (3001 831-7761 + 6AX, (3003 831-7762 + 10751 HAMIN 9619-1069 TELEPHONE, (3001 831-7761 + 6AX, (3003 831-7762 + 10751 HAMIN 9619-1069 TELEPHONE, (3001 831-7761 + 6AX, (3003 831-7762 + 10751 HAMIN 9619-1069 TELEPHONE, (3001 831-7761 + 6AX, (3003 831-7762 + 10751 HAMIN 9619-1069 TELEPHONE, (3001 831-7761 + 6AX, (3003 831-7762 + 10751 HAMIN 9619-1069 TELEPHONE, (3001 831-7761 + 6AX, (3003 831-7762 + 10751 HAMIN 9619-1069 TELEPHONE, (3001 831-7761 + 6AX, (3003 831-7762 + 10751 HAMIN 9619-1069 TELEPHONE, (3001 831-7761 + 6AX, (3003 831-7762 + 10751 HAMIN 9619-1069 TELEPHONE, (3001 831-7761 + 6AX, (3003 831-7762 + 10751 HAMIN 9761 HAMIN 9771 HAMIN 9771 HAMIN 9761 HAMIN 9761 HAMIN 9761 HAMIN 9761 HAMIN 9761 HAMIN 9771 HAMIN 9761 HAMIN 9771 HAMIN 9761 HAMIN 9771 HAMIN 9761 HAMIN 9771 HAMIN 97	ATTLO K. LEOMBO	JOHN CLARK DOPUTY FIRE CHIEF	February 25, 2005		t Shigekuni, Principal ii	avings Bank Tower pricer, Suite 650	Hawaii 96813 higekuni:	invironmental Impact Statement Preparation Notice Jniversity of Hawaii - West Oahu	Tax Map Keys: 9-1-016: 120, 127, and 129	d your letter dated January 31, 2005, requesting our comments on the above-mentioned	ulu Fire Department has no additional comments. Please refer to our letter of January 10, revious comments.	have any questions, please call Battalion Chief Lloyd Rogers of our Fire Prevention Bureau at	Sincerely,	ATTLICK. LEONARDI Fire Chief	-	of Environmental Quality Control rsity of Hawaii, Office of Capital Improvements	
	MUFI HANNEMANN MAYDR			;	Mr. Vince PBR Haw	American 1001 Bish	Honolulu, Dear Mr. !	Subject:		We receiv subject.	The Hono 2003, for J	Should yo 831-7778.			AKL/SY:	cc: Offic Univ	

June 16, 2006		Mr. Boisse P. Correa, Chief of Police Police Department	City and County of Honolulu 801 South Breteriania Street 17		Auti: MII. Mari Godsey, Assistant Chief of Police, Support Services Bureau	SUBJECT: UNIVERSITY OF HAWAI'I WEST OAHU (UH WEST O'AHU) ENVIRONMENTAL IMPACT STATEMENT PREPARATION	NOTICE (EISPN)	Dear Chief Correa:	Since filing the UH West O'ahu EISPN in January 2005, the University has selected a development partner to purchase approximately 287 acres of land within the	approximately 500-acre property. The University will own the remaining 214 acres of land The Drivete Development Londo will occurs of common 2014 acres of	profits (sale income) from these lands will be used to construct the first phase of the	cutifus.	Thank you for your letter dated March 2, 2005 (your reference CS-KP). We acknowledge that the Police Department believes that calls for service to the area will be	impacted and there will be vehicular and traffic concerns. These issues will be addressed in the Draft EIS. We understand that the HPD may have more specific concerns as the plans for the UH West O'ahu campus and residential and commercial areas within the	property are developed.	Thank you again for your participation in the Environmental Impact Statement process. If you have any questions resarding this molect please do not besitate to contact me at	521-5631.	Sincerely,	PBR HAWAII		Nuiezo R. Huge-	Vincent Shigekuni Vice President	cc Gene Awakuni/UH West O'ahu o/10B1311322.05EIStDraft EIStEISPN Comments-Responses/BL-15 HPD response.doc
PBR	LAND FLANRO LANDSCAPE ARCHTECTURE ENVIROPALENTAL STUDIES		WM. FRANK BRANDT, FASLA Chairaan	THOMAS S. WITTEN, ASLA President	R. STAN DUNCAN, ASLA EXECUTIVE VICE-PRESIDENT	RUSSELL Y.J. CHUNG, ASLA EXECUTIVE VICE-PRESIDENT	VINCENT SHIGEKUNI VICE PRESIDENT	GRANT MURAKAMI, AICP PRINCIPAL		TOM SCHNELL, AJCP SENIOR ASSOCIATE	Raymond T, Higa, ASLA Senidr Associate	KEVIN NISIIIKAWA, ASLA Associate	KIMI MIKAMI YUEN ASSOCIATE	SCOTT ABRIGO ASSOCIATE SCOTT MIJBAY AMI	ASSOCIATE		Hovolucia OFFICE	1001 BISHOP STREET ASB TOWER, SUITE 650 HONGLULL, HAWAI' 9681 3-3484 TEL: (808) 521-5631	EAX: (808) 523-1402 E-MARC: sysadmin@pbrhawaii.com	Tto a America	JILLO DYFICE 101 AUPUNI STREET HILD LAGOON CONTER, SUITE 310 HILD LAGOON CONTER, SUITE 310	Fax: (808) 961-4939 Fax: (808) 961-4939	WAILJARU OFFICE: 1787 WILT'A, LOOR, SHITE'A WALLUKC, HANNY 19579-11271 THL: (6603) 342-3578 FAX: (8693) 342-3592
CITY AND COUNTY OF HONOLULU BOI SOUTH BERETANIA STREET HONOLULU, HAWAII 96813 - AREA CODE (808) 529-3111 http://www.honoluluc.gov www.honolulu.gov	MUEI INNIVERANN NAVOR DATA CORREA		OUR REFERENCE CS-KP	March 2, 2005		Mr. Vincent Shigekuni	PBR Hawaii ASB Tower, Suite 650	1001 Bishop Street Honolutu, Hawaii 96813	Dear Mr. Shigekuni:	Thank vou for the opportunity to review and comment on the Environmental Immost	Statement Preparation Notice for the University of Hawaii West Oahu.	As we stated in our response to the Draft Environmental Assessment/Environmental Impact	outerinant worke or interparation for this project, we believe that calls for service to the area will be impacted and that there will be vehicular and pedestrian traffic concerns.	We will probably have more specific concerns as additional detailed plans for the University of Hawaii campus and the residential and commercial areas are developed.	If there are any questions, please call Major Michael Tamashiro of District 8 at 692-4253 or Me. Carol Scienting of the Science Science District at ano or of the science o	MS. Carol Souelarin of the Support Services Bureau at 528-3658.	Sincerely,	BOISSE P. CORREA Chief of Police		By hall lang	KARL GODSEY	Support Services Bureau	cc: Ms. Genevieve Salmonson, OEQC Ms. Jan Yokota, University of Hawaii

Serving and Protecting with Aloha

9.0 List of Preparers

9.0 LIST OF PREPARERS

This EIS was prepared by PBR HAWAII, 1001 Bishop Street, ASB Tower, Suite 650, Honolulu, Hawai'i 96813. The staff involved in the preparation of this document is listed below.

W. Frank Brandt, FASLA Grant Murakami Vincent Shigekuni <u>Michael Shibata</u> Lacey Kazama Christine Chaplin Vivaswan Verawudh Chris Chavez Kanai'a Nakamura Dionne Talia Principal-in-Charge Principal Campus Planner EIS Project Manager <u>Planner Researcher and Writer</u> Planner Researcher and Writer Planner/GIS Analyst Planner/GIS Analyst Graphic Designer Cultural Impact Advisor Administrative Assistant

Several key technical consultants were employed to provide specific assessments of environmental factors for this project. These consultants, their company affiliation, and their specialty are listed below.

Name	Firm	Area of Expertise
Phillip L. Bruner		Faunal Studies
Winona P. Char	Char & Associates	Botanical Studies
Dana Dorsch	D.L. Adams Associates, Ltd.	Acoustical Studies
David Miyamoto	Engineering Concepts, Inc.	Civil Engineering
Barry D. Neal	B. D. Neal	Air Quality Studies
Wayne Yoshioka	Parsons Brinckerhoff Quade & Douglas	Traffic Engineering

(This page intentionally left blank.)

10.0 GLOSSARY

10.0 GLOSSARY

The following is a list of terms, abbreviations, and acronyms used in the EIS.

A	
AAQS	Ambient Air Quality Standards
ADA	Americans with Disabilities Act
ALISH	Agricultural Lands of Importance to the State of Hawai'i
ASTM	American Society for Testing and Materials
В	
BLNR	State of Hawai'i Board of Land and Natural Resources
BMP	Best Management Practice
BOE	State of Hawai'i Board of Education
BPNAS	Barbers Point Naval Air Station
BWS	City and County of Honolulu Board of Water Supply
C	
CFRC-NFRAP	CERCLIS No Further Remedial Action Planned
CERCLIS	Comprehensive Environmental Response. Compensation, and
CERCEIC	Liability Information System
cfs	cubic feet per second
City	City and County of Honolulu
ĊŴŔM	State of Hawai'i Department of Land and Natural Resources
	Commission on Water Resource Management
CZM	State of Hawai'i Coastal Zone Management Program
D	
DA	United States Department of the Army
dBA	A-weighted decibels
DBEDT	State of Hawai'i Department of Business, Economic Development,
	and Tourism
DEA	Draft Environmental Assessment
DEIS	Draft Environmental Impact Statement
DHHL	State of Hawai'i Department of Hawaiian Home Lands
DLNR	State of Hawai'i Department of Land and Natural Resources
DOE	State of Hawai'i Department of Education
DOH	State of Hawai'i Department of Health
DOT	State of Hawai'i Department of Transportation
DPA	Development Plan Area
DPR	City and County of Honolulu Department of Parks and Recreation
DPP	City and County of Honolulu Department of Planning and
	Permitting
DTS	City and County of Honolulu Department of Transportation Services
E	
EA	Environmental Assessment
EaB	Ewa Silty Clay Loam, 3 to 6 percent slopes
EDR	Environmental Data Resources, Inc.

UNIVERSITY OF HAWAI'I WEST O'AHU

FINAL DRAFT ENVIRONMENTAL IMPACT STATEMENT

	EIS	Environmental Impact Statement						
	EISPN	Environmental Impact Statement Preparation Notice						
	EMS	Emergency Medical Services						
	EPA	Environmental Protection Agency						
	ERNS	Emergency Response Notification System						
	Ewa DP	Ewa Development Plan						
	Ewa DPA	Ewa Development Plan Area						
	EwC	Ewa Stony Silty Clay, 6 to 12 percent slopes						
F								
•	FFA	Final Environmental Assessment						
	FEIS	Final Environmental Impact Statement						
		Fodoral Emergency Management Agency						
		Federal Highway Administration						
		Federal Index System						
		Federal Index System						
	FIRM	Flood insurance Kale Map						
	FONSI	Finding of No Significant impact						
~	FIA	Federal Transit Administration						
G								
	GPD	gallons per day						
	gpm	gallons per minute						
Н								
	HAR	Hawaii Administrative Rules						
	НСДСН	State of Hawai'i Housing and Community Development						
		Corporation of Hawaii						
	HCDCH Fast Kapolei FIS	East Kapolei Master Plan Einal Environmental Impact Statement						
	HCP	Habitat Conservation Plan for Abutilon menziesii at Kapolei						
	HECO	Hawaiian Electric Company, Inc.						
	HED	City and County of Honolulu Fire Department						
		Honolulu High Conscitu Transit Project						
		City and County of Honoluly Police Department						
		Honolulu Program of Weste Energy Receivery						
		Homolulu Program of Waste Energy Recovery						
		Hawaii Keviseu Statutes						
		U.s. Department of Housing and Orban Development						
		Honouliuli Vvater Recycling Facility						
		Honouliuli Clay, 0 to 2 percent slopes						
	НХВ	Honouliuli Clay, 2 to 6 percent slopes						
J								
	JTS	Joint Tactical Support						
Κ								
••	KARP	Kanolei Area Rikeway Plan						
	KV/	kilo-volt						
		Kilo-volt						
L								
		Leeward Community College						
	L _{dn}	Day-night equivalent sound level measured in dBA						
	LEED	Leadership in Energy and Environmental Design						
	LOS	Level of Service						

UNIVERSITY OF HAWAI'I WEST O'AHU Final Draft Environmental Impact Statement

-		
	LPA	Locally Preferred Alternative
	LRDP	University of Hawai'i West O'ahu Long Range Development Plan
		(2004)
	LRDP Update	University of Hawai'i West O'abu Long Range Development Plan
	ener opulie	Undate (2006)
		State of Hawai'i Land Use Commission
		City and County of Honolulu Land Lise Ordinance
		Looking Underground Storogo Tonk
		Leaking Onderground Storage Talik
N	1	
	MG	million gallons
	mg/m³	milligrams per cubic meter
	MGD	million gallons per day
	MOA	Memorandum of Agreement
	mph	miles per hour
	msl	mean sea level
	MVA	megavolt ampere
	MW	mega-watt
N	l	0
		No Eurthor Pomodial Action Plannod
		National Bollutant Discharge Elimination System
		National Political Discharge Emmination System
~	INPL	National Priorities List
U		
	OEQC	State of Hawai'i Office of Environmental Quality Control
	OMPO	Oahu Metropolitan Planning Organization
	OR&L	Oahu Railway and Land Company
	ORTP	Oahu Regional Transportation Plan
	OSCo	Oahu Sugar Company, Ltd.
Р		
•	Private Development Lands	Lands controlled by private development partner
	PRI 1	Plan Review Lise
		Primary Urban Contor
п	100	Thinkiy Orban Center
К		
	RCRIS-SQG	Resource Conservation and Recovery Information System-Small
		Quantity Generator
	RECs	Recognized Environmental Conditions
S		
	SHPD	State of Hawai'i Department of Land and Natural Resources Historic
		Preservation Division
	SHW/S	State Hazardous Waste Site
	Site Selection Study	University of Hawai'i West O'abu Campus Site Selection Study
	SMA	Special Management Area
	SORA	Southorn O'abu Basal Aquifor
		State or Local Poloase Natifications
	SI ILLS State	State of Howaiti
		State OFFIdWall
		Statewide Transportation Improvement Plan
	SVVLF	Sond vvaste land fill

UNIVERSITY OF HAWAI'I WEST O'AHU Final Draft Environmental Impact Statement

Т	
ТМК	Tax Map Key
U	
UH	University of Hawai'i
UH Mānoa	University of Hawai'i Mānoa
UH West Oʻahu	University of Hawai'i West O'ahu
UH West Oʻahu Lands	Lands controlled by University of Hawai'i West O'ahu
UIC	Underground Injection Control
UST	Underground Storage Tank
W	
WASC	Western Association of Schools and Colleges
WkA	Waialua Silty Clay, 0 to 3 percent slopes
WOCD	West Oahu Campus Development, LLC
WWTP	Wastewater Treatment Plant
WzA	Waipahu Silty Clay, 0 to 2 percent

11.0 References

11.0 REFERENCES

- Bauer, Glenn R. (1996) Reevaluation of the Ground-Water Resources and Sustainable Yield of the Ewa Caprock Aquifer – Final Report. Honolulu, Hawai'i. Prepared for the State
 Department of Land and Natural Resources, Commission on Water Resource Management.
- Blakeman, Karen. *Honolulu Advertiser*. "H-Power upgrade sought". 22 September 2004. Honolulu, Hawai'i. Available at:

http://the.honoluluadvertiser.com/article/2004/Sep/22/In/In14a.html (December 2004).

- Bruner, Phillip L. (2005) Avifaunal and Feral Mammal Field Survey of Lands Involved in the University of Hawaii West O'ahu. Lā'ie, Hawai'i. Prepared for PBR HAWAII.
- B.D. Neal & Associates. (2006) *Air Quality Study for the Proposed University of Hawaii West Oahu Project*. Kailua-Kona, Hawai'i. Prepared for PBR HAWAII.
- Char & Associates. (2003) *Botanical Survey University of Hawai'i, West O'ahu*. Honolulu, Hawai'i. Prepared for PBR HAWAII.
- Chiogioji, Mitchell, and Hammatt. (2004) *Cultural Impact Evaluation in Support of the North-South Road and Kapolei Parkway Project, Honouliuli Ahupua'a, 'Ewa District, O'ahu.* Honolulu, Hawai'i. Prepared for Parsons, Brinkerhoff, Quade & Douglas, Inc.
- City and County of Honolulu Department of Planning and Permitting. (2004) Annual Report on the Status of Land Use on Oahu: Fiscal Year 2003. Honolulu, Hawai'i. Available at: http://honoluludpp.org/planning/dpar2/dpar2003.pdf (May 2005).
- City and County of Honolulu Department of Planning and Permitting. (Revised 2000) *Ewa Development Plan*. Honolulu, Hawai'i. Available at: <u>http://www.honoluludpp.org/Planning/DevSust_Ewa.asp</u> (December 2004).
- <u>City and County of Honolulu Department of Public Works. (1986)</u> *Storm Drainage Standards.* <u>Honolulu, Hawaii.</u>
- City and County of Honolulu Department of Land Utilization. (1987) *Coastal View Study*. Honolulu, Hawai'i.
- City and County of Honolulu Department of Transportation Services. (2006) Honolulu On The Move: An Update on the Honolulu High-Capacity Transit Corridor Project. May/June 2006 Newsletter. Honolulu, Hawai'i.
- City and County of Honolulu Department of Transportation Services. (2006) *Honolulu High-Capacity Transit Corridor Project*. Honolulu, Hawai'i. Available at: <u>http://honolulutransit.org/project_overview</u> (June 2006).
- D.L. Adams Associates, Ltd. (2006) Environmental Noise Assessment Report University of

Hawaii West Oahu Campus. Kailua, Hawai'i. Prepared for the University of Hawai'i.

- Engineering Concepts, Inc. (2006) University of Hawaii West Oahu Infrastructure Study. Honolulu, Hawai'i. Prepared for PBR HAWAII.
- Federal Highway Administration. (2005) *Finding of No Significant Impact (FONSI) for North-South Road and Kapolei Parkway*. Honolulu, Hawai'i.
- Geolabs, Inc. (2002) Preliminary Assessment for Site Selection University of Hawai'i West O'ahu Campus, Kapolei Makai Alternative Site, 'Ewa, O'ahu, Hawai'i. Honolulu, Hawai'i. Letter report to Mr. Grant Murakami, AICP, PBR HAWAII.
- HPOWER. HPOWER prepares for upcoming year in turning O'ahu's trash to renewable energy. Press Release, 10 March 2004. Honolulu, Hawai'i. Available at: http://www.honoluluhpower.com/uploads/20040310B.pdf (February 2005).
- Jurik, S.P. (1998) Atlas of Hawaii: 3rd edition. University of Hawaii Press. Honolulu, Hawai'i.
- Kamakau, Samuel M. (1991) <u>Ka Po'e Kahiko: The People of Old</u>. Bishop Museum Press. Honolulu, Hawai'i.
- Kua, Crystal and Richard Borreca. *Honolulu Star-Bulletin*. "The \$2.6 billion system includes light rail and requires a tax hike." 28 October 2003. Honolulu, Hawai'i. Available at: <u>http://starbulletin.com/2003/10/28/news/story2.html</u> (December 2004).
- Kuykendall, Ralph S. (1967) <u>The Hawaiian Kingdom</u>. Volume 3. University of Hawai'i Press. Honolulu, Hawai'i.
- Mink, John F. and Stephen Lau. (1990) *Aquifer Identification and Classification for Oahu; Groundwater Protection Strategy for Hawaii*. Water Resources Research Center Technical Report No. 179.
- Mink and Yuen, Inc. (1993) *Hydrogeological Impacts Proposed Expansion of the Barbers Point* Harbor. Prepared for the State of Hawai'i Department of Transportation.
- Nakuina, Emma M. (1904) <u>Hawaii Its People and Their Legends</u>. Hawai'i Promotion Committee. Honolulu, Hawai'i.
- Parsons Brinckerhoff Quade & Douglas, Inc. (2006) *Traffic Study University of Hawaii West Oahu*. Honolulu, Hawai'i. Prepared for PBR HAWAII.
- PBR HAWAII. (2005) *DHHL East Kapolei Development Parcel B Draft Environmental Assessment*. Honolulu, Hawai'i. Prepared for the State of Hawai'i Department of Hawaiian Home Lands.
- PBR HAWAII. (2005) DHHL East Kapolei Development Parcel B Final Environmental

Assessment. Honolulu, Hawai'i. Prepared for the State of Hawai'i Department of Hawaiian Home Lands.

- PBR HAWAII. (1998) *East Kapolei Master Plan Final Environmental Impact Statement*. Honolulu, Hawai'i. Prepared for the State of Hawai'i Housing and Community Development Corporation of Hawaii.
- PBR HAWAII. (2004) *Habitat Conservation Plan for Abutilon menziesii at Kapolei*. Honolulu, Hawai'i. Prepared for the State of Hawai'i Department of Transportation.
- PBR HAWAII. (2002) University of Hawai'i West O'ahu Alternative Sites Evaluation. Honolulu, Hawai'i. Prepared for the University of Hawai'i.
- PBR HAWAII. (2002) University of Hawai'i West O'ahu Campus Site Selection Study. Honolulu, Hawai'i. Prepared for the University of Hawai'i.
- PBR HAWAII. (2004) University of Hawai'i West O'ahu Long Range Development Plan. Honolulu, Hawai'i. Prepared for the University of Hawai'i.
- PBR HAWAII, John Hara Associates Inc., David J. Neuman, FAIA in association with West O'ahu Campus Development, LLC. (2006) *University of Hawai'i West O'ahu Long Range Development Plan Update*. May Draft. Honolulu, Hawai'i. Prepared for the University of Hawai'i.
- PBR HAWAII. (2005) University of Hawai'i West O'ahu Environmental Impact Statement Preparation Notice. Honolulu, Hawai'i. Prepared for the University of Hawai'i.
- State of Hawai'i Department of Agriculture. (1977) *Agricultural Lands of Importance to the State of Hawaii*. Honolulu, Hawai'i.
- State of Hawai'i Department of Business, Economic Development and Tourism. (1991) *The Data Book*. Honolulu, Hawai'i.
- State of Hawai'i Department of Business, Economic Development and Tourism (1993) *Hawai'i Model Energy Code*. Honolulu, Hawai'i.
- State of Hawai'i Department of Business, Economic Development and Tourism Office of Planning. (1989) *The Hawaii State Plan*. Honolulu, Hawai'i.
- State of Hawai'i Department of Education. (2006) *School Status and Improvement Report School Year 2004-05*. Available at: <u>http://arch.k12.hi.us/school/ssir/2005/leeward.html</u> (June 2006).
- State of Hawai'i Department of Education. (2005) "Public School 2005-06 Enrollment Slightly Lower." 2005-06 Official Enrollment table from news release by Greg Knudsen, 15 September 2005. Available at:

UNIVERSITY OF HAWAI'I WEST O'AHU Final Draft Environmental Impact Statement

http://lilinote.k12.hi.us/STATE/COMM/DOEPRESS.NSF/a1d7af052e94dd120a2561f7000a0 37c/4ca2735ab6a02f4e0a25707d0067d092?OpenDocument (June 2006).

- State of Hawai'i Department of Education. (2004) "Enrollment Count: Leeward Area and Charters Continue to Grow." 2004-05 Official Enrollment table from news release by Greg Knudsen, 16 September 2004. Available at: <u>http://lilinote.k12.hi.us/STATE/COMM/DOEPRESS.NSF/0/67969154fb48f1430a256f110029</u> <u>c57d</u> (November 2004).
- State of Hawai'i Department of Transportation. (1991) *State Transportation Functional Plan*. Honolulu, Hawai'i.
- State of Hawai'i Department of Transportation and City and County of Honolulu Department of Transportation Services. (2004) North South Road and Kapolei Parkway Final Environmental Assessment/FONSI. Honolulu, Hawai'i.
- State of Hawai'i Department of Transportation and City and County of Honolulu Department of Transportation Services. (2004) North South Road and Kapolei Parkway Revised Draft Environmental Assessment. Honolulu, Hawai'i.
- State of Hawai'i Office of Hawaiian Affairs. Letter to Mr. Vincent Shigekuni regarding DHHL East Kapolei Development Parcel B, Kapolei, Honouliuli, 'Ewa District, O'ahu, Hawaii, TMK: (1) 9-1-016: 108 (portion). 6 May 2005.
- State of Hawai'i University of Hawai'i Land Study Bureau. (1963) *Detailed Land Classification, Island of Oahu*. Honolulu, Hawai'i.
- Sterling, Elspeth P., and Summers, Catherine. (1978) <u>Sites of Oahu</u>. Bernice P. Bishop Museum. Honolulu, Hawai'i.
- Yuen, George A. L. and Associates, Inc. (1989) *Groundwater Resources and Sustainable Yield Ewa Plain Caprock Aquifer*. Honolulu, Hawai'i.
- United States Department of Agriculture Soil Conservation Service. (1972) *Islands of Kauai, Oahu, Maui, Molokai, and Lanai, State of Hawaii.* Honolulu, Hawai'i.

Thrum, Thomas G. (1923) More Hawaiian Folk Tales. A.C. McClurg & Co. Chicago, Illinois.

12.0 Comments on the Draft

COMMENTS ON THE DRAFT ENVIRONMENTAL IMPACT STATEMENT AND RESPONSES

12.0 COMMENTS ON THE DRAFT ENVIRONMENTAL IMPACT STATEMENT AND RESPONSES

<u>The UH West O'ahu Draft Environmental Impact Statement was sent to the following agencies, organizations and individuals. The public comment period on the Draft EIS was from July 8, 2006 to August 22, 2006. Agencies, organizations, or individuals that submitted comments on the Draft EIS are listed in bold.</u>

	AGENCY	DEIS MAIL DATE	<u>DATE OF</u> <u>COMMENTS</u>
Stat	ie	· · · · · · · · · · · · · · · · · · ·	
1	Department of Accounting and General Services	07-03-06	07-25-06
<u>2</u>	Department of Business, Economic Development and Tourism	<u>07-03-06</u>	<u></u>
<u>3</u>	Department of Business, Economic Development and Tourism – Land Use Commission	07-03-06	<u>07-25-06</u>
<u>4</u>	Department of Business, Economic Development and Tourism – Office of Planning	<u>07-03-06</u>	
<u>5</u>	Department of Business, Economic Development and Tourism – Strategic Industries Division	<u>07-03-06</u>	<u>07-12-06</u>
<u>6</u>	Department of Business, Economic Development and Tourism – Hawaii Housing Finance and Development Corporation	<u>07-03-06</u>	<u>08-17-06</u>
7	Department of Defense	07-03-06	<u></u>
8	Department of Education	07-03-06	08-10-06
9	Department of Hawaiian Home Lands	<u>07-03-06</u>	<u>07-13-06</u>
<u>10</u>	Department of Health – Environmental Planning Office	<u>07-03-06</u>	08-21-06
<u>11</u>	Department of Health – Office of Environmental Quality Control	<u>07-03-06</u>	<u>08-21-06</u>
12	Department of Land and Natural Resources	07-03-06	<u></u>
<u>13</u>	Department of Land and Natural Resources – Historic Preservation Division	07-03-06	<u>08-08-06</u>
14	Department of Transportation	07-03-06	08-22-06
<u>15</u>	Department of Transportation – Airports Division	<u>07-03-06</u>	
16	Office of Hawaiian Affairs	07-03-06	07-31-06
<u>17</u>	State Legislature – Representative Mark Moses	07-03-06	
<u>18</u>	State Legislature – Representative Kymberly Pine	<u>07-03-06</u>	<u></u>
<u>19</u>	<u> State Legislature – Senator Brian Kanno</u>	<u>07-03-06</u>	<u></u>
<u>20</u>	<u> State Legislature – Senator Will Espero</u>	<u>07-03-06</u>	<u>08-31-06</u>
<u>21</u>	<u> State Legislature – Representative Rida Cabanilla</u>	<u>07-03-06</u>	<u>07-26-06</u>
22	University of Hawai'i – Environmental Center	07-03-06	08-21-06

Table 12.1 – Draft EIS Comment Letters

UNIVERSITY OF HAWAI'I WEST O'AHU Final Draft Environmental Impact Statement

	<u>AGENCY</u>	<u>DEIS</u> MAIL DATE	DATE OF COMMENTS
23	University of Hawai'i Water Resources Research Center	07-03-06	
24	Hawai'i State Library	07-03-06	
25	'Ewa Beach Public and School Library	07-03-06	
<u>26</u>	Kapolei Public Library	07-03-06	
<u>27</u>	Pearl City Regional Library	07-03-06	
28	Kaimukī Regional Library	07-03-06	
29	Kaneohe Regional Library	07-03-06	
<u>30</u>	Hilo Regional Library	07-03-06	
31	<u>Kahului Public Library</u>	07-03-06	
32	Lihue Regional Library	07-03-06	
33	<u>University of Hawai'i – Hamilton Library</u>	07-03-06	
<u>34</u>	<u>Department of Business, Economic Development and</u> Tourism Library	07-03-06	
35	Legislative Reference Bureau	07-03-06	
<u>36</u>	City and County of Honolulu Department of Customer Services Library (formerly the Municipal Reference and Records Center)	07-03-06	
Fed	eral		
37	Department of the Army – Army Engineer District	07-03-06	07-27-06
<u>38</u>	Department of the Interior – Fish and Wildlife Service	07-03-06	<u></u>
<u>39</u>	Federal Legislature – Senator Daniel Inouye	0/-03-06	
40	Federal Legislature – Senator Daniel Akaka	07-03-06	
<u>41</u>	<u>Federal Legislature – Kepresentative Neil Abercrombie</u>	0/-03-06	
<u>City</u>		07.02.00	07.24.00
<u>42</u>	Board of Water Supply	07-03-06	07-24-06
<u>43</u>	County Legislature – Mayor Muti Hannemann	07-03-06	
44	City Council – Councilmember Todd Apo	07-03-06	
<u>45</u>	Department of Community Services	0/-03-06	08-02-06
<u>46</u>	Department of Design and Construction	0/-03-06	08-22-06
<u>4/</u>	Department of Environmental Services	0/-03-06	
<u>48</u>	Department of Facility Maintenance	0/-03-06	08-21-06
<u>49</u>	Department of Parks and Recreation	<u>07-03-06</u>	07-14-06
<u>50</u>	Department of Planning and Permitting	<u>07-03-06</u>	<u>08-21-06,</u> <u>08-22-06</u>
<u>51</u>	Department of Transportation Services	<u>07-03-06</u>	<u>08-28-06</u>
<u>52</u>	<u>'Ewa Neighborhood Board, No. 23</u>	<u>07-03-06</u>	
53	Fire Department	<u>07-03-06</u>	<u>07-18-06</u>
<u>54</u>	Makakilo/Kapolei Neighborhood Board, No. 34	<u>07-03-06</u>	
55	Police Department	<u>07-03-06</u>	
56	<u>Wai'anae Neighborhood Board, No. 24</u>	07-03-06	
Oth	er Organizations/Individuals		

UNIVERSITY OF HAWAI'I WEST O'AHU

Final Draft Environmental Impact Statement

	<u>AGENCY</u>	DEIS MAIL DATE	DATE OF COMMENTS
57	<u>Hawaiian Electric Company, Inc.</u>	<u>07-03-06</u>	08-22-06
58	Hawaiian Telcom	<u>07-03-06</u>	<u></u>
<u>59</u>	Palehua Community Association	<u>07-03-06</u>	
<u>60</u>	The Estate of James Campbell	07-03-06	08-22-06
<u>61</u>	The Gentry Companies	<u>07-03-06</u>	<u></u>
<u>62</u>	D.R. Horton - Schuler Division	<u>07-03-06</u>	08-07-06
<u>63</u>	HASEKO (Ewa), Inc.	<u>07-03-06</u>	08-21-06
64	Kapolei Golf Course	07-03-06	<u></u>
65	The Honolulu Advertiser	07-03-06	
66	The Honolulu Star-Bulletin	07-03-06	<u></u>

R HAWAII Ociates, inc.	Mr. Ernest Y. W. Lau, Public Works Administrator State of Hawai'i Department of Accounting and General Services	Honolulu, Hawai'i 96810 SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS)	Dear Mr. Lau: Thank you for your letter dated July 25, 2006. We acknowledge that this project will not impact any of the Department of Accounting and General Services' projects or existing facilities, and therefore, the department has no comments to offer.	Thank you again for your participation in the Environmental Impact Statement process for this project. Your letter will be included in the Final EIS under Section 12.0.	If you have any questions regarding this project, please do not hesitate to contact me at 521-5631.	PBR HAWAII	Vincent R. Shigekuni Vice President cc: Office of Environmental Quality Control Gene Awakuni/UFI West O'ahu	O:VOB13N1322.09ELSVFinal Responses to DEIS Comments/BL-06 DAGS response 11-20.doc	
PB ans	W. FRANKI, IRANDT, FASIA Charnum THEMASS, WITTES, ASIA Predident K-STANDANSA, MA	RESALTATION FOR A CONTRACT AND A CON	Penergust TOM SCHNELL-AICP Senior Associate RANNOND T.HIDA, SELA Senior Avoiente	KEVIN K, NISHIKAWA, ASI.A Avstrate	Associates Annual FULS, 19412 AP Associates Scorr Actists Abbulgo Associates Coversition 1 Annual FUL Acti	Associate	HONOLUZU UFFICE (OD) Biology Vece ASI Frace, Safre (SA) Browch, Safre (SA) Browch, Safre (SA) Browch, Safre (SA) Frace, Safre (SA) Frace, Safre (SA) Frace, Safre (SA)	HLO OFFICE III AppenSivert IIIA AppenSivert Correcting IIIA Appendition Correcting IIIA Appendition Correcting IIIA Appendition Appendition IIIA Appendition II	
RUSS K. SATTO OQUPTROLLER KATHERINE H. THOMASON DEPUTY COMPTROLLER	3.147.6					loes not impact any g facilities, and we	t of the Planning	IECEIVED . JUL 26 2006 BR HAWAII	
-	STATE OF HAWAII DEPARTMENT OF ACCOUNTING AND GENERAL SERVICES P.O. BOX 119, HONOLULU, HAWAII 96810	JUL 2 5 2006	ihigekuni, Vice President itreet, Suite 650 waii 96840	gekuni:	University of Hawaii West Oahu Draft Environmental Impact Statement TMK: 9-1-016: 120, 127, and 129	the opportunity to review the subject project. This project on the of Accounting and General Services' projects or existing tents to offer at this time.	oquestions, please have your staff call Mr. Allen Yamanoha 0488. Sinccrely,	Public Works Administrator Public Works Administrator Parevieve Salmonson, OEQC	¢
LINDA LINGLE Governor			Mr. Vincent ? PBR Hawaii 1001 Bishop ? Honolulu, Hay	Dear Mr. Shi	Subject:	Thank you for of the Depart have no comm	If you have ar Branch at 586	AY,mo c: Mís, Gi	

•

PLANNING + LANDACAPE ARCHITECTHRE + ENVIRONMENTAL STUBLES + ENTIFIEMENTS PERMITTING + GRAPHIC DESIGN

GOVERNOR



STATE OF HAWAII DEPARTMENT OF BUSINESS, ECONOMIC DEVELOPMENT & TOURISM LAND USE COMMICSION PLAND USE COMMISSION PLANGUL, Havat 98904.2139 Handulu, Havat 98904.2139 Fear 398367-3827 Fear 398367-3827 Fear 398367-3827

July 25, 2006

Mr. Vincent Shigekuni, Vice President PBR Hawaii ASB Tower, Suite 650 1001 Bishop Street Honoluku, Hawaii 96813-3429

Dear Mr. Shigekuni:

Subject: University of Hawaii-West Oahu Draft Environmental Impact Statement (DEIS)

We have reviewed the DEIS for the subject project and have the following comments:

- 1) As we pointed out in our January 21, 2003, comment letter on the Draft Environmental Assessment/Environmental Impact Statement Preparation Notice, the Land Use Commission (LUC) urbanized the project site under LUC Docket No. A99-728/Housing and Community Development Corporation of Hawaii, State of Hawaii (HCDCH). The LUC imposed 27 conditions upon the redastification pursuant to Findings of Fact, Conclusions of Law, and Decision and Order (Decision and Order) dated September 8, 1999! Given that the conditions upon the land and are binding upon the HCDCH and each and eacy subsequent owner, lesses, sub-lessee, transferee, grantee, asignee, or developer, the development of the University of Hawaii-West Oalu lands and the Private Development Lands is subject to these conditions. As such, we request that there be discussion of the proposed uses on these lands and their impacts within the context of the applicable conditions.
 - 2) As we also pointed out in the comment letter, the development of the University of Hawaii-West Oahu campus on the project site represents a significant departure to the representations that were made before the LUC in its proceedings on the abovementioned docket so as to require that a motion to amend the Decision and Order be

ANTHONY J.H. CHING EXECUTIVE OFFICER Mr. Vincent Shigekuni, Vice President July 25, 2006 Page 2 filed with our office.² To that end, we request that LUC approval of this motion be included on the list of required permits and approvals that is referenced on pages xii, 2, and 122 of the DEIS. In accordance with section 11-200-17(h), Hawaii Administrative Rules, the status of each identified permit or approval should also be described.

- 3) Chapter 6.0 fails to include the State's previous proposal to build the University of Hawaii-West Oahu campus on approximately 991 acress mauka of the H-1 Freeway as an alternative to the proposed action. Given that this location was once *the* site for the development of the campus and provided the impetus for the urbanization of 1,300 acres comprised by the East Kapolei Master Plan (EKMP), including the project site, we believe that a discussion of this alternative location is warranted.
- 4) As the DEIS correctly notes, the project site was previously planned for a range of uses under the EKMP, including two elementary schools on sites of 12 acres each. A third elementary school on 12 acres within the EKMP site was planned to the east of the proposed North-South Road. We note that under the current Conceptual Land Use Plan for the project site, only one elementary school on 12 acres is proposed. Clarification should be provided as to whether this single elementary school and the one proposed on Department of Hawiian Home Lands property outside of the project site will adequately accommodate the student population expected from the residential development on the Private Development Lands and surrounding residential development.

We have no further comments to offer at this time. Thank you for the opportunity to comment on the subject DEIS. Should you have any questions, please feel free to call me or Bert Saruwatari of our office at 587-3822.

ANTHONY J. Η. Φ Executive Office incerely,

c: Office of Environmental Quality Control

¹ The LUC subsequently amended Condition No. 3 pursuant to Order Granting Pelitioner's Request To Medity Condition 3 And Findings C Fact 31, 63, 44 128 Of The Findings Of Fact Conclusions of Effav, And Decision And Order Issued By The Land Use Commission On September 8, 1995, Speciating Factorsing And Community Development Corporation Of Hawali's Pelition For Land Use District Boundary Amendment Filed March 5, 1999 dated March 5, 2000.

² We note Condition No. 22 of the Decision and Order states that "Tetitioner or londowners shall develop the Petition Area in substantial compliance with the representations made to the Commission. Feilure to do so may result in reversion of the Petition Area to its former classification, or a change to a more appropriate classification."



December 18, 2006

W, FRANK DRAND C, FASLA Chairman

FIOMASS, WITTEN, ASLA President

R. STAN DUNCAN, ASLA Executive View President

RUSSEL V.J. CHUNGANSA Executive Vire-President VINCENT SHIGENUNI Une-President GRANT TAU'RAKAMI, ARP Principal

Senar Associate

÷

KEVIN K. NISHIKAWA, ASI A Associate

The original intent of the State Land Use District Boundary Amendment was to entitle approximately 1,300 acres of land in East Kapolci, which would be then be sold to private developers. It was intended that the proceeds of the land sales would be used to fund the construction of a UHWO campus site mauka of the H-1 Freeway. Subsequent to the LUC approval of LUC Docket No. A99-728, Hawai'i's economy suffered a serious

During this period, it was determined that the construction of a new campus mauka of the H-I Freeway would be more expensive than the current site (makai of Parrington Highway). Since the marketability of the petition area was non-existent, and a site

selection study determined that the current site was best suited for a new UHWO campus,

downturn, and there was little or no market demand for new residential development.

KIMI MBKAMI YUTN, LITBU'AP

Associate

Tel. (808) 524-5631 Eav. (868) 524-1402 E-mail: Aysadmine; phehaw aut.om HONOLULU OFFICE 1081 Richop Street ACH Tower Strife 6720 Tel (6681 524 5611 741 (6681 524 5611

HILD OFFICE 101 Auponi Street 1161 Auponi Street 1161 Augeon Carter, Vane 310 1161 Auxart 94720 1241 0041 004 33.111 1522 (283) 061 7330

Mr. Anthony J. H. Ching, Executive Officer Land Use Commission State of Hawai'i P.O. Box 2359

Honolulu, Hawai'i 96804-2359

UNIVERSITY OF HAWAIT WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS)

Dear Mr. Ching:

TOM SCHNELS, AICP

hand Associate

SCOTT ALIKA ABRIGO

WAJEUKU OFFICE 1787 Wilf Pal oup, Suite 1 Wadubu, Rowart 9679 8-1251 Job (8081 242, 2378

SUBJECT:

Thank you for your letter dated July 25, 2006. We have reviewed your letter and offer the following responses to your comments:

KAYMOND T.HIGA, ASI A

Associate

NCOTT MURAKAMI, ASLA

UHWO was allocated 500 acres and most of the remainder of the 1,300 acres which were "urbanized" was granted to DHHL. The current owners (UHWO, DHHL, DOT, DLMR) then, are significantly different from those originally envisioned (private residential As you have learned, UHWO intends to file a motion to amend the Decision and Order in a few months. Of course, at that time, the motion to amend will address the differences between the original plans for the land and the current plans, and their impacts within the context of the applicable conditions. developers).

As recommended, approval of a motion to amend the Decision and Order will be included on the list of required permits and approvals that is referenced in the Final EIS. In addition, the status of each identified permit or approval will be described, in accordance with Section 11-200-17(h), Hawai'i Administrative Rules. The table showing the required approvals and permits on page xii and in Section 5.4 of the Draft EIS will be revised as follows: d

Mr. Anthony J.H. Ching SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Page 2

Application to be submitted with an application for Subdivision Approval. Application to be filed after the Zone Change application Motion to be filed at the time Application to be submitted FEIS concurrently with the after the acceptance of the FEIS concurrently with the PRU Permit application. Application to be submitted after the Zone Change Currently coordinating with HECO. Will be approved prior to occupancy. Application to be submitted <u>Application to be submitted</u> after the acceptance of the Zone Change application prior to Grading Permits application is approved. the FEIS is submitted. SUTATUS is approved State Public Utilities Commission Required Approvals and Permits Department of Planning and Permitting / City Council Department of Planning and Permitting / City Council Department of Planning and Department of Planning and Department of Planning and Governor, State of Hawai'i State Department of Health AUTHORIT Land Use Commission Permitting Permitting Permitting Environmental Impact Statement Plan Review Use (PRU) Permit Installation of Power Lines and Substations Motion to Amend Decision and PERMIT OR APPROVAL National Pollutant Discharge Elimination System (NPDES) (EIS) in compliance with Building/Grading Permits Subdivision Approval Chapter 343, HRS Park Dedication Zone Change <u>Order</u> Permit

In addition, Section 1.1 Project Profile, Permits/Approvals Required of the FEIS will be revised as follows:

Permits/Approvals Required:

Compliance with Chapter 343, Hawaii Revised Statutes (HRS) powerlines National Pollutant Discharge Elimination System (NPDES) Building/Grading Permits Public Utilities Commission approval of substations and Motion to Amend the Decision and Order Plan Review Use (PRU) Permit Subdivision Approval Park Dedication Zone Change Permit

PLANNING + LANDYCAPP ARCHITECTURE + ENVIRONMENTAL STUDIES + ENTITLEMENTS - PERMITTING + GRAPHIC DEGEN

Based on UHWO's and DHHL's discussions with DOE, it was determined that two elementary schools and one middle school (with one elementary school on the UHWO portion of the petition area) were warranted. During the public review period, the DOE wrote that: "...there will be approximately 1,771 public school students living within the project. That would be enough The first two paragraphs of Section 4.11.1 Educational Facilities, Anticipated Impacts and elementary school students to fill one elementary school...the DOE estimates that a fair-share contribution for the UH urban center would include sufficient acreage for an [emphasis added] enotiment in the schools and the projected number of students that will grow beyond the enotiment in the schools and the projected number of students that when the UH project is mature, that there will be approximately 1.771 public school students living within the project and that would be enough elementary school students to fill one <u>On September 13, 2002, the UH Board of Regents approved the 500-acre Kapolei Makai</u> site as the permanent location for the new UH West O'ahu campus and requested the transfer of title to the 500 acres from DLNR. The Board of Land and Natural Resources The University of Hawai'i still retains the 991-acre site mauka of the H-1 Freeway, but has no current plans for it other than being generally viewed as developable in the future The UH West O'ahu will likely increase the number of students enrolled in public schools. Based on the table provided by the DOE (Table 4.65), in 2009, three elementary schools (Barbers Point, Iroquois Point, and Kaimiloa) will be under capacity by approximately 867 students, and Kapolei Middle School and Kapolei Middle School will be over capacity by approximately 328 and 859 students, respectively. <u>During the public</u> review period the DOE wrote that the data provided in the DEIS illustrates the growing addition. DOE wrote that it "recognizes UH's acknowledgement of the need for a contribution to offset the impacts of the proposed project and we look forward to working together." As such, one possibility is the development of a "Lab School" on the compus The 550 students and 60 faculty and staff. During the public review period, the DOE also location of the school would be in close proximity to the proposed residential areas within the property and the planned DHHL residential subdivision directly south of the property. The school is projected to accommodate a typical DOE elementary school for wrote that it estimated that a school fair-share contribution for the proposed project <u>ide sufficient acreage for an elementary school and a cash contribution. In</u> UH has held preliminary discussions with DOE to address the demand of future school. aged children that would be generated by the proposed project. To help meet the demana for public educational facilities in the area, a 12-acre elementary school is proposea within the southern portion of the UH West O'ahu property (see Figure 2.1). Mr. Anthony J.H. Ching SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) approved the transfer of title of the 500- acre site to the UH in fee simple. Mitigating Measures of the EIS will be revised to read as follows: as an expansion area for the UHWO campus. elementary school and a cash contribution." elementary school. December 18, 2006 would Page 4 4 3. In response to your comments, we have revised the first paragraph of Section 6.2 Atternative In addition, we have added the following three paragraphs to the end of Section 6.2 Alternative Sites, Preferred Site of the EIS: In 1967, the UH Board of Regents initiated efforts to establish a four-year college in Central or West O'ahu. Over the years, the Regents approved varving sites. Wai aver Ridge (1970), Honouliult/Even (1973) and in 1993, Kapolei Makai (the current site). In 1996, it was decided that the West O'ahu, compute site would be moved to 991 acres mauka of the H-1 Freeway in Kapolet. This was based on a strategy by the State to entitle approximately 1,300 acres of land in East Kapolet, which would be then be sold to private developers. As a result, the Housing and Community Development Corporation of Hawaii (HCDCH) initiated a master planning effort for 1,300 acres including the of Hawaii (HCDCH) initiated a master planning effort for 1,300 acres including the UHWO Kapolei Makai site for mostly residential uses. It was intended that the proceeds of the land sales would be used to fund the construction of a UHWO campus site mauka however. Hawai'i's economy suffered a serious downturn, and there was little or no market demand for new residential development. During this period, it was determined that the construction of a new campus matka of the H-J Freeway would be more During the fall of 2001, the UH Administration, the Estate of James Compbell, and the Governor explored the possibility of relocating the UH West O'ahu campus from the 991-acre manka site to the City of Kapolei. acres, from Agricultural use to Urban use for the master-planned community known as East Kapolei. By Findings of Fact, Conclusions of Law and Decision and Order filed on September 8. 1999, the State Land Use Commission granted this petition. Subsequently, The University of Hawai'i conducted the University of Hawai'i West O'ahu Campus Site Selection Study (2002) (Site Selection Study) to assist the Administration and Board of three sites were initially listed: the City of Kapolei, Leeward Community College, and 1) the compilation of information on the potential of each site; 3) order-of-magnitude cost estimates and scheduling; and 4) Since the petition area was not marketable, and a site selection study determined that the current site was best suited for a new UHWO campus, UHWO was allocated 500 acres and most of the remainder of the 1,300 acres which were "urbanized" was transferred to DHHL. of the H-1 Freeway. On March 5, 1999, HCDCH filed a Petition for Land Use District Boundary Amendment to reclassify approximately 1,300 acres, including the subject 500 Regents in the selection of a permanent site for the UH West O'ahu campus. Sites <u>the</u> Kapolei Sports Complex, the City of Kapolei and while the Kapolei Makai (the proposed site) (see Figure 6.1) was added based on community input. A detailed characteristics of each site studied; 2) an evaluation of the physical development recommendations on a selected site for the UH West O'ahn campus. Six community considered for a-permanent campus included: Based upon the site evaluation criteria <u>meetings were held which indicated strong support for the 500-acre Kapolei Makai site</u> Mr. Anthony J.H. Ching SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) expensive than the current site (makai of Farrington Highway). evaluation of these sites included: Sites, of the EIS as follows:

December 18, 2006 Page 3

	RECEIVED
Me Anthenne 111 Otting	JUL 13 2006
SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Page 5	ECONOMIC DEVELOPMENT & TOURISM MAKKANDESS
Thank you again for your participation in the Environmental Impact Statement process. If you have any ounstitune remeasing the variant of any horizon of any content and for the second	STRATECIC INDUSTRIES DIVISION SISS South Beretanda Street, Leobogae A Kamehametha Bidg, 5 [°] Floor, Honoulu, Hawaii 08613 Totemhone: (883) 555-2535 Mailing Address: P.O. Box 2359, Honoulu, Hawaii 89804
any successful togething this project, prease do not nestrate to contact me at >21-5651. Sincerely,	
PBR HAWAII	July 12, 2006
Vince R. Shigh-	PBR HAWAII ASB Tower, Suite 650 1001 Bishon Street
v mean s. sugestim Vice President	Honolulu, Hawaii 96813
cc: Gene Awakuni/UH West O'ahu	Attn: Vincent Shigekuni
O.MOB1311322.05/E15/Final EIS/Final Responses to DEIS Comment/BL/08 LUC response 11-20.40c	Re: Draft Environmental Impact Statement (DEIS) University of Hawaii West Oahu Tax Map Key: 9-1-016: 120, 127, and 129
	In response to your July 3, 2006, notice, thank you for the opportunity to provide comments on the DEIS for the University of Hawaii West Oahu. The project will be a 7,600 student campus, with student housing, workforce/affordable housing, mixed use, residential, elementary school, parks, electrical substations, detention basin, and roads. We note that the first phase of the project will be accomplished through a public/private sector partnership.
	We would like to call your attention to: (1) State energy conservation goals; and, (2) energy and resource efficiency and renewable energy and resource development.
	1. State energy conservation goals. Project buildings, activities, and site grounds should be designed and/or retrofit with energy saving considerations. The mandate for such consideration is found in Chapter 344, HRS ("State Environmental Policy") and Chapter 226 ("Hawaii State Planning Act"). In particular, we would like to call to your attention HRS 226 18(c) (4) which includes a State objective of promoting all cost-effective energy conservation through adoption of energy-efficient practices and technologies.
	We recommend that you consult the City & County of Honolulu Energy Code early on in your project. Hawaiian Electric Company, Inc. may also have demand-side management programs that offer rebates for installation of energy efficient technologies.

PBR HAWAII July 12, 2006 Page 2	
E N S	

commitment to implement innovative and resource efficient operations and management and to design and construct buildings to meet and receive certification Bnergy and resource efficiency and renewable energy and resource development. Administrative Directive No. 06-01, which is also reflected in Act 96, 2006 SLH, install energy reduction, energy savings, or energy producing efforts and technologies to lessen electrical consumption or to increase efficiencies in using Design (LEED), among others. Second is Act 160, 2006 SLH, relating to the State Budget, which requires that each executive department shall plan for or for U.S. Green Building Council's Leadership in Energy and Environmental We refer you to two important directives affecting state agencies. First is which encourages State agencies and programs to increase their leadership electrical energy. ų.

not hesitate to contact Carilyn Shon, Energy Efficiency Branch Manager, at telephone number information on guidelines, directives and statutes, as well as studies and reports on aspects of energy efficiency (http://wrww.hawaii.gov/dbedt/info/energy/efficiency/state). Please also do We would like to call your attention to our website which provides detailed 587-3810, for additional information on implementation of Act 96 and the Governor's Administrative Directive.

This is a great opportunity for the University of Hawaii to "Lead by Example!"

Sincerely

Maurice H. Kaya Mur No

Chief Technology Officer

OEQC PBR HAWAII

3

HONOJJJJJ OFFICE 201 Blaup Street 202 Blaup Street 203 Blaup Street 203 Blaup Street 203 Street 203

BILO OFFICE 101 Aupuni Stevel Hit-1 agrees Screet Suite 310 Hito, 1 agrees Screet 200 Hito, 1 agrees 200 Tel: Rets 964, 1311 Fax: (816) 964, 1983

WAILUKU OFFICE ("52 With Pal Logy, Sony 1 % allelu, Hawart 96293-123 (eth. 6886) 247-3558

PBR HAWAI A ssociates, inc.

CHOMASS, WHITEN, ASLA R. FRANK IIRANDU FASLA Chaimmu

R. STAN DUNCAN, ASLA Exemitie Tree-President Pre-lalent

AUSSIDE V.J. CHUNG, FAN A

Executive Vice-Prostdeat PARCINT/SHIGERUNI

GRANT TAU'RAKAMI, AICP Vier President

TOM SCHNELL, AICP Senior Associate Principal

KEVIN K. NISHIKAWA, ASLA ANMOND THIGA, ASI A Servine Avoreinte

AIMI MIKAMI VUCN, LITUT'AP Associate

SCOTT ALIKA ABRIGO Issueime

Associate

COLT MURANAMI, ASEA Southers

December 18, 2006

Mr. Maurice H. Kaya, Chief Technology Officer State of Hawai'i

Department of Business, Economic Development & Tourism Strategic Industries Division Honolulu, Hawai'i 96804 P.O. Box 2359

SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS)

Dear Mr. Kaya:

Thank you for your letter dated July 12, 2006. We have reviewed your letter and offer the following responses to your comments: The following text will be added to the end of Section 2.6 Sustainability of the EIS will be revised as follows: ÷

Economic Development & Tourism (DBEDT), Strategic Industries Division, wrote to call attention to: 1) State energy conservation goals; and 2) energy During the DEIS public review period, the State Department of Business. and resource efficiency and renewable energy and resource development.

designed and/or retrofit with energy saving "considerations." DBEDT further noted that the mandate for such consideration is found in Chapter 344. HRS ("State Favironmental Policy") and Capter 226. HRS ("Havani 344. Branching Act"). In porticular, DBEDT noted HRS 226. 18(3) (4) which includes a State objective of promoting all cost-effective energy conservation through adoption of energy-efficient practices and technologies. DBEDT noted that project buildings, activities, and site grounds should be

others. The other directive DBEDT noted was Act 160, 2006 SLH, relating to the State Budget, which requires that each executive department shall plan for or install energy reduction, energy savings or energy-producing efforts and technologies to lessen electrical consumption or to increase efficiencies in using electrical energy. As stated above: Administrative Directive No. 06-01. which is also reflected in Act 96, 2006 Session Laws of Hawaii (SLH), which encourages State agencies and programs to increase their leadership commitment to implement innovative and resource efficient operations and management and to design and construct buildings to meet and receive certification for U.S. Green Building Council's Leadership in Energy and Environmental Design (LEED), among DBEDT noted two important directives affecting State agencies. The first is

FJANNING + FANDSCAPT ARCHITECTURE + ENVIRONMENTAL STUDILS + ENTITLUMENTS - FERMITTING - GRAPHIC DESIGN

Mr. Maurice H. Kaya SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Drafi EIS) December 18, 2006 Page 3	<u>DBEDT recommended that UHWO consult the City and County of Honolulu Energy</u> code early in the project. <u>DBEDT also noted that HECO may also have demand-side</u> manugement programs that offer rebates for installation of energy-efficient technologies.	<u>Based on DBEDT's recommendations. UHWO's mechanical and electrical</u> consultants. in consultation with UHWO's sustainability consultant, will be directed to review the City and County of Honolulu's Energy code early in the project and to consult with HECO on demand-side management programs that offer rebates for	instaliation of energy-efficient technologies. While current plans for the development of the UHWO campus includes both energy efficient design and the use of electricity developed and sold by HECO, the site is suited for the use of renewable energy technologies such as wind turbines and photovoltaics.	3. Section 5.2.2 State Environmental Policy, Chapter 344, Hawaii Revised Statutes of the EIS will be revised as follows:	ENERCY (A) Encourage the efficient use of energy resources.	DISCUSSION: The UH West O'ahu compus will help to conserve existing energy resources by reducing commuting time and distances for students and faculty restiding in the 'Ewa region. West O'ahu residents will have better access to higher education facilities, and ultimately, the campus will connect to a transit corridor planned adjacent to the North-South Road. Within the campus, pedestrian walkways, facilities for bicycles, and facilities for distance education will encourage non- motorized vehicular forms of transportation and reduce fossil fuel consumption. The use of enervy-efficient equinoment for highing systems and how water hearing and	cooling will also be encouraged within the campus. The State's Model Energy Code will be considered during the detailed design phases of project development.	Sustainability guidelines have been established for the UH West O'ahu Lands (see Section 2.6). The guidelines strive to achieve a Leadership, in Energy and Environmental Design (LEED) NC silver standard for campus development and community performance standards haved on the LEED NC v2.2 rating system. The Private Development Lands will strive to achieve the applicable design criteria in the UH West O'ahu LRDP and the recommended community performance standards.	4. The following text will be added at the end of the discussion of Section 5.2.5 Hawaii State Plan, Chapter 226, Hawaii Revised Statutes of the EIS will be revised as follows:	During the DEIS public review period, the State Department of Business, Economic Development, & Tourtism (DBEDT), Strategic Industries, Division, wrote to call attention to: 1) State energy conservation goals: and 2) energy and resource efficiency and renewable energy and resource development.
Mr. Maurice H. Kaya SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Page 2	Sustainability guidelines have been established for the UH West O'ahu. The guidelines strive to achieve a Loadership in Energy and Environmental Design (LEED) NC silver standard for compus development and community performance standards baved on the LEED NC v2.2 rating system.	DBEDT recommended that UHWO consult the City and County of Honolulu Energy code early in the project. DBEDT also noted that HECO may also have demand-side manugement programs that offer rebates for installation of energy-efficient technologies.	Baxed on DBEDT's recommendations. UHWO's mechanical and electrical consultants, in consultation with UHWO's sustainability consultant, will be directed to review the City and County of Honolulu's Energy code early in the project and to consult with HECO on demand-side management, programs that offer rebates for installation of energy-efficient technologies.	2. The following text will be added to the end of Section 4.10.6 <i>Electrical Facilities, Anticipated Impacts and Mitigative Measures</i> of the EIS will be revised as follows:	During the DEIS public review period, the State Department of Business, Economic Development & Tourism (DBEDT), Strategic, Industries, Division, wrote to call attention to: 1) State energy conservation goals; and 2) energy and resource efficiency and renewable energy and resource development.	DBEDT noted that project buildings, activities, and site grounds should be designed and/or retrofit with energy saving "considerations." DBEDT further noted that the mandate for such consideration is found in Chapter 344, HRS ("State Environmental Policy") and Chapter 226, HRS ("Hawaii State Planning Act"). In particular DBEDT noted HRS 226 18(c) (4) which includes a State Objective of promoting all cost-effective energy conservation through adoption of energy-efficient practices and technologies.	<u>DBEDT</u> noted no important directives affecting State agencies. The first is <u>Administrative Directive No. 06-01</u> , which is also reflected in Act 96, 2006 Session	LANK OF ITANKOL NUTCH ENCOURGES State agencies and programs to increase their leadership comminent to implement innovative and resource efficient operations and management and to design and construct buildings to meet and receive certification for U.S. Green Building Council's Leadership in Energy and Environmental Design (LEED), among others. The other directive DBEDT noted was Act 160, 2006 SLH, relating to the State Budget, which requires that each executive department shall plan for or install energy, reduction, energy savings or	energy-producing, efforts and technologies to lessen electrical consumption or to increase efficiencies in using electrical energy. As stated in Section 2.6 of the Draft E1S:	Sustainability suidelines have been established for the UH West O'ahu. The guidelines strive to achieve a Leadership in Energy and Environmental Design (LEED) NC silver standard for campus development and community performance standards based on the LEED NC v2.2 rating system.

Mr. Maurice H. Kaya SUBHECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Page 4

DBEDT noted that project buildings, activities, and site grounds should be designed and/or retrofit with energy saving "considerations." DBEDT further noted that the mandrae for such consideration is found in Chapter 344. HRS ("State Environmental Policy") and Chapter 226. HRS ("Havaii State Planning, Act"). In particular, DBEDT noted HRS 226. IRS (14) which includes a State objective of promoting all cost-effective energy construction through addressed in the Draft BIS. DBEDT noted two important directives affecting. State agencies. The first is Administrative Directive No. 06-01, which is also reflected in Act 96, 2006 Sassion Lows of Hawaii (SLH), which encourages State agencies and programs to increase their leadership commitment to implement innovative and resource efficient operations and management and to design and construct buildings to meet and receive certification for U.S. Green Building Council's Leadership in Energy and Environmental Design (LEED), among others. The other directive DBEDT noted was Act 160, 2006 SLH, relating to the State Budget, which requires that each executive department shall plan for or install energy reduction, energy savings or encetwire department shall plan for or install energy reduction, energy savings or energy-producing efforts and technologies to lessen electrical consumption or to increase efficiencies in using electrical energy. As stated in Section 2.6 of the Draft EIS.

<u>Sustainability guidelines have been established for the UH West O'ahu. The</u> guidelines strive to achieve a Leadership in Energy and Environmental Design (LEED) NC silver standard for campus development and community performance standards based on the LEED NC v2.2 rating system. DBEDT recommended that UHWO consult the City and County of Honolulu Energy code early in the project. DBEDT also noted that HECO may also have demand-side management programs that offer rebates for installation of energy-efficient technologies.

Based on DBEDT's recommendations. UHWO's mechanical and electrical consultants. in consultation with UHWO's sustainability consultant, will be directed to revew the City and County of Honolulu's Energy code early in the project and to consult with HECO on demand-side management programs that offer rebates for installation of energy-offecient technologies. Thank you very much for calling our attention to your website which provides detailed information on guidelines, directives, and even as studies and reports on aspects of energy efficiency (http://www.hawaii.gov/dbed/infe/energy/efficiency/state). We will not hesitate to contact Carlyn Shon, Energy Efficiency Branch Manager, for additional information on implementation of Act 96 and the Governor's Administrative Directive.

In closing, we concur that this is a great opportunity for the University of Hawai't to "Lead by Example!"

Mr. Maurice H. Kaya SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Page 5

Thank you again for your participation in the Environmental Impact Statement process. If you have any questions regarding this project, please do not hesitate to contact me at (808) 521-5631.

Sincerely,

PBR HAWAII

Vire & Brigh

Vincent R. Shigekuni Vice President cc Office of Environmental Quality Control Gene Awakuni/UH West O'ahu O:UOB13N322.05/EIS/Finul EIS/Final Responses to DEIS Comments/BL-01 DBEDT response 11-20.doc

DR HAWAII Associates, inc.	December 18, 2006	Mr. Otlando "Dan" Davidson, Executive Director	Department of Business, Economic Development & Tourism	rawent routing runnice and Development Corporation 677 Queen Street, Suite 300 A Honolulu, Hawai'i 96813	SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) Dear Mr. Davidson: Thank you for your letter dated August 17, 2006. We have reviewed your letter and offer the following responses to your comments:			A name you for your tener dated August 1/1, 2006. We have reviewed your letter and offer the following responses to your comments:	Thank you very much for the expression of support of the proposed project. We strongly concurnate that the UH West O'ahu campus will complement the directed growth of residential housing in Kapolei and provide various educational opportunities for the communities of West O'ahu.	UH West O'ahu shares Hawaii Housing Finance and Development Corporation's concerns regarding maximizing the provision of alfordable housing, but must weigh this concern with the goal of having a private partner fund the construction of most of the first phase of the UHWO campus. The private partner will make available a large portion of the non-student housing units to be affordable under the City and County of Honotuhu's affordable housing guidelines. In addition, UHWO will eventually include anoroximately 760 student housing units	Thank you again for your participation in the Environmental Impact Statement process. Your comments will be included in the Final EIS.	If you have any questions regarding this project, please do not hesitate to contact me at 521-5631.	Sincrely,	Vure R. Swell	Vincent R. Shigekuni	Vice President cc: Office of Environmental Quality Control Gene A watenni/11 Ware O'shin	OvtoB131132.05/EIS\Final EIS\Response to DEIS Comments\HIFDCresponse 11-20.doe	
		W. FRANK MRANDT, FASLA (<i>Lantun</i>	TROMASS, WITTEN, ASIA Predition	Executive Free Prediction Executive Vice Prediction Executive Vice President	VINCENT'SHIGGKUNA Vice-President	GRANTTAIURAKAMEAU 0-4-0-0	TTRUTH IN T	RAYMONDT: II6A, ASLA Schner Avwelate	KEVINK, MISHIKAWA, ASI, Awotatif Yivii Mir And Yefyy Leed	Associate SCOTT ALIAA ABRIGO Avsociate Vediate Kentiate Associate			1004 Bishop Steera 1004 Bishop Steera ASR Fower, Suite 670 Ronolofic "Hawafi 9683 4 4381 Ronolofic "Hawafi 9683 4 4381	ret (2013) 5.24 - 56.01 Lave (1918) 5.254 - 1.60 E-mail, sysachmusée pluchawan.c	HILO OFFICE 101 August States 1106 Laguar Center, Saite 110 1105 August 2012 02 1362	teb (8034 261-2123 Earc (8884 264: 1959	WAIRUKU OFFICE USE Will burney, Seite 1 Walhuk, Hawai 96295,123 Defekter 247, 2575	
ORLANDO "DAN" DAVIDSOH EKECUIVE DREGTOR	IN REPLY REFER TO:	06:DEV/0330								wed the ersity of project as provide	residential	ned number the Petition	ion Area be sidential					
	OF HAWAII	DOWNED DEVELOPMENT & LUDRISM ND DEVELOPMENT CORPORATION STREET SUITE 300	809) 587-0500 F イハーンハハム		(III) ALLEY	416 1 7 206	HAWAD		ihu, Ewa, Oahu, Hawaii <u>statement</u>	ment Corporation (HHFDC) revie (IS) dated July 2006 for the Univo . We are very supportive of this sidential housing in Kapolei and mmunities of West Oahu.	to provide approximately 3,280 i of which 355 units are earmarke	rsity campus, and an undetermin out the residential parcels within	ig, we recommend that the Petiti aximum number of affordable rec					strol
	STATE DEPARTMENT OF RUSINESS EV	HAWAII HOUSING FINANCEA FORCEA FORCEA	Fax: Attente		t Shigekuni, Vice President	r, Suite 650	p Street Iawaii 96813	higekuni:	University of Hawaii West Oa <u>Draft Environmental Impact S</u>	i Housing Finance and Developi onmental Impact Statement (DE st Oahu campus (Petition Area) lement the directed growth of re ucational opportunities for the α	at the Petition Area is projected ding 761 student housing units)	within the boundary of the univer- e units to be distributed through	with the intent of producing a m feasible	for the opportunity to comment.	-	I. I.	an" Davidson Nrector	se of Environmental Quality Cor
LINDA LINGLE GOVERNOR					Mr. Vincen PBR Heuro	ASB Towe	1001 Bisht Honolulu, }	Dear Mr. S	Subject:	The Hawai Draft Envin Hawaii We it will comp various edu	We note th units (includ	of affordable	developed i units where	Thank you t	Sincerely,	A A	Orlando "Da Executive D	c: The Offi

.

14 ANNING + LANINGAFI-ARCHITECTHRE + FNAIRONNINTAL STUDIES + ENTITHAGNTS - RERNITTAG + GRAPHIC DESIGN

LINDA LINGLE GOVERNOR



PATRICIA HAMAMOTO SUPERNTENDERT

> DEPARTMENT OF EDUCATION P.O. BOX 2300 HONCLULU, HAWAIT 95804

> > office of the superintendent August 10, 2006

Mr. Vincent Shigekuni, Vice President PBR Hawaii Pacific Tower, Suite 650 1001 Bishop Street Honolulu, Hawaii 96813

Dear Mr. Shigekuni:

Subject: University of Hawaii Draft Environmental Impact Statement (DEIS) for West Oahu Campus and Mixed Use Urban Development, East Kapolei, Ewa, Oahu TMK: 9-1-016: 120, 127, & 129 The Department of Education (DOE) has reviewed the Draft Environmental Impact Statement (DEIS) for the University of Hawaii's (UH) West Oahu campus and urban development on 500 acres within the former Housing and Community Development Corporation of Hawaii East Kapolei site.

The DEIS provides lists of actual enrollment in most of the Eva region's public schools in the past several years, enrollment projections, and the facility capacity of those schools. The data illustrates the growing enrollment in the schools and the projected number of students that will grow beyond the capacity of the schools' classrooms.

The DOE estimates that when the UH urban center is mature, there will be approximately 1,771 public school students living within the project. That would be enough elementary school students to fill one elementary school. To offset the impacts of the UH urban center the DOE requests that the State Land Use Commission (SLUC) impose a school fair-share contribution by using the standard condition language which reads as follows:

"The Applicant shall contribute to the development, funding, and/or construction of school facilities, on a fair-share basis, as determined by, and to the satisfaction of, the Department of Education. Terms of the contribution shall be agreed upon in writing by the Applicant and the Department of Education prior to obtaining zoning approval for any portion of the development."

Mr. Vincent Shigckuni Page 2 August 10, 2006 Based on the final number of residential units where public school students can reside, the DOE estimates that a school fait-share contribution for the UH urban center would include sufficient acreage for an elementary school and a cash contribution. The DOE recognizes UH's acknowledgement of the need for a contribution to offset the impacts of the proposed project and we look forward to working together.

If you have any questions, please call Heidi Meeker of the Facilities Development Branch at 733-4862.

Their Very truly yours,

Patricia Hamamoto Superintendent

PH:jmb

 Randolph Moore, Acting Assistant Superintendent, OBS Duane Kashiwai, Public Works Managet, FDB Mamo Carteria, CAS, Campbell/Kapolei/Waiamac Complex Areas Genevice Salamonson, OEQC Anthony Ching, SLUC

AN AFFIRMATIVE ACTION AND EQUAL OPPORTUNITY EMPLOYER

\$

Ĭ
VA
AV INC
INTES,
BR

December 18, 2006

Ms. Patricia Hamamoto, Superintendent State of Hawai' P.O. Box 2360 THOMASS, WITTEN, ASLA W. FRANK BRANDT, FASI, A

Channan Pre-tule at R.STANDENCAN, AMA Faculty: Vice-President

RUNNLLY, CHUNG, DASLA Executive Vice-President VINCENTSHIGERUNI Vice President GRANTT, MURAKAMI, AICP TONI SCHNELL, ALCP Principal

KAYMOND T, IDGA, ASLA Senar A sociate

Senior Associate

KEVIN K. NIVIIKAWA, ASLA Associate KIMI MIKAMI YI TNALIFI D'AP

SCOTT ALIKA ABRIGO Associate handing SCOFT MURANAMI, ASLA Issociate teut fostop Street ASD Tower, Suite 650 Horolog, Jawer 9581 5 181 Hor 8880 521 5631 fav (1986) 521 5631 fav (1986) 521 462 Ernalt sysaltrivite pfedats att om

HONOLULU DEFICE

RILO OFFICE [0] Aupuni Street [11] Aupuni Street [11] A. Las aff 967.201 476.5 [14] 1850 964.1333 [564,0885) 964.1333

WAILUKU OFFICE 1787 Wilt Da Loop, Suite 3 Waddar, Hawan 796791-1271 1646 (800) 242-2521

Honolulu, Hawai'i 96804 Department of Education

UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) SUBJECT:

Dear Ms. Hamamoto:

Thank you for your letter dated August 10, 2006. We have reviewed your letter and offer the following response to your comments: We acknowledge your concern regarding the projected increasing encollment for public schools in the 'Ewa/Kapolei region. We strongly concur that to offset the impact of the proposed project, a school fair-share contribution will be needed to provide sufficient acreage for an elementary school and a cash contribution. ...**:**

The first two paragraphs of Section 4.11.1 Educational Facilities, Anticipated Impacts and Mitigating Measures of the EIS will be revised to read as follows:

The UH West O'ahu will likely increase the number of students enrolled in public schools. Based on the table provided by the DOE (Table 4.65), in 2009, students, respectively. <u>During the public review period the DOE wrote that the</u> data provided in the DEIS illustrates the growing enrollment in the schools and the projected number of students that will grow beyond the capacity of the existing schools' classrooms. The DOE estimated that when the UH project is mature, that there will be approximately 1.771 public school students to fill within the project and that would be enough elementary school students to fill three elementary schools (Barbers Point, Iroquois Point, and Kaimiloa) will be under capacity by approximately 867 students, and Kapolei Middle School and Kapolei High School will be over capacity by approximately 328 and 859 one elementary schoo.

UH has held preliminary discussions with DOE to address the demand of O'ahu property (see Figure 2.1). The location of the school would be in close school is projected to accommodate <u>a typical DOE elementary school for</u> 550 students and 60 faculty and staff. <u>During the public review period, the DOE also wrote that it estimated that a school fair-share contribution for the</u> future school-aged children that would be generated by the proposed project. To help meet the demand for public educational facilities in the area, a 12acre elementary school is proposed within the southern portion of the UH West planned DHHL residential subdivision directly south of the property. The proximity to the proposed residential areas within the property and the proposed project would include sufficient acreage for an elementary school and a cash contribution. In addition, DOE wrote that it "recognizes UH's acknowledgement of the need for a contribution to offset the impacts of the proposed project and we look forward to working together." As such, one possibility is the development of a "Lab School" on the campus site. PLANNING + LANDSCAPF ARCHITECTURF + INTROUMENTAL STUDIES + INTITIEMENTS - FEMALTTING + GRAPHEC DESIGN

SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) Ms. Patricia Hamamoto December 18, 2006 Page 2 In regards to your Department's suggested language for the State Land Use Commission to use as condition of the project, your Department may not be aware that the land underlying this project was amended from Agriculture to Urban and had the following condition regarding schools: Ч,

school and University of Hawai't facilities as determined by and to the satisfaction of the State Department of Education (DOE) and the University of Hawai't. The Petitioner, the DOE, and the University of Hawai't shall enter into agreements on this matter prior to Petitioner shall contribute to the development, funding, and/or construction of public the Petitioner obtaining approval for the City and County of Honolulu zoning.

the prior Decision and Order. At that time, we understand that all of the agencies referred to in existing condition and those in your letter are very similar and we will not know what will be decided by the Land Use Commission, we will not make any revisions to the EIS regarding this Since the original petition to amend the State Land Use District Boundary did not include the Decision and Order would be contacted by the Land Use Comunission. The Land Use Commission may decide to: 1) leave the condition as stated above, as is: 2) amend the condition per your Department's recommended language; or 3) develop a different condition that would incorporate the key points of your Department's recommendation. Since the language of the particular topic. However, your letter and this response will be included in their entirety in UHWO as a prime land use, UH is planning to petition the State Land Use Commission to amend Section 12.0 of the Final EIS.

Thank you again for your participation in the Environmental Impact Statement process. Your comments will be included in the Final EIS.

If you have any questions regarding this project, please do not hesitate to contact me at 521-5631.

Sincerely,

Vire & R. Bug PBR HAWAII

Vincent R. Shigekuni Vice President Office of Environmental Quality Control Gene Awakuni/UH West O'ahu ij

0:UOB13\1322.05\EISVFinal EIS\Response to DEIS Comments\DOE response 11-20.doc

SSOCIATES, INC.	December 18, 2006			Mr. Micah A. Kane, Chairman	State of Hawai'i Hawaiian Homes Commission	Department of Hawaiian Home Lands P.O. Box 1879	Honolulu, Hawai'i 96805	SUBJECT: UNIVERSITY OF HAWALI WEST O'AHII (UH WEST O'AHII)	DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EKS)	Dear Mr. Kane:	Thank you for your letter dated July 13. 2006. We have reviewed your letter and offer the	following responses to your comments:	 Thank you very much for the expression of support of the proposed project. We strongly concur that the West O'ahu campus will give your homestead lessees (as 	well as other native Hawaiians) fiving in Kapolei and the Leeward Coast the opportunity to learn and work close to home.	2. Thank you also for your statement that the DHHL shall continue to cooperate with the University in the development of water eccess and terremission and unscender	collection off-site infrastructure.	3. According to the UH West O'atu, the construction of the 11.2-acre detention basin mauka of the DHHL property line will likely occur in the first phase of the project.	Your letter and this response will be included in Section 12.0 of the Final EIS.	Thank you again for your participation in the Environmental Impact Statement process. If	you have any questions regarding this project, please do not hesitate to contact me at 521-5631.	" Sincereły,	PBR HAWAII	Vire & Bugler	<i>.</i>	Vincent R. Shigekuni Vice President	cc: Office of Environmental Quality Control Gene Awalami/ITH West O'shu	
MANNEN & A			W. FRANK IRANDE, EASLA (<i>Justingiu</i>	VISVNALLIM'S SYMOHI.	President	R. STAN DUNCAN, ASI A L'Actutive Vire-President	RUSSELLY, CHUNG FASLA Executive View-Prevident	VINCLAT SHIGEKUN	View President	GRANT'T AIURAKAMI, AICP Principal	TOM SCHNILL, AICP Senior Associate	RAYAIOND T. HIGA, ASI,A Armir Asimune	KUVIN K, NISHIKAWA, ASLA Associa	KAMI MIKAMI VUTSy,LITD'AI Associato	SCOTI ALKA ABRIGO Avvecinte	SCOTT MURAKAMII, ASLA . Teordae				1003 Birlon Street ASIT Taver, Saire 620 Doublin, Raeafi 96813-5184	Tet (AUS) 523-564 1.5ee (AUS) 523-1412 5.mail: sysadmus; phythawair.co	HILO OFFICE 101 Anpuni Street	Hile Lagoon Conter, Suite Mu Hile, Hawait 967-24 1264 3(c) 16081 964-3113	FAX: [KCS] 484 FAX:	WAILUKU OPFICE 1787 With Pa Loop, Suite 4 Waikht, Diversity 2020 J 127	1412 757 MANDI 1991	
MICAH A. KANE CKANKIAN (LAWARAN HOVES CORDISION	BIGN FERMIERSON DGZUTY TO THE CHARMAN	KAULANA H. PARK EXECUTIVE ASSISTANT										at	mment on the subject	HHL) reiterates its	campus will give our it the opportunity to		ne university in the on, and wastewater	is the schedule for in marticular. the	11.2-acre detention	<pre>improvements we would f our single-family </pre>	subdivision. We are s included in Phase I	felopment Division at		,o,	RECENT.	Chairman JUL 18 2006 Commission	Lun HAWAII
		STATE OF HAWAII	DEPARTMENT OF HAWAILAN HOME LANI	P.O. BOX 1879	JULY 13, 2006		i, Vice President				and the set the set of the set of the set	erstry of navall west oallu ft Environmental Impact Stateme	allowing us to review and co Tunact Statement (DETS)	: of Hawaiian Home Lands (D	osed project. The West Oahu 1 Kapolei and the Leeward Coas	complex.	contrante to cooperate with t ter storage and transmissi infrastructure.	mcern with the proposed plan the drainage improvements.	Gulch and construction of the	broperty line. Without those j svelop approximately fifty o	out 1278 of our residential (see that the detention basin is chedule.	. Darrell Ing in our Land Dev	any questions.	Aloha and mahal	MACH	ficah A. Ane. Hawaiian Homes	mental Quality Control
CLINDA LINGLE GOVERNOR STATE OF MAWAII							Mr. Vincent Shigekun PBR Hawaii	ASB Tower, Suite 650	1001 Bishop Street Honolulu, HI 96813	Dear Mr. Shi <i>c</i> ekuni:	Cubact.	Draf	Thank you for Draft Environmental	The Department	support of the prop homestead lessees in learn and work close	The DHH chart	development of wal	Our primary co implementation of	diversion of Kaloi (Dasin mauka of our p not be able to de homestead lots _ ab	thus encouraged to s of the development s	Please call Mr	587-6451 if you have	÷			c: Office of Environ

.

PBR HAWAII & ANAII

• • . O:UOB131(322.05)EIS/Einal EfStResponse to DEIS Comments/BL-03 DHHL response 11-20.dox

PLANNING + LAS.DSCAPL ARCHITECTURF + ENVIRONMENTAL STUDIES + ENTITLEMENTS - FERMITTING + GAAPHIC DEGIG

LINDA LINGLE GOVERNOR OF HAWAS



CHIYOME L. FUKINO, M.D. DRECTOR OF HEALTH

RECEIVED

STATE OF HAWAII DEPARTMENT OF HEALTH P.O. Bex 378 HOROLUU,HAWAII 5565-1378

August 16, 2006

In reply, planse refer to: AUG 2 3 2006 PBR HAWAII

EPO-06-121

Mr. Vincent Shigekuni, Vice President Honolulu, Hawaii 96813 ASB Tower, Suite 650 1001 Bishop Street **PBR HAWAII**

Dear Mr. Shigekuni:

Draft Environmental Impact Statement for the Proposed University of Hawaii TMK: (1) 9-1-016: 120, 127 and 129; 500. 327 acres West Oahu Campus at Kapolei, Oahu, Hawaii SUBJECT:

Thank you for allowing us to review and comment on the subject document. The document was routed to the various branches of the Environmental Health Administration. We have the following comments.

Wastewater Branch

We have reviewed the document on the subject project which proposes to develop a 7,600 student campus, student housing, workforce/affordable housing, mixed use, residential, clementary school, parks, electrical substations, detention basin and roads.

Oahu Wastewater Advisory Committee where no new cesspools will be allowed. The majority The project is located in the Critical Wastewater Disposal Area (CWDA) as determined by the of the parcels lie in the Pass Zone. However, there are parcels in the No Pass Zone.

connection, we do not have any objections to the proposed project. Further, the developer should work with the City and utilize recycled water for irrigation and other non-potable water purposes connection to the City & County Sewer Service System. Areas in the No Pass Zone are usually denied development unless connection can be made. Therefore, due to City and County sewer We note that the wastewater generated from the project will be disposed of through the in the open spaces and for landscaping areas.

Administrative Rules, Chapter 11-62, "Wastewater Systems." We do reserve the right to review questions, please contact the Planning & Design Section of the Wastewater Branch at 586-4294. All wastewater plans must conform to applicable provisions of the Department of Health's the detailed wastewater plans for conformance to applicable rules. Should you have any

August 16, 2006 Mr. Shirekuni Page 2

Safe Drinking Water Branch (SDWB)

Kapolei developments (including the proposed new campus). We request that further details of We have reviewed the Draft EIS for the subject project. We understand that based on discussions with the Board of Water Supply (BWS), potable water is available to serve the proposed new campus and the BWS will construct new facilities to accommodate the East these new facilities are discussed in the Final EIS. The document also mentions that the BWS plans to extend the existing non-potable water system irrigation. All projects which propose the use of dual water systems or the use of a non-potable water system in proximity to an existing potable water system to meet irrigation or other needs must be careful in the design and operation of these systems to prevent the cross-connection of Hawaii Administrative Rules, Title 11, Chapter 11-21 entitled "Cross Connection and Backflow with warning signs to prevent inadvertent consumption of non-potable water. Compliance with to the North-South Road corridor and the proposed campus will use this non-potable water for proper operation. Further, all non-potable spigots and irrigated areas should be clearly labeled these systems and prevent the possibility of backflow of water from the non-potable system to potable water supply. In addition backflow devices must be tested periodically to assure their the potable system. The two systems must be clearly labeled and physically separated by air gaps or reduced pressure principle backflow prevention devices to avoid contaminating the Control" is required.

Should you have any questions, please contact Mr. Kumar Bhagavan of the SDWB's Compliance Section at 586-4258 in Honolulu.

Clean Water Branch (Standard Comments)

The Department of Health (DOH), Clean Water Branch (CWB) has reviewed the limited information contained in the subject document and offers the following comments:

- discharge into the navigable waters..." (enphasis added). The term "discharge" is defined in CWA, Subsections 502(16), 502(12), and 502(6); Title 40, Code of Federal Regulations required for "[a]ny applicant for Federal license or permit to conduct any activity including, Pursuant to Federal Water Pollution Control Act (commonly known as the "Clean Water Act" (CWA) Paragraph 401(a)(1), a Section 401 Water Quality Certification (WQC) is The Army Corps of Engineers should be contacted at (808) 438-9258 for this project. but not limited to, the construction or operation of facilities, which may result in any (CFR), Section 122.2; and Hawaii Administrative Rules (HAR), Chapter 11-54.
- In accordance with HAR, Sections 11-55-04 and 11-55-34.05, the Director of Health may require the submittal of an individual permit application or a Notice of Intent (NOI) for сi

Mr. Shigekuni August 16, 2006 Page 3 general permit coverage authorized under the National Pollutant Discharge Elimination System (NPDES).

- a. An application for an NPDES individual permit is to be submitted at least 180 days before the commencement of the respective activities. The NPDES application forms may also be picked up at our office or downloaded from our website at http://www.hawaii.gov/health/environmental/water/cleanwater/forms/indiv-index.html
- b. An NOI to be covered by an NPDES general permit is to be submitted at least 30 days before the commencement of the respective activity. A separate NOI is needed for coverage under each NPDES general permit. The NOI forms may be picked up at our office or downloaded from our website at:

http://www.hawaii.gov/health/environmental/water/cjeanwater/forms/genl-index.html.

- Storm water associated with industrial activities, as defined in Title 40, CFR, Sections 122.26(b)(14)(i) through 122.26(b)(14)(ix) and 122.26(b)(14)(xi). [HAR, Chapter 11-55, Appendix B]
- ii. Construction activities, including clearing, grading, and excavation, that result in the disturbance of equal to or greater than one (1) acre of total land area. The total land area includes a contiguous area where multiple separate and distinct construction activities may be taking place at different times on different schedules under a larger common plan of development or sale. An NPDES permit is required before the commencement of the construction activities. [HAR, Chapter 11-55, Appendix C]
- iii. Discharges of treated effluent from leaking underground storage tank remedial activities. [HAR, Chapter 11-55, Appendix D]
- iv. Discharges of once through cooling water less than one (1) million gallons per day. [HAR, Chapter 11-55, Appendix E]
- v. Discharges of hydrotesting water. [HAR, Chapter 11-55, Appendix F]
- vi. Discharges of construction dewatering effluent. [HAR, Chapter 11-55, Appendix G]
- vii. Discharges of treated efftuent from petroleum bulk stations and terminals. [HAR, Chapter 11-55, Appendix H]
- viii. Discharges of treated effluent from well drilling activities. [HAR, Chapter 11-55, Appendix I]

Mr. Shigekuni August 16, 2006 Page 4

- Discharges of treated effluent from recycled water distribution systems. [HAR, Chapter 11-55, Appendix J]
- Discharges of storm water from a small municipal separate storm sewer system. [HAR, Chapter 11-55, Appendix K]
- Discharges of circulation water from decorative ponds or tanks. [HAR, Chapter 11-55, Appendix L]
- 3. In accordance with HAR, Section 11-55-38, the applicant for an NPDES permit is required to either submit a copy of the new NOI or NPDES permit application to the State Department of Land and Natural Resources, State Historic Preservation Division (SHPD), or demonstrate to the satisfaction of the DOH that the project, activity, or site covered by the NOI or application has been or is being reviewed by SHPD. If applicable, please submit a copy of the request for review by SHPD or SHPD's determination letter for the project.
- Any discharges related to project construction or operation activities, with or without a Section 401 WQC or NPDES permit coverage, shall comply with the applicable State Water Quality Standards as specified in HAR, Chapter 11-54.

The Hawaii Revised Statutes, Subsection 342D-50(a), requires that "InJo person, including any public body; shall discharge any water pollutants into state waters, or cause or allow any water pollutant to enter state waters except in compliance with this chapter, rules adopted pursuant to this Chapter, or a permit or variance issued by the director." If you have any questions, please contact Mr. Alec Wong, Supervisor of the Engineering Section, CWB, at (808) 586-4309.

Clean Air Branch

A significant potential for fugitive dust emissions exists during all phases of construction and operations. Proposed activities that occur in proximity to existing residences, businesses, public areas or thoroughizares, exacerbate potential dust problems. It is recommanded that a dust control management plan be developed which identifies and addresses all activities that have a potential to generate fugitive dust. The plan, which doses not require the Department of Health (DOH) approval, would help with recognizing and minimizing the dust problems from the proposed project.

Activities must comply with the provisions of Hawaii Administrative Rules, §11-60.1-33 on Fugitive Dust. In addition, for cases involving mixed land use, we strongly recommend that buffer zones be established, wherever possible, in order to alleviate potential nuisance problems.

Mr. Shigekuni August 16, 2006 Page 5 The contractor should provide adequate measures to control the fugitive dust from the road areas and during the various phases of construction. Examples of measures that can be implemented to control dust include, but are not limited to, the following:

- a) Planning the different phases of construction, focusing on minimizing the amount of dust-generating materials and activities, centralizing on-site vehicular traffic routes, and locating potential dust-generating equipment in areas of the least impact;
 b) Providing an adequate water resource at the site prior to start-up of construction
 - activities; c) Landscaping and providing rapid covering of bare areas, including slopes, starting from
 - d) Minimizing dust from shoulders and access roads;
- Providing adequate dust contractors during weekends, after hours, and prior to daily start-up of construction activities; and
 - f) Controlling dust from debris being hauled away from the project site.

If you have any questions, please contact the Clean Air Branch at 586-4200

Hazard Evaluation & Emergency Response Office (HEER)

- Phase I Environmental Site Assessment (ESA) should be conducted for developments or redevielopments. If the investigation shows that a release of petroleum, hazardous substances, pollutaris or contaminants occurred at the site, the site should be properly characterized through an approved Hawaii State Department of Health (DOH)/ HEBR soil and or groundwater sampling plan. If the site is found to be contaminated, then all removal and remodial actions to clean up hazardous substance or oil releases by past and present owners/tenants must compily with Chapter 128D, Environmental Response Law, HRS, and Title 11, Chapter 451, HAR, State Contingency Plan.
- 2. All lands formerly in the production of sugarcane should be characterized for arsenic contamination. If arsenic is detected above the US EPA Region 9 preliminary remediation goal (PRG) for non-cancer effects, then a removal and or remedial plan must remediated to the DOH HEER office for approval. The plan must comply with Chapter 128D, Environmental Response Law, HRS, and Title 11, Chapter 451, HAR, State Contingency Plan.
- 3. If the land has a history of previous releases of petroleum, hazardous substances, pollutants, or contaminants, we recommend that the applicant request a " no further action" (NFA) letter from the DOH HEER office prior to the approval of the land use change or permit approval.

Mr. Shigekuni August 16, 2006 Page 6 4. We also note that the Department of Hawaiian Homelands East Kapolei Affordable Housing Brownfield project which surrounds the listed Ewa Sugar Mill/Oahn Sugar Company Pesticide Mixing and Loading Site, was not included on page 53 Table 4.1 Hazardous Waste Sites. A Phase I environmental site assessment was conducted and a Phase II was recommended for the site.

Should you have any questions, please contact HEER office at 586-4250.

We strongly recommend that you review all of the Standard Comments on our website: <u>www.state.ht.us/heaith/environmental/env-planning/landuse/landuse.html</u>. Any comments specifically applicable to this project should be adhered to. If there are any questions about these comments please contact Jiacai Liu with the Environmental Planning Office at 586-4346.

Sincerely,

KELVIN H. SUNADA, MANAGER Environmental Planning Office

c EPO CAB SDWB SDWB OHQC OHQC
KIMI MIKAMI VI'EN,LEED^aad *Asociate*

KEVEN K. NISHIKAWA, ASI A Assirtate

RAVMOND T. HIGA. ASLA Settor Associate TOM SCHNELL, AICP Scalar Associate

SCOIT AILKA ABRIGO Asow iau SCOIT MURAKAMI, ASLA Asovéiate

R. STANDERCAK, AMA Exernity Tre-President RUSSEL YJ, CHUNG, IASIA Exernity Vize President VINCENT SHIGERUNI Ver President

W. FRANK IRANDT, FASLA Chuirman THOMASS, WITTEN, ASI,A Prevident GRANTT, MURAKAMI, AICP Principul

HONOLULU OFFICE MODI Bahy WOFFICE ASB Energissure for Develop. Jana (FASE) 1184 Brenden, Jana (FASE) 1184 Eres (SE) 251 402 Ereadi syndamer plehtus at com

H1LO OFFICE 101 Aupunt Street 116 - Layeous Cruter, Suite 310 116 - Llyavit W-C-30 126.2 14 - (203) 961 - 7057 15 - 50 (203) 961 - 7057 15 - 50 (203) 961 - 7057

WAIIJUKU OFFICE 1787 Will 15 Loop, Suite 4 Windidan, Jhanaf 196793, 1271 664 OND2 242, 2378

Kelvin H. Sunada BJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT VIRONMENTAL IMPACT STATEMENT (Draft EIS) vember 20, 2006	Mr. Kelvin H. Sunada SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) November 20, 2006
<u>this chapter, rules adopted pursuant to this Chapter, or a permit or variance issued by the</u> director."	rage 4 <u>property be characterized for arsenic contamination. The UH West O'ahu will conduct a Phase I</u> FSA and wile review on a second
<u>Air Branch</u>	essection with testing once an agreement with the private developer is reached.
meur that there is a significant potential for fugitive dust emissions during construction and greatly ziate the information provided. We have revised the third paragraph of Section 3.4 <i>Soils</i> , pated Impacts and Mitigation Measures of the EIS to read as follows:	7. We will notify the applicant that they should check with HEER if the site has a history of previous teleases of petroleurn, hazardous substances, pollutants and contaminants (which would be discovered during the Phase 1 ESA) and if so, request a "No Further Action" (NFA) letter from HEER prior to the approval of the land use change or other permit approvals.
watering program will be implemented to minimize soil loss through fugitive dust emissions tring construction. Other dust and erosion control measures include cleaning job-site ustruction equipment and establishing ground cover as quickly as possible after grading. using the public review period, the Department of Health Clean M. Baucht Precommended that dust control management be prepared. The DOH CAB also movided additional set	8. Table 4.1 has been revised to include as "Hazardous Waste Sites," the Department of Hawaiian Homelands East Kapolei Affordable Housing Brownfield Project which surrounds the 'Ewa Sugar Mill/Oahu Sugar Company Pesticide Mixing and Loading Site (although this latter site is not located within the property).
astures that can be implemented during construction: where possible, for areas involving mixed land uses, buffer zones be established to alleviate potential dust mixence mrchbens.	The following responses are offered to the Standard Comments on your website: <u>http://www.state.hi.us/health/environmental/env-planning/landuse.html</u> for the Environmental Planning Office:
planning the different phases of construction, focusing on minimizing the amount of dust- generating materials and activities, contralizing on-site vehicular traffic routes, and locating potential dust-generating equipment in areas of the least impact; providing an adequate water resource and watering program at the site prior to start-up of construction activitier.	• Waterbody type and class: The proposed project is located within the middle reaches of the Kalo't Gulch watershed. Kalo't Gulch and Hunchune Gulch, which converge within the property, are ephemeral, flowing in response to storm events that are significant cough to generate direct runoff. The proposed project does not contain any other wetlands, streams, or known waterbodics onsite.
the initial grading phase. minimizing dans throughers and access roads. providing adequate dust control macasures during weekends, after hours, and prior to daily start-up of construction activities; and controlling dust from debris heing hauled away from the project site.	 National Pollutant Discharge Blimination System (NPDES): A NPDES permit is required for this project. The need for this permit is stated in Section 5.4 of the EIS. The preliminary engineering report includes a drainage master plan and is attached as an appendix to the EIS.
Evaluation & Emergency Response Office (HEER)	• Water quality management: The drainage master plan for the project is described in the medianinary environments and is altoched as Amondus 1.56 the DNS The medianing
^{hase} I Brwironmental Site Assessment (ESA) will be conducted for development of the proposed ject. The following text will be added to the second paragraph of Section 4.6, <i>Man-made ands, Existing Conditions:</i>	drainage system calls or detention basins to help miligate and filter runoff generated onsite. The drainage system is summarized in Section 4.10.4 of the FIS.
PBR HAWAII used Environmental Data Resources, Inc. (EDR) to search major Federal, State, and local regulatory agency lists of RECs in January 2023. No hazardous suise vere identified on the UH West O'ahu istroperty. During the mubits review pariod the DOH	 Impaired waters of Hawai'i: No water bodies on the current 2004 List of Impaired Waters in Hawaii Prepared under Clean Water Act Section 303 (d) are expected to be impacted by the proposed project.
Harard Evaluation & Emergency Response Office (HEER) recommended that a Phase 1 ESA be conducted and the property he characterized for assenic contamination. The UH West O chu, will conduct a Phase 1 ESA and soils testing once an agreement with the private developer is reached.	 We acknowledge your suggestion that we identify and analyze potential project impacts at a watershed scale. Section 4.10.4 <i>Drainage Facilities, Anticipated Impacts and Mitigation Measures</i> of the EIS and the preliminary infrastructure report describe the proposed drainage system which include detention basins that are consistent with your recommendation for implement Alternative and error projection consistent to mitigate
recommended, all lands formerly in the production of sugarcane will be characterized for arsenic tamination. The following text will be added to the second paragraph of Section 4.6, Man-made cards, Existing Conditions:	runoff and improve water quality and the project's potential impact on aquatic and riparian ecosystems.
PBR HAWAII used Environmental Data Resources, Inc. (EDR) to search major Federal, State, and local regulatory agency lists of RECs in January 2003. No hazardous sites were identified on the UH West O ahu property. <u>During the public review period the DOH Hazard Evaluation</u> & Emergency Response Office (HEER) recommended that a Plusse I ESA be conducted and the	The Standard Comments for the Hazard Evaluation and Emergency Response Office listed on your website are the same as the comments included in your letter and are answered above.

Mr. Kelvin H. Sunada SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) November 20, 2006 Page 5

We offer the following responses to the Standard Comments for the Clean Air Branch:

- Construction/Demolition Involving Asbestos: Since the majority of the site is currently
 undeveloped, cultivated for many years for sugarcane, and most recently for diversified
 agriculture, it is not expected that there may be asbestos on site. However, if asbestos is
 found, the applicant or subsequent developers will contact the Asbestos Abatement
 Office in the Noise, Radiation and Indoor Air Quality Branch prior to
 construction/demolition.
- Control of Fugitive Dust: Discussion of this issue and proposed mitigation measures are provided in Section 3.4 Soils, Anticipated Impacts and Mitigation Measures of the EIS.

We offer the following responses to the Standard Comments for the Safe Drinking Water Branch:

- Public Water Systems: The Safe Drinking Water Branch directly responded to the DEIS in your August 16, 2006 letter and included comments applicable to the project. The responses to those comments are provided earlier in this letter.
- Underground Injection Control: No underground injection wells are proposed for this project.
- Groundwater Protection Program: No 9- or 18-hole golf courses are proposed for this project.

We offer the following responses to the Standard Comments for the Solid and Hazardous Waste Branch:

 Detailed discussion on solid waste management for the project is provided in Section 4.10.5 of the EIS. In addition, we will include the following paragraphs after the first paragraph in Section 4.10.5 Solid Waste Disposal Facilities, Anticipated Impacts and Mitigation Measures of the EIS:

"Per comments received from the State Office of Environmental Quality Control during the Public Review Period. a construction wastle receveling plan will be prepared before construction is initiated. All solid wastle generated during project constructions that he directed to a Department of Health permitted solid wastle constructions that he directed to a Department of Tealth permitted solid wastle disposal. or recycling facility. Also, all highway, and road construction improvement projects funded by the State or a county or roadways that are to be accepted by the State or a sophic road shall utilize a minimum of ten percent crushed glass aggregate as specified by the Department of Transportation in all base-course (treated or untreated) and sub-base when the glass is available to the quary or contractor at a price no greater than that of the equivalent aggregate." "Recycling shall be encouraged within the project including the reuse and recycling of green waste generated during construction clearing and grubbing activities, the use of recycled construction and demolition wastes and the use of materials made from recycled routies, the use of locally produced compost as available for landscaping, and the provision of space for recycling bins in the detailed design of the community."

Mr. Kelvin H. Sunada SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) November 20, 2006 Page 6

The Standard Comments for the Wastewater Branch listed on your website are the same as the comments included in your letter and are answered above.

For Noise, Radiation, and Indoor Air Quality Branch Standard Comments, the proposed project will comply with the Administrative Rules of the Department of Health.

Thank you again for your participation in the Environmental Impact Statement process for this project. Your comments will be included in the Final EIS.

if you have any questions regarding this project, please do not hesitate to contact me at 521-5631.

Sincerely,

PBR HAWAII

Vizzek.

Vincent R. Shigekuni Vice President cc: Office of Environmental Quality Control Gene Awakuni/UH West O'ahu O:VOB1311322.05/EISUDFinal EISVFinal Responses toDEIS CommentsBL-19 DOH response 11-20.doc

608 555 4155 OFC ENVIRONMENTAL	09:00:54 08-21-2006 1 /2	208 565 4185 DFU: ENVIRUMAENTAL 05-01-20 06-21-2006
цена циста сопенная ника сопенная ника сопенная с с с с разли сон обятся с никала с с с с с с с с с с с с с с с с с с	GRATIFIC SALADOSON DISECTOR All MALITY CONTROL	Gene Awakuni August 21, 2006 Page 2
August 21, 2006 Gene Austrust Chancellor		, identify public viewpoinds of the project site from which visual impacts may occur, especially or marks and makar viewplanes. Show duese impacts by superimposing a rendering of the propose facilities (or major portions thereof) onto photographis taken from public vantage points. In the final Ediscuss underground phacement of utility lines and, if this option has not be seleted, the reasons why.
University of Hawaii - West Oahu 96-129 Ala Ike		<u>Parks</u> : Will a park dedication be required? In the final EIS discuss the details of this.
Pearl City, Hawaii 96782		Construction impacts. We recommend establishing a construction waste recycling plan.
Dear Mr. Awakum:		Campus services:
Subject: Draft Environmental Impact Statement (E Include the following in the final EIS:	EIS), UH - West Oahu	A. Lodging: The 3/14/05 letter from the Department of Planning & Permiliting discusse "lodging in Commercial Placel A" and that currently this type of lodging would only be allowed at RG Ollina and Ocean Pointe marina. Is there rowision for any kind of
<u>Permits and approvals</u> : In table 5.1, list the status of each applications have not yet been made, indicate the expect	ch permit or approval. If some ted dates of application.	temporary ledging on campus for visiting faculty or confetence-goers? That would be a logical accessory service for a university campus.
<u>Conformance to policies</u> : Section 5.2.2 enumerates clern discussion. Add this to the final EIS.	aents of HRS 344 but there is no	B. Transit service: Will there be a shuttle service on campus for those not wishing to drive, cycle or walk?
Paving and Innécaping: Hawaii Revised Statotes 103D-408 paving materials whenever possible, and HRS 103D-408 flora whenever and wherever possible. We also recomme final EA indicate if you will follow these requirements.	407 requires the use of recycled glass in 8 requires the use of native Rawaiian end the use of untitity plants. In the	If you have any questions, call Nancy Heiurich at 586-4185. Sincerely,
Consultation:		GENEVIEVE SALARONSON
 A. In the community presentations section of cha raised at these meetings. Transcripts are not requir B. Correct table 8.2 by adding the Department of 	apter 8 include a symopsis of issues inten, just a symopsis. f Design & Construction as a	Ditector c: Vince Shigekuni
commentor. C. Your 6/16005 response to the Department of Pi includes the abbreviation WOCD. Add this to you D. In section 8 indicate briefly wity the 1998 EiS new one submitted February 2005.	Planning & Permitting's letter of 3/14/05 ur glossary. 5 preparation notice was withdrawn and a	

Visual impacts: Include drawings or diagrams of proposed buildings and any proposed landscaping that show the final appearance of at least major portions of the campus and the development lands.

717

R HAWA
$\operatorname{PB}_{\operatorname{sag}}$

December 18, 2006

Office of Environmental Quality Control Ms. Genevieve Salmonson, Director State of Hawai'i

> W. FRANK BRAND'E FASLA Chairman THOMAS S, WITTEN, ASI, A

K.STAN DUNCAN, ASLA Freethyr Vire-Pasident Presiden

RUSSELLY, J. CHUNG, FASLA

Executive Vice-Pustilent VINCENT SHIGEKUNI Vice President GRANTT. MURAKAMI, AICP TOM SCRINELL, AICP Poincipul

RAYMOND T. HIGA, ASLA Senior Associate Sentary Associate KEVEN K, NISHEKAWA, ASEA **SAPCERIE** KUMI MIKAMI VETEN, UTI D'AP

SCOTT ALIKA ABRIGO Associate Associate

SCOTT MURAKAMI, ASLA Associate

1091 Bitchup Street ASIT Tower, Sinter 670 Hendlin, Hwardt Will 1184 Let (018) 221 2641 Lax (8165 224 21 102 Canalle vy.adhostne, phehawahr, onn BONOLULU OFFICE

HILO OFFICE 281 Aupuri Street 860 - Lasson Screet 860 - Lasson Socie 310 164-18481 96-201 - 1962 164-18481 96-1 - 1989 15-0 (208) 96-1 - 1989 WAILUKU OFFICE 1787 Will Da (1990, Suite 1 Waihdar, Hawail 9679 & 1271 Ich 18083 242 2578

235 South Beretania Street, Suite 702 Honolulu, Hawai'i 96813

UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) SUBJECT:

Dear Ms. Salmonson:

Thank you for your letter dated August 21, 2006. We have reviewed your letter and offer the following responses to your comments: Table 5.2 of the Draft EIS will be updated to include the status of each permit and/or approval. As much as we would like to include the expected dates of the various applications, unfortunately we are reluctant to do so as this project has endured numerous delays to date. Table 5.2 will be revised to read as follows:

Table 5.2 – Required Approvals and Permits

PERMIT OR APPROVAL	AUTHORITY	STATUS
Errvironmental-Impact Statement (EIS) in compliance with-Chapter 343, HRS	Governor, State of Hawai'i	
Motion to Amend Decision and Order	Land Use Comnission	<u>Motion to be filed at the</u> time the FEIS is submitted.
Plan Review Use (PRU) Permit	Department of Planning and Permitting / City Council	Application to be submitted after the acceptance of the FEIS concurrently with the Zone Change application.
Zone Change	Department of Planning and Permitting / City Council	Application to be submitted after the acceptance of the FEIS concurrently with the PRU Permit application.
Subdivision Approval	Department of Planning and Permitting	Application to be submitted after the Zone Change application is approved.
Park Dedication	<u>Department of Planning and</u> Permitting	Application to be submitted with an application for Subdivision Approval.
Building/Grading Permits	Department of Planning and Permitting	Application to be filed after the Zone Change application is approved.
Installation of Power Lines and Substations	<u>State Public Utilities</u> Commission	Currently coordinating with HECO. Will be approved prior to occupancy.
National Pollutant Discharge Elimination System (NPDES) Permit	State Department of Health	Application to be submitted prior to Grading Permits.

PERMITTING - GRAPHIC DESIGN PLAKNING • LANDSCAPF ARCHITECTHRE • INVIRONMENTAL STUDILS • ENTERTMENTS

O'AHU (UH WEST O'AHU) DRAFT Ms. Genevieve Salmonson SUBJECT: UNIVERSITY OF HAWAI'I WEST (ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Section 5.2.2 of the Final EIS will be updated to include an expanded discussion in reference to guidelines of the HRS 344 that are applicable to the development of UH West O'ahu as follows: ų

Chapter 344, Hawaii Revised Statutes (HRS) establishes a State policy to encourage productive and enjoyable harmony between people and their environment, promote efforts that prevent or eliminate damage to the environment and biosphere and stimulate the health and welfare of humanity, and enrich the understanding of ecological systems and natural resources important to the people of Hawai't. Guidelines of the policy that are applicable to the UH West O'ahu development are discussed below.

LAND, WATER, MINERAL, VISUAL, AIR, AND OTHER NATURAL RESOURCES

- Encourage management practices which conserve and fully utilize all natural resources. ₹
- Promote irrigation and waste water management practices which conserve and fully utilize vital water resources. (B)
- Promote the recycling of waste water. Ũ
- Promote the optimal use of solid wastes through programs of waste prevention, energy resource recovery, and recycling so that all our wastes become utilized. 6

DISCUSSION: Sustainability guidelines have been established for the UH West O'ohu. The guidelines strive to achieve a Leadership in Energy and Environmental Design (LEED) NC silver standard for campus development and community performance standards based on the LEED NC v2.2 rating system. The Private Development Lands will also strive to achieve the applicable design criteria in the UH West O'ahu LRDP and the recommended community performance standards.

FLORA AND FAUNA

- (A) Protect endangered species of indigenous plants and animals and introduce new plants or animals only upon assurance of negligible ecological hazard.
- Foster the planting of native as well as other trees, shrubs, and flowering plants compatible to the enhancement of our environment. (B)

identified in the Hohtat Conservation Plan for Abutilon menziesii at Kapolei (HCP) prepared in consultation with the State DLNR. On April 8, 2004, the BLNR unanimously approved the HCD and accompanying lanctedual Toke License and Certificate of Induction. The University of Hawai' is currently working with the DOT to be induced in the Certificate of Inclusion for the Incidental Take License. Long term impact on the Abutilon menziesti DISCUSSION: A single plant of endangered ko'oloa'ula (Abutilon menziesii) was found within the property in 1997 flora survey but has since died. In comparison larger plant clusters were found within the alignment of the North-South Road and adjoining DHHL. affect the Abutilon menziesti plant clusters. DOT prepared a Habitat Conservation Plan. In accordance with both. State and Federal regulations, mitigation, measures have been properties. Since the North-South Road was the first project to be constructed which woul

M5. Genevicve Salmonson SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Page 4	TRANSPORTATION (A) Encourage transportation systems in harmony with the lifestyle of the people and	environment of the State. (B) Adopt guidelines to alleviate environmental degradation caused by motor vehicles.	DISCUSSION: The plan for UHWO addresses the need to expand and improve the transportation. system and close the gap between where people live and work through decentralization, mixed zoning, and related initiatives. The UH West O'ahu will provide education and employment opportunities in the 'Ewa region. The compust is conveniently located near major transportation facilities and residential developments, thus encourasing	public transportation and valking or bicycling. West O'ahu residents will be able to live, work, and attend school within this region, alleviating traffic to other universities and workplaces in Honolulu.	<u>The UH West O'ahu campus will help to alleviate environmental degradation by reducing</u> commuting time and distances for students, staff and faculty residing in the 'Ewa region,	West O dut residents will have better access to higher elucation facilities, and utimately, the compus will connect to a transit corridor planned adjacent to the North-Sauth Road, Within the campus, placetarian valkoarus, facilities for bicycles, and facilities for distance education will encourage non-motorized vehicular forms of transportation and reduce fossil	j <u>uet consumption.</u> ENERGY	(A) Encourage the efficient use of energy resources.	DISCUSSION: The UH West O'ahu campus will help to conserve existing energy resources by reducing commuting time and distances for students and faculty residing in the 'Evar region. West O'ahu residents will have better access to higher education facilities, and ultimately the campus will connect, to a transit corridor planned algacent to the North-Scinth Road. Within the campus, pedestrian walkways, facilities for bicycles, and facilities for thicycles, and facilities for this education will encourge non-motived vehicular forms of transportation and reduce fossit fuel consumption. The use of energy-efficient equipment for lighting systems and hor water fuel consumption.	<u>uearny, and coorny wut aiso of encouraged within the compus. The State's Model therevy</u> Code will be considered during the detailed design phases of project development.	Sustainability guidelines have been established for the UH West O'ahu Lands (see Section 2.61. The guidelines strive to achieve a Leadership in Energy and Environmental Design OF FDD) NG vitage actions for the section of the sectio	terior INC states stated of compart are everyphene and community performance standards based on the LEED VC 22 anging system. The Private Development Lands will strive to achieve the applicable design criteria in the UH Wext O and the recommended community performance standards.	COMMUNITY LIFE AND HOUSING	(A) Foster lifestyles compatible with the environment; preserve the variety of lifestyles traditional to Hawaii through the design and maintenance of neighborhoods which reflect the culture and mores of the community.
Ms. Genevieve Salmonson SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Page 3	population in Kapolei will be beneficial, as proposed mitigation measures in the HCP will ensure the future propagation of new plants. If an off-site preserve can be established, some of the plants would remain in situ for many years. Because the protection, propagation, and	relocation of the kc olog thin plant is a long-term undertaking. final implementation of the HCP will extend well into the project's construction period. Once construction and build-out of the campus is complete, use of the ko olog vul in project inducating and computed we of the ko olog vula nutrisery for morroordin will enviro a much house and viscome securities.	of the ko'oloor luta than would have occurred without development of the veryone population property. The HCP also proposes long-term management of the populations to occur concurrently with project development, over a period of approximately 20 years. The successful implementation of the HCP would significantly increase the numbers of new plants on O altu and improve the quality of existing populations on adjacent properties.	As noted in Section 2.6 of this Draft EIS, one of the sustainability guidelines for UHWO is to "create habitat that relies on native species, enhances the campus experience and celebrates the region's agrarian roots."	PARKS, RECREATION, AND OPEN SPACE	(A) Establish, preserve and maintain scenic, listoric, cultural, park and recreation areas, including the shorelines, for public recreational, educational, and scientific uses.	(C) Promote open space in view of its natural beauty not only as a natural resource but as an enviobiling, living environment for its people.	DISCUSSION: A hierarchy of open spaces will be provided within the campus and surrounding, residential developments. The commus onen score elements include a control	plata, informal courtyards, and open, spaces. Two predering more actions manues to comp flor safety, consistent paving patterns and materials, and site furnishings will also help to define the open space network. The campus will be connected to regional open space systems adjacent to the compus through the Humehune and Kaloi 9 greenwoy, which will extend from the campus to the surrounding community. Maukarmakai views, thom the property will be enhanced. focusting on views of the Wai'came and Ko'o lau. Mountains, and landscape treatments and building design will mitigate visual impacts of the campus.	ECONOMIC DEVELOPMENT	(B) Promote and foster the agricultural industry of the State; and preserve and conserve productive agricultural lands.	<u>DISCUSSION: Currently cultivated lands within the property will be withdrawn from</u> agricultural production for the proposed development. This will result in some loss in revenues, jobs, or pervolt. However, the State and Civ have loss planned for new development in the project and some newsivity within the Libert for new	District. Tenaus has project one or projectly a write, or organ state data one or District. Tenaus have been fully aware, for quite some time, that the proposed project area would be used to accommode for future thereforment in the regime.	

Ms. Genevieve Salmonson SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Page 6	CITIZEN PARTICIPATION	(A) Encourage all individuals in the State to adopt a moral ethic to respect the natural environment; to reduce waste and excessive consumption; and to fulfil the responsibility as trustees of the environment for the present and succeeding generations.	<u>DISCUSSION: The University of Hawai't has been a leader in recycling and will continue this</u> effort at its UHWO compus.	 Thank you for referring us to HRS 103D-407. As required, the University will utilize the use of recycled glass in paving materials whenever possible. This information will be provided in a new Section 5.2.7, 103D-407 Harvaii Revised Statures of the Final FIS and read as follows: 	Havaii Revised Statues 103D-407 requires the use of recycled glass in paving materials wherever possible, specifically;	(a) When purchasing roadway materials or other high-value, end-use applications for public projects, state agencies shald, and county agencies may, purchase materials with minimum recycled glass content meeting specifications adopted by the policy board which, at a minimum, shall provide for:	(1) A minimum recycled glass content of ten per cent crushed aggregate in treated ar untreated basecourse in paving materials that shall not reduce the quality standards for highway and road construction: and	<u>(2) The use of one hundred per cent aspregate in nonstructural capital improvement</u> <u>applications.</u>	(b) All highway and road construction and improvement projects funded by the State or a county or roadways that are to be accepted by the State or a county as public roads shall utilize a minimum of ten per cent crutched placs accreate as specified by the devorment of	transportation in all basecourse (treated or untreated) and subbase when the glass is available to the quarry or contractor at a price no greater than that of the equivalent aggregate.	(c) All state and county construction projects calling for nonstructural backfill shall utilize one hundred per cent crushed plass when available at a cost equal to or lower than the	<u>equivalent assergate.</u> It is the Hiniwersity of Houorit's intent to convely with the above have	4. Thank you for referring us to HRS 103D-408. As required, the University will landscape the property with native Hawaiian flora whenever and wherever possible. As recommended, the University will consider the use of un-thirsty plants whenever possible. This information will be provided in a new Section 5.2.8, 103D-408 Hawaii Revised Statutes of the Final EIS and read as follows.
Ms. Genevieve Salmonson SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Page 5	(B) Develop communities which provide a sense of identity and social satisfaction in	harmony with the environmental and provide internal opportunities for shopping, employment, education, and recreation. (C) Encourage the reduction of environmental pollution which may deprode a	community. (D) Foster safe, sanitary, and decent homes.	(E) Recognize community appearances as major economic and aesthetic assets of the counties and the State; encourage green belts, plantings, and landscape plans and designs in urban area; and preserve and promote mountain-10-ocean vistas.	DISCUSSION: The proposed project presents a unique opportunity to create a new campus that addresses the need to create more of a resident campus with more student housing opportunities than normally provided by most universities and a "covm-covm' compute form lifestyle, as well as housing to opportunities to attract the hest faculty and stuff. The mixersity	compus-focus of a residential mixed-use project will attract residents who are interested in not only in their childrens' education, but in cultural events and life-long learning opportunities that UHWO will present. The project has been designed to provide internal opportunities for shapping, employment, education and recreation.	As noted above, one of the goals of the project is to achieve LEED NC silver standard as a way to reduce environmental pollution. All housing on the project site will be designed to foster safe, sanitary and decent homes.	Kalo [†] Greenway and Hunehune Greenway will include a pedestrianbike path that will connect the campus with residential and mixed-use commercial developments surrounding the campus. The greenway system is also envisioned to link the two proposed transit stations	on the eastern portion of the property (adjacent to the North-South Road) with the University and surrounding mixed-use and residential communities. The greenway system, for the most part, will be unimpeded by vehicular traffic.	Plazas, courtyards, other landscaped areas, and the 10-acre detention basin will provide open space within the campus, allowing views towards the Wai'anae and Ko'olau mountain ranges.	EDUCATION AND CULTURE	(A) Poster culture and the arts and promote their linkage to the enhancement of the environment.	DISCUSSION: While the residential population within Kapolei has grown as planned, it lacks much of the culture and the arts which are available within the Primary Urban Center. The university campus-focus of a residential mixed-use project will attract residents who are interested in not only in their childrens' education, but in cultural events and life-long learning opportunities that UHWO will present.

Ms. Genevieve Salmonson SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Page 8	campus size, and the UH West O'ahu curriculum. The issue was referred to the 'Ewa Neighborhood Board Planning and Zoning Committee.	 'Ewa Neighborhood Board Planning and Zoning Committee (July 23, 2002) <u>– Issues</u> raised during the meeting focused on the positive and negative aspects of each site under consideration, including concerns regarding reaffic, development cost, onlich sites previously considered, roadswys, infrastructure, liming of construction, building heights and hutre campus considered, roadswite, Other issues raised included conserns regarding the UH West O'ohu curriculum. The Committee voted in support of the Kapolei Makai (current) 	one. • Makakilo/Kapolei/Honokai Hale Neighborhood Board (July 31, 2002) <u>- Issues regarding</u> <u>various sites for consideration, including future compus stze and expansion potential</u> ,	utatus <u>e of construction, reatific, building hergens, endangered plants, costs, parking, and</u> <u>funding, were raised during the meeting. Other issues raised included concerns regarding</u> <u>community input from other areas, and student housing. At the time, no voie was regarding</u> Baard members vortheir raisions chout the mycociety formation	the Kapolet Makai (current) site, • Makakito/Kapolet Honokai Hale Neighborhood Board (August 28, 2002) <u>- The Board</u> decided to take a position on the Incrition of the IIH West O'ahu Cumure The Board	 voted in support of the Kapolei Makai (current) site. University of Hawai't Board of Regents (September 13, 2002) - The UH Board of Regents approved the designation of the 500-acre Kapolei Makai (current) site as the location for a permanent UH West O'alu (Campa). 'Ewas discretes the physical Board (Phyruar D) 2001 - Icures discretes during the maximum of the section of the 2001 - Icures discretes during the maximum. 	included the feasibility of various sites for the location of the campus. UH West O'ahu curriculum, and funding initiatives for the campus. The Board voted in support of a resolution, strongly endorwing the need to invasion commence highly a summence	 <u>four-year UH West Or aim Compus.</u> <u>University of Howai'i Board of Regents (July 16, 2004) - The UH Board of Regents</u> approved in principle the 2004 Long Range Development Plan (LRDP). <u>Makakila/KapoleiHorokai Hale Neizhborhood Board (Aurust 25, 2004) - Ixeus raised</u> 	during, this meeting, included concerns regarding the location of the compus, phasing, project funding, development partnerships, student housing, roadways, infrastracture, and future compassion.	e <u>transmentationen taate Negritorin Joan Januery 1, 2005) – Kues raised</u> during hits meeting included concents regarding roadwors, student housing, project funding, the UH West O ahu curriculum, accreditation, increasing, attendance,	development partnership, expansion on other UH West O'ahu property, and providing support at the legislature for the campus. • Mai'arae Neighborhood Board (March 1, 2005) – Issues raised during the meeting	included concerns, regarding, project, generated, traffic, available residential/work force housting, student houving, and the location of the compus. • University of Hawai? Board of Reservis (hug 20-21, 200) – The UH Board of Regents approved an undate of the 1114 Weer O'chin Comvue	 MakakitoKapolei(Honokai Hale Neighborhood Board (July 26, 2006) – The University of Hawai'i West O' anu and its consultants gave a presentation regarding the UH West O'ahu's intent to pursue a Plan Review Use and Zone Change Application. Issues raised during the meeting included concerns regarding roadways, parking on compus, development of other UH West O'ahu property, campus site, compus, architecture, open accession of the set of the UH West O'ahu property. 	space and actention areas, project pupans, available studentwork force housing.
Ms. Genevieve Salmonson SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Page 7	<u>Hawaii Revised Stantes 103D-408 requires the use of native Hawaiian flora whenever and</u> wherever possible, specifically:	(a) Wherever and whenever feasible, all plans, designs, and specifications for new or removated lendscoping of any building, complex of buildings, facility, complex of facilities, or housing developed by the State with public moneys shall incorporate indigenous land plant species as defined in section 195D-2, and plant species brought to Hawaii by Polymerians before European contact, such as the kukui, noni, and coconut, provided that:	(1) Suitable cultivated plants can be made available for this purpose without jeopardizing wild plants in their natural habitat; and	(2) Wherever and vytenever possible, indigenous, plants shall be used for landscaping on the island or islands on which the species originated.	(b) Each plant or group of plants used pursuant to subsection (a) shall be clearly identified with signs for the edification of the general public. [L 1999, c 149, pt of 82]	It is the University of Hawait's intent to comply with the above law and to further comply with sustainability principles by: 1) selecting plants suited to the climate of the project site. 2) specifying plant materials, that would be irrigated, with similar irrigation requirements; and 3) using non-drinkable water for irrigation, where fassible.	5. Section 8.1.1 Individuals/Organizations Consulted During the Site Selection Process of the EIS will be revised to read as follows:	The University of Hawai'i and its consultants made presentations to various community groups, neighborhood boards, community associations, and government agencies as described below:	•-University of Hawai': Board of Regents; • Makakilo/Kapolei/Honokai Hale Neighborhood Board Planning and Zoning Committee (June 6, 2002) <u>- Issues raised during the meeting focused on the positive and negative</u>	uspects of each site, including concerns related to impacts from surrounding land uses, building relations, refister, rotadivers, parcel size, estrantion potential, feasibility, of development on other sites, notivery on commes and notion. Only review encloaded	concerns regarding project funding, student housing, and the proposed UH West O'ahu curriculum. The Committee noted that they would like to have a presentation given to the Makakilo/Karolei/Honokai Hale Neichthorhood Rorrd	 Villages of Kapolei Association (June 26, 2002) — Issues raised during the meeting focused on the positive and negative aspects of each site under consideration, including concerns trearding traffic, roadways, pollution impacts, parcel size and development 	 cours. Unlet Using Instant and the UH West O'ahu curriculum. Although now other areas, the ultimate community input from other areas, the ultimate community in a strain of the UH West O'ahu curriculum. Although no vote was taken, the Kapoleh Mokai (current) site appeared to be the most popular choice. 'Ewa Neighborhood Board (Juty 11, 2002) – Issues raised during the meeting focused on the positive and argative aspects: of each site anuder consideration, including concerns researding traffic and roadwors, comput, growth and expansion, costs, and building concerns 	heights. Other issues raised included concerns regarding student housing, the ultimate

Ms. Genevieve Salmonson SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Page 9

- information on public presentations, the proposed DOE school, and the proposed UH West O'ahu curriculum. The Board voted in fravor to support the UH West O'ahu Campus. Even Neighbortood Board voted in fravor to support the UH West O'ahu Campus. Bud its consultants give a presentation regarding the UH West O'ahu's intent to pursue a Plan Review Use and Zone Change Application, fesues raised during the meeting included concerns regarding roadways, opportunities for community input, and affordable housing. •
- As recommended, Table 8.2 of the Final EIS, *Environmental Impact Statement Preparation Notice* (EISPN) Comment Letters will be updating to include the Department of Design & Construction as a commenter as follows: ġ.

omment Letters	
EISPN C	
Table 8.2 -	

AGENCY AGENCY DATE DATE DATE OF ic DAPPOINTMENTS DATE COMMENTS DATE COMMENTS ic Depentment of Business, Economic Development and Tourism 02-01-05 Department of Education 02-01-05 03-07-05 Department of Headth Office of Planning Office 02-01-05		1	r—	1	T	T	····	1	1	1	1		T	·····	Τ	T	1-	1-	—	F	r	r	T	T		···· F	_	
AGENCY AGENCY MAIL te Department of Business, Economic Development and Tourism 02.01.05 Department of Hautiscies Division 02.01.05 Department of Health - Environmental Planning Office 02.01.05 Department of Health - Office of Environmental Quality 02.01.05 Department of Health - Office of Environmental Quality 02.01.05 Department of Health - Office of Environmental Quality 02.01.05 Department of Itanyoision 02.01.05 Department of Health - Environmental Quality 02.01.05 Department of Transportation 02.01.05 Department of Transportation 02.01.05 Department of Transportation 02.01.05 Department of Transportation 02.01	DATE OF COMMENTS		:		03-07-05			03-02-05	-		02-08-05	02-07-05	a +	:		**			03-04-05	-	•	:					-	ł
te Department of Business, Economic Development and Tourism Department of Business, Economic Development and Department of Business, Economic Development and Tourism – Office of Planning Department of Business, Economic Development and Tourism – Office of Planning Department of Business, Economic Development and Tourism – Office of Planning Department of Business, Economic Development and Department of Education Department of Haudih – Office of Environmental Quality Control Department of Haudh – Office of Environmental Quality Control Department of Tanagoration Department of Transportation Department of Business, Economic Development and Tourism Department of Business, Economic Development and Tourism	EISPN MAIL DATE		02-01-05	02-01-05	02-01-05	02-01-05	02-01-05	02-01-05	02-01-05	02-01-05	02-01-05	02-01-05	02-01-05	02-01-05	02-01-05	02-01-05	02-01-05	02-01-05	;	02-01-05	02-01-05	02-01-05	02-01-05	02-01-05	02-01-05		50-10-20	02-01-05
3 3 33 <u>33 3 3 8 1 9 2 7 3 3 5 9 8 1 0 2 8 1 0 7 7 8 8 1 0 7 8 8 1 0 8 1 0 1 0</u>	AGENCY	State	1 Department of Business, Economic Development and Tourism	2 Department of Business, Economic Development and Tourism - Land Use Commission	3 Department of Business, Economic Development and Tourism - Office of Planning	4 Department of Business, Economic Development and Tourism - Choncer Induction Division	5 Department of Defense	6 Department of Education	7 Department of Hawaiian Home Lands	8 Department of Health – Environmental Planning Office	9 Department of Health – Office of Environmental Quality Control	10 Department of Land and Natural Resources	11 Department of Land and Natural Resources – Historic Preservation Division	12 Department of Transportation	13 Department of Transportation – Airports Division	14 Office of Hawaiian Affairs	15 State Legislature – Representative Mark Moses	16 State Legislature – Senator Brian Kanno	17 State Legislature – Senator Will Espero	18 University of Hawai'i – Environmental Center	19 Hawai'i State Library	20 'Ewa Beach Public and School Library	21 Kapolei Public Library	22 University of Hawai'i – Hamilton Library	23 Department of Business, Economic Development and Tourism		44 Trefinitive Reference Durcan	25 City and County of Honolulu Department of Customer Services Library (formerly the Municipal Reference and Records)

Ms. Genevieve Salmonson SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006

Page 10

AGENCY	EISPN MAIL DATE	DATE OF COMMENTS
Center)		
Federal		
26 Department of the Army - Army Engineer District	02-01-05	03-01-05
27 Department of the Interior – Fish and Wildlife Service	02-01-05	1
28 Department of the Navy	02-01-05	-
City		
29 Board of Water Supply	02-01-05	02-14-05
30 City Council – Councilmember Todd Apo	02-01-05	;
31 Department of Community Services	02-01-05	02-22-05
32 Department of Design and Construction	02-01-05	03-21-05
33 Department of Environmental Services	02-01-05	
34 Department of Facility Maintenance	02-01-05	03-21-05
35 Department of Parks and Recreation	02-01-05	02-14-05
36 Department of Planning and Permitting	02-01-05	03-14-05
37 Department of Transportation Services	02-01-05	03-11-05
38 'Ewa Neighborhood Board, No. 23	02-01-05	-
39 Fire Department	02-01-05	02-25-05
40 Makakilo/Kapolei Neighborhood Board, No. 34	02-01-05	1
41 Police Department	02-01-05	03-02-05
42 Wai'anae Neighborhood Board, No. 24	02-01-05	
Other Organizations/Individuals		
43 Hawaiian Electric Company, Inc.	02-01-05	1
44 Palehua Community Association	02-01-05	-
45 The Estate of James Campbell	02-01-05	
46 The Gentry Companies	02-01-05	1

West O'abu Campus Development, LLC (WOCD) will be included in a list of terms, abbreviations, and acronyms in Section 10.0 of the Final EIS as follows: ~

	Western Association of Schools and Colleges	Waialua Silty Clay, 0 to 3 percent slopes	West Oahu Campus Development, I.LC	Wastewater Treatment Plant	Waipahu Silty Clay, 0 to 2 percent
W	WASC	WkA	WOCD	WWTP	V2M

We acknowledge that an EISPN was withdrawn in 1998 and a new EISPN was submitted in February 2005. As recommended, a summary of the events that resulted in the withdrawal and re-submittal of the EISPN will be inserted in front of the first paragraph of Section 8.1.1. Individuals/Organizations Consulted During the Pre-Consultation Process of the Final EIS as follows: ø

In 1967, the UH Board of Regents initiated efforts to establish a four-year college in Central or West O'ahu. Over the years, the Regents approved varying sites: Wai'awa Ridge (1970), Honouliuli/Ewa (1973) and in 1993, Kapolei Makai (the current site), In 1996, it was

Ms. Genevieve Salmonson SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Page 12	panoramic photos on a revised Figure 4.1. As mentioned in the paragraph above, there are no available drawings or diagrams of the proposed buildings and any proposed landscaping that would show the final appearance of even a small portion of the campus and development lands. However, we believe from the additional photographs that we have taken, that the proposed development would have a greater visual impact on makai viewplanes from Farrington Highway, than the H-1 Freeway, because the H-1 Freeway is elevated above the UHWO site. There are no other public roads that abut the project site other than the Kapolei Golf Course entry road, and that road is tree-lined with canopy trees. Please refer to the attached revised Figure 4.1a	11. The placement of underground utility lines will be discussed in Section 4.7, Visual Resources and Open Space, Anticipated Impacts and Mitigation Measures of the Final EIS and the discussion will be revised to read as follows: The visual appearance of the UH West O'ahu property will change from vacant scrub and cultivated vegetation to a compus and mixed-use community. Distant views of the Shoreline from the H-1 Freeway may be impacted by the proposed development; however, the UH West O'ahu will schange from water as an important visual landmark for the 'Ewa region. Within the property of a cump of ethylicine development; however, the UH West O'a cump of the science of the Mest O'a the science of the Shoreline from the H-1 Freeway may be impacted by the proposed development; however, the UH West O'a cump of ethylicine of the Mest O'abu the A-1 freeway may be impacted by the proposed development; however, the UH West O'a the visual of the Shoreline from the H-1 Freeway may be impacted by the proposed development; however, the UH West O'abu will science of the Mest O'abu will be science of the Mest O'abu the the transforment from the H-1 Freeway may be impacted by the proposed development; however, the UH West O'abu will science of the Mest O'abu the transforment from transforment from the transforment from transforment from transforment	The compus will be oriented to comparize our views of and/orms such as performent parameters parameters provident in the compus will be oriented to considered to the extent possible. Extensive landscaping, campus view corridors, and thoughtful architectural design will add to the visual character of the area. HECO has yet to determine whether power lines to the substations and power lines. extending out from the substations will be overlead or underground. Should HECO decide on implementing overhead power lines, there will be possible visual impacts on the project prostory and the possible visual impacts on the project prostory.	12. Park dedication is required and will be addressed in Sections 4.8.2 and 4.11.5 of the Final EIS. The University and/or the private developer of residential units associated with the project will meet the affected City departments to discuss plans to satisfy the City's Park Dedication Ordinance requirements. The second paragraph of Section 4.8.2. <i>Housing, Anticipated Impacts and Mititactive Measures</i> of the The Second paragraph of Section 4.8.2. <i>Housing, Anticipated Impacts and Mititactive Measures</i> of the Park Second paragraph of Section 4.8.2. <i>Housing, Anticipated Impacts and Mititactive Measures</i> of the Park Second paragraph of Section 4.8.2. <i>Housing, Anticipated Impacts and Mititactive Measures</i> of the Park Second paragraph of Section 4.8.2. <i>Housing, Anticipated Impacts and Mititactive Measures</i> of the Park Second paragraph of Section 4.8.2. <i>Housing, Anticipated Impacts and Mititactive Measures</i> of the Park Second paragraph of Section 4.8.2. <i>Housing, Anticipated Impacts and Mititactive Measures</i> of the Park Second paragraph of Section 4.8.2. <i>Housing, Anticipated Impacts and Mititactive Measures</i> of the Park Second paragraph of Section 4.8.2. <i>Housing, Anticipated Impacts and Mititactive Measures</i> of the Park Second paragraph of Section 4.8.2. <i>Housing, Anticipated Impacts and Mititactive Measures</i> of the Park Second paragraph of Section 4.8.2. <i>Housing, Anticipated Impacts and Miticactive Measures</i> of the Park Second paragraph of Section 4.8.2. <i>Housing, Anticipated Impacts and Miticactive Measures</i> of the Park Second paragraph of Section 4.8.2. <i>Housing, Anticipated Impacts and Miticactive Measures</i> of the Park Second paragraph of Section 4.8.2. <i>Housing, Anticipated Impacts and Miticactive Measures</i> of the Park Second Pa	EIS will be revised as follows: Park areas and open space will be provided throughout the property in compliance with the City's Park Dedication Ordinance. <u>The University and/or the private developer will meet</u>	will allected Lity departments to discuss specific plants to satisfy the park dedication requirements. For meet the City's affordable housing requirements, approximately 355 units, a large portion of which will be affordable nonsing multi be provided within the UH West O'ahu Eands. The remainder of the affordable units required for the development will be distributed throughout residential porcels (and not concentrated in one or two arcord) in the Private Development Lands.	The project will address the provision of affordable housing by coordinatine with the Department of Planning and Permitting and the Department of Community Services. UHWO is particularly molivated that the project provides antainable housing for faculty and staff as it is difficult to attract the best condidents without it. In addition, UHWO will eventually include approximately 200 less student housing with approximately 3 beds each. Thus there will be approximately 2.280 less students that would be competing for affordable rentals in will be approximately 2.280 less students that would be competing for affordable rentals in
Ms. Genevieve Salmonson SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Page 11	decided that the West O'ahu campus site would be moved to 991 acres mauka of the H-1 Freewory in Kapolei. This war based on a strategy by the State to entitle approximately, 1,300 acress of land in East Kapolei, which would be then be sold to private developers. As a result, the Housing and Community Development Corporation of Havati (HCDCH) initiated a matter planuing effort for 1,300 acress including the UHWO Kapolei Makai site for mostly residential uses. It was intended that the proceeds of the land safer would be used to find the construction of a UHWO compus site marka of the H-1 Freeway. In 1998, an ElS Preparation Matter 5, 1000 HCDCH field of Destributed to the Environmental Notice. On March 5, 1000 HCDCH field of Destributed to the Environmental Notice.	Arrentation of the second second second and a feature of second s	During the full of 2001, the UH Administration, the Estate of James Campbell, and the Governor explored the possibility of relocating the UH West O'ahu campus from the 991- acre mauka site to the City of Kapolei. Since the marketability of the petition area was non-existent, and a site selection study determined that the current site was best suited for a new UHWO campus, UHWO was allocated 500 acres and most of the remainder of the 1,300 acres which were "urbanized"	man transfer ten to 2010. On September 13, 2002, the UH Board of Resents approved the 500-acre Kapolei Makai site as the the S00 acres from DLNR. The Board of Land and Natural Resources approved the transfer of title of the 500, acres site to the UH in fee simple. As a result, 1998 EIS Preparation Notice was withdrawn and new one submitted in February 2005.	The University of Hawai't still retains the 991-acre site manka of the H-J. Freeway, but has no current plans for it other than being generally viewed as potentially being developed in the future as an expansion area for the UHWO compus,	9. While a few drawings and/or diagrams of the proposed buildings and landscaping have been prepared that illustrate the possible appearance of portions of the proposed project, these older renderings will not be included in the Final EIS as UHWO's current architect, John Hara & Associates, strongly feels that the inclusion of such graphics would be misleading and will not be representative of the final appearance of even a small portion of the campus. Also, since the agreement with the private developer has not been consummated as of this writing, there are no drawings or diagrams of	proposed buildings and any proposed landscaping that would show the final appearance of the development lands. development lands. 10. We concur that distant views of the shoreline from manka and makai view planes may be impacted by the proposed development. In response to your comments, we have taken additional photographs of the project site from both Farrington Highway and the H-1 Freeway and will be including these

 MS. Genevieve Salmonson U) DRAFT DRAFT SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Page 14 	cted Cir. plans to plans to protection <i>"reative geness"</i> . The the plans the plans of the plans of the plans of the plans plans the plans of the plans plans of the plans of the plans plans of the plans of the p	 ver could 15. When demand warrants it, UHWO will be providing a shuttle service on campus as an alternative form of transportation. The following one-sentence paragraph will be added at the end of Section 4.10.1 Roadways and Traffic, Anticipated Impacts and Mitigation Measures, Public Transit of the Final EIS: 	vs. The diffics in the service on compus as an alternative form of the providing a shuttle service on compus as an alternative form of the service is a critical mass of students, faculty and staff.	e private the park the park duestions regarding this project, please do not hesitate to contact me at 52.1-5631. Students, Courts, Sincerely,	ng paragraph , Anticipated	<i>t before</i> <i>t before</i> <i>At before</i> <i>Vice</i> President <i>Vice</i> President <i>St. Also</i> Encl: Figure 4.1a – Visual Resources	uilize a uilize a Bene Awakuni/UH West O'ahu glass té uivaleni vuivaleni	lanuing and sisible within t this area as Ko Olina are ees, vendors ees, vendors
Ms. Genevieve Salmonson SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AH ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Page 13	the open market. The University and/or the private developer will meet with after departments, including the Department of Community. Scrvices to discuss specific sonisfy the affordable housing requirements. During public review period, the Depa Planning and Permitting words the following, regarding the potential of developing flat" type revist units in proposed the following, regarding the potential of developing flat" to consider the benefits (i.e., low cost student remals and increased housing unit affordability) through this type of housing to be substantial. It appears than an opportunity for th unit exists where the "alley loaded" low-density residential development on Parcels where such units could take the form of studio units above the game. This approv of these studies the flat the need for approximately 2.280 beats for the Z600-student population of little or no cost to the University to construct "gramy flats," low you	to meet interving residential district standards for two (2) dwellings, or a develop use the Planned Development Housing (PD-H) option." The first paragraph of Soction 4.11.5 Recreational Facilities, Anticipated Impacts an Measures of the EIS will be revised as follows:	The demand for recreational facilities in 'Ewa will increase as the population gro proposed UH West O'ahu development will provide additional fac- counting and the Civit on the District of the counter	computative with the Uty's Fark Dedication Uramance. In University and/or the developer will meet with affected City departments to discuss specific plans to satisfy dedication requirements, as required by Section 22-7. Review Orthinances of Honolu recreational facilities within the proposed campus are planned for use primarily by faculty, and staff. Such facilities could include tennis courts, basketballycolleybal and jogging paths.	13. As recommended, a construction waste recycling plan will be established. The followin will be inserted after the fitter paragraph of Section 4.10.5 Solid Waste Disposal Facilities Impacts and Mitigation Measures of the Finat EIS:	Per comments received from the State Office of Environmental Quality Control du Public Review Period. a construction waste recycling plan will be prepared construction is initiated. All solid waste generated during project construction directed to a Department of Heath permitted solid waste disposal for recycling facilit all bioway and condervation	roadways that are to be accompany unprovement protects protect or are or a county as public roads shall ninimum of ten percent crushed glass aggregate as specified by the Depart Transportation in all base-course (treated or untreated) and sub-base when the aggregate.	14. Per your letter, we have discussed this possible land use with the Department of P Permitting (DPP) and at least preliminary, DPP believes an extended-stay facility is pos- tife campus area and permitted with a Plan Review Use approval. This use is necessary in the closest resort-zoned area is Ko Olina, and the type of ploging facilities available in more expensive than the budget usually allowed visiting professors, conference attend and other visitors to the campuses of the University of Hawai'i. The description of Stud

FETRA Y, YOJMG CONSTRUCTION OF A CONSTRUCT RESOLUTION CONSTRUCT AN OLDER OF A CONSTRUCT RESOLUTION CONSTRUCT AN OLDER OF A CONSTRUCT AND A CONSTRUCT AN OLDER OF A CONSTRUCT ANA OLDER AND A CONSTRUCT ANA OLDER OF A CONSTRUCT ANA OLDER AND A CONSTRUCT ANA OLDER OF A CONSTRUCT ANA OLDER AND A CONSTRUCT ANA OLDER OF A CONSTRUCT ANA OLDER OF A CONSTRUCT ANA OLDER AND A CONSTRUCT ANA OLDER OF A CONSTRUCT ANA	LOG NO: 2006.2659 DOC NO: 0608AJ03 Arohaeology	-[]-[]-	eived a copy of the draft 06. We apologize for the of Hawai'i, West O'ahu	issance and assessment of tober 23, 1996 (Appendix <i>ec and Assessment of the</i> <i>archaeological inventory</i> <i>dditionally, portions of</i> <i>le at the State Historic</i> , <i>traditional and historic</i> , <i>ect</i> (APE) associated with <i>ly-significant remnants of</i>	ssuance of permits for the us a listing of local firms consultants to contact us, warded to us for review) is of HAR Chapter 13-13-
STATE OF HAWAI DEPARTMENT OF LAND AND MATURAL RESOURCES STATE HISTORIC PRESERVATION DIVISION STATE HISTORIC PRESERVATION DIVISION SOLIEL, HAWAII 95007	RECEIVED Au6 10 2005 PBR HAWAII	r GE-8 Historic Preservation Review [University of I uvironmental Impact Statement if of Hawai'i – West O'ahu Campus li'uli Ahupua'a, 'Ewa District, Jaland of O'ahu 1) 9-1-016:120, 127, and 129	prtunity to review the aforementioned project. We rev Statement (EIS) and a brief cover fatter on July 6, 20 dertaking involves the development of the University subject property.	n provided in the draft EIS, an archaeological reconna conducted by Scientific Consultant Services Inc. on O. iment (Spears, R.L 1996. Archaeological Reconnaissa uelopment Project) does not meet the requirements of a awai'i Administrative Rules (HAR) 13-13-276. A larve been demonstrated, by various reports on f SHPD), to contain historically-significant resources s/burials. Also, we believe that the area of potential efi as a moderate to high potential of containing historica neutre (e.g., irrigation systems).	haeological inventory survey is warranted prior to the he Hawai' State Preservation Division website conta dibr/hpd/archeon.htm). We recommend archaeological sare a basic inventory survey plan (which can be for are a basic inventory survey plan (which can be for in order to ensure that the study meets the requirement
Transverse construction and the second secon	August 8, 2006 Mr. Vincent Shigekuni PBR Hawai'i ASB Tower, Suite 650 1001 Bishop Street Honolulu, Hawai'i 9681 Dear Mr. Shigekuni:	SUBJECT: Chapte Draft E Univers Hono'u IMME.(Thank you for the oppo Environmental Impact S delay. The proposed un Campus on the 500-acre	According to informatio the subject property was E). However, this assess <i>HEDC_Exet Kapolet</i> Det survey pursuant to H Hono ^u li'uli <i>aniupua</i> ' Preservation Division (including human remain the current undertaking i sugar plantation infrastri	For these reasons, an arc proposed undertaking. T (http://www.hawali.gov/ or, alternatively, to prep <i>before</i> starting the work, 276.





D. View looking makai toward the proposed University of Hawai'i West O'ahu Campus site from Farrington Highway (180 degree panoramic view)



Figure 4.1a Visual Resources University of Hawai'i West O'ahu



NOT TO SCALE



PBR HAWAII & ASSOCIATES, INC.	December 18, 2006	W FRANK IRANDT, EASTA Chamman THOMASS WITTEX, ASIA Predian Restant Restant Vircent, Administrator Predian Restant Vircent, Administrator Ms. Melanie Chinen, Administrator Rash Administrator Administrator Restant Rest	VINCRETISHIGHAUNI In Providing SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AH Universitier DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) Privilyal DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) Privilyal Dear Ms. Chinen: Sound Alsoch Dear Ms. Chinen:	KEYTK KNINIKAWA, AMA We acknowledge that an archaeological inventory survey (AIS) is warranted prior to Aurearch Aurearin issuance of permits for the proposed undertaking. UH West O'ahu has contracted KMM MANUTURS.LITUPAR archaeological consultant to conduct an AIS, pursuant to HAR 13-13-276, which will Associate Associate completed before the issuance of any permits. SCOTTAINAADBACOO The text in Section 4.1 Archaeological and Historic Resources of the EIS will be reprint. Auronate Extisting Conditions	<u>During the DEIS Public Review Period, the Office of Hawaiian Affairs (OHA)</u> <u>wrate:</u>	HONOLULU OFFICE A brief search of OHA's records generated the following: Anti Bang Nucc Anti Anti Anti Bang Nucc Anti Anti Anti Bang Nucc Anti Anti Anti Anti Anti Anti Anti Anti	HLO OFFICE ACCORDING to records at the Bishop Museum pertaining to inventories conducted that Laswark conversions and the condition of the Native American Groves Protection and Repatriation Act that Transverses 2022, 2020, buried sites in Homoulinti and in "Ewa in general have been documented in the past including:	WALUNG OFFICE In 1938, human remains representing six individuals from Honouliuli. Tenson, Source 1. "Ewa, O'calu were collected by Remath P. Emory and William A. Lesso wathen 10: 25% out acquired by the Bishop Museum. Museum documentation indicates these remains were in a shallow cryst burial one mile from the coast;	<u>In 1933, human remains representing three individuals from stone pits at</u> 'Ewa, O'ahu were collected by J.W. Barrington and Edwin H. Bryan;
ł	(8) 692-8015 if you have any questions or concerns regarding this		Affairs						

Mr. Shigckuni Page 2

.

Aloha,

Rollin Houcen For Melanie Chinen, Administrator State Historic Preservation Division

AJ:

cc: Jesse Yorck, Office of Hawaiia

Ms. Melanie Chinen SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS)
December 18, 2006
Page 2

In 1942, human remains representing two individuals from Kualakai, 'Ewa Beach, O'ahu were donated to the Bishop Museum: In 1959, human remains representing seven individuals from "Eva, O'ahu were donated to the Bishop Museum by the Anthropology Club of the University of Howai? (from Standard Oil Refinery land); In 1980, human remains representing nine individuals from Honouliuli. O and were collected and donated to the Bishop Museum by Albert. Borthwick and Folk. Donor information indicates these human remains were recovered from coral sinkholes. In the last decade, unmarked burial sites have been found in the area of St. Francis West, West Loch Estates, Old Fort Weaver Road, Kalaelon, One vila Beach, Campbell Estate, Ko Olina and other areas in the vicinity of this project.

The depth of grading activities and the likelihood of adversely impacting any sub-surface cultural sites or deposits is contingent upon understanding the original surface grade as it may have existed prior to agricultural activities such as integrations. Native Hawaiian burial sites have been found just on and under the surface to dephis of eight-or nine feet depending upon the nature of the terrain. Furthermore, the nature of documented interments in the "Ewa area (stone pits sinkholes, crypts, etc.) could lead to the survival of these sites despite intensive agricultural activities on the survival of these sites despite intensive agricultural activities on the survival. OHA is recommending that the project area undergo an Archaeological liventory Survey (AIS), as a new study or supplement to the previous, in which all aspects of applicable 6E administrative rules and revised statutes are statisfied. It is our recommendion that the AIS include substantial subsurface inventory as morth of the surface architecture may have been distroyed during previous commercial agriculture activities. Also during the DEIS public review period, the State Historic Preservation Division wrote:

"...portions of Honouliuli alupuaa have been demonstrated, by various reports on file at the State Historic Preservation Division (SHPD), to contain historically-significant resources, traditional and historic, including human remainsformals. Also, we believe that the area of potential effect (APE) associated with the current undertaking has a moderate to high potential of associated with the current undertaking has a moderate to high potential of clear, irrigation prstents)." <u>For these reasons, an archaeological inventory survey is warranted prior to the issuance of permits for the proposed undertaking.</u>

Ms. Melanie Chinen SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Drafi EIS) December 18, 2006 Page 3

Anticipated Impacts and Mitigation Measures

<u>As recommended by OHA and SHPD, an archaeologist has been contracted to conduct an archaeological inventory survey. This AIS will be completed prior to the issuance of grading and building permits for the proposed project.</u>

Thank you again for your participation in the Environmental Impact Statement process for this project. Your comments will be included in the Final EIS.

If you have any questions regarding this project, please do not hesitate to contact me at \$221-5631.

Sincerely,

PBR HAWAII

Vine R. Suga

Vincent R. Shigekuni Vice President cc: Office of Environmental Quality Control Gene Awakuni/UH West O'ahu O:UOB13\1322.05\EISWinal EIS\Final Responses to DEIS Comments\BL-13 DLNR response 11-20.doc

SCIATES, INC. December 18, 2006	Mr. Rodney K. Haraga Director of Transportation State of Hawai'i Dependentment of Transportation 869 Puncthowi Street Honolulu, Hawai'i 96813-5097	SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) Dear Mr. Haraga:	Thank you for your letter dated August 22, 2006 (your reference number STP 8.2247). The following arc our responses to your comments. 1. It is acknowledged that the proposed development of the UHWO campus will have a	significant impact on the roadway facilities under the jurisdiction of the State DOT. It is unclear from DOT's letter if DOT views the impact as positive or negative. We would concur that a University facility will generate traffic, gradually, as it is ultimately developed, but we would like to think the sting of UFWOI in East Kapoloi will have a positive impact on roadware facilities teleph-aude Theortuolly mone of these living is <i>Control</i> and Lowers	O'ahu who would have commuted to the UFWO at Leeward Community College and UF Manoa, would have commuted to the UFWO at Leeward Community College and UF Manoa, would be instead attending classes at UHWO (at the proposed site). In addition, the UHWO project will be significantly different from University of Hawai'i at Manoa or the other colleges in the University of Hawai'i system as it is planned to provide a greater percentage of its student population housing for faculty and staff, so vehicles trips commuting to and from the campus will be reduced.	2. This is to confirm that it is UHWO's intention to allow private entities to develop improvements on UH property. The last paragraph of Section 2.4.3 <i>Private Development Lands</i> of the Draft EIS will be revised as follows: <i>Roads C, D, E, F, and G. Roads within the Private Development Lands total 24 acres.</i> <i>Roads C, D, E, F, and G. Roads within the Private Development Lands total 24 acres.</i> <i>Highway (Road F). Road S with connect the Private Development Lands total UH West O'ahu Lands. Road E will provide access to the campus from North-South West O'ahu Lands. Road E will provide another access to the campus from North-South West O'ahu Lands. Road E will provide another access to the campus from North-South West O'ahu Lands. Road E will provide another access to the campus from North-South West O'ahu Lands. Road E will provide another access to the campus from North-South West O'ahu Lands. Road E will provide another access to the campus from North-South West O'ahu Lands. Road E will provide another access to the campus from North-South West O'ahu Lands. Road E will provide another access to the campus from North-South West O'ahu Lands. Road E will provide another access to the campus from North-South West O'ahu Lands. Road E will provide access to the campus from North-South West O'ahu Lands. Road E will provide another access to the campus from North-South West O'ahu Lands. Road E will provide access to the campus from North-South West O'ahu Lands. Road E will provide access to the campus from North-South West O'ahu Lands. Road E will provide access to the campus from North-South West O'ahu Lands. Road E will provide access to the campus from North-South West O'ahu Lands. Road E will provide access to the campus from North-South West O'ahu Lands. Road E will provide access to the campus from North-South West O'ahu Lands. Road E will provide access to the campus from North-South West O'ahu Lands. Road E will provide access to the campus from North-South West O'ahu Lands. Road E will provide access to the </i>	Road. Road F will connect Farington Highway and North-South Road through the Private Development Lands. Road G will provide access to the Private Development Lands from the DHHL Lesideniate tubdivision to the south. These roadways are further described in Section 4.10.1. Pert the State Department of Transportation's comments during the public review period, the developer(s) of the Private Development Lands will be informed that they will need to coordinate with DOT's flighways Division regarding the responsibility for the provision of any required roadway improvements to mitigate impacts on the State roadways in the area.	
PH Kens	W. FRANK, BRANDT, FASLA Chorrana THOMAS, WITTEN, ASLA President R. STAN DUNCAN, AM.A FACUITE Un-President RUSALL, CHYNG, TASLA RUSALL, CHYNG, TANJA Excentive Une-breadent	VINCES-USHIGIRUNI Vice President GRANTTAURAKAMI.AICP Principal	TIMI SCHWILL AICP Scuine Associate RAYMOND T. IRGA, ASLA Seane Associate	KIYUN K. NISHIKAWA, ASI A Asoverate Kami mikaani yu ya da da Ap Asoverate	SCOTT ALIKA ABRIGO Avariah: Acottan:Araali, ASLA Asonian:	HONDJULJJOFFICE HONDJULJJOFFICE (101 Bickory Sterre Ga ASB Theory Sterre Ga ASB Theory Sterre Ga ASB 21 5641 Tot (1988) 22 5641 Tot (1984) 22 5641	HILO OFFICE III August Scene: Sur- Biol Laport Ocenes: Sur- Biol Laport Ocenes: Sur- Biol Laport, Sur- Biol Laport, Sure J Andria Hasari ways (LID Cardiah Hasari ways (LID	
RODNEY X, HARAGA DRECTOR DEMU DREGGA FRANCS MULL REENO AMARY FUKUNAGA BREAMAL NEDROKA BREAMAL NEDROLOH	RECEIVITY MARCHARER TO: 6 AUG 23 P2:25 Uduity U.M.:k.		HWO) snt (DEIS)	project.	l of approximately 500 acres, f, will have a significant State DOT. provements on UH property onsibility for the provision n the State roadways in the	affic impact analysis report th approval. The applicant clinate the use of the trence increased aircraft arational.	for our review.	
	STATE OF HAWAII DEPARTMENT OF TRANSPORTATION 889 PUNCHBOWL STREET HONOLULU, HAWAII 96813-6097 0 August 22, 2006 (iFc son, Director	ar cuanty cuttor reef, Suite 702 13	Subject: University of Hawaii West Oahu (U Draft Environmental Impact Stateme TMK: 9-1-016: 120, 127, and 129	nsmittal requesting our review of the subject blows:	evelopment of the UHWO campus comprised ort 7,600 students and 1,040 faculty and staff adway facilities under the jurisidiction of the anding to allow private entities to develop im- ordinate with our Highways Division the resp roadway improvements to mitigate impacts o	taff indicated that assumptions used in the tra- treet and need to be revised for our review an tact our Highways Planning Section and coor imptions for their review. If indicated that the subject project will exper- sloa Airport becomes fully developed and ope-	Amments until the revised TIAR is submitted	
INDA LINGLE GOVERNOR	Ms. Genevieve Salmor Office of Fravieweeve	235 South Beretania Sl Honolulu, Hawaii 968 Dear Ms. Salmonson:		Thank you for your tra Our comments are as ft	 The proposed d which will supp impact on the rc if UHWO is intr they need to coc thay required. 	 Our Highways s Our Highways s (TIAR) are inco will need to con appropriate assu 4. Our Airports sta activity as Kalaa 	We will defar further or Very truly yours, all, Haztadd, RODNRYK, HAZAGA	

PLANNING + LANDSCAPE ARCHIFTCTURE + ENVIRONMENTAL STUDIES + ENTITIEMENTS + PSRMITTING + GRAPHIC DESIGN

	PHONE (808) 594-1888		FAX (808) 594-1885
Mr. Rodnoy K. Haraga SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) ENVIRONMENTAL IMPACT STATEMENT PREPARATION NOTICE (EISPN) December 18, 2006 Page 2	ž	STATE OF HAVAY' STATE OF HAVAY' STATE OF HAVAAY' I KOPTOLUU, HAVALI 9615	
Per your comments, the traffic engineering consultant who prepared the Traffic Impact Analysis Report (TIAR), Parsons Brinckerhoff, is contacting DOT's Highways Planning Section to clarify the assumptions used in the TIAR.	July 31, 2006	RECEIVED	HRD06/2523
4. We thank you for the information provided that the project will experience increased aircraft activity as Kalaeloa Airport becomes fully developed and operational. This is consistent with a letter we received from the DOT Airports Administrator on January 17, 2003 (DOT's reference number AIR-P 03.0014).	Vincent Shigekuni PBR Hawaii ASB Tower, Suite 650 1001 Bishop Street Honohun FT 0523	AUS 07 2006 PBR HAWAII	
Thank you again for your participation in the Environmental Impact Statement process for this project. Your letter will be included in the Final EIS under Section 12.0.	RE: Draft Environmental Assessme	ent for the Proposed University of Ha	wai'i West O'ahu
If you have any questions regarding this project, please do not hesitate to contact me at 521-5631. Sinceredy	Campus, 'Ewa, O'ahu, TMIK 9-1-016:) Dear Mr. Shigekuni,	120 and 129.	
	The Office of Hawaiian Affairs (OHA) is following comments:	in receipt of your July 3, 2006 submissio	n and offers the
Vuren R. Hunghen- Vincent R. Shigekuni Visco Dosistori	Our staff is concerned that the archaeolog suffice for a project of this scale and cons following:	gical field check and literature review (Spe equence. A brief search of OHA's record	ars, 1996) will not s generated the
ece: Office of Environmental Quality Control Gene Awakuni/UH West O'ahu	The 'Ewa plain has historically been kno remains have been encountered. These si heavily cultivated for agricultural uses.	wn to contain sinkholes in which human inkholes can continue to exist in areas th	skeletal and avi-faunal at have been graded or
O:UOB13/1322.05/ElS/Final ElS/Response to DEIS CommentsBL-28 DOT Response 11-20 doc	According to records at the Bishop Muse Native American Graves Protection and I in general have been documented in the pr	um pertaining to inventories conducted f Repatriation Act of 1990, burial sites in F ast including:	or compliance with the anouliuli and in 'Ewa
	In 1938, human remains represen collected by Kenneth P. Envory Museum Auseum documentation one mile from the coast;	ting six individuals from Honouliuli, 'Ew , and William A. Lessa and acquired , indicates these remains were in a shalla	a, O'ahu were by the Bishop w crypt burial
	In 1933, human remains represe were collected by J.W. Barrington	nting three individuals from stone pits a 1 and Edwin H. Bryan;	t 'Ewa, O'ahu
	In 1942, human remains represen were donated to the Bishop Muser	ting two individuals from Kualakai, 'Ewa um;	Beach, O'ahu
	In 1959, human remains represen to the Bishop Museum by the A Standard Oil Refinery land);	ting seven individuals from 'Ewa, O'ahu Inthropology Club of the University of	were donated Hawaii (from
	In 1980, human remains repres collected and donated to the Bis information indicates these human	enting nine individuals from Honouliul shop Museum by Albert, Borthwick and 1 remains were recovered from coral sink	, Oʻahu were I Folk. Donor holes
	•		

.

PBR HAWAII & ASSOCIATES, INC.	December 18, 2006	W FRANK FRANDE FASLA Chaittean THUMASS, WITTESA ASLA ME. Clyde W. Nämu'o, Administrator President Official of the analijan Afficial	Restrative States Allowed Allowed Allowed Allowed Supervises Allowed Allowed Supervises Allowed Allowed Supervises Allowed Allowed Allowed Supervises Allowed	VINCENTRHIGKUN SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) VILO Predicial DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) GRANTLAUVANA.AILOP Dear Mr. Nämu'o: Proprint UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU)	Semin Associate A strain Associate RANNOND THEA.ASLA RANNOND THEA.ASLA Semin Associate Semin Associate We acknowledge that the project area in guestion, particularly the historic properties,	Asserved. In response to concerns raised hyperational and SHPD, UH West O'ahu has contracted an archaeological consultant to KNIMMANI VUEX.IHD'AR Conduct an archaeological inventory study (AIS) that will comply with Chapter 6E, Asserved. Hawai'i Revised Statutes. As recommended, a substantial subsurface inventory will be scorr Alles included in the AIS.	Acourt NURANANIANIA The text in Section 4.1 Archaeological and Historic Resources of the EIS will be Acourted to the teplaced with the following:	Existing Conditions	During the DEIS Public Review Period, the Office of Hawaiian Affairs (OHA) wrote:	BORDAULUOFFICE <u>A brief search of OHA's records generated the following:</u> ANT module the <u>ANT of the Stan plain has historically have brown to convert eightedes in ultich human</u> Impolabilitatives: 11si <u>The 'Ewa nhin has historically have brown to converte eightedes in ultich human</u>	tet and 223 561 Section 2010 Section and the section of the section statement and the section and the section and the section of the section	MLO OFFICE MLO OFFICE MLO ACCORDING 10. FECORIS 10. FECORIS at the Bishop Museum pertaining to inventories conducted for the Mean WAR (2011) MLO MEAN (2011) MLO MEAN (2011) MLO MEAN (2011) MLO MEAN (2011) MLO MEAN (2011) MLO MLO MARKED (2011) MLO MLO MUSEUM (2011) MLO MLO MLO MLO MLO MLO MLO MLO MLO MLO	past including: past including: WAUUNOFFICE In 1938, human remains representing six individuals from Honouliult. 'Ewa, O'ahu Development in the second structure of the second st	In 1933, human remains representing three individuals from stone pits at 'Ewa, O'ahu were collected by J.W. Barrington and Edwin H. Bryan;
Viterane Outscheret	vincent singekuni July 31, 2006 Page 2	In the last decade, unmarked burial sites have been found in the area of St. Francis West, West Loch Estates, Old Fort Weaver Road, Kalaeloa, One ula Beach, Campbell Estate, Ko'Olina and other areas in the vicinity of this project.	The depth of grading activities and the likelihood of adversely impacting any sub-surface cultural sites or deposits is contingent upon understanding the original surface grade as it may have existed prior to agricultural activities such as sugarcane.	Native Hawaiian burial sites have been found just on and under the surface to depths of eight or nine fect depending upon the nature of the terrain. Furthermore, the nature of documented interments in the 'Ewa area (stone pils, sinkholes, crypts, etc.) could lead to the survival of these sites despite intensive agricultural activities on the surface.	Our staff feels that the project area in question, particularly the historic properties, descroes a higher level of scrutiny than has been afforded. The letter report regarding the archaeological work (Spears, 2006) does not contain a scope of work, methodology, background research or even a bibliography. It is with this in mind that OHA is recommending that projects area undergo an Archaeological Inventory	Survey (AIS), as a new study or supplement to the previous, in which all aspects of applicable 6E administrative rules and revised statutes are satisfied. It is our recommendation that the AIS include a substantial subsurface inventory as much of the surface architecture may have been destroyed during previous commercial agriculture activities. Our staff will be happy to review the report as a supplement to the Daft Environmental Assessment (DEA) when completed. Until the aforementioned literature is	available, our stalf cannot concur with a finding of no significant impact for the DEA. Thank you for the opportunity to comment. If you have further questions or concerns, please contact Jesse Yorck, Native Rights Policy Advocate, at (808) 594-0239 or <u>jessey@oha.org</u> .		Alota,	Clyde M. Nämu'o Administrator	CC: Genevieve Salmonson Office of Environmental Quality Control	235 South Beeetunga Street, Suite 702 Honolulu, HI 96813 Melanie Chinen	Department of Land and Natural Resources State Historic Preservation Division 601 Karnokila Blvd., Roorn 555 Kapolei, HI 96797	

.

Mr. Clyde W. Nāmu'o SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Page 2

<u>In 1942, human remains representing two individuals from Kualakai, 'Ewa Beach, O'ahu were donated to the Bishop Museum:</u>

In 1959, human remains representing seven individuals from "Eva., O'ahu were donated to the Bishop Museum by the Anthropology Club of the University of Hawai'i (from Standard Oil Refinery land); In 1980, human remains representing nine individuals from Honoutiufi, O'aim were collected and donated to the Bishop Museum by Albert, Borthwick and Folk. Donor information indicates these human remains were recovered from coral sinkholes. In the last decade, unmarked buriel sites have been found in the area of St. Francis West, West Loch Estates. Old Fort Weaver Road. Kalaeloa, One'ula Beach. Compbell Estate, Ko'Olina and other areas in the vicinity of this project. The depth of grading activities and the likelihood of adversely impacting any sub-surface cultural sites or deposits is contingent upon understanding the original surface grade as it may have existed prior to agricultural activities such as sugarcame.

Native Hawaiian burial sites have been found inst on and under the surface to depths of eight-or nine feet depending upon the nature of the terrain. Furthermore, the nature of documented interments in the Twa area (stone pils, sinkholes. crypts, etc.) could lead to the survival of these sites despite intensive agricultural activities on the surface... OHA is recommending that the project area undergo an Archaeological Inventory Survey (MS), as a new study or supplement to the previous, in which all aspects of applicable 6E administrative rules and revised statutes are sustified. It is our recommendation that the AlS include substantion insustantory as much of the surface architecture may have been destroyed during previous commercial agriculture activities.

Also during the DEIS public review period, the State Historic Preservation Division wrote:

"...portions of Honoulinti aduptuan have been demonstrated, by various reports on file at the State Historic Preservation. Division (SHPD), to contain historically-rignificant resources, traditional and historic, including human remains/buriols. Also, we believe that the area of potential effect (APE) associated with the current underlook has a moderate to high potential of containing historically-significant remnants of sugar plantation infrastructure (e.s., irrigation systems)." For these reasons, an archaeological inventory survey is warranted prior to the issuance of permits for the proposed undertaking.

Anticipated Impacts and Mitigation Measures

As recommended by OHA and SHPD, an archaeologist has been contracted to conduct an archaeological inventory survey. This AIS will be completed prior to the issuance of grading and building permits for the proposed project.

Mr. Clyde W. Nämu'o SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Page 3

Thank you again for your participation in the Environmental Impact Statement process for this project. Your comments will be included in the Final EIS.

If you have any questions regarding this project, please do not hesitate to contact me at \$211-5631.

Sincerely,

PBR HAWAII

Mier & R. Smigh

Vincent R. Shigekuni Vice President cc: Office of Environmental Quality Control Gene Awakuni/UH West O'ahu 0:VOB13V1322.05/EISVFinal EISVFinal Responses to DEIS Comments/BL-11 OHA response 11-20/doc

PBR HAWAII & ASSOCIATES, INC.	December 18, 2006	UDITASIA TTENASIA TEL ASIA TEL COLLO	A Rise Senate A State of Hawai'i President State of Hawai'i	rysc. Mala Diate Capitol Prustern Honolulu, Hawai'i 96813	BUUNI SUBJECT: UNIVERSITY OF HAWAF'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft BIS)	Dear Senator Espero:	Thank you for your letter dated August 31, 2006. We have reviewed your letter and offer to have the following responses to your comments:	kww. AshA 1. Thank you for your support of the development of UH West O'ahu.	 We strongly concur that the UH West O'ahu campus is the missing link to the 'EwarKapolei region which is currently lacking a quality four-year higher seaco education institution. 	MALASLA This letter will be included in the Final EIS under Section 12.0.	Thank you again for your participation in the Environmental Impact Statement process. If you have any questions regarding this project, please do not hesitate to contact me at 521-5631.	Sincerely,		en Nurez R. Hugh	Vincent R. Shigekuni Vice President	cc: Office of Environmental Quality Control Gene Awakuni/UH West O'ahu	CE 0.VOB1311322.05VEISVFinal EISResponse to DEIS CommentsVE-nator Esperio response 11-20.doc Sate-4 60311171
The Senate	The Twenty-Third Legislature	STATE CAPITOL STATE CAPITOL HONOLULU, HAWAII 96913	August 31, 2006 August 31, 2006 August 31, 2006	Vincent Shigekuni Eaouño Tac	VINCINTISHIE ASB Tower, Suite 650 1001 Bishop Street GAMATTANIH	Towischer Toward Towa Toward Toward	Scatter Association Scatter Association Statement (Draft EdS) Kateward Statement (Draft EdS) Scatter Association Scatter Association Scatter Association Scatter Association Scatter Scat	I am writing in support of a new UHWO campus for the following reasons:	 The existing UH Martoa campus is nearing its capacity. It would accelerate and support the growth of the area as Oahu's New City, scort Aluka, as well as foster a college town atmosphere. The accention accelerate on atmosphere. 	J. Lue campus is a priority project of the Uri Student Caucus Which represents the online student propulation of the university system	4. A campus in <i>Ewark</i> apoler would reduce tratic heading to the main Manoa campus and community colleges in downtown Honolulu, as well as ease demand for more parking and housing in Manoa. 5. The university will be a major employer for West Oahu residents, as well as	provide construction jobs and income during the building phase. 6. A UHWO campus could offer degree programs in high-demand	professions, such as teachers, nurses, and para-professionals. 7. As a legislator from the area, not one constituent has voiced his or her from the area, not one constituent has voiced his or her from the area.	opposition to the project. Interest campus is the missing link to this region currently lacking a E-ant-readour	quality 4 year higher education institution. But approvention of the art of t	International way and a second s	Will Espero

KGEERT BURGA Present Ware Freesen Ware Freesen Ware Freesen Ware Freesen Ware Freesen Koonte Freesen Hand States H

PLANNING + LANDIGAPE ARCHITECTURE + ENTIRONMENTAL STUDIES + ENTILEMENTS - PERMITTING + GRAPHIC DESIGN

FIRST DASTRECT LOBRACET ACORRECT LOBRACET RECORD DISTRICT RECORD DISTRICT RECORD DISTRICT THEND DISTRICT THEND DISTRICT THEND DISTRICT RECORD DISTRICT THEND DISTRICT RECORD DISTRICT

|--|

HOUSE OF REPRESENTATIVES STATE OF HAWAII STATE CAPITOL HONOLULU, HAWAII 96813

омогиги, намаії эєвіз July 26, 2006

> PBR HAWAII ASB Tower, Suite 650 1001 Bishop Street Honolulu, Hawaii 96813

Dear PBR Hawaii,

RE: UNIVERSITY OF HAWAII WEST OAHU DRAFT ENVIRONMENTAL IMPACT STATEMENT

I have two observations for comment regarding this project:

1. USE OF AREA CANE HAUL ROADS

Suggestion: It is imperative that existing east-west connector roads (privately owned cane roads) that could potentially link the North-South Road to both Ft. Barrette Road and Ft. Weaver Road be constructed as soon as possible before the University of Hawaii West Oahu housing is made available; as well as cane roads that run in a north-south direction that could link Farrington Highway with Renton Road, Roosevelt Boulevard, and Old Ft. Weaver Road.

Concern: A private developer now has ownership over the majority of these connector roads and there is no obligation, commitment, timeline, or otherwise, being provided by the developer as to when these roads would be improved to City and County of Honolulu standards and available for public usage.

Conclusion: The DEIS fails to make reference as to when additional feeder/or arterial roads will be constructed that could serve the campus (excluding the North-South Road). Reliance upon just two roads- Farrington Highway and the North-South Road to serve the campus is ill advised and will prove to be insufficient. The east-west connector road (Mango Tree Road) should be constructed before the campus is open for learning rather than years afterwards.

Page 2

:

2. HOURS OF OPERATION

Suggestion: Institute hours for classroom instruction to commence after 9:00 a.m.

Concern: The residential matrix of housing units currently in existence from Ewa, Ewa Beach, Makakilo, through Kapolei proper, have exceeded the carrying capacity upon the very roads that are to serve these residential areas.

Conclusion: The State of Hawaii has granted the University of Hawaii the authority to schedule its hours of operations as the University sees fit. The DEIS fails to take into consideration the plausibility of staggering University operations so that during the morning commute rush hours, travel to the campus is minimized.

Many people over the years have opined that if a university were built on the Leeward side, it would resolve our traffic woes. This concept is flawed. A university in Kapolei will undoubtedly lessen traffic flow from the H-1 freeway heading east while yet navigating within Kapolei itself will implode with the additional jobs, housing, and activities a university campus brings.

Therefore, every right-of-way and potential service route to and from the university itself that can be built must be incorporated into the plan now. I fully support UHWO being located in Kapolei. However, I have deduced that the DEIS underestimates traffic patterns that most likely will be emanating from the addition of a four-year institution of higher learning upon a landscape already lacking a network of roads. The addition of the North-South Road alone to serve the campus will not suffice.

tida Cabanilla pectfullv

CC: Office of Environmental Quality Control Enc: Copy of Letter to Mayor Hannemann Dated July 26, 2006 لبًا المحافظة المحا محافظة المحافظة المحاف المحافظة ال محافظة المحافظة المحاف محافظة المحافظة الحاف محافظة



HOUSE OF REPRESENTATIVES STATE OF HAWAII STATE CAFITOL HONOLULU, HAWAII 96833

July 26, 2006

Honorable Mayor Mufi Hannemann 530 South King Street Honolulu, Hawaii 96813

Dear Mayor Hannemann,

RE: UNIVERSITY OF HAWAII WEST OAHU DRAFT ENVIRONMENTAL IMPACT STATEMENT

EAH AND ST. FRANCIS DEVELOPMENT IMPACT ON RENTON ROAD EWA VILLAGES A constituent concern is after EAH and St. Francis have completed their housing projects and Renton Road is opened to Kapolei Parkway, the aftermath will produce a level of traffic through Renton Road of which Renton Road is not properly designed and definitely not equipped to handle. I have a solution. The Ewa Development Plan references an east-west connector road (also known as Bill Balfour Drive and/or Mango Tree Road) that is to become a City thoroughfare one-day. It makes perfect sense to connect the east west connector road to Park Row in order to access Renton Road. With the addition of a north-south aligned road through Ewa Villages Municipal Golf Course, increased traffic congestion on Renton Road can be minimized. The good news is that you own the golf course. It can be easily reconfigured to permit a road across it just like North Road does over the New Ewa Beach Golf Course and Geiger Road over Coral Creek Golf Course. The costs for the new road I am proposing are estimated to be under \$5 million. It is less than an eighth of a mile in distance from Park Row through the golf course to the east-west connector road. In return for such a short road, we get major traffic relief and at minimal costs to boot.



The illustration above depicts how the extension of Park Row would connect to the east-west connector road. Proceeding north from the east-west connector road is another cane haul road that would take pedestrians and motorists alike all the way to Farrington Highway. This connection should not take place after all the development that is planned for has been completed. Rather, it should be done now in preparation and prior to the massive influx of development planned for this area. Having a college carrpus in one's backyard means accommodating it with an ample and adequate supply of roads to navigate through and around it. As you know, the State is building the North-South Road to serve the campus. The City should be constructing additional tributaries such as in my proposal to augment that road.

	PB And Association	R HAWAII ociates, inc.
	τ	December 18, 2006
to assist you in getting the east-west ind connected with the Park Row extension I tain that the area infrastructure is completed anticipated growth- and not afterwards by ow.	W FRANK BRASIDT EASLA Channum THOMASS WITTEX ASLA Preddon R STAN DYSCAN ASLA FACUNG UTE-Preddon	Ms. Rida Cabanilla Statc Representative State Capitol, Room 303 Honolulu, Hawai 'i 96813
ppreciated. I can be reached at 586-6080.	RUSSFILLY, CHUNG RALA Exercitiv Vice-Prevident VISCINT BALGERUSI Vice Date Bandered	SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) Dear Ms. Cabanille:
ida Cabanilla	GRANTTANUARAMALAICP Prondpad TOMASCHNTLAUCP Science Associate KAYNOND THIGAASLA Science Associate KERYIN KANSHIKAWA ANIA Associate KIMAMIKANI VI TSA ITIP'AP Associate SCOTTANARAMI, ASLA Associate SCOTTANARAMI, ASLA Associate	Thank you for your letter dated July 26, 2006. We have reviewed your letter and offer the following responses to your comments: following responses to your comments: We acknowledge your concern that the East-West Connector Road (EWCR) will undoubtedly benefit the 'EwwKapolei region. Since this road has been planned since the original <i>Ewa Development Plan</i>, it should have received a higher priority for CIP funding in recent years. This especially true since many of your constituents need another alternative to Geiger Road/Roosevelt Avenue/Fort Barrette Road/Kapolei Interchange/H- I Freeway. The description of Road "F" in Section 4.10.1 <i>Roadways and Traffic</i>. Anticipated Impacts and Mitigation Measures, Roadways of the EIS will be revised as follows: Road F will <i>service the regional roadway network, providing access within the approximately 500-acre property as well as to the surrouting</i>
nental Impact Statement Comment/Letter	HONOLULU OFFICE HONOLULU OFFICE AGI Taking Veren AGI Taketi Veren AGI Taketi Veren AGI Taketi Veren AGI Taketi Veren AGI Taketi Agita Taketi Agita Honol Agita Hon	 EworKapolei region. Road F will have a right-of-woy width of 104 feet, with four travel lanes (no in each direction), a 16-foot-wide median, like lanes, 10-foot-wide planting strips, and 6-foot-wide sidewalks. This road is scen as the western end of the nhanned "East-West Connector" (connecting. Farrington Highwey and Fort Weaver Road). Construction of the EWCR and a "north-south connector road" will require land acquisition from various parties including the City and Conny of Honolbult (Ewa Villages Golf Course, DHHL, D.R. Horton and possibly others), an Environmental Assessment on Farvionmental Assessment of the new varies including the City and Conny of Honolbult (Ewa Villages Golf Course, DHHL, D.R. Horton and possibly others), an Environmental Assessment on Farvionmental Impact Statement, and the design of the road itself, and we would be very supportive of your efforts to secure CIP funding for these roads. You may be interested to know that we will be requesting that the EWCR be included in an updated list of projects to be funded under Ewa Highway Master Plan (Section 33A-1.6(bt)) of the Revised Ordinances of Honolulu, when it undergoes its scheduled five-year review in 2007. You are correct that we failed to make reference to other regional roadway improvements have undergoes the compuse (besides North-South Road). Also planned are the channel of K-moles of Panchasis. North-South Road). Also planned are the channel of K-moles North-South Road).
entative Rida Cabanilla 303 • Honolulu, Hawaii 96813 coo coo - m-sh. maria	u ao wui 124 2005 ang wu 1 Madhal (haad 2002 1121 1 et 1868 219 2558	computed on Kapoter Farkway and koad U. Ine tollowing will be added at the end of the first paragraph of Section 4.10.1 Roadways and Traffic, Anticipated Impacts and Mitigation Measures, Roadways of the EIS:

Please let me know what I can do to connector road open for public use and am proposing. We need to make certai ahead of time in preparation for the anti playing catch up like we are doing now.

•

Page 3

.

Your favorable consideration is ap Thank you.

ഷ

Attachment: UHWO Draft Environm

د

V State Representative Rida Cabanilla State Capitol, Room 303 • Honolulu, Hawaii 96813 Telephone: (808) 586-6080 • Fax: (808) 586-6081 • Email: <u>representilla@capitol.hawaii.gov</u>

FFANNING + LANDSCIPF ARCHITICTURE + ENTRONMENTAL STUDIES + ENTITLEMENTS - FLEMITTING + GRAPHIC DISIGN

BR HAWAII NSSOCIATES, INC.	December 18, 2006		Ms. Rída Cabanilla State Representative	State Capitol, Room 303 Honolulu, Hawai'i 96813	SUBJECT: YOUR LETTER TO MAYOR HANNEMANN DATED JULY 26, 2006	Dear Ms. Cabanilla:	We were asked to comment on your letter addressed to Mayor Hannemann dated July 26, 2006, which was attached to your comments on the University of Hawai'i Wess O'ahu (UHWO) Draft Environmental Impact Statement (dated July 26, 2006, which we	will be replying to in a separate letter). We have reviewed your letter to the Mayor and offer the following responses to your comments:	 Since your first five paragraphs were directed toward the Mayor, we will no comment on the possible addition of a north-south aligned road through the Para 	Villages Municipal Golf Course as it is City property.	2. In regards to the diagram on page 2 of your letter, please be assured that the opening of UHWO will occur (2009) after the planned completion of the North-South Road (2008), and the initial phase of the campus will be accessed via	Farrington Highway. In addition, the proposed campus roadway system incorporates the western end of the proposed East-West Connector Road (EWCR)	3. The University is actively addressing regional traffic-related concerns by coordinating with various landowners in the East Kapolei region. Construction of the PAVEP (2 new ener word body with according to a provide the second body and the second body and the second body according to a provide the second body according to a p	the burlet of the weat-west load that will extend to the manuaged angleway though the UHWO site and beyond, eventually connecting with Fort Weaver Road and a "horth-counth connector road" will require land architection from	various parties including the City and County of Honolulu (Ewa Villages Golf Course, DHHI, D.R. Horton and possibly others) an Provincemental Accessment	or Environmental Impact Statement, and the design and construction of the road itself. We would be very supportive of your efforts to secure CIP funding for	these roads. In addition, the University will be requesting that the EWCR be included in an updated list of projects to be funded under the Ewa Highway Master Plan (Section 33A-1.6(h) of the Revised Ordinances of Honolulu, when it	undergoes its scheduled five-year review in 2007.
P ()		W. FRANK DRAND'L FASLA Chaitnin	THOMASS, WITTEN, ASLA President	R. STAN DUNCAN, ASI A Executive Vice-President	RUSSIAL Y, J. CHUNG, FASIA Executive Vice-Dreadent	VINCINT SHIGEKUNI Vec President GRANT TAURAKANI, AICI	renapai TAM SCHNEL, AICP Kenine Associate	RAVMOND T. HIGA, ASLA Semer-Associate KEVIN K. NISHIKAWA, ASLA	ADVERTIGE KIMI MIKAMI YUTN, LEED'A	Associatie Scott Alika Abrigo	Aswelate SCOTT MCRARAMLASLA Azeretate			ROMANDULU DEFELI 1012 BUSING VIECT ASIF Towers, Suite (20 Hemolofu, Flawari' 968) 5–1383	Feb (808) 521 - 50.31 Fax: (808) 723 - 1402 E-araft: synadionne plodarwadan	HILO OFFICE IN Aupuni Sice 4	1411, ASSAN WC 2014, 2012 1110, ASSAN WC 2014, 2012 1414 (858) 964 - 1713 1414 (858) 964 - 1939	WAILUKU OFFICE 1757 Will LLARD, Store 1 Valled LlARD, Store 1 Ref (8887212, 2538
Ms. Rida Cabanilla SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Page 2		The roadway network in the vicinity of the UH West O'alu property is expected to change significantly in the future. Phase I of North-South Road and its interchange with the H-1 Freeway will be constructed by late 2008. Consistent with the ORTP, the H-1	Freeway and Farrington Highway will be widened. The DOT and DTS have been consulted in the timing of these projects. <u>It is assumed that the portion of Kapalei</u>	Parkway between Kapolei Middle School and the North-South Road will be timed to be completed by the Department of Havaiin Home Lands with the completion of the first phone of North-Couth Good Science, it command that the connected of Science Science and	<u>parter of vertity open spear, without a toxetter that the segment of Aufwert Farkway</u> through 'Eva Villages will be completed which will allow 'Eva residents an alternate (to Geiger RoadRoosevelt Avenue) route to Fort Barrette Road.	The description of Road "G" in Section 4.10.1 Roadways and Traffic, Anticipated Impacts and Mitigation Measures, Roadways of the EIS will be revised as follows:	Road G will provide access to the property from the DHHL residential subdivision to the south. Road G will be a collector roadway with parking lanes. <u>This road will provide an</u> alternative north-south route connecting the campus with Kapolei Parkway.	2. Your recommendation to start is instruction no earlier than 9.00 am each work week is creative and UHWO will study its merit and impact of academic scheduling and operations. Section		<u>Traffic Domand Management</u>	During the DEIS public review period. State Representative Rida Cabanilla suggested that classes do not commence until after 9:00. The University of Hawai'i will study this suggested change in operation and assess the impact it might have on its operations.	Thank you again for your participation in the Environmental Impact Statement process. If you have any questions regarding this project, please do not hesitate to contact me at (808) 521-5631.	Sincerely,	V - A + Q - B	Vureex K. Ohrefu-	Vincent R. Shigekuni Vice President	cc: Office of Environmental Quality Control Gene Awakuni/UH West O'ahu	ONOB1311322.05/E[S\Final EJSResponse to DEIS Comments\BL-02 State Rep Cabanilla response 11-20.doc

PLANNING + LAVINGAPL ARCHITTICTURE + ENVIRONMENTAL STUDILS + ENTITLEMENTS - PERMITTING + GRAVIIIC DEGIS

	AUG222006 TUE 10:4	7 am uh-eavironnental ontr. 96	9563980 P.	01
Ms. Rida Cabanilla SUBJECT: YOUR LETTER TO MAYOR HANNEMANN DATED JULY 26, 2006 December 18, 2006 Page 2		FAX TRANSMITTAL SH	EET	,
Thank you again for your participation in the Environmental Impact Statement process. If you have any questions regarding this project, please do not hesitate to contact me at (808) 521-5631. Sincerely, PBR HAWAII	2	BNVIRONMENTAL CENTE University of Hawaii 500 Dole Street, Krauss Annex 19, Honol Telephone: (808) 956-7361 Fax: (80	38 hilu, HI 96822 28) 956-3980	
Vuezzo K. Otugue Vincent R. Shigekuni Vice President	Han A M	2006/16/18		F
cc: Office of Environmental Quality Control Gene Awakuni/UH West O'ahu Mayor Mufi Hannemann/City and County of Honolulu ov10B1311322.054ESVFinal EISVFinal E	FROM	Peter Rappa (W. Environmental Review Coordinator		
		Vincent Shigekuni, PBR Hawaii (5' Chancellor Gene Awakuni, UH We OEQC (586-4186)	28-1402) ss: Oahu (453-6078)	
	SUBJECT	Review Comments Draft Environmental Impact Statem UH West Oalm		
	No. of Pages: inclu	ding cover sheet: 7		ł

AIKG-22-2006 TUE 10:48 AK UH-ERVIRONMENTAL	CNTR, 9956	3980 P. 02	· AUG22-2006 TUE 10:48 AB UH-ENVIRONWENTAL CNTR. 99563980	P. 0
UNIVERSITY OF M Environmental Center	1.IV M V	AT MANOA	August 21, 2006 Page 2 of 6	
Angust 21, 2006 RE:0752			General Commonts Our reviewers feit that the DEIS was very professionally prepared and most	it the
Mr. Vincent Shigekuni Vice-President PBR Hawaii			important issues were well covered. There were, however, three general areas whic not adequately addressed, including justification for building the West Oatun of the construction of the new earnpus, and the treatment of cumulative impacts.	we felt were the funding
ASB Tower, Suite 650 1001 Bishop Street Honolulu, HI 96813	· · · · · · · · · · · · · · · · · · ·		Need for the West Oahu Campus. It is not clear from the text that there is develop a campus in West Oahu. The DEIS does not cite any projection indicating develop a campus for students seeking to attend the University of Hawaii. Indeed, there a seem construction construction of the Oakard set of the construction of the Con	t need to n increase in to be a lack of
Dear Mr. Shigetomi: Draft Environne University of	rtal Impact Statemor Hawaii West Oahu		The provides of the product of support of the productions into any number is contender to a contender of a second plan on the page of the DEIS, but there are no supporting statistic there is a deforment of a compute in West Oahu. There are approximately 830 student West Oahu. The proposed new campus will accommodate 7,600 when fully built, a fold increase in the student body. An analysis of where those students will come from the student body.	ates to a numb- showing that now attending almost ten- is warranted.
The University of Hawnii West Oahu (U acres of land in the Ewa district of Oahu. The p north, the future North-South Road to the east,	JHWO) property con property is bordered l a future Department	oprises approximately \$00 by Farrington Highway to the of Hawaiian Home Lands	Funding. Funding the development of URWO seems to be dependent on the and sale of private housing on the site. At the moment, final agreements for this priconstruction have not been concluded and, therefore, the financial viability of the puncertain.	construction ate ject seems
restortural suportysion to the south, and ure Ang selected a dovelopment partner to purchase appl approximately 500-acre property. The Universi proposed UFWO campus will be a four-year un opportunities. The campus will support 7,600 a provide much needed higher cducation opportut collere-aread bonulation of West Oahu.	ooler vool Course of roximately 287 acres ity will own the rema itiversity offering a p itudents (and 1,040 f nitics and employma	the west. The Duriversity of land within the totad array of educational aoulty and staff) and will nt opportunities for the growing	Cumulative Impacts. The other area we felt needed a more thorough analy cumulative impacts of $\underline{\mathbf{MI}}$ the proposed new development in the Kapolei area. We up there may be no legal requirement that UH or PBR prepare such at analysis, but su assessment of the cumulative impacts to which this project contributes is surely nee the following:	is is the derstand that an ad. We nofe
The Private Development Lands will ser income) from these finds will be used to constrain will include a 7,600-student campus, University commus extransion work force/affordath, bouch	rve as revenue-produ ucr the first phase of v Village, mixed-use ing an electrical subs	teing lands and the profits (sale compus. The UHWO lands purcels, student housing or sations site. and rondways. The	 Traffle. The DEIS indicates that if needed interchange are built and w Farrington occurs, the traffle from the project would rekult in service fer But what if all the traffle from proposed projects is included? The analy indicate what the ournulative impacts are. 	lening of 1 "D" (p. 66). is doesn't
Private Development Lands will include residen elementary school, a detention basin, and an ele development will provide approximately 4,041 units) and approximately 842,900 square feet of	ritai parcels, mixed- cetrical substation sit residential units (inc f commercial space	ise parcels, parks, sn 2. In total, the UFWO luding 761 student housing	 Water. The DBIS indicates that new water domand would result in the upgraded storage and distribution lines. However, as marry as we can t reference to water supply. Rather the DBIS refers to the "Water Use an Plan" and the "State Water Projects Plan" which, we assume, indicate w 	aed for 1, there is no Development ter sources and
This review was prepared with the assis Planning: Stephen Marlin, UFM Botany;	tance of Kem Lowny and Lynn Hodgson,	, UHM Urban and Regional UHWO Natural Science.	distribution. In our search through the agency letters at the end of the U not find the BWS statement on water supply.	IS, we could
1			 Wastewater. This project alone would, according to the DEIS, more the capacity of the Honouliuli treatment plant. Again, what is the total wast discharge associated with planned Kapolei build out—and to what exter expansions of Honouliuli adequate to meet the treatment needs? 	ı exhaust the water are planned
2600 Dolo Streel, Krause Annex 10, Honeluiu, Hewei'i 06822. Tolophono: (808) 950-7381 + Facsimile: (808) 966-3980	-2313 -			
An Equal Opportunity/Affirmative Action Institution				

P. 03

.

.

· AUG-22-2006 TUE 10:49 AN UN-ENVIRONMENTAL CNTR. 99563980 P. 05	August 21, 2006 . Page 4 of 6	landscaped sections. As sugar cane agriculture continues to disappear on the island, perinaps wo will see other displaced plant species make a come-back. They also suggest that in addition to removing the populations to the <i>Abutilon menzievii</i> to the three nuisery sites that an attempt is made to leave some plants integrated with other native plants around the comput.	Our reviewers also fait that native plant species should be bacd for landscaping where appropriate and that the xeriphytic plants be used. The Honolulu Roard of Watsr Supply has a Xeriscape garden in Halawa and might be a good source for finding appropriate plants.	Agricultural Impact (Section 4.3, p. 40-41)	In this section on Anticipated Impacts and Mitigation Meadures, the DEIS state that "cultivated lands within the property will be withdrawn from agricultural production for the proposed development." The DEIS further states that the lost of agricultural production will not conferently inverse meaning and account of the states that the lost of agricultural production will not	argumentary imputs agreement production statewards and inster is for indicest times easewards or farming. Flowever, the DBRS fails to mention the fate of the farmel's working the land. How many farmers are there and what are the impacts of the project on them? [Will they be compensated for their loss? Will they be reflocted to other suitshie lands? The final FIS should contain some	reference to the farmers and their ultimate fate. Noise (Section 4.4, p. 41-42)	In this section, there is a discussion of three locations labeled S1, L1, and L2. While the results of the testing are detailed for sites L1 and L2, there is no results for the testing are detailed for sites L1 and L2, there is no results for the test at this ait?	Construction Noise (5, 42)	In the section on the Anticipated Impacts and Mitigation Measures of construction noise, the DEIS states that the "Construction activities will comply with State DOEI noise regulations." How will anyone know if the contractors will on somely multes there is an on-site monitor or	at least forme provision for occessional monitoring? It is hard for usite believe to un the the construction commany will report a violation in the monitoring monitoring was and and the file construction	should be out three monitoring the construction site to make sure that contractors comply with applicable noise standards.	Air Quality (p.49)	In the section on the Long-term Impacts and Mitigation of bir quality impacts, the DEIS states that "the UrITWO will encourage public transit, as it is located near planned transit nodes."	How will UHWO encourage the use of public transify. Are they contemplating any programs such as free bus passes for faculty and staff to take the bus or any other programs? The use of public transit is a very important component of reducing congestion in the Bwa plain. Saying that it will be encourated in not ennuch. The University should be working on that and that any that	
99563980 P. 04		f solid wayte is forecast (p. 76). Waste offill can be expanded to accommodate e, but politically dubious. The but doesn't diminish the issue about where	nė specifi¢ issues:	· (21	Kapolei whter supply system is for potable Are there two 215-foot water supply	anacement, vour cuideline is to "Reduce	a should choisider the use of permeable able surfaces on parking lofs, walkways er runoff while increasing infiltration of	races is gaining popularity in construction dence by a well-attended Sea Grant		states that "The island of Oahu originated om two volcanoes, the Waianac and		encourage or utilize the Kaloi and ipus, zather than trying to make them vo landscavion may serve as refinita for a	a to an and the two the family standards for		id oncouraged at how many native plants nany years of sugar cane agriculture. They ans, especially <i>Abutiton menziesit</i> , in the and the schees of the campus. or even in the	
AUG-22-2006 TUE 10:48 AM UH-ENVIRONMENTAL CNTR.	Амgust 21, 2006 Раде 3 оf б	 Solild waste. Something like 40 tons a day of collection assumes that Waimanalo Gulch lan additional solid waste. That's technically true commitment to project recycling is laudable, additional wastes will go. 	In addition to our general comments, we note som	Water Supply and Distribution (Section 2.5.2.2, p. 16-1	In this section, the DEIS states that the 215-foot K water on page 16 and for non-potable water on page 17. J systems or is one of the statements in error?	Sustainability (Section 2.6, p. 17) In this section under the builtet noint on Water Ma	overall potable water consumption storm water." You surfaces as a guideline where possible. The use of perma- and courtyard can assist in the management of storm wate	rain water into the water table. The use of permeable surt on the mainland and interest in growing in Hawati as evid workshop on the topic held in Maui earlier this year.	Topogrephy (Section 3.2, p. 22)	In the first line in Existing Conditions, the DBIS s as a volcano flows." The island of Oahu originated fro Koolau volcances.	Drainage (pp.23-24 & 74-75)	We are pleased that the planners have decided to e Hunehune guiches on the property as a fasture of the carn disannear. Even if smail in area, those fastures and nearly	surprising number of plants and antimals.	Habitat Conservation Plan (p. 31-32)	Our botanist reviewers ware actually surprised and and animals were found in the surveys, considering the mu suggest first UHWO make places for displaced native plat planned botanical garden section of the new cammus. aroun	

P. 02	Angust 21, 2006 Page 6 of 6	energy dernande?) Is there any consideration of alternative energy sources? What about photovoltaic or wind derived energy? Since UFIWO is committed to sustainability, why a alternative sources of energy not even considered in the plan?	The DOB has raised issues about the availability and finding of new schools. The DOB has raised issues about the availability and finding of new schools. The notes that regulatory authorities are demanding that provide them. In one plact, the DEIS identifies I lands and other sites as potential achool sites. Later, the DEIS indicates that a site for a site and for a site and for a site and for a site and for a site as potential achool sites. Later, the DEIS indicates that a site for a site as potential achool sites. Later, the DEIS indicates that a site for a site and for a site and for a site and for a site as potential achool sites. Later, the DEIS indicates that a site for a site and for a site and for a site and for a site as potential achool sites. Later, the developer will provide sufficient of astrooms, but it appears some of the are unresolved.	Alternatives to the Proposed Action (p. 123) At the risk of repeating ourselves, it very difficult to disouss alternatives in the ab any data on the number of students anticipated for the UH system as a whole, where Wes students will come from, what the impacts is on the other schools in the UH system, etc. believe that this discussion must have gone on somewhere within the University and it sh presented in the DBIS.	ffic Thank you for the opportunity to review this DEIS Sincerely, Sincerely, Peter Rappa Friend Review Condinator	to co: OEQC Chancelor Gene Awakuni, UHWO fames Moncut Kem Lowry Mark Merlin Lynn Hodgson
13980 P.		art would be the UHM ag at traffic congestions for so	orporate energy conservation o mitigate impact." What ergy conservation design le in terms of energy savings ergy conservation design y saving designs into their new y saving designs into their new.	able bousing. Kow many sr than "a large portion" of the	feashtes of roadways and traff way in question. It was 19 and the discussion on pages y. It was probably more : pages 64 and 65 showing all	the City and County's LUO? tion Bragincors? How is the one parking stall for every two stall for approximately three some olarification in the final some olarification in the final trical facilities, the DBIS state also care of the damand for how will UHWO meet it
je 02:55 pm uh-environmental ontr. 9956	22, 2006 af 6	ut into place as the new campus opens. A good place to sh tent of Urban and Regional Planning which has been lookin	at Demund and Solid Waste Generation (p. 50) The DEIS states in this section the UFIWO will strive to if catures and promote conservation and recycling programs(will administrators at UFIWO do to encourage the use of ch of it there some type of report or manual on what is availably Will someone at UFIWO is assign the task of being the en- try UVII someone at UFIWO is assign the task of being the en- try UVII someone at UFIWO is assign the task of being the en- try UVII someone at UFIWO is assign the task of being the en- try UVII someone at UFIWO is a poor record of incorporating feature at UFIWO administrators should be weighing design feature ?	(p. 56-57) The last section on page 56 is unclear with regards to afford to units are required? If its 355 then won't all of them rafit able? ys and Traffic (p. 64-65)	The discussion on the Anticipated Ympacts and Mitigation A re been improved by providing a map of the proposed road to shuffle back and forth to Figure 2-1 bokween pages 8 and 5 on the madways proposed for contraction in file hard cop co do so in the DVD version. Futing a new figure betweel reys would be helpful in the final EIS. (p. 70)	That are the number of stalls that would be required under to number of stalls suggested by the institute of Transport ber of stalls arrived at since the DEMS states that a ratio of was utilized for the initial 1,520-student campus. This section needs of the ultimate 7,600-student campus. This section needs the section on Antiopated Impacts and Mitigation for effect of it planning to build a 110-megawatt generator that will UHWO. What happens if FIGCO cannot build the facility.



December 18, 2006

THOMASS, WITTEN, ASLA President F. FRANK DRANDT, DASLA Charman

R. STAN DUNCAN, ASLA Exernity Vice-President

RUSSELY, CHUNG FASLA Executive Vice-President INURAL SHIPPING SHARES GRANT'E MURAKAMI, AICP Principal

tue President

KAYMOND T. HIGA, ANI, A FOM SCRINELL, AICP Sentiar Associate Senne Associate

KEVIN K. NISHHKAWA, ASLA Issurate KIMI MIKANI VUUM LEEDTAP SCOTT ALJKA ABRIGO Associate

SCOTT MURAKAMI, ASLA

AUNTRAL

1001 flutuop Street ASIT Jower, Supte 6 Jo Ronadia, Jawafi 1964 1 Artis 1983 521-563 Jawa (1988) 521-51 (10) Famili Aysadiruog, Pichawar, con HONOLULU OFFICE

HILO OFFICE 101 Auparti Street 11th, Lagoon Correc Suite 139 11th, Likoardy 962-01 155 74-05819 061-0111 Ear (1898) 961-1982

20,000 2,270 1,633 Source: Windward North Share WAILUKU OFFICE 1787 Will Patropy Suite 4 Walidar, Hawai? 9779411271 Jek (808) 237.2578

Mr. Peter Rappa, Environmental Review Coordinator 2500 Dole Street, Krauss Annex 19 Honolulu, Hawai'i 96822 Environmental Center University of Hawai'i

UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) SUBJECT:

Dear Mr. Rappa:

Thank you for your letter dated August 21, 2006. We have reviewed your letter and offer the following responses to your comments: Need for the West O'ahu Campus. The following text and tables will be added before the last paragraph in Section 2.2 Need for the Project of the EIS: ÷

West O'ahu campus is expected to target students in the surrounding regions to alleviate the burden on the UH Manoa campus. The relative need for With the expected population increase in the 'Ewa region, the proposed UH postsecondary education based on projected changes in population from the year 2000-2020, are shown in Tables 2.1 and 2.2. Table 2.1 Relative Need for Postsecondary Education (Based on Projected Change in Population (#), 2000-2020

Relative Need for Postsecondary Ed & Training Based on Projected Change in Population (#), 2000–2020 University of Hewel's Second Decade



PIANNING + LANDSCAPE ARCHITTCTURE + ENVEROMENTAL STUDLES + ENTITEIMENTS / PERMITTING + GRAPHIC PESIGN

Mr. Peter Rappa SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Page 2

Table 2.2 Relative Need for Postsecondary Education (Based on Projected Change in University of Hawai'i Second Decade Population (%), 2000-2020

Based on Projected Change in Population (%), 2000–2020 Relative Need for Postsecondary Ed & Training



Funding. The University is currently entertaining a number of options available for the initial development of the campus, including the sale of land to a private developper. It is acknowledged that final agreements for the sale of the land have not been finalized (as of this writing). In addition to the monies expected from the land sale, UHWO will seek funding from the upcoming 2007 Legislative Session. Also, the University is seeking any from the upcoming 2007 Legislative Session. Also, the University is seeking any opportunity to share the costs of infrastructure development with surrounding landowners such as DHHL and D.R. Horton. Section 2.8 Order of Magnitude Cost of the FEIS will be revised as follows: c,i

The current estimated cost for Phase 1, including earthwork; landscaping; internal roadways; sever, potable water, non-potable water, and drainage facilities; and inprovements to Farrington Highway for access is approximately \$150 million.

financial burden on the University. Final agreements for private construction have not been finalized, however, the University and the private developer are taking the Due to limited State resources, a private developer was sought out to alleviate the necessary steps to ensure that the projected phasing of the project remain on schedule. Cumulative Impacts - Traffic. Please be assured that the cumulative impacts of surrounding projects were included in developing the scenario of future traffic without the project. As stated in Section 4.10.1 Roadways and Traffic, Anticipated Impacts and Mitigation Measures, Background Traffic, of the Draft EIS: с.

Mr. Peter Rappa SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DR/ ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Page 4	6. Cumulative Impacts - Solid Waste. We acknowledge your concern regarding additional solid waste that would be generated by the UH West O'ahu. The sec paragraph of Section 4.10.5 Solid Waste Facilities, Anticipated Impacts and Mitiga Measures of the EIS will be revised as follows: At full build-out, the solid waste generated by the project is estimated to average	approximately 80,926 pounds per day. It should be noted that this estimate does not account for solid waste that would be recycled, which would be a considerable amount. Sustainability guidelines have been established for the UH West O'ahu Lands (see Section 2.6). The goal for waste management is to appropriately reduce,	reuse and recycle materials, to munute generation of solid waste and achieve diversion from landfills. The Private Development Lands will strive to achieve the applicable design criteria and the recommended LEED community performance standards. Therefore, the increase in solid waste generated by the UH West O 'ahu' expected to be accommodated by existing solid waste disposal facilities. <u>UH West</u> O'ahu will, promote the optimal use of solid wastes through, programs of waste prevention, energy resource recovery, and recycling with the goal that all wastes are utilized pursuant to Chapter 344-4.2, HRS.	 Water Supply and Distribution. A dual potable and non-potable waster system, to completed by the Board of Water Supply, is planned to run along North-South Road. Sect 2.5.2.2 Water Supply and Distribution of the EIS will be revised as follows: To accommodate the planned developments in East Kapolei, the existing 215-fool Kapolei potable water system will be upgraded and a new 440-foot potable water system will be constructed. For the 215-foot potable water system, a 4.0-million 	gaton (NO) reservoir and a trainsmission main in North-South Koad will be completed with ongoing developments in the area. For the 440-foot potable water system, one 5.0-MG reservoir or two 2.5-MG reservoirs and a transmission main in North-South Road will be constructed concurrently with the development of the UH West O'ahu.	A <u>dual potable and non-potable water system is planned for the 215-foot elevation</u> <u>service zone</u> . The 215-foot non-potable water system is expected to be completed by the Board of Water Supply (BWS) before completion of the UH West O'ahu development.	8. Sustainability. We appreciate your low impact design suggestion and please be assured 1 UHWO will consider the installation of permeable paving surfaces on parking lots, walkw and courtyards. The second paragraph of Section 4.10.4 Drainage Facilities, Anticipa Impacts and Mitigative Measures, Proposed Improvements of the EIS will be revised follows:	The proposed drainage system for the campus will consist of grate inlets in parking lots and landscaped areas, curb inlets along roadways, underground pipe/box drains, and a 10-acre detention/water quality basin with a flow control structure. In addition, as suggested by the UH Environmental Center during the Public Review period.
Mr. Peter Rappa SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Page 3	Background Traffic. Background traffic volumes are not directly associated with the proposed development. They include sub-regional traffic on Farrington Highway and the future North-South Road, future trips associated with the DHHL East Kapolei developments makai and east of the UH West O'ahu property, and future trips associated with D.R. Horton's proposed Ho'opili mixed-use development. The projected background traffic volumes for the year 2015 are shown in Figure 8 of the traffic study (see Appendix H).	4. Cumulative Impacts - Water. A reference to the water supply to the UH West O'ahu campus is illustrated in Section 4.10.2 Water Supply Facilities, Existing Conditions, Potable Water of the EIS as follows:	The UH West O'ahu campus lies over the Pearl Harbor Aquifer and the 'Ewa Caprock Aquifer, and is within the BWS' 440- and 215-foot elevation service zones. Based on discussions with the BWS, water is available to both service zones from the existing 215-foot Kapolei potable water system. Two major water transmission mains (30- and 36-inch) in Farrington Highway provide water to the 215-foot Kapolei and Barbers Point reservoirs via the Honouliui line booster and Kapolei line booster. A 4-million gallon (MC) reservoir for the 215-foot potable water system is planued to accommodate East Kapolei developments including notrions of the UH	West O'ahu property). Currently, there are no existing 440-foot potable water system facilities in the project area and a new system will need to be constructed. Water will be conveyed from the 215-foot system to a proposed 440-foot system reservoir through booter pumps. The UH West O'ahu water system will be part of the East Kapolei regional water system. Which will accommodate East Kapolei and the future demand of the Kalaeloa area. The water master plan for East Kapolei has been submitted to and approved by the BWS.	 Cumulative Impacts - Wastewater. The Honouliuli Wastewater Treatment Plant is expected to increase solids treatment capacity which would accommodate for future build- out in the region. The first paragraph of Section 4.10.3 Wastewater Facilities, Existing Conditions of the EJS will be revised as follows: 	The UH West O'alu property is within the service area of the Honouliuli Wastewater Treatment Plant (WWTP), which has a treatment capacity of 38 MGD. There are future plans to expand the plant's capacity to 51 MGD. The Makakilo Interceptor Sever and the recently completed Kapolei Interceptor Sever currently transport wastematic from existing completed Kapolei Interceptor Construction from sever	WWTP. According acceleration years of the WWTP is limited by the WWTP. According to the City DPP, the capacity of the WWTP is limited by the capacity of the solids handling treatment units, which have a current capacity of approximately 27 to 29 MGD average flow (see Section 8.2). <u>A planned project to</u> add <u>anaetobic digesters</u> , which will increase the solids treatment capacity is tentatively scheduled to be completed by early 2007. <u>After completion of this project</u> the overall WWTP solids handling concervity will be 38 MGD	During the DEIS Public Review Period, the Department of Planning, Site Development Division, Wastewater Branch wrote that the "projected average flow of 1.68 mgd is included in the approved Wastewater Master Plan for East Kapolei, dated June 2006."

Mr. Peter Rappa SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Drafi EIS) December 18, 2006 Page 6	into the project's construction period. Once construction and build-out of the computs is complete, use of the ko'oloa'ula in project landscaping and continued use of the ko'oloa'ula nursery for propagation will ensure a much larger and vigorous population of the ko'oloa'ula than would have occurred without development of the UH West O'ahu property. The HCP also proposes long-term management of the populations to occurrently with project development, over a period of	approximatery 20 years. Ine successivit implementation of the HCP would significantly increase the numbers of new plants on O'ahu and improve the quality of existing populations on adjacent properties.	Also stated in Section 2.4.2 <i>UH West O'ahu Lands</i> of the EIS: <i>LANDSCAPE</i>	As part of the outdoor learning environment, the campus will incorporate xeriscape techniques aimed at creating a sustainable landscape that complements the dry climate, pays tribute to the region's agricultural past, and incorporates planting of native vegetation	12. Agricultural Impact. In response to your comments, the first paragraph of Section 4.3 Agricultural Impact, Anticipated Impacts and Mitigative Measures of the EIS will be revised as follows:	<u>controps of the "</u> utrentry cuturbated totals within the property with be withdrawn from agricultural production <u>as portions of the project site are developed</u> for the proposed configuration of the project site are developed for the proposed	development, <u>which would nave a negative uppact</u> on the above insted agricultural lessees. This will result in some loss in revenues, jobs, of and payoll generated by agricultural activities. However, the State and City have long planned for new development in the project aen, and the property is within the Urban State Land Use District. Tenants have been fully aware, for quite some time, that the proposed project area would be used to accommodate for future development in the region. It is our understanding with the closure of the Del Monte Fresh Produce, there are 3.100 acres of vacant lands in Kunia available for cultivation.	13. Noise. Location S1 measured current noise measurements which were then used to calculate the projected long-term noise levels for Locations L1 and L2. In response to your question, the following paragraph will be added after the first paragraph of Section 4.4 Noise, Existing Conditions, Noise Measurement of the Final FIS.	Location S1. An approximate 30 minute ambient noise level measurement was conducted in the morning at 8 am and in the evening at 5 pm. The davime (8:00 am to $8:30$ am) short-term noise measurement resulted in a 71.4 A-weighted decibel (dBA). The nighttime (5:00 pm to 5:30 pm) short-term noise measurement resulted in a 64.9 dBA. Simultaneous traffic counts were also conducted. These figures were to	pe used in calculating the long-term noise levels for Locations L1 and L2.
Mr. Peter Rappa SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Page 5	UHWO will consider the installation of permeable paving surfaces on parking lots, walkways and courtrards to assist in the management of storm water runoff while increasing infiltration of rain water into the water table. A release structure will be designed to control the discharge rates to the existing peak flows for the 10- and 100- year storm events. A perforated pipe riser will be designed to meet the City-specified releasing time period for water quality control.	9. Topography. We greatly appreciate the information provided. The first paragraph of Section 3.2 Topography, Existing Conditions of the EIS will be revised as follows:	The island of O'alu originated as a voleane from two volcanoes. Wai'unae and $\underline{Ko'olau_a}$ and is characterized by underlying basaltic flows. The 'Ewa Plain is an emerged coral is efformed during the Pleistocene Period, when the ocean level was at higher elevation.	buffs croded by Honouliuf Stream. The plan is underlain by material that has been modified over the millennia and is hard but extremely permeable. This hard, permeable caprock of sedimentary deposits forms a wedge that retards the seaward movement of fresh groundwater from the inland baselifc aquifer. At higher elevations, the ground surface is made of altuvium and sedimentary deposits that	washed down over the millennia. 10. Drainage. We concur that the utilization of portions of Kalo'i Gulch will serve as an open space feature within the campus. In addition, this open space element could serve as ar efuge for a number of nhants and animals. While nhans for Einschma Gulch have as have	finalized, it may have to be channelized or put into a box culvert.	11. Habitat Conservation Plan. We strongly concur that native plant species be used for landscaping, where appropriate, and that xeriphytic plants be used. We also recognize your suggestion that in addition to removing the populations to the <i>Abutilon menziesii</i> (xo'oloa'ula) to the three nursery sites that an attempt is made to leave some plants integrated with other native plants around campus. As stated in Section 3.7 Flora, Anticipated Impacts and Mitigative Measures of the EIS:	Seeds that may be present in the soils and the few individual plants along the southeastern boundary of the property will be removed. Howver, the long-term impact on the ko'olod ula population in Kapolei would be beneficial, as proposed mitigation measures in the HCP will ensure the future propagation of new plants.	Under DLNR's interim management program, three new populations of ko'oloa'ula have been initiated at Koko Crater Botanical Garden, Ka'ena Point State Park, and the Honoultuli Unit of the Pearl Harbor National Wildlife Refuge. The Ka'ena Point State Park and Pearl Harbor National Wildlife Refuge are wild sites, and additional sites will continue to be pursued to yield three successful wild sites for propagation of ko'oloa'ula.	If an off-site preserve can be established, some of the plants would remain in situ for many years. Because the protection, propagation, and relocation of the ko'oloa'ula plant is a long-term undertaking, final implementation of the HCP will extend well

Mr. Peter Rappa SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAF ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Page 8	technical assistance to the Energy Branch, SID, DBEDT, to enable it to provide assistance to State facilities (including presumably the University of Hawai'i) to assist in meeting th Leadership in Energy and Environmental Design (LEED) silver or two green globes ratin system or another comparable state-approved, nationally recognized, and consensus-base guideline, standard or "green building rating system."	17. Housing. Subsequent to the publication of the Draft EIS, there has been a change in the project and the 355 affordable housing units proposed within the UHWO campus is no longe being proposed. As a result, the second to the last paragraph of Section 4.8.2 <i>Housing Existing Conditions</i> of the EIS will be revised as follow:	Park areas and open space will be provided throughout the property in compliance with the City's Park Dedication Ordinance. The University and/or the private developer vill meet with affected City departments to discuss specific plans to satisfy the park dedication requirements. For meet the City's affordable housing requirements, approximately 355-units, a large portion of vihiel vull be affordable housing units, vull be provided vihiln the UH West Ordin Lands. The remainder of the affordable units required for the development will be distributed throughout residential-pareels (and not concentrated in one or two areas) in the Private Development Lands:	The project will address the provision of affordable housing by coordinating with the Department of Planning and Permitting and the Department of Community Services. UHWO is particularly motivated that the project provides attentiable housing for faculty and staff as it is difficult to attract the best candidates without it. In addition. UHWO will eventually include approximately 760 student. housing in with approximately 3 beds each. Thus there will be approximately 2.280 less students that would be competing for affordable renals in the open market. The University and/or the study of the open market. The open market. The University and/or the study of the open market. The open market and or the study of the open market. The University and/or the study of the study of the open market. The University and/or the study of the study of the open market. The University and/or the study of the study of the open market. The University and/or the study of the study of the study of the open market. The University and/or the study of the study of the study of the open market. The University and/or the study of the study of the open market. The University and/or the study of the study of the study of the open market. The University and/or the study of the study of the study of the open market. The University and/or the study of the study of the study of the study of the open market.	The private acteroper with meet with affected Liry departments, including the Department of Community Services to discuss specific plans to satisfy the affordable housing requirements. During the public review period, the Department of Planning and Permitting, wrote the following regarding the potential of developing "gramny flat"-type remtal units in proposed housing development. "The DPP considers the potential benefits (i.e. low cost student remtals and increased housing unit affordability) available through this type of housing to be substantial. It appears than an opportunity for this type of unit exists where the "alley loaded" low-density residential development on Parets C and D where the "alley loaded" low-density and on opportunity for this type of unit exists where the "alley loaded" low-density residential development on Parets C and D where the "alley loaded" low-density and on opportunity for this type of unit exists where the "alley loaded" low-density and on opportunity of this type of unit exists where the "alley loaded" low-density and on opportunity of this type of the dot where the "alley loaded" low-density and on the operation of the provident of the dot dot dot density and operation of the operation of the dot down of the dot	and the second second and the second second contained and the second second and an easily particular that the second seco	Mitigation Measures of roadways and traffic could have been improved by providing a map of the proposed roadways. As such, Figure 4.2 will be provided after Section 4.10. Roadwarw and Traffic, Anticipated Improve and Mitianton Measures, Boodwarw in the
Mr. Peter Rappa SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Drafi EIS) December 18, 2006 Page 7	14. Construction Noise . The Environmental Center raises a good point, and we believe that while the State DOH establishes noise level guidelines, which contractors must abide by, the reality is that if nearby residents are adversely impacted by construction noise, they will complain to the DOH, and DOH will send its own, unbiased staff to monitor the alleged excessive noise. If the noise standards are exceeded, than the contractor will be forced to apply for a permit and/or take measures to reduce construction noise.	15. Air Quality. Per the Environmental Center's comments, the second paragraph of Section 4.5 Air Quality, Anticipated Impacts and Miligative Measures, Long-Term Impacts and Mitigation of the EIS will be revised to read as follows:	Measures to mitigate long-term, traffic-related air pollution include improvements to roadways, reduction of traffic, and reduction of individual vehicular emissions. Given that the more stringent State standards would likely be met during worst-case conditions, implementing any air quality mitigation measures for long-term traffic- related impacts is probably unnecessary and unwarranted. Nevertheless, the UH West 0 and will encourage public transit into the design of its campus as it is located near planned rail transit nodes. The addition of buffer zones between walkways and roadways can help mitigate potential air quality impacts, and the UH West 0 ann will include acres of parks and open space. In addition, all internal roads are currently planned to include planting strips or tree wells.	A comprehensive transportation master plan, which incorporates various modes of travel, including transit, vehicle, bicycle and pedestrian will be developed for inclusion in upcoming zone change and PRU applications. Part of the goal of preparing this plan is to reduce traffic congestion in the 'Bwa plain and to reduce air pollution. A new paragraph will be added to Section 4.10.1 <i>Roadways and Traffic, Anticipated Impacts and Mitigation Measures</i> of the EIS to read as follows:	As requested by the Department of Planning and Permitting (DPP) Traffic Review Branch, a comprehensive transportation master plan, which incorporates various modes of travel, including transit, vehicle, bicycle and pedestrian will be developed and included with zone change and PRU applications to DPP. This plan will be designed to establish and promote a safe and efficient balance between the various modes of travel, such as grade separated bicycle and pedestrian facilities, convenient and contrally located transit stops and terminals, traffic calming devices and other methods, as necessary.	16. Electrical Demand and Solid Waste Generation. As noted on page 50 of the Draft EIS, sustainability guidelines (including energy conservation) have been established for the proposed UHWO campus. To that end, UHWO has hired John Hara & Associates, Inc. (the architectural designer of the Punahou Case Middle School, which achieved a LEED Gold standard) and they in turn have hired Arup, an engineering firm who's acknowledged as a leader in LEED and who have developed their own tool in incorporating ustainability	features. In addition, the Department of Business, Economic Development, and Tourism (DBEDT) has released a RFP entitled "Technical Assistance for Green Buildings, Solicitation No DFD-071.04 STD," Approach to account the second of this marked second seco

Mr. Peter Rappa SUBBECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Page 9	Mr. Peter Rappa SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Page 10
19. Parking. The City's LUO does not have off-street parking requirement for colleges and universities. A ratio of one parking stall for every two students was utilized by UH West O'ahu, to accommodate the parking needs of students, faculty, and administrative staff. For the initial 1520-student campus, 760 parking stalls (at 0.5 per student head count) are assiminated to be included.	parking standards for the ultimate development of the campus could be reduced should transit access to the area be enhanced through the City's proposed HHCTP. The University will continue to work with the City during the Plan Review Use (PRU) permitting process to ensure that the design of future parking facilities adequately meets future needs.
estimated to be included. And utilitate 7,000-student campus is expected to include 2,850 parking stalls (at 0.375 per student head count). The traffic engineering consultant compared these ratios with the Institute of Transportation Engineers average rates for parking generations (at 0.37 per student head count). The proposed number of parking stalls exceeds the number of stalls suggested by the Institute of Transportation Engineers.	Table 4.4 shows the initial phase student population and the ultimate student population and the proposed number of parking stalls that is being proposed for each phase of campus development. Table 4.4 also provides the estimated number of parking stalls using methodology provided by the Institute of Traffic Envineers and if one work of commone with the number of eacle worked of to
Section 4.10.1 Roadways and Traffic, Anticipated Impacts and Mitigative Measures, Parking of the EIS will be revised as follows:	comparable number of students at the UH Mānoa Campus. Table 4.4 - Proposed UH West O'altu Parking Stalls
the City's Land Use Ordinance (LUO) establishes a minimum of one parking stall for every tan students, phis one stall for every 400 square feet of office space. Based on this standard, a ratio of one parking stall for every two students was utilized to accommode the parking needs of students, faculty, and administrative was to be seen.	Year Student UILWO ITE UILManoa based Population Stalls ITE UILManoa based Provided Stalls ITE Itead Count (Students) (Stalls) (Stalls) (Stalls)
ач <u>ар — For-mer сталия - 7.20 у нисет сатрик, - vor ранхену заань стан ост рок 5 нисен</u> head count) are estimated - to be-required. — The utimate 7,600 student campus in expected to require _ 2,830 ранкну, сцави (461 0,875 рак-student - head count). — The	2009 1.520 760 502 464 2015 7.600 2.812 2.812 2.318
proposed-number of parking stalls exceeds the LUO requirement and the number of stalls suggested by the Institute of Fransportation Engineors— The University will continue to work with the City during the Plan Review Use (PRU) permitting process-to-ensure that the design-of future parking facilities advantely mean	It is possible that the proposed UHWO campus is more likely to have impacts on on-street parking on existing residential neighborhoods nearby. For the initial 1.520 student campus, the project site has adequate land available if each of the students were that their pair own over the hand more some day. To the
future needs—Future factors that may affect parking demand are: _}} plans to provid e student housing equivalent to _30_percent of the ultimate full-time equivalent student population , and 2) plans for sail and bus transit .	student campus is expected to include 2.812 parking statis (at 0.37 per student student campus is expected to include 2.812 parking stalls (at 0.37 per student head count). The proposed number of parking stalls equals the number of stalls suggested by the Institute of Transportation Engineers (at 0.37 per student head
For the initial 1.520-student campus, 760 parking stalls (at 5 per student head count) are estimated to be required. This requirement is higher than both the Institute for Traffic Engineers (ITE) standard and the ratio utilized for the	count. Since the nearest existing, restatential neighborthoods are located in the Villages of Kapolei and the Villages of Kapolei is separated from UHWO by the Kapolei Golf Course, without any connecting streets, any students who park in the Villages of Kapolei would have to either walk, bike or moped onto either Kapolei
uturate 1.500 student campus. 1ms numer parking requirement was uturzed to compensate for the lack of development of alternative modes of transportation and lack of student housing in the immediate vicinity of the campus, which would result in higher automobile use during the initial phase. For the utlimate 7,600 student campus conversioned of 28.10 metrins each of 37 ms endow hoodsoned	Parkway or Farrington Highway to access the campus. The closest proposed residential neighborhoods are the private development lands surrounding the campus and it is hoped that students from these areas will either walk or bike to the campus. It is possible that in the long-term future, if there is inadequate
will be provided. This number is equal to the Institute of Traffic Engineers (ITE) suggested parking requirement of 2.812 stalls.	parking provided on the UHWO campus, commuting students will park on-street in the private development lands or on surrounding proposed projects (DHHL East Kapolei 1 - which is immediately makei of the 500 acre UHWO site, and DHHL East Kapolei 2 and the Ho'ronili majori – which are both across the
Factors that may impact the estimated parking requirement for the 7,600 student campus include: 1) the level of transit service to the UH West O'ahu campus (less parking required if a significant portion of the UHWO students, faculty and staff use transit). 2) the percendage of distance learning, students, 3) the	proposed North-South Road from UHWO). Development of the UH West O'ahu Campus Lands will comply with the City's requirements for off-street loading, pursuant to Section 21-6,100, LUO.
percenter of matern strong in ton-arytime courses, and 11 ne amount of nearby and on-campus student housing which is projected to be equal to 30 percent of the total student population. Parking standards could be adjusted, as necessary, to accommodate for any of the factors identified above. For example,	20. Electrical Facilities. The Environmental Center raises important points, and we believe that UHWO's architectural designer, John Hara & Associates, Inc. and its sustainability

Mr. Peter Rappa SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Page 12	that a school fair-share contribution for the proposed project would include sufficient acreage for an elementary school and a cash contribution. In addition, DOE wrote that it "recognizes UH's acknowledgement of the need for a contribution to offset the impacts of the proposed project and we look forward to working together." As such, one possibility is the development of a "Lab School" on the compus site.	22. Alternatives to the Proposed Action. An analysis of the University of Hawai'i system will be included in the Final EIS, highlighting the need to develop a UH West O'ahu campus. Section 6.3 "No-Action" Alternative of the EIS will be revised as follows:	ALLE ADJUSTION ALCTIONED TO ALLAND IN ALCONGING TO ADJUSTIVE OF THE PLOPASE ACCION AS discussed in Section 2.2. Need for the Project, with the expected population increase in the Swa region, the proposed UH West O'ahu campus is expected to target students in the surrounding regions to alleviate the burden on the UH Mānoa campus. The relative need for postsecondary education based on	<u>projected changes in population from the year 2000–2020, are shown in Tables</u> <u>2.1 and 2.2.</u> The "no-action" alternative would not be consistent with stated governmental policies of directing future growth toward the 'Ewa Plain (as in the Ewa Development Plan). It would also be inconsistent with the University of	Havai'i - West O'ahu Strategic Plan 1997-2007, which seeks to establish new higher education and employment opportunities in the region. Under this alternative, the property would remain as agricultural and vacant land, surrounded by planned urban developments in the area. After weighing the socio-economic benefits of the campus against the loss of actively cuttivated land	on portions of the property, the "no-action" alternative was rejected. Thank you again for your participation in the Environmental Impact Statement process for this project. Your comments will be included in the Final EIS.	If you have any questions regarding this project, please do not hesitate to contact me at 521-5631. Sincerely,	PBR HAWAII Vuried R. Shugh-	U Vincent R. Shigekuni Vicc President	encl: Figure 4.2 Conceptual Land Use Plan	CC. OLIVE OF ANARUNI/UH West O'ahu Gene Awakuni/UH West O'ahu o.∪08131322055EIStfinal EIStfinal Responses to DEIS Comments/BL-17 UHEC response 11-204∞
Mr. Peter Rapa SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Page 11	consultant, Arup, will consider both photovoltaic and wind-derived as alternative energy sources. The University of Hawai'i is already studying such technology at its Coconut Island marine research facility. HECO is partnering with the Hawai'i Institute of Marine Biology (HIMB) to develop energy efficiency and renewable energy applications for the pland. Apparently renewable energy technologies such as wind turbines and photovoltaics will be explored. For the Environmental Center's comments we are revising Section 4.10.6 <i>Electrical Facility</i> .	EIS by adding the following paragraph at the end of this section: While current plans for the development of the UHWO campus includes both energy efficient design and the use of electricity developed and sold by HECO, the	and photovoltaics. 21. Educational Facilities. Please be assured that the University of Hawai'i is proposing a minimum of one DOE elementary school site on its site, but that the burden of providing	DOE school sites for UHWO (or DHHL) does not fall on either curity alone. To address the currulative impacts of school-aged children being generated by the residential components of both entities, we mentioned that DHHL is proposing an elementary school and a separate middle school site on its property immediately east of the UH West O'ahu	and that DOE is contemplating a high school on DHHL land immediately north of the UH West O'ahn. During the public review period, the DOE provided comments which will be incorporated into the first two paragraphs of Section 4.11.1 Educational Facilities, Anticipated Improvements and Mitigative Measures of the EIS as follows:	The UH West O'ahu will likely increase the number of students errolled in public schools. Based on the table provided by the DOE (Table 4.65), in 2009, three elementary schools (Barbers Point, Iroquois Point, and Kaimiloa) will be under capacity by approximately 807 students, and Kapolei High School will be a second to so a second the second s	respectively. During the public review by approximately 2.50 and 0.59 statents, respectively. During the public review period the DOE wrote that the data provided in the DEIS fluctuates the growing enrollment in the schools and the projected number of students that well grow beyond the capacity of the existing schools' classrooms. The DOE estimated that when the UFH project is mature.	that there will be approximately 1.771 public school students living within the project and that would be enough elementary school students to fill one elementary school.	<u>UH has held preliminary discussions with DOE to address the demand of future</u> <u>school-aged children that would be generated by the proposed project.</u> To help meet the demand for public educational facilities in the area, a 12-acre elementary school is proposed within the southern portion of the UH West O'ahu	property (see Figure 2.1). The location of the school would be in close proximity to the proposed within the property and the planned DHHL	restantiates subsystom attenty sound of the property. Ane school is projected to accommodate <u>a typical DOE elementary school for</u> 550 students and 60 faculty and staff. <u>During the public review period, the DOE also wrote that it estimated</u>



Water Resource Engineering and conclude that "research into the history of the Gulch reveals there has never been a <u>natural</u> outlet connecting the Gulch to the ocean." While the Corps does not dispute this conclusion, it should be noted that a man-made water conveyance can establish a tributary connection to a water of the U.S. This is supported by case law (<u>United States v</u>. Newdum, 344 F.34 407 (4th Cir. Sept. 10, 2003)). The report itself goes on to describe how the For the record, we would like to clarify several misleading statements made in your November 28, 2005 letter and in Mr. Wright's report that we feel result from a misunderstanding of the Corps' regulatory program and a selective interpretation of the information in Corps files. sump to an ocean outlet. According to the "Drainage and Soil Erosion Control Report for Haseko the gulch onto makai lands, ending approximately one mile from shore at a sump for the shallow P. 12, states that, "A drainage ditch runs from the cane fields through the project area terminating An earlier report, however, describes a continuation of the so-called Kaloi ditch from this Under I(A)(1) of your letter you reference a 1997 report by PBR Hawaii and Tom Nance ever in fact overtopped the Coral Creek system and discharged to the Pacific Ocean. The storm water management systems associated with developments within the lower reaches of the Kaloi Gulch watershed have been engineered such that the upper reaches of the gulch have been, on a gulch had been modified by sugar plantations in the early to mid-twentieth century. According Bwa Marina Community Project, Volume I" by Paul H. Rosendahl, Ph.D., Inc., December 1995, at the occan within One'ula Park. This trench is lined with the terrigenuous soils from the sugar to the authors, the modifications consisted of constructing parallel levees to direct flows within 1. Absence of an historical, natural hydrological connection between Kaloi Gulch and Pacific Also, the "Final Report, Phase II – Data Recovery, Archaeological Mitigation Program, (Ewa), Inc. Golf Course" by Wilson Okamoto & Associates, Inc., September 1994, P.10: The existing Kaloi Gulch irrigation ditch is used for delivery of irrigation The irrigation ditch in the Haseko (Ewa), Inc. golf course site will remain The Kaloi irrigation ditch empties onto an existing sump area in a kiawe forest before it outlets to the Pacific Ocean via a very small drainage ditch near as it exists until sugar cultivation in the region is terminated and the need of the water. It does not generally function as a drainage channel since the channel irrigation facility ceases. The irrigation ditch will be removed eventually as the City and County of Honolulu's Oneula Beach Park. practical scale, effectively isolated from the Pacific Ocean. invert is higher than the adjoining topography. properties upstream are developed. .. . caprock well known as EP-27. cane fields." .**g** Hawaii (UH), including a report by Wayne Wright of GeoEngineers dated November 7, 2005, in support of your request that the Corps re-evaluate its assertion of DA jurisdiction over the Kaloi Gulch in the vicinity of the proposed UH West Oahu Campus. Subject: Request for review of Department of the Army (DA) jurisdictional determination for This office has reviewed the materials you submitted on behalf of the University of File No. POH-2005-89 U. S. ARMY ENGINEER DISTRICT, HONOLULU FT, SHAFTER, HAWAII 96359-5440 DEPARTMENT OF THE ARMY February 16, 2006

that although an emergency overflow channel was constructed in 2000 to allow overflows from a very large, infrequent storm event to overtop the detention/retention basins of the Coral Creek golf course and proceed towards the Pacific Ocean in the vicinity of Oneula Beach Park (see details below), a site inspection revealed there are no physical indications that storm flows have

Kaloi Gulch, Ewa, Oahu, Hawaii

Dear Ms. Kuwaye:

Imanaka Kudo & Fujimoto

Naomi U. Kuwaye

REPLY TO ATTENTION OF

Regulatory Branch

745 Fort Street, 17th Floor **FOPA Financial Center**

Fort Street Tower

Honolulu, HI 96813

0

to be an exhaustive file review. Much of this material had already been reviewed by my staff in staff on November 4, 2005, and, on November 8, 2005, the Corps provided copies of the documents she requested (copy of transmittal letter attached). The files presented to Ms. Uechi for her review were intended to provide a starting point and were not represented as or intended In support of your request, a file review was conducted by Ms. Stephanie Uechi of your making our original determination of jurisdiction.

drainage basin, addittonal reports associated with the Haseko Ocean Pointe Marina development, As part of its recvaluation of its jurisdictional determination (JD), this office reviewed your report, the files referenced in your letter as well as others pertaining to the Kaloi Gulch acrial photographs, additional site inspections, and applicable case law.

connection to waters of the U.S. Therefore, a DA permit pursuant to Section 404 of the Clean While the Corps does not concur with the rationale behind your conclusion, we agree with your conclusion that the upper reaches of Kaloi Gulch do not have a regulated tribulary Water Act (CWA) will not be required for the proposed fill associated with the UH West Campus or the North-South Road As a result of our re-examination of the facts associated with Kaloi Gulch, we have found

The existence of this small remnant drainage ditch was confirmed by Connie Ramsey and

Farley Watanabe of my staff during a site inspection on November 1, 2005. Further, it is

depicted on a map in the "Final Report, Phase II – Data Recovery, Archaeological Mitigation Program, Bwa Marina Community Project, Volume III: Illustrations" by Paul H. Rosendåhl, Ph.D., Inc., December 1995, Figure 3.4, "Historic Impacts to Project Area" (attached).

ŝ

(~.

Therefore, the statement in the 1997 PBR Hawaii/Tom Nance report that "no channel, man-made or natural, has ever existed as a continuous conveyance all the way to the shoretine" appears to be incorrect. In his report, Mr. Wright states that "Yoo evidence was found that Kalo'i Gulch is connected to the Pacific Ocean at any time." An Environmental Impact Statement Preparation Notice (EISPN) for the Kaloi Gulch Drainageway Improvement at Oncula Beach Park, preparation Notice (EISPN) for the Kaloi Gulch Drainageway Improvement at Oncula Beach Park, preparation Notice (EISPN) for the Kaloi Gulch Drainage outlet is located at the easternmost portion of the existing Orienta Beach Park" (p.11). The EISPN further explains that "the portion of Gulch which now connected to the project site, is the result of the construction of an enorgency trief edamnel construction in June 2000" (p.23).

The stated purpose of the proposed drainageway improvements at Oneula Beach Park is to complete the interim measures of the initial emergency relief channel constructed to avoid recurrence of the flooding event that impacted Ewa Villages in November 1996, and, ultimately, to meet the City & County of Honolulu's drainage standard for a 100-year flow.

As further clarification, the "Kaloi Guloh Drainageway Improvements at Oneula Beach Park, Draft Environmental Impact Statement" by R.M. Towill, December 2004 (p. 4-7), states:

As the Ewa Plan has undergone development over the past quarter contury, conversion from agriculture to urban has both increased and concentrated stormwater runoff within the watershed. Developers in the Kaloi Guleh Watershed have installed basins for retention and detention to contain and store large amounts of storm water within their developments where it infiltrates into the ground.

Runoff during periods of prolonged heavy rainfall, however, can exceed the retention capacity of these features and overflow onto makai lands, including Ocean Pointe and Oneula Beach Park. To accommodate this overflow and to provide flood relief for upland communities, HASEKO in June 2000 constructed an overflow emergency channel through the Ocean Pointe site. This emergency channel can clearly be seen in the acrial photographs (Figure 2) in Mr. Wright's report. Thus, the Corps acknowledges that the downstream reaches of the Kaloi are usually dry, but the possibility remains for the upstream retention basins to be breached and for storm waters to move downstream within this wide, shallow drainageway to the Pacific Ocean.

As additional background, the 1997 PBR Hawaii/Tom Nance report also describes the separation of Kaloi's irrigation and drainage functions in the early 1990's as the City's redevelopment of Ewa Villages removed the levees in the project vicinity to direct stormwater into on-site retention-detention areas within what is now known as the Ewa Villages golf course. The irrigation function for plantation operations was maintained by a newly constructed

irrigation ditch that ended as described in Page 2 above. The stormwater flows were to be impounded within the Ewa Villages golf course with no outlet makai, which resulted in significant flooding of Ewa Villages from two major storms in November, 1996. As a result of these flooding events, the City and County of Honolulu initiated a task force to focus on the drainage issues within the Kaloi Gulch Watershed basin. As a result of regional planning, in the 1990's, a bridge was constructed under the OR&L raiload right-of-way, which, until that point, had acted as a berm to concentrate floodwares in the Ewa Villages. The bridge construction included the re-alignment of the Kaloi channel, connecting it downstream to Ewa by Geniry's Corris Course, which had incorporated retention basins within its design to accommodate storm flows.

(

Ċ.

-

It should be noted that since the date of the 1997 PBR Hawaii/Tom Nance report, further modifications to the Kaloi alignment have occurred (including those described above) as a result of continuing development of the Ewa plain as a major suburban region. This is also relevant to your discussion of the US Geological Survey topographic map "blue line" depiction of Kaloi Gulch, which indicates there is no surface connection to the Pacific Ocean. The USGS topographic map in Mr. Wright's report (Figure 1) is dated 1998, and since then changes have occurred within the watershed, which is why the USGS topographic maps are generally used by this District primarily as a guideline and not as conclusive records as to the extent of waters of the United States.

2. Corps jurisdictional determinations are inconsistent

Section III of your letter asserts that Corps' jurisdiction calls are inconsistent because we have allegedly not asserted jurisdiction over lower reaches and yet have asserted over upper reaches of the gulch. You support your point with a summary of jurisdictional determinations for areas downstream of the project location <u>only</u>. However, in reviewing these files, the situations vary and we believe it inappropriate to represent these as being inconsistent.

To further explain, I would like to note that there are essentially two aspects to Corps jurisdictional determinations: 1) the geographic determination of the presence and extent of waters of the United States; and 2) whether the proposed work involves the discipations of the determination that no Permit is required for a proposed project to mean the subject water body is not regulated, when, in fact, the waterbody is determined to be a "water of the U.S." however, for of Corps jurisdictional decisions are associated with the Specific location and of Corps jurisdiction. Jurisdictional decisions are associated with the specific location and project scope and are necessarily case-by-case. Below are the cases you reference in your letter, with more specific information experimentance:

Corns File No. 990000285: The Corns letter dated May 10, 1999 addresses jurisdictional requirements for the construction of a channel for the relocated Kaloi Gulch to connect an existing retention pond (Ewa Villages Golf Course) to the Coral Creek Golf Course pond. The letter states that "connection to the jurisdictional portion of Kaloi Gulch will involve excavation

~-
ر. ۲	•	subsequent subsequent wher mark berefore, coups File No. <u>990000173</u> Corps letter dated February 23, 1999 providing co relation to Environmental Assessment for the proposed North-South Road state "a DA po required for work done in Kaloi Gulch at the road crossings."	r reaches in the Corps File No. 980000294 Corps letter dated September 2, 1998 providing co EISPN for the UH West Oahu Campus state that "any impact to Kaloi or Hon water may require a DA permit." I would like to note that all these files were reviewed by Ms. Uechi of exception of the Wilson Okamoto & Associates "Drainage and Soil Erosion C	 Corps Corps Corps Corps Pacific Paci	Campus project. If you object to thus determination, you may request an admit ter of the contrast of the second second and the second seco	 D.S. Army Corps of Engineers indicated indicated Pacific Ocean Division Part Part 331.5, and tha televed Part 331.5, and tha received by the Division Office on within 60 days of the AAP. Should mpus 	last I do not object to this jurisdictional determination, there is no need on the Pacific Ocean Division.	
ر		ouly. There will be no discharge of dredged or fill material to the existing gulch," A letter dated July 13, 1999 states that "all work will take place above the ordinary high in Kaloi Gutch. Therefore a Department of the Army permit will not be required." T the Corps did not require a DA permit based on the nature and location of the work in a water of the U.S., not because the discharge of fill material below the OHWM with Gutch would not have been regulated.	Corps File No. 99000316 A Corps letter dated January 23, 1998 states that "the upp of Kaloi Guloh are considered jurisdictional waters of the U.S. However, Kaloi Guld vicinity of the Coral Creek Golf Course project site does not exhibit an ordinary high mark, and therefore, is not considered a [jurisdictional] water of the United States at the location. We reaffirm our initial determination that a DA permit will not be required project."	Corps File No. 200400466 The Corps letter dated November 22, 2004 states that "Th considers Kaloi Gulch to be a water of the U.S. based on its tributary connection to the Cocam, however, a site inspection conducted by a member of my staff on November 1 revealed the absence of an Ordinary High Water Mark (OHWM) by which to establist jurisdictional line for this section of the gulch. Any discharge of dredged material or it portion of the gulch will be required."	Further, a particular aquatic feature may provide a tribulary connection to a wa U.S. without itself being regulated. An aquatic feature, such as downstream Kaloi Gu not necessarily need to have a defined bed and bank or an OHWM to act as a hydrolog connection. Therefore, what you interpret from our jurisdictional calls to indicate a st connection and represent inconsistencies is in fact meither. Your assertion that the Con declining jurisdiction over the downstream portions of Kaloi severs the tributary conne upstream portions is erroneous.	In response to Mr. Wright's statement that, "A recent review of the Corps' file, that, in fact, this is the first instance where the Corps has asserted jurisdiction over Kal despite an extensive history of various project that involved the Gulch," the Corps wou refer you to the below Corps jurisdictional determinations for which the records were by your firm, but were not mentioned in your letter. The records do in fact show cons over time in our jurisdictional determinations for Kaloi Gulch in the UFI West Oahu ca project area:	Corps File No. 970000161 The August 12, 1997 Corps letter in this file addresses the I Kapolei Master Plan, which is where the proposed UH West Oahu campus is located, a "Kaloi Gulch at the subject area is considered a jurisdictional water of the U.S. This determination was based on the identification of Kaloi Gulch as an internittent stream U.S. Geological Survey map and the presence of an ordinary high water mark. The determination was also based upon the condition of the gulch at the project site. Previo	

•

.

.

BR HAWAII vsociates. inc.	December 18, 2006	Mr. George P. Young, P.E. Chief, Regulatory Branch Department of the Army U.S. Army Engineer District, Honolulu Fort Shafter, Hawaii 96858-5440	SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) Dear Mr. Young:	I hank you for your letter dated July 27, 2006. We have reviewed your letter and offer the following response to your comments: Thank you for the information that a DA permit will not be required for the proposed project. We understand that this determination does not relieve UH West O'ahn of the responsibility to obtain all other applicable federal, state or local permits. We are confirming that this project also involves the development of off-site infrastructure, including potable water, wastewater and drainage facilities to accommodate the primary development activities, and this work has been the subject of other environmental reviews, and, as such, it is not addressed by your correspondence (File No. POH-2005- 89). Your letter and this response will be included in Section 12.0 of the Final EIS.	Thank you again for your participation in the Environmental Impact Statement process. If you have any questions regarding this project, please do not hesitate to contact me at 521-5631. Sincerely, PBR HAWAII PBR HAWAII Vincent R. Shigekuni Vice President	cc: Office of Environmental Quality Control Gene Awakuni/UH West O'ahu o:uOB1341322.05EISWTaai EISResponse to DEIS CommentSBL 09 DOA response 11-20 dec
P s, P		W. FRASK BRANDT, FASIA Chorman THRMASS, WITTBA, ASIA President R. STANDUSCAN, ASIA F. STANDUSCAN, ASIA F. STANDUSCAN, ASIA	RUSSILA FLARVIG, FANIA Esconter Vices Preddout VINCENTER PRIMERUN Une President GRANTT MURAKAMI, AICP Printpal	TOM SCHNETL, AICF Scalar Absociate RAYMONE HIGA ASIA Neur About the KEVINK, NISHIKAWA, ASIA Aswith KIMI AIRANII VU'S, HTD'A Aswith Aswith SCOTT AILKA AIRREO SCOTT AILKA AIRREO SCOTT AILKA AIRREO SCOTT AILKA AIRREO Aswith SCOTT AICHAKAMI, ASIA Aswith	IGNNOLULU OFFICE 1041 Halanp Storet AND Jacob State (2) AND Jacob State (2) AND Jacob State (2) Jacob Jacob State (2) And Jaco	WAILKU OFFICE 1-97 With Long. State 4 Wallita (Kasari 97791 J.27) 14-10983 27: 5528
		For more information on our regulatory program, please visit our web site at <u>http://www.poh.usace.amy.mil/regulatory.asp.</u> If you need further assistance, please contact Ms. Ramscy by phone at 438-2039, by facsimile at 438-4060, or by cleetronic mail at <u>Connie.L. Ramscy@usace.army.mil</u> ? Please refer to file number above for further inquiries regarding this project. Thank you for your cooperation with our regulatory program.	George P. Young, P.E. Chief, Regulatory Branch	Attachments: Transmittal letter from Corps dated November 8, 2005 Figure 3.4, Histonic Impacts to Project Area, "Final Report, Phase II – Data Recovery, Archaeological Mitgation Program, Eva Marina Community Project, Volume III: Illustrations" by Paul H. Rosendahl, Ph.D., Inc., December 1995 Notification of Administrative Appeals Options and Process and Request for Appeal Form		1.

.

•

FFANNING + IANDSCAFE ARCHITICTURE + FAV180NKUNTAL STUDIIS + FNTITLEMENTS / FERMITTING + GRAPHIC DIVIGN

SR HAWAII sociates, inc.	December 18, 2006		Mr. Keith S. Shida, Principal Executive Customer Care Division Board of Water Supply City and County of Honolulu Gib South Bertania Street	HOHORHIR, FIAWAI 1 2004-5 SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) Dear Mr. Shida:	Thank you for your letter dated July 24, 2006. We have reviewed your letter and offer the following responses to your comments:	 We acknowledge that UH West O'ahu and/or private developer will be required to install the necessary water system improvements to serve the proposed development. 	 We acknowledge that UH West O'ahu will be required to pay the BWS Water System Facilities Charges for resource development and transmission. 	The second paragraph of Section 4.10.2 Water Supply Facilities, Anticipated Impacts and Mitigation Measures, Proposed Improvements of the EIS will be revised as follows:	Upgrades to the 215-foot potable water system include the installation of a 4.0-MG reservoir and transmission main in North-South Road. These upgrades will be completed with ongoing developments in the area. The new 440-foot elevation system will include a 5.0-MG reservoir or two 2.5- MG reservoirs and a transmission main in North-South Road. These facilities will be completed concurrently with development of the UH West O'ahu. Construction of the 215-foot non-potable water system to the UH West O'ahu. Construction of the 215-foot non-potable water system to the UH West O'ahu. Property is expected to be completed by the BWS before completion of the UH West O'ahu. <u>UH West O'ahu and/or a private</u> developer will be required to poy the BWS Water System Facilities Charges for resource development and transmission. The fourth paragraph of Section 4.10.2 Water Supply Facilities, Anticipated	impacis and initigation measures, rroposed improvements of the Els Will be revised as follows:
PB Anter a Asse		W. FRANK IRAND'I, FASI A (Tauthin	THOMAS SWITTEN ASLA Predden R. STANDENCAN ASLA Faendre Tier-Preddent RUSSTLL P. LOHL YG, PASLA	Ecounty Tree Provident VIXCESY BHCBAUNI The Prevident GRANT 7, MI KM KAMI, ALCP Privident TOM SCHNITL, ALCP	Scution As occurate RAYMOND THIGA, ASLA Scutter Associate	KEVINK, NISHIKAWA, AMI.A Ausu ato Kimi Mikami Yuyy, 11719'ad Aswedato	SCOYT ALIKA ABRIGO Acoriate Mentate Astati Asi A	. Ison tide	MONOLULU OFFICE (NOT Bidgen Street, State (NOT Bidgen Street, State) (NOT Bidgen Street, State) (State) (State) (State) (State) (State) (State) (State) (State) (State) (State) (State	1:23 Wile Jacop Sate 4 Waldon Haoa 1 W 29:1-127 Ré-1886 - 12 28:8
kufi hannemann, kayar	RANDAL Y. S. CHUNG, Creuman Herbery S. K. KAOPUA, SR. SANUEL T. HATA ALLY J. PARK	RODNEY K. HARAQA, Ex-Oficio LAVERNE T. HIGA, Ex-Oficio	CLIFFCRO P. LUM Manage and Chei Ei-gineer DONNA FAV X. KIYOSAN Depuy Manager and Chei Eigilneer		ff Environmental Immede	MK: 9-1-016: 120, 127 &	iect,	system improvements to	to pay our Water System on. 5440. A Bivision	
)	July 24, 2006		ssident	lith 3, 2006 Benerding the De	versity of Hawaii West Oahu, TI	o comment on the proposed pro	ts to otter. ed to install the necessary wate. ment.	le, the applicant will be required ce development and transmissi se contact Robert Chun at 748- Very truly yours KEITH S. SHID Principal Execu Customer Care	
BOARD OF WATER SUPPLY	CITY AND COUNTY OF HONOLULU 630 SOUTH BERETANIA STREET HONOLULU, HI 98843			Mr. Vincent Shigekuni, Vice Pri PBR Hawaii ASB Tower, Suite 650 1001 Bishop Street Honolulu, Hawaii 96813	Dear Mr. Shigekuni: Subiect: Your Letter Dated	Statement for Unit	Thank you for the opportunity to We have the following common	 The developer will be require serve the proposed developit 	2. When water is made availat Facilities Charges for resour If you have any questions, plea RECEIVET	PBR HAWAII

Water for Life . . Ka Wei Ola

PEANWING + LANDACAPE ARCHEFFCTHRE + ENVIRONMENTAL STUDIES - ENTITIEMENTS / PERMITTING + GRAPHIC DESIGN

DANILO "DANNY" AGSALOG SENIOR ADVISOR DEBORAH KIM MORIKAWA DIRECTOR persons with special needs. We actively encourage the development of housing opportunities include a phasing plan or schedule to demonstrate that the housing will be constructed in time An institution the size of the University of Hawaii West Oahu will generate a demand for housing to support faculty and staff, as well as student housing. Oahu's current housing region. The study should indicate the housing demand to be generated by UH West Oahu in market can be characterized by low vacancy rates, high demand, high rental rates, and sales Thank you for the opportunity to review the Draft Environmental Impact Statement prices and target markets (affordable, market, elderly, special needs), and how this mix will prices. The UH West Oahu will have an impact on the local housing market in the EwarKapolei region but this impact is not described in the DEIS. Since the DEIS does not meet the demand for housing generated by UH West Oahu. The housing study should also include a detailed housing study or market study, it is not possible to fully assess what the impacts of UH West Oahu will be on the cost and availability of housing in the area. This We strongly recommend that the DEIS include a detailed housing study that would identify the specific impacts of UH West Oaku on the housing market in the Ewa/Kapolei (single family versus multifamily, number of bedrooms, and tenancy-sales versus rentals), principally concerned with affordable housing for low to moderate income families and program to be undertaken as part of the UH West Oahu project in terms of housing mix for the University of Hawaii West Oahu. The Department of Community Services is terms of quantity and cost. The housing study should more fully describe the housing RECEIVED IBM HAWAII 715 SOUTH XINS STREET, SUITE 311 . HONOLULU, HAIWAI \$6813 • ABEA CODE 803 • PHONE: 522-5311 • FAX: 527-5453 AUG 04 2006 CITY AND COUNTY OF HONOLULU Subject: University of Hawaii West Oahu (UH West Oahu) DEPARTMENT OF COMMUNITY SERVICES Draft Environmental Impact Statement (DEIS) August 2, 2006 represents a material weakness in the DEIS. Mr. Vincent Shigekuni, Principal Honolulu, Hawaii 96813 ASB Tower, Suite 650 for these populations. Dear Mr. Shigekuni: 1001 Bishop Street MUFI HANNEMANN PBR MAYOR Mr. Keith S. Shida SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) Thank you again for your participation in the Environmental Impact Statement process for this If you have any questions regarding this project, please do not hesitate to contact me at 0:VOB13\1322.05\EISVFinal EIS\Response to DEIS Comments\BL-07 BWS response 11-20.doc roads are planned to be dedicated to the BWS where practicable. Water lines within the campus will not be dedicated to the BWS, unless water line easements are established for which will be able to provide the adequate fire flow. A backflow prevention device will <u>UH West O'ahu and/or a private developer will be required to install the necessary water</u> system improvements to serve the proposed development. Water lines within dedicable such purposes. The campus is intended to be metered by several master water meters, project. Your comments will be included in the Final EIS. be required by the BWS at all meter locations. Office of Environmental Quality Control Gene Awakuni/UH West O'ahu

Vires R. Sug

PBR HAWAII

Sincerely, 521-5631.

December 18, 2006

Page 2

Vincent R. Shigekuni

Vice President

ö

Prage 2 Prage 2 Image of the project will be sufficient to meet the need and demand generated by UH West Oahu. The study should also demonstrate how the student housing proposed for the project will be sufficient to meet the need and demand for student housing to be generated by UH West Oahu and should be based on the experience of UH Manoa or other objective and reliable data. w. FRANK IRANDIA TARA TARA TARA TARA TARA TARA TARA TA	December 18, 2006
to meet the demand generated by UH West Oahu. The study should also demonstrate how the student housing proposed for the project will be sufficient to meet the need and demand for student housing to be generated by UH West Oahu and should be based on the experience of UH Manoa or other objective and reliable data. We would also like to see that the housing developed provide a mix of unit types and prices to provide a broad range of housing opportunities for special needs populations such as three deerly and persons with disabilities. These are issues of concern to our department as we strive to facilitate the continued development of affordable and special needs housing on Oahu. Thank you for providing us the opportunity to comment on the University of Hawaii West Oahu Draft Environmental Impact Statement. Please contact Mr. Keith Ishida at 527- 5092 or Mr. Stephen Karel at 523-4690 should you have any questions regarding this matter, stere to be should by and you have any questions regarding this matter, stere to be apported by and you have any questions regarding this matter, stere to be apported by and you have any questions regarding this matter, stere to be apported by and you have any questions regarding this matter, stere to be apported by and you have any questions regarding this matter, stere to be apported by and you have any questions regarding this matter, stere to be apported by and you have any questions regarding this matter, stere to be apported by and you have any questions regarding this matter, stere to be apported by and you have any questions regarding this matter, stere to be apported by and you have any questions regarding this matter, stere to be apported by and at 527- tere to be apported by a stere to be apported by any apported by a stere to be apported b	
We would also like to see that the housing developed provide a mix of unit types and prices to provide a broad range of housing opportunities for special needs populations such as the elderly and persons with disabilities. These are issues of concern to our department as we strive to facilitate the continued development of affordable and special needs housing on oahu. Revent Ansolution as the elderly and persons with disabilities. These are issues of concern to our department as we strive to facilitate the continued development of affordable and special needs housing on Oahu. Revent Ansolution (Ansolution and Ansolution at 527- 5592 or Mr. Stephen Karel at 523-4690 should you have any questions regarding this matter. New Ansolution Ansolution and Ansolution Ansolution and Ansolution and Ansolution and Ansolution and Ansolution and Ansolution and Ansolution Ansolution Ansolution and Ansolution and Ansolution Ansolutio	Ms. Deborah Kim Morikawa, Director City & County of Honolutu Department of Community Services s. wrr118x.Ast.A 715 South King Street, Suite 311 Honolulu, Hawai'i 96813 Usech. Ast.A
Thank you for providing us the opportunity to comment on the University of Hawaii Tonk Schwerk, MCr West Oahu Draft Environmental Impact Statement. Please contact Mr. Keith Ishida at 527- RANNOND THEAA RANNOND THEA	CLICHTORINGLINEL SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT Tre-Predent ENDINENTAL IMPACT STATEMENT (Draft EIS) SHICANINI Dear Ms. Morikawa: MI'RAKAMLART Thank you for your letter dated August 2, 2006. We have reviewed your letter and offer the following responses to your comments:
KEVIN K. MISHIKANA, ASLA	 VEAL, ART V. U.H. West O'ahu shares the Department of Community Services' (DCS) concerns regarding V.T.HIGA, ARLA U.H. West O'ahu shares the Department of Community Services' (DCS) concerns regarding V.T.HIGA, ARLA U.H. West O'ahu shares the Department of Moriable housing opportunities for most housing needs sectors, V.H. West O'ahu shares the Department can understand that UHWO must weigh the need to address VISHIGA, ASLA VISHIGA, ASLA
Sincerely, sincerely, down and a second	The most of the first phase of the UHWO campus. DCS might not be aware that the original land use plan for the non-campus land was primarily focused towards commercial development use plan for the non-campus land was primarily focused towards commercial development (with very filter housing). While commercial development would provide needed employment opportunities in the "Ewa region (and reduce traffic toffrom Downtown HKAABAGO HADONILIU) and provide a source of long-term income to find operations and mainenance and pay down construction debt, UH decided to seek a solution where a private residential development opportunities in the "Ewa region (and reduce traffic toffrom Downtown HKAABAGO HADONILU) and provide a source of long-term income to find operations and mainenance and pay down construction debt, UH decided to seek a solution where a private residential developer would be able to provide a significant amount of immediate as the should be significantly more residential than commercial. In that regard, there should be significantly less impact on the housing inventory in the "Ewa region, than if UHWO implemented is original land use plan. Finally, UHWO is particularly motivated that the project provides workforce housing for faculty and staff as it is difficult to attract the best candidates without it.
	 WOFICK: The following paragraph will be added to Section 6.4 Alternatives Related to Different Environmental District Distributions Which Would Present Different Environmental Impacts of the Environmental I



housing study. As one who is personally involved in the County of Hawait's 1,000± unit Waikoloa Employee Housing project, I have learned that the type of study that DCS is recommending is very time-sensitive and should probably not be undertaken until after this project is farther along in the entilement process (the project still needs rezoning and Plan Review Use approval). As discussed with one member of the DCS staff (upon receipt of your letter), UHWO looks forward to working with DCS in addressing the concerns expressed in your letter. The following paragraph will be added before the last paragraph of Section 4.8.2 *Housing, Anticipated Impacts and Minigation*:

The project will address the provision of affordable housing by coordinating with the Department of Planning and Permitting and the Department of Community Services. UHWO is particularly motivated that the project provides workforce housing for faculty and staff ast is is difficult to aftract the best condidates without it. In addition, UHWO will ensumably include approximately 760 studen housing units with approximately 3 beds each. Thus,

2. We regret to inform you that we disagree with DCS recommendation that the EIS include a detailed

SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (DEIS) December 18, 2006

Ms. Deborah Kim Morikawa

Page 2

Thank you again for your participation in the Environmental Impact Statement process. If you have any questions regarding this project, please do not hesitate to contact me at \$21-5631.

the surge. This approach could at least partially fulfill the need for approximately 2.280 beds for the 7.600-student resident population at little or no cost to the University...to construct "gramy flats," lots would need to meet underlying residential district standards for two (2) dwellings, or a developer could use the Planned Development Housing (PD-H)

opportunity for this type of unit exists where the "alley loaded" low-density residential <u>development on Parcels C and D where such units could take the form of studio units above</u>

affected City depertments, including the Department of Community Services to discuss specific plans to satisfy the affordable housing requirements. During public review period, ende Department of Planning, and Permitting vyrole the polynomy regarding the polential of developing "gramment" the renal autis in proposed housing development: "The DPP considers the ponential benefits (i.e. low cost student renals and increased housing un dordability) available through this type of housing to be substantial. It appears than an

rentals in the open market. The University and/or the private developer will meet with

there will be approximately 2,280 less students that would be competing for affordable

Sincerely,

option.

Mitel & Buig PBR HAWAII

Vincent R. Shigekuni Vice President Office of Environmental Quality Control Gene Awakuni/UH West O'ahu 8

O:VOB13V1322.05/EISVFinal EIS/Responses to DEIS Comments \DCS response 11-20.doc

|--|

PLANNING + FANDSCAPF ARCHITTCTHRF + FNYIROAMENTAL STUDJES + FNYITLEMENTS - PERMETING + GRAPHIC DFOIGN

2/2

Aug-23-2006 D8:28 AM City & County of Honolulu 1234567890

.

. • •

ation AL INFACT STATEMENT (Drait ELS) 8, 2006 first paragraph of Section 4.11.5 <i>Recreational Facilities, Anticipated Impacts and Mitigative</i> <i>surves</i> of the ElS will be revised as follows:	ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Page 3 divided roadway between Kamokila Boulevard and the Wai'awa Interchang Approximately 5.5 acres of land with the property will be used for improvements Approximately 5.5 devices of land with the property will be used for improvements
es of the bits with the recreational facilities in 'Eva will increase as the population grows. the proposed UH West O'ahu development will provide additional recreational facilities compliance with the City's Park Dedication Ordinance. <u>The University and/or the</u> <i>date developer will meet with affected City dataments to discuss specific plans to</i> <u>sity the park dedication requirements, cas regulated by Section 22-7. Revised</u> <i>timances of Honolulu.</i> New recreational facilities within the proposed campus are <i>inved for superimentary by students, faculty, such staff, Such facilities could include</i> <i>incodes the polabellation review of staff, Such facilities could include</i> <i>incodes the polabellation review of staff</i> . Such facilities could include	Intringion Highway. The design of Farmington Highway frontage improvements and access intersections will be coordinated with DDC, and has undergone preliminate review by the DPP Traffic Review and Subdivision Branches. Thank you again for your participation in the Environmental Impact Statement process for this p Your comments will be included in the Final Eff. If you have any questions regarding this project, please do not hesitate to contact me at 521-5631.
ommended, City roadways affected by the proposed development will be analyzed for cy of street lighting illumination per the Illuminating Engineering Society of North a standards for roadway lighting.	PBR HAWAII Vuied R. Buigh
ays of the EIS will be revised as follows: a roadway network in the vicinity of the UH West O'alu property is expected to mge significantly in the future. Phase I of North-South Road and its interchange with H-I Freeway will be constructed by late 2008. Consistent with the ORTP, the H-I eway and Farrington Highway will be widened. The DOT and DTS have been studied in the timing of these projects. It is assumed that the portion of Kapolei	Vincent R. Shigckuni Vice President cc: Office of Environmental Quality Control Gene Awakuni/UH West O'ahu ov008131322.05ElSNFnal ElSNFnal Responses to DEIS CommentsBL-21 DDC response 11
Nevey between Kapole Middle School and the North-South Road will be timed to be pleted by the Depertment of Havalian Home Lands with the completion of the first as of North-South Road. Simularly, it assumed that the segment of Kapolei Parlsouro ugh. Eva Villages will be completed which will allow "Eva residents an alternate (to get ReaddRoosevelt Around) route to Fort Barreite Road. A description of project rovennest at a pletnate state of the matternate for a set ReaddRoosevelt Around) route to Fort Barreite Road. A description of project rovennest are a particular corridors and a lineframe sto when the amficipated rovenness are particular corridors and a lineframe sto when the amficipated rovenness are particular corridors and a lineframe sto when will be requesting and PRU licentions. UH West O'ahu and its private development partner will be requesting and PRU licenting and PRU licentings and PRU licenting and the Kapolei Goff Course ess Road Including additional rereel lights, it varranted), and the construction of "East-West Connector" (Road F a new east-vest road that extends from Farrington linear theorem fort Weaver review in 2007.	

All surrounding roadways controlled by the City, affected by the proposed project, will be analyzed for adequacy of street lighting illumination per Illuminated Engineering Society of North America standards for roadway lighting.

As recommended, the design of Farrington Highway frontage improvements and access intersections will be coordinated with DDC. ŝ

The paragraph describing Fartington Highway in Section 4.10.1 Roadways and Traffic, Anticipated Impacts and Mitigative, Roadways of the EIS will be revised as follows:

Farrington Highway, between the Kapolei Golf Course access road and Fort Weaver Road, is planned to be widened from a 2-lane, undivided roadway to a 4-lane, divided roadway. This road widening would make Farrington Highway a continuous 4-lane,

Mr. Eugene C. Lee SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS)

୍ ତ ସାଧ

roject.

--20.doc

PBR HAWAII 8 ASSOCIATES. INC.	December 18, 2006	ta Ms. Laverne Higa, P.E.	Director and Chief Engineer Department of Facility Maintenance City and County of Honolulu	kapolei, Hawai'i 96707	AUD: MIT. LAUTY LEOPATOL SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT MUCI' ENVIRONMENTAL IMPACT STATEMENT (Draft EIS)	Dear Ms. Higa:	A Thank you for your letter dated August 21, 2006. We have reviewed your letter and offer the following responses to your comments:	The University of Hawai'i's current plans are to dedicate nearly all the roads shown on Figure 2.1	Conceptual Lana Use Plan of the Urfat Eds. Besides the usual concerns about liability, this decision can be attributed to: 1) the Department of Planning and Permitting specifying revisions to the roadway alignments within the 500-acte project and new intersections with bordering roads to provide resident concientivity in 0.0 Mar concernents concentration of 0.0 Mar concernents bade will include concienciate and concentrate the provide residencial concentrates and concentrate bade will include concentrates and concentrates the provide residencial concentrates and concentrate bade will include concentrate and concentrates the provided residencial concentrates and concentrates the provided residencial concentrates and concentrates the provided residencial concentrates the provision of the provided residencial concentrates	commercial uses that will generate property tax revenue to the City and Country of Honolulu (thus, it is felt that the non-campus roads should be allowed to be dedicated to the Country).	Based on your comments, the last two sentences of Section 2.5.1.1 Roadways and Traffic of the EIS will be revised as follows:	Roads within the UH_West O'ath Lands with either be controlled by the University or dedicated to the City. Roads within the Private Development Lands with the private Development Lands	w ur oc aconcorco to the City as actermined during the enithement phase of the projectIt is proposed that all roads (and water, sewer and drainage systems) within the project site will be dedicated to the City and County of Honolulu,	181 except Road "A" and the portion of Road "B" westward of the intersection with Road "D". During the public review period, the City and County of Honolulu Department of Excelity Mentenance worles. "Roads Avivage facilities and their	related infrastructure should be owned and maintained by the State of Howoii for this State facility, similarly to that at the Manoa Campus. The City and County of	Honolulu will not accept the dedication or maintenance of this facility." This will be identified as an unresolved issue since the Department of Planning and Permitting is specifying roads within the 500-acre project to provide regional	connectivity, and the non-campus lands will include residential and commercial uses that will generate property tax revenue to the City and County of Honolulu, it is felt that the non-campus roads should be allowed to be dedicated to the	<u>County.</u>
		W, FRANK BRANDT, FASI (<i>lantmun</i> THOMAASS, WITTEN, AS	President President R. STAN DUNCAN, ASI.A	Executive Vice-Presiden	VINCENT RHKARUN Une-Presiden GRANTI, MURAKAMI, J Principal	TOM SCHNELL, AICP Scrint Associate	RANMOND T.HIGA, ASI, Secure Associate	KEVIN K. NISHIKAWA, A Asarciate	KINI MIKAMI VUTS, LFI Asociate George Alter A buller	Associate Associate Associate SCOLT MURARAMI, ASI	Associate		HONOLULU OFFICE 1880 Diduop Street ASB Jiner, Suite 620	Efmontolo, Lavedi 1983] 3 140-1889] 5.21 5.5.1 141. (888) 5.23 1.01 1.41. (888) 5.23 1.01 E-tualt. sveatmore, phyton.	HILO OFFICE 112 Aupunt Steet 114a Lawan Coder Suite	1146, 11.86, 11.86, 26 1262 Tel: 62081 964-5444 Eace (82081 964-5445	WAILUKU OFFICE 1787 Will Pa Loop, 2000-3 Walltha, Bayan 26291 12 (ef (200) 212 2578	
RT CEIVED AUG 2 8 2005	IAWAII HAWAII	LAVERNE HIGA, P.E. Orector and chef Byghter Geore "rean" myamoto Deputy Drector	IN REPLY REFER TO: ORM OG:346							ct DEIS.	e owned and to that at the t the dedication or		opardi, Chief of		N	leer		
DEPARTMENT OF FACILITY MAINTENANCE CITY AND COUNTY OF HONOLULU 1000 Undels Streat, Suin 215, Karelei, Havail 86707 Phone: (808) 852-6544 - 15-Ex: (808) 692-5657 Weblie: Www.biochdu.gov			August 21, 2006		lekuni, Vice President	e 650 eet	196813-3484	uni:	nvironmental Impact Statement (DEIS) sity of Hawaii West Oahu	ving us the opportunity to comment on the subje	facilities and their related infrastructure should by e State of Hawaii for this State facility, similarly The City and County of Honolulu will not accept	his facility.	any further information, please contact Larry Le oad Maintenance, at 484-7600.	Sincerely,	Farmer Hig	Laverne Higa, P.E. Director and Chief Engir		
		Hannemann Hayor			Mr. Vincent Shiş PBR Hawaii	ASB Tower, Sui 1001 Bishop Str	Honolulu, Hawa	Dear Mr. Shigek	Subject: Draft E Univer	Thank you for g	Roads, drainage maintained by th Manoa Campus.	maintenance of	Should you need the Division of R					

MUFI HANNEMANN MAYOR

PERNNING + FANINCAPE ARCHITECTURE + ENVIRONMENTAL STUDIES + ENTITIEMENTS / PERMITTING + GRAFHIE DESIGN

DEPARTMENT OF PARKS AND RECREATION CITY AND COUNTY OF HONOLULU Kerdet HALE - 1000 LUUDIAN STREET. SUIT 2007 PARADLE, IMMAI 95707 TELEPHONE: (4030 922-5551 - 1747, 1603) 932-511 - 1077 EMBERT. www.abab.gov	ИЛТ НАИНЕМИНИ ИЛТ НАИНЕМИНИ ИЛТ НАИНЕМИНИ ВОССИНИТ ВОССИНИ ВОССИНИ ВОССИНИТ ВОССИНИТ ВОССИНИТ ВОССИНИТ ВОССИНИТ ВОССИНИТ ВОССИНИТ ВОССИНИТ ВОССИНИТ ВОССИНИ ВОССИНИТ ВОССИНИТ ВОССИНИТ ВОССИНИТ ВОССИНИТ ВОССИНИТ ВОССИНИТ ВОССИНИТ ВОССИНИТ ВОССИНИТ ВОССИНИ ВОССИНИ ВОССИНИ ВОССИНИТ ВОССИНИТ ВОССИНИТ ВОССИНИТ ВОССИНИТ ВОССИНИТ ВОССИНИТ ВОССИНИТ ВОССИНИТ ВОССИНИ ВОССИНИТ ВОССИНИТ ВОССИНИТ ВОССИНИТ ВОССИНИТ ВОССИНИТ ВОССИНИТ ВОССИНИ ВОССИНИТ ВОССИНИТ ВОССИНИТ ВОССИНИТ ВОССИНИТ ВОССИНИТ ВОССИНИТ ВОССИНИТ ВОССИНИТ ВОССИНИТ ВОССИНИТ ВОССИНИТ ВОССИНИ ВОССИНИ ВОССИНИ ВОССИНИТ ВОССИНИ ВОССИНИТ ВОССИНИТ ВОССИНИ ВОССИНИ ВОССИНИ ВОССИНИ ВОССИНИ ВОССИН ВОССИНИ ВОССИНИ ВОССИНИ	July 14, 2006	Mr. Vincent Shigekuni, Vice President PBR HAWAII ASB Tower, Suite 650 1001 Bishop Street Honolulu, Hawaii 96813 Dear Mr. Shigekuni:	Subject: Draft Environmental Impact Statement University of Hawaii, West Oahu	Thank you for the opportunity to review and comment on the Draft Environmental Impact Statement relating to the development of the University of Hawaii West Oahu. The Department of Parks and Recreation has no comment.	Should you have any questions, please contact Mr. John Reid, Planner, at 692-5454. Sincerely,	LESTER K. C. CHANG Director	LKCC:mk (162619)	cc: Office of Environmental Quality Control	RECEIVED Jul 1 9 2005 PBR HAWAH
Ms. Lavente Higa, P.E. SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Page 2	The last paragraph of Section 4.10.1 Roadways and Traffic, Anticipated Impacts and Mitigation Measures, Roadways of the EIS will be revised as follows: Roads within the UH West Oahu Lands will be controlled by the University, and roads within the Private Development Lands will be dedicated to the City and County of <u>Homohuke. It is proposed that all roads will be dedicated to the City and County of</u>	the project site will be dedicated to the City and County of Honoluth, except Road "A" and the portion of Road "B" westward of the intersection with Road "D". During the public review period, the City and County of Honolulu Department of Facility Maintenance wrote: "Roads, draindes chcilitiss and their related infrastructure should	be owned and maintained by the State of Howaii for this State facility, similarly to that at the Manoa Campus. The City and Councy of Honolutu will not accept the dedication or maintenance of this facility." This will be identified as an unresolved issue since the Department of Planning and Permitting is specifying roads within the 500-acre project to provide regional connectivity, and the non-campus lands will include residential and commercial uses that will generate property tax revenue to the City and Country of Honolulu, it is felt that the non-campus should be allowed to be dedicated to the Country.	A paragraph will be added to the third to the last paragraph of Section 7.4.2 Unresolved Issues to read as follows:	Dedication of Roads, Drainage Facilities and Their Related Infrastructure. During the public review period, the City and County of Honolulu Department of Facility Maintenance wrote: "Roads, drainage facilities and their related infrastructure should be owned and maintained by the State of Hawaii for this State facility, similarly to that at	the Manoa Campus. The City and County of Honolulu will not accept the dedication or maintenance of this facility." This will be identified as an unresolved issue since the Department of Planuma and Permitting is specifying roads within the 500-acre project to provide regional connectivity and the non-campus lands will include residential and commercial uses that will generate property tax revenue to the City and County of Homohult, it is felt that the non-campus roads should be allowed to be dedicated to the County.	Thank you again for your participation in the Environmental Impact Statement process for this project.	If you have any questions regarding this project, please do not hesitate to contact me at 521-5631. Sincerely,	PBR HAWAII Vuiz D. D. Sunglen- Vincent R. Shigekuni	Vice President cc: Office of Environmental Quality Control Gene Awakuni/UFi West O'ahu o:008130132205UBSFarat EISVErial Responses to DEIS Comments/DFM response 11-20.40c



Senior Associate Sential: Associate

Associate

Associate Associate 1seacratic

lice-President

Principal

Chairman President PLANNING + LANDSCAPF ARCHITECTURE + ENVIRONMENTAL STUDIES + ENTITIEMENTS - FERMITTING + GRAPHIC DESEGN

Mr. Vincent Shigekuni August 21, 2006 Page 2

.

Please note that the Honolulu City Council has decision-making authority over zone changes. Reconing approval will be required for the portion of the project that will be developed privately, and which is not related to the university (the "non-campus lands").

The Final EIS should include a proposed zoning map that shows the zoning districts and corresponding acres for the approximately 500-acre U.H. West Oahu property. An estimate of the number of proposed housing units also should be included.

- 6. The Final Environmental Impact Statement (FEIS) should identify, list and discuss the zoning districts to which both the campus and the non-campus lands are proposed to be rezoned.
- The FEIS should contain figures that clearly depict the areas, and delineate the boundaries, of the project site that are subject to PRU approval and rezoning approval, respectively.
- 8. Section 21-2.120-2 of the Land Use Ordinance (LUO) requires an applicant for a PRU to submit a proposed master plan spanning at least five (5) years. The FEIS should discuss the Five (5)-Year Master Plan for the University of Hawaii West Oahu that will be submitted with the PRU application. This discussion should indeke a description of how the development of the campus lands will be phased, and the timelines in which such sequential development will occur.
- 9. The FEIS should discuss how development of the campus lands will or will not comply with LUO development standards, such as for height, setbacks and density, for the proposed underlying zoning districts in which the university and university-related uses will be located.
- 10. The LUO does not have a parking standard for colleges and universities. The discussion of parking requirements should be expanded to explain how the standard of one (1) stall for every ten (10) students, plus one (1) stall for every 400 square feet of office space, was obtained. The FEIS should list the number of stalls suggested by the institute of Transportation Engineers and discuss how, and why, the proposed 2,850 parking stalls for the campus lands exceeds that standard and it in the number of stalls suggested by the institute of Transportation Engineers and discuss how and why, the proposed 2,850 parking stalls for the campus lands exceeds that standard and the number of recommended parking stalls. There should also be discussion of how the parking to be provided relates to transit use.
- 11. The FEIS should discuss how development of the campus lands will comply with LUO requirements for off-street loading.

Mr. Vincent Shigekuni August 21, 2006 Page 3 Interim Planning Division

Community Action Plans Branch

- On page 57, the DEIS should explain the basis for the statement "If more executive housing were available, business owners buying executive housing in Ewa would likely relocate their businesses to Ewa." This appears to be speculative than supported by the current housing market, which has at least 30 homes in the Ewa area, including Ko Olina, Makakilo, Ewa Beach (Gentry Homes), with prices of \$1 million or more.
- 2. On page 60, Agricultural impacts, please discuss what alternatives are available to the existing lessees to replace lands being used for pasture or crop production and whether the loss of the subject area would have a negative impact on existing farming operations.
- 3. On pages 63 and 64, the DEIS considers Level of Service (LOS) E as acceptable. However, in the footnote, LOS E is described as traffic experiencing very long delays. The DEIS should explain why LOS D cannot be achieved given that the entire Kapolei area is a developing planned community and funding for traffic improvements is, in part, based on participating developers' contributions.
- 4. On page 86, regarding expected improvements to traffic into Honotulu, the DEIS should provide a description of projected improvements at particular corridors and a timeframe as to when the anticipated improvements are to occur.
- 5. The DEIS should expand on the anticipated impacts, short-term and long-term, the West Oahu Campus would have on on-street parking in the local residential neighborhoods, if any. The DEIS should also discuss whether existing or proposed affordable housing unlis would be adequate to support the anticipated demand generated by the West Oahu Campus, in both short- and longer-term decending.

Development Plan and Zone Change Branch

 Initial Development Phase – (Ref. Para 1 of consultant's response letter to the Environmential Impact Statement Preparation Notice (EISPN) – Phasing plan Environmential Impact Statement Preparation Notice (EISPN) – Phasing plan shown in the EISPN has been revised and is included in the DEIS. The consultant's response of June 16 to the EISPN states that the University of Hawaii West Oahu (UHWO) will be constructed in eight (8) phases. The Newert, indicates only two (2) phases; Initial Phase and Build Out Phase. The FEIS should address this discrepancy and clarify the development phasing. The inclusion of an implementation timeline including infrastructure (even if preliminary) would be useful in providing such clarification.

Mr. Vincent Shigekuni August 21, 2006 Page 4

- Dormitory Development (Ref. Para 3 of consultant's response letter) Student housing is planned for Phase 4 (Year 2011), however no details have been provided for the components of Phase 4 since Phase 4 was not mentioned in the DEIS.
- 3. Student Housing Development (Ref. Para 4 of consultant's response letter) Consultant has stated that no "gramy flat" type rental units will be included in proposed housing development. The DPP considers the propertial benefits (i.e. low cost student rentals and increased housing unit affordability) available through this type of housing to be substantial and would like to know why this form of housing unit has been excluded. It appears that an opportunity for this type of unit has been excluded. It appears that an opportunity for this type of unit has been excluded. It appears that an opportunity for this type of unit exists with the "alley loaded" low-density residential development on Parcels C and D where such units could take the form of studio units above the garage. This approach could at least partially fulfil the need for approximately 2.280 beeds for the 7,600-student resident population at little or no cost to the University. Flease note that to construct "granny flats", lois would need to met underlying residential district standards for two (2) dwellings, or a developer could use the Planned Development Housing (PD-H) option.
- 4. Transit Nodes (Ref. Para 6 & 7 of consultant's response letter) The DEIS has given consideration to the two (2) transit nodes indicated in the Ewa Development Plan (DP) and proposes to link the campus to one (1) of these nodes with housing linked to the second. As recontly discussed, consideration should be given to having a transit station on campus.
- Alternatives to Auto Use (Ref. Para 11 of consultant's response letter) The University Village is intended as a transit-oriented development (TOD) near the transit node that will serve the campus. Pedetarin/bike pathways were not completely discussed in the DEIS, but should extend throughout the project and be included in the PRU application support materials.
- Connectivity (Ref. Para 8 of consultant's response letter) The Draft EIS indicates linkage with key roads and developments surrounding the site. An internal ring road appears to provide a suitable level of internal connectivity. Details of pedestrian and bikeway connectivity should be included in the PRU application and in requests for zone charges.
- 7. Scope of Commercial Development (Ref. Para 10 of consultant's response letter) – Given the proximity of the high density residential uses on Parcels A and F (925 units) to the University Village, it is likely that residents may do much of their shopping there and possibly at Mixed Use Parcels A & B instead of driving to the Mixed Use Parcel C on the west side of the campus. This possibility should therefore indude and accommodate easy pedestrian/bike tinkage between these parcels.

Mr. Vincent Shigekuni August 21, 2006 Page 5

Policy Planning Branch

- Page 115, Other Community Facilities. This section should disclose why the former mauka site is no longer the preferred site.
- Page 120, Land Use Map, 1st sentence. We suggest the following revision: "The currently proposed U.H. West Oahu property is designated for Low and Medium Density Residentia!, and High Density Residential uses by the Ewa Development Plan Urban Land Use Map (see Figure 5.1)."
- Page 105, Section 5.3.1, General Plan. Please discuss the potential residential population of the project and the impact and consistency with the General Plan residential population distribution policy.
- Page 137, Section 7.4.2, Unresolved tssues. The DEIS indicates three (3) unresolved issues and lists Funding and Public/Private Partnership. Only two (2) are discussed. Please clarify the number of unresolved issues.
- Page 3, Section 1.6, Surrounding Land Uses. Please provide more detail on the surrounding land uses, including a map, and any potential compatibility issues. Because the development plans of adjacent owners may not yet be final, please indicate their current plans.
- Page 92, Section 5.2.3, State Land Use Law, Chapter 205, Hawaii Revised Statutes. Please elaborate on any substantive conditions of approval associated with the boundary reclassification to Urban. Also, if any amendment is contemplated, this should also be discussed.

Site Development Division

Civil Engineering Branch:

- The documents listed under Section 4.10.4 (Drainage Facilities-Anticipated Impacts and Mitigation Measures) should be listed under Section 11.0 (References).
- Drainage: Should provide/include a reference for the referenced material under "Anticipated Impacts and Miligation Measures" (p. 74).

<u>Subdivision Branch</u>

 Under Permits Required, park decircation should be listed, as required by Section 22-7, Revised Ordinances of Honolulu.

Ē	PB	R HAWAII OCIATFS. INC. December 18, 2006
t DEIS's projected average flow of 1.68 mod is included in the	W. FRANK BRANDU FASLA Chantaun Tatomass. Witten, Asla President	Mr. Henry Eng., FAICP, Director Department of Planning and Permitting City and County of Honolutu 650 South King Street, 7 th Floor Honolutu, Hawai'i 96813
astewater Master Plan for East Kapole), dated June 2006. A Site t Master Application for Sewer Connection is required. This project liable for payment of a Wastewater System Facility Charge.	R. STANDY XXAN, ASIA Exerutive Vicy-President RUSSILL V, CHYWG FASI.A Escentive Vice-President	SUBJECT: UNIVERSITY OF HAWALI WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS)
on Armitage (527-6274) or Matthew Higashida (527-6056) of our questions.	VINCEND'SHIGBAUNI Vac Prvadati GRANFT AIT'RAKAMI, AICP Prinaid	Dear Mr. Eng: Thank you for your letter dated August 21, 2006. We have reviewed your letter and offer the following responses to your comments:
Activity yours,	TOM SCHNELLAICP Sanny Associate Raynord Thinga, Asla	Land Use Permits Division Zoning Reculations and Permits Branch
Henry Eng. #AICP//Director Department/of Planning and Permitting	venine Avorente KEVIN K. NISHIKAWA, ASI A Absorate	 We recognize that the UJH West O'ahu and/or private developer will be subject to pay impact fees, pursuant to Ordinance 02-52. As such, UH West O'ahu and its private
of Hawaii	KIMEARKAMEYULN, LITD'AP Asacinte Asacinte	development partner will be requesting an amendment to Ordinance 02-52 (Section 33A- 1.6 (h) of the Revised Ordinances of Honolully to include widening and improvements of Enclosed the revision of the revision of the review of
DirectorythalsontRequests for CommentstEIStdets itr uthwo.doc	SCOTT ALIKA ABRIGO Jsowiate SCOTT AUBRAKAMI, ASLA	ratinguin ringiway rom two to jour lattes exervent in rort weaver froad and the kapolci Golf Course Access Road (including additional street lighting, if warranted), and the construction of a new east-west road (netween Fort Weaver Road and Farrington Highway) when Ordinance 02-52 undergoes its five-year review in 2007.
	Shinker.	The first paragraph of Section 4.10.1 Roadways and Traffic. Anticipated Impacts and Mitigative Measures, Roadways of the EIS will be revised as follows:
		The roadway network in the vicinity of the UH West O'ahu property is expected to change significantly in the future. Phase 1 of North-South Road and its interchange with the H-1 Freeway will be constructed by late 2008. Consistent
	HONOLULU OFFICE 1001 Biding Street ASB Daves Suite 620	with the ORTP, the H-1 Freeway and Farrington Highway will be widened. The DOT and DTS have been consulted in the timing of these projects. <u>It is assumed</u>
	Horniau, Phona P. 2021 1891 141 (2021 22) 26.31 25 v. (2005 22) 25.1 (102 E-mail: sy-adhrone, phylony ni cum	tuat the portion of Kapolet Parkway between Kapolet Middle School and the North-South Road will be timed to be completed by the Department of Hawaiian Hone Lands with the completion of the first phase of North-South Road.
	HILO OFFICE MILO OFFICE IM Algourd Autor (Contensional Contensional Management (Management (M	Similarly, it assumed that the segment of Kapolei Parkway through. Ewa Villeges will be completed which will allow "Bwa residents an alterinate (to Gitger RondRossevelt Arenuel: Doute to Fort Barrette Road. A description of project improvements at particular corridors and a timetrame as to when the anticipated improvements are scheduled to occur will be included in the zone provention of the context of the second provention of the second prov
	WAILUKU OFFICE TST VBB (13 Janp, 5mr. 1 Nadda, 11 Panet (97-25 J.27 (4 +1804, 212, 2025	clattice and FKU applications. UNI West U and its private development partner will be requesting on annubuent 10 Ordinance 02-52 to include widening and improvements of Farington Highway from Iwo 10 four lanes between Fort Weaver Road and the Kappelei Golf Course Access Road finchuling additional street lighting. If varranisable, and the construction of the "East-West Connector" (Road F, a new east-west road that extends from Fortivern
		Highwav through the UHWO site and beyond, eventually connecting with Fort Weaver Road) when Ordinance 02-52 undergoes its five-year review in 2007.

PLANNING + LANDYCAPE ARCHITICTURE + FAVIRONALMTAL STUDIES + ENTITIENTS - PERMITTING + GRAPHIC DESIGN

Mr. Vincent Shigekuni August 21, 2006 Page 6

-

Wastewater Branch

The project's DEI approved Wastew Development Mas may also be liable ÷

Please contact Nelson / staff if you have any que

HE:cs cc: Governor, State of I

G:\LandUse\PosseWorkingDirect

Mr. Heary Eng, FAICP SUBBECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Drafi EIS) December 18, 2006 Page 2

- 2. We acknowledge that Tax Map Key 9-1-16:129 will be subject to varying streat/road widening setbacks along Fartington Highway. During the Public Review Branch, the DPP Site Development Division Traffic Review Branch wrote: "The Department of Design and Construction is currently working on the roadway cross-section for Fartington Highway fronting the Computery." UHWO has set aside 5.5 acres along Farrington Highway for its widening. We also note that Tax Map Keys (TMKs) 9-1-16:120 and 127 are not subject to setbacks for street/road widening.
- 3. We recognize your confirmation that the portion of the project that consists of a university in AG-1 Restricted Agricultural district (the "campus lands") will need a Plan Review Use (PRU) approval from the Honolulu City Council. The last two paragraphs of Section 5.3.3 Land Use Ordinance of the EIS will be consolidated and revised as follows:.

The UH West-Oahu campus is permitted as a public use and structure and is consistent with the existing zoning classification for the property upon approval of the PRU permit application. However, a classification for the property upon approval of the Auton application. However, a classofic in zoning will be required for development of the nonapplicative classifier of the property. Under the Gip's LUO- the UH West Oahn property is currently zoned-AG-1. Resented Agreentine (see Figure 5.2). Universities and colleges are pornited with a PRU permit Use (PRU) approved-for any zoning district. The University will seek a PRU permit for the compuse and endown housing: however, a zone change will be required for the remaining lands within the property.

Under the City's LUO, the UH West O'ahm property is currently zoned AG-I. Restricted Agriculture (see Figure 5.2). The University will seek a PRU permit for the compus and student housing londs. PRU approval from the Honolutu City Council will only apply to the University and uses directly related to the university. Auditionally, DPP requires that the "Compus londs" be regord to a district other than the currently zoned AG-I, and as such, a zonous londs" will be required to be processed concurrently with the PRU Rezoning approval will olso be needed for the portion of the project that will be developed privately, and which is not related to the university.

- 4. As noted above, the Final EIS will mention that the UH West O'ahu campus lands will be rezoned to a district other than the agricultural district; and will be processed concurrently with a PRU application.
- As noted above, the Final EIS will mention that rezoning approval will be required for the portion of the project that will be developed privately, and which is not related to the university.

The following paragraph will be inserted at the end of Section 5.3.3 Land Use Ordinance of the BIS:

Figure 5.3 is a proposed zoning map that shows the proposed zoning districts and corresponding acress for the approximately 500-acre UHWO property. Table 5.0 identifies, lists and discusses the zoning districts to which both the computs and the noncomputs lands are proposed to be rezoned. With rezoning and PRU approval and at full development, the proposed project will include 4.041 housing units.

A copy of the proposed zoning map is attached.

Mr. Henry Eng, FAICP

SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) Docember 18, 2006 Page 3

 As requested, a new Table 5.1, *Proposed Plan Review Use Permit/Zone Change Land Use* Acreage has been included in the Final EIS which identifies, lists and discusses the zoning districts to which both the campus and the non-campus lands are proposed to be rezoned.

<u> Table 5.1 – Proposed Plan Review Use Permit/Zone Change Land Use Acreage</u>

LAND USE	ACREAGE	UNITS/	RESIDENTIAL	COMMERCIAL
		ACRE	UNITS	SQUARE
		-	(approximate)	FUUIAGE
1 ands Coversed 1 Inder the Plan Review 1 fee P	UT VENUA	CUNAL D		
		10-4 000		
7,000 Stutent Campus Incruang 10- acre detention basin and possible "Lab	c.601	'	•	ł
) SCROOL /				
Student Housing or Campus Expansion Parcel B	12.1	61	230	1
Roads	3.7	ı		•
Subtotal for PRU (Zoned R-5)	119.3		230	•
Business Mixed Use District (BMX-3)				
Mixed Use (Retail, Office, Residential)	15.1	10	151	164,439
Campus Expansion/Multi-Family Housing/Mixed Use	22.2	16	355	•
Student Housing/Mixed Use or	38.0	21	646	331.056
Campus Expansion Parcel A (531				
student housing units, 115 residential units)				
Mixed Use Parcel C	10.2	10	102	111.078
HECO Substation	1.0	1	-	
Roads	8.0	,	•	-
Subtotal for BMX-3 District	94.5	-	1,254	606,573
Total for UH West O'ahu Lands	213.8		1,484	606,573
PRI	VATE DEVELOP	MENT LANDS		
Business Mixed Use Community District (BM	X-3)		-	
Mixed Use Parcel A	10.5	10	105	114,345
Mixed Use Parcel B	11.2	10	112	121,968
Detention Basin	11.2	,		
HECO Substation	1.0	-		3
Roads	3.9	,		
Subtotal for BMX-3 District	37.8	•	217	236,313

Mr. Henry Eng. FAICP SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Page 5	the 500-acre property will be developed in Phase 1, providing 616 residential units. <u>The</u> current estimated date for opening of the campus is Fall 200 <u>9.</u>	Build-Out Phase. Ultimately, the UH West O'ahu Lands will include a 7 600,-student	campus (including a 10-acre detention basin); University Village (student housing/nized use or campus expansion); work-foreofolfordable multi-family housing- a HECO	substation; and roads. The Private Development Lands will include single- and multi- family residential uses: mixed-use areas: a detention bosin: an elementer school, o	HECO substation; and roads. The current estimated year for completion of the project is 2015.	9. The FHS will discuss how development of the cammus lands will or will not commus with 1110	developments standards, such as for height, schedeks, and density. For he monocod underlying	zoning districts in which the university and university-related uses will be located. The	tollowing three paragraphs will be added at the end of Section 5.3.3 Land Use Ordinance:	Durthe the public review period the Department of Planning and Demutrium Lond Iter	Permits Division. Zoning Regulations and Permits Branch wrote: "The FEIS should	<u>discuss how development of the campus lands will or will not comply with UUO</u> <u>development standards, such as for height, setbacks and density. For the proposed</u>	underlying zoning districts in which the university and university-related uses will be	tocated. While the exact "exemptions" and "variances" are not known at this time.	<u>UDIWO A REQUECT, ONN HATAR & KNOCLERES, INE. MAS PROVIDED HIE GIONENTE FREIMMINT</u> ODDIATA ON Ionw Hor CANNAR APOGUN MAY AP MAY WAY AND ANALAMAAN	standards for R-5 coning. Building heights on campus will yory, but will typically exceed	the 25-30 foot maximum height required by R-5 zoning. Buildings in the first phase will be approximately of 1 feet high at the rideo with the communic tensor washing.	approximately 85 feet above finish grade. Setbacks of buildings will meet R-5 zoning	requirements. The buildings will be set back more than 30 feet at the front and more than	15 feet at the side and rear property lines.		The height setbacks will also be met. Any portion of a structure exceeding 15 feet will be	set back from every side and rear buildable area boundary line at least one foot for each two feet of additional height over 15 feet. Any portion of a structure exceeding 20 feet will be set back from the front buildable area boundary line at least one foot for every two feet of additional height over 20 feet.	<u>Density of the site will meet R-5 zoning requirements. The maximum building areas will</u> not exceed 50 percent of the zoning lot.	10. We recognize that the LUO does not have a parking standard for colleges and universities. Section 4.10.1 Roadways and Traffic, Anticipated Impacts and Mitigation Measures, Parking will be revised to read as follows:	The Giy's Land Use Ordinance (LUO) establisher a minimum of one parking stall for	very-ten students, plus one stall for every-fold square foel of office space. Based on this standard: a ratio of one parking stall for every two students was utilized to accommodate the parking neodo of students, faculty, and administrative staff. For the initial 1,520 student-compus, 760 parking stalls (at 0.5 per student head-count) are estimated to be required. The utilizate 7,600 student compute is expected to require stalls (at 0.375 per student head count). The proposed number of parking stalls exceeds the LUO requirement and the number of stalls and solved by the Institute of Transportation Engineers. The University will continue to work with the City during the Plan-Review
IU) DRAFT	COMMERCIAL SQUARE FOOTAGE		P	1	•	1		•				4			r		,	171111	•	-	236,313	842,886	rly depict the and rezoning feveloper has	is subject to	itted with the he Draft EIS le 2.4 which The last two	lows:	struction Iniversity and the Private 5, F, and Hision-of acres of
(UH WEST 0'AI	RESIDENTIAL UNITS (approximate)		530	241	395	248	*	1,414			183	182	e	,	365		561		•	561	2,557	4,041	been propared to clea are subject to PRU the selected private o	l and non-PRU land	st O'ahu will be subm Ordinance (LUO). T e developed, and Tak lands will be phased.	crevised to read as fol	u Lands includes con assin); a portion of U portion of Road D; ? I development of th statential Parcets D. 1, and E; and t he csp , Approximately 110.5
r oʻahu	UNITS/ ACRE		16	12	16	12	•	4		Ļ	¢	6	•	,			10	-	•	1	1		ng Map, has cct site that reement with	een the PRL	, Land Use of the campus line	<i>Action</i> will be	H West O'al e detention l B, and a 2.26). Phass 2.26,
'AI'I WES' VT (Drafi EIS	ACREAGE		33.1	20.1	24.7	20.7	9.9	108.5			c.0c	30.3	12.0	4.3	77.1		56.1		7.0	63.1	286.5	500.3	oposed Zoni of the proj since the ag	ry line betw	aster Plan foi n 21-2.120-2 he first initia celopment of	d I umug of I	tent of the U ding a 5-acr coads A aucr y (see Figura struction of p th Farringto
Mr. Henry Eng. FAICP SUBJECT: UNIVERSITY OF HAW ENVIRONMENTAL IMPACT STATEMER December 18, 2006 Page 4	LAND USE	Apartment Medium Density District (A-2)	Residential Parcel A (High Density, Multi-Family)	Residential Parcel B (Medium Density, Multi Family)	Residential Parcel F (High Density, Multi Family)	Residential Parcel G (Medium Density, Multi Family)	Roads	Subtotal for A-2 District	Beritania District A PA	Decidential District (K-3)	Single Family)	Residential Parcel D (Low Density, Single Family)	Elementary School	Roads	Subtotal for R-5 District	Recidential District (B. 2 5)	Residential Parcel E (Medium-Low	Density, Multi-Family)	Koads	Subtotal for R-3.5 District	Total Private Development Lands	GRAND TOTAL	 As noted above, a new Figure 5.3 P₁ areas, and delineate the boundaries, approval. Please note, however, that 	not been consummated, the bounda change.	 We acknowledge that a Five-Year M PRU application, pursuant to Section included Figure 2.6 which depicts the includes a description of how the device deviced. 	paragraphis of Section 2.1 Fridsing an	Initial Phase. Phase I developm of a 1.520-student compus (inclu Village: a HECO substation; J expansion of Farrington Highwa Development Lands includes com G; an 1.1.2-acre detention basin, improvements at intersections wi

O'AHU (UH WEST O'AHU) DRAFT Mr. Henry Eng. FAICP SUBJECT: UNIVERSITY OF HAWAI'1 WEST ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Page 6

adoquately-meets future-needs- Future factors that may affect parking demand are: 1) plans to provide student housing equivalent to 30 percent of the utlimate full time equivalent student population, and 2) plans for rail and bus transit. Use (PRU) permitting process to ensure that the design of future parking facilities

For the initial 1.520-student campus. 760 parking stalls (at .5 per student head count) are estimated to be required. This requirement is higher than both the Institute for Traffic Engineers (ITE) standard and the ratio utilized for the ultimate 7,600 student campus. This higher parking requirement was utilized to compensate for the lack of initial <u>phase.</u> For the ultimate 7.600 student compus approximately 2.812 parking stalls (at. <u>37 per student headcount) will be provided. This number is equal to the Institute of</u> Traffic Engineers (11E) suggested parking requirement of 2.812 stalls. development of alternative modes of transportation and lack of student housing in the mmediate vicinity of the campus, which would result in higher automobile use during the

Eactors that may impact the estimated parking requirement for the 7.600 student compus include: 1) the level of transit service to the UH West O'ahu compus (less parking required if a significant portion of the UHWO students, faculty and staff use transit). 2) the percentage of distance learning students. 3) the percentage of students enrolled in non-daytime courses, and 4) the amount of nearby and on-compus student housing which is projected to be equal to 30 percent of the total student population. Parking standards could be adjusted, as necessary, to accommodate for any of the factors identified above. HHCTP. The University will continue to work with the City during the Plan Review Use (PRU) permitting process to ensure that the design of future parking facilities adequately For example, parking standards for the ultimate development of the campus could be reduced should transit access to the area be enhanced through the City's proposed meets future needs. We are not able to provide the requested table that would depict parking calculations for the campus lands by use. floor area (square feet), the applicable parking standard and the number of recommended parking stalls. As the campus design is in its very preliminary stage, the floor area is Table 4.4 will also provide the estimated number of parking stalls using methodology provided by the Institute of Traffic Engineers and if one were to compare with the number of stalls provided for a comparable number of students at the UH Mänoa Campus. The following text and Table will be added to the end of Section 4.10.1 Roadways and Traffic, Anticipated Impacts and Mitigation presently not available. However, to address your request, a new table (Table 4.4) showing initial phase student population and the ultimate student population and the proposed number of parking stalls that is being proposed for each phase of campus development will be provided in the Final EIS. Measures, Parking:

and the proposed number of parking stalls that is being proposed for each phase of campus development. Table 4.4 also provides the estimated number of parking stalls using methodology provided by the Institute of Traffa. Engineers and if one were to compare with the number of stalls provided for a comparable number of students at the UH Mānoa Campus. Table 4.4 shows the initial phase student population and the ultimate student population

UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) Henry Eng, FAICP December 18, 2006 SUBJECT: Page 7

Table 4.4 - Proposed UH West O'ahu Parking Stalls

<u>7.600</u> 2.812 2.812 2.318	Control Control <t< th=""><th>Student UHWO ITE UHMON Population Proposed Parking Suggested Parking baread on</th></t<>	Student UHWO ITE UHMON Population Proposed Parking Suggested Parking baread on
--------------------------------	--	---

11. The development of the campus lands will comply with off-street loading requirements, pursuant to Section 21-6.100, LUO. The following text has been added at the end of Section 4.10.1 Roadways and Traffic, Anticipated Impacts and Mitigative Measures, Parking of the EIS:

Development of the UH West O'ahu compus lands will comply with the City's requirements for off-street loading, pursuant to Section 21-6.100, LUO. Development of the UH West O'ahu compus

Interim Planning Division

Community Action Plans Branch

- The statement: "If more executive housing were available, business owners buying executive housing in 'Ewa would likely relocate their businesses to 'Ewa," is taken out of context (the previous sentence read: "One challenge to attracting businesses to 'Ewa has been that most of the housing inventory provided has been affordably- and moderately-priced housing, rather than executive housing...") is intended to illustrate the potential positive impacts of executive housing can play in sitting of businesses. While we concur that there may be 30 homes in the 'Ewa area with prices of \$1 million or more, this statement is based on an observation that the defacto center of employment on O'ahu is the Central Business District and Waikiki and not surprisingly, many business owners live east of the Central Business District. ŀ.
- Regrettably, as portions of the project site are developed, there will be corresponding loss of land available on the project site for existing farming operations. Fortunately for existing lessees there is an adequate amount of lands available on O'ahu and other islands that may be used for pasture or crop production. The first paragraph of Section 4.3 Agricultural Impact, Anticipated Impact and Mitigative Measures of the EIS will be revised as follows: c'i

agricultural production <u>as portions of the project site are developed</u> for the proposed development, which would have a negative impact on the above listed agricultural <u>lessees</u>. This will result in some loss in revenues, jobs, or and payroll <u>generated by</u> <u>agricultural</u> activities. However, the State and City have long planned for new area would be used to accommodate for future development in the region. It is our understanding with the closure of the Del Monte Fresh Produce, there are 3.100 acres of development in the project area, and the property is within the Urban State Land Use District. Tenants have been fully aware, for quite some time, that the proposed project Portions of the currently cultivated lands within the property will be withdrawn from vacant lands in Kunia available for cultivation.

within the discussion pertains to the existing, two-way undivided highway LOS. There is no existing intersection at this location. In this case, LOS is determined as a function of "percent time following" and "average travel speed." Because Farrington Highway in the vicinity of the UH West O'ahu site is mainly a "no passing zone," the "percent time following" is relatively The discussion on pages 63 and 64 focuses on existing traffic conditions. The LOS E reported ŝ

Mr. Henry Eug, FAICP SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Page 9	their own cars. In the longer term, the ultimate 7.600-student computs is expected to include 2.812 parking stalls (at 0.37 per student head count). The proposed number of parking stalls equals the number of stalls suggested by the Institute of Transportation Engineers (at 0.37 per student head count). Since the nearest existing residential methorhoods are located in the Villages of Kapolei and the Villages of Kapolei is	separated from UHWO by the Kapolei Golf Course, without any connecting streets, any students who park in the Villages of Kapolei would have to either wolk, hise or moped onto either Kapolei Parkwovo r Farrington Hishway to access the campus. The closest proposed residential neighborhoods are the private development lands surrounding the campus, and it is hoped that students from these areas will either work or bike to the campus. It is possible that in the long-term future, if there is inadequate parking provided on the UHWO campus, commuting proposed projects (DHHL East Kapolei 1 - which is immediately moded or on surrounding proposed projects (DHHL East Kapolei 1 - which is immediately moded of the 500 acre UHWO vie, and DHH. Fork Knohei 7 and the	Ho'opili project – which are both across the proposed North-South Road from UHWO). We can understand the Interim Planning Division's Community Action Plans Branch's comments whether proposed affordable housing units would be adequate to support the anticipated demand generated by the West O'ahu Campus, but we hope the Branch can understand that UHWO must weigh the need to address affordable housing needs with the goal of faving a private partner fund the construction of the UHWO campus. The Interim Planning Division's Community Action Plans Branch might not be aware that the original land use plan for the non-campus land was	purtainty recussed towards commercial development (virtu very nucle commercial development would provide needed employment opportunities in the 'Bwa region (and reduce traffic toffrom Downtown Honolulu) and provide a source of long-term income to fund operations and maintenance and pay down construction debt, UH decided to seek a solution	where a private residential developer would be able to provide a significant amount of immediate cash to help offset the cost of constructing the first phase of the campus. As a result, the phan now reflects significantly more residential than commercial. In that regard, there should be significantly less impact on the housing inventory in the "Ewa region, than if UHWO implemented its original land use plan. Finally, UHWO is particularly motivated that the project provides workforce housing for faculty and staff as it is difficult to attract the best candidates	when the account to the EIS will be revised to read as follows: Mitigative Measures of the EIS will be revised to read as follows:	Park areas and open space will be provided throughout the property in compliance with the City's Park Dedication Ordinance. The University and/or the private developer will meet with aftered City deportments to discuss specific plans to satisfy the value developer will meet with a complexity and/or the private developer will meet with a complexity equivements. The operation of discuss specific and the private developer will meet the City edjordable housing requirements approximately 355 units a discuss a specific plans. Los satisfy the UNIVERS of the second of the ediored of the compression of the state of the ediored behavior and the compression of the ediored of the ediored behavior and the UNIVERS of the ediored behavior and the the UNIVERS of the ediored behavior and the complexity of the ediored behavior and the complexity of the ediored behavior and the the UNIVERS of the ediored behavior and the the UNIVERS of the ediored behavior and the ediored behavior and the the the UNIVERS of the ediored behavior and the the the the ediored behavior and the the the the ediored behavior and the the the theorem and the theorem and the the theorem and the the theorem and	The project will address the provision of affordable housing by coordinating with the Department of Planning and Permitting and the Department of Community Services. UHWO is particularly mativated that the project provides workforce housing for faculty and staff as it is difficult to attract the best candidates without it. In addition, UHWO will eventually include approximately 760 student housing units with approximately 3	peats each. Inus, there will be approximately 2.280 less students that would be competing for affordable renats in the open market. The University and/or market private developer will meet with affected City departments, including the Department of Community Services to discuss, specific plans to satisfy the affordable housing
Mr. Heiry Eng, FAICP SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Page 8	high, therefore translating into an LOS E condition during the peak hours. Under these conditions, LOS E is considered acceptable. Future traffic conditions will involve an intersection at this location and improvements strive to achieve LOS D at intersections for major movements. The footnote on page 69 will be moved to the appropriate page that discusses future intersection house of a section.	4. The Oaku Metropolitan Planning Organization (OMPO) Oahu Regional Transportation Plan (ORTP) describes fiscally-constrained long-range transportation improvements that are projected to be completed by year 2030. The timing of these projects between now and 2030 are not identified. The Transportation Improvement Program (TIP) lists projects that are programmed for federal Fiscal Years 2006 through 2008. Projects on the TIP in the UH West O'ahu area include:	 North-South Road between Kapolei Parkway and H-1 Freeway; Kapolei Parkway between Renton Road and North-South Road; Fort Barrette Road widening right-of-way (ROW) acquisition; Kapolei Interchange complex between Makakilo and Palalai Interchanges; Freeway Management System Phases I.A and IB; H-1 Freeway Eastbound Widening design between Waiau and Halawa Interchanges; Oahu Intra-Island Ferry Design planning and design. 	Additionally, projects recently completed or atready in implementation include the H-1 Freeway Waimalu Viaduct widening, AM Zipper Lane Extension, and Fort Weaver Road widening between Laufamui and Geiger Road.	Implementation of these projects will help to manage the growth in vehicular travel in the future and, in that way, would result in better traffic conditions than if they are not implemented. Similarly, the partial reoricantation of university trips toward UH West O'thu from UH Manoa would also help to manage the growth in veticular travel and would result in better traffic conditions than if it does not occur.	Therefore, the last paragraph of Section 4.11.6 Public Transit, Anticipated Impacts and Mitigation Measures will be rewritten as follows:	The UH West O'ahu campus will be a higher education facility and major employment center. West O'ahu residents will have the opportunity to enroll in University courses and work near their homes. A s such regific conditions into and within Hombulu are also e spected to improve, as fewer people would commute to the UH Mānoa, other UH Community Sedleges, and workplaces outside of West O'ahn. This opportunity will help to re-orient travel to private Uhrwertities. UH Mānoa, other CH O'ahn to UH West O'chu. As usch, UH Wānoa, and workplaces outside of West O'ahn to UH West O'chu. As usch, UH Wānoa, and workblaces outside of West O'ahn to UH West O'chu. As usch, UH Wānoa, and would help to manage the growth in vehicular travel on routes into and within Honolulu and would result in better traffic	conditions than if it did not occur. 5. As suggested, the anticipated impacts of UHWO on on-street parking will be addressed by adding the following text to the end of Section 4.10.1 Roadways and Traffic, Anticipated Impacts and Mitigation Measures, Parking:	It is possible that the proposed UHWO camput is more likely to have impacts on on- street parking on existing residential meighborhoods nearby. For the initial 1,520 student campus, the project site has adequate land available if each of the students were to take

Mr. Henry Eng. FAICP SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Page 11	 where the "alley loaded" low-density residential development on Parcels C and D where such units could take the form of studio units above the sarrace. This approach could at least particulty fulfill the need for approximately 2.280 beds for the 7.600-student resident population at little or no cost to the University to construct "srawny flats," loss would used to meet underlying residential district standards for Nuo (2) dwellings. or a developer could use the Planned Development Housing (PD-H) option." 4. UH West O'ahu will study the implications of including a transit station on campus. The following will be added to the end of the second paragraph of Section 4.10.1 Roadways and Traffic. Anticipated Impacts and Mitgative Measures, Public Transit of the EIS: During the public review period the Department of Planning and Permiting transit station on campus. The prevention of the number of the Department of the EIS: 	5. The University Village is intended as a transit-oriented development (TOD) near the mauka-most transit node (that will primarily serve the campus). Pedestrian/bike pathways will extend throughout the project and as requested, a discussion of such a network will be included in the PRU application.	6. We appreciate your comments that the Draft EIS reflects linkages with key roads and development surrounding the UHWO site. Subsequent to your Department's letter we have met with DPP staff and have studied various changes to the master plan roads to further increase connectivity. Details of pedestrian and bikeway connectivity will be included in the PRU application and in requests for zone changes. Section 4.10.1 Roadways and Traffic, Anticipated Impacts and Mitigative Measures of the EIS will be revised as follows:	As requested by the Department of Planning and Permitting (DPD) Traffic Review Branch, a comprehensive transportation master plan, which incorporates various modes of travel, including transit, vehicle, bicycle and pedestrian will be developed and included with the zone change and PRU applications to DPP. This plan will be designed to establish and pronote a safe and efficient balance between the various modes of travel such as concrete contracted birvelo and advariant fossilities commission and contract.	The concernent of the second s	Parcels A and F (>22 vunits) to the University Vilage, it is likely that residents may do much of their shopping there and possibly at Mixed Use Parcels A & B firstead of driving to the Mixed Use Parcel C on the west side of the campus. A such, we will include and accompodate easy pedestriam/Nike linkawe herver hive narrels. Details of nodestriam and history connocievity	Will be included in the PRU and zone change applications, and will be mentioned in Section 4.10.1 Roadways and Traffic, Anticipated Impacts and Mitigative Measures of the EIS as stated in response to your previous comment (see above). Policy Plannine Pranch	 As requested, Section 5.3.2 Ewa Development Plan, Other Community Facilities, Discussion of the EIS has been revised to include the following text: 	<u>Discussion: In 1967, the UH Board of Regents initiated efforts to establish a four-year</u> college in Central or West O'ahu. Over the years, the Regents approved varving sites:
Mr. Henry Eng. FAICP SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Page 10	requirements. During the public review period, the Department of Planning and Permitting vvrote the following regarding the potential of developing. "granm flat"-type returd, units in proposed housing development: "The DPP considers the potential benefits (i.e. low cost student renatis and increased housing unit affordshilly) available through this type of housing to be substantiolIt appears than an opportunity for this type of unit stype of housing to be substantiolIt appears than an opportunity for this type of unit stype of housing to be substantionIt appears than an opportunity for this type of unit stype of housing to be substantionIt appears than an opportunity for this intersects C and D where such units could take the form of studio units above the garage. This approvel could at least portially fulfil the need for approximately 2.280 beds for the 7.600-student population at little or no cost to the University construct "granty flats." lost would need to meet underlying, residential district standards for the 7.600-student resident population at little or no cost to the University construct option."	 My apologies for the confusion on phasing. The phasing described in the Draft EIS best describes the phasing of the project. The initial phase of the campus includes the development of a 1,520 student campus by Fall of 2009 and some of the development of the private development lands. The Build-Out Phase includes the remainder of the private development commbtion vear of 2015. The set was remainder of the private development 	of the EIS has been revised as follows: Initial Phase. Phase 1 development of the UH West O'ahu Lands includes construction of a 1.520-student campus (including a 5-acre detention basin); a portion of University Village; a HECO substantion; Roads A and B, and a portion of Road D; and the expansion of Farrington Highway (see Figure 2.6). Phase 1 development of the Private Development Lands includes construction of portions of Residenting Parcels D, E, F, and G; an 11.2-acre detention basin:	portions of Roads C, D, and E; and the expansion of improvaments at intersections with Farrington Highway. Approximately 110.5 acres of the 500-acre property will be developed in Place 1, providing 616 residential units. The current estimated date for opening of the campus is Fall 2009. Build-Out Phase – Ultimonely, the UH West O'chin Londs with include a 7600-and and	compute functioning a Powen-19, where the normal powen in the area of 2000-states of compute functions a Powen-force-despendable <u>multi-family</u> housens, a HECO substation; and or campus expansion); work-force-despendable <u>multi-family</u> housens, a HECO substation; and roads. The Private Development Lands will include single- and multi-family residential uses; mixed-use areas; a detention basin; an elementary school; a HECO substation; and roads.	<i>the current examined year for completion of the project is 2013.</i> 2. As noted above (and will be in the Final EIS) student housing is planned for the Build-Out Phase.	3. We appreciate the Development Phan and Zone Change Branch's encouragement of the potential development of "granny flat"-type rental units within the project. We concur that there may be potential benefits and will revisit the potentials of this use and encourage the private developer to seriously consider such units in their portion of the project. As noted in our resparse above, the second to the last narrarent of <i>Society</i> . <i>A 20 Havino Amilierated Immerse and Anisotics</i> .	Measures of the ElS will be revised to include the following text:	putonum <u>e resonance the potential of developing "granty lial"-oppe rential units in</u> proposed housing development: "The DPP considers the potential benefits (i.e. low con student rentals and increased housing unit affordability) available through this type of housing to be substantial(It appears than an opportunity for this type of unit exists

- As noted above (and will be in the Final EI) сi
- development of "granny fiat"-type rental un potential benefits and will revisit the potenti seriously consider such units in their portion second to the last paragraph of *Section* 4 *Measures* of the EIS will be revised to inclue We appreciate the Development Plan and Z ŝ

Mr. Henry Eng. FAICP SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Page 13	POPULATION	Objective C. Policy 2: Encourage development within the secondary urban center at Kapolei and the Ewa and Central Oahu urban-fringe areas to relieve devolvmental pressures in the remaining urban-fringe and rural areas and the most brane areas to the second	presented in the primary tabon center. readily provided in the primary tabon center.	Objective C. Policy 4: Seek a year 2010 distribution of O'ahu's residential population	Discussion: According to the DPP, the population of the Eva Development Plan Area (Eva DPA), grew from 42,231 in 1920 to 68,218 in 2000, representing an increase of 60.1 percent (DPP 2004). In comparison, the population for the City as a valuele interacted onty 4,8 percent (from 336,531 in 1990 to 376,156 in 2000. Revialmin	development in the 'Eva region continues to increase in population, and the City expects the population of the Eva DPA to reach 141,864 and 164,462 by 2020 and 2025 as the secondary urban center, respectively. The City's population is expected to reach 1.037,250 and 1.078,050 in 2020 and 2025, respectively.	At full build-out, the UH West O'abu development will introduce new residents to the Eva region and will provide approximately 4.041 residential units (including 761 student housing units). Using an estimate of 2.9 persons per household, the 3.280 residential units (not including student housing units) will provide homes to	approximately 9.512 people. This population increase is consistent with stated proternmental policies of directing future growth toward the 'Ewa Plain. Within the property, the proposed elementary school and UH West O tohu compute with provide for the educational needs of students residing within the property and the surrounding area.	4. We have clarified the number of unresolved issues. Section 7.4.2 Unresolved Issues of the EIS has been revised as follows:	In compliance with Section 11-200-17(11), HAR, this chapter describes the unresolved issues issues associated with the UH West O'ahu development. These unresolved issues	primarity deal with future actions and decisions of governmental bodies that cannot be determined at this time. The following sections discuss how such issues will be resolved and what overriding reasons exist for proceeding with the project.	There are three <u>six</u> currently unresolved issues for the proposed project. Each issue is in the process of being resolved by the University of Havai'i or other agencies with jurisdiction. N one of the unresolved issues require acceptance of this EIS prior to parmit	аррчоча н от сонзітисніон.	Funding. The University is currently entertaining a number of funding strategies for the initial 1.520 student compus. These strategies include a development partnership, land sale and/or legislative funding, or a combination of these options. Funding for compus	<u>development beyond the initial phase is currently an unresolved issue. The initial</u> campus for 1,520 students will be developed by the University's development partner.	сон ытиснот ор Ик- соприя Боуона Ик- инна! рраге will оссиг он ап аз поедед Базіз інчонз к а рибне рачиаганір анд <i>юх funding from the State</i> .	Public/Private Partnership. The University and its private development partner are currently working on an agreement for the UH West O'ahu development. <u>The exact</u> acreage allocated to the Private Developer shown in this EIS is approximate and will not be known until the agreement with the private development partner is fundized. In any
Mr. Heary Eng. FAICP SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Page 12	<u>Wai'ava Ridge (1970). Honouliulit'Eva (1973) and in 1993, Kapolei Makai (the current</u> site). In 1996, it was decided that the West O'chu campus site would be moved to 901	acress manker of the H-1 Freeway in Kapolei. This was based on a strategy by the State to entitle approximately 1,300 acress of land in East Kapolei, which would then be sold to private developers. As a result, the Housing and Community Development Corporation	of Hawaii (HCDCH) initiated a master planning effort for 1.300 acres, including the UHWO Kapolei Makai site for mostly residential uses. It was intended that the proceeds	of the lond safes would be used to fund the construction of a UHWO campus site marka of the H-1 Freeway. On March 5, 1999, HCDCH filed a Petition for Land Use District	poundator. A menomento rectassity approximatery 1,500 acres, including the subject 500 acres, from Agricultural use of Urban use the master-planned community known as East Kapolei. By Findings of Fact, Conclusions of Low and Decision and Order filed on September 8, 1999, the State Lond Use Commission, granted this perinton. (This marka comments from the state Low Low Exact Concerns and Decision and Order filed on comments for the state Low Use Commission, stranted this perinton. (This marka	interview of the section of the current case of the one were the part of the p	During the full of 2001, the UH Administration, the Estate of James Campbell, and the Governor explored the possibility of relocating the UH West O'ahu campus from the 991-acre matka site to the City of Kapolei.	Since the marketability of the petition area was non-existent, and a site selection study determined that the current site was best suited for a new UHWO compus. UHWO was allocated 500 acres and most of the remainder of the 1,300 acres which were "urbanized" was transferred to DHHL	On September 13, 2002, the UH Board of Regents approved the 560-acre Kapolei Makai site as the permanent location for the new UH West O ahu compus and requested the transfer of	tue to the Joy dores Hom DJAN. The Board of Land and Natural Resources approved the transfer of title of the 500- acre site to the UH in fee simple.	The University of Hawai'i still retains the 991-acre site mauka of the H-1 Freeway, but has no current plans for it other than being generally viewed as potentially being developed in the future as an expansion area for the UHWO campus.	 As suggested, Section 5.3.2 Eva Development Plan, Public Facility Investment Priorities, Land Use Map, of the EIS has been revised as follows: 	Land Use Map. The currently proposed U.H. West O'aliu property is designated for Low and Medium Density Residential, and Hieh Density Residential uses by the 'Four	Development Plan Urban Land Use Map (see Figure 5.1). A symbol for the UH West O'altu campus is shown matka of the H-1 Freeway, reflecting a previously proposed campus site. Currently, it is believed that the Ewa DP Five-Year Review will show the	UH West O'ahu at the currently proposed site.	Discussion of the potential residential population of the proposed project and the impact and consistency with the General Plan residential population distribution policy will be included after	ue msi paragraph in Section S.3.1 <i>General Plan</i> of the EIS as follows:

Mr. Henry Eng. FAICP SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Page 14

<u>cose, adequate land area will be set aside for a campus than can eventually accommodate up to 7,600 students in the future.</u>

Completion of Archaeological Inventory Survey Report. OHA recommends that the project area undergo an Archaeological Inventory Survey (AIS), as a new study or antiperate to the previous, in which all aspects of applicable 6E administrative rules and revised Rotates are satisfied. The AIS will include a substantial substantiace inventory as much of the surface architecture may have been destroyed during previous commercial agriculture activities. UH West O'ain has contracted a professional archaeological firm and the firm is currently conducting its field survey. Dedication of Roads. Drainage Facilities and Their Related Infrastructure. During the public review period, the City and County of Honolulu. Department of Facility Manitenance wrote: "Reads, drainage facilities and their related infrastructure should be owned and maintained by the State of Honolulu will not accept the dedication or the Manoa Compus. This City and County of Henolulu will not accept the dedication or maintenance of this facility." This will be identified as an nuresolved issue since the Department of Planning and Permitting is specifying roads within the 500-acre project to provide regional connectivity. and the non-cannus lands within the 500-acre project to connection acception is specifying roads within the 500-acre project to permercial uses that will generate property tox revenue to the City and County of Homolutu, it is felt that the non-cannus rough should be allowed to be dedicated to the County. Plan Review Use (PRU) Permit. Pursuant to Section 3.1.60-1.110. the University will usek a PRU Permit for the compus and student housing lands. As required by DPP. the University will submit a PRU Permit application concurrently with the Zone Change application.

<u>Zone Change. The UH West O'ahu property is currently zoned AG-I, Restricted</u> <u>Agriculture. The Department of Planning and Permitting requires that the "campus</u> <u>chands" be required to a district other than the currently with the PRU. Rezoning approval</u> <u>change will also be needed for the protessed concurrently with the PRU. Rezoning approval</u> will also be needed for the portion of the project that will be developed privately, and which is not related to the university.

- 5. As suggested, a map of surrounding land uses (see attached) will be included as Figure 1.7. UHWO is surrounded by the Kapolei Golf Course, Farrington Highway, the proposed North-South Road and DHHL East Kapolei 1 (mostly residential), with no existing adjoining residences. Since the fall of 2005, UHWO has been meeting monthly with DHHL, HCDA and D.R. Horton to ensure roadway connectivity, the sharing of infrastructure development, alternative transit rail alignments and other issues. As such, there are no known potential compatibility issues with surrounding land uses.
- 6. The original intent of the State Land Use District Boundary Amendment was to entitle approximately 1.300 acres of land in East Kapolei, which would then be sold to private devolopers. It was intended that the proceeds of the land sales would be used to find the construction of a UHWO campus site mauka of the H-1 Freeway. Subsequent to the LUC approval of LUC Docket No. A95-728, Hawai't's economy suffered a serious downum, and there was little or no market demand for new residential development. During this period, it was determined that the construction of a new campus muka of the H-1 Freeway would be more expensive than the current site (makai of Farrington Highway). Since the marketability of the petition area was non-existent, and a site selection study determined that the current site (makai of Farrington Highway).

Mr. Henry Eng, FAICP SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Drafi EIS) December 18, 2006 Page 15 suited for a new UHWO campus, UHWO was allocated 500 acres and most of the remainder of the 1,300 acres which were "urbanized" was granted to DHHL. The current owners (UHWO, DHHL, DOT, DLMR) then, are significantly different from those originally envisioned (private residential developers). As recommended, a discussion of the proposed uses of the UH West O'ahu lands and their potential impacts within the context of the applicable conditions is attached and will be included in Section 5.2.3 of the Final EIS. In addition, the following text will be added to the paragraph in Section 5.2.3:

Chapter 205, HRS establishes the State Land Use Commission (LUC) and classifies all lands in the State into jour districts: Urban, Rural, Agricultural, and Conservation. lands in the UH West Ordin property is located within the Urban District and the proposed land uses are permitted (see Figure 1.6). During the public review period, the Land Luc Commission vorter. "...the Land Luse Connection (LUC) Unbanized the proposed land uses are permitted (see Figure 1.6). During the public review period, the Land Luce Commission vorter. "...the Land Luse Commission (LUC) urbanized the proper site under LUC Docker No. A99-128/ficussing and Community Development Corporation of Hawaii. State of Hawaii (HCDCH). The LUC imposed 27 conditions upon the reclassification pursuent to Findings of Fact. Conclusions of Luw, and Decision auton the reclassification pursuent to Findings of Fact. Conclusions of Luw, and Decision and Order. Decision and Order) dated September 8, 1999. Given that the conditions ru with the load on are building upon the HCDCH and each and evelopment of the University of Hawaii. Yest Oohu lands Repeated Development Lands is subject to Unsective of Hawaii. As such, we request that the conditions. As such uses on these conditions. As such, we request that the conditions.

The original intent of the State Land Use District Boundary Amendment was to entitle approximately 1,300 acres of land in East Kapolei, which would then be sold to private developers. It was intended that the proceeds of the land safes would be used to fund the construction of a UHYOO campus site proceeds of the H-1 Freeway. Subsequent to the LUC approval of a UHYOO campus site and safes would be used to fund the construction of a UHYOO scape. So the neuka of the H-1 Freeway. Subsequent to the LUC approval of LUC Docket No. A99-728. Hawai'ts economy sufficiend a various downturn, and there was little or no marker demand for new residential development. During this period, it was determined that the construction of a new computs marked of the H-1 Since the morket helm the current site (makin of Farrington Highway). Since the morket beince expensive than the current site (makin of Garrington Highway). Since the morket bein of the petition area was non-existent, and a site selection study determined that the current site was best suited for a new UHWO campas. UHWO was allocated 500 acres and most of the remainder of the I,300 acres which were "unbaated" was granted to DHHL. The subsequent owners (IUHWO. DHH, DOT DLMR, and eventually DOE) then, are significantly different from those originally envisioned furd retricted deresh. UH West O'ahu intends to request an amendment to the Decision and Order within a few months.

Site Development Division

Civil Engineering Branch

 Unfortunately we were unable to find all of the references mentioned on pages 74 and 75 of the Draft EIS. The first two sentences of Section 4.10.4 *Drainage Facilities, Anticipated Impacts* and Mirigation Measures will be revised to read as follows:

Mr. Henry Eng. FAICP SUBJECT: UNIVERSITY OF ENVIRONMENTAL IMPACT STA December 18, 2006 Page 16	r HAWAI'I WEST O'AHU (U (TEMENT (Drafi EIS)	H WEST O'AHU) DRAFT	Mr. Henry Eng. FAICP SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Drafi EIS) December 18, 2006 Page 17
Design guidelines conu evaluate the proposed d Harneti- end-the-East K civil engineering consu professional chil envine	ained in the City's Storm Drainag rainage facilities. The Brosion and, a polei Drainage Master Plan vores , <u>diant. Enstineering Concepts, Inc.</u>	te Standards were used to Sediment-Control-Guide-for t ued as references <u>UHWO's</u> <u>used commonity accepted</u> stative and monosed rundf	<u>Wastewater Branch</u> 1. We appreciate the information provided and will revise Scction 4.10.3 Wastewater Facilities, by adding the following text to the end of the last paragraph in this section:
characteristics. 2. As suggested, Section 11.0 R	leferences, will be revised to include	the following reference:	During the DEIS Public Review Period, the Department of Planning, Site Development Division, Wastewater Branch wrote that the "projected average flow of 1.68 mgd is included in the approved Wastewater Master Plan for East Kapolei, dated June 2006, A
<u>City and County of F</u> <u>Standards. Honolulu</u>	<u> Ionolulu Department of Public W</u> . Hawai'i.	lorks. (1986) Storm Drainage	<u>Site Development Master Application for Sewer Connection is required. This project</u> may also be liable for payment of a Wastewater System Facility Charge."
<u>Subdivision Branch</u>			Thank you again for your participation in the Environmental Impact Statement process for this project. Your comments will be included in the Final EIS.
 As suggested, "Park Dedicat Honolulu (ROH), will be incl 	tion Requirement." pursuant to Secti luded in a revised Table 5.2 - Requir	on 22-7, Revised Ordinances of ed Approvals and Permits.	If you have any questions regarding this project, please do not hesitate to contact me at 521-5631.
Ta	ble 5.2 – Required Approvals and Pr	ernuks	purcerery. PBR HAWAII
PERMIT OR APPROVAL	AUTHORITY	STATUS	
Environmental Impaet Statement (EIS) in compliance with Chapter 343, 1485	Governor, State of Hawai'i		Vuriet R. Durgh-
<u>Motion to Amend Decision and</u> Order	Land Use Commission	Motion to be filed at the time the FEIS is submitted.	vincent K. Sngekuni Vice President
Plan Review Use (PRU) Permit	Department of Planning and Permitting / City Council	Application to be submitted after the acceptance of the FEIS concurrently with the Zone Charse amplication	encl: Figure 1.7 – Surrounding Land Uses Figure 5.3 – Proposed Zoning Map
Zone Change	Department of Planning and Permitting / City Council	Application to be submitted after the acceptance of the FEIS concurrently with the PRU Permit application.	CC: OLICE OI EXVIROMENTAL QUALITY CONTROL Gene Awakuni/UH West O'ahu O-UOB1M1322.05HEISVFiral EISResponse to DEIS CommentSUPP Response 11dos
Subdivision Approval	Department of Planning and Permitting	Application to be submitted after the Zone Change application is approved.	
Park Dedication	<u>Department of Planning and</u> <u>Permitting</u>	<u>Application to be submitted</u> with an application for Subdivision Approval.	
Building/Grading Permits	Department of Planning and Permittine	Application to be filed after the Zone Change application	

Currently coordinating with HECO, Will be approved prior to occupancy. Application to be submitted prior to Grading Permits.

State Public Utilities Commission

Installation of Power Lines and Substations

National Pollutant Discharge Elimination System (NPDES)

Permit

State Department of Health

PERMIT OR APPROVAL Environmental Anpact Statement (EBS) in compliance with Chapter 343, URS

Department of Planning and Permitting

is approved.





AUG-22-2006 TUE 05:49 PM PLANNING & PERMITTING	G FAK NO, 808 527 6743 P. 01	AUG-22-2006 TUE 05:49 PM PLANNING & PERMITTING FAX NO. 808 527 6743
DEPARTMENT OF PLA CITY AND COUN 55 2001 Nins STRET 791 75.001 Nins Stret 791 75.001 Nins Stret 701 75.001 Nins Stret 701 Nins Stret	ANNING AND PERMITTING 1TY OF HONOLULU PROGREMOND, HAMI SANS MAJARY INTERNET, WARDADALON MAJARY INTERNET, WARDADALON	Mr. Vincent Stigekuni August 22, 2006 Pege 2
NIVIANA RUM	HENRY ENG, FALOP MARTINA DAMATINA DAMADA DAM	 roadway. This should be used to generally determine and define the hiterarchy streets, as it relates to a typical "grid" pattern of roadways. The nadway sectio follow the City's Subdivision Stutes and Regulations to the greatest extent practi TLRR should also determine the need and estimated length of left and right turn large addreg all the mejor roadways. The Department of Design and Construction is currently working on the roadway section the condways.
Mr. Vincent Shigekunl Vice President PBR Hawai 1001 Bishop Street ASB Tower, Suite 650 Honotulu, Hawaii 96813 Dear Mr. Shigekuni:	Post-It* Fax Note 7671 Date 3/22/66 Dages 2 To VINCe From 0.00	should incorporate various design elements, such as a separated bloyde path, Campus and chanelized exulting janes and pedeshan islands at all major inte. Plasse contact Mel Hirayama of our staff at 523-4119 if you have any questions regard above comments. Very truty yours, Hamy Eng, FALCP Director Department of Haming and Permittue
Re: Draft Environm University of He <u>Tax Map Kevs</u> The following are additional comments from the	nental Impact Statement lawaii West Dahu - Ewa <u>19-1-16:120, 127, 129</u> he Traffic Review Branch of the Department of	HE.pl cc: The Honorable Linda Lingle GuandussPossWentor of the State of Hawali GuandussPossWenthoptweon/Netsonheavest kin Commentatistadet for univer dees
Planning and Permitting relating to the Draft Er above-referenced project. Site Development Division	invironmental Impact Statement (DEIS) for the	
Traffic Review Branch:		
 A comprehensive transportation master travel, including transit, vehitcle, bicycle processing the scan change and the Pit designed to establish and promote a sai travel modes, such as grade separated centrally located transit stops and termit as necessary. 	Is plan, which incorporates various modes of a and pedestrian should be developed prior to lan Review Use (PRU). The plan should be ate and efficient balance between the various a bloycle and pedestrian facilities, convenient and inals, traffic calming devices and other methods,	

•

Jurisdiction over roadways should be discussed in the FEIS. Those that would be under the University should be identified. This would be similar to the distinction between Dole Street, that is under the jurisdiction of the City, and East-West Road, which is controlled. by the University at the Manoa Campus.

c,i

•

ei .

.

•

A roadway master plan, incorporating "smart growth" and "connectivity" concepts, should be developed in concert with the Department of Planning and Permitting and the Department of Transportation Services prior to processing the zone change and the PRU. The roadway master plan should be based on a Traffic Impact Analysis Report (TIAR) which establishes projected traffic volumes along each major segment of

.

.

P. 02

- of ons should lical. The 1 auxiliary
- ry cross-f Hawaii within the rsections.

ing the

AW/ NC
CIMTES. 1
PBF

December 18, 2006

W.FRANK BRANDT.FASLA Charman

THOMASS, WITTEN, ASI A Prevident R.STAN INVICAN, ASI A FACENTIFF President

Exemptor Vice-President RUSELLY,LCHUNG, IASIA Exemptor Vice-President

VINCENTSHIGERUM Ver President GRANT T. MP. RAKAMI, AICP Principal TOM SCHNELL, AICP Senter Ason line

RAYAIOND THIGA, ASI, A Semine Ason min KEVIN K. NISHIKAWA, ASI A Asocate kimi mikami ya itya lefid'ap

Lasochute

SCOTY ALJKA ABRIGO Associate SCOTT ANT A STAND AS

SCIPTUMURAKAMI, ASI.A Associate RONORULU OFFICE AN Bishop Street ASI Fours, Suite 650 Roundin, Basafi 9881, 1481 Ter enso 521, 520 E-rable systellinge phylosista and

IILO OFFICE IN Append Science fould IN III.0.12, and Certer, fould IN III.0.18450195501262 (eff 12808) 966-2159 [act (488) 966-2159

WAILUKU OFFICE 1787 Wile Loop, Suite 5 Walloku, Bawar 96794-1271 Jef-19883 242, 2578

Mr. Henry Eng, FAICP, Director Department of Planning and Permitting City and County of Honolulu 650 South King Street, 7th Filoor Honolulu, Hawai'i 96813 SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS)

Dear Mr. Eng:

Thank you for your letter dated August 22, 2006. We have reviewed your letter and offer the following responses to your comments:

 As recommended, a comprehensive transportation master plan will be developed prior to processing the zone change and the Plan Review Use (PRU). A new paragraph will be added to Section 4.10.1 Roadways and Traffic, Anticipated Impacts and Mitigative Measures of the EIS as follows:

As requested by the Department of Planning and Permitting (DPP) Traffic Review Branch, a comprehensive transit, vehicle, biocvele and pedestrian will be developed and included with the zone change and PRU applications to DPP. This plan will be designed to establish and promote a safe and efficient halance between the various modes of travel, such as prade segmented bickele and deficient halance between the various and scientally located transit, stops and terminals, traffic calming devices and other methods, as necessary.

As recommended, jurisdiction over roadways will be discussed in the Final EIS.

The second to the last paragraph of Section 4.10.1 Roadways and Traffic, Anticipated Impacts and Mitigative, Roadways of the Draft EIS will be revised as follows:

Reads within the UH West Oalus Lands will be controlled by the University, and readis within the Private Development Lands will be dedicated to the City and County of Homolulus—It is propored that all roads (and water, sever and drainage systems) within the proviet site will be dedicated to the City and County of Homolulu, except Road "A" and the portion of Road "B" westword of the intersection with Road "D". During the public review Network of the intersection with Road "D". Proving the public review period, the City and County of Homolulu Department of Proving the public review period. The City and County of Homolulu Department of infrastructure should be owned and maintained by the State of Homolulu benearing infrastructure should be owned and maintained by the State of Homolulu infrastructure should be owned and maintained by the State of Homolulu infrastructure should be owned and maintained by the State of Homolulu and netic review of the Department of Planning and Penniting is specifying coads within the 500-acre project to provide regional connectivity, and the noncompute states ince the City and County of Homolulu, it is felt that the noncompute states ince the City and County of Homolulu. PIANNING + LANDSCAFF ANCHITFCTHRE + FANIRONMENTAL STHDIGS + FNTITLIMENTS - FERMITTING + GRAFFIC DISIGN

Mr. Henry Bng SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Drafi EIS) December 18, 2006 Page 2

Section 2.5.1.1 Roadways and Traffic of the Draft EIS will be revised as follows:

Honolulu will not accept the dedication or maintenance of this facility." This will be identified as an unresolved issue since the Department of Planning and Permitting is specifying roads within the Solvene project to provide regional connectivity, and the non-comput lands will include residenticat and commercial uses that will generate property to revenue to the City and County of Honolulu, it is felt that the non-computs roads should be è feet. Seven other internal roadways are planned within the property, with right-of-way widths varying from 66 feet to 104 feet. Roads-within the UH West-Oahu Lands will either be <u>Lands will be dedicated to the City as determined during the emitlement phase of the project.</u> will be dedicated to the City and County of Honolulu. except Road "A" and the portion of Road "B" weatward of the intersection with Road "D". During the public review period, the City and County of Honolulu Department of Facility Maintenonce wrote: "Roads, drainage facilities, and Lonuty of Honolulu Department of Pacific Maintenonce and State of facilities, and maintenier of New County of Maintenonce wrotes." The only existing public roadway providing access to property is Farrington Highway. During the public review period, the Department of Planning and Permitting, Traffic Review improvements to Farrington Highway. A loop road will provide the major access to various access throughout the campus. The campus loop road will have a right-of-way width of 56 controlled by the University or dedicated to the City. Roads within the Private Development it is proposed that all roads (and water, sewer and drainage systems) within the project site Hawaii for this State facility, similarly to that at the Manoa Campus. The City and County of Branch wrote that the Department of Design and Construction is currently working on the roadway cross-section for the widening of Farrington Highway fronting the Campus property. Approximately 5.5 acres of land within the property has been set aside areas of the campus. A loop road encircling the major University buildings will proallowed to be dedicated to the County.

- 3. The UHWO and its consultants have been meeting with various City and County of Honolulu agencies including DPP and the Department of Transportation Services (DTS) in developing a roadway master plan that incorporates "smart growth" and "connectivity" concepts. We acknowledge that a roadway master plan is being developed in concert with the DPP and the DPT prior to processing the zone change and PRU. A copy of the requested roadway master plan reflecting coordination with your Department and others will be submitted with the applications for a zone change and PRU. The zone change and PRU. A copy of the requested roadway master plan reflecting coordination with your Department and others will be submitted with the applications for a zone change and PRU. The zone change and PRU applications will include proposed roadways excitons that follow the City's Subdivision Rules and Regulations to the greatest extent practical. UHWO's tarffic engineering constituation with DPP and other appropriate plan dright una maxiliary planes along projecting constituation with the submitted length of greaters and regulations to a transfer plan will define the hierarchy of streets as it relates to a typical "grid" pattern of coadways. The zone change and Regulations to the greatest extent practical. UHWO's tarffic engineering constituate with DPP and other appropriate City agencies prior to submitting applications for a zone change and PRU.
- 4. We acknowledge that the Department of Design and Construction is currently working on the roadway cross-section for Farrington Highway fronting the Campus property. The second to the last paragraph of Section 2.4.2 UH West Oahu Lands of the EIS has been revised as follows:

Farrington Highway Expansion. Farrington Highway is a major arterial roadway that provides regional and sub-regional mobility. The highway runs in an east-west direction adjacent on the UH West O'anu property. During the public review period. the Department of Planning and Permitting. Traffic Review Brench worde that the Department of Design and Construction. is currently working on the roadway cross-section for the widening of Farriston Highway fronting the Campary. Approximately 4.4 acres within the northern portion of the property will be used for improvements to Farrington Highway.

Mr. Henry Eng SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Page 3 As recommended, the University will incorporate various design elements within the Campus. The sixth paragraph of Section 4.10.1 Roadways and Traffic, Anticipated Impacts and Mitigative, Roadways of the EIS will be revised as follows.

As recommended by the Department of Planning and Permitting Traffic Review Branch (during the public review period). UHWO will incorporate various design elements, such as separated bicycle paths within the Campus and channelized auxiliary lanes and pedestrian islands at major intersections. A hierarchy of roadwars, bikeways, and pedestrian paths will be designed within the property. A loop, road ways and a pedestrian and ho of the compus. A hierarchy of roadways and a pedestrian and ho will be provided united within the property. A loop, road ways and a pikeway-system will be provided united within the property. A loop, road ways and a pedestrian and hikeway-system will be provided unitin the property. A loop, road with the radio United and bikeway system will be provided united within the property. A loop, road ways and a pedestrian and bikeway system will be provided united within the property. A loop, road ways comprising a total of 31.3 acres are planned within the property (see Figure <u>4.2</u>.4.1) and are described below. Thank you again for your participation in the Environmental Impact Statement process for this project. Your comments will be included in the Final EIS.

If you have any questions regarding this project, please do not hesitate to contact me at 521-5631.

Sincerely,

PBR HAWAII

Vire & R. Su

Vincent R. Shigekuni

Vice President

Enclosure: Proposed Roadway Sections

cc Office of Environmental Quality Control Gene Awakuni/UH West O'ahu O:UOB13V1322.05/EISVFnail EISResponse to DEIS Conuncuis UBL-22 DPP TRB response 11-20.dos





DEPARTMENT OF TRANSPORTATION SERVICES CITY AND COUNTY OF HONOLULU 680 SOUTH KING STREET, 2RD FLOOR Phone. (2013) 523-4239-FEAR (2013) 523-4730-fear (2013) 523-4730-fear (2013)

RECEIVED

MUFI HANNEMANN MAYOR



MELVIN N. KAKU DRECTOR

--- 3 0 2006 Plan הואיייהיי,

TP7/06-162181R

Mr. Vincent Shigekuni, Vice President PBR HAWAII ASB Tower, Suite 650 1001 Bishop Street Honolulu, Hawaii 96813

Dear Mr. Shigekuni:

Subject: University of Hawaii West Oahu

Thank you for the July 3, 2006 letter from PBR HAWAII, requesting our review of and comments on the draft environmental impact statement for the subject project. We have the following comments regarding the document:

- The location of Road F shown in Figure 2.1 is better than previously proposed.
- 2. The proposed elementary school site is discussed on Page 14. The school should be situated where potential pedestrian conflicts with vehicles will be minimized. We would prefer pathways and/or sidewalks that do not traverse major roadways. Planning for school access should be done early so that an access plan that is agreeable to all parties can be formulated.
- 3. On Page 16, the first sentence in Section **2.5.2.1 Roadways** should be revised to state that the City Department of Design and Construction (DDC), not the City DTS, plans to widen Farrington Highway.
- On Page 64, the third paragraph in the discussion on <u>Roadways</u> should be corrected to state that the HOV lanes on the H-1 Preeway are planned for extension.

Mr. Vincent Shigekuni, Vice President Page 2 August 28, 2006

- 5. Pages 64 and 65 discuss the anticipated impacts of the project on <u>Roadways</u>. The design of the project should ensure that TheHandi-Van vehicles have access to the project buildings. Presently, the tallest vehicles is 127 inches high. In addition to facilitating TheHandi-Van vehicle is 127 inches high. In addition to facilitating TheHandi-Van vehicle circulation, mobility features should be integrated into the design of public spaces on campus and among the private development land areas to promote accessibility for persons with disabilities. The "blank slate" of the project site affords the University a unique opportunity to take a proactive approach to developing an accessible community.
- 6. On Page 66, the first sentence in the second paragraph should be revised to state that three of the four possible alignments in the Kapolei/Ewa section of the fixed guideway alternative would be located immediately adjacent to the proposed project. The following sentence should reflect this change and should state that the three possible alignments would each include <u>two</u> stations adjacent to the
- 7. Section 4.11.6 Public Transit (Pages 85-86) includes a discussion of the Honolulu High-Capacity Transit Corridor Project. As related in Comment 6, three of the four alignment options in Section 1 -Kapolei/Ewa will be adjacent to the project site; one on Farrington Highway and the other two on North-South Road. All three options include two stations adjacent to the project site.
- 8. The discussion on Pages 86 and 87 is titled **Anticipated Impacts and Mitigation Measures** on **Public Transit**. This section should discuss the impacts of the project on the public transportation systems, rather than presenting public transportation systems as mitigation for traffic impacts.
- Appendix H is the traffic study prepared for the project. The following comments relate to this study:
- Some of the project roads shown in Figure 7 (Page 18) are mislabeled.

Mr. Vincent Shigekuni, Vice President August 28, 2006 Page 3

- roadway system in East Kapolei. This is not consistent with its Figure 8 (Page 20) shows Road "E" as part of the background description on Page 12. o.
- Figure 9 (Page 21) also shows Road "E" as described in Comment 9b above. പ
- consistent with the Phase 1 roadway system described on Page 37. Figures 17 (Page 39), 18 (Page 40) and 19 (Page 41) are not ч.

Should you have any questions regarding these comments, please contact Ms. Faith Miyamoto of the Transportation Planning Division at 527-6976.

MELVIN N. KAKU Sincerely, Director

Office of Environmental Quality Control Ms. Genevieve Salmonson, Directbr ö



W. FRANKIRAND'E FASLA Channau

President

VINCENTSHIGERUNI

Principal

Senior A succute

Asumiate

basedate

SCOTT MA'RAKAMII, ASI A Asociate

WAILUKU OPFICE 1747 Wild J. Leep, Suite J Waduka, Hawari 96293-1251 Tef-18001 242-2558

December 18, 2006

TROMASS, WRITCN, ASLA

R. STAN DUNCAN, ASLA Executive Vice-President

RUSSIAL Y.J. CHUNG, FASIA Exerctive Vice-President

GRANT'E MURAKAMI, AICP Une-President

TOM SCHWILL, AICP Senior Associate RAYMOND T. HIGA, ASLA

SCOLT ALAKA ABRIGO

aver strong weeks ASII Tawer, Suite 630 Henedulti, Hawari 9681,3–3184 Tel (1898) 523–5613 Taw (1898) 524,1102 F. attail, Sycadiumey, phythas ad coun

HILO OFFICE 281 Aupuri Steed 1810 Jageon Curte, Suite 310 1810, Jhavafi 967 20 1202 Tel: (3039-961-971) Tav. (3038-961-952)

Department of Transportation Services City and County of Honolulu 650 South King Street, 3rd Floor Mr. Melvin N. Kaku, Director

Honolulu, Hawai'i 96813

KEVIN K. NISHJKAWA, ASLA

KIMI MIKANI VITEN, LEDTAP Ausocinty.

HONOLULU OFFICE

requirements.

UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) SUBJECT:

Dear Mr. Kaku:

Thank you for your letter dated August 28, 2006. We have reviewed your letter and offer the following responses to your comments:

- 1. We acknowledge that the location of Road F, shown in Figure 2.1. is better than locations illustrated in previous proposals.
- We strongly concur that the proposed elementary school should be situated where potential pedestrian conflicts with vehicles will be minimized. The description of the elementary school site in Section 2.4.3 *Private Development Lands* of the EIS will be revised as follows: à

Elementary School. A 12-acre elementary school site will be located adjacent to low-density residential parcels and near the DHHL residential subdivision. The school will help to address the denand for public educational facilities in the region and will be set behind the commercial area to buffer it from North-South Road, while and will be set behind the commercial area to buffer it from North-South Road, while making it directly accessible to the communy. The school is within walking or biking distance of residences and accessible via an internal pedestrian path. During the public review period. the Department of Transportation Services (DTS) vorte: "The school should be situated where potential pedestrian conflicts with vehicles will be minimized. We would prefer pathwark and/or sidevalls that do not traverse major roadwarys. Planning for school access should be done early so that an access plan that is agreeable to all parties can be formulated." The Draft ElS did not mentiorlemphastice that the stitug of the elementary school was driven by noo stituge of the school at the makei boundary is constrained by an air installation compatible use zone AICUZ) emanating from the former Barbers Point Naval Air Station (MSI). While there are efforts to remove the AICUZ (aud efforts by some parties to retain the AICUZ), with the issue of the AICUZ is resolved, the stitug of the school is not final. UHWO has coordinated with DOE on the siting of the elementary important factors: 1) it was intended that the elementary school stite was sited as close as possible to the makai boundary of the UHWO site in order to better serve the conflicts. UHWO has met with City and County of Honohulu agencies on a number of occasions, including at least one with DTS present, and to date, there had been no concerns raised about the location of the elementary school site. It is possible that a "Lab School" may be developed on the campus lands in order to address DOE of the elementary school in terms of the potential for pedestrian and vehicular school as currently shown, and assumes that DOE was comfortable with the location adjoining DHHL East Kapolei 1 residential development immediately makai:

PLANAING - LANDACAPI ARCUITECTIIRE - ENVIRONMENTAL ATUDIES - ENTITLEMENTS - FERMITTING - GRAPHIC DESIGN

Ms. Melvin N. Kaku SUBJECT: UNIVERSITY OF HAWAI'I WEST OʻAHU (UH WEST OʻAHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Drafi EIS) December 18, 2006 Page 3	6. As recommended, the second paragraph of Section 4.10.1 Roadways, Anticipated Impacts and Mitigative Measures, Public Transit of the EIS has been revised as follows:	At this time, two litree of the four possible alignments in the Kapoleti'Ewa section of the HHCTCP Fixed Guideway Atternative would be located immediately adjacent to the proposed UH West O'ahu, either on Farrington Highway or on North-South Road. Both All littee alignments include one two stations at <u>adjacent</u> to the UH-West O'ahu compus <u>proposed Proposed Proposed Functions</u> and adjacent to the UH West O'ahu compus <u>proposed Proposed Pr</u>	preferred option includes an adjament that travels along Farington Highway and down the North-South Road, next to the UH West O'ahu property. I finit adjament is selected, the DTS has indicated that the winitive of the Uhiverview reverses, the remote consider consider consider consider	elevated and located within the median of the North-South Road right-of-way. The DTS has also indicated that two transit stops may be considered for the UH West O'ahu property, one located near the intersection of Farrington Highway and the North-South Road, and the other in the southern portion of the property, near the southern access from the North-South Road.	7. As recommended, the third to the last paragraph of Section 4.11.6 Public Transit, Existing Conditions of the EIS has been revised as follows:	At this time, Areo three of the four possible alignments in the Kapoleit'Ewa section of the HHCTCP Fixed Guideway Atternative would be focated immediately adjacent to the proposed IHW West O'har sinher softwaring Historyman Secon Month Each Dack All	proposed project site. <u>Proposed project site</u> .	8. We appreciate your suggested revisions to Section 4.11.6 Public Transit, Anticipated Impacts and Mitigative Measures of the EIS. As a result, the first two paragraphs have been revised as follows:	As a result of the UH West O'ahu development, traffic on local, collector, and regional transportation systems will increase as students, faculty, and staff commute to the proposed transportation systems will be more all to have properly is bounded to the north by frartington tilghway and to the east by the proposed North-South Road, near planned transit nodes and corridors. As such, the commus is being planned to be transit-ready and the use of abilit transportation systems will be encouraged. With current plans to include UH Matton abilit transportation systems will be encouraged. With current plans to include UH Matton abilit transportation systems will be encouraged. With arrent plans to include UH Matton abilit transportation systems and the end of each radi transit alignment alternative. The development of the UH West Of alm compusitie attend to provide greater indership numbers, than if the site was developed as originally planned (mostly lower-density residential). Ridership is not addition, traffic constitutions in the project vicinity are expected to improve with the future completion of Kapolei Parkway and Construction of North-South Road, which will connect the areas between the Kapolei Parkway and construction of North-South Road, which will connect the areas between the Kapolei Parkway and construction of North-South Road, which will connect the areas between the Kapolei Parkway and construction of North-South Road, which will connect the areas between the Kapolei Parkway and construction of North-South Road, which will connect the areas between the Kapolei Parkway and construction of North-South Road, which will connect the incourage walking and bicycling within the area. This will help to reduce vehicular traffic in the vicinity of the compus. Ridership nouse of the provine riders to carry their biocycles with them.
Ms. Melvin N. Kaku SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Page 2	The following new paragraph will be inserted after the first two paragraphs in Section 4.11.1 Educational Facilities, Anticipated Impacts and Mitigative Measures of the Final EIS:	During the public review period, the DTS wrote that the proposed elementary school should be situated where potential pedestrian conflicts with vehicles will be minimized. DTS stated a preference that pathways and/or sidewalks do not traverse major roadways. As such, UHWO will continue to coordinate with DOE, DTS, DPP and other appropriate agencies so that an agreeable access plan can be formulated.	3. As recommended, the first sentence in Section 2.5.2.1 Roadways of the EIS has been revised as follows:	The <u>City and County of Honolulu Department of Design and Construction (PFSDC)</u> plans to widen Farrington Highway to two lanes in each direction with a right-of-way width of 100 Jeet between the Kapolei Golf Course and Fort Weaver Road, taking land along the UH West O'ahu property frontage.	4. As recommended, the third paragraph in Section 4.10.1 Roadways and Traffic, Anticipated Impacts and Mitigative Measures, Roadways of the EIS has been revised as follows:	<u>HOV lanes on</u> File H-1 Freeway is <u>are</u> planned to be extended from the Wai'awa Interchange to the Makakilo Interchange, according to the Oahu Regional Transportation Plan (2025).	5. We recognize that the design of applicable roadways and drop-off/pick-up areas should ensure TheHandi-Van vehicle accessibility. In addition, we concur that the design of public spaces on campus and among the private development hand areas be integrated to promote accessibility for	persons with disabilities. As such, the following new paragraph will be added at the end of Section 4.10.1 Roadways and Traffic, Anticipated Impacts and Mitigative Measures, Roadways of the EIS:	During the public review period the Department of Transportation Services wrote. "The design of the project should ensure that TheHandi-Van vehicles have access to the project buildings. Presenthy, the allest vehicle, is 127 includes high. In addition: to facilitating TheHandi-Xan vehicle circulation, mobility feature should be integrated into the design of public, spaces, on campus, and annums, the private development land areas to promote accessibility for perrons with disabilities" As TheHandi-Van vehicle site provides para-transit service for campus, and all campus the private development land areas to promote accessibility for perrons with disabilities" As TheHandi-Van vehicle accessibility and ADA accession of and annus, the campus with disabilities" As TheHandi-Van vehicle accessibility and ADA and all campus parking areas will include wheelchair accessibility and ADA accessibility to project buildings.Section 5.1 Federal, Americans with disabilities Act of the BIS will be revised as follows:The Americans with Disabilities Act of the BIS will be revised as follows:Section 5.1 Federal, Americans with disabilities. To the extent required by requisitors is struct by individuals with disabilities. To the extent required by requisitors is struct by federal agencies, these guidelines covered by the ADA. It should be noted that the is relatively flat and thus in auronally conducive to accessibility to buildings and garcies, these guidelines covered by the ADA. It should be noted that the relatively flat and thus is naturally conducive to accessibility. The design of buildings and continent of the provide the provident of the the the context of the the context of the the context of the the that the is relatively flat and this is naturally conducive to accessibility to public space and annus the private development land areas will be integrated to promote accessibility for persons with disabilities.

HONOLULU FIRE DEPARTMENT CITY AND COUNTY OF HONOLULU 586 500111 STREET + HOMOLULU, HAMMI 2613 TELËHONE: (2001 723-7139 • FAX: (2003 723-711 (• 1117ERUET; WMMAAAAAAA	ИПГТАЛИРЕМАНИ КОНТЕРИСТИИ КОНТЕРИСТИИ КОНТЕРИСТ 5 ЦИА		OPALITY FIRE CHART	July 18, 2006	Mr. Vincent Shigekuni, Vice President PBR Hawaii Suite 660	oute obu, American bavings bank tower 1001 Bishop Street Monolulu, Hawaii 96813	Dear Mr. Shigekuni:	Subject: Draft Environmental Impact Statement University of Hawaii West Oahu Ewa, Oahu, Hawaii Tax Map Keys: 9-1-016: 120, 127, and 129	In response to your letter of July 3, 2006, regarding the above-mentioned subject, the Honolulu Fire Department (HFD) reviewed the material provided and requires that the following be complied with:	 Provide a fire apparatus access road for every facility, building, or portion of a building hereafter constructed or moved into or within the jurisdiction when any portion of the facility or any portion of an exterior wall of the first story of the building is located more than 150 feet (45 720 mm) from a fire apparatus access road as measured by an approved route around the exterior of the building or facility. (1997 Uniform Fire Code, Section 902.2.1.) 	 Provide a water supply, approved by the county, capable of supplying the required fire flow for fire protection to all premises upon which facilities or buildings, or portions thereof, are hereafter constructed or moved into or within the county. 	On-site fire hydrants and mains capable of supplying the required fire flow shall be provided when any portion of the facility or building is in excess of the 150 feet (45 720 mm) from a water supply on a fire apparatus access road, as measured by an approved route around the exterior of the facility or building. (1997 Uniform Fire Code, Section 903.2, as annended.)	JUL 2.1. 2006 JR HAVVALI	
Ms. Melvin N. Kaku SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Page 4	addressing your comments will be included in the FEIS. The revised TIAR has also been sent to you under a separate cover letter.	This letter will be included in the Final EIS under Section 12.0.	Thank you again for your participation in the Environmental Impact Statement process. If you have any questions regarding this project, please do not hesitate to contact me at \$21-5631.	Sincerely,	PBR HAWAII Vuiz-8 R. Suger-	Vincent R. Shigekuni Vice President	cc: Office of Environmental Quality Control Gene Awakuni/UH West O'ahu	O.VOB13\1322.05\EISUFinal EISUFinal Responses to DEIS Comments\BL-27 DTS response 11-20.doc						

	W. FRANK HEANDJ, FAMAA (<i>Jaatahum</i>)	Mr. Kenneth G. Silva, Fire Chief
¹ D for review and approval.	THOMAS 5. WITTEN. ASLA President	Totaonuu rue Department City and County of Honolulu 636 Sonth Kines Street
ss is:	R. STAN DUNCAN, AM. A Executive Vice-President	Honolutu, Hawai'i 96813-5007
e Department	RUSSELL, Y.J. CHUNG, FASLA Executive Vice-President	SUBJECT: UNIVERSITY OF HAWAL'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMEVTAL IMPACT STATTEMENT (IL FIS)
iwaii 96813-5007	VINCENT SHIGERUMI Vie: President	Doar Mr. Silva:
Battalion Chief Lloyd Rogers of our Fire Prevention	GRANTT, AU'KAKAAII, AICP J'ruripad	Thank you for your letter dated July 18, 2006. We have reviewed your letter and offer the following response to your comments:
Sincerely,	'TOM SCHERL, AICP Sentor Associate	I. Fire apparatus access roads shall be designed and constructed in accordance with the
all o al	RAYMOND T. HIGA, ASILA Venne: Associate	Uniform Fire Code, Section 902.2.1.
KENNETH G. SILVA	KEVIN K. MSHBKAWA, ASEA Avocade	 Water infrastructure shall be designed and installed in accordance with the Uniform Fire Code, Section 903.2, as amended.
Fire Chief	KIMI MIKAMI YETSAJI D'AP Associate	As noted in the Draft EIS, civil drawings will be submitted to your Department for your review and annoval.
	SCOTT ALIKA ABRIGO	Contributed of the second s
ŭ 1. Construct	SCOIT SR'RAAMB, ASLA Vermee	section 4.10.2 water suppy radiines, Anticipated Impacts and Mitigation Measures, Fire Protection of the EIS will be revised as follows:
	1200-100 S. C. T.	In compliance with Honolulu Fire Department (HFD) requirements, a water
		system in which an appurenences, ityarani spacing, and fire flow requirements meet the BWS standards will be provided. <u>Water infrastructure shall be designed</u>
		<u>and installed in accordance with the Uniform Fire Code. Section 903.2. as</u> amended. Fire hydrants will be spaced throughout the campus within 150 feet of
		all sides of unsprinklered buildings and 150 feet of the face of sprinklered
	HONDIALU OFFICE INT Bidep Steer	buildings. All multi-story buildings <u>are</u> assumed to be sprinklered.
	ACB from the fraction fraction Rearbidies [Jacosci 9983] 3 [81] Jef (8083 523 3 613) Jacosci 8583 553 3 10 Jaco (and fraction fraction fraction fraction fraction fraction	Section 4.1.1.3 Fire Protection, Anticipated Impacts and Mitigation Measures of the EIS will be revised as follows:
	ī	Urban structures and landscaping for the UH West O'ahu will eliminate the
	HILO OFFICE 301 Amount Street	potential fire hazard posed by the existing scrub vegetation. Although all
	Hile Lapon Center, salte JD	ouncarress with the equipped with modern fire control devices and access for fire
	1144-1144-1144-1144-1144-1144-1144-114	apperates, water suppry, and valuary, construction way conjorm to existing codes and standards, an occasional and unavoidable increase in demand for fire
	A DAY I HAVE JAMAN TANK I	protection services is likely to result. According to the Ewa Development Plan, to
	WAILUKU OFFICE 1782 Wilt Part Controls	meet the projected population and economic growth in 'Ewa by 2020, three fire stations are planned at 'Ewa Villages, Ko Olina, and Makañwa Hills, but service
	Wathbar, Hawari 36794 1271 164 (888) 247 2328	dates have not been determined. The UH West O'ahu would be serviced by the
		proposed 'Ewa Villages Fire Station in Tenney Village. As the population of
		Eva grows and the planned fire stations are estavitshed, the UH West U ahu development will be adequately protected from the unavoidable occurrence of
		fire and as such, no significant impacts to fire protection facilities or services are
		expected to result from the development.

PBR HAWAII & ASSOCIATES, INC.

December 18, 2006

PLANNING + LANDSCAFL ARCHITECTHRE + ENVIRONMENTAL STUDIES + INTELLEMENTS - FLAMETTING + GRAPHE DESIGN

_1

Mr. Vincent Shigekuni, Vice President Page 2 July 18, 2006

.

3. Submit civil drawings to the HF

In addition, please note that our new address

Honolulu Fire 636 South Stre Honolulu, Hav

Should you have any questions, please call Bureau at 723-7151.

KGS/SK:bh

cc: The Honorable Linda Lingle, Governor c/o Office of Environmental Quality

.

.

.

Kenneth G. Silva JECT UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT TRONMENTAL IMPACT STATEMENT (Draft EIS)		
ber 18, 2006		RECEIVED
	August 22, 2006	AUG 2 4 2006
required by the ATV.), a private water system in which all appurtenomices, hydrant spacing, Fire flow requirements meet BWS standards will be provided. <u>Water infrastructure shall be</u> <u>fined and invatiled in accordance with the Uniform Fire Code, Section 903.2, as antended</u> . A department access road within 150 feet of the first floor of the most remote structure will also	Ð	PBR HAWAII
rovided. Fire apparatus access roads shall be designed and constructed in accordance with Uniform Fire Code. Section 902.2.1. This access will have a minimum vertical clearance of eet and 6 inches, be constructed of an all-weather driving surface that complies with the City comdrate. In a constant of current of a university of the city of the city of the city	Mr. Vincent Shigekuni Vice President	
communds, be exprove of supporting the manuan 00,000-points weight of the apparatus, be of a gradient not exceeding 20 percent. The unobstructed with of the access road will a City requirements, and all dead-and fire apparatus access roads in excess of 150 feet in th will be provided with an approved turnaround having a radius complying with DTS	A BT Tower - Suite 650 1001 Bishop Street Honolulu, HI 96813	
idards. Fire hydrants will be spaced throughout the campus within 150 feet of all sides of prinklered buildings and 150 feet of the face of sprinklered buildings. All multi-story dings <u>are</u> assumed to be sprinklered. In addition, civil drawines will be submitted to the	Dear Mr. Shigekuni:	
D for review and approval. you again for your participation in the Environmental Impact Slatement process for this project.	Re: University of Hawaii West Oahu (UHWO) Draft Environmental Impact Statement	
omments will be included in the Final EIS. ave any questions regarding this project, please do not hesitate to contact me at 521-5631.	Thank you for the opportunity to comment on the above-referenced Company, Inc. (HECO) has no objections at this time.	roject. Hawaiian Electric
у,	To serve the UHWO development and surrounding electrical loads, bringing power to the substations, and power lines extending c installation of the power times and substations will require the arr	there will be power lines of from the substations.
iners R. Shielen	Utilities Commission, as well as other applicable state and city approves Should HECO have existing facilities/easements on the subject proper	ls and permits. v. we will need continued
R. Shigekuni sidem	access for maintenance purposes. In addition, we reserve the oppo on the protection of existing power lines and electric power facilities the project.	unity to further comment at may be affected by the
Office of Environmental Quality Control Gene Awakuni/UH West O'ahu 0:V0B13N32205E(SFinal EISFinal Responses to DEIS Comments/BL-05 HFD 11-20 response.doc	As the project develops and construction plans are finalized, ple informed so that we may be better able to evaluate any effects on point of contact for this project, and the originator of these comm Project Manager, Engineering Department (543-7194). I suggest d continued PECO's continuity invest in this project.	se continue to keep us ur system facilities. Cur nts, is Kathy Yonamine, aling directly with her to
	Sincerely,	·
	Kirk S. Tomita Senior Environmental Sci	ntist
	cc: Ms. Genevieve K.Y. Salmonson (OEQC) K. Yonamine	
	. y WINNIER C	HHE EDISON AWARD

R HAWAII OCIATES. INC. December 18, 2006			Mr. Kirk S. Tomita SUBJECT: UNIVERSITY OF ENVIRONMENTAL IMPACT STA December 18, 2006 Page 2	² HAWAI'I WEST O'AHU (U VTEMENT (Drafi EIS)	H WEST O'AHU) DRAF
Mr. Kirk S. Tomita, Senior Envi Hawaiian Electric Company, Inc	ronmental Scientist		Section 1.1 Project Profile,	Permits/Approvals Required of the E	S will be revised as follows:
P.O. Box 2750 Honolulu, Hawai'i 96840-0001			Permits/Approvals Reautred	Compliance with Chapter 31 Statute (11DC)	3, Hawaii Revised
Attn: Ms. Kathy Yonamine, Pro	iject Manager			Motion to Amend the Decision Plan Review Use (PRU) Per	<u>m and Order</u> nit
SUBJECT: UNIVERSITY ENVIRONME	OF HAWAI'I WEST O'AB NTAL IMPACT STATEME	HU (UH WEST O'AHU) DRAFT 'NT (Draft EIS)		Zone Change Subdivision Approval Part Dedication	
Dear Mr. Tomíta:				Building/Grading Permits Public Utilities Commission	a annraval of substations o
Thank you for your letter dated following responses to your com	l August 22, 2006. We have ments:	reviewed your letter and offer the		powerfines National Pollutant Discharg Permit	e Elimination System (NPDE
 We acknowledge that a approval of the State Pu 	the installation of power line ablic Utilities Commission, as	es and substations will require the s well as other annlicable state and	Table 5.2 Required Approva	<i>ls and Permits</i> of the EIS will be revi	sed as follows:
city approvals and perm revised as follows:	uits. The last page of the Exer	cutive Summary of the EIS will be	T	<u>able 5.2 - Required Approvals and P</u>	<u>rmits</u>
,	Required Approvals and Perm	nits	PERMIT OR APPROVAL Environmental-Impact Statement	AUTHORITY	STATUS
PERMIT OR APPROVAL	AUTHORITY	STATUS	(EUS) in compliance with Chapter 343, HRS	Governor, State of Hawai'i	
Ervertonmenten-impace Statement (ElS) in compliance-with Chapter	G overnor, State of Hawai' i		Motion to Amend Decision and Order	Land Use Commission	Motion to be filed at the tim the FEIS is submitted.
945, HRS Motion to Amend Decision and Order	Land Use Commission	Motion to be filed at the time the FEIS is submitted.	Plan Review Use (PRU) Permit	Department of Planning and Permitting / City Council	after the acceptance of the <i>FEIS concurrently with the</i> <i>Zone Change ambivation</i>
Plan Review Use (PRU) Permit	Department of Planning and Permitting / City Council	Application to be submitted after the acceptance of the FES concurrently with the Zone Change application.	Zone Change	Department of Planning and Permitting / City Council	Application to be submitted after the acceptance of the FEIS concurrently with the DDII Dominication
Zone Change	Department of Planning and Permitting / City Council	Application to be submitted after the acceptance of the FEIS concurrently with the PRII Permit andication	Subdivision Approval	Department of Planning and Permitting	LANCE FRAME approximation. Application to be submitted after the Zone Change application is approved.
Subdivísion Approval	Department of Planning and Permitting	<u>Anot in apparenton.</u> <u>Appendication to be submitted</u> <u>after the Zone Change</u> amplication is amorened	Park Dedication	<u>Department of Planning and</u> Permitting	Application to be submitted with an application for Subdivision Approval.
Park Dedication	Department of Planning and Permitting	Application to be submitted with an application for Subdivision A menoul	Building/Grading Permits	Department of Planning and Permitting	Application to be filed after the Zone Change application is approved.
Building/Grading Permits	Department of Planning and Permitting	<u>Application to by Stata</u> <u>Application to be filed after the</u> <u>Zone Change application is</u> approved.	Installation of Power Lines and Substations	<u>State Public Utilities Commission</u>	Currently coordinating with HECO. Will be approved prior to accupancy.
Installation of Power Lines and Substations	State Public Utilities Commission	Currently coordinating with HECO. Will be approved wior to occurancy.	National Pollutant Discharge Etimination System (NPDES) Permit	State Department of Health	Application to be submitted prior to Grading Permits.
National Pollutant Discharge Elimination System (NPDES) Permit	State Department of Health	Application to be submitted prior to Grading Permits.			

PLANNING + LANDICAFL ARCHITECTHRE + FMURONMENTAL STUDIES + ENTITLEMENTS - FERMETTING + GRAPHIC DESIGN

W.FRANK DRANDT, EASLA Charman

THOMASS, WITTEN, ASLA President

R. STAN DUNCAN, ASLA F vendive Vice-President RUSSILLY, CHUNG, IAM.A Executive Vice-President

VINCENYSHIGERUNI Vier President

GRANT TAIPRAKAMI, AICP Principul TOM SCHNELL, AICP Script Associate

RAYMOND T. HIGA, ASLA Sentine Avois inte

KIMI MIKAMI VETN, LED'AP Åssociate KEVIN K. NISIUKAWA, ASLA Associate

SCOTT ALIKA ABRIGO Åswrånde SCOTT MURAKAMI, ASLA Assoriate

HONOLUEU OFFICE AND Holpsy Viete Ca AND Thore Suite Ca Lennedan, Llawai Wist V 195 Lenned Ca Ca Ca Ca La v (SNS 723 - 1912 La v (SNS 723 - 1912 La mail' sysadining pletawai com

HILO OFFICE DIL Augunt Street DILL Agent Criter, Suite 310 DILS, L1Aa 21 967,0, 1562 (del (2021 964, 1313 Jaze (2020 964, 1953)

WAILUKU OFFICE 1787 Wah ?a Loop, Suite 1 Wadidat, (1886/1795/93-127) Def (5081-212-2578

Mr. Kirk S. Tomita SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Page 3		AUG 2 MM COMPANNA LUC PBR H	awali
The first paragraph of Section 4.10.6 Etectrical Facilities, Anticipated Impacts and Mitigative Measures of the EIS will be revised as follows:		August 22, 2006	
Phase I of the development is expected to generate an electrical demand of 18 megavolt amperes (MVA) for conventional development (without Energy Star appliances and other energy-saving design measures). At full build-out, the estimated electrical demand is energy-saving design measures). At full build-out, the estimated electrical demand is estimated to be 55 MVA. HECO calculations indicate that there is insufficient concerts) from the existing Eva Nut, Kamokila, and Fort Weave rules into serve the projected load. To accommodate a portion of this increase in load, HECO is propositing to construct non new system distribution substations on the UH West O'ahu property, Land for a 1-acre substation site will be stated near the conthern boundary of the property. To serve the UHNO development and surrounding electrical loads, there will be prover fines binging	Mr. Vicent Shigt Vice President PBR Hawaii ASB Tower, Sui 1001 Bishop Stre Honolulu, HI 96 Dear Mr. Shigek	ekuni te 650 set 3813 uni	
power to the substations, and power lines extending out from the substations. HECO has yet to determine whether the power lines to the substations will be overhead or		<u>University of Hawaii West Oahu</u>	
<u>unaerground. UHWO's preference is for the power lines to be installed</u> underground, subject of PUC approval. The installation of power lines and substations will require approval by the State Public Unities Commission, as well as other applicable state and cirv approvals and permits.	Thank you for th (DEIS) for the U	e opportunity to comment on the Draft Environmental Impact Statement niversity of Hawaii West Oahu (UHWO).	
 As the project develops and construction plans are finalized, UHWO, the private developer and/or its consultants will continue to keep HECO informed so that HECO may be better able to evaluate any effects on its system ficalities. 	We continue to b supportive of its	elieve that the UHWO will be a positive effect on the community and are development.	
Thank you again for your participation in the Environmental Impact Statement process for this project.	There are two co our contribution	nditions in our agreement with the State of Hawaii which was made as par of 200 acres for the UHWO that should be considered.	t of
If you have any questions regarding this project, please do not hesitate to contact Grant Murakami or myself at 521-5631. Sincerely,	Condition I:	UH must designate at least 200 acres of the mauka property ("University Parcel") for use solely for University Purposes by the earlier to occur of December 31, 2005, or the date of any conveyance of any portion of the	
РВК НАМАП	-	mauka property to any third party.	
Wie-BR. Right-		This condition was met.	
U Vincent R. Shigekuni Vice President	Condition II:	The State shall have secured funding, entered into contracts and commence the construction of all of the administration, classroom and related support	q
cc: Office of Environmental Quality Control Gene Awakuni/UH West O'ahu		actuities necessary to accommodate an enrollment of 2.750 students "Campus"), provided that such commencement date shall be extended to Occember 31, 2011, if the State Legislature appropriates funds for the	
O:UOB13/1322.05EESUFinal EISUFinal Responses to DEIS Conneents/BL-24 HECO response 11-20.doc	Η	planning and design of the University of Hawaii-West Oahu Campus on th University Parcel by December 31, 2006.	4
		The Legislature did fund planning and design. The deadline for constructi s through December 31, 2011.	ц
	1001 Kamokila Boulevard, I	^c apolei, Hawaii 96707 Phone: (808) 674-6674 Fax: (808) 674-3111 Website: www.jamescampi	ell.com

Ma 172-2004 (No. 10-10-10-10)	PBF ABF	R HAWAII Siates, inc.
Mr. vicent Singekuni August 22, 2006 Page 2		December 18, 2006
	W. FRANK HRANDT, FASLA	
If construction of the Campus is not commenced by December 31, 2006, or by December 31, 2011, as appropriate, then the Stare, at its election shall:	Chaerwoor THOMASS, WITTEN, ASLA Psetdiout	Mr. David W. Rae, Vice President Public Affairs James Carmoholt Commany, 1.1,C
Convey to Campbell Estate a single, agreed upon parcel of land that is located makai of	R. STAN DUNCAN, ASI, A Executive Vice-President	1001 Kamokila Boulevard Kapolei, Hawai'i 96707
H-1, zoned for agriculture with urban potential, and that has value equal to the 1994 value of the 200 acres of land conveyed to the State in 1994;	RUSSELLY, L. CHUNG, FASLA Executive Vice-President	SUBJECT: UNIVERSITY OF HAWAIT WEST O'AHU (UH WEST O'AHU) DRAFT
Or	VINCENTSHIGEKUNI Vice-President	ENVIRONMENTAL IMFACT STATEMENT (Draft EIS) Deat Mr. Rae:
Pay the Estate the current fair market value of the 200 acres of land conveyed to the State in 1994.	GRANT LAU RAKAMI, AICP Principud TOM SCHNELL, AICP Scular Associate	Thank you for your letter dated August 22, 2006. We have reviewed your letter and offer the following responses to your comments:
Тћапќ уои.	RAVAIOND T. HIGA, ASLA Scutior Avox tale	 Thank you very much for the expression of support of the proposed project. We strongly concur that the UH West O'ahu campus will have a positive effect on the community.
Sincerely,	KEVIN K. NISHIKAWA, ASLA Avortato	 We acknowledge the two conditions set forth in your agreement with the State of Hawai'i as part of your generous contribution of 200 across for LHH West O'shu
David W. Rae View Douvil Artice Artice	KIMI MIKAMI YUPS, LITD'AP Aswelate SCOTT ALIKA ABRIGO Aswe fate	 We concur that the University has met the first of two conditions of the agreement between the Estate of James Campbell and the State of Hawai'i by designating at least 200 acres of the mauka property for use solely for University Purposes.
	scort MrkakaMl, ASIA Associate	4. We acknowledge that the State Legislature has appropriated funds for the planning and design of the UH West O'ahu Campus on the University Parcel, and recognize that the deadline for construction of the proposed project is through December 31, 2011.
ga:000015001K10032		The following text will be added to Section 6.3 No Action Alternative of the EIS:
cc: The Honorable Linda Lingle	RONORULU OFFICE (041 Bitshop Street) (041 Bitshop Street) (11 Parenty Bartel) (12 Parenty Street) (12 Parenty Parenty Parenty (12 Parenty Parenty Parenty (12 Parenty Parenty Parenty Parenty (12 Parenty Parenty Parenty Parenty Parenty (12 Parenty Parenty Parenty Parenty Parenty Parenty Parenty Parenty (12 Parenty Pare	Under one of the conditions and terms of agreement set forth between the James Campbell Company. LLC, and the State of Hawai's, in regards to the acquisition of fond for the proposed UH West O datu campus the State shall have secured funding, enter contracts and commenced the construction of all of the administration, classroom and related support focilities mecessary to accommodate on enrollment of 2750 students, provided that such
	HILO OFFICE Bit Auput Street Bits Laynon Center Sute 10 Mar Hawward And	commencement and state when we extended to Decomplet 31, 2011, If, it is state Legislature appropriates funds for the planning, and design of the UH West O'ahu campus on the mauka property by December 31, 2006.
	THE FARM TWATE THE FARM THE FA	The State has appropriated funds for the planning and design of the property for the UH West O'ahu, compus, however, if construction of the UH West O'ahu compus is not commenced by December 31, 2011, then the State, at its election shall either:
	14:1848) 242 255 K	 Convex to Campbell Estate a single, agreed upon parcel of land that is located makai of H-1, zoned for agriculture with urban potential, and that hex value equal to the 1994 value of the 200 acres of land conveyed to the State in 1994; or

•

PJANNING + JANDACAPI ARCHITECTURE + ENVIRONMENTAL STUDIGS + ENTITIEMENTS - FERMITTING + GRAPHIC DESIGN

D·R·HORION" 影響 America's Builder	SCHULER DIVISION	August 7, 2006 RECEIVES	Mr. Vincent Shigekuni, Vice President PBR Hawaii ASB Tower - Suite 650 1001 Bishop Street Honolulu, Hawai'i 96813 Dear Mr. Shigekuni,	Subject: University of Hawaii West Oahu (UHWO) Draft Environmental Impact Statement (DEIS)	We appreciate this opportunity to provide our comments on the UHWO DEIS.	D.R. Horton has long been a supporter of Kapolei as Oahu's planned second city and we believe that UHWO will complete the Estate of James Campbelt's long-term vision for	Kapolei, allowing Ewa Plain residents to truiy live, work, play, shop and study within Kapolei, without traveling outside of the area.	As an adjoining property owner and developer, we have been meeting almost monthly for a year with representatives from UHWO, DHHL and HCDA (Kalaeloa) to discuss	regional planning issues (such as the Ewa Development Plan and the Honolulu High Capacity Transit Corridor project), and potentials for joint development of necessary infrastructure (such as water, sewer, drainage, connecting roads) and I am sure I am not	speaking out of turn when I say that all of us who attend those meetings are very supportive of the development of UHWO and are looking forward to its first opening day.	As modest as the initial student population planned to be served in UHWO's first phase is, we believe that UHWO will not only benefit Kapolei, Makakilo, Ewa Beach, Wainahu Wainana Crart and Control Oahu incincents, endows we is used	play a greater role in reducing traffic from those areas to UH Manoa, Kapiolani Community College, Honolulu Community College and Leeward Community College.	I ne (trattic demand management) benefit will not only be regional, but island-wide.	Building unique homes in diverse neighborhoods throughout Hawaii Oahu Maui Hawaii Kauai	828 Fort Street Mall • 4th Floor • Honohlu, Hawaii 96813 • (808) 521-5661 • Fax: (808) 538-1476 www.drhortonhawaii.com
Mr. David W. Rae SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2005 Page 2	 Pay the Campbell Estate the current fair market value of the 200 acres of land conveyed to the State in 1994. 	The following text will be added at the end of Section 6.6 Alternative of Postponing Action P ending Further Study of the EIS:	Under conditions and terms of agreement set forth between James Comphell Compony, LLC, and the State of Hawai's, in regents to the acquisition of land for the proposed UP Wew O'thu compuse: 1) Uff must devignate at least 200 acres of the manka proposed VD use solely for University Purposed by the earlier to occur of December 31, 2005, or the date of any conversance of any portion of the manka property to any third porty, and 2) the State shall have secured funding, entered into contracts and commerced the construction of all of the administration. classroom and related support facilities necessary to accommodate an enrollment of 2.350 students, proded that such commercement date shall be extended to December 31. 2011, if the State Legislance appropriates funds for the planning and design of the UH West O'ahu compus on the	nausa property by December 31, 2009. The State has appropriated funds for the planning and design of the property for the UH West O'ahu campus, however, if construction of the UH West O'ahu campus is not	commenced by December 31, 2011, then the State, at its election shall either:	 Convey to Campbell Estate a single, agreed upon parcel of land that is located maked of H-1, zoned for agriculture with urban potential, and that has value equal to the 1994 value of the 200 acres of land conveyed to the State in 1994; or 	<u>2. Pay the Campbell Estate the current fair market value of the 200 acres of land</u> conveyed to the State in 1994.	Thank you again for your participation in the Environmental Impact Statement process for this project. Your comments will be included in the Final EIS.	If you have any questions regarding this project, please do not hesitate to contact me at \$21-5631. Sincerely,	PBR HAWAII Vuizes R. Suigh-	Vincent R. Shigekuni Vice President	cc: Office of Environmental Quality Control Gene Awakuni/UH West O'ahu	O:VOB1311322.05)EISVFinal EISVFinal Responses to DEIS CommentablL-231. Campbell response 11-28.40c		

.

,

.

.
SR HAWAII sociates. Inc.	December 18, 2006	Mr. Michael T. Jones, Division President	Mr. Michael T. Jones, Division President D.R. Horton – Schuler Division 828 Fort Street Mall, Fourth Floor Honolulu, Hawai'i 96813 SUBJECT: JUNUVEDSTIV OF HAMARY WEET OALTHE GUT WEET		DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) Dear Mr. Jones:	Thank you for your letter dated August 7, 2006 . We have reviewed your letter and offer the following responses to your comments:	1. Thank you for your support of the development of UH West O'ahu.	 We strongly concur that UH West O'ahu will not only benefit Kapolei, Makakilo, 'Ewa Beach, Waipahu, Wai'anae Coast and Central O'ahu university students, but it will eventually play a greater role in reducing traffic from those areas to UH Mánoa, Kapiolani Community College, Honolulu Community College and Leeward Community College. 	This letter will be included in the Final EJS under Section 12.0.	Thank you again for your participation in the Environmental Impact Statement process. If you have any questions regarding this project, please do not hesitate to contact me at 521-5631.	Sincerely,	PBR HAWAII	Vurex K. Onegu-	Vincent R. Shigekuni Vice President	cc Office of Environmental Quality Control Gene Awakuni/UH West O'ahu	0/UOB1311322.0%EJSFinal EiSResponse to DEIS Comments/BL-12 DR Horton response 11-20.doc	AFT ARCHITT GTHRE + ENVTRORMENTAL STUDIES + ENTITIEMENTS - FFRMITTING + GAAFUIC DEVICO	
P. S.		W, FRANK BRANDT, LASLA (baitnur THOMASS, WITTES, ASLA	President R. STAN DUNCAN, ASI.A Executive Vier-President	RUSAILLY, CHTNG, FASIA Executive Vice-President MICENERMERED	Vice Presulent GRANT T.MI.RAKAMI, AICP Principal	TOM SCHNELL, AICP Senior Associate	RATMOND THIGA, ASLA Scriur Associate KEVIN S. NISHIKAWA, ASLA	Alsortativ RIMM ABRAMII VI'I'S, ITI'D'AU Associatic SCOTT ALIKA ABBIGO Assoviato	NCOPT MURARAMI, ASLA Associate		HONOLULU OFFICE 1001 Dickey Street	A 54 Tower, Suite 6.40 Houdda, Hawrit 964 1 185 For 6880 524 504 Lay 6885 524 102 É-mall-Ayadume phylaryatom	HILO OFFICE 101 Ammint Stated	Filler Jagwar Coller, Suile 310 (10), Alaxail 97, 2012 (10), Alaxail 97, 2012 (10), 100, 11, 2012 (10), 100, 100, 100, 100, 100, 100, 100,	WAILUKU OFFICIE E'ST WILL'I A Every, suite 1 Waithin, Massari We'St J.T.	8767 747 (MN934 (D))	ANNING - LANDSC.	
If you have any questions, please do not hesitate to call Bob Bruhl or myself.	Sincerely.	J ~ J	Michael T. Jones Division President	D.R. Horton – Schuler Division cc: Office of Environmental Ouality Control														

Mr. Gene Awakuni August 21, 2006 Page Two	Watershed, and HASEKO has agreed to process the permits required for this outlet to the ocean with City and State agencies. In this regard, in December of last year, the City accepted a Final Environmental Impact Statement describing a proposed surface channel over Oneta Beach Park.	HASEKO believes this effort is for the benefit for all the landowners and developers in the Kaloi Regional Drainage Basin, and solicits your support and input to permitting this outlet to the ocean. Issues regarding equitable participation in park restoration following major storm events needs to be addressed, and upland retention/detention basins to protect offshore water quality will be encountered in applications dealing with State	waters. Resolution of these issues and conditions to approvals to these permits may be important to the future planning of your projects, and it is our intent to provide you with updates of our progress.	We welcome continued open communication with you on UHWO's future plans, especially as it relates to regional drainage and the myriad of opportunities the campus	and surrounding FITVate Development Lands will create for the benefit of all of West Oahn. Again, thank you for the opportunity to comment on your Draft Environmental Impact Statement.	Sincerely, HASEKO(Ewa), Inc.	Nelson W. G. Dee Executive Vice President cc: Office of Environmental Quality Control PHR Hawaii (Vincent Shinekuni)		VDowntownNWGL Docs/Client Files(Occan PointetDrainage/08 18 06 Haseko - Responso to UHWO DEIS nl.doo
RECEIVED Aug 23 2006 PBR HAWAH		al Impact Statement		Environmental UHWO) dated July) campus. As an w the campus as ffic woes of the area, education and lifestyle	nakai of the proposed ed, we are particularly m all mauka projects	ons of the Kaloi unoff caused by the amount of i in Appendix I, xribes detention basins <i>mection to the Pacific</i>	e times over the last let at Oneula Beach ar the Kaloi Gulch	•5005
HASEKO (Ewa), Inc.	August 21, 2006	Mr. Gene Awakuni University of Hawaii West Oahu 96-129 Ala Ike Pearl City, Hawaii 96782 Re: <u>University of Hawaii West Oahu Draft Environmenta</u>	Dear Mr. Awakuni:	Thank you for the opportunity to review and comment on the Draft Impact Statement (DEIS) for the University of Hawaii West Oahu (I 2006.	HASEKO fully support efforts to construct the long-awaited UHW(active member of the West Oatu community of Ewa Beach, we vier playing an important role in providing much-needed relief to the trai new job opportunities in the region, and increased access to higher e learning.	As our Ocean Pointe Project (formerly known as Ewa Marina) lies r UHWO project site and at the terminus of the Kaloi Gulch Watershe interested in the impacts from potential increased surface runoff fror that are part of the ongoing urbanization of the Watershed.	We are pleased your agency recognizes the current drainage limitati Gulch Watershed, and are planning to retain any increase in storm r UHWO development within the UHWO property, and maintain the downstream storm water flow at predevelopment levels as described <i>Infrastructure Study</i> . by Engineering Concepts, Inc. The DEIS desc to which runoff will be conveyed " <i>until a downstream drainage con</i> <i>Ocean its scientishied at Oreada Beach Park.</i> "	Plans for this connection to the Pacific Ocean have changed multipl 16 years due to factors beyond our control. Currently, an ocean outl Park appears to be the City's preferred regional drainage solution fo	۹۱-1001 Kaimalie Sheet, Suile 205 ، قدم 8each, Hawaii 96706 تافاصاحمه (2018) نام 205 ، قدم 1939 مرم، 2757 ، 1

7

. ,

Fox (808) 689-5757 N 91+1001 Kaimalie Sheet. Telephone (80 .

\leq
\geq
- Si
\triangleleft
~≦i
┝┷┷┥╘╴
2
<u>5</u>

December 18, 2006

W. FRANK IRANDU, FASLA

THOMAS S. WITTEN, ASLA Preddent

R. STAN DUNCAN, ASLA Freetive Vice-President

WISSELLY LCHUNG, FASLA Executive Vice-President VINCENT'SHIGIKUN) ter President GRAN FT: MURAKAMI, AICP Principal

RAYMOND T. HIGA, ASI A hentine Associate KUVIN K. NISHIKAWA, ASI A 1350Crothe

KIMD AUKAND YUEN, UT D'AP Accordate

SCOTT MURAKAMI, ASLA bsociate

1storiate

HONOLULU OFFICE 1001 Hadop Vires 2011 Tower Sinfe 670 Frondulu, Hawart 9664 § 1184 1-47 (1986) 521 1019 Frankle systallyhillyg (behavaai com

Mr. Nelson W.G. Lee, Executive Vice President 91-1001 Kaimalie Street, Suite 205 Ewa Beach, Hawai'i 96706 HASEKO (Ewa), Inc.

UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) SUBJECT:

FOM SCHNILL, AICP Serior Associate

SCOTT ALIKA ABRIGO

HILO OFFICE 001 Augund Street 11860 Jageon Caener, buite J10 11860 Jageon Caener, buite J10 1186, Jahawalf 962, 9430 7a b (2018) 964-9389 Face (R28) 961-9389

WAILUKU OFFICE 1287 Wild Pallogip, Soite 1 Walloku, Hawail 96293 1251 164 (2003) 242-2378

Dear Mr. Lee:

Thank you for your letter dated August 21, 2006. We have reviewed your letter and offer the following responses to your comments:

- O'ahu campus. We strongly concur that the UH West O'ahu campus will play an Thank you very much for the expression of support of the long-awaited UH West important role in providing much-needed relief to the traffic woes of the area, new job opportunities in the region, and increased access to higher education and lifestyle learning. ÷
- limitations, and the UH West O'ahu is planning to retain any increase in storm runoff caused by the UH West O'ahu development within the UH West O'ahu property, and maintain the amount of downstream storm water flow at predevelopment levels until a downstream drainage connection to the Pacific We acknowledge that the Kalo'i Gulch Watershed experiences current drainage Ocean is established at Oneula Beach Park. d'
- We support your efforts to process the permits required for the drainage outlet at Oneula Beach Park. We recognize that issues need to be addressed, regarding equitable participation in park restoration following major storm events needs to be addressed, and upland retention/detention basins to protect offshore water quality will be encountered in applications dealing with State waters. ы.

of Section 4.10.4 Drainage Facilities, Proposed The fourth paragraph of Section 4.10.4 Dru Improvements of the EIS will be revised as follows:

Flood control detention areas will be required until a downstream drainage connection to the Pacific Ocean is established at Oneula Beach Park. According to comments received by Hareko (Evad). Inc. (HASEKO), an ocean outlet at Oneula Beach Park appears to be the City's preferred regional drainage solution for the Reach Park appears to be the City's preferred regional drainage solution for the Reach Park appears to be the City's preferred to process the permits required for this outlet to the ocean with City and sate agencies. In this regard, a Final Environmental Impact Statement describing, a proposed surface channel over Oneula Beach Park has been accepted by the City (December 2005).

Mr. Nelson W. G. Lee SUBJECT: UNIVERSITY OF HAWAI'I WEST O'AHU (UH WEST O'AHU) DRAFT ENVIRONMENTAL IMPACT STATEMENT (Draft EIS) December 18, 2006 Page 2

Thank you again for your participation in the Environmental Impact Statement process for this project. Your comments will be included in the Final BIS.

If you have any questions regarding this project, please do not hesitate to contact me at 521-5631.

Sincerely,

Mich & R. Bu PBR HAWAII

Vincent R. Shigekuni Vice President Office of Environmental Quality Control Gene Awakuni/UH West O'ahu 넝

0:VOB13V1322.05/EISVTinal EISVResponse to DEIS Comments/BL-20 HASEKO response 11-20.400

PLANNING + LAMINGAPF ARCHITFGTIRF + FNVIRGNMENTAL STUDIES + INTITLEMENTS - FERMITTING - GRAFIIIC DESIGN